DATA-DRIVEN SCIENCE AND ENGINEERING

Machine Learning, Dynamical Systems, and Control

Steven L. Brunton · J. Nathan Kutz

Data-Driven Science and Engineering

Data-driven discovery is revolutionizing the modeling, prediction, and control of complexsystems. This textbook brings together machine tearning, engineering mathematics, and mathematical physics to integrate modeling and control of dynamical systems with modern methods is data sizence. It highlights many of the recent advance is scientific computing that enable data-driven methods to be applied to a diverse range of complex systems such as intufficience. But including methods (see final strengther systems such as intufficience, but including scientification).

Aimed at advanced undergraduate and beginning graduate students in the engineering and physical sciences, the text presents a range of topics and methods from introductory to state of the art.

Steret L foruthe is Associate Professor of Mechanica Engineering at the University of Wonlington. He is also Adjunct Associate Professor of Applied Mathematics and a Dan-Science Fellow at the elscience Institute. His research applies data science and machine learning for dynamical systems and control to find dynamics, holocomotion, optics, energy systems, and numericativity. He is an analos of two textbooks, received the Ampli-Wonlinetty College of Environment textbooks and a professional and the structure of the structure o

J. Baban Mirgi is the Robert Bolles and Yasako Endo Porlessor of Applied Mathematics as the University of Wahalington, and servers of adoptime training unit 2015. He is also Adjunct Porlessor of Electrical Engineering and Physics and a Seaior Data-Science Fellow at the Ensearch interests are incomplex systems and data analysis where machine income the systems and data analysis where the machine income the systems and data analysis where the state of the

Data-Driven Science and Engineering

Machine Learning, Dynamical Systems, and Control

STEVEN L. BRUNTON

University of Washington

J. NATHAN KUTZ University of Washington



CAMBRIDGE UNIVERSITY PRESS

University Printing House, Cambridge CB2 8BS, United Kingdom

One Liberty Plaza, 20th Floor, New York, NY 10006, USA

477 Williamstown Read, Port Melbourne, VIC 3207, Australia

314-321. 3rd Floor. Plot 3. Solendor Forum. Jasola District Centre, New Delhi - 110025. India

79 Anson Road, #06-04/06, Singapore 079906

Cambridge University Press is part of the University of Cambridge.

It furthers the University's mission by disseminating knowledge in the pursuit of education, learning, and research at the highest international levels of excellence.

www.cambridge.org Information on this title: www.cambridge.org/978110842209 DOI: 10.1017/9781108380990

O Steven L. Brunton and J. Nathan Kutz 2019

This publication is in copyright. Subject to statutory exception and to the provisions of subvant cellscrive licensing agreements, no reproduction of any part may take place without the written permission of Cambridge University Procs.

First published 2019

Printed and bound in Great Britain by Clays Ltd, Elcograf S.p.A.

A catalogue record for this publication is available from the British Library.

Library of Coursess Cataloning-in-Publication Data

Name: Tensus, Stovet, L. (Stovet, Luc, 1984. - andre ; Enzi, Jon Nahau, andre The Dara -drugs core: and enginesting: annualish learning, dynamical opnome, The Dara -drugs core and enginesting: annualish learning, dynamical opnome, Description: Catholicy, Catholicy University of Washington. Description: Catholicy, Catholicy University (2020) Annualistic, 180, paper) Mathematical analysis. Mathematical analysis. Mathematical analysis.

ISBN 978-1-108-42209-3 Hardback

Cambridge University Pross has no responsibility for the persistence or accuracy of URLs for external or third-party Internet Web vites referred to in this publication and does not guarantee that any content on such Web sites is, or will remain, accurate or assocritate.

Contents

	Pres	lace	page ix
	Con	unon Optimization Techniques, Equations, Symbols, and Acronyms	xiii
Part I	Dime	nsionality Reduction and Transforms	1
1	Sing	ular Value Decomposition (SVD)	3
	1.1	Overview	3
	1.2	Matrix Approximation	7
	1.3	Mathematical Properties and Manipulations	10
	1.4	Pseudo-Inverse, Least-Squares, and Regression	15
	1.5	Principal Component Analysis (PCA)	21
	1.6	Eigenfaces Example	25
	1.7	Truncation and Alignment	30
	1.8	Randomized Singular Value Decomposition	37
	1.9	Tensor Decompositions and N-Way Data Arrays	41
2	Fou	rier and Wavelet Transforms	47
	2.1	Fourier Series and Fourier Transforms	47
	2.2	Discrete Fourier Transform (DFT) and Fast Fourier Transform (FFT)	56
	2.3	Transforming Partial Differential Equations	63
	2.4	Gabor Transform and the Spectrogram	69
	2.5	Wavelets and Multi-Resolution Analysis	75
	2.6	2D Transforms and Image Processing	77
3	Spa	rsity and Compressed Sensing	84
	3.1	Sparsity and Compression	84
	3.2	Compressed Sensing	88
	3.3	Compressed Sensing Examples	92
	3.4	The Geometry of Compression	95
	3.5	Sparse Regression	98
	3.6	Sparse Representation	103
	3.7	Robust Principal Component Analysis (RPCA)	107
	3.8	Sparse Sensor Placement	110

Part II Machine Learning and Data Analysis			115
4	Reg	ression and Model Selection	117
	4.1	Classic Curve Fitting	118
	4.2	Nonlinear Regression and Gradient Descent	123
	4.3	Regression and Ax = b: Over- and Under-Determined Systems	130
	4.4	Optimization as the Cornerstone of Regression	136
	4.5	The Pareto Front and Lex Parsimoniae	140
	4.6	Model Selection: Cross-Validation	143
	4.7	Model Selection: Information Criteria	148
5	Clus	tering and Classification	154
	5.1	Feature Selection and Data Mining	155
	5.2	Supervised versus Unsupervised Learning	160
	5.3	Unsupervised Learning: k-means Clustering	164
	5.4	Unsupervised Hierarchical Clustering: Dendrogram	168
	5.5	Mixture Models and the Expectation-Maximization Algorithm	172
	5.6	Supervised Learning and Linear Discriminants	176
	5.7	Support Vector Machines (SVM)	180
	5.8	Classification Trees and Random Forest	185
	5.9	Top 10 Algorithms in Data Mining 2008	190
6	Neu	ral Networks and Deep Learning	195
	6.1	Neural Networks: 1-Layer Networks	196
	6.2	Multi-Layer Networks and Activation Functions	199
	6.3	The Backpropagation Algorithm	204
	6.4	The Stochastic Gradient Descent Algorithm	209
	6.5	Deep Convolutional Neural Networks	212
	6.6	Neural Networks for Dynamical Systems	216
	6.7	The Diversity of Neural Networks	220
Part II	Dyn	amics and Control	227
7	Data	-Driven Dynamical Systems	229
	7.1	Overview, Motivations, and Challenges	230
	7.2	Dynamic Mode Decomposition (DMD)	235
	7.3	Sparse Identification of Nonlinear Dynamics (SINDy)	247
	7.4	Koopman Operator Theory	257
	7.5	Data-Driven Koopman Analysis	266
8	Line	ar Control Theory	276
	8.1	Closed-Loop Feedback Control	277
	8.2	Linear Time-Invariant Systems	281
	8.3	Controllability and Observability	287
	8.4	Optimal Full-State Control: Linear Quadratic Regulator (LQR)	292

- 14

	8.5 Optimal Full-State Estimation: The Kalman Filter	296
	8.6 Optimal Sensor-Based Control: Linear Quadratic Gaussian (LC	(G) 299
	8.7 Case Study: Inverted Pendulum on a Cart	300
	8.8 Robust Control and Frequency Domain Techniques	308
9	Balanced Models for Control	321
	9.1 Model Reduction and System Identification	321
	9.2 Balanced Model Reduction	322
	9.3 System identification	336
10	Data-Driven Control	345
	10.1 Nonlinear System Identification for Control	346
	10.2 Machine Learning Control	352
	10.3 Adaptive Extremum-Seeking Control	362
Part IV	/ Reduced Order Models	373
11	Reduced Order Models (RDMs)	375
	11.1 POD for Partial Differential Equations	375
	11.2 Optimal Basis Elements: The POD Expansion	381
	11.3 POD and Soliton Dynamics	387
	11.4 Continuous Formulation of POD	391
	11.5 POD with Symmetries: Rotations and Translations	396
12	Interpolation for Parametric ROMs	403
	12.1 Gappy POD	403
	12.2 Error and Convergence of Gappy POD	409
	12.3 Gappy Measurements: Minimize Condition Number	413
	12.4 Gappy Measurements: Maximal Variance	418
	12.5 POD and the Discrete Empirical Interpolation Method (DEIM)	423
	12.6 DEIM Algorithm Implementation	426
	12.7 Machine Learning ROMs	429
	Glossary	436
	Bibliography	443
	Index	471

Preface

This bok in shoch the growing interaction of data-direct methods, applied optimization, and the calculated Bok of generizing mathematical application. We have undergoted and have been applied on the strength of the strengt

Data-drive descorey is correctly revolutionizing how se endod, predic, and commo project systems. These maps provide endormal end

Drving modern data sciences is the availability of var and increasing quantities of data, emaled by premarkabile invorvation in low consense, orders-of angaindusis increases in comparational poper, and virtually unilimited data storage and transfer capabilities. Such vare quantities of data ardiording engineers and scientifica serves all disciplicities were supportainties for data-driven discovery, which has been referred to as the fourth paradigm of scientific discovery (2013). This forth paradigm is the natural collimation of the three paradigms: empirical experimentation, analytical derivation, and comparational investgation. The intergration of these techniques provides a transformative framework of the scientific discovery. data-dirow discovery efforts. This process of scientific discovery is not new, and indeer miniscs the efforts of leading fuguess of the scientific revolution: Johannes Kepfer (1571– 1630) and Sir baar. Newton (1642–1727). Each physel a critical role in developing the theoretical undergomings of crientian theoretisms, based on a combination of empirical data driven and milytical approaches. Data science is not replacing millementical physics of a transissory that a servelution.

Data science tred is not seen, honigo theor proposed new chan 30 years, any 10 Marty horizon beneficient of the science of a sciencific broaded at homatify the data (here you horizontee) of the science of the the science of the sci

Themes of This Book

There are a number of key themes that have emerged throughout this book. First, many complex systems exhibit dominant low-dimensional patterns in the data, despite the rapidly increasing resolution of measurements and computations. This underlying structure enables efficient sensing, and compact representations for modeling and control. Pattern extraction is related to the second theme of finding coordinate transforms that simplify the system. Indeed, the rich history of mathematical physics is centered around coordinate transformations (e.g., spectral decompositions, the Fourier transform, generalized functions, etc.), although these techniques have largely been limited to simple idealized geometries and linear dynamics. The ability to derive data-driven transformations opens up opportunities to generalize these techniques to new research problems with more complex geometries and boundary conditions. We also take the perspective of dynamical systems and control throughout the book, applying data-driven techniques to model and control systems that evolve in time. Perhaps the most pervasive theme is that of data-driven applied optimization, as nearly every topic discussed is related to optimization (e.g., finding optimal lowdimensional patterns, optimal sensor placement, machine learning optimization, optimal control, etc.). Even more fundamentally, most data is organized into arrays for analysis, where the extensive development of numerical linear algebra tools from the early 1960s onward provides many of the foundational mathematical underpinnings for matrix decompositions and solution strategies used throughout this text.

Acknowledgments

We are indebted to many wonderful students, collaborators, and colleagues for valuable feedback, suggestions, and support. We are especially grateful to Joshua Proctor, who was instrumental in the origination of this book and who helped guide much of the framing and organization. We have also henclifted from extensive interactions and conversations with Bing Branton, Igor Mezić, Bernd Noack, and Sam Taiza. This work would also not be possible without our many great colleagues and collaborators, with whom we have worked and whose research is fearmed throughout this hook.

We would also like to thank our publisher Lauren Cowles at Cambridge University Press for being a reliable supporter throughout this process.

Online Material

We have designed this book to make extensive use of online supplementary material, including codes, data, videos, homeworks, and suggested course syllabi. All of this material can be found at the following website:

databookuw.com

In addition to come resources, all of the code and data used in the book are available. The codes colline are more extensive than those persented in the book, including code used to generate publication quality figures. Data visualization was ranked as the top used data-science methods in the Kaggle 2017 The State of Data Science and Machine Learning study, and so we highly encourage readers to download the online codes and make full use of these plotting commands.

We have also recorded and posted video lectures on YouTube for most of the topics in this book. We include supplementary videos for students to fill in again in their background on scientific computing and foundational applied mathematics. We have designed this text both to be a reference as well as the material for several courses at various bevies of student preparation. Most chapters are also modular, and may be converted into stand-alone *hoot* composition of the student of material science.

How to Use This Book

Our intereded audience includes beginning graduate sudents, or advanced undergraduates, in englancering and science. As such the muchine learning methods are introduced at a beginning level, whereas we assume students know how to model physical systems with differential equations and isamize then we in shorts such as added. The developing the student student student students are students about the student student science roblems. All homorehysical homorehysical as a student student science roblems. Morehensity is a student student student student student science roblems. Morehensity is a student student student student student science roblems. Morehensity is a student student student student student science roblems. Morehensity is the boot student student student student science roblems. Morehensity is the boot student student student student student science roblems. Student student student student student student student student science roblems. Student student student student student student student student science roblems. Student science roblems. Student science roblems. Student st statistics students who often have limited knowledge of dynamical systems and control. Various courses can be designed from this material, and several example syllabi may be found on the book website; this includes homework, data sets, and code.

First and foremosit, we want this book to be fun, inspiring, cyc-opening, and empowening for young scientificat and engineers. We have attempted to make everything as simple as possible, while still providing the depth and breadth required to be sarchil are insearch. Muny of the chapter points in this text could be entire books in their count of the analyse of them are. However, we also warmed to be as comprehensive as may be transmitly expected for methods, and charges the work of white model data a science?

Common Optimization Techniques, Equations, Symbols, and Acronyms

Most Common Optimization Strategies

Least-Squares (discussed in Chapters 1 and 4) minimizes the sum of the squares of the residuals between a given fitting model and data. Laraer least-quares, where the residuals are linear in the unknown, has a closed form solution which can be computed by taking the derivative of the residual with negative to each unknown and setting it to zero. It is commonly used in the engineering and applied sciences for fitting polynomial functiona, the nonlinear least-news with in fitness the sciences rest of the readout the model and the science of the science of the fitting polynomial functiona. The nonlinear least-news with in fitness the sciences rest of the readout of the model of the sciences rest of the readout of the science of the readout of the science of the readout of the science of the readout of the readout of the science of the readout of the science of the readout of the science of the readout of the re

Gradient Devent (discussed in Chapters 4 and 6) is the industry leading, convect optimization method for high-dimensional systems. In minimizer scriedlor that by computing the gradient of a given fitting function. The iterative procedure updates the colution by moving downlift in the residual ages. The Neuran-Laphone method is a suc-dimensional version of pradient docent. Since it is often applied in high-dimensional settings, it is proze to full only local minime. There is a substantiation of the state of the state of the state of the descent and the backpropagatos algorithm which makes the optimization amenable to commoding the residue is (seff.

Alternating Descent Method (ADM) (discussed in Chapter 4) avoids computations of the aprilent by optimizing in one unknown at a time. Thus all unknowns are bed constant while a line search (non-convex optimization) can be performed in a single variable. This begins and a single variable is the single search of the unknowns is updated. The iterative procedure continues through all unknowns and the iteration procedure is repeated unit al desired break of accuracy is abselved.

Augmented Lagrange Method (ALM) (discussed in Chapters 3 and 8) is a class of algorithms for solving constrained optimization problems. They are similar to penalty methods in that they reglace a constrained optimization problem by a series of unconstrained problems and add a penalty term to the objective which helps enforce the distribuconstraint. ALM adds another term designed to mimic a Lagrange multiplier. The augmented Lagrangian is not the same as the method of Lagrange multiplier.

Linear Program and Simplex Method are the workhore algorithms for convex optmization. A linear program has an objective function which is linear in the unknown and the constraints consist of linear inequalities and equalities. By computing its feasible region, which is a convex optopyte, the linear programming algorithm finds a point in the polyhelmow where this function has the smallest for largesty value if such a point exists. The simplex method is a specific herarite testings for this are programs which align to take a given basic feasible solution to another basic feasible solution for which the objective function is smaller, then producing an iterative procedure of optimizing.

Most Common Equations and Symbols

Linear Algebra

Linear System of Equations

$$Ax = b.$$
 (0.1)

The matrix $\mathbf{A} \in \mathbb{R}^{p \times n}$ and vector $\mathbf{b} \in \mathbb{R}^{p}$ are generally known, and the vector $\mathbf{x} \in \mathbb{R}^{n}$ is unknown.

Eigenvalue Equation

$$AT = TA$$
. (0.2)

The columns ξ_k of the matrix **T** are the eigenvectors of $\mathbf{A} \in \mathbb{C}^{n \times n}$ corresponding to the eigenvalue $\lambda_k : \mathbf{A} \xi_k = \lambda_k \xi_k$. The matrix **A** is a diagonal matrix containing these eigenvalues, in the simple case with *n* distinct eigenvalues.

Change of Coordinates

$$x = \Psi a.$$
 (0.3)

The vector $\mathbf{x} \in \mathbb{R}^n$ may be written as $\mathbf{a} \in \mathbb{R}^n$ in the coordinate system given by the columns of $\Psi \in \mathbb{R}^{n \times n}$.

Measurement Equation

$$y = Cx.$$
 (0.4)

The vector $\mathbf{y} \in \mathbb{R}^p$ is a measurement of the state $\mathbf{x} \in \mathbb{R}^n$ by the measurement matrix $\mathbf{C} \in \mathbb{R}^{p \times n}$.

Singular Value Decomposition

$$\mathbf{X} = \mathbf{U} \mathbf{\Sigma} \mathbf{V}^* \approx \mathbf{U} \mathbf{\Sigma} \mathbf{V}^*$$
. (0.5)

The matrix $\mathbf{X} \in \mathbb{C}^{n\times nn}$ may be decomposed into the product of three matrices $\mathbf{U} \in \mathbb{C}^{n\times nn}$, $\mathbf{\Sigma} \in \mathbb{C}^{n\times nn}$, and $\mathbf{V} \in \mathbb{C}^{n\times nn}$. The matrices \mathbf{U} and \mathbf{V} are unitary, so that $\mathbf{U}^n = \mathbf{U}^* \mathbf{U} = \mathbf{L}_{n\times n}$, and $\mathbf{V}^{n'} = \mathbf{V}^* \mathbf{V} = \mathbf{I}_{n\times nn}$, where ' denotes complex conjugate transpose. The columns of \mathbf{U} (resp. \mathbf{V}) are orthogonal, called left (resp. right) singular vectors. The matrix $\mathbf{\Sigma}$ contains decreasing, nonnegative diagonal metrics called singular values.

Often, \tilde{X} is approximated with a low-rank matrix $\tilde{X} = \tilde{U}\tilde{\Sigma}\tilde{V}^*$, where \tilde{U} and \tilde{V} contain the first $r \ll n$ columns of U and V, respectively, and $\tilde{\Sigma}$ contains the first $r \times r$ block of Σ . The matrix \tilde{U} is often denoted Ψ in the context of spatial modes, reduced order models, and sensor placement.

Regression and Optimization

Overdetermined and Underdetermined Optimization for Linear Systems

$$\operatorname{argmin}(\|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2 + \lambda g(\mathbf{x}))$$
 or (0.6a)

$$\underset{\mathbf{x}}{\operatorname{argmin}} g(\mathbf{x}) \quad \operatorname{subject to} \|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2 \le \epsilon, \quad (0.6b)$$

Here $g(\mathbf{x})$ is a regression penalty (with penalty parameter λ for overdetermined systems). For over- and underdetermined linear systems of equations, which result in either no solutions or an infinite number of solutions of $\mathbf{A} = \mathbf{b}$, a choice of constraint or penalty, which is also known as *resultarization*, must be made in order to moduce a solution.

Overdetermined and Underdetermined Optimization for Nonlinear Systems

$$\operatorname{argmin}(f(\mathbf{A}, \mathbf{x}, \mathbf{b}) + \lambda g(\mathbf{x}))$$
 or (0.7a)

$$\operatorname{argmin}_{\mathbf{x}} g(\mathbf{x}) \text{ subject to } f(\mathbf{A}, \mathbf{x}, \mathbf{b}) \leq \epsilon \qquad (0.7b)$$

This generalizes the linear system to a nonlinear system $f(\cdot)$ with regularization $g(\cdot)$. These over- and underdetermined systems are often solved using gradient descent algorithms.

Compositional Optimization for Neural Networks

$$\operatorname{argmin}_{\mathbf{A}_{j}} \left(f_{M}(\mathbf{A}_{M}, \cdots, f_{2}(\mathbf{A}_{2}, (f_{1}(\mathbf{A}_{1}, \mathbf{x})) \cdots) + \lambda g(\mathbf{A}_{j}) \right) \qquad (0.8)$$

Each A_k denotes the weights connecting the neural network from the *k*th to (*k* + 1)th layer. It is typically a massively underdetermined system which is regularized by *g*(**A**₁). Composition and regularization are critical for generating expressive representations of the data as well as preventing overfitting.

Dynamical Systems and Reduced Order Models

Nonlinear Ordinary Differential Equation (Dynamical System)

$$\frac{d}{dt}\mathbf{x}(t) = \mathbf{f}(\mathbf{x}(t), t; \boldsymbol{\beta}). \quad (0.9)$$

The vector $\mathbf{x}(t) \in \mathbb{R}^n$ is the state of the system evolving in time t, $\boldsymbol{\beta}$ are parameters, and \mathbf{f} is the vector field. Generally, \mathbf{f} is Lipschitz continuous to guarantee existence and uniqueness of solutions.

Linear Input-Output System

$$\frac{d}{dx} \mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} \qquad (0.10a)$$

$$v = Cx + Du$$
. (0.10b)

The state of the system is $\mathbf{x} \in \mathbb{R}^n$, the inputs (actuators) are $\mathbf{u} \in \mathbb{R}^n$, and the outputs (sensors) are $\mathbf{y} \in \mathbb{R}^n$. The matrices A, \mathbf{B} , \mathbf{C} , \mathbf{D} define the dynamics, the effect of actuation, the sensing structure, and the effect of actuation feed-through, respectively. Nonlinear Map (Discrete-Time Dynamical System)

$$\mathbf{x}_{k+1} = \mathbf{F}(\mathbf{x}_k).$$
 (0.11)

The state of the system at the kth iteration is $\mathbf{x}_k \in \mathbb{R}^n$, and \mathbf{F} is a possibly nonlinear mapping. Often, this map defines an iteration forward in time, so that $\mathbf{x}_k = \mathbf{x}(k\Delta r)$; in this case the flow map is denoted $F_{\Delta r}$.

Koopman Operator Equation (Discrete-Time)

$$K_{rg} = g \circ \mathbf{F}, \implies K_{rg} = \lambda \varphi.$$
 (0.12)

The linear Koopman operator K_{τ} advances measurement functions of the state $g(\mathbf{x})$ with the flow F_{τ} . Eigenvalues and eigenvectors of K_{τ} are λ and $\psi(\mathbf{x})$, respectively. The operator K_{τ} operates on a Hilbert space of measurements.

Nonlinear Partial Differential Equation

$$u_t = N(u, u_x, u_{xx}, \dots, x, t; \beta).$$
 (0.13)

The state of the PDE is u, the nonlinear evolution operator is N, subscripts denote partial differentiation, and x and r are the spatial and temporal variables, respectively. The PDE is parameterized by values in β . The state u of the PDE may be a continuous function $u(x, \tau)$, or it may be discretized at several spatial locations, $\mathbf{u}(t) = [u(x_1, t) \quad u(x_2, \tau) \quad \cdots \quad u(x_{n-1})^T \in \mathbb{R}^n$.

Galerkin Expansion

The continuous Galerkin expansion is:

$$u(x, t) \approx \sum_{k=1}^{r} a_k(t)\psi_k(x).$$
 (0.14)

The functions $a_k(t)$ are temporal coefficients that capture the time dynamics, and $\psi_k(x)$ are spatial modes. For a high-dimensional discretized state, the Galerkin expansion becomes: $u(t) \approx \sum_{k=1}^{k-1} a_k(t) \psi_k$. The spatial modes $\psi_k \in \mathbb{R}^n$ may be the columns of $\Psi = \tilde{U}$.

Complete Symbols

Dimensions

- K Number of nonzero entries in a K-sparse vector s
- m Number of data snapshots (i.e., columns of X)
- *n* Dimension of the state, $\mathbf{x} \in \mathbb{R}^n$
- p Dimension of the measurement or output variable, $\mathbf{y} \in \mathbb{R}^{p}$
- q Dimension of the input variable, $\mathbf{u} \in \mathbb{R}^{4}$
- r Rank of truncated SVD, or other low-rank approximation

Scalars

- s Frequency in Laplace domain
- 1 Time
- δ learning rate in gradient descent
- Δr Time step
 - x Spatial variable
- Δx Spatial step
- σ Singular value
- λ Eigenvalue
- λ Sparsity parameter for sparse optimization (Section 7.3)
- λ Lagrange multiplier (Sections. 3.7, 8.4, and 11.4)
- r Threshold

Vectors

- a Vector of mode amplitudes of x in basis Ψ, a ∈ R⁴
- b Vector of measurements in linear system Ax = b
- b Vector of DMD mode amplitudes (Section 7.2)
- O Vector containing potential function for PDE-FIND
- r Residual error vector
- s Sparse vector, s ∈ ℝⁿ
- u Control variable (Chapters 8, 9, and 10)
- u PDE state vector (Chapters 11 and 12)
- w Exogenous inputs
- wd Disturbances to system
- wa Measurement noise
- w, Reference to track
- x State of a system, $x \in \mathbb{R}^n$
- x2 Snapshot of data at time r2
- \mathbf{x}_i Data sample $j \in \mathbb{Z} := \{1, 2, \dots, m\}$ (Chapters 5 and 6)
- \tilde{x} Reduced state, $\tilde{x} \in \mathbb{R}^{\prime}$, so that $x \approx \tilde{U}\tilde{x}$
- x Estimated state of a system
- y Vector of measurements, $\mathbf{y} \in \mathbb{R}^{p}$
- y_j Data label $j \in Z := \{1, 2, \dots, m\}$ (Chapters 5 and 6)
 - y Estimated output measurement
 - z Transformed state, x = Tz (Chapters 8 and 9)
 - 6 Error vector

xnli

Vectors, continued

- β Bifurcation parameters
- Eigenvector of Koopman operator (Sections 7.4 and 7.5)
- ♦ DMD mode
- Y Vector of PDE measurements for PDE-FIND

Matrices

- A Matrix for system of equations or dynamics
- Reduced dynamics on r-dimensional POD subspace
- Ax Matrix representation of linear dynamics on the state x
- Ay Matrix representation of linear dynamics on the observables y
- (A, B, C, B) Matrices for continuous-time state-space system
- (Ad, Bd, Cd, Bd) Matrices for discrete-time state-space system
 - $(\hat{\mathbf{A}}, \hat{\mathbf{B}}, \hat{\mathbf{C}}, \hat{\mathbf{B}})$ Matrices for state-space system in new coordinates $\mathbf{z} = \mathbf{T}^{-1}\mathbf{x}$
 - (Ã, B, Č, B) Matrices for reduced state-space system with rank r
 - B Actuation input matrix
 - C Linear measurement matrix from state to measurements
 - C Controllability matrix
 - F Discrete Fourier transform
 - G Matrix representation of linear dynamics on the states and inputs [x^Tu^T]^T
 - H Hankel matrix
 - H' Time-shifted Hankel matrix
 - I Identity matrix
 - K Matrix form of Koopman operator (Chapter 7)
 - K Closed-loop control gain (Chapter 8)
 - K / Kalman filter estimator gain
 - Kr LQR control gain
 - L Low-rank portion of matrix X (Chapter 3)
 - O Observability matrix
 - P Unitary matrix that acts on columns of X
 - Q Weight matrix for state penalty in LQR (Sec. 8.4)
 - Q Orthogonal matrix from QR factorization
 - R Weight matrix for actuation penalty in LQR (Sec. 8.4)
 - R Upper triangular matrix from QR factorization
 - S Sparse portion of matrix X (Chapter 3)
 - T Matrix of eigenvectors (Chapter 8)
 - T Change of coordinates (Chapters 8 and 9)
 - U Left singular vectors of X, $U \in \mathbb{R}^{n \times n}$
 - \hat{U} Left singular vectors of economy SVD of $X, U \in \mathbb{R}^{n \times m}$
 - \tilde{U} Left singular vectors (POD modes) of truncated SVD of $X, U \in \mathbb{R}^{n \times r}$
 - V Right singular vectors of $X, V \in \mathbb{R}^{m \times m}$
 - \tilde{V} Right singular vectors of truncated SVD of $X, V \in \mathbb{R}^{m \times d}$

Matrices, continued

- Σ Matrix of singular values of X. Σ $\in \mathbb{R}^{n \times m}$
- $\hat{\Sigma}$ Matrix of singular values of economy SVD of $X, \Sigma \in \mathbb{R}^{m \times m}$
- $\hat{\Sigma}$ Matrix of singular values of truncated SVD of $X, \Sigma \in \mathbb{R}^{r \times r}$
- W Eigenvectors of A
- Wc Controllability Gramian
- W_a Observability Gramian
- X Data matrix, $X \in \mathbb{R}^{n \times m}$
- \mathbf{X}' Time-shifted data matrix, $\mathbf{X}' \in \mathbb{R}^{m \times m}$
- Y Projection of X matrix onto orthogonal basis in randomized SVD (Sec. 1.8)
- Y Data matrix of observables, Y = g(X), $Y \in \mathbb{R}^{p \times m}$ (Chapter 7)
- \mathbf{Y}' Shifted data matrix of observables, $\mathbf{Y}' = \mathbf{g}(\mathbf{X}')$, $\mathbf{Y}' \in \mathbb{R}^{p \times m}$ (Chapter 7)
- Z Sketch matrix for randomized SVD, $Z \in \mathbb{R}^{n \times r}$ (Sec. 1.8)
- Θ Measurement matrix times sparsifying basis, Θ = CΨ (Chapter 3)
- Θ Matrix of candidate functions for SINDy (Sec. 7.3)
- Γ Matrix of derivatives of candidate functions for SINDy (Sec. 7.3)
- E Matrix of coefficients of candidate functions for SINDy (Sec. 7.3)
- E Matrix of nonlinear snapshots for DEIM (Sec. 12.5)
- A Diagonal matrix of eigenvalues
- Υ Input snapshot matrix, $Υ \in \mathbb{R}^{q \times m}$
- Φ Matrix of DMD modes, $Φ ≜ X'VΣ^{-1}W$
- Orthonormal basis (e.g., Fourier or POD modes)

Tensors

(A, B, M) N-way array tensors of size $I_1 \times I_2 \times \cdots \times I_N$

Norms

- || ℓ₀ pseudo-norm of a vector x the number of nonzero elements in x
- $\|\cdot\|_1 = \ell_1$ norm of a vector **x** given by $\|\mathbf{x}\|_1 = \sum_{i=1}^n |x_i|$
- $\|\cdot\|_2 = \ell_2$ norm of a vector **x** given by $\|\mathbf{x}\|_2 = \sqrt{\sum_{i=1}^{n} (x_i^2)}$
- $\|\cdot\|_2$ 2-norm of a matrix X given by $\|X\|_2 = \max_{X} \frac{\|XX\|_2}{\|Y\|_2}$
- $\|\cdot\|_F$ Frobenius norm of a matrix X given by $\|X\|_F = \sqrt{\sum_{i=1}^{n} \sum_{j=1}^{m} |X_{ij}|^2}$
- $\|\cdot\|_{*}$ Nuclear norm of a matrix X given by $\|X\|_{*} = \text{trace}\left(\sqrt{X^{*}X}\right) = \sum_{i=1}^{m} \sigma_{i}$ (for $m \leq n$)
 - (\cdot, \cdot) Inner product. For functions, $(f(x), g(x)) = \int_{-\infty}^{\infty} f(x)g^*(x)dx$
 - (·, ·) Inner product. For vectors, (u, v) = u*v.

Operators, Functions, and Maps

- F Fourier transform
- F Discrete-time dynamical system map
- F1 Discrete-time flow map of dynamical system through time r
 - f Continuous-time dynamical system
- Gabor transform

xbx

Operators, Functions, and Maps, continued

- G Transfer function from inputs to outputs (Chapter 8)
- g Scalar measurement function on x
- g Vector-valued measurement functions on x
- J Cost function for control
- 2 Loss function for support vector machines (Chapter 5)
- K Koopman operator (continuous time)
- K1 Koopman operator associated with time t flow map
 - Laplace transform
 - L Loop transfer function (Chapter 8)
 - L Linear partial differential equation (Chapters 11 and 12)
 - N Nonlinear partial differential equation
- O Order of magnitude
- S Sensitivity function (Chapter 8)
- T Complementary sensitivity function (Chapter 8)
- W Wavelet transform
- μ Incoherence between measurement matrix C and basis Ψ
- K Condition number
- φ Koopman eigenfunction
- ∇ Gradient operator
- Convolution operator

Most Common Acronyms

- CNN Convolutional neural neural network DL Deep learning DMD Dynamic mode decomposition FFT Fast Fourier transform ODE Ordinary differential equation FCA Principal components analysis DE partial differential equation FOD Proper orthogonal decomposition FOM Reduced order model
- SVD Singular value decomposition

Other Acronyms

ADM	Alternating directions method
AIC	Akaike information criterion
ALM	Augmented Lagrange multiplier
ANN	Artificial neural network
ARMA	Autoregressive moving average
ARMAX	Autoregressive moving average with exogenous input
BIC	Bayesian information criterion
BPOD	Balanced proper orthogonal decomposition
DMDc	Dynamic mode decomposition with control
CCA	Canonical correlation analysis
CFD	Computational fluid dynamics
CoSaMP	Compressive sampling matching pursuit
CWT	Continuous wavelet transform
DEIM	Discrete empirical interpolation method
DCT	Discrete cosine transform
DFT	Discrete Fourier transform
DMDc	Dynamic mode decomposition with control
DNS	Direct numerical simulation
DWT	Discrete wavelet transform
ECOG	Electrocorticography
cDMD	Extended DMD
EIM	Empirical interpolation method
EM	Expectation maximization
EOF	Empirical orthogonal functions
ERA	Eigensystem realization algorithm
ESC	Extremum-seeking control
GMM	Gaussian mixture model
HAVOK	Hankel alternative view of Koopman
JL.	Johnson-Lindenstrauss
KL.	Kullback-Leibler
ICA	Independent component analysis

00

Other Acronyms, continued

- KLT Karhunen-Loève transform
- LAD Least absolute deviations
- LASSO Least absolute shrinkage and selection operator
 - LDA Linear discriminant analysis
 - LQE Linear quadratic estimator
 - LOG Linear quadratic Gaussian controller
 - LQR Linear quadratic regulator
 - LTI Linear time invariant system
- MIMO Multiple input, multiple output
 - MLC Machine learning control
 - MPE Missing point estimation
- mrDMD Multi-resolution dynamic mode decomposition
- NARMAX Nonlinear autorerressive model with exorenous inputs
 - NLS Nonlinear Schrödinger equation
 - OKID Observer Kalman filter identification
 - PBH Popov-Belevitch-Hautus test
 - PCP Principal component pursuit
- PDE-FIND Partial differential equation functional identification of nonlinear dynamics
 - PDF Probability distribution function
 - PID Proportional-integral-derivative control
 - PIV Particle image velocimetry
 - RIP Restricted isometry property
 - rSVD Randomized SVD
 - RKHS Reproducing kernel Hilbert space
 - RNN Recurrent neural network
 - RPCA Robust principal components analysis
 - SGD Stochastic gradient descent
 - SINDy Sparse identification of nonlinear dynamics
 - SISO Single input, single output
 - SRC Sparse representation for classification
 - SSA Singular spectrum analysis
 - STFT Short time Fourier transform
 - STLS Sequential thresholded least-squares
 - SVM Support vector machine
 - TICA Time-lagged independent component analysis
 - VAC Variational approach of conformation dynamics

Part I

Dimensionality Reduction and Transforms

The singlet value decomposition (SVD) is among the most important matrix factorization of the comparisonic arc providing a foundation for reardy al of the data methods in this book. The SVD provides a summicrical value matrix documposition that can be useful a value of the system approximations to matrix and to profering provide-interves of mosquare matrices is the approximation to matrix and the system provide system of the system matching is applied in the system of the system of the system of the system is decomposed unto its most statistically descriptive factors SVDPCA has been applied to a value variety of produces the system of the system of the system of the system of the system is decomposed unto its most statistically descriptive factors SVDPCA has been applied to a value variety of produces the system of the system of the system of the system of the system is decomposed unto its most statistically descriptive factors SVDPCA has been applied to a system of the system of t

In a sense, the SVD generalizes the concept of the fast Fourier transform (FFT), which will be the subject of the next chapter Many engineering texts begin with the FFT, as it is the basis of many classical analytical and numerical results. However, the FFT works in disculted settings, and the SVD is an one generic data-driven technique. Because this book is focused on data, we begin with the SVD, which may be thought of as providing a basis in an individual setting and the specific data, as opposed to the FFT, which provides a poweric basis.

Is may domain, complex systems will generate data that is summity arranged in the generatives, at more presentation and the system of the system of the system in the system of the system of the system of the system of the system for measurement at a given time. The data at each instant is time is under demonstration, the system of the data may is high dimensional column vector, for reading of the system of the function of a system of the system o

1.1 Overview

Here we introduce the SVD and develop an intuition for how to apply the SVD by demonstrating its use on a number of moviaring examples. The SVD will provide a foundation for many other techniques, developed in this book, including classification methods in Chaper 5, the dynamic mode decomposition (OMD) in Chapter 7, and the proper orthogonal decomposition (POD) in Chapter 11. Detailed mathematical properties are discussed in the following sections: High dimensionality is a common challenge in processing data from complex systems. These systems may involve large measured data sets including and/a, image, or video data. The data may also be generated from a physical system, such as neural recordings from a brain, of high deviceity measurements from a simulation or experiment. In many manually occurring systems, it is observed that data exhibit dominant patterns, which may be characterized by a low-dimensional attractor or manifold [22, 23, 21].

As an example, consider images, which repically contain a large number of messarement (pitcsh), and an effective (example of the hybrid messarian view power. However, near hanges are highly compressible, messing that the relevant automation may be repdonced in the hybrid point of the hybrid messarian view point of the discoveral in dipth hybrid methods that hybrid messarian view point of the number of the hybrid messarian view point of the hybrid messarian sphere or the numbers which has a broched account of the approximation point of the numbers of the hybrid messarian view point of the hybrid fluid simulations, typically require at last millions or billion of digrass of fluidon, the sheaf hybrid disclosed in the systemic at the structure.

The SVD provides a systematic way to determine a low-dimensional approximation to high-dimensional data in terms of dominant patterns. This technique is domina-driven in that patterns are discovered purely from data, without the addition of expert knowledge or imitiation. The SVD is immerically asthem and provides a hierarchical representation of the data in terms of a new coordinate system defined by dominant correlations within the data. Mereover, the SVD is immarked by astisf or any matrix, unlike the eigendecomossion.

The SVD has many powerful applications byout dimensionality relativities of highdimensional data in its used to compare the pseudo-inverse of mon-square matrices, providing solutions to underdetermined or overdetermined matrix equations, $\mathbf{Ax} = \mathbf{b}$. We will also use the SVD to do-noise data sets. The SVD is kinesis important to characterize the lings and coupte geoseney of a linear map between vector spaces. These applications will all be explored in this chapter, providing an innition for matrices and high-dimensional data.

Definition of the SVD

Generally, we are interested in analyzing a large data set $X \in \mathbb{C}^{n \times m}$:

$$\mathbf{X} = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{x}_1 & \mathbf{x}_2 & \cdots & \mathbf{x}_n \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}. \quad (1.1)$$

The columns $x_{k} \in \mathbb{C}^{M}$ may be measurements from simulations or experiments. For example, columns may represent images that have been related and into column vectors with as many elements as pixels in the image. The column vectors may also represent the state of a physical system that is evolving in time, such as the fluid velocity at a set of discrete points, a set of mercul measurements, or the state of a weather simulation with one square kinemeter resolution.

The index k is a label indicating the k^{th} distinct set of measurements. For many of the examples in this book, **X** will consist of a *time-series* of data, and $x_k = \mathbf{x}(k\Delta t)$. Often the state-dimension n is very large, on the order of millions of billions of decrees of freedom. The columns are often called *swapshots*; and *m* is the number of snapshots in **X**. For many systems $n \gg m$, resulting in a *kall-skinny* matrix, as opposed to a *short-far* matrix when $n \ll m$.

The SVD is a unique matrix decomposition that exists for every complex-valued matrix $\mathbf{X} \in \mathbb{C}^{n \times m}$:

$$X = U\Sigma V^{*}$$
 (1.2)

where $\mathbf{U} \in \mathbb{C}^{n \times m}$ and $\mathbf{V} \in \mathbb{C}^{n \times m}$ are awitary matrices¹ with orthonormal columns, and $\mathbf{\Sigma} \in \mathbb{R}^{n \times m}$ is a matrix with real, nonnegative entries on the diagonal and zeros off the diagonal. Here * denotes the complex conjugate transpose². As we will discover throughout this charter, the condition that \mathbf{U} and \mathbf{V} are unitary is used extensively.

When $n \ge m$, the matrix Σ has at most m nonzero elements on the diagonal, and may be written as $\Sigma = \begin{bmatrix} \hat{\Sigma} \\ 0 \end{bmatrix}$. Therefore, it is possible to *exactly* represent X using the *economy* SVD:

$$\mathbf{X} = \mathbf{U} \mathbf{\Sigma} \mathbf{V}^* = \begin{bmatrix} \hat{\mathbf{U}} & \hat{\mathbf{U}}^{\perp} \end{bmatrix} \begin{bmatrix} \hat{\mathbf{\Sigma}} \\ \mathbf{0} \end{bmatrix} \mathbf{V}^* = \hat{\mathbf{U}} \hat{\mathbf{\Sigma}} \mathbf{V}^*.$$
 (1.3)

The full SVD and economy SVD are shown in Fig. 1.1. The columns of \hat{U}^{\perp} span a vector space that is complementary and orthogonal to that spanned by \hat{U} . The columns of U are called left singular vectors of X and the columns of W are right singular vectors. The diagonal elements of $\tilde{\Sigma} \in \mathbb{C}^{n\times n}$ are called *singular values* and they are ordered from largeret to smallex. The rank of X is could be the number of nonzero singular values and they are observed.

Computing the SVD

The SVD is a convention of comparisonal wineter and engineering, and the muterical implementator of the SVD is the hiperturnal and the SVD is the muterical implementatory of the SVD is the muterical implementatory of the SVD is the muterical implementatory of the SVD is the software and a single immediate least in the software of the SVD is the software of the SVD is the

Matlab. In Matlab, computing the SVD is straightforward:

¹ A source matrix U is unitary if UU* = U*U = L

² For real-valued matrices, this is the same as the regular transpose $X^{\pm} \equiv X^{T}$.



Figure 1.1 Schematic of matrices in the full and economy SVD.

For non-square matrices X, the economy SVD is more efficient:

>>[Uhat,Shat,V] = mvd(X,'econ'); & economy mixed SVD

Python

```
>>> X support numpy as np
>>> X = np.random.rand(5, 1) % create random data matrix
>>> U, S, V = np.linslg.avd(X,full_matricessTrue) % full SVD
>>> Uhat, Shat, Vhat = np.linslg.avd(X, full_matricessTalse)
% economy SVD
```

```
R
```

```
> X <- replicate(3, rnorm(5))
> z <- svd(X)
> U <- s$u
> S <- diag(s$d)
> V <- s$v
```

Mathematica

In:= X=RandomReal[{0,1}, {5,3}]
In:= {U,S,V} = SingularValueDecomposition[X]

Other Languages

The SVD is also available in other languages, such as Fortran and C++. In fact, most SVD implementations are based on the LAPACK (Linear Algebra Package) [13] in Fortran. The SVD routine is designated DGESVD in LAPACK, and this is wrapped in the C++ libraries Armadillo and Eigen.

Historical Perspective

The SVD has a long and rich history, ranging from early work developing the theoretical frontidions to modern work on comparisonal stability and efficiency. There is an excellent historical review by Stewart [502], which provides context and many important details. The review forsues on the early faceretical work of Belmmi and Joadan [1737, JSyvester (1899), Schmid (1997), and Weyl (1917). It also discusses mere enter work, including the semiant comparison during of Golda and collaborators [212, 211]. In additon, there are many excellent endness on the SVD in moder texts [524, 17, 106].

Uses in This Book and Assumptions of the Reader

The SVD is the basis for many related techniques in dimensionality reduction. These methods include principal component analysis (PCA) in stratical (FHZ 58, 257). Kathunen-Love transform (KLT) [203, 349], empirical orthogonal functions (EOFs) in clamac [144], the projer orthogonal docomologion (POD) in third dynamics [21], and canonical correlation analysis (FCA) [131]. Although developed independently in a magropresessed. There is an excellent discussion about the relationship between the SVD, the KLT and FCAN of correlation [200].

The SVD is also widely used in system identification and control theory to obtain reduced order models that are balanced in the sense that states are hierarchically ordered in terms of their ability to be observed by measurements and controlled by actuation [388].

For this chapter, we assume that the reader is familiar with linear algebra with some experience in computation and numerics. For review, there are a number of excellent books on numerical linear algebra, with discussions on the SVD [524, 17, 316].

1.2 Matrix Approximation

Perhaps the most useful and defining property of the SVD is that it provides an optimal low-rank approximation to a matrix X. In fact, the SVD provides a hierarchy of low-rank approximations, since a rank-r approximation is obtained by keeping the leading r singular values and vectors, and discarding the rest.

Schmidt (of Gram-Schmidt) generalized the SVD to function spaces and developed an approximation theorem, establishing truncated SVD as the optimal low-rank approximation of the underlying matrix X [476]. Schmidt's approximation theorem was rediscovered by Eckart and Young [170], and is sometimes referred to as the Eckart-Young theorem.

Theorem 1 (Eckart-Young [170)) The optimal rank-r approximation to X, in a leastsquares sense, is given by the rank-r SVD transation \tilde{X} :

$$\underset{\hat{\mathbf{X}} \text{ st. mak}(\hat{\mathbf{X}}) = \mathbf{r}}{\operatorname{argmin}} \|\mathbf{X} - \hat{\mathbf{X}}\|_{F} = \hat{\mathbf{U}}\hat{\mathbf{\Sigma}}\hat{\mathbf{V}}^{*}. \quad (1.4)$$

Here, \tilde{U} and \tilde{V} denote the first r leading columns of U and V, and $\tilde{\Sigma}$ contains the leading $r \times r$ sub-block of Σ . $\|\cdot\|_{F}$ is the Frobenius norm.

Here, we establish the notation that a truncated SVD basis (and the resulting approximated matrix \tilde{X}) will be denoted by $\tilde{X} = \tilde{U}\tilde{\Sigma}\tilde{V}^*$. Because Σ is diagonal, the rank-r SVD approximation is given by the sum of r distinct matrix imatrices:

$$\hat{\mathbf{X}} = \sum_{k=1}^{r} \sigma_{k} \mathbf{u}_{k} \mathbf{v}_{k}^{*} = \sigma_{1} \mathbf{u}_{1} \mathbf{v}_{1}^{*} + \sigma_{2} \mathbf{u}_{2} \mathbf{v}_{2}^{*} + \dots + \sigma_{r} \mathbf{u}_{r} \mathbf{v}_{r}^{*}. \quad (1.5)$$

This is the so-called dyadic summation. For a given rank r, there is no better approximation for X_i in the ℓ_2 sense, than the truncated SVD approximation \hat{X} . Thus, high-dimensional data may be well described by a few dominant patterns given by the columns of \hat{U} and \hat{V} .

This is an important property of the SVD, and we will remark to it many times. These memory campuing of this section to constrain high-demonsion mecoursements, resulting in a large data matrix X. However, there are often dominant to dimensional patterns in high-dimensional material and the section of the section of the section of the dimensional patterns in the section of the section of the section of the section of the benefit of characterial metal section of the section of the section of the section of the benefit of characterian dimension of large data sect, yielding a tractable basic for a dominer (and the section of the Charger 77, and fits SVD hasin protokes a latenachy of massis that characteria the dominer of the metal dominer (2).

Truncation

The truncated SVD is illustrated in Fig. 1.2, with $\hat{U}, \hat{\Sigma}$ and \hat{V} denoting the truncated matrices. If X does not have full rank, then some of the singular values in $\hat{\Sigma}$ may be zero. and the truncated SVD may still be exact. However, for truncation values is that are similler than the number of nonzero singular values (i.e., the rank of X), the truncated SVD only approximents X:

$$\mathbf{X} \approx \hat{\mathbf{U}} \hat{\mathbf{\Sigma}} \hat{\mathbf{V}}^*$$
. (1.6)

There are numerous choices for the truncation rank r, and they are discussed in Sec. 1.7. If we choose the truncation value to keep all non-zero singular values, then $\mathbf{X} = \hat{\mathbf{U}}\hat{\mathbf{\Sigma}}\hat{\mathbf{V}}^{*}$ is exact.

Example: Image Compression

We demonstrate the idea of matrix approximation with a simple example: image compression. A recentry the metherophone this bok is that large data area then contain underlying patterns that functions. Sharing images present a simple and image increasing the intercent compressibility. A graycase it image may be hought of as a treal-vialeed matrix $\mathbf{X} \in \mathbb{R}^{2008}$, where n and n are the number of picels in the vertical and horizontal directions, presently of the picels in the vertical materia bottorismal directions, respectively.¹ Depending on the basis of presenting tipe-dispective for the picels in the vertical materia bottorismal directions, respectively.² Depending on the basis of presenting tipe-dispective. Fourier frequency domain, SVD transform coordinates), images may have very compact amovimations.

³ It is not uncommon for image size to be specified as horizontal by vertical, i.e. X^T = R^{m×m}, although we stick with vertical by horizontal to be consistent with generic matrix notation.



Figure 12 Schematic of truncated SVD. The subscript 'rem' denotes the remainder of $\hat{U}, \hat{\Sigma}$ or V after truncation.

Consider the image of Montecut the same dap in Fig. 1.3. This image has 2000 s 1500 pitch. It is possible to take the SVD of this image and plot the diagonal singular values, as in Fig. 1.4. Figure 1.3 shows the approximate matrix **X** for various truncation values F(y) = 100. But the image variance. The SVD immutation results in a compression of the absorbed SVD of the image variance. The SVD immutation results in a compression discover and the structure of the SVD immutation results in a compression discover and the structure of the SVD immutation results in a compression discover and the structure of the SVD immutation of the SVD immutation discover and the structure of the SVD immutation of the SVD immutation of the SVD immutation discover and the SVD immutation of the SVD immutation of the SVD immutation of the SVD immutation discover and the SVD immutation of the SVD immutation of

First, we load the image:

```
Asimread('.../DATA/dog.jpg');
Xadouble(rgbZgray(A)) % Convert REG-sgray, 256 bit-sdouble.
nx = size(X,1); ny = size(X,2);
imagesc(X), axis of colormap gray
```

and take the SVD:

[U,S,V] = avd(X);

Next, we compute the approximate matrix using the truncated SVD for various ranks (r = 5, 20, and 100):

```
for r=[5 20 100]; # Truncation value
    Xapprox = U(:,l:r)_s((l:r,l:r)_sV(:,l:r)'; # Approx. image
    figure, images(Xapprox), axis off
    title(['r=',numlstr(r,'%d'),']);
    and
```



Figure 1.3 Image compression of Mondecai the snow dog, truncating the SVD at various ranks r. Original image resolution is 2000 × 1500.

Finally, we plot the singular values and cumulative energy in Fig. 1.4:

```
subplot(1,2,1), semilogy(diag(S),'k')
subplot(1,2,2), plot(cussum(diag(S))/sum(diag(S)),'k')
```

1.3 Mathematical Properties and Manipulations

Here we describe important mathematical properties of the SVD including geometric interpretations of the unitary matrices U and V as well as a discussion of the SVD in terms of



Figure 1.4 (a) Singular values σ_{2} . (b) Cumulative energy in the first k modes.



Figure 1.5 Correlation matrices XX* and X*X for a matrix X obtained from an image of a dog. Note that both correlation matrices are symmetric.

dominant correlations in the data X. The relationship between the SVD and correlations in the data will be explored more in Section 1.5 on principal components analysis.

Interpretation as Dominant Correlations

The SVD is closely related to an eigenvalue problem involving the correlation matrices XX* and X*X, shown in Fig. 1.5 for a specific image, and in Figs. 1.6 and 1.7 for generic matrices. If we plug (1.3) into the row-wise correlation matrix XX* and the column-wise correlation matrix X*X, we find:

$$\mathbf{X}\mathbf{X}^{*} = \mathbf{U}\begin{bmatrix} \hat{\mathbf{\Sigma}} \\ \mathbf{0} \end{bmatrix} \mathbf{V}^{*}\mathbf{V}\begin{bmatrix} \hat{\mathbf{\Sigma}} & \mathbf{0} \end{bmatrix} \mathbf{U}^{*} = \mathbf{U}\begin{bmatrix} \hat{\mathbf{\Sigma}}^{2} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} \end{bmatrix} \mathbf{U}^{*}$$
 (1.7a)

$$\mathbf{X}^{*}\mathbf{X} = \mathbf{V}\begin{bmatrix} \hat{\mathbf{\Sigma}} & \mathbf{0} \end{bmatrix} \mathbf{U}^{*}\mathbf{U}\begin{bmatrix} \hat{\mathbf{\Sigma}} \\ \mathbf{0} \end{bmatrix} \mathbf{V}^{*} = \mathbf{V}\hat{\mathbf{\Sigma}}^{2}\mathbf{V}^{*}.$$
 (1.7b)



Figure 1.8 Correlation matrix XX* is formed by taking the inner product of rows of X.



Figure 1.7 Correlation matrix X*X is formed by taking the inner product of columns of X.

Recalling that U and V are unitary, U, Σ , and V are solutions to the following eigenvalue problems:

$$XX^*U = U\begin{bmatrix} \hat{\Sigma}^2 & 0\\ 0 & 0 \end{bmatrix}, \quad (1.8a)$$

$$X^*XV = V\hat{\Sigma}^2$$
. (1.8b)

In other words, each nonzero singular value of X is a positive square root of an eigenvalue of X'X and of XX^{*}, which have the same nonzero eigenvalues. It follows that if X is selfadjoint (i.e. X = X^{*}), then the singular values of X are equal to the absolute value of the eigenvalues of X.

This provides an intuitive interpretation of the SVD, where the columns of U are eigenvectors of the correlation matrix XX^* and columns of V are eigenvectors of X^*X . We choose to arrange the singular values in descending order by magnitude, and thus the columns of U are hierarchically ordered by how much correlation they capture in the columns of U are hierarchically cordered by how much correlation they capture in the

Method of Snapshots

It is often impractical to construct the matrix XX* because of the large size of the statedimension n. let alone solve the eigenvalue problem: if x has a million elements, then XX*



Figure 1.8 Geometric illustration of the SVD as a mapping from a sphere in R⁴¹ to an ellipsoid in R⁴⁰.

has a trillion elements. In 1987, Sirovich observed that it is possible to bypass this large matrix and compute the first *m* columns of U using what is now known as the method of sumshots (1490).

Instead of computing the eigen-decomposition of XX^{*} to obtain the left singular vectors (), we only compute the eigen-decomposition of XX^{*} thick is much smaller and more manageable. From (1.8b), we then obtain V and $\tilde{\Sigma}$. If there are zero singular values in $\tilde{\Sigma}$, then we only keep the r non-zero part, $\tilde{\Sigma}$, and the corresponding columns \tilde{V} of V. From these matrices, it is then possible to approximate \tilde{U} , the first r columns of U, as follows:

$$\tilde{U} = X \tilde{V} \tilde{\Sigma}^{-1}$$
(1.9)

Geometric Interpretation

The columns of the matrix U provide an orthonormal basis for the column space of X. Similarly, the columns of V provide an orthonormal basis for the row space of X. If the columns of X are spatial measurements in time, then U encode suporal patterns, and V encode temporal patterns.

One property data makes the SVD particularly useful is the fact that hold . and V are mainly matrices such that UU = $|U| = 1 - k_{cost}$ and VV = $|V| = - k_{cost}$. This means that solving a system of equations involving U or V is as simple as multiplication by the transpose, which is such as $O(n^2)$. As most in the previous section and in f(T), the SVD is inverse, which such as $O(n^2)$. As most in the previous section and in f(T), the SVD is an $V = - k_{cost}$ and $V = - k_{cost}$.

The SVD of X may be interpreted geometrically based on how a hypersphere, given by $S^{r-1} \triangleq \{x | | x | x = 1 | \subset \mathbb{R}^{n}$ magn into an ellipsical, $|y| = X to f x < z^{r-1} | \subset \mathbb{R}^{n}$, through X. This is shown graphically in Fig. 13 for a sphere in \mathbb{R}^{2} and a mapping X with three non-zero singular values. Because the mapping through X (i.e., marks multiplication) is linear, knowing how it mays the unit sphere determines how all other vectors will map.
For the specific case shown in Fig. 1.8, we construct the matrix X out of three rotation matrices, R_x , R_y , and R_z , and a fourth matrix to stretch out and scale the principal axes:

$$\begin{split} \mathbf{X} = & \underbrace{ \begin{bmatrix} \cos(\theta_{1}) & -\sin(\theta_{2}) & 0 \\ \sin(\theta_{2}) & \cos(\theta_{2}) & 0 \\ 0 & 1 \end{bmatrix} }_{\mathbf{R}_{i}} \begin{bmatrix} \cos(\theta_{2}) & 0 & \sin(\theta_{2}) \\ 0 & 1 & 0 \\ -\sin(\theta_{2}) & 0 & \cos(\theta_{2}) \end{bmatrix} \\ \times & \underbrace{ \begin{bmatrix} 1 & 0 & 0 \\ 0 & \cos(\theta_{1}) & -\sin(\theta_{2}) \\ 0 & \sin(\theta_{1}) & \cos(\theta_{2}) \end{bmatrix} }_{\mathbf{R}_{i}} \begin{bmatrix} \sigma_{i} & 0 & 0 \\ 0 & \sigma_{2} & 0 \\ 0 & 0 & \sigma_{3} \end{bmatrix}. \end{split}$$

In this case, $\theta_1 = \pi/15$, $\theta_2 = -\pi/2$, and $\theta_2 = -\pi/20$, and $\sigma_1 = 3$, $\sigma_2 = 1$, and $\sigma_3 = 0.5$. These rotation matrices do not commute, and so the order of rotation matters. If one of the singular values is zero, then a dimension is removed and the ellipsoid collapses onto a lower-dimensional subspace. The product $\mathbf{R}_4\mathbf{R}_5\mathbf{R}_6$ is the unitary matrix U in the SVD of X. The matrix V is the identity:

Code 1.1 Construct rotation matrices

```
Lists = [gr(152, -gr(2g) = [1/20])
impose = dam([1/15, -gr(2g)])
impose = [1/20] impose =
```

Code 1.2 Plot sphere

```
[x, y, z] = sphere(25);
hl=surf(x, y, z);
```

Code 1.3 Map sphere through X and plot resulting ellipsoid

```
 \begin{split} & x = 0 \, \text{st}_{1} \ \ y \equiv 0 \, \text{st}_{1} \ \ x \equiv 0 \, \text{st}_{1} \\ & \text{for isized}_{1,2} \\ & \text{for isized}_{1,2} \\ & \text{state}_{1,2} \\ & \text{st}_{1,2} \\ & \text{
```

Invariance of the SVD to Unitary Transformations

A such property of the SVD is that if we left or right-multiply our data matrix **X** by a minuty tandomization, it preserves the term in the SVD, except for the corresponding left or right unitary matrix **U** or *Y*, respectively. This has important implications, since the discover Fourier transmon (DFF) see Calputer 2017; *J* is a minuty matrix moder of the modes of the SVD of Aax = FX will be each of PCU is adding the matrix of each YDD to minuty matsformation canable the use of compressed measurements to reconstruct SVD matches that are space in some than 60 minute (see Calputer).

The invariance of SVD to unitary transformations is geometrically intuitive, as unitary transformations rotate vectors in space, but do not change their inner products or correlation structures. We denote a left unitary transformation by C, so that Y = CX, and a right unitary transformation by P_{ν} so that $Y = XP_{\nu}$. The SVD of X will be denoted $U_X \Sigma_X V_X^c$ and the SVD of Y will be $U_X \Sigma_V V_{\nu}^c$.

Left Unitary Transformations

First, consider a left unitary transformation of X: Y = CX. Computing the correlation matrix Y*Y, we find

$$Y^{*}Y = X^{*}C^{*}CX = X^{*}X.$$
 (1.10)

The projected data has the same eigendecomposition, resulting in the same V_X and Σ_X . Using the method of snapshots to reconstruct U_Y , we find

$$U_Y = YV_X \Sigma_X^{-1} = CXV_X \Sigma_X^{-1} = CU_X.$$
 (1.11)

Thus, $U_Y = CU_X$, $\Sigma_Y = \Sigma_X$, and $V_Y = V_X$. The SVD of Y is then:

$$Y = CX = CU_X \Sigma_X V_X^*. \quad (1.12)$$

Right Unitary Transformations

For a right unitary transformation $Y = XP^*$, the correlation matrix Y^*Y is:

$$Y^{*}Y = PX^{*}XP^{*} = PV_{X}\Sigma_{X}^{2}V_{X}^{*}P^{*}$$
, (1.13)

with the following eigendecomposition

$$Y^*YPV_X = PV_X\Sigma_X^2.$$
 (1.14)

Thus, $V_Y = PV_X$ and $\Sigma_Y = \Sigma_X$. We may use the method of snapshots to reconstruct U_Y :

$$U_Y = YPV_X\Sigma_X^{-1} = XV_X\Sigma_X^{-1} = U_X.$$
 (1.15)

Thus, $U_Y = U_X$, and we may write the SVD of Y as:

$$Y = XP^* = U_X \Sigma_X V_X^* P^*. \qquad (1.16)$$

1.4 Pseudo-Inverse, Least-Squares, and Regression

Many physical systems may be represented as a linear system of equations:

$$4x = b$$
, (1.17)

where the constraint matrix A and vector b are known, and the vector x is unknown. If A is a square, invertible matrix (i.e., A has nonzero determinant), then there exists a unique solution x for every b. However, when A is either singular or nectangular, there may be one, none, or infinitely many solutions, depending on the specific b and the column and row spaces of A.

First, consider the underdetermined system, where $\Lambda \in \mathbb{C}^{noni}$ and $n \ll m$ (i.e., Λ is a short-far matrix), so that there as fewer equations than unknowns. This type of system is likely to have full column mak, since it has many more columns than are required for a since the since the since state of the since the since s

Similarly, consider the overdetermined system, where $n \gg m$ (i.e., a tall-skinny matrix), so that there are more equations than unknowns. This matrix cannot have a full column rank, and so it is guaranteed that there are vectors b that have no solution x. In fact, there will only be a solution x if b is in the column space of A, i.e. b $\leq ol(ol A)$.

Technically, there may be some choices of **b** that admit infinitely many solutions **x** for a tall-skinny matrix **A** and other choices of **b** that admit zero solutions even for a short-fat matrix. The solution space to the system in (1.17) is determined by the four fundamental subspaces of $\mathbf{A} = \vec{\mathbf{U}} \vec{\mathbf{\Sigma}} \vec{\mathbf{V}}$, where the rank *r* is chosen to include all nonzero singular values:

- The column space, col(A), is the span of the columns of A, also known as the range. The column space of A is the same as the column space of U;
- The orthogonal complement to col(A) is ker(A*), given by the column space of Û⊥ from Fig. 1.1;
- The row space, row(A), is the span of the rows of A, which is spanned by the columns of V. The row space of A is equal to row(A) = col(A^{*});
- The kernel space, ker(A), is the orthogonal complement to row(A), and is also known as the null space. The null space is the subspace of vectors that map through A to zero, i.e., Ax = 0, given by col(V[±]).

More precisely, if $\mathbf{b} \in \mathrm{col}(\mathbf{A})$ and if dim (ker(\mathbf{A})) $\neq 0$, then there are infinitely many solutions \mathbf{x} . Note that the condition dim (ker(\mathbf{A})) $\neq 0$ is guaranteed for a short-fat matrix. Similarly, if $\mathbf{b} \notin \mathrm{col}(\mathbf{A})$, then there are no solutions, and the system of equations in (1.17) are called inconsistent.

The fundamental subspaces above satisfy the following properties:

$$col(\mathbf{A}) \oplus ker(\mathbf{A}^*) = \mathbb{R}^n$$
 (1.18a)

$$col(\mathbf{A}^{*}) \oplus ker(\mathbf{A}) = \mathbb{R}^{n}$$
. (1.18b)

Remark 1 There is an extensive literature on random matrix theory, where the above stereotypes are almost certainly true, meaning that they are true with high probability. For example, a system Ax = b is extremely unlikely to have a solution for a random matrix $A \in \mathbb{R}^{n\times n}$ and random vector $b \in \mathbb{R}^n$ with n > m, since there is little chance that b is in

⁴ It is easy to construct degenerate examples where a short-fat matrix does not have full column rank, such as $\mathbf{A} = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & 1 & 1 & 1 \end{bmatrix}.$

the column space of A. These properties of random matrices will play a prominent role in compressed sensing (see Chapter 3).

In the overdetermined case when no solution exists, we would often like to find the solution **x** that minimizes the sum-squared error $\|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2^2$, the so-called *loast-squares* solution. Note that the least-squares solution also minimizes $\|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2$. In the underdetermined case when infinitely many solutions exist, we may like to find the solution **x** with minimum norm 184's so that $\mathbf{A}\mathbf{x} = \mathbf{b}$ the so-called minimus-norm solution.

The SVD is the technique of choice for these important optimization problems. First, if we substitute an exact truncated SVD $\mathbf{A} = \tilde{\mathbf{U}}\tilde{\mathbf{x}}\tilde{\mathbf{V}}^{2}$ in for \mathbf{A} , we can "invert" each of the matrices $\tilde{\mathbf{U}}, \tilde{\mathbf{x}}$, and $\tilde{\mathbf{V}}^{4}$ in turn, resulting in the Moore-Penrose *left pseudo-inverse* [425, 426, 433, 572] \mathbf{A}^{1} of \mathbf{A} :

$$\mathbf{A}^{\dagger} \stackrel{\circ}{=} \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} \tilde{\mathbf{U}}^{*} \implies \mathbf{A}^{\dagger} \mathbf{A} = \mathbf{I}_{m \times m}.$$
 (1.19)

This may be used to find both the minimum norm and least-squares solutions to (1.17):

$$A^{\dagger}A\bar{x} = A^{\dagger}b \implies \bar{x} = \tilde{V}\tilde{\Sigma}^{-1}\tilde{U}^{*}b.$$
 (1.20)

Plugging the solution x back in to (1.17) results in:

$$A\tilde{x} = \tilde{U}\tilde{\Sigma}\tilde{V}^*\tilde{V}\tilde{\Sigma}^{-1}\tilde{U}^*b$$
 (1.21a)

$$= UU^{b}b.$$
 (1.21b)

Note that $\hat{U}\hat{U}^*$ is not necessarily the identity matrix, but is rather a projection onto the column space of \hat{U} . Therefore, \hat{x} will only be an exact solution to (1.17) when **b** is in the column space of \hat{U} , and therefore in the column space of **A**.

Comparing the pseudo-inverse A¹ is computationally efficient, after the expensive upfront cost of computing the SVD. Inverting the uniary matrices \hat{U} and \hat{V} involves matrix multiplication by the transpose matrices, which are $O(n^2)$ operations. Inverting \hat{z} is even more efficient since it is a diagonal matrix, requiring O(n) operations. In contrast, inverting a dense square matrix would require an $O(n^2)$ operation.

One-Dimensional Linear Regression

Regression is an important statistical tool to relate variables to one another based on data [360]. Consider the collection of data in Fig. 1.9. The red \times 's are obtained by adding Gaussian white noise to the black line, as shown in Code 1.4. We assume that the data is linearly related, as in (1.7), and we use the pseudo-inverse to find the least-squares solution for the shoes r below (blue dashed line, shown in Code 1.5.

$$\begin{bmatrix} \mathbf{I} \\ \mathbf{b} \\ \mathbf{J} \end{bmatrix} = \begin{bmatrix} \mathbf{I} \\ \mathbf{a} \\ \mathbf{a} \end{bmatrix} \mathbf{x} = \tilde{\mathbf{U}} \tilde{\mathbf{V}}^* \mathbf{x}. \quad (1.22a)$$

$$\Rightarrow x = \tilde{V}\tilde{\Sigma}^{-1}\tilde{U}^{*}b.$$
 (1.22b)

In (1.22b), $\hat{\Sigma} = \|\mathbf{a}\|_2$, $\hat{V} = 1$, and $\hat{U} = \mathbf{a}/\|\mathbf{a}\|_2$. Taking the left pseudo-inverse:

$$x = \frac{\mathbf{a}^* \mathbf{b}}{\|\mathbf{a}\|_2^2}$$
 (1.23)



Figure 1.9 Illustration of Encar repression using noisy data.

This makes physical sense, if we think of x as the value that best maps our vector a to the vector b. Then, the best single value x is obtained by taking the dot product of b with the normalized a direction. We then add a second normalization factor $||\mathbf{a}||_2$ because the a in (1.22a) is not normalized.

Note that strange things happen if you use row vectors instead of column vectors in (1.22). Also, if the noise magnitude becomes large relative to the slope x, the pseudoinverse will undergo a phase-change in accuracy, related to the hard-thresholding results in subsequent sections.

Code 1.4 Generate noisy data for Fig. 1.9.

Code 1.5 Compute least-squares approximation for Fig. 1.9.

The procedure above is called *linear regression* in statistics. There is a regress command in Matlab, as well as a **piny** command that may also be used.

Code 1.6 Alternative formulations of least-squares in Matlab.

```
xtilde1 = V inv(S) U' inv(S) with xtilde2 = pinv(a) inv(a) inv(a) inv(b) xtilde3 = regress(b,a)
```



Figure 1.10 Heat data for cement mixtures containing four basic ingredients.

Multilinear regression

Example 1: Cement heat generation data

Find, we begin with a simple built-in Matlah dataset that describes the barg generation for various centerm transverse comprised for those ingredients. In this problem, we are solving (1.17) where $A \in \mathbb{R}^{12/4}$, since there are four ingredients and here the exponention of the four ingredients to the least generation. It is possible to find the minimum error of the four ingredients to the least generation. It is possible to find the minimum error also explores.

Cade 1.7 Multilinear regression for cement heat data.

```
        land huld; # Loud Portlant Common dataset

        A = injections

        A = injections

        D = North

        D =
```

Example 2: Boston Housing Data

In this example, we explore a larger data set to determine which factors best predict prices in the Boston housing market [234]. This data is available from the UCI Machine Learning Repository [24].

There are 13 attributes that are correlated with house price, such as per capita criten rate and property-ast. and: These fatures are regressed onto the price data, and the best fit price prediction is plotted against the true house value in Fig. 1.11, and the regression coefficients are shown in Fig. 1.12. Although the house value is not percedy predicted, the trend agrees quite well. It is often the case that the highest value outliers are not well-captured by simple linear fits, as in the scannels.

This data contains prices and attributes for 506 homes, so the attribute matrix is of size 506 x 13. It is important to nad this matrix with an additional column of ones, to take



Figure 1.11 Multilinear regression of home prices using various factors. (a) Unsorted data, and (b) Data sorted by home value.



Figure 1.12 Significance of various attributes in the regression.

into account the possibility of a nonzero constant offset in the regression formula. This corresponds to the "y-intercept" in a simple one-dimensional linear regression.

Code 1.8 Multilinear regression for Boston housing data.

```
Last homotop, data

\beta = homotop(1, 24) = \hbar homotop values in 22000

A = homotop(1, 24) = \hbar other factors,

\beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = homotop(1, 24) = \hbar other factors, \beta = hom
```

Caution

In general, the matrix U, whose columns are left-singular vectors of X, is a unitary square matrix. Therefore, $U^{*}U = UU^{*} = I_{axxx}$. However, to compute the pseudo-inverse of X, we must compute $X^{*} = \tilde{V} \tilde{\Sigma}^{-1} \tilde{U}^{*}$ since only $\tilde{\Sigma}$ is invertible (if all singular values are nonzero), although $\tilde{\Sigma}$ is no invertible in external in fact, it is senerally not even square).

Until now, we have assumed that $\mathbf{X} = \tilde{U}\tilde{\Sigma}\tilde{\mathbf{V}}^*$ is an exact SVD, so that the rank r includes all nonzero singular values. This guarantees that the matrix $\tilde{\Sigma}$ is invertible.

A complication arises when working with a truncated basis of left singular vectors \tilde{U} . It is still true that $\tilde{U}^{\tilde{U}} = I_{r,r}$, where r is the rank of X. However, $\tilde{U}^{\tilde{U}} \neq I_{wca}$, which is easy to verify numerically on a simple example. Assuming that $\tilde{U}^{\tilde{U}}$ is equal to the identity is one of the most common accidental misuses of the SVD².

1.5 Principal Component Analysis (PCA)

Phiciple components analysis (PC)s is one of the central uses of the SND, providing a dire driven, hierarchica coloradian system to respect high-dimensional correlated data. This coordinate system involves the correlation matrices discribed in Sec. 13. Importantly, PCA perspectises the data by sense asthracian a direting the variance to using before performing the SND. The geometry of the resulting coordinate system is determined by the product of the SND. The geometry of the resulting coordinate system is determined by the product of the system material correlation with the measurements. This theory was developed at 1001 per secand of reference texts.

Tyisality, a number of measurements are collected in a single experiment, and these measurements are arranged into a row vector. The measurements may be denotes of an observable, such as demographic features of a specific human individual A number of measurements are arranged as a specific human individual A number of measurements and the specific human individual A number of measurements. It is the example, the control of the partners' via perfision, being the specific human individual and the partners' via perfision. The the measurement is the constant with PCA literature in the are arranged as columns. However, we choose to be constant with PCA literature is not an example of the trace as *n*_maloud in two phere areas of an example. We take the specific human is the specific human is a specific human in the observable in the specific human is a specific human in the observable in the specific human is a specific human in the specific human i

Computation

We now compute the row-wise mean \tilde{x} (i.e., the mean of all rows), and subtract it from X. The mean \hat{x} is given by

$$\tilde{\mathbf{x}}_{j} = \frac{1}{n} \sum_{i=1}^{n} \mathbf{X}_{ij},$$
 (1.24)

5 The authors are not immune to this, having mistakenly used this fictional identity in an early version of 1961.

and the mean matrix is

$$\mathbf{\tilde{x}} = \begin{bmatrix} 1 \\ \vdots \\ 1 \end{bmatrix} \mathbf{\tilde{x}}.$$
 (1.25)

Subtracting X from X results in the mean-subtracted data B:

$$B = X - B.$$
 (1.26)

The covariance matrix of the rows of B is given by

$$C = \frac{1}{n-1}B^*B.$$
 (1.27)

The first principal component u1 is given as

$$u_1 = \underset{\|u_1\|=1}{\operatorname{argmax}} u_1^* B^* B u_1,$$
 (1.28)

which is the eigenvector of B*B corresponding to the largest eigenvalue. Now it is clear that u₁ is the left singular vector of B corresponding to the largest singular value.

It is possible to obtain the principal components by computing the eigen-decomposition of C:

$$CV = VD$$
, (1.29)

which is guaranteed to exist, since C is Hermitian.

pca Command

In Matlab, there the additional commands pca and princomp (based on pca) for the principal components analysis:

>> [V,score,s2] = pcs(X);

The matrix V is equivalent to the V matrix from the SVD of X, up to sign changes of the columns. The vector S4 contains eigenvalues of the covariance of X, also known as principal component variances; these values are the squares of the singular values. The values have singuly contains the coordinates of each row of B (the mean-subtraced data) in the principal component directions. In general, we often prefer to use the s4d command with the variance percoressing steps described earlier in the section.

Example: Noisy Gaussian Data

Consider the noisy cloud of data in Fig. 1.13 (a), generated using Code 1.9. The data is generated by selecting 10, 000 vectors from a two-dimensional normal distribution with years mean and unit variance. These vectors are then scaled in the x and y directions by the values in Table 1.1 and rotated by $\pi/3$. Finally, the entire cloud of data is translated so that it has a nonzero enter $y_{ce} = [2, 1]^2$.

Using Code 1.10, the PCA is performed and used to plot confidence intervals using multiple standard deviations, shown in Fig. 1.13 (b). The singular values, shown in Tabbe 1.1, match the data scaling. The matrix U from the SVD also closely matches the rotation matrix, up to a sign on the columns:



Table 1.1 Standard deviation of data and normalized singular values.

Figure 1.13 Principal components capture the variance of mean-subtracted Gaussian data (a). The first three standard deviation ellipsoids (red), and the two left singular vectors, scaled by singular valaes $(\sigma_1 a) + a_{cc}$ and $\sigma_{2c} + a_{cc} - syan)$, are shown in (b).

$$\mathbf{R}_{\pi/3} = \begin{bmatrix} 0.5 & -0.8660 \\ 0.8660 & 0.5 \end{bmatrix}$$
, $\mathbf{U} = \begin{bmatrix} -0.4998 & -0.8662 \\ -0.8662 & 0.4998 \end{bmatrix}$

Code 1.9 Generation of noisy cloud of data to illustrate PCA.

 mf = [2]1];
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 0
 <td

Cade 1.10 Compute PCA and plot confidence intervals



Figure 1.14 Simular values for the Ovarian cancer data

```
plot(Xavg(1)+Xstd(1,:),Xavg(2) + Xstd(2,:),'r-')
plot(Xavg(1)+2*Xstd(1,:),Xavg(2) + 2*Xstd(2,:),'r-')
plot(Xavg(1)+3*Xstd(1,:),Xavg(2) + 3*Xstd(2,:),'r-')
```

```
Finally, it is also possible to compute using the pea command:
```

```
>> [V,score,s2] = pcs(X);
>> morm(V_score' - B)
ans =
2.2878e-13
```

Example: Ovarian Cancer Data

The covarian cuerer data set, which is had into Mathia provides a more realistic sample to infranze the bowers of PCA. This cauging bounds of grout data for glotness, 121 data covariant of the provided set of the provided set of the provided set of the data covariant glotness of the data for the provided set of the provided

More importantly, patients with ovarian cancer appear to chatter separately from patients without cancer when pointed in the space spanned by the first three PCA modes. This is shown in Fig. 1.15, which is generated by Code 11. This is induced chatter comparison of data by cancepts a is domaintoined decent of machine teaming and pattern tercognition. For example, we will see in Sec. 16 that images of different human faces will form Contents. The sec of these classes will be explored in genetate dual in Contents.



Figure 1.15 Clustering of samples that are normal and those that have cancer in the first three mincinal commences coordinates.

Code 1.11 Compute PCA for ovarian cancer data.

```
land covariancemers; # Load ovarian cancer data
(p.5, v) = avd(b, *cont);
for isinite(ba, 1)
x = V(1,2)*col(1,1)*;
y = V(1,2)*col(1,1)*;
y = b(1,2)*col(1,1)*;
if(pri()=c-concer)
plot(x,y,z,*rx*,`Limstich*,2);
alse
plot(x,y,z,*b*,`Limstich*,2);
and
mat mat
```

1.6 Eigenfaces Example

One of the most striking demonstration of SVDPCA is the so-called cigrathese cample final hange both PCA is SVD to mean-solution between large. The result of the fixed magnetic strength is most domain contribution between large. The result of the the probability of the solution of the solution of the solution of the principal components. It will be down in Chapter 5 that images of the nume person tend to chapter in the eigenberg security making the solution immediated by Sizowick and Kings control (10). B) The eigenberg problem was from standed by Sizowick and Kings components (10). B) The eigenberg problem was from standed by Sizowick and Kings components (10). B) The eigenberg problem was from standed by Sizowick and Kings was researched The Ander Strain In (17) That.

Here, we demonstrate this algorithm using the Extended Yale Face Database B [203], consisting of cropped and aligned images [327] of 38 individuals (28 from the extended dutabase, and 10 from the original database) under 9 poses and 64 lighting conditions⁶.

6 The database can be downloaded at http://vision.ucsd.edu/-iskwak/ExtYaleDatabase/ExtYaleB.html.



Figure 1.16 (left) A single image for each person in the Yale database, and (right) all images for a specific person. Left runel remerated by Code (1.12).

Each image is 192 pixels tail and 108 pixels wide. Utilize the pervious image example in Section 1.2, each of the facial images in our library have been reshaped into a large column vector with 192 × 108 = 32, 256 dements. We use the first 36 people in the database (feb panel of Fig. 11.61) on our training data for the eigenfaces cample, and we hold back two people as a test set. An example of all 64 images of one specific person are shown in the right panel. These images are loaded and plotend using Code 1.12.

Code 1.12 Plot an image for each person in the Yale database (Fig. 1.16 (a))

```
lad ...RXX/direct.att
later to the second seco
```

As metrioned before, each image is reshaped into a large column vector, and the average face is computed and subtracted from each column vector. The mean-subbracted image vectors are then stacked horizontally as columns in the data matrix **X**, as shown in Fig. 1.17. Thus, taking the SVD of the mean-subbracted matrix **X** results in the PCA. The columns of U are the eigenfaces, and they may be reshaped back into 192 × 168 images. This is illustrated in Code 1.13.



Figure 1.17 Schematic procedure to obtain eigenfaces from library of faces.

Code 1.13 Compute eigenfaces on mean-subtracted data.

```
B We use the first is people for training data
trainingTones face(; Limm(faces(1)40));
wrgTone = mamm(trainingTones(2)) & bits over by 1;
b Compte eigenfaces on max-motherated training data
X = trainingTones-wrgTonescones(Latin (trainingTones(2));
(D,S,V] = wd(X, = emot);
images(rembage(wrgTones,n,m)) > Pich say face
images(rembage(wrgTones,n,m)) > Pich size i domface
```

Using the eigenface library, \dot{U}_i obtained by this code, we now attempt to approximately represent an image that was not in the training data. At the beginning, we held back two individuals (the 37th and 38th people), and we now use one of their images as a test image, x_{auc} . We will see how well a rank- σ SVD basis will approximate this image using the following projection:

$$\tilde{x}_{text} = \tilde{U}\tilde{U}^*x_{text}$$
.



Figure 1.18 Approximate representation of test image using eigenfaces basis of various order r. Test image is not in training set.

The eigenface approximation for various values of r is shown in Fig. 1.18, as computed using Code 1.14. The approximation is relatively poor for $r \le 200$, although for r > 400it converses to a nassable representation of the test image.

It is interesting to note that the eigenface space is not only useful for representing human faces, but may also be used to approximate a dog (Fig. 1.19) or a cappoccino (Fig. 1.20). This is possible because the 1600 eigenfaces span a large subspace of the 3225 dimensional image space corresponding to bread, smooth, nonlocalized spatial features, such as check; horebach, monthy, etc.

Cade 1.14 Approximate test-image that was omitted from training data.

```
testFaceMS = testFace - avgFace;
for re(25 50 100 200 400 800 160]
reconface = avgFace + (U(1,1:r)*(U(1,1:r)*testFaceMS));
imagesc(reshape(reconFace,n,m))
ed
```

We further investigate the use of the eigenfaces is a coordinate system, defining an eigenface space. Its projecting an image are one for at r < R modes, use obtain a set of coordinates in this space: K = U'. Some principal comparements may expurse the most mouse neutral for distinguishing between hardwards. Additional principal components with the more model for distinguishing between hardwards. Additional principal components with the principal component of the principal component with the component of the principal component of the principal component. The principal of the initial component of the principal component of the principal component of the principal component of the principal component, generated by Code 115 in the basis for many encountion and Control component of the principal component of the princ



Test image r = 25r = 50r = 100r = 200r = 1600r = 400r = 800

Figure 1.20 Approximate representation of a carpraccino using eigenfaces.



Figure 1.21 Projection of all images from two individuals onto the 5th and 6th PCA modes. Projected images of the first individual are indicated with black diamonth, and projected images of the second individual are indicated with red triangles. Three examples from each individual are circled in blac, and the corresponding image is shown.

Code 1.15 Project images for two specific people onto the 5th and 6th eigenfaces to illustrate the potential for automated classification.

```
Figure 2, 1 = Dyram makes 2

Figure 2, 1 = Dyram makes 2

Figure 2, 2 = Dy
```

1.7 Truncation and Alignment

Deciding how many singular values to keep, i.e. where to truncate, is one of the most important and controls decisions where using the SVD. There are many factors, heading specifications on the distinct rank of the system, the magnitude of noise, and the distribution of the singular values. Offore, one truncauses the SVD at a rank *r* that captures a distribution of the singular values. Offore, one truncauses the SVD at a rank *r* that captures a truncation, Although crude, this technique is commonly used. Other techniques involve distribution, which were one "sace" in the traingur value distribution, which may denote the transition from singular values that represent important patterns from those that represent motion. Transition may be viewed as a fund threshold on singular values, where muraned. Research larger than a threshold *r* are kept, while remaining singular values are runnated. Received, work by Gavish and Donolo [200] provides an optimal transition values, or and threshold under certain conditions, providing a principled approach to obtaining low-rank matrix approximations using the SVD.

In addition, the alignment of data significantly impacts the rank of the SVD approximation. The SVD essentially relies on a separation of variables between the columns and rows of a data matrix. In many situations, such as when analyzing traveling waves or misaligned data, this assumption breaks down, resulting in an artificial rank inflation.

Optimal Hard Threshold

A recent theoretical breakthrough determines the optimal hard threshold r for singular value truncation under the assumption that a matrix has a low-rask structure commanized with Gaussian white noise [200]. This work builds on a significant literature surrounding valuous techniques for hard and soft thresholding of singular values. In this section, we summarize the main results and demonstrate the thresholding on various examples. For more details, see [200].

First, we assume that the data matrix X is the sum of an underlying low-rank, or approximately low-rank, matrix X_{mut} and a noise matrix X_{noise}:

$$X = X_{true} + \gamma X_{noise}. \qquad (1.30)$$

The entries of X_{nclos} are assumed to be independent, identically distributed (i.i.d.) Gaussian random variables with zero mean and unit variance. The magnitude of the noise is characterized by y, which deviates from the notation in [200]?.

When the noise magnitude y is known, there are closed-form solutions for the optimal hard threshold r:

1. If $X \in \mathbb{R}^{n \times n}$ is square, then

$$r = (4/\sqrt{3})\sqrt{n\gamma}$$
. (1.31)

 If X ∈ R^{n×m} is rectangular and m ≪ n, then the constant 4/√3 is replaced by a function of the aspect ratio β = m/n:

$$\tau = \lambda(\beta)\sqrt{n\gamma}$$
, (1.32)

$$h(\beta) = \left(2(\beta + 1) + \frac{8\beta}{(\beta + 1) + (\beta^2 + 14\beta + 1)^{1/2}}\right)^{1/2}$$
. (1.33)

Note that this expression reduces to (1.31) when $\beta = 1$. If $n \ll m$, then $\beta = n/m$.

When the noise magnitude γ is unknown, which is more typical in real-world applications, then it is possible to estimate the noise magnitude and scale the distribution of singular values by using σ_{mash} , the worldaw singular value. In this case, there is no closedform solution for r, and it must be approximated numerically.

⁷ In [200], σ is used to denote standard deviation and γ_k denotes the k^{th} singular value.

For unknown noise γ, and a rectangular matrix X ∈ ℝ^{w×w}, the optimal hard threshold is given by

$$r = \omega(\beta)\sigma_{mel}$$
 (1.34)

Here, $\omega(\beta) = \lambda(\beta)/\mu_{\beta}$, where μ_{β} is the solution to the following problem:

$$\int_{(1-\beta)^2}^{\mu_{\beta}} \frac{\left[\left((1+\sqrt{\beta})^2 - t\right)\left(t - (1-\sqrt{\beta})^2\right)\right]^{1/2}}{2\pi t} dt = \frac{1}{2}.$$

Solutions to the expression above must be approximated numerically. Fortunately [200] has a Matlab code supplement⁸ [151] to approximate u.s.

The new method of optimal hard thresholding works remarkably well, as demonstrated on the examples below.

Example 1: Toy Problem

In the first example, shown in Fig. 1.22, we artificially construct a mats 2 matrix (Code 1.16) and we contaminate the signal with Gaussian white noise (Code 1.17). A de-noised and dimensionally reduced matrix is then obtained using the threshold from (1.31) (Code 1.13), as well as using a 90% energy transcristic (Code 1.19). A discussion of the singular values (Code 1.20) in Fig. 1.23, it is clear that there are two values that are above threshold from the transcription of the singular values (Code 1.20) in Fig. 1.23, it is clear that there are two values that are above threshold.

Cade 1.16 Compute the underlying low-rank signal. (Fig. 1.22 (a))

```
clear all, close all, clc
t = (-1:.01:3)';
Utrus = [cos(17+t).exp(-t.^2) sin(11+t)];
Strus = [ 2 (0 . 5];
Vtrus = [sin(5+t).exp(-t.^2) cos(13+t)];
X = Utrus,Strus,Verus';
figures, inshow(X);
```

Code 1.17 Contaminate the signal with noise. (Fig. 1.22 (b))

```
sigma = 1;
Xnoisy = X+sigma_randn(size(X));
figure, imshow(Xnoisy);
```

Code 1.18 Truncate using optimal hard threshold. (Fig. 1.22 (c))

8 http://parl.stanford.edu/vg705qn9070



Figure 1.22 Underlying rank 2 matrix (a), matrix with noise (b), clean matrix after optimal hard threshold $(4/\sqrt{3})\sqrt{n\sigma}$ (c), and truncation based on 90% energy (d).

Code 1.19 Truncate using 90% energy criterion. (Fig. 1.22 (d))

```
rdS = cumsum(diag(S))./sum(diag(S));  # Oursulative energy
r90 = min(find(cdS>0.90));  # Find r to capture 90% energy
X90 = U(:,l:r90)*S(1:r90,l:r90)*V(:,l:r90)';
figure, imshow(X90)
```

Code 1.20 Plot simular values for hard threshold example. (Fig. 1.23)

```
semilogy(diag(S),'-ok','LineWidth',1.5), hold on, grid on
semilogy(diag(S(1:r,1:r)),'or','LineWidth',1.5)
```

Example 2: Eigenfaces

In the second example, we revixit the eigenfaces problem from Section 1.6. This provides a new typical example, since the data matrix X is rectanging, with aspect ratio $\beta = 3/4$, and the noise magnitude is unknown. It is also not clear that the data is contaminated with white noise. Nonetheless, the method determines a threshold r, above which columns of U appear to have strong facial features, and below which columns of U consist mostly of noise shown in Fig. 1.24.



Figure 1.23 Singular values σ_t (a) and cumulative energy in first r modes (b). The optimal hard threshold $\tau = (4/\sqrt{5})\sqrt{n}\sigma$ is shown as a red dashed line (--), and the 90% cutoff is shown as a blue dashed line (--). For this case, n = 600 and $\sigma = 1$ so that the optimal cutoff is approximately $\tau = 56.6$.



Figure 1.24 Hard thresholding for circufaces example.

Importance of Data Alignment

Here, we discuss common pitfalls of the SVD associated with minal/pield data. The 64biomic scanaple is designed to librarize one of the certar Warkenses of the SVD data dimensionally reduction and otherest feature extraction in data. Consider a matrix of zeros with a restrugative at block constaint of orders. As a intrage, this would look like a utility restruction of the structure of the structure of the structure of the structure of the restructure of the structure of the structure of the structure of the structure single with the z-a of z-ass of the flucture, then the SVD is simple, howing only one sourcero implative value σ_i (see Fig. 1.25 (ci)) and corresponding singular vectors \mathbf{u}_i and \mathbf{v}_i the define the width and beingt of the white restands.



Figure 125 A data matrix consisting of ones with a square sub-block of zeros (a), and its SVD spectrum (c). If we rotate the image by 10°, as in (b), the SVD spectrum becomes significantly more complex (d).

When we begin to rotate the inner rectangle so that it is no longer aligned with the image axes, additional non-zero singular values begin to appear in the spectrum (see Figs. 1.25 (bd) and 1.26).

Code 1.21 Compute the SVD for a well-aligned and rotated square (Fig. 1.25).

```
[n = 1000; 1000 = 2000 papers

0 = 1000; 1000 = 2000 papers

0 = 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1000; 1
```



Figure 1.28 A data matrix consisting of zeros with a square sub-block of ones at various rotations (a), and the corresponding SVD spectrum, diao(S), (b).

The reaso that this cample breaks down is that the SVD is fundamentally geometry, meaning that it depends on the coordinate yours in which the data is represented. As we have seen earlier, the SVD is only generically invariant to unitry transformations, meaning that the transformation preserves the inter product. That fars may be viewed as both a neurophic and a weakness of the method. First, the dependence of SVD to the inner duration is neuronal to the avoines media generation. However, the SVD has the distribution of the strain tender generation. However, the SVD has of the data in fact, the SVD rate edgeds when objects in the columns transfils rotation of the data in fact, the SVD rate edgeds when objects in the columns transfils rotation.

For instance, the eigenfaces example was built on a library of images that had been meticulously cropped, centered, and aligned according to a stencil. Without taking these important pre-processing steps, the features and clustering performance would be underwhelmine.

The insbilly of the SVD to coprare translations and contains of the data is a major limiton. For example, the SVD is still membed of choice for the low-stark accomposition of data from partial differential equations (PDEia), as will be explored in Chapters 11 and 12. However, the SVD is fundamentally a data-drene sequenciane of variables, which we Generalized decompositions that retain the fravolable properties and are applicable to data with symmetries is a simplicable to data with symmetries is a simplicable to data with symmetries is a simplicable to data with symmetries in the simplicable to data with symmetries in the simplicable to data w

Cade 1.22 SVD for a square rotated through various angles (Fig. 1.26).

```
bhogies +2; + seep through 12 angles, from 0:4:44
fort +3;
for +3;
for -4;
for +2;
for -4;
for -4
```

subplot(1,2,1), imagesc(Xrot), colormap([0 0 0; cm])
subplot(1,2,2), semilogy(disg(5),'-o','color',cm(j;;))

1.8 Randomized Singular Value Decomposition

The accurate and efficient decomposition of large data nucleos is one of the correspondtone many energieting the section and many and the section of the control of the section of the data does not decrease approximate, one of the the section of the section of the data does not decrease approximate, or the section of the data does not decrease approximate, we can also the section of the data does not decrease approximate, we can also the section of t

Randomized Linear Algebra

Randomized linear algebra is a much more general concept than the treatment presented here for the SVD. In addition to the randomized SVD [164, 377]. In anominer algorithms have been developed for principal component analysis [154, 229], the pivoted LU decomposition [183], the pivoted QP decomposition [162], and the dynamic mode decomposition [173]. Most randomized matrix decompositions can be broken into a for common steps, as described here. There are also several exactivity arrays can be the pitclip 324. 223, 334, 177]. We assume that we are working with tall-kainy matrices, so that n > m, atlongth the hory result) several to a transfer.

- Step 0: Identify a target rank, r < m.
- Step 1: Using random projections P to sample the column space, find a matrix Q whose columns approximate the column space of X, i.e., so that X ~ QQ*X.
- Step 2: Project X onto the Q subspace, Y = Q*X, and compute the matrix decomposition on Y.
- Step 3: Reconstruct high dimensional modes U = QU_Y using Q and the modes computed from Y.

Randomized SVD Algorithm

Over the past two decades, there have been several randomized algorithms proposed to compute a low-rank SVD, including the *Monte Carlo* SVD [190] and more robust approaches based on random projections [464, 335, 371]. These methods were improved by incorporating structured sampling matrices for faster matrix multiplications [559]. Here, we use the randomized SVD algorithm of Halko, Martinsson, and Topp [232].

37

which combined and expanded on these previous algorithms, providing favorable error bounds. Additional analysis and numerical implementation details are found in Voronin and Martinsson [544]. A schematic of the rSVD algorithm is shown in Fig. 1.27.

Step 1: We construct a random projection matrix $\mathbf{P} \in \mathbb{R}^{m \times r}$ to sample the column space of $\mathbf{X} \in \mathbb{R}^{n \times m}$:

$$Z = XP$$
. (1.35)

The matrix **Z** may be much smaller than **X**, especially for low-rank matrices with $v \ll m$. It is highly unlikely that a random projection matrix **P** will project out important components of **X**, and so **Z** approximates the column space of **X** with high probability. Thus, it is possible to compute the low-rank QR decomposition of **Z** to obtain an orthonormal basis for **X**:

$$Z = QR.$$
 (1.36)

Step 1







Figure 1.27 Schematic of randomized SVD algorithm. The high-dimensional data X is depicted in red, intermediate steps in gray, and the outputs in blue. This algorithm requires two passes over X.

Step 2: With the low-rank basis Q, we may project X into a smaller space:

$$Y = Q^* X.$$
 (1.37)

It also follows that $X \approx QY$, with better agreement when the singular values σ_k decay rapidly for k > r.

It is now possible to compute the singular value decomposition on Y:

$$Y = U_Y \Sigma V^*$$
. (1.38)

Because Q is a orthonormal and approximates the column space of X, the matrices Σ and V are the same for Y and X, as discussed in Section 1.3.

Step 3: Finally, it is possible to reconstruct the high-dimensional left singular vectors U using Uv and O:

$$U = QU_{Y}$$
. (1.39)

Oversampling

Most matrices X do not have an exact low-rank structure, given by r modes. Instead, there are notizeo singular values n_0 for k-r and the sketch V will not exactly point in column space of X. In general, increasing the number of columns in P from ro r + p, significantly improves results, crease with p adding around $5 \circ rol columns [370]$. This is known as oversampling, and increasing p decreases the variance of the singular value spectrum of the sketched matrix.

Power Iterations

A second challenge in using randomized algorithms is when the singular value spectrum decays slowly, so that the remaining truncated singular values contain significant variance in the data X. In this case, it is possible to preprocess X through q power iterations [454, 228, 224] to create a new matrix X⁽⁴⁾ with a more rapid singular value decay:

$$X^{(q)} = (XX^*)^q X.$$
 (1.40)

Power iterations dramatically improve the quality of the randomized decomposition, as the singular value spectrum of X^(q) decays more rapidly:

$$X^{(q)} = U\Sigma^{2q-1}V^*$$
(1.41)

However, power iterations are expensive, requiring q additional passes through the data X. In some extreme examples, the data in X may be stored in a distributed architecture, so that every additional pass adds considerable expense.

Guaranteed Error Bounds

One of the most important properties of the randomized SVD is the existence of tunable error bounds, that are explicit functions of the singular value spectrum, the desired rank r, the oversampling parameter p and the number of power iterations q. The best attainable error bound for a deterministic absorbitm is:

$$\|X - QY\|_2 \ge \sigma_{r+1}(X).$$
 (1.42)

In other words, the approximation with the best possible rank-r subspace Q will have error greater than or equal to the next truncated singular value of X. For randomized methods, it is possible to bound the expectation of the error:

$$\mathbb{E}(\|\mathbf{X} - \mathbf{Q}\mathbf{Y}\|_2) \le \left(1 + \sqrt{\frac{r}{p-1}} + \frac{e\sqrt{r+p}}{p}\sqrt{m-r}\right)^{\frac{1}{2q+1}} \sigma_{k+1}(\mathbf{X}),$$
 (1.43)

where e is Euler's number.

Choice of random matrix P

There are several suitable choices of the random matrix P. Gaussian random projections (e.g., the elements of P are i.i.d. Gaussian random variables) are frequently used because of favorable mathematical properties and the richness of information extracted in the sketch Z. In particular, it is very unlikely that a Gaussian random matrix P will be chosen hadly so as to project out important information in X. However, Gaussian projections are expensive to generate, store, and compute. Uniform random matrices are also frequently used, and have similar limitations. There are several alternatives, such as Rademacher matrices, where the entries can be +1 or -1 with equal probability [532]. Structured random projection matrices may provide efficient sketches, reducing computational costs to O(nmlog(r)) [559]. Yet another choice is a sparse projection matrix P which improves storage and computation, but at the cost of including less information in the sketch. In the extreme case, when even a single pass over the matrix X is prohibitively expensive, the matrix P may be chosen as random columns of the m × m identity matrix, so that it randomly selects columns of X for the sketch Z. This is the fastest option, but should be used with caution, as information may be lost if the structure of X is highly localized in a subset of columns, which may be lost by column sampling.

Example of Randomized SVD

To demonstrate the randomized SVD algorithm, we will decompose a high-resolution image. This particular implementation is only for illustrative purposes, as it has not been optimized for speed, data transfer, or accuracy. In practical applications, care should be taken [228, 177].

Code 1.23 computes the randomized SVD of a matrix X, and Code 1.24 uses this function to obtain a rank-400 approximation to a high-resolution image, shown in Fig. 1.28.

Code 1.23 Randomized SVD algorithm.



Figure 128 Original high-resolution (left) and rank-400 approximations from the SVD (middle) and rSVD (right).

```
[UY,S,V] = mvd(Y,'econ');
U = Q_UY;
```

Cade 1.24 Compute the randomized SVD of high-resolution image.

1.9 Tensor Decompositions and N-Way Data Arrays

Low-rank decompositions can be generalized beyond matrices. This is important as the SVD requires that diparate types of data the futures (in an single vector in order to evaluate correlated structures. For instance, different time simplosts (columns) of a matrix may include measurements in diverse as trajectorizen pressure, concentration of a matrix may make seme. Ultimately, what is desired in to preserve the various data structures and press in deter own, independent directions. Matrices can be generalized to N-way arrays, or in their own. Independent directions. Matrices can be generalized to N-way arrays, or



Figure 1.28 Comparison of the SVD and Ternor decomposition frameworks. Both methods produce an approximation to the original data matrix by sums of outer products. Specifically, the tensor decomposition generalizes the concept of the SVD to N-way arrays of data without having to flatten (vectorize) the data.

tensors, where the data is more appropriately arranged without forcing a data-flattening process.

The construction of data tensors requires that we revisit the notation associated with tensor addition, multiplication, and inner products [299]. We denote the ruh column of a matrix A by u. Given matrices A $\in \mathbb{R}^{d\times K}$ and B $\in \mathbb{R}^{d\times K}$, their Khatri-Rao product is denoted by A \odot B and is defined to be the $H \times K$ matrix of column-wise Kronecker products, namely

$$A \odot B = (a_1 \otimes b_1 \cdots a_K \otimes b_K).$$

For an N-way tensor A of size $I_1 \times I_2 \times \cdots \times I_N$, we denote its $i = (i_1, i_2, \dots, i_N)$ entry by a_i .

The inner product between two N-way tensors A and B of compatible dimensions is given by

$$\langle \mathcal{A}, \mathcal{B} \rangle = \sum_{i} a_{i}b_{i}.$$

The Frobenius norm of a tensor A, denoted by $\|A\|_{E}$, is the square root of the inner product of A with itself, namely $\|A\|_{E} = \sqrt{|A, A|}$. Finally, the mode-n matricization or unfolding of a tensor A is denoted by $mA_{h(n)}$.

Let M represent an N-way data tensor of size $I_1 \times I_2 \times \cdots \times I_N$. We are interested in an R-component CANDECOMP/PARAFAC (CP) [124, 235, 299] factor model

$$\mathbf{M} = \sum_{r=1}^{R} \lambda_r \mathbf{m} \mathbf{a}_r^{(1)} \circ \cdots \circ \mathbf{m} \mathbf{a}_r^{(N)}, \qquad (1.44)$$



Figure 120 Example N-way series data set created from the function (1.45). The data matrix is $\mathbf{A} \in \mathbb{R}^{121 \times 101 \times 105}$, A CP tensor decomposition can be used to extract the two underlying structures that produced the data.

where represents our product and $\mathbf{m}_{i}^{(2)}$ represents the rh column of the factors matrix $\mathbf{m}_{i}^{(2)}$ of size $I_{i} \times R$. The CP decomposition refers to CANDECOMPPARAFAC which stand for parallel factors analysin (PRAFAC) and comovid decomposition (CANDE) COMP respectively. We refer to each summand as a component. Assuming each factor matrix has been constrained to have unit Encidents neighbor verifies to the λ_{i} sector that λ_{i} as wrights. We will use the shorthand notation where $\lambda = (\lambda_{i}, \ldots, \lambda_{d})^{2}$ [25] A tensor that has CP decomposition is simelinesr efference to as a favoidal strends.

For the rest of this chapter, we consider a 3-way CP tensor decomposition (See Fig. 1.29) where two modes index state variation and the third mode indexes time variation:

$$\mathcal{M} = \sum_{r=1}^{R} \lambda_r \mathbf{A}_r \circ \mathbf{B}_r \circ \mathbf{C}_r$$

Let $\mathbf{A} \in \mathbb{R}^{I_2 \times R}$ and $\mathbf{B} \in \mathbb{R}^{I_2 \times R}$ denote the factor matrices corresponding to the two state modes and $\mathbf{C} \in \mathbb{R}^{I_2 \times R}$ denote the factor matrix corresponding to the time mode. This 3-way decomposition is compared to the SVD in Fig. 1.29.

To illustrate the tensor decomposition, we use the MATLAB N-way toolbox developed by Rasmus Bro and coworkers [84, 15] which is available on the Mathworks file exchange. This simple to use package provides a variety of tools to extract tensor decompositions and evaluate the factor models generated. In the specific example considered here, we generate duat from a spatic-stemporal function (See Fig. 10)

$$F(x, y, t) = \exp(-x^2 - 0.5y^2) \cos(2t) + \operatorname{sech}(x) \tanh(x) \exp(-0.2y^2) \sin(t).$$
 (1.45)



Figure 13: 3-way tensor decomposition of the function (1.45) discretized so that the data matrix is $A \in \mathbb{R}^{121\times101\times313}$. A CP tensor decomposition can be used to extract the two underlying structures that produced the data. The first factors is in blue, the second factors is real. The three distinct directions of the data (parallel factors) are illustrated in (a) the y direction, (b) the x direction, and (c) the time t.

This model has two spatial modes with two distinct temporal frequencies, thus a two factor model should be sufficient to extract the underlying spatial and temporal modes. To construct this function in MATLAB the following code is used.

Code 1.25 Creating tensor data

```
x=-5:0.1:5; y=-6:0.1:6; t=0:0.1:10*pi;
[X,Y,T] mmashgrid(x,y,t);
Amarp(-(X,^2+0.5*Y,^2)).*(cos(2*T))+...
(sech(X).*tanh(X).*exp(-0.2*Y,^2)).*sin(T);
```

Note that the **meshgrid** command is capable of generating N-way arrays. Indeed, MAT-LAB has no difficulties specifying higher-dimensional arrays and tensors. Specifically, one can easily generate N-way data matrices with arbitrary dimensions. The command A = randul(10, 10, 10, 10, 10) generates a 5-way hypercube with random values in each of the five directions of the array.

Figure 130 shows eight snapshots of the function (1.45) discretized with the code above. The N-way array data generated from the MATLAB code produces $\mathbf{A} \in \mathbb{R}^{[21+101\times33]}$, which is of total dimension 10°. The CP tensor decomposition can be used to extract a two factor model for this 3-way array, thus producing two vectors in each direction of space x, space y, and time t.

The N-way toolbox provides a simple architecture for performing tensor decompositions. The PARAFAC command structure can easily take the input function (1.45) which is discretized in the code above and provide a two-factor model. The following code produces the output as model.

```
Code 1.28 Two factor tensor model.
```

```
modelsparafac(A,2);
[A1,A2,A3]=fac2let(model);
subplot(3,1,2), plot(y,A1,'Linewidth',[2])
subplot(3,1,2), plot(x,A2,'Linewidth',[2])
subplot(3,1,3), plot(x,A3,'Linewidth',[2])
```

Note that in this code, the **fac2let** command turns the factors in the model into their component matrices. Further note that the **meshgrid** arrangement of the data is different from **parafac** since the x and y directions are switched.

Figure 13.1 shows the results of the N-say trans decomposition for the prescribed two three models. Specification, the two volces and and on the three discretions of the array factor models. Specification, the two volces and and the network of the array from the much 2 model (1.45). The first sort of a two nedes (indeg the original) denotion include a Gaussian proceedings. The scored set of two models (indeg the original) denotion include a Gaussian for the first function, and the anti-symmetries scot(s) (much) (finde functions of the structure). The score of two the origin of the two functions (or (2)) and ultit), respectively. Thus, the two factor model produced by highdemension distant much score of the two match much much moder of the score dimension distant much score of the score functions. The two functions can be a dimensioned that much score of the score function of the two dimension distant much score of the score functions of the two dimension distant much score of the score function of the two dimensioned dimensioned the high-

Recert thereards and comparison all solutions in N-way decompositions are opening in the period information of period information of the N ange, such decompositions in my discreptional period in the comparison of the comparison

Suggested Reading

Texts

(1) Matrix computations, by G. H. Golub and C. F. Van Loan, 2012 [214].

Papers and reviews

- (I) Calculating the singular values and pseudo-inverse of a matrix, by G. H. Golub and W. Kahan, Joannal of the Society for Industrial & Applied Mathematics, Series B: Numerical Analysis, 1965 1221.
- (2) A low-dimensional procedure for the characterization of human faces, by L. Sizovich and M. Kirby, *Journal of the Optical Society of America A*, 1987 [491].

45

- (3) Finding structure with randomness: Probabilistic algorithms for constructing approximate matrix decompositions, by N. Halko, P.-G. Martinsson, and J. A. Tropp, SIAM Review, 2011 [230].
- (4) A randomized algorithm for the decomposition of matrices, by P.-G. Martinsson, V. Rohlin, and M. Tygert, Applied and Computational Harmonic Analysis, 2011 [371].
- (5) The optimal hard threshold for singular values is 4/√3, by M. Gavish and D. L. Denoho, IEEE Transactions on Information Theory, 2014 [200].

A count occurs of amheritaria physics and engineering mathematics involves the marmetanian of equations in an earlyst. This is a common these throughout this body are member to computation and analyst. This is a common these throughout this body is a second to the engineering of the conduct systems by constraining and an discretulativity. Perhaps the next found and subplotuses conduct materian terms and intervention of the engineering of the englishing of the engineering (the engineering of the engineering of the engineering of the advancement of the engineering of the engineering of the engineering of the engineering of the programment of the engineering of the engineering of the engineering of the programment of the engineering of the engineering of the engineering of the programment of the engineering of the engineering of the engineering of the generation of the engineering of the engineering of the engineering of the generation of the engineering of the generation of the engineering of the engineering of the generation of the generation of the engineering of the generation of the gener

Fourier's semial work provided the nuthematical foundation for Hibber spaces, opertant theory, approximation theory, and the subsequent revolution is analysical and compartational mathematics. Fast forward two handled years, and the fast Fourier transform has become the correstrost of comparational mathematics, cannille graval-thic image and and compression, global communication networks, modern devices and hardware, numerical applysis and engineering at scale, and advanced data. analysis. Simply, rath. be far Spectru transform has had a more significant and profound rede in shaping the modern world than any other algorithms to date.

With increasingly complex problems, data sets, and computational geometrics, simple Fourier sine and concise house have given way to addree blocks, such as the data driven SVD. In fact, the SVD basis can be used as a direct analogue of the Fourier basis for adving-PBs with complex geometrics, as will be discussed later. In addition, related functions, called wavelets, have been developed for advanced signal processing and compression called wavelets, have been developed for advanced signal processing and compression wavelets, have been developed for advanced signal processing and compression strandorms, we will demonstrate a for of the many use of Fourier and wavelet transforms.

2.1 Fourier Series and Fourier Transforms

Before describing the computational implementation of Fourier transforms on vectors of duth, here we infractors the malytic Fourier series and Fourier transform, defined for continuous functions. Naturally, the discrete and continuous formulations about much may find of dut with infinitely that resolutions. The Fourier series and transform are aritminizely generative the rotion of vector spaces to include functions with infinitely many degrees of freedom. Thus, we begin with an introduction to functions with infinitely many degrees of freedom. Thus, we begin with an introduction to functions vectors.



Figure 2.1 Discretized functions used to illustrate the inner product.

Inner Products of Functions and Vectors

In this section, we will make use of inner products and norms of functions. In particular, we will use the common Hermitain inner product for functions f(x) and g(x) defined for x on a domain $x \in [a, b]$:

$$(f(x), g(x)) = \int_{a}^{b} f(x)\tilde{g}(x) dx$$
 (2.1)

where g denotes the complex conjugate.

The inner product of functions may seem strange or unmotivated at first, but this definition becomes clear when we consider the inner product of vectors of data. In pairticular, if we discretize the functions $f_1(x)$ and $g_1(x)$ into vectors of data. In a sin Fig. 21, we would like vectorin inter product to converge to the function inter product as the sampling resolution is increased. The inner product of the data vectors $\mathbf{I} = \begin{bmatrix} f_1 & f_2 & \cdots & f_n \end{bmatrix}$ and $\mathbf{g} = \begin{bmatrix} g_1 & g_2 & \cdots & g_n \end{bmatrix}$ is defined by:

$$\langle \mathbf{f}, \mathbf{g} \rangle = \mathbf{g}^{*} \mathbf{f} = \sum_{k=1}^{n} f_{k} \bar{g}_{k} = \sum_{k=1}^{n} f(x_{k}) \bar{g}(x_{k}).$$
 (2.2)

The magnitude of this inner product will grow as more data points are added; i.e., as n increases. Thus, we may normalize by $\Delta x = (b - a)/(n - 1)$:

$$\frac{b-a}{n-1}(\mathbf{f}, \mathbf{g}) = \sum_{k=1}^{n} f(x_k)\bar{g}(x_k)\Delta x,$$
 (2.3)

which is the Riemann approximation to the continuous function inner product. It is now clear that as we take the limit of $n \rightarrow \infty$ (i.e., infinite data resolution, with $\Delta x \rightarrow 0$), the vector inner product converges to the inner product of functions in (2.1). This inner product also induces a norm on functions, given by

$$||f||_2 = ((f, f))^{1/2} = \sqrt{(f, f)} = \left(\int_a^b f(x)\hat{f}(x) dx\right)^{1/2}$$
. (2.4)

The set of all functions with bounded norm define the set of square integrable functions, detored by $L^2([0, k])$, this is also known as the set of Lebegue imegrable functions. The interval [0, k] may also be chosen to be infinite $(e_2, -(\infty, \infty))$, semi-finitine $(e_3, -(\infty, \infty))$, estical $(e_3, -(\infty, \infty))$, semi-finitine $(e_3, -(\infty, \infty))$, semi-finitine $(e_3, -(\infty, \infty))$, semi-finitine $(e_3, -(\infty, \infty))$, semi-finite $(e_3,$

As in finite-dimensional vector spaces, the inner product may be used to project a function into an new coordinate system defined by a basis of orthogonal functions. A Fourier series reserventation of a function *f* is precisely a projection of this function notes the orthogonal set of sine and cosine functions with integer period on the domain [a, b]. This is the subject of the following sections.

Fourier Series

A fundamental result in Fourier analysis is that if f(x) is periodic and piecewise smooth, then it can be written in terms of a Fourier series, which is an infinite sum of cosines and sines of increasing frequence. In particular, if f(x) is 2π -periodic, it may be written as:

$$f(x) = \frac{a_0}{2} + \sum_{k=1}^{\infty} (a_k \cos(kx) + b_k \sin(kx)). \quad (2.5)$$

The coefficients ag and bg are given by

$$a_k = \frac{1}{\pi} \int_{-\pi}^{\pi} f(x) \cos(kx) dx$$
 (2.6a)

$$b_k = \frac{1}{\pi} \int_{-\pi}^{\pi} f(x) \sin(kx) dx,$$
 (2.6b)

which may be viewed as the coordinates obtained by projecting the function onto the orthogonal cosine and sine basis {cos(kx), sin(kx)}^{*}_{ku0}. In other words, the integrals in (2.6) may be re-written in terms of the inner product as:

$$a_k = \frac{1}{\|\cos(kx)\|^2} \langle f(x), \cos(kx) \rangle$$
 (2.7a)

$$b_k = \frac{1}{\|\sin(kx)\|^2} \langle f(x), \sin(kx) \rangle,$$
 (2.7b)

where $\|\cos(kx)\|^2 = \|\sin(kx)\|^2 = \pi$. This factor of $1/\pi$ is easy to verify by numerically integrating $\cos(x)^2$ and $\sin(x)^2$ from $-\pi$ to π .

The Fourier series for an L-periodic function on 10. L) is similarly given by:

$$f(x) = \frac{a_0}{2} + \sum_{k=1}^{\infty} \left(a_k \cos \left(\frac{2\pi kx}{L} \right) + b_k \sin \left(\frac{2\pi kx}{L} \right) \right), \quad (2.8)$$
with coefficients ak and bk given by

$$s_k = \frac{2}{L} \int_0^L f(x) \cos \left(\frac{2\pi kx}{L}\right) dx \qquad (2.9a)$$

$$b_k = \frac{2}{L} \int_0^L f(x) \sin\left(\frac{2\pi kx}{L}\right) dx. \qquad (2.9b)$$

Because we are expanding functions in terms of sine and cosine functions, it is also natural to use Euler's formula $e^{ikx} = \cos(kx) + i \sin(kx)$ to write a Fourier series in complex form with complex coefficients $c_k = a_k + i\beta_k$:

$$\begin{split} f(x) &= \sum_{k=-\infty}^{\infty} \alpha x^{kk} = \sum_{k=-\infty}^{\infty} (a_k + i\beta_k) \left(\cos(kx) + i \sin(kx) \right) \\ &= (a_k + i\beta_k) + \sum_{k=1}^{\infty} \left[(a_{-k} + a_k) \cos(kx) + (\beta_{-k} - \beta_k) \sin(kx) \right] \\ &+ i \sum_{k=1}^{\infty} \left[(\beta_{-k} + \beta_k) \cos(kx) - (\alpha_{-k} - a_k) \sin(kx) \right]. \end{split}$$
(2.10)

If f(x) is real-valued, then $\alpha_{-k} = \alpha_k$ and $\beta_{-k} = -\beta_k$, so that $c_{-k} = \tilde{c}_k$.

Thus, the functions $\psi_k = e^{ikx}$ for $k \in \mathbb{Z}$ (i.e., for integer k) provide a basis for periodic, complex-valued functions on an interval $[0, 2\pi)$. It is simple to see that these functions are orthogonal:

$$\langle \psi_j, \psi_k \rangle = \int_{-\pi}^{\pi} e^{ijx} e^{-ikx} dx = \int_{-\pi}^{\pi} e^{i(j-k)x} dx = \begin{bmatrix} \frac{e^{i(j-k)x}}{i(j-k)} \end{bmatrix}_{-\pi}^{\pi} = \begin{cases} 0 & \text{if } j \neq k \\ 2\pi & \text{if } j = k. \end{cases}$$

So $(\psi_1, \psi_2) = 2\pi \delta_{jk}$, where δ is the Kronecker delta function. Similarly, the functions $e^{i2\pi k_1/L}$ provide a basis for L^2 ([0, L]), the space of square integrable functions defined on $x \in [0, L]$.

In principle, a Fourier series is just a change of coordinates of a function f(x) into an infinite-dimensional orthogonal function space spanned by sines and cosines (i.e., $\psi_R = e^{Rx} = \cos(kx) + i \sin(kx)$):

$$f(x) = \sum_{k=-\infty}^{\infty} c_k \psi_k(x) = \frac{1}{2\pi} \sum_{k=-\infty}^{\infty} \langle f(x), \psi_k(x) \rangle \psi_k(x).$$
 (2.11)

The coefficients are given by $c_1 = \frac{1}{2\pi} (f(x), \psi_1(x))$. The factor of $1/2\pi$ normalizes the projection by the square of the norm of ψ_1 ; $l_c = |\psi_1|^2 = 2\pi$. This is consistent with our standard filtrai-dimensional notesion of change of basis, as in Fig. 2.2. A vector f may be written in the (\bar{x}, \bar{y}) or (\bar{u}, \bar{v}) coordinate systems, via projection onto these orthogonal bases:

$$\vec{f} = (\vec{f}, \vec{x}) \frac{\vec{x}}{\|\vec{x}\|^2} + (\vec{f}, \vec{y}) \frac{\vec{y}}{\|\vec{y}\|^2}$$
(2.12a)

$$= (\vec{f}, \vec{u}) \frac{\vec{u}}{\|\vec{u}\|^2} + (\vec{f}, \vec{v}) \frac{\vec{v}}{\|\vec{v}\|^2}.$$
 (2.12b)



Figure 22 Change of coordinates of a vector in two dimensions.

Example: Fourier Series for a Continuous Hat Function

As a simple example, we demonstrate the use of Fourier series to approximate a continuous hat function, defined from $-\pi$ to π :

$$f(x) = \begin{cases} 0 & \text{for } x \in [-\pi, \pi/2) \\ 1 + 2x/\pi & \text{for } x \in [-\pi/2, 0) \\ 1 - 2x/\pi & \text{for } x \in [0, \pi/2) \\ 0 & \text{for } x \in [\pi/2, \pi). \end{cases}$$
(2.13)

Because this function is even, it may be approximated with cosines alone. The Fourier series for f(x) is shown in Fig. 2.3 for an increasing number of cosines.

Figure 2.4 shows the coefficients a_k of the even cosine functions, along with the approximation error, for an increasing number of modes. The error decreases monotonically, as expected. The coefficients b_k corresponding to the odd sine functions are not shown, as they are identically zero since the hat function is even.

Code 2.1 Fourier series approximation to a hat function.

```
i Define domain
dr = 0.001;
a = 0.101;
a = (-1-decide(1)+i;
a = (-1-decide(1)+i;
a = 1-angth(0); nquart = floor(n/4);
i Define hat function
f(opart:1-engart) = 4.4(1)nquart)=1/An;
f(opart:1-anguart) = 1.4(0)nquart)=1/An;
plot(x, f.'-k', 'LingVith(1', 1.5), hold on
- Commute origin angular
```



Figure 2.3 (top) Hat function and Fourier cosine series approximation for n = 7. (middle) Fourier cosines used to approximate the hat function, and (bottom) zeom in of modes with small amplitude and high frequency.

Example: Fourier Series for a Discontinuous Hat Function

We now consider the discontinuous square hat function, defined on [0, L), shown in Fig. 2.5. The function is given by:

$$f(x) = \begin{cases} 0 & \text{for } x \in [0, L/4) \\ 1 & \text{for } x \in [L/4, 3L/4) \\ 0 & \text{for } x \in [3L/4, L). \end{cases}$$
(2.14)

The truncated Fourier series is plagued by ringing oscillations, known as Gibbs phenomena, around the sharp comers of the step function. This example highlights the challenge of applying the Fourier series to discontinuous functions:



Figure 2.4 Fourier coefficients (top) and relative error of Fourier cosine approximation with true function (bottern) for hat function in Fig. 2.3. The n = 7 approximation is highlighted with a blue circle.



Figure 2.5 Gibbs phenomena is characterized by high-frequency oscillations near discontinuities. The black curve is discontinuous, and the red curve is the Fourier approximation.

```
ht = 0.01; t = 10;
a = 0.04; i;
a = length(x); nguart = floor(n/4);
f(nguart(l=nguart) = 1;
AO = num(f.=news(nise(x)))=dn=2/L;
for = AO/2;
for kaling
b = num(f.=news(nise(x)))=dn=2/L;
b = num(f.=news(nise(x)))=dn=2/L;
b = num(f.=news(nise(x)))=dn=2/L;
```

```
\label{eq:stars} \begin{array}{l} fTS = fTS + \lambda k*cos(2*k*pi*x/L) + Bk*sin(2*k*pi*x/L);\\ end\\ \\ plot(x,f_1)k'_2(timeKidth',2), hold on\\ \\ plot(x,fTS,'t=','LimeKidth',2,2) \end{array}
```

Fourier Transform

The Fourier series is defined for periodic functions, so that conside the domain of definition, the function repeated in iterard in segment the function repeated in the function repeated by the limit of a Fourier ransform iterargal is essentially the limit of a Fourier series as the length of the domain goes to inflativ, which allows us to define a function defined on $(-\infty,\infty)$ with where repeating, as shown in Fig. 2.6. We will consider the Fourier series on a domain $x \in [-L, L)$, and then let $L \to \infty$. On this domain, the Fourier series is

$$f(x) = \frac{a_0}{2} + \sum_{k=1}^{\infty} \left[a_k \cos\left(\frac{k\pi x}{L}\right) + b_k \sin\left(\frac{k\pi x}{L}\right) \right] = \sum_{k=-\infty}^{\infty} c_k e^{ik\pi x/L} \qquad (2.15)$$

with the coefficients given by:

$$c_k = \frac{1}{2L}(f(x), \psi_k) = \frac{1}{2L} \int_{-L}^{L} f(x)e^{-ik\pi x/L} dx.$$
 (2.16)





Figure 2.8 (top) Fourier series is only valid for a function that is periodic on the domain [-L, L). (bottom) The Fourier transform is valid for generic nonperiodic functions.

Restating the previous results, f(x) is now represented by a sum of sines and cosines with a discrete set of frequencies given by $a_0 = \pm \pi/L$. Taking the limit as $L \rightarrow \infty$, these discrete frequencies become a continuous range of frequencies. Define $\omega = k\pi/L$, $\Delta\omega = \pi/L$, and take the limit $L \rightarrow \infty$, so that $\Delta\omega \rightarrow 0$:

$$f(x) = \lim_{\Delta \omega \to 0} \sum_{k=-\infty}^{\infty} \frac{\Delta \omega}{2\pi} \underbrace{\int_{-\pi/\Delta \omega}^{\pi/\Delta \omega} f(\xi) e^{-ik\Delta \omega \xi} d\xi}_{(f(x), \bar{\psi}_{h}(x))} e^{ik\Delta \omega x}.$$
 (2.17)

When we take the linit, the expression $(f(x), \psi_k(x))$ will become the Fourier transform of f(x), denoted by $\hat{f}(\omega) \triangleq \mathcal{F}(f(x))$. In addition, the summation with weight $\Delta \omega$ becomes a Riemann integral, resulting in the following:

$$f(x) = \mathcal{F}^{-1}(\hat{f}(\omega)) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \hat{f}(\omega)e^{i\omega x} d\omega$$
 (2.18a)

$$\hat{f}(\omega) = \mathcal{F}(f(x)) = \int_{-\infty}^{\infty} f(x)e^{-i\omega x} dx.$$
 (2.18b)

These two integrals are known as the *Fourier transform pair*. Both integrals converge as long as $\int_{-\infty}^{\infty} |f(x)| dx < \infty$ and $\int_{-\infty}^{\infty} |\hat{f}(\omega)| d\omega < \infty$; i.e., as long as both functions belong to the space of Lebesgue integrable functions, $f, \hat{f} \in L^1(-\infty, \infty)$.

The Fourier transform is particularly useful because of a number of properties, including linearity, and how derivatives of functions behave in the Fourier transform domain. These properties have been used extensively for data analysis and scientific computing (e.g., to solve PDEs accurately and efficiently), as will be explored throughout this chapter.

Derivatives of Functions The Fourier transform of the derivative of a function is given by:

$$F\left(\frac{d}{dx}f(x)\right) = \int_{-\infty}^{\infty} \frac{dx}{f'(x)} e^{-i\omega x} dx \qquad (2.19a)$$

$$= \left[\underbrace{f(x)e^{-i\omega x}}_{w} \right]_{-\infty}^{\infty} - \int_{-\infty}^{\infty} \underbrace{f(x)}_{v} \left[\underbrace{-i\omega e^{-i\omega x}}_{dx} \right] dx \qquad (2.19b)$$

$$= i\omega \int_{-\infty}^{\infty} f(x)e^{-i\omega x} dx \qquad (2.19c)$$

$$= i\omega F(f(x)).$$
 (2.19d)

This is an extremely important property of the Fourier transform, as it will allow us to turn PDEs into ODEs, closely related to the separation of variables:

$$u_{tt} = cu_{tx} \xrightarrow{\mathcal{F}} \hat{u}_{tt} = -c\omega^2 \hat{u}.$$
 (2.20)
(PDE) (ODE)

Linearity of Fourier Transforms The Fourier transform is a linear operator, so that:

$$F(\alpha f(x) + \beta g(x)) = \alpha F(f) + \beta F(g).$$
 (2.21)

$$F^{-1}(\alpha \hat{f}(\omega) + \beta \hat{g}(\omega)) = \alpha F^{-1}(\hat{f}) + \beta F^{-1}(\hat{g}).$$
 (2.22)

Parseval's Theorem

$$\int_{-\infty}^{\infty} |\hat{f}(\omega)|^2 d\omega = 2\pi \int_{-\infty}^{\infty} |f(x)|^2 dx. \quad (2.23)$$

In other words, the Fourier transform preserves the L_2 norm, up to a constant. This is closely related to unitarity, so that two functions will retain the same inner product before and after the Fourier transform. This property is useful for approximation and truncation, providing the ability to bound error at a given truncation.

Convolution The convolution of two functions is particularly well-behaved in the Fourier domain, being the product of the two Fourier transformed functions. Define the convolution of two functions f(x) and g(x) as $f \neq g$:

$$(f * g)(x) = \int_{-\infty}^{\infty} f(x - \xi)g(\xi) d\xi.$$
 (2.24)

If we let $\hat{f} = F(f)$ and $\hat{g} = F(g)$, then

$$\mathcal{F}^{-1}\left(\hat{f}\hat{g}\right)(x) = \frac{1}{2\pi}\int_{-\infty}^{\infty}\hat{f}(\omega)\hat{g}(\omega)e^{i\omega x}d\omega$$
 (2.25a)

$$= \int_{-\infty}^{\infty} \hat{f}(\omega)e^{i\omega x} \left(\frac{1}{2\pi} \int_{-\infty}^{\infty} g(y)e^{-i\omega y} dy\right) d\omega \qquad (2.25b)$$

$$= \frac{1}{2\pi} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} g(y) \hat{f}(\omega) e^{i\omega(x-y)} d\omega dy \qquad (2.25c)$$

$$= \int_{-\infty}^{\infty} g(y) \left(\underbrace{\frac{1}{2\pi} \int_{-\infty}^{\infty} \hat{f}(\omega) e^{i\omega(x-y)} d\omega}_{-\infty} \right) dy \qquad (2.25d)$$

$$= \int_{-\infty}^{\infty} g(y)f(x - y) \, dy = g * f = f * g. \tag{2.25e}$$

Thus, multiplying functions in the frequency domain is the same as convolving functions in the spatial domain. This will be particularly useful for control systems and transfer functions with the related Laplace transform.

2.2 Discrete Fourier Transform (DFT) and Fast Fourier Transform (FFT)

Until now, we have considered the Fourier series and Fourier transform for continuous functions f(t). However, when computing or working with real-data, it is necessary to approximate the Fourier transform on discrete vectors of data. The resulting discrete Pourier transform (DFT) is essentially a discretized version of the Fourier series for vectors of data $f = [f_1 \ f_2 \ f_3 \ f_4 \ homedown \ h_1^2, 22, 23]$, where $f(t) = f(t) \ f(t) \ h_1^2, t)$ is regularized by the point of the point f(t) at a regular special, σ_i , σ_i shown in this f(t) = 2.

The DFT is tremendously useful for numerical approximation and computation, but it does not scale well to very large $n \gg 1$, as the simple formulation involves multiplication by a dense $n \times n$ matrix, requiring $O(n^2)$ operations. In 1965, James W. Cooley (IBM) ad John W. Takey (Princeton) developed the revolutionary (art Fourier transform (FFT)



Figure 2.7 Discrete data sampled for the discrete Fourier transform.

algorithm [137, 156] that scales as O(nlog(n)). As n becomes very large, the Ig(n) component grows show, and the algorithm approaches a larger scaling. The island was based on a fractal symmetry in the Fourier transform that allows an a dimensional DFT to be solved with a number of annual fermiosinal DFT comparations. Although the different comparation of the other than the DFT implement values may seen the different comparation of the different comparations. Although the different comparation of the different comparation of the different PFT in real-dimense communication. Issues of nuclei on all image commension (539).

It is important to note that Cooky and Takey did out invest the idea of the FFT, as there were checked on their works, developing excitations, although they provided the three were checked on their works, and the constraints of the second second second by Gians new TSI years carlot in 1005 supprovident the other of the annexide Fahau and the algorithm, and during a second second second second second second is an algorithm, and during a second second second second second break second second second second second second second second break second second second second second second second second break second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second second from second second second second second second second second second from second second second second second second second second from second second second second second second second second from second second second second second second second second second from second from second s

Discrete Fourier Transform

Although we will always use the FFT for computations, it is illustrative to begin with the simplest formulation of the DFT. The discrete Fourier transform is given by:

$$\hat{f}_{k} = \sum_{j=0}^{n-1} f_{j} e^{-i2\pi j k/n},$$
 (2.26)

and the inverse discrete Fourier transform (iDFT) is given by:

$$f_k = \frac{1}{n} \sum_{j=0}^{n-1} \hat{f}_j e^{i2\pi j k/n}. \quad (2.27)$$



Figure 2.8 Real part of DFT matrix for n = 256.

Thus, the DFT is a linear operator (i.e., a matrix) that maps the data points in f to the frequency domain \hat{f} :

$$(f_1, f_2, \dots, f_n) \xrightarrow{\text{DOT}} (\hat{f}_1, \hat{f}_2, \dots, \hat{f}_n).$$
 (2.28)

For a given number of points n, the DFT represents the data using sine and cosine functions with integer multiples of a fundamental frequency, $\omega_n = e^{-2\pi t/n}$. The DFT may be computed by matrix multiplication:

$$\begin{bmatrix} f_1 \\ f_2 \\ f_3 \\ \vdots \\ 1 \\ 1 \\ m_{q}^{d-1} = \frac{a_1^{d} \cdots a_q^{d-1}}{a_1^{d} \cdots a_q^{d-1}} \begin{bmatrix} f_1 \\ f_2 \\ \vdots \\ \vdots \\ \vdots \\ \vdots \\ f_n \\ f_n$$

The output vector \hat{f} contains the Fourier coefficients for the input vector \hat{f} , and the DFT matrix F is a unitary Vandermonde matrix. The matrix F is complex-valued, so the output \hat{f} has both a maximude and a balase, which will both have useful obviscual interoretations.

The real part of the DFT matrix \mathbf{F} is shown in Fig. 2.8 for n = 256. Code 2.2 generates and plots this matrix. It can be seen from this image that there is a hierarchical and highly symmetric multiscale structure to \mathbf{F} . Each row and column is a cosine function with increasine frequency. Code 2.2 Generate discrete Fourier transform matrix

Fast Fourier Transform

As methods carlier, multiplying by the DPT matrix F involves $O(n^2)$ operations. The forefore transform scales as O(n) legg(n), ending a tremmodule range of opplications, including and major compression in MPP and PPT formati, streaming Valos, tatleting the ending of the ending

To see the remendous beends of the FT, consider the transmission, storage, and decoding of an ando signal. We will see later that many signals are highly compressible in the Fourier transform domain, meaning that most of the coefficients of a resumal and can be doubted. This southies much more efficient storage and transmission of the compressed doubted that the storage of the storage storage storage storage storage accessary to reputly encode and decode the compressed Fourier signal by computing the TT and surverse FT of IPT. This is accessed from the signal complexity of the transformed storage storage

The basis idea behind the FFT is that the DFT may be implemented much more efficiently if the number of data points n is a power of 2. For example, consider $n = 1024 = 2^{10}$. In this case, the DFT mutrix **F**₁₀₂₄ may be written as:

$$\hat{\mathbf{f}} = \mathbf{F}_{1024} \mathbf{f} = \begin{bmatrix} \mathbf{I}_{512} & -\mathbf{D}_{512} \\ \mathbf{I}_{512} & -\mathbf{D}_{512} \end{bmatrix} \begin{bmatrix} \mathbf{F}_{512} & \mathbf{0} \\ \mathbf{0} & \mathbf{F}_{512} \end{bmatrix} \begin{bmatrix} \mathbf{f}_{rvm} \\ \mathbf{f}_{odd} \end{bmatrix}, \quad (2.30)$$

where f_{even} are the even index elements of f, f_{odd} are the odd index elements of f, I_{512} is the 512 × 512 identity matrix, and D_{512} is given by

$$\mathbf{D}_{512} = \begin{bmatrix} 1 & 0 & 0 & \cdots & 0 \\ 0 & \omega & 0 & \cdots & 0 \\ 0 & 0 & \alpha^2 & \cdots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & 0 & \cdots & \alpha^{511} \end{bmatrix}.$$
 (2.31)

This expression can be derived from a careful accounting and recognization of the terms in (226) and (220). If $n = 2^{-}$, this process can be represented by Eq., which can then be represented by Eq. $h \to h_{21} \to -h_{31} \to -h_{32} \to -h_{31} \to$

FFT Example: Noise Filtering

To gain familiarity with how to use and interpret the FFT, we will begin with a simple example that uses the FFT to denoise a signal. We will consider a function of time f(t):

$$f(t) = sin(2\pi f_1 t) + sin(2\pi f_2 t)$$
 (2.32)

with frequencies $f_1 = 50$ and $f_2 = 120$. We then add a large amount of Gaussian white noise to this simul, as shown in the top panel of Fig. 2.9.

It is possible to compute the fast Fourier transform of this noisy signal using the $\Pi''_{\rm T}$ command. The power spectral density (1251) is the normaline departed magnitude of $\Gamma_{\rm c}$ and indicates how much power the signal contains in each frequesty. In Fig. 2 (middle), it is clear that the noisy signal contains to long peaks at 2014 at 2014 $E_{\rm c}$ is possible to zero more components that have been being a block at 2014 and 2014 $E_{\rm c}$ is a possible to zero more components that the force being a block at 2014 $E_{\rm c}$ is possible to zero more components that the force being a block at 2014 $E_{\rm c}$ is a possible to zero more components that the force being at the first block at 2014 $E_{\rm c}$ block more block at 2014 $E_{\rm c}$ and 2014 $E_{\rm c}$ block more code 2.2 medium code that we and folts the results.

Code 2.3 Fast Fourier transform to denoise signal.

```
64 = 0.001
64 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7 = 0.001
7
```



Figure 2.9 De-neising with FFT. (top) Neise is added to a simple signal given by a sum of two sine waves. (middle) In the Fourier domain, dominant peaks may be selected and the noise filtered. (bottom) The de-neised signal is obtained by inverses Fourier transforming the two dominant peaks.

FFT Example: Spectral Derivatives

For the next example, we will demonstrate the use of the FFT for the fast and accurate computation of derivatives. As we saw in (2.19), the continuous Fourier transform has



Figure 210 Comparison of the spectral derivative, computed using the FFT, with the finite-difference derivative.

the property that $F(d/d_{1}) = (nF_{1}T)$. Similarly, the numerical derivative of a vector of discretized data can be well approximately pumiliplying each component of the discrete Fourier transform of the vector P by is, where $s = 2\pi k/n$ is the discrete wavenumber associated with that component. The accuracy and efficiency of the spectral derivative makes it particularly useful for solving partial differential equations, as explored in the next section.

To demonstrate this so-called spectral derivative, we will start with a function f(x)where we can compute the analytic derivative for comparison:

$$f(x) = \cos(x)e^{-x^2/25} \implies \frac{df}{dx}(x) = -\sin(x)e^{-x^2/25} - \frac{2}{25}xf(x).$$
 (2.33)

Fig. 2.10 compares the spectral derivative with the analytic derivative and the forward Euler finite-difference derivative using n = 128 discretization points:

$$\frac{df}{dx}(x_k) \approx \frac{f(x_{k+1}) - f(x_k)}{\Delta x}.$$
(2.34)

The error of both differentiation schemes may be reduced by increasing en, which is the mass adscensing bar. A towaver, the error of the spectral derivative improves more npidly with increasing a than finite-difference schemes, as shown in Fig. 21. The forward likely differentiation is noteriously inaccurate, with error proportional to $O(\Delta x)$; however, even increasing the softer of a faint-difference scheme was that as yield the same accuracy tred to $O(\Delta x)$. And and commarks the tree differentiation is noted, adding during the dots and commarks the tree differentiation schemes.

Code 2.4 Fast Fourier transform to compute derivatives.

```
 \begin{array}{ll} n = 128 ; \\ i = 30 ; \\ ds = -i/2\pi i s_1 i / 2 - \delta s_1 ; \\ ds = s_1 / 2\pi i s_1 i / 2 - \delta s_1 ; \\ s = \cos(s_1 \cdot s_1 - s_2 \cdot s_1 - s_1 ); \\ df = cos(s_1 \cdot s_1 - s_2 \cdot s_1 - s_1 ); \\ df = b rear(s_1 \cdot s_2 \cdot s_2 - s_1 ); \\ df = b rear(s_1 \cdot s_2 - s_1 ); \\ df = s_1 \cdot s_2 \cdot s_1 - s_2 \cdot s_1 , \\ df = s_1 \cdot s_1 \cdot s_2 \cdot s_1 \cdot s_1 , \\ df = s_1 \cdot s_1 \cdot s_2 \cdot s_1 \cdot s_1 , \\ df = s_1 \cdot s_1 \cdot s_2 \cdot s_1 \cdot s_1 \cdot s_1 \cdot s_1 \cdot s_1 \\ df = s_1 \cdot s_1 \cdot s_2 \cdot s_1 \cdot
```



Figure 2.11 Benchmark of spectral derivative for varying data resolution.

```
for keypelineghtd():

market():

dfp(deg):

dff(deg):

dff(de
```

If the derivative of a function is discontinuous, then the spectral derivative will exhibit Gibbs phenomena, as shown in Fig. 2.12.

2.3 Transforming Partial Differential Equations

The Fourier transform was originally formulated in the 1000s as a change of coordinates for the heat equation into an eigenfluctuation coordinate system where the dynamics decouple. More generally, the Fourier transform is useful for transforming partial differential equations (PDEs) into ordinary differential equations (ODEs), as in (12:20). Here, we will treatment or dispertial methods for PDEs, see Trieferten [533]; extensions also exist for util PDEs [232].

Heat Equation

The Fourier transform basis is ideally suited to solve the heat equation. In one spatial dimension, the heat equation is given by

$$u_t = \alpha^2 u_{xx}$$
 (2.35)



Figure 2.12 Gibbs phenomena for spectral derivative of function with discontinuous derivative.

where u(t, x) is the temperature distribution in time and space. If we Fourier transform in space, then $F(u(t, x)) = \hat{u}(t, \omega)$. The PDE in (2.35) becomes:

$$\hat{u}_{t} = -\alpha^{2}\omega^{2}\hat{u}$$
 (2.36)

since the two spatial derivatives contribute $(i\omega)^2 = -\omega^2$ in the Fourier transform domain. Thus, by taking the Fourier transform, the PDE in (2.35) becomes an ODE for each fixed frequency ω . The solution is given by:

$$\hat{u}(t, \omega) = e^{-\omega^2 \omega^2 t} \hat{u}(0, \omega),$$
 (2.37)

The function $\hat{n}(0, \omega)$ is the Fourier transform of the initial temperature distribution u(0, x). It is now clear that higher frequencies, corresponding to larger values of ω , decay more rapidly as time evolves, so that sharp corners in the temperature distribution rapidly mondo out. We may take the inverse Fourier transform using the convolution property in (2.24), vieldine:

$$u(t, x) = \mathcal{F}^{-1}(\hat{u}(t, \omega)) = \mathcal{F}^{-1}\left(e^{-a^{2}\omega^{2}t}\right) * u(0, x) = \frac{1}{2a\sqrt{\pi t}}e^{-\frac{a^{2}}{\omega^{2}t}} * u(0, x).$$
 (2.38)

To simulate this PDE numerically, it is simpler and more accurate to first transform to the frequency domain using the FFT. In this case (2.36) becomes

$$\hat{u}_{t} = -\alpha^{2} \kappa^{2} \hat{u}$$
 (2.39)

where s' is the discretized frequency. It is important to use the **fftshift** command to re-order the wavenumbers according to the Matlab convention.

Code 2.5 simulates the 1D heat equation using the FFT, as shown in Figs. 2.13 and 2.14. In this example, because the PDE is linear, it is possible to advance the system using ode45



Figure 2.13 Solution of the 1D heat equation in time for an initial condition given by a square hat function. As time evolves, the sharp corners rapidly smooth and the solution approaches a Gaussian function.



Figure 2.14 Evolution of the 1D heat equation in time, illustrated by a waterfall plot (left) and an x-t diagram (right).

directly in the frequency domain, using the vector field given in Code 2.6. Finally, the plotting commands are given in Code 2.7.

Figs. 21 and 2.14 does seemd different views of the temperature distribution $w(x_1)$ and the two levels in time E_{12} to have the distribution at everal interse ordering and this same data is simulated in Fig. 2.14 in a second plot (circl) and in an x-diagonin (right). In all the highest vacuumbars. The temperature is the highest vacuumbars. The temperature results are subscription, which is a solution of highest vacuumbars. The event shows that the highest vacuumbars. The event second results of the highest vacuumbars. The event second results was a second result of the highest vacuumbars of thighest vacuumbars of the highest vacuumbars of the highest

Code 2.5 Code to simulate the 1D heat equation using the Fourier transform.

Cade 2.8 Right-hand side for 1D heat equation in Fourier domain, du/dt.

function duhatdt = rhsHeat(t,uhat,kapps,s) duhatdt = -s^2*(kapps.^2)'.*uhat; % Linear and disgonal

Code 2.7 Code to plot the solution of the 1D heat caustion.

```
figure, waterfall((u(1:10:end,:)));
figure, imagesc(flipud(u));
```

One-Way Wave Equation

As second example is the simple linear PDE for the one-way equation:

$$u_r + cu_r = 0.$$
 (2.40)

Any initial condition u(0, z) will simply propagate to the right in time with speed z, as u(z, z) = u(t), z = v(t), z = v(t). Set $Z \ge 3$ simulates this PER for an initial condition given by a Gaussian pulse. It is possible to integrate this equation in the Fourier transform domain, as holescar using the vector field pixeu by Code 2.2. Bit however, it is also possible to and then variantism back, as in Code 2.10. The solution u(t), is plotted in Figs. 2.15 and z(f, as heffere.

Code 2.8 Code to simulate the 1D wave equation using the Fourier transform.



Figure 2.15 Solution of the 1D wave equation in time. As time evolves, the Gaussian initial condition moves from left to right at a constant wave speed.



Figure 2.16 Evolution of the 1D wave equation in time, illustrated by a waterfall plot (left) and an x-r diagram (right).

Code 2.9 Right hand side for 1D wave equation in Fourier transform domain.

function duhatdt = rhsWave(t,uhat,kapps,c)
duhatdt = -c+i+kapps.+uhat;



Figure 2.17 Solution of Burgers' equation in time. As time evolves, the leading edge of the Gaussian initial condition steepens, forming a shock front.

Code 2.10 Right hand side for ID wave constion in spatial domain.

```
function dudt = rhsMaveSpatial(t,u,kapps,c)
uhat = fft(u);
dubat = i_kappa.uhat;
du = ift(dubat);
dudt = -c.du;
```

Burgers' Equation

For the final example, we consider the nonlinear Burgers' equation

$$u_t + uu_x = vu_{xx}$$
 (2.41)

which is a simple 1D example for the nonlinear convection and diffusion that gives rise to shock waves in fluids [253]. The nonlinear convection aw_c essentially gives rise to the behavior of wave steepening, where portions of u with larger amplitude will convect more randity, causing a shock front to form.

Code 2.11 simulates the Biogrev equation giving rise to Figs. 2.17 and 2.18. Biogrev equation is an interesting scample to solve thin PFT because the confinently request us to major to and out of the Fourier domain with the FFT. because the confinently request us to major that and the scample, usen up has the Fourier transform domain to compute at and u_{ex} and h_{ex} and then map back to the equividing domain to compute the product any, Figs. 2.17 and 2.18 dourly show the wave stepenging effect that gives rise to a hock. Without the damping term u_{ex} this shock would become infinitely steep, but with damping, it maintains a faite withh

Code 2.11 Code to simulate Burgers' equation using the Fourier transform.



Figure 2.18 Evolution of Burgers' equation in time, illustrated by a waterfall plot (left) and an x-r diagram (right).

```
de = 1/2

de = -1/2 dirit/2-de; = Define a dumain

2 = pdfine discrete avenuabler

happe = dig2/df2, (2018)/2-1];

happe = dig2/df2, (2018)/2-1];

D finial condition

0 = meth(z);

4 = discrete discr
```

Code 2.12 Right hand side for Burgers' equation in Fourier transform domain.

```
function dudt = rhsBurgers(t,u,kapps,nu)
uhat = fft(u);
duhat = -(kapps.+uhat;
duhat = -(kapps.*2).+uhat;
du = ifft(duhat);
ddu = ifft(duhat);
ddut = -u.+du + nu+ddu;
```

2.4 Gabor Transform and the Spectrogram

Although the Fourier transform provides detailed information about the frequency content of a given signal, it does not give any information about when in time theorem frequencies occur. The Fourier transform is only able to characterize truly periodic and stationary signals, as time is stripped out visi the integration in [12,18], For a signal with notrattionary frequency content, such as a musical composition, it is important to simultaneously characterize the frequency content and its evolution in time.

The Gabor transform, also known as the short-time Fourier transform (STFT), computes a windowed FFT in a moving window [437, 262, 482], as shown in Fig. 2.19. This STFT



Figure 2.19 Illustration of the Gabor transform with a translating Gaussian window for the short-time Fourier transform.

enables the localization of frequency content in time, resulting in the spectrogram, which is a plot of frequency versus time, as demonstrated in Figs. 2.21 and 2.22. The STFT is given by:

$$\hat{g}(f)(t, \omega) = \hat{f}_{g}(t, \omega) = \int_{-\infty}^{\infty} f(\tau)e^{-i\omega\tau}\tilde{g}(\tau - t) d\tau = (f, g_{t,\omega})$$
 (2.42)

where $p_{r,m}(\mathbf{r})$ is defined as

$$g_{r,w}(\mathbf{r}) = e^{i\omega\tau}g(\mathbf{r} - t).$$
 (2.43)

The function g(r) is the kernel, and is often chosen to be a Gaussian:

$$g(t) = e^{-(t-\tau)^2/a^2}$$
(2.44)

The parameter a determines the spread of the short-time window for the Fourier transform, and τ determines the center of the moving window.

The inverse STFT is given by:

$$f(t) = G^{-1}(\hat{f}_g(t, \omega)) = \frac{1}{2\pi \|g\|^2} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} \hat{f}_g(\tau, \omega)g(t - \tau)e^{i\omega t} d\omega dt.$$
 (2.45)

Discrete Gabor Transform

Generally, the Gabor transform will be performed on discrete signals, as with the FFT. In this case, it is necessary to discretize both time and frequency:

$$v = j\Delta \omega$$
 (2.46)

$$r = k\Delta t$$
. (2.47)

The discretized kernel function becomes:

$$g_{j,k} = e^{i2\pi j\Delta\omega t}g(t - k\Delta t) \qquad (2.48)$$



Figure 2.20 Power spectral density of quadratic chirp signal

and the discrete Gabor transform is:

$$\hat{f}_{j,k} = \langle f, g_{j,k} \rangle = \int_{-\infty}^{\infty} f(\tau) \tilde{g}_{j,k}(\tau) d\tau.$$
 (2.49)

This integral can then be approximated using a finite Riemman sum on discretized functions f and $\tilde{e}_{\ell,k}$.

Example: Quadratic Chirp

As a simple example, we construct an oscillating cosine function where the frequency of oscillation increases as a quadratic function of time:

$$f(t) = \cos(2\pi t\omega(t))$$
 where $\omega(t) = \omega_0 + (\omega_1 - \omega_0)t^2/3t_1^2$. (2.50)

The frequency shifts from ω_0 at t = 0 to ω_1 at $t = t_1$.

Fig. 2.20 shows the power spectral density obtained from the FPT of the quadratic chirp signal. Although there is a clear peak at 50 Hz, there is no information about the progression of the frequency in time. The code to generate the spectrogram is given in Code 2.13, and the resulting spectrogram is plotted in Fig. 2.21, where it can be seen that the frequency content shifts in time.

Code 213 Spectrogram of quadratic chirp, shown in Fig. 2.21.

```
% = 0:0:0:1;
6 = 50;
1 = 20;
3 = shirp(t, 0; 1; 1; 'quadratic');
a = shirp(t, 0; 1; 1; 'quadratic');
a = ca(pqit...(6) = ([1:(0)t...2((3+t)^2]));
1 = there is a type in Melia documentation...
1 = ...divide by 1 = 0 derivative amplitude natches frequency
spectrogram(t, 15, 20; 20; 1; 4; c), 'vrats')
```

71



Figure 2.21 Spectrogram of quadratic chirp signal. The PSD is shown on the left, corresponding to the integrated power across rows of the spectrogram.

Example: Beethoven's Sonata Pathétique

It is possible to analyze richer signals with the spectrogram, such as Beethoven's Sonata Pathetique, shown in Fig. 2.2.2. The spectrogram is widely used to analyze music, and has recently been leveraged in the Shazam algorithm, which searches for key point markers in the spectrogram of songs to enable rapid classification from short clips of recorded musics [545].

Fig. 2.22 shows the first two bars of Beethoven's Sonata Pathétique, along with the spectrogram. In the spectrogram, the various chords and harmonics can be seen clearly. A zoom-in of the frequency shows two octaves, and how cleanly the various notes are excited. Code 2.14 loads the data, computes the spectrogram, and plots the result.

Cade 2.14 Compute spectrogram of Beethoven's Sonata Pathétique (Fig. 2.22).

```
    Deschard approach from http://www.antheorie.monitalizationtes.j.
(F.F.M.2007, COV). a privated (http://www.antheorie.monitalizationtes.j.
(F.F.M.2007, COV). a privated (http://www.antheorie.monitalization). A strate of anomala
prilitient (http://www.antheorie.monitalization). A strate of anomala
prilitient (http://www.antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/antheorie.com/anthe
```

To invert the spectrogram and generate the original sound:

[x_istft, t_istft] = istft(S, h, FS/4, FS); sound(x_istft,FS);



Figure 222 First two bars of Beethoven's Sonata Pathétique (No. 8 in C minor, Op. 13), along with annotated spectrogram.

Artists, such as Aphex Twin, have used the inverse spectrogram of images to generate music. The frequency of a given plano key is also easily computed. For example, the 40th key frequency is given by:

```
freq = 0(n) (((2^(1/12))^(n-49))*440);
freq(40) % frequency of 40th key = C
```

Uncertainty Principles

In time-frequency analysis, there is a fundamental uncertainty principle that limits the ability to simultaneously main high re-orders in both offset time and frequency domains. In the extreme limit, a time series is perfectly resolved in time, but provides no information about frequency constant, and the Fourier transformer perfects protosyste frequency constant, but provides no information about when in time these frequencies domains to provide no information about when in time these frequencies constant. The series both time and frequency informations, but while over endouslin in each domain, as illustrated in Fig. 22.2. An alternative approach, based on a multi-resolution analysis, will be the subtext of the next section.

Stated mathematically, the time-frequency uncertainty principle [429] may be written as:

$$\left(\int_{-\infty}^{\infty} x^2 |f(x)|^2 dx\right) \left(\int_{-\infty}^{\infty} \omega^2 |\hat{f}(\omega)|^2 d\omega\right) \ge \frac{1}{16\pi^2}.$$
 (2.51)

This is true if f(x) is absolutely continuous and both sf(x) and f'(x) are square integrable. The function $x^2|f(x)|^2$ is the dispersion about x = 0. For real-valued functions, this is the second moment, which measures the variance if f(x) is a Gaussian function. In other works, a function f(x) and its Fourier transform cannot both be arbitrarily localized. If the



Figure 2.23 Illustration of resolution limitations and uncertainty in time-frequency analysis.

function f approaches a delta function, then the Fourier transform must become broadband, and vice versa. This has implications for the Heisenberg uncertainty principle [240], as the position and momentum wave functions are Fourier transform pairs.

In time-frequency analysis, the uncertainty principle has implication for the ability to localize the Fourier transform in time. These uncertainty principles are known as the Galor limit. As the frequency content of a signal is resolved more finally, we loss information about when in time these events occur, and view event. Thus, there is a finalmental tradeoff between the simultaneously animable resolutions in the time and frequency domains, surport, maintain that they are localized, as stated in Bereachiet, thereem 18, 511.

2.5 Wavelets and Multi-Resolution Analysis

Werefers [20], 132] centred the concepts in Fourier analysis to more general enloquing, how and partially concepts the uncertained particle decound does by explosing a conduct different time and frequency folders in different frequency back, which hape the structure of the structure of particles in the structure of the structur

The basic idea in wavelet analysis is to start with a function $\psi(t)$, known as the mother wavelet, and generate a family of scaled and translated versions of the function:

$$\psi_{a,b}(t) = \frac{1}{\sqrt{a}}\psi\left(\frac{t-b}{a}\right). \quad (2.52)$$

The parameters a and b are responsible for scaling and translating the function ψ , respectively. For example, one can imagine choosing a and b to scale and translate a function to fit in each of the segments in Fig. 2.23 (d). If these functions are orthogonal then the basis may be used for projection, as in the Fourier transform.

The simplest and earliest example of a wavelet is the Haar wavelet, developed in 1910 [227]:

$$\psi(t) = \begin{cases} 1 & 0 \le t < 1/2 \\ -1 & 1/2 \le t < 1 \\ 0 & \text{otherwise.} \end{cases}$$
(2.53)

The three Haar wavelets, $\psi_{1,0}$, $\psi_{1/2,0}$, and $\psi_{1/2,1/2}$, are shown in Fig. 2.24, representing the first two layers of the multi-resolution in Fig. 2.23 (d). Notice that by choosing each higher frequency layer as a bisection of the next layer down, the resulting Haar wavelets are enthronousl, newviding a hierarchical basis for a simul.

The orthogonality property of wavelets described above is critical for the development of the discrete wavelet transform (DWT) below. However, we begin with the continuous



Figure 2.24 Three Haar wavelets for the first two levels of the multi-resolution in Fig. 2.23 (d).

wavelet transform (CWT), which is given by:

$$W_{\psi}(f)(a, b) = \langle f, \psi_{a,b} \rangle = \int_{-\infty}^{\infty} f(t) \tilde{\psi}_{a,b}(t) dt,$$
 (2.54)

where $\tilde{\psi}_{a,b}$ denotes the complex conjugate of $\psi_{a,b}$. This is only valid for functions $\psi(t)$ that satisfy the boundedness property that

$$C_{\psi} = \int_{-\infty}^{\infty} \frac{|\hat{\psi}(\omega)|^2}{|\omega|} d\omega < \infty. \quad (2.55)$$

The inverse continuous wavelet transform (iCWT) is given by:

$$f(t) = \frac{1}{C_{\psi}} \int_{-\infty}^{\infty} \int_{-\infty}^{\infty} W_{\psi}(f)(a, b)\psi_{a,b}(t) \frac{1}{a^2} da db.$$
 (2.56)

New wavelets may also be generated by the convolution $\psi + \phi$ if ψ is a wavelet and ϕ is a bounded and integrable function. There are many other popular mother wavelets ψ beyond the Haar wavelet, designed to have various properties. For example, the Mexican hat wavelet is given by:

$$\psi(t) = (1 - t^2)e^{-t^2/2}$$

(2.57a)

$$\hat{\psi}(\omega) = \sqrt{2\pi}\omega^2 e^{-\omega^2/2}$$
. (2.57b)

Discrete Wavelet Transform

As with the Fourier transform and Gabor transform, when computing the wavelet transform on data, it is necessary to introduce a discretized version. The discrete wavelet transform (DWT) is given by:

$$W_{\psi}(f)(j, k) = \langle f, \psi_{j,k} \rangle = \int_{-\infty}^{\infty} f(t) \tilde{\psi}_{j,k}(t) dt$$
 (2.58)

where $\psi_{j,k}(t)$ is a discrete family of wavelets:

$$\psi_{j,k}(t) = \frac{1}{a^j}\psi\left(\frac{t-kb}{a^j}\right). \quad (2.59)$$

Again, if this family of wavelets is orthogonal, as in the case of the discrete Haar wavelets described earlier, it is possible to expand a function f(t) uniquely in this basis:

$$f(t) = \sum_{j,k=-\infty}^{\infty} (f(t), \psi_{j,k}(t))\psi_{j,k}(t).$$
 (2.60)

The explicit computation of a DWT is somewhat introbed, and is the subject of several excellent papers and text [59], 145, 53, 100, 373, 100, sover, the gud here is not to provide computational details, but rather to give a high-revel data of shut the wavelet transform accomplishes, by coaling and translating a given shape accoust a signal, it is possible to efficiently extract multi-scale structures in an efficient hierarchy that provides an optimal tradoff brevenes time and frequency resolution. This general proceedure is widely used in and/o and image processing, compression, scientific computing, and machine learning, to mare a few examples.

2.6 2D Transforms and Image Processing

Although we analyzed both the Fourier transform and the wavelet transform on onedimensional signals, both methods readily generalize to higher spatial dimensions, such as two-dimensional and three-dimensional signals. Both the Fourier and wavelet transforms have had termendous impact on image processing and compression, which provides a compelling example to investigate higher-dimensional transforms.

2D Fourier Transform for Images

The two-dimensional Fourier transform of a matrix of data $X \in \mathbb{R}^{n\times m}$ is achieved by first applying the one-dimensional Fourier transform to every row of the matrix, and then applying the one-dimensional Fourier transform to every other during of the intermediate matrix. This sequential row-wise and column-wise Fourier transform is shown in Fig. 2.25. Switching the order of taking the Fourier transform of row and columns does not change the result.

Cade 2.15 Two-dimensional Fourier transform via one-dimensional row-wise and column-wise FFTs.

77



Figure 225 Schematic of 2D FFT. First, the FFT is taken of each row, and then the FFT is taken of each column of the resulting transformed matrix.

```
Cwhitf(j,l) = fftanhit(fft(N(j,l)));
mc(j,l) = (fft(N(j,l)));
mc(j,l) = (fft(N(j,l)));
mc(j,l) = (fft(C,l)) = (Compute column-wise FFT
for j=lision(C,l) = (Compute column-wise FFT
mcd
ashpio(c(j,l), images(fftanhift(log(abs(D)))))
D = fft2(Ab). Hoch more efficient to use fft2
```

The two-dimensional FFT is effective for image compression, as many of the Fourier coefficients are small and may be neglected without loss in image quality. Thus, only a few large Fourier coefficients must be stored and transmitted.

Code 2.16 Image compression via the FFT.

```
Bintfill(B); 1 & is groupeld image from above
Bintfill(B); 1 & is groupeld image from above
Bintfill south and a set of the set of the set of the
property of the set of the set of the set of the set of the
fill set of the set of the set of the set of the set of the
Allow set. Set of the set of the set of the set of the
allow set of the set of the set of the second image
allows (hard set of the set of the second image
allows (hard set of the second image image image
allows (hard set of the second image image image
allows (hard set of the second image image image image
and the set of the second image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image image image image image image
and the second image image
and the second image im
```

Finally, the FPT is extensively used for denoising and filtering signals, so it is straightforward to isolate an imaginate particular frequency band. Code 2.17 and Fig. 2.27 demonstrate the use of a FPT threshold filter to denoise an image with Claussian noise add. In this example, it is observed that the noise is especially pronounced in high frequency modes, and we therefore zero out any Fourier coefficient outside of a given radius containing to frequencies.





1.0% of FFT

0.2% of FFT



Figure 228 Compressed image using various thresholds to keep 5%, 1%, and 0.2% of the largest Fourier coefficients.

Code 2.17 Image denoising via the FFT.

```
 \begin{aligned} & \text{motions } n = \text{wind}\left(200 - \text{model}\left(\sin(n)\right)\right) + \frac{1}{2} \text{ dist sour motions} \\ & \text{motival}\left(100 - 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100 + 100
```

79



Figure 227 Denoising image by eliminating high-frequency Fourier coefficients outside of a given radius (bottom right).

```
prahifile = Btahifi..ind;
Yfile 1 o(gdw (Eshififile)+1); # Put FFT on log-scale
subplot(2,2,4), images(Ffile) # Plot filtered FFT
pfile = iffet(Bfile);
Will = iffet(Bfile);
Will = iffet(Bfile);
Mile = iffet(Bfile);
```

2D wavelet Transform for Images

Similar to the FFT, the discrete wavelet transform is extensively used for image processing and compression. Code 2.18 computes the wavelet transform of an image, and the first



Figure 2.28 Illustration of three level discrete wavelet transform.

three levels are illustrated in Fig. 2.28. In this figure, the hierarchical nature of the wavelet decomposition is seen. The upper left corner of the DWT image is a low-resolution version of the image, and the subsequent features add fine details to the image.

Code 2.18 Example of a two level wavelet decomposition

```
All Annual Antomycettem (2 (1997) (2 (1997)))

All Annual Antomycettem (2 (1997)))

All Annual (2 (1997))

Annual (2 (1
```

Fig. 2.29 shows several versions of the compressed image for various compression ratios, as compated by Code 2.19. The hierarchical representation of data in the wavelet transform is ideal for image compression. Even with an aggressive truncation, retaining only 0.5% of the DWT coefficients, the coarse features of the image are retained. Thus,



Figure 229 Compressed image using various thresholds to keep 5%, 1%, and 0.5% of the largest wavelet coefficients.

when transmitting data, even if bandwidth is limited and much of the DWT information is truncated, the most important features of the data are transferred.

Code 2.19 Wavelet decomposition for image compression.

```
[C,S] = vavedec2(n,4,'db:');
Caort = sort(sbs(C(i))); & Sort by magnitude
for keep = [.1.05.01.005]
threats = Cort(floor((1-keep),length(Caort)));
ind = sbs(C)threats;
Cfilt = C.sind; & Threshold small indices
```



Suggested Reading

Texts

- (1) The analytical theory of heat, by J.-B. J. Fourier, 1978 [185].
- (2) A wavelet tour of signal processing, by S. Mallat, 1999 [357].
- (3) Spectral methods in MATLAB, by L. N. Trefethen, 2000 [523].

Papers and reviews

- An algorithm for the machine calculation of complex Fourier series, by J. W. Cooley and J. W. Tukey, Mathematics of Computation, 1965 [137].
- (2) The wavelet transform, time-frequency localization and signal analysis, by L Daubechies, IEEE Transactions on Information Theory, 1990 [145].
- (3) An industrial strength audio search algorithm, by A. Wang et al., Ismir, 2003 [545].

The laberer structure observed in natural data implics that the data data is aquare regress data data and appropriate the structure of the st

On the decosing on paralog and compressed similary still accountly involve the critically comparent fields of optimizers and matrices. Space 1 and a properties in process means the start of parameters of the start of the start of the start of the start within a start of the start optimizers of the start of the start of the start start of the start of the start optimizers of the start optimizers of the start optimizers of the start optimizers of the start optimizer approximation of the start optimizer approximation of the start of the start of the start of the start optimizer approximation of the start of the start of the start of the start optimizer approximation of the start of the start of the start of the start optimizer approximation of the start of the start of the start of the start optimizers of the start optimizers of the start of the start of the start of the start optimizers of the start optimizers of the start of the start optimizers of the start of the start optimizers optimizers of the start optimizers of the start optimizers of the start optimizers optimizers of the start optimizers optimizers optimizers of the start optimizers optimizers of the start optimizers opt

3.1 Sparsity and Compression

Most natural signals, such as images and andic, are highly compressible. This compressibility means that when the signal is written in an appropriate basis only a few modes are active, thus reducing the number of values that must be stored for an accurate representation. Sid another way, a compressible signal $X \in \mathbb{R}^{2n}$ may be written as a sparse vector $s \in \mathbb{R}^{2n}$ containt mostly zerosi to a transform basis $v \in \mathbb{R}^{2n}$.

Specifically, the vector s is called K-sparse in Ψ if there are exactly K nonzero elements. If the basis Ψ is generic, such as the Fourier or wavelet basis, then only the few active terms in s are required to reconstruct the original signal x, reducing the data required to store or transmit the signal.

Images and ando signaba are both compressible in Fourier or wavelet bases, so that that taking the Fourier overdet transform, more correlations are sumd and may be set exactly equal to zero with medigable loss of quality. These for active coefficients may the constraint of the standard of the original place dimensional ignal. This is necessarily the original signal is the analysis of gas of the standard dimensional ignal standard dimension of the standard dimensional ignal fractions are standard constantly to the standard dimension of the standard dimension of the standard dimension of the standard dimension of the standard dimension standard dimension of the standard dimension of the standard dimension of the spaces coefficients in a Thie is the foundation of PEG compression for images and MP3 compression for making the standard dimension of the standard dimensis dimension of the standard dimension of the standard

The broater modes and swardsen are gravies or anizorab bases, and hence seen that angult and main language considerable are space in the bases. Therefore, once a signal is compressed, one needs only since or maximum the space vectors at there than the neuron model. The start of start of the start of th

Although the majority of compression theory has been driven by andie, image, and video applications, there are many implications for engineering systems. The solution to a highdimensional system of differential equations typically evolves on a low dimensional material differential equations typically evolves on a low dimensional material differential equations typically evolves on a low dimensional procession of the evolution of the evolution of the evolution of the two hoodband phenomes, such as trathetic, may be institutionedly characterized by a garare representation. This has a profound impact on how to sense and compare, as will be described throughout this chapter and the remainder of the book.

Example: Image Compression

Compression is relatively simple to implement on images, as described in Section 2.6 and revisited here (see Fig. 3.1). First, we load an image, convert to grayscale, and plot:

```
Asimread('jelly', 'jpeg'); & Load image
Abwergb2gray(A); & Convert image to grayscale
imabow(Abw). & Plot image
```

Next, we take the fast Fourier transform and plot the coefficients on a logarithmic scale:

```
At=fft2(Abw);
F = log(abs(fftshift(At))+1); * put FFT on log-scale
inshbw(mat2gray(F),[]);
```

85


Figure 3.1 Illustration of compression with the fast Fourier transform (FFT) F.

To compress the image, we first arrange all of the Fourier coefficients in order of magnitude and decide what percentage to keep (in this case 5%). This sets the threshold for truncation:

```
Bt = sort(abs(At(:)));
keep = 0.05;
threah = Bt(floor((1-keep)*length(Bt)));
ind = abs(At)>threah;
Atlow = At.sind;
```



Figure 3.2 Compressed image (left), and viewed as a surface (right).

Finally, we plot the compressed image by taking the inverse FFT (iFFT):

```
Alowsuint8(ifft2(Atlow));
imshow(Alow)
```

To understand the role of the sparse Fourier coefficients in a compressed image, it helps to view the image as a surface, where the height of a point is given by the brightness of the corresponding pixel. This is shown in Fig. 3.2. Here we see that the surface is relatively simple, and may be represented as a sum of a few spatial Fourier modes.

```
Anew = imresize(Abw,.2);
surf(double(Anew));
shading flat, view(-168,86)
```

Why Signals Are Compressible: The Vastness of Image Space

It is important to note that the compressibility of images in related to the overwhelming dimensionality of image space. For even a simple 20 × 20 pixel black and white image, there are 2ⁿ⁰⁰ distinct possible images, which is larger than the number of nucleons in the kanow universe. The number of images is considerably more staggering for higher resolution images with prestra could expth.

In the space of one megapite images (*a*_1000 × 1000 pitch), there is an image of me call being born, on the ryping this instrume, and of you realing it. However van the space of these minumal images, they ecceptly a tiny, minuscube fraction of the total image space. The implicit of the images is image space represent random noise, executining television static, and the space of the total provides of each of the pitch. With exceedingly high probability, the resulting image value theo, this noise, which no approxet significance. Yue could draw these random images for an approxame of the space of the pitch is a space of the space of the total pitch is a space of the space of



Figure 3.3 Illustration of the vastness of image (pixel) space, with natural images occupying a vanishinely small fraction of the space.

entire lifetime and never find an image of a mountain, or a person, or anything physically recognizable¹.

In other words, natural images are extremely rare in the vantuess of image space, as illustrated in Fig. 3.3. Because so may images are unstructured or random, most of the dimensions used to encode images are only necessary for these random images. These dimensions is not rendmatur if all we care about vue seconding natural images. An important implication is that the images we care about Ca, natural images in the highly compressible, if we find a suitable transformed have shore the rendmated immensions are easily identified.

3.2 Compressed Sensing

Despite the considerable success of compression in real-world applications, is still related to having access to this hydrometicanit measures. The neural measures of compressed seming [153, 111, 111, 113, 115, 109, 109, 114, 60] tunts the compression paradiap aspidasming [154, 121, 111, 113, 115, 109, 109, 114, 60] tunts the compression paradiap aspidated optimal paradiapped seminary of the paradiapped seminary of the parameasurements and then later what the sparse representation is in the transformed basis to take basis compared seminary in the sparse representation is in the transformed basis recently finding the sparse sector constants with measurements was a non-polynomial recently finding the sparse sector constants with measurements was a non-polynomial in applied sciences results of the sparse representation in Tanework's the produce conditions.

¹ The summer of signal space was described in Borgies', "The Library of Buble" in 1944, where he describes a Heavy continuing all gradefit looks it and only low strate, of which strate observes book exceepy a narguing and traction (1944). In Borgies', Bhorgy with the strate observes book exceepy a narguing strations on this indige summers. Another hannes variations on this indige summers. Another hannes variations and the shear consider hand in given among homedey typing on energies hypowritaric, new world eventually recruited by works of Makanguare. These of the definest related descriptions of these combined here in a single strates, and the single same strategies are an advected by the single strates. The shear the single strates is the single strates in the single strates in the single strates.

² Interestingly, the incredibly important collaboration between Emmanuel Candés and Terrance Tao began with them discussing the odd properties of signal reconstruction at their kids' daycare. for when it is possible to reconstruct the full signal with high probability using convex algorithms.

Mathematically, compressed sensing explosits the sparsity of a signal in a generic basits to achieve full signal construction from superinging for measurements. If a signal is is K-sparse in Ψ , then instead of measuring x directly (in measurements) and then compressing, it is possible to colder durantizally freezer randomly chosen co-composer dimensurements and then solve for the neutron elements of a sin the transformed coordinate system. The measurements $P_{\rm eff}(\Phi_{\rm eff}) = P_{\rm eff}(\Phi_{\rm eff}) = P_{\rm eff}(\Phi_{\rm eff})$ are given by

$$y = Cx.$$
 (3.2)

The measurement matrix $C \in \mathbb{R}^{n\times n}$ represents as of p linear measurements on the state x. The choice of measurement matrix C is of critical importance in compressed sensing, and is discussed in Section 3.4. Typically, measurements may consist of random projections of the state, in which case the entries of C are Gaussion on Bernould distributed variable variables. It is also possible to measure individual entries of x (i.e., single pixels if x is an immary limit. The effective effective entries of x (i.e., single pixels if x is an immary limit, in which case C consists of random rows of the identity matrix.

With knowledge of the sparse vector s it is possible to reconstruct the signal x from (3.1). Thus, the goal of compressed sensing is to find the sparsest vector s that is consistent with the measurements y:

$$y = C\Psi s = \Theta s.$$
 (3.3)

The system of equations in (3.3) is underdetermined since there are infinitely many consistent solutions s. The sporsest solution s satisfies the following optimization problem:

$$\hat{s} = \operatorname{argmin} \|s\|_0$$
 subject to $y = C\Psi s$, (3.4)

where $\|\cdot\|_0$ denotes the ℓ_0 pseudo-norm, given by the number of nonzero entries; this is also referred to as the cardinality of s.

The optimization in (3.4) is non-convex, and in general the solution can only be found with a better-force search that is combinization in a mail K. In particular, all possible Ksparse vectors in \mathbb{R}^n must be checked; if the exact level of sparsity K is micknown, the search is even through Encount this search is combinationi, solving (3.4) is instactable for even moderately large n and K, and the prospect of solving larger problems does not improve with Moore's law of expenses middly lincersing comparisonal power.

Fortunately, under certain conditions on the measurement matrix C, it is possible to relax the optimization in (3.4) to a convex ℓ_1 -minimization [112, 150]:

$$\hat{s} = \operatorname{argmin} \|s\|_1$$
 subject to $y = C\Psi s$, (3.5)

where || - ||1 is the £1 norm, given by

$$\|\mathbf{s}\|_1 = \sum_{k=1}^{n} |s_k|.$$
 (3.6)

³ In the compressed sensing literature, the measurement matrix is often denoted Φ; instead, we use C to be consistent with the output equation in control theory. Φ is also already used to denote DMD modes in Charter 7.



Figure 3.4 Schematic of measurements in the compressed sensing framework.



Figure 3.5 *l*₁ and *l*₂ minimum norm solutions to compressed sensing problem. The difference in solutions for this regression are further considered in Chapter 4.

The ℓ_1 norm is also known as the taxicab or Manhattan norm because it represents the distance a taxi would take between two points on a rectangular grid. The overview of compressed sensing is shown schematically in Fig. 3.4. The ℓ_1 minimum-norm solution is sparse, while the ℓ_2 minimum norm solution is not, as shown in Fig. 3.5.

There are very specific conditions that must be met for the ℓ_1 -minimization in (3.5) to converge with high probability to the sparsest solution in (3.4) [109, 111, 39]. These will be discussed in detail in Sec. 3.4, although they may be summarized as:

- 2. The number of measurements p must be sufficiently large, on the order of

$$p \approx O(K \log(n/K)) \approx k_1 K \log(n/K).$$
 (3.7)

The constant multiplier k1 depends on how incoherent C and V are.



Figure 3.6 Schematic illustration of compressed sensing using ℓ_1 minimization. Note, this is a dramatization, and is not actually based on a compressed sensing calculation. Typically, compressed sensing of immess requires a similicant number of measurements and is computationally redubitive.

Roughly speaking, these two conditions guarantee that the matrix $C\Psi$ acts as a unitary transformation on K sparse vectors s, preserving relative distances between vectors and enabling almost certain signal reconstruction with ℓ_1 convex minimization. This is formulated precisely in terms of the restricted isometry property (RIP) in Sec. 3.4.

The last of composed sensing may be committante or a first, opecularly prior to they bound by the sensitive sensitive sensitive sensitive sensitive sensitive bounds by the sensitive sensitive sensitive sensitive sensitive that is a support of a sensitive compresent descent sensitive compresent descent sensitive sen

Disclaimer

A cough schematic of compressed sensing is shown in Fig. 3.6. However, this schematic is a dramatration, and is not texturly based on a compressed sensing calculation since using compressed sensing for image reconstruction is comparisonally prohibitive. It is important to note that for the majority of applications in imaging, compressed sensing is non-practical. However, images are often still used to motivate and explain compressed sensing because of the scate midlexciton.

Upon closer inspection of this image example, we are analyzing an image with 1024×768 pixels and approximately 5% of the Fourier coefficients are required for accurate compression. This puts the sparsity level at $K = 0.05 \times 1024 \times 768 \approx 40,000$. Thus,

a back of the envelope estimate using (3.7), with a constant multiplier of $k_1 = 3$, indicates that we need $p \approx 350,000$ measurements, which is about 45 % of the original pixels. Even if we had access to these 45 % random measurement, inferring the correct sparse vector of Fourier coefficients is computationally prohibitive, much more so than the efficient FFT based image compression in Section 3.1.

Compressed sensing for images is typically only used in special cases where a reduction of the number of measurements is significant. For example, an early application of compressed sensing technology was for infant MRI (magnetic resonance imaging), where reduction of the time a child must be still could reduce the need for dangerous heavy sedation.

However, it is easy to see that the number of measurements p scales with the sparsity level K, so that if the signal is *move* sparse, then fewer measurements are required. The viewpoint of sparsity is still vialuable, and the mathematical innovation of convex relaxation of combinatorially hard f_0 problems to convex f_1 problems may be used much more broadly than for compressed sensing of images.

Alternative Formulations

In addition to the f_-minimization in (3.5), there are alternative approaches based on greedy algorithms [525, 526, 528, 527, 530, 243, 529, 207, 531, 205, 398, 206] that determine the sparse solution of (3.3) through an iterative matching parsuit problem. For instance, the compressed sensing matching parsait (CoSAMP) [398] is computationally efficient, easy to implement, and freely available.

When the measurements y have additive noise, say white noise of magnitude s, there are variants of (3.5) that are more robust:

$$\hat{s} = \operatorname{argmin} \|s\|_1$$
, subject to $\|C\Psi s - y\|_2 < \epsilon$. (3.8)

A related convex optimization is the following:

$$\hat{s} = \operatorname{argmin} \|C\Psi s - y\|_2 + \lambda \|s\|_1,$$
 (3.9)

where $\lambda \ge 0$ is a parameter that weights the importance of sparsity. Eqs. (3.8) and (3.9) are closely related [528].

3.3 Compressed Sensing Examples

This section explores concrete examples of compressed sensing for sparse signal recovery. The first example shows that the ℓ_1 norm promotes sparsity when solving a generic underdetermined system of equations, and the second example considers the recovery of a sparse two-one and/o signal with compressed sensing.

21 and Sparse Solutions to an Underdetermined System

To see the sparsity promoting effects of the ℓ_1 norm, we consider a generic underdetermined system of equations. We build a matrix system of equations $\mathbf{y} = \mathbf{\Theta}\mathbf{s}$ with p = 200rows (measurements) and n = 1000 columns (unknowns). In general, there are infinitely mary solutions of that are consistent with these equations, unless we are very unfortunate and the row equations are linearly dependent while the measurements are inconsistent in these rows. In fact, this is an excellent example of the probabilistic thinking used more generally in compressed sensing; if we generate a linear system of equations at random, that has sufficiently many more unknowns than knowns, then the resulting equations will have infinitely many solutions with help probability.

In MATLAR, it is straightforward to solve this underdetermined linear system for both the minimus t_1 norm and minimum t_1 porom solutions. The minimum t_2 norm solutions is obtained using the pseudo-inverse (related to the SVD from Chapters 1 and 4). The mintimus t_1 norm solution is obtained with the exx (ConvEyO pointization package, Fig. 3.7 shows that the t_1 -minimum solution is also that one the exerce of the system of the solution of the end of the solution of the end of the solution is both and the ext (ConvEyO primitation package, Fig. 3.7 while the t_2 -minimum solution is also, with a bit of energy in each vector coefficient.

Code 3.1 Solutions to underdetermined linear system y = Os.

```
% Bollow y= Thete.s = for y="
a 1000; i Binacim of s =
1000; i Binacim of s =
y = mandh(y=1)
y = mandh(y=1)
y = mandh(y=1)
y = mandh(y=1)
i Li minima norm solution s _Li
exclusion __Li(s)
visibility __Li(s)
visibility __Li(s)
visibility __Li(s)
Thetes_Li = y;
ever_mandj
```



Figure 3.7 Comparison of l₁-minimum (blue, left) and l₂-minimum norm (red, right) solutions to an underdetermined linear system.

Recovering an Audio Signal from Sparse Measurements

To illustrate the use of compressed sensing to reconstruct a high-dimensional signal from a sparse set of random measurements, we consider a signal consisting of a two-tone audio signal:

$$x(t) = \cos(2\pi \times 97t) + \cos(2\pi \times 777t).$$
 (3.10)

This signal is clearly sparse in the frequency domain, as it is defined by as use of exactly true consides waves. The higher frequency present is 10^{-7} Res to atthe typoids ampling rate is 1554R. However, becarging the sparsity of the signal in the frequency domain, we cancentrify coccounts the signal with random samples that are appeared sampling rate of 221R, which is well follower the Vypain sampling rate Fig. 35 does well as generated from -2 to $t_{-} = 100$ to -100 to -100 to -100 to -100 to -100 to samples -100 to the method point of -100 to -100 to

Code 3.2 Compressed sensing reconstruction of two-tone cosine signal.

Generate signal, DCT of signal n = 4096; # points in high resolution signal





```
[* slopes], slopes
[* slopes], slopes
* fittal |* for slopes transformation (space)
WE = strategy (slopes transformation (space)
# strategy (slopes) = formation (slopes)
p = 138 | some constant sampling (slopes)
p = 138 | slopes | slopes (slope slope slope slope)
p = slopes | slopes (slopes) | slope slope slope slope slope
p = strategy (slopes) | slopes (slope slope slope slope) | slopes (slopes) | slopes (slopes) | slopes slope slope slopes) | slopes (slopes) | slopes) | slopes) | slopes (slopes) | slopes) | slop
```

It is important to note that the p = 128 measurements are randomly chosen from the 4, 098 resolution signal. Thus, we have the precise iming of the space measurements at a much higher resolution that no our sampling rate. If we chose p = 128 measurements uniformly in rine, the compressed sensing algorithm that. Secielically, it we compare the PSD directly from these uniform measurements, the high-frequency signal will be aliased resulting in emonosus frequency peaks.

Finally, it is also possible to replace the matching pursuit algorithm

```
s = cosamp(Thets,y',10,1.e-10,10); # CS via matching pursuit
```

```
with an l1 minimization using the CVX package [218]:
```

In the compressed sensing matching pursuit (CoSaMP) code, the desired level of sparsity K must be specified, and this quantity may not be known ahead of time: The t_1 minimization routine does not require knowledge of the desired sparsity level a priori, although convergence to the sparset solution relies on having sufficiently many measurements <math>p, which indirectly desends on K.

3.4 The Geometry of Compression

Compressed sensing can be summarized in a relatively simple statement: A given signal, if it is sufficiently space in a known basis, may be recovered (with high probability) using significantly lever measurements as the signal length. If there are sufficiently many measurements and these measurements are sufficiently mandum. Each part of this susment can be gootnery of space vectors, and how these vectors are transformed through rankom measurements. Specifically, neough people measurement was untered in the system of the state of the system of the syst that preserves the distance and inner product structure of spanse vectors s. In other words, we seek a measurement matrix C to nut Φ acts as a next ionnerry may on sparse vectors. Isometry literally means same distance, and is closed related to unitarity, which not only preserves: distance, but also angles between vectors. Where Φ acts as a rar isometry, it is possible to solve the following equation for the sparsest vector s using convex ℓ_1 minimization:

$$y = \Theta s.$$
 (3.12)

The remainder of this section describes the conditions on the measurement matrix C that are required for Θ to act as a near isometry map with high probability. The geometric properties of various norms are shown in Fig. 3.9.

Determining how many measurements to take is related by single. If the signal is a base of measurements that all the of events are more, then its maskes of multiplier $|_{1}$, which defines a carbo by measurements are needed, depind to all the constraints of the signal of the measurements are needed, depind to the measurements are needed, depind to the measurement are needed, depind to the measurement are needed with the signal transfer and the signal probes of the



Figure 33 The minimum norm point on a line in different ℓ_{μ} norms. The blue line represents the solution set of numler-determined system of equations, and the red curves represent the minimum-norm local sets that intersect this blue line for different norms. In the norms between ℓ_{μ} and ℓ_{μ} , the minimum-norm solution also corresponds to be separate toddings, with only one coordinate active. In the ℓ_{μ} and higher norms, the minimum-norm solution is not sparse, but has all

quency response. The more incoherent the measurements, the smaller the required number of measurements p.

The incoherence of measurements C and the basis Ψ is given by $\mu(C, \Psi)$:

$$\mu(\mathbf{C}, \Psi) = \sqrt{n} \max_{j,k} |\langle \mathbf{c}_k, \Psi_j \rangle|, \qquad (3.13)$$

where e_k is the kth row of the matrix C and ψ_j is the *j*th column of the matrix Ψ . The coherence μ will range between 1 and \sqrt{n} .

The Restricted Isometry Property (RIP)

When measurements are incoherent, the matrix CV satisfies a restricted isometry property (RIP) for sparse vectors s,

$$(1 - \delta_K) \|\mathbf{s}\|_2^2 \le \|\mathbf{C} \Psi \mathbf{s}\|_2^2 \le (1 + \delta_K) \|\mathbf{s}\|_2^2$$

with restriction lossery constar $k_{\rm E}$ (14). The constar $k_{\rm E}$ is defined in the sumface there that statistics for the lengality of R^2 suprave textures. Where $k_{\rm E}$ is small, then C we take a lengal constrainty on R approxements. The statistical is computed to the lengal statistical properties have been been been been been to more describe to textualized approxeme based to be boundon on $k_{\rm E}$. For a statistic is non-induced to textualized approxeme based in the boundon on $k_{\rm E}$ for the statistical properties that the statistical properties have a statistical for the statistical properties that the boundow on $k_{\rm E}$ for a specific C. Generally, increasing the number of measurements will decreme the constraint $k_{\rm E}$ into property of C W to at isometically on sparse versions. When these are solution on the C constraint demans of the Anaphy between K is that is case, these are bound on the constraint $k_{\rm E}$ more and the BH constraint [15] [14].

Incoherence and Measurement Matrices

Another significant result of compressed sensing is that there are generic sampling matrices: C that are sufficiently incoherent with respect to nearly all transform bases. Specifically, Bernouli and Gaussian random measurement matrices satisfy the RP for a generic basis Ψ with high porbability [113]. There are additional results generalizing the RIP and investigning incoherence of sparse matrices [205].

In many engineering applications, it is advantageous to represent the signal x in a generic basis, such as Fourier or wavelet. One key advantage is that single-point measurements are incoherent with respect to these bases, exciting a broadbaud frequency response. Sampling at random point leactions is appealing in applications where individual measurements are expensive, such as in occam monitoring. Examples of random measurement matrices, includure single role. Gaussian, Bernoulli, and source random, are shown in Firs. 3.10.

A particularly useful transform basis for compressed sensing is obtained by the SVD⁴, resulting in a tailored basis in which the data is optimally sparse [316, 80, 81, 31, 98]. A truncated SVD basis may result in a more efficient signal recovery from fewer measurements. Provers has been unde developing a compressed SVD and PCA based on the

⁴ The SVD provides an optimal low-rank matrix approximation, and it is used in principal components analysis (PCA) and resource theorem decomposition (POD).



Figure 3.10 Examples of good random measurement matrices C.

Johnson-Lindenstrauss (JL) lemma [267, 187, 436, 206]. The JL lemma is closely related to the RIP, indicating when it is possible to embed high-dimensional vectors in a lowdimensional space while preserving spectral properties.

Bad Measurements

So far we have described how to take good compressed measurements. Fig. 3.11 shows a particularly goor choice of measurements C, corresponding to the tast ρ columns of the sparsifying basis Ψ . In this case, the product $\Theta = C\Psi$ is a $\rho \times \rho$ identity matrix gaded with zeros on the left. In this case, any signal s that is not active in the tast ρ columns of Ψ is in the null-space of Θ , and is completely invisible to the measurements γ . In this case, these measurements incur similarity information to show the column of events.

3.5 Sparse Regression

The use of the ℓ_1 norm to promote sparsity significantly produce compresed sensing. In fact, many benefits of the \ell_1 norm were well-known and of-used in statistic decades calific. In this section, we show that the ℓ_1 norm may be used to regularize statistical regression, both to penalize statistical califies and also be promote parainomicon statistical models with as few factors as possible. The role of ℓ_2 versus ℓ_1 in regression is further dealied in Chapter 4.

Outlier Rejection and Robustness

Least squares regression is perhaps the most common statistical model used for data fitting. However, it is well known that the regression fit may be arbitrarily corrupted by a single large outlier in the data; outliers are weighted more heavily in least-squares regression because their distance from the liftnie is sourced. This is shown schematically in Fig. 31.2.

In contrast, ℓ_1 -minimum solutions give equal weight to all data points, making it potentially more robust to outliers and corrupt data. This procedure is also known as least absolute deviations (LAD) terrestoin, among other names. A script demonstration the



Figure 3.11 Examples of a bad measurement matrix C.

use of least-squares (ℓ_2) and LAD (ℓ_1) regression for a dataset with an outlier is given in Code 3.3.

Code 3.3 Use of l1 norm for robust statistical regression.

```
s = ser(+(rand(z,z)-z)) + Random dast from (-2,2)
b = .9ee - 1-remod(siss(z)) + Random dast from (-2,2)
strum = N(z) = 1 - Rand-remotion (-2,2)
h(rand) = .5;
= Random dast from (-2,2) = Random dast
= Random dast from (-2,2) = Random dast from (-2,2) = Random dast
= Random dast from (-2,2) = Random dast from (-2,2)
```



Figure 3.12 Least-squares regression is sensitive to outliers (red), while minimum ℓ_1 -norm regression is robust to outliers (blue).

Feature Selection and LASSO Regression

Interpretability is important in statistical models, as these models are often communicated to a non-technical audience, including business leaders and policy makers. Generally, a regression model is more interpretable if it has fewer terms that bear on the outcome, motivating yet another perspective on sparsity.

The bars houses shufting and selection operator (LASOS) is at (a possible preprint on testinging the black model complexity on physical (S13). This is not testinging the black model complexity on physical (S13). This is a possible complexity on physical (S13) and S13 a

Given a number of observations of the predictors and outcomes of a system, arranged as rows of a matrix A and a vector b, respectively, regression seeks to find the relationship between the columns of A that is most consistent with the outcomes in b. Mathematically, this may be written as:

$$Ax = b.$$
 (3.14)

Least-squares regression will tend to result in a vector x that has nonzero coefficients for all entries, indicating that all cohumns of A must be used to predict b. However, we often believe that the statistical model should be *simpler*, indicating that x may be sparse. The LASSO adds an *t*₁ penalty term to *regularize* the least-squares regression problem; i.e., to prevent overfitting:

$$\mathbf{x} = \underset{\mathbf{x}'}{\operatorname{argmin}} \|\mathbf{A}\mathbf{x}' - \mathbf{b}\|_2 + \lambda \|\mathbf{x}\|_1.$$
 (3.15)

Typically, the parameter λ is varied through a range of values and the fit is unfiltered against a test set of boldout data. If there is not enough due to have a sufficiently large training and test set, it is common to repeatedly train and test the model on random selection of the data (often 849) for training and 20% for testing), reading an evaluation of the formance. This cross-validation procedure enables the selection of a parsimonious model that has relatively for uterms and avoid overfitting.

Many statistical systems are overdetermined, as there are more observations than candid perdetices. This is not possible to use standard composed sensing, as measurement noise will pursance than to exact sparse solution cuists that minimize [Ax = b]_2. There is a statistical problem, making it as the sparse of the statistical problem, making a to explain the sparsity-promoting nature of the \ell₁ norm was presented in Thubizmi's 1996 perept [518].

LASSO regression is frequently used to build statistical models for disease, such as cancer and heart failure, since there are many possible predictors, including mompaphics, fifestyle, biometrics and genetic information. Thus, LASSO represents a clever version of the knhwn-sind approach, whereby meanly all possible predictive information is thrown into the mix, and afterwards these are then sifted and sieved through for the truly relevant redictors.

As a simple example, we consider an artificial data set consisting of 100 observations of an outcome, arranged in a vector $\mathbf{b} \in \mathbb{R}^{300}$. Each outcome in **b** is given by a combination of exactly 2 out of 10 candidate predictors, whose observations are arranged in the rows of a matrix $\mathbf{A} \in \mathbb{R}^{100\times101}$.

The vector x is sparse by construction, with only two nonzero entries, and we also add noise to the observations in b. The least-squares regression is:

```
>>xL2 = pinv(A)*b
xL2 = -0.0232
-0.3395
-0.3595
-0.1777
0.2912
-0.0525
-1.2720
-0.0413
-0.0413
-0.0500
```

Note that all coefficients are nonzero.

Implementing the LASSO, with 10-fold cross-validation, is a single straightforward command in MATLAB:

[XL1 FitInfo] = lasso(A,b,'CV',10);

The lasso command sweeps through a range of values for λ_i and the resulting x are each stored as columns of the matrix in XLI. To select the most parsimonious model that describes the data while avoiding overfitting, we may plot the cross-validated error as a function of λ_i as in Figs. 3.13:

lassoPlot(XL1,FitInfo,'PlotType','CV')

The green point is at the value of λ that minimizes the cross-validated mean-square error, and the blue point is at the minimum cross-validated error plus one standard deviation. The resultine model is found via FitInfo.IndexISE:





Figure 313 Output of lassoPlot command to visualize cross-validated mean-squared error (MSE) as a function of λ .

Note that the resulting model is sparse and the correct terms are active. However, the regression values for these terms are not accurate, and so it may be necessary to *de-bias* the LASSO by applying a final least-squares regression to the nonzero coefficients identified:

```
>>xLlDeBiazed = pinv(A(:,abz(xL1)>0))*b
xLlDeBiazed = 1.0980
```

3.6 Sparse Representation

Implicit no car discussion on sparshy is the fact that when high-dimensional signals exhibit boo-dimensional structure, they admit any sparse representation in an appropriate basis or dictionary. In addition to a signal being sparse in an SVD or Fourier basis, it may also be sparse in an overcomplete dictionary whose columns consist of the training data itself. In essence, in addition to a set signal being sparse in generic feature Bheary U from the SVD, $X = U \nabla V$, it may also have a sparse representation in the dictionary X

Wright et al. [560] demonstrated the power of sparse representation in a dictionary of test signals for robust classification of human faces, despite significant noise and occlusions. The so-called sparse representation for classification (SRC) has been widely used in image processing, and more recently to classify dynamical regimes in nonlinear differential equations (98, 433, 191, 308).

Code 3.4 Load Yale faces data and build training and test sets.



Figure 3.14 Schematic overview of sparse representation for classification.



Code 3.5 Downsample training images to build O library.

```
H = mise(Train,2);
Torka = serce(100,M);
for kait#
temponal is inversise(sepp.[12:10],'lancrost');
Thista(i,k) = reakape(tempional,100,1);
ad
for kait# # Normalize columns of Theta
(Theta(i,k) = Theta(i,k)/norm(Theta(i,k));
```



Figure 3.15 Sparse representation for classification demonstrated using a library of faces. A clean test image is correctly identified as the 7th person in the library.



Figure 3.18 Sparse representation for classification demonstrated on example face from person #7 occluded by a fake mustache.



Figure 3.17 Sparse representation for classification demonstrated on example image with 30% occluded pixels (randomly chosen and uniformly distributed).



Figure 3.18 Sparse representation for classification demonstrated on example with white noise added to image.

Code 3.6 Build test images and downsample to obtain y.

Code 17 Search for sparse representation of test image. The same code is used for each of the test images y1 through y4.

```
gr = f(1);
gr = f(1);
wrisial = 10(); program vector of coefficients
undget to (1, 1);
andget to (1, 1);
andget (1, 1);
angget (compared for a second for a second for a second
grant (1, 1);
angget (compared for a second for a second for a second
grant (1, 1);
angget (compared for a second for a second for a second
grant (1, 1);
angget (compared for a second for a second for a second
grant (1, 1);
angget (compared for a second for a second for a second
grant (1, 1);
angget (compared for a second for a second for a second for a second
grant (1, 1);
angget (compared for a second fo
```

3.7 Robust Principal Component Analysis (RPCA)

As mentioned earlier in Section 3.5, least-squares regression models are highly susceptible to outliers and converged data. Principal component analysis (PCA) suffers from the same weakness, making it /ngs/w with respect to outliers. To ameliorate this sensitivity, Candes et al. [110] have eleveloped a robust principal component analysis (PPCA) that seeks to decompose a data matrix **X** into a structured low-rank matrix **L** and a sparse matrix **S** containing outliers and corrupt data.

$$K = L + S.$$
 (3.16)

The principal components of L are robust to the outliers and cerupt data in S. This decomposition has profound implications for many modern problems of interest, including video surveillance (where the background objects appear in L and foreground objects appear in S), face recognition (eigenfaces are in L and shadows, occlasions, etc. are in S), natural language processing and latent semantic indexing, and ranking problems².

Mathematically, the goal is to find L and S that satisfy the following:

$$\min_{\mathbf{L},\mathbf{S}} \operatorname{rank}(\mathbf{L}) + \|\mathbf{S}\|_0 \text{ subject to } \mathbf{L} + \mathbf{S} = \mathbf{X}. \quad (3.17)$$

However, neither the rank(L) nor the $||S||_0$ terms are convex, and this is not a tractable optimization problem. Similar to the compressed sensing problem, it is possible to solve for the optimal L and S with high probability using a convex relaxation of (3.17):

$$\min_{\mathbf{L},\mathbf{S}} \|\mathbf{L}\|_{*} + \lambda \|\mathbf{S}\|_{1} \text{ subject to } \mathbf{L} + \mathbf{S} = \mathbf{X}. \quad (3.18)$$

Here, $\|\cdot\|_{n}$ denotes the nuclear norm, given by the sum of singular values, which is a proxy for rank. Remarkably, the solution to (3.18) converges to the solution of (3.17) with high probability if $\lambda = 1/\sqrt{\max(n, m)}$, where n and m are the dimensions of X, given that L and S satisfy the following conditions:

- 1. L is not sparse
- S is not low-rank; we assume that the entries are randomly distributed so that they do not have low-dimensional column space.

The convex problem in (3.17) is known as principal component pursuit (PCP), and may be solved using the augmented Lagrangia multiplier (ALM) algorithm. Specifically, an augmented Lagrangia may be constructed:

$$\mathcal{L}(\mathbf{L}, \mathbf{S}, \mathbf{Y}) = \|\mathbf{L}\|_{*} + \lambda \|\mathbf{S}\|_{1} + \langle \mathbf{Y}, \mathbf{X} - \mathbf{L} - \mathbf{S} \rangle + \frac{\mu}{2} \|\mathbf{X} - \mathbf{L} - \mathbf{S}\|_{F}^{2}.$$
 (3.19)

A general solution would solve for the L_4 and S_4 that minimize \mathcal{L} , update the Lagrange multipliers $\mathbf{Y}_{k+1} = \mathbf{Y}_k + \mu(\mathbf{X} - \mathbf{L}_k - \mathbf{S}_k)$, and iterate until the solution converges. However, for this specific system, the alternating directions method (ADM) [337, 566] provides a simple procedure to find L and S.

First, a shrinkage operator $S_t(x) = sign(x) max(|x| - t, 0)$ is constructed (MATLAB function shrink below):

```
function out = shrink(X,tau)
    out = sign(X).*max(abs(X)-tau,0);
end
```

Next, the singular value threshold operator $SVT_{\tau}(X) = US_{\tau}(\Sigma)V^{s}$ is constructed (MAT-LAB function SVT below):

```
function out = SVT(X,tau)
[U,S,V] = svd(X,'econ');
out = U+shrink(S,tau)+V';
end
```

⁵ The ranking problem may be thought of in terms of the Netflix prize for matrix completion. In the Netflix prize, a large matrix of preferences is constructed, with rows corresponding to users and columns

corresponding to movies. This matrix is sparse, as most users only rate a handful of movies. The Netflix prize socks to accurately fill in the missing entries of the matrix, revealing the likely user rating for movies the user has not seen.

Finally, it is possible to use S_T and SVT operators iteratively to solve for L and S:

Code 3.8 RPCA usine alternatine directions method (ADM).

```
 \begin{split} & \text{Restriction} \left\{ \begin{array}{ll} \text{Restriction} & \text{RES}(1) \\ \text{Restriction} & \text{Restriction} \\ \text{Restriction}
```

This is demonstrated on the eigenface example with the following code:

load allFaces.mat
X = faces(:,l:nfaces(1));
[L.S] = RPCA(X);

In this example, the original columns of \mathbf{X}_{n} along with the low-mark and sparse components, are shown in Fig. 3.19. Notech that in this example, RPCA effectively fills in cocluded regions of the image, corresponding to shadows. In the low-mark component such that the state of the



Figure 3.19 Output of RPCA for images in the Yale B database.

3.8 Sparse Sensor Placement

Until now, we have investigated signal reconstruction in a generic basis, such as Fourier or wavelets, with random measurements: This provides considerable flexibility, as no prior structure is assumed, except that the signal is space in a known basis. For example, conpresed ensity works equally well for reconstructing an image of a mominia, rate, or a opt ordine. However, if we known that will be reconstructing an human face, we can dramatically reduce the number of sensor required for reconstruction or classification by optimizing sensors for a particular forms in leave $\mathbf{v}_{i} = \mathbf{0}$ that from the SVD.

Thus, it is possible to design autored sensors for a particular library, as contrast to the previous approach of modes mesors in a gravite library. Nass explaints arent locations may be obtained using fast preedy procedures that scale well with large signal dimension, et al. [186] descritation. The following discussion will coloridy fibre Mandard et al. [186] and B. Bussons et al. [196], and de reader is encouraged to find some detail. The Similar approach will be used for deficient sampling of modes/scales models following for scasses and actuators phenoments in control [165], based on the balancing tamfindancing discussion [Contert 9].

Optimizing sensor locations is important for nearly all downstream tasks, including consideration, prediction, estimation, nonelleng, and control. Nonever, identifying optimal locations involves a brate force search through the combinatorial choices of p reasors of p possible contoins in space. Record grade and the search through semiciple control is search and the second force states of the second second second second second factor states estimation for the statescy, biochardwide control.

Sparse Sensor Placement for Reconstruction

The goal of optimized sensor placement in a tailored library $\Psi_r \in \mathbb{R}^{n \times r}$ is to design a sparse measurement matrix $\mathbf{C} \in \mathbb{R}^{p \times n}$, so that inversion of the linear system of equations

$$y = C\Psi_{f}a = \theta a$$
 (3.20)

is a well-conditioned as possible. In other words, we will design C to minimize the condition number of $C_{ij} = \emptyset_{ij}$ on that may be inverted to identify the low-rank coefficience a piren noisy measurements γ . The condition number of a matrix θ is the ratio of its maximum and minimizing and values. Indicating how screening matrix θ is the ratio of the invertigation of the strength o

$$\theta(\mathbf{a} + \epsilon_{\mathbf{a}}) = \sigma_{\min}\mathbf{a} + \sigma_{\max}\epsilon_{\mathbf{a}}.$$
 (3.21)

Thus, the signal-to-noise ratio decreases by the condition number after mapping through θ . We therefore seek to minimize the condition number through a principled choice of C. This is shown schematically in Fig. 3.20 for p = r.

When the number of sensors is equal to the rank of the library, i.e. p = r, then θ is a square matrix, and we are choosing C to make this matrix as well-conditioned for inversion



Figure 3.20 Least squares with r sparse sensors provides a unique solution to a, hence x. Reproduced with permission from Manohar et al. [366].

a possible. When p > r, we see the support the conflict of $M = P^2$, which is interval to the post-interval is in the post-ion terms of a post-ion terms of a post-ion terms of a post-ion term of post-ion term of a post

Random sensors: In general, randomly placed sensors may be used to estimate mode coefficients a However, then p = n and the number of sensors is equal to the number of modes, the condition number is typically very large. In fact, the matrix Θ is often numerically singular and the condition number is not 10^{10} . Voreampling, as in Sec. 13, rapidly improves the condition number, and even p = r + 10 usually has much better reconstruction performance.

QR Proteing for sparse sensors. The greedy matrix QR factorization with column price of α of α , exploring to physical price of the physical provides a particularly simple and effective sensor optimization. The QR pixeling method is fast, adapted to implement and physical energy optimil arrows tailout on a specific SVDFPOD basis. SPArenet matrix is by the physical for any column physical physical computing librarias, including Matther for the relative set obtained. The reduced matrix QR fastorization with column protong decomposes a matrix A c filewise into a unimary matrix Q, an upper strangular matrix R and a column permutations matrix C^2 such that $AC^2 = QR$. The pivoting procedure protokes an approximate greacy solution method to maintize the matrix volume, which is the absolute value of the determinant QR column protog increments the volume of the submatrix constructed from the pivoted columns by selecting a new pivot column with maximal 2-norm, then subtracting from every other columns in contengonal protocien onothe pivot column.

Thus QR factorization with column pivoting yields r point sensors (pivots) that best sample the r basis modes Ψ_r

$$\Psi_{i}^{T}C^{T} = QR.$$
 (3.22)

Based on the same principle of pivoted QR, which controls the condition number by minimizing the matrix volume, the oversampled case is handled by the pivoted QR factorization of Ψ, Ψ .

$$(\Psi_r \Psi_r^T)C^T = QR.$$
 (3.23)

The code for handling both cases is give by

```
if (pest) % QE sensor selection, per
(D,R,pivel) = q(?Ni._','vector');
=lasi(por) % Oversampled QE sensors, per
(D,R,pivel) = qc(Pai_r*Pai_r*/vector');
end
C = secos(p,n);
for jsip
C(j,pivet(j))=1;
```

Example: Reconstructing a Face with Sparse Sensors

To demonstrate the concept of signal reconstruction in a tailored basis, we will design optimized sparse sensors in the library of eigenfaces from Section 1.6. Fig. 3.21 shows the QR sensor placement and reconstruction, along with the reconstruction using random sensors. We use p = 100 sensors in a r = 100 mode library. This code assumes that



Figure 3.21 (left) Original image and p = 100 QR sensors locations in a r = 100 mode library. (middle) Reconstruction with OR sensors. (right) Reconstruction with random sensors.



Figure 3.22 Schematic illustrating SVD for feature extraction, followed by LDA for the automatic classification of data into two categories A and B. Reproduced with permission from Bai et al. [29].

the faces have been loaded and the singular vectors are in a matrix U. Optimized QR sensors result in a more accurate reconstruction, with about three times less reconstruction error. In addition, the condition number is orders of magnitude smaller than with random sensors. Both QR and random sensors may be improved by oversampling. The following code computes the QR sensors and the approximate reconstruction from these sensors.

```
i = 100 p = 100 i # # of modes r, # of semence p
pist = U(1)i()
r = U(1)i()
r = (1) pist()
r = r =
```

Sparse Classification

For image classification, even fewer sensors may be required than for reconstruction. For example, space sensors may be selected that contain the most discriminating information to characterize two categories of data [89]. Given a library of s SVD modes Ψ_{c} , it is done possible to identify a vector w e^{2} is this subspace that maximulty distinguishes, between two categories of data, as described in Section 5.6 and showa in Fig. 322. Space are found to the subspace to the single set of the single devices, projecting and all device information, are found by:

$$\mathbf{s} = \underset{\mathbf{s}'}{\operatorname{argmin}} \|\mathbf{s}'\|_1 \quad \text{subject to} \quad \Psi_{\mathbf{s}'}^T \mathbf{s}' = \mathbf{w}.$$
 (3.24)

This sparse sensor placement optimization for classification (SSPOC) is shown in Fig. 3.23 for an example classifying dogs versus casts. The library Ψ_{c} contains the they of opportra and the vector w identifies the key differences between dogs and cars. Note that this vector does not care about the degrees of freedom that characterize the various features with the dog or car clusters, but rather only the differences between the two outgoestics. Optimized sensors are aligned with regions of interest, such as the syste, nose, moth, and east.



Figure 3.23 Sparse sensor placement optimization for classification (SSPOC) illustrated for optimizing sensors to classify dogs and cats. Reproduced with permission from B. Branton et al. (89).

Suggested Reading

Papers and Reviews

- Regression shrinkage and selection via the lasso, by R. Tibshirani, Journal of the Royal Statistical Society B, 1996 (518).
- (2) Robust uncertainty principles: exact signal reconstruction from highly incomplete frequency information, by E. J. Cambes, J. Romberg, and T. Tao, *IEEE Trans*actions on Automatic Control, 2006 [111].
- (3) Compressed sensing, by D. L. Donoho, IEEE Transactions on Information Theory, 2006 [150].
- (4) Compressive sensing, by R. G. Baraniuk, IEEE Signal Processing Magazine, 2007 [39].
- (5) Robust face recognition via sparse representation, by J. Wright, A. Yang, A. Ganesh, S. Sastry, and Y. Ma, IEEE Transactions on Pattern Analysis and Machine Intelligence, 2009 [560].
- (6) Robust principal component analysis?, by E. J. Candès, X. Li, Y. Ma, and J. Wright, Journal of the ACM, 2011 [110].
- (7) Signal recovery from random measurements via orthogonal matching pursuit, by J. A. Tropp and A. C. Gilbert, *IEEE Transactions on Information Theory*, 2007 [529].
- (8) Data-driven sparse sensor placement, by K. Manohar, B. W. Brunton, J. N. Kutz, and S. L. Brunton, *IEEE Control Systems Magazine*, 2018 [366].

Part II

Machine Learning and Data Analysis

All of machine learning revolves around optimization. This includes regression and model selection frameworks that aim to provide parsimonious and interpretable models for data [266]. Curve fitting is the most basic of regression techniques, with polynomial and evonential fitting resulting in solutions that come from solvine the linear system.

$$Ax = b$$
. (4.1)

When the model is not prescribed, then optimization methods are used to select the best model. This changes the underlying mathematics for function fitting to either an overdetermined or underdetermined optimization problem for linear systems given by:

$$\operatorname{argmin}(\|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2 + \lambda g(\mathbf{x}))$$
 or (4.2a)

argmin
$$g(\mathbf{x})$$
 subject to $\|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2 \le \epsilon$, (4.2b)

where g(x) is a given penalization (with penalty parameter λ . for overdetermined systems), for over and underetermined linear systems of quanticos, which result in either in solutions or an infinite number of solutions of (4.1), a choice of constraint or penalty, which is also known as regularization, must be made in order to produce a solution. For instance, one chan entry z_1 are minined by the solution of the solution of the solution of the solution models, then the overell mathematical solution of the solution of the solution of the solution models, then the overell mathematical solution of the models, then the overell mathematical intersorts that be the more enterial form

$$\operatorname{argmin}(f(\mathbf{A}, \mathbf{x}, \mathbf{b}) + \lambda g(\mathbf{x}))$$
 or (4.3a)

argmin
$$g(\mathbf{x})$$
 subject to $f(\mathbf{A}, \mathbf{x}, \mathbf{b}) \le \epsilon$ (4.3b)

which are often solved using gradient descent algorithms. Indeed, this general framework is also at the center of deep learning algorithms.

In addition to optimization straingings, a central concern in data science is understanding in proposed model have or for or under of the data. These coss-subdates multiple and the data science is additional to the science in additional to the science in additional to the science in additional to the science is additional to the science in additional to the science is additional to the scie



Figure 4.1 Prototypical behavior of over- and under-fitting of data. (a) For over-fitting, increasing the model complexity or training epochs (iterations) leads to improved reduction of error on training data while increasing the error on the withheld data. (b) For under-fitting, the error performance is limited due to rostrictions on model complexity. These canonical graphs are ubiquitous in data science and of arramount invortunce, when evaluation a model.

the ability to achieve a good model as shown in Fig. 4.10b. However, it is not always clear if you are under-fining or if the model can be improved. Cross-validation is of such paramount importance that it is automatically included in most machine learning algorithms in MATLAB. Importantly, the following mantra holds: if you don't cross-validate, you is donth.

The net few chapters will coultine how optimizations and cross-validation arise in partice, and will adjuffin the choices that need to be made in applying meaningful constraints and structure to g(t) so as to achieve interpretable solutions. Indeed, the objective (loss) instruction (r) can arguitarization g(r) are equally important in determining comparisonal transition generations strategies. Other times, proxy four and regularization functions are choices determining the objective (loss) and the objective (loss) choices determining the objective (loss) and a strategies of the objective (loss).

4.1 Classic Curve Fitting

Core, first, or os of the more basic and foundational tools in data science. Them on effect shortward aregorizations in the englishering analysis discretes. Subscience 1995, bronnel first, was absociated for understanding the daminant trends in the large the science basic and the science basic and the 1993, with a semidar control first and the science basic applications are science of the science basic applications are science of the science basic applications are science basic and context. This one can argue that data science basic prevales by planets and cortexts. This one can argue that data science basic planets are science basic and the science basic applications are science by the science basic and the science basic application are science basic applications are science basic applications are science basic applications and the science basic applications are science basic applicating applic

A broader mathematical viewpoint of curve fitting, which we will advocate throughout this text, is *regression*. Like curve fitting, regression attempts to estimate the relationship among variables using a variety of statistical tools. Specifically, one can consider the general relationship between independent variables X, dependent variables Y, and some unknown parameters β :

$$Y = f(X, \beta)$$
 (4.4)

where he regression function *f*(*i*) is typically prescribed and the parameters *J* are found by optimizing the geodence-off-of this function to data. In bat follows, we will consider curve fitting as a special case of regression. Importantly, regression and curve fitting dicover relationships among variables by optimization. Broody pergission, machine learning is framed around regression techniques, which are themselves framed around optimization bedown dues the association of the structure of the structure

Least-Squares Fitting Methods

To illustrate the concepts of regression, we will consider classic least-squarespolynomial fitting for characterizing trends in data. The concept is straightforward and simple: use a simple function to describe a trend by minimizing the sum-square error between the selected function $f(\cdot)$ and its fit to the data. As we show here, classical curve fitting is formulated as a simple solution of $Ax = \mathbf{b}$.

Consider a set of n data points

$$(x_1, y_1), (x_2, y_2), (x_3, y_3), \dots, (x_n, y_n).$$
 (4.5)

Further, assume that we would like to find a best fit line through these points. We can approximate the line by the function

$$f(x) = \beta_1 x + \beta_2$$
 (4.6)

where the constants β_1 and β_2 , which are the parameters of the vector β of (4.4), are chosen to minimize some error associated with the fit. The line fit gives the *linear regression* model $\mathbf{Y} = f(\mathbf{A}, \beta) = \beta_1 \mathbf{X} + \beta_2$. Thus the function gives a linear model which approximates the data, with the approximation error at each point given by

$$f(x_{k}) = y_{k} + E_{k}$$
 (4.7)

where y_k is the true value of the data and E_k is the error of the fit from this value.

Various error metrics can be minimized when approximating with a given function f(x). The choice of error metric, or norm, used to compute a goodness-of-fit will be critical in this chapter. Three standard possibilities are often considered which are associated with the $t \sim (hear-sources), t = and t \sim norms.$ These are defined as follows:

$$E_{\infty}(f) = \max_{1 \le k \le n} |f(x_k) - y_k|$$
 Maximum Error (ℓ_{∞}) (4.8a)

$$E_1(f) = \frac{1}{\pi} \sum_{k=1}^{n} |f(x_k) - y_k|$$
 Mean Absolute Error (ℓ_1) (4.8b)

$$E_2(f) = \left(\frac{1}{\pi}\sum_{k=1}^{n} |f(x_k) - y_k|^2\right)^{1/2}$$
 Least-squares Error (ℓ_2) . (4.8c)



Figure 4.2 Line fits for the three different error metrics E_{∞} , E_1 and E_2 . In (a), the data has not outliers and the three linear models, although different, produce approximately the same model. With outliers, (b) shows that the predictions are significantly different.

Such regression error metrics have been previously considered in Chapter 1, but they will be considered once again here in the framework of model selection. In addition to the above norms, one can more broadly consider the error based on the ℓ_o -norm

$$E_{\rho}(f) = \left(\frac{1}{n}\sum_{k=1}^{n} |f(x_k) - y_k|^{\rho}\right)^{1/\rho}$$
. (4.9)

For different values of p, the best fit line will be different. In most cases, the differences are small. However, when there are outliers in the data, the choice of norm can have a similicant innext.

When fitting a curve to a set of data, the root-mean square (RMS) error (4.82) is often chosen to be minimate. This is called *Last-squares*, for $f_{\rm eff} = 2.4$ explicit three line firs that minimize the errors $E_{\rm eff}$, $f_{\rm eff}$ lasted previously. The $E_{\rm eff}$ are line fit is strongly dimensional by the call point which due to in it the error $H_{\rm eff}$ and $E_{\rm eff}$ line the indep the data has no outliers. The linear models for these three error metrics are commuted using MATLAR¹. Minimearch communit The Core for all three in given as follows:

Code 4.1 Regression for linear fit

```
# The data
xx[1 2 3 4 5 6 7 8 9 10]
yx[0.2 0.5 0.3 3.5 1.0 1.5 1.8 2.0 2.3 2.2]
plefminsearch('fit1', [1 1], [],x,y);
```

For each error metric, the computation of the error metrics (4.8) must be computed. The fminsearch command requires that the objective function for minimization be given. For the three error metrics considered, this results in the following set of functions for fminsearch:

Code 4.2 Maximum error t_...

```
function Esfit1(x0,x,y)
Esmax(abs( x0(1)*x+x0(2)-y ));
```

Code 4.3 Sum of absolute error l1.

function Esfit2(x0,x,y) Essum(abs(x0(1)*x+x0(2)-y));

Code 4.4 Least-squares error (2.

function E=fit3(x0,x,y) E=sum(abs(x0(1)*x+x0(2)-y).^2);

Finally, for the outlier data, an additional point is added to the data in order to help illustrate the influence of the error metrics on producing a linear repression model.

Code 4.5 Data which includes an outlier.

x= [1 2 3 4 5 6 7 8 9 10] y= [0.2 0.5 0.3 0.7 1.0 1.5 1.8 2.0 2.3 2.2]

Least-Squares Line

Least-squares fitting to linear models has critical advantages over other norms and metrics. Specifically, the optimization is inexpensive, since the error can be computed analytically. To show this explicitly, consider applying the least-square fit criteria to the data points (α_{i}, γ_{i}) where $k = 1, 2, 3, \cdots, n$. To fit the curve

$$f(x) = \beta_1 x + \beta_2$$
 (4.10)

to this data, the error E_2 is found by minimizing the sum

$$E_2(f) = \sum_{k=1}^{n} |f(x_k) - y_k|^2 = \sum_{k=1}^{n} (\beta_1 x_k + \beta_2 - y_k)^2.$$
 (4.11)

Minimizing this sum requires differentiation. Specifically, the constants β_1 and β_2 are chosen so that a minimum occurs. Thus we require: $\partial E_2/\partial \beta_1 = 0$ and $\partial E_2/\partial \beta_2 = 0$. Note that athrough a zero derivative can indicate either a minimum or maximum, we know
this must be a minimum of the error since there is no maximum error, i.e. we can always choose a line that has a larger error. The minimization condition gives:

$$\frac{\partial E_2}{\partial \beta_1} = 0$$
: $\sum_{k=1}^{n} 2(\beta_1 x_k + \beta_2 - y_k)x_k = 0$ (4.12a)

$$\frac{\partial E_2}{\partial \beta_2} = 0$$
: $\sum_{k=1}^{n} 2(\beta_1 x_k + \beta_2 - y_k) = 0$. (4.12b)

Upon rearranging, a 2 × 2 system of linear equations is found for A and B

$$\begin{pmatrix} \sum_{k=1}^{n} x_k^2 & \sum_{k=1}^{n} x_k \\ \sum_{k=1}^{n} x_k & n \end{pmatrix} \begin{pmatrix} \beta_1 \\ \beta_2 \end{pmatrix} = \begin{pmatrix} \sum_{k=1}^{n} x_k y_k \\ \sum_{k=1}^{n} y_k \end{pmatrix} \longrightarrow \mathbf{A}\mathbf{x} = \mathbf{b}. \quad (4.13)$$

This linear system of equations can be solved using the backslash command in MATLAB. Thus an optimization procedure is unnecessary since the solution is computed exactly from a 2×2 matrix.

This method can be easily generalized to higher polynomial fits. In particular, a parabolic fit to a set of data requires the fitting function

$$f(x) = \beta_1 x^2 + \beta_2 x + \beta_3 \qquad (4.14)$$

where now the three constants β_1 , β_2 , and β_3 must be found. These can be solved for with the 3 × 3 system resulting from minimizing the error $E_2(\beta_1, \beta_2, \beta_3)$ by taking

$$\frac{\partial E_2}{\partial \beta_1} = 0$$
 (4.15a)

$$\frac{\partial E_2}{\partial \beta_2} = 0$$
 (4.15b)

$$\frac{\partial E_2}{\partial B_1} = 0.$$
 (4.15c)

In fact, any polynomial fit of degree k will yield a $(k + 1) \times (k + 1)$ linear system of equations Ax = b whose solution can be found.

Data Linearization

Although a powerful method, the minimization procedure for general fitting of arbitrary functions results in equations which are nontrivial to solve. Specifically, consider fitting data to the exponential function

$$f(x) = \beta_2 \exp(\beta_1 x)$$
. (4.16)

The error to be minimized is

$$E_2(\beta_1, \beta_2) = \sum_{k=1}^{n} (\beta_2 \exp(\beta_1 x_k) - y_k)^2.$$
 (4.17)

Applying the minimizing conditions leads to

$$\frac{\partial E_2}{\partial \beta_1} = 0$$
: $\sum_{k=1}^{n} 2(\beta_2 \exp(\beta_1 x_k) - y_k)\beta_2 x_k \exp(\beta_1 x_k) = 0$ (4.18a)

$$\frac{\partial E_2}{\partial \beta_2} = 0$$
: $\sum_{k=1}^{n} 2(\beta_2 \exp(\beta_1 x_k) - y_k) \exp(\beta_1 x_k) = 0.$ (4.18b)

This in turn leads to the 2 × 2 system

$$\beta_2 \sum_{k=1}^{n} x_k \exp(2\beta_1 x_k) - \sum_{k=1}^{n} x_k y_k \exp(\beta_1 x_k) = 0$$
 (4.19a)

$$\beta_2 \sum_{k=1}^{n} \exp(2\beta_1 x_k) - \sum_{k=1}^{n} y_k \exp(\beta_1 x_k) = 0.$$
 (4.19b)

This system of equations is nonlinear and cannot be solved in a straightforward fashion. Indeed, a solution may not even exist. Or many solution may exist. Section 4.2 describes a possible iterative procedure, called gradient descent, for solving this nonlinear system of equations.

To avoid the difficulty of solving this nonlinear system, the exponential fit can be linearized by the transformation

$$Y = \ln(y)$$
 (4.20a)

$$\beta_3 = \ln \beta_2$$
. (4.20c)

Then the fit function

$$f(x) = y = \beta_2 \exp(\beta_1 x)$$
 (4.21)

can be linearized by taking the natural log of both sides so that

$$\ln y = \ln(\beta_2 \exp(\beta_1 x)) = \ln \beta_2 + \ln(\exp(\beta_1 x)) = \beta_3 + \beta_1 x \implies Y = \beta_1 X + \beta_3. \quad (4.22)$$

By fitting to the natural log of the y-data

$$(x_i, y_i) \rightarrow (x_i, \ln y_i) = (X_i, Y_i)$$
 (4.23)

the curve fit for the exponential function becomes a linear fitting problem which is easily handled. Thus, if a transform exists that linearizes the data, then standard polynomial fitting methods can be used to solve the resulting linear system Ax = b.

4.2 Nonlinear Regression and Gradient Descent

Polynomial and exponential curve fitting admit analytically tractable, beed if least-quares obtainess. However, such curve fitting tably specialized and none general mathematical framework is necessary for solving a broader set of problems. For instance, one may with to its a nonlinear function of the form $(1-p) \neq (n, onl)_{k} + p + l_{k}$ to a data set. Instead of nonlinear equations. The general theory of nonlinear regression assumes that the fitting function takes the general form.

$$f(x) = f(x, \beta)$$
 (4.24)

where the m < n fitting coefficients $\beta \in \mathbb{R}^{m}$ are used to minimize the error. The root-mean source error is then defined as

$$E_2(\beta) = \sum_{k=1}^{n} (f(x_k, \beta) - y_k)^2 \qquad (4.25)$$

which can be minimized by considering the $m \times m$ system generated from minimizing with respect to each parameter β_i

$$\frac{\partial E_2}{\partial \beta_j} = 0$$
 $j = 1, 2, \dots, m$. (4.26)

In general, this gives the nonlinear set of equations

$$\sum_{k=1}^{n} (f(x_k, \beta) - y_k) \frac{\partial f}{\partial \beta_j} = 0 \quad j = 1, 2, 3, \dots, m. \quad (4.27)$$

There are no general methods svalulable for solving such nonlinear systems. Indeed, nonliaear systems can have no solutions, severa in sinitations, ore ven an infinite number of solutions. Most attempts at solving nonlinear systems are based on iterative schemes which require a good initial geneses to converge to the global minimum error. Regardless, the general fitting procedure is straightforward and allows for the construction of a best for curve to much the data. It such a solution procedure, it is importive that an ascontheli initial general be provided for by the user. Otherwise, rapid convergence to the desired root may not be activesed.

Fig. 4.3 shows two example functions to be minimized. The first is a cover function (Fig. 4.3a). Cover functions are sided in the quarantes of covergence exits for may algorithm, and gradient descent can be tuned to perform exceptionally well for such functions. The second fluctures a monocover. Function and shows many of the typical problem associated with gradient descent, including the fact that the function hose molecules to all matters are also strated as the second strategies are difficult to strately compared to the initial conditions of the gradient sec difficult to strately compare. It is the initial conditions of the gradient descent distribution and the area many advances of the initial conditions of the gradient descent distribution.



Figure 43 Two objective function landscapes representing (a) a convex function and (b) a nonconvex function. Convex functions have many guarantees of convergence, while nonconvex functions have a variety of prihills that can finit the success of gradient docent. For nonconvex functions, local minima and an inhibity to compute gradient directions (derivatives that are near zero) make it challenting for our diminization.

around gradient descent for restarting and ensuring that one is not stuck in a local minima. Recent training algorithms for deep neural networks have greatly advanced gradient descent innovations. This will be further considered in Chapter 6 on neural networks.

Gradient Descent

For high-dimensional systems, we generalize the concept of a minimum or maximum, i.e. an extremum of a multi-dimensional function $f(\mathbf{x})$. At an extremum, the gradient must be zero, so that

$$\nabla f(\mathbf{x}) = 0$$
, (4.28)

Since saddles exist in higher-dimensional spaces, one must test if the extremum point is a minimum or maximum. The idea behind gradient descent, or steepest descent, is to use the derivative information as the basis of an iterative algorithm that progressively converges to a local minimum point of f(x).

To illustrate how to proceed in practice, consider the simple two-dimensional surface

$$f(x, y) = x^2 + 3y^2 \qquad (4.29)$$

which has a single minimum located at the origin (x, y) = 0. The gradient for this function is

$$\nabla f(\mathbf{x}) = \frac{\partial f}{\partial x} \hat{\mathbf{x}} + \frac{\partial f}{\partial y} \hat{\mathbf{y}} = 2x \hat{\mathbf{x}} + 6y \hat{\mathbf{y}}$$
 (4.30)

where \hat{x} and \hat{y} are unit vectors in the x and y directions, respectively.

Fig. 4.4 illustrates the gradient steepest descent algorithm. At the initial guess point, the gradient V(r) is computed. This gives the direction of network does minimum point of $f(x_0)$, i.e. the minimum is located in the direction gives by $-\varphi'(x_0)$. The point of the state of the state

$$\mathbf{x}_{k+1}(\delta) = \mathbf{x}_k - \delta \nabla f(\mathbf{x}_k) \qquad (4.31)$$

where the parameter 3 dicates how far to move along the gradient docent curve. This formula represents a generalization of a Norton method where the derivative is used to compute an update in the heration scheme. In gradient docectit, it is crucial to determine how much to step forward according to the computed gradient, so that the algorithm is always is going downkill in an optimal way. This requires the determination of the correct value of 3 in the algorithm.

To compute the value of δ , consider the construction of a new function

$$F(\delta) = f(\mathbf{x}_{l+1}(\delta))$$
 (4.32)

which must be minimized now as a function of δ . This is accomplished by computing $\partial F/\partial \delta = 0$. Thus one finds

$$\frac{\partial F}{\partial \delta} = -\nabla f(\mathbf{x}_{k+1})\nabla f(\mathbf{x}_k) = 0.$$
 (4.33)



Figure 4.4 Gradient descent algorithm applied to the function $f(x, y) = x^2 + 3y^2$. In the top panel, the contours are plotted for each successive value (x, y) in the iteration algorithm given the initial gradient. The bottom panel demonstrates the rapid convergence and error (E) to the minimum (optimal) solution.

The geometrical interpretation of this result is the following: $\nabla f(\mathbf{x}_k)$ is the gradient direction of the current iteration point and $\nabla f(\mathbf{x}_{k+1})$ is the gradient direction of the future point, thus δ is chosen so that the two gradient directions are orthogonal.

For the example given above with $f(x, y) = x^2 + 3y^2$, we can compute this conditions as follows:

$$\mathbf{x}_{k+1} = \mathbf{x}_k - \delta \nabla f(\mathbf{x}_k) = (1 - 2\delta)\mathbf{x} \ \mathbf{\dot{x}} + (1 - 6\delta)\mathbf{y} \ \mathbf{\dot{y}}.$$
 (4.34)

This expression is used to compute

$$F(\delta) = f(\mathbf{x}_{k+1}(\delta)) = (1 - 2\delta)^2 x^2 + 3(1 - 6\delta)^2 y^2$$

(4.35)

whereby its derivative with respect to δ gives

$$F'(\delta) = -4(1 - 2\delta)x^2 - 36(1 - 6\delta)y^2$$
. (4.36)

Setting $F'(\delta) = 0$ then gives

$$\delta = \frac{x^2 + 9y^2}{2x^2 + 54y^2}$$
(4.37)

as the optimal descent step length. Note that the length of δ is updated as the algorithm progresses. This gives us all the information necessary to perform the steepest descent search for the minimum of the eiven function.

```
Code 4.6 Gradient descent example.
```

```
 \begin{array}{l} (1) \leq j \ (1) \leq j
```

As is clearly evident, this descent search algorithm based on derivative information is similar to Newton's method for root finding both in on-dimension as used in a higherdimensions. Fig. 4.4 shows the rapid convergence to the minimum for this convex function. Moreover, the graniend secsoral algorithm is the core algorithm of advanced textures solvers such as the bi-conjugate gradient descent method (biogstab) and the generalized method of reidulus (grane) [200].

In the example above, the gradient could be computed analytically. More generally, given just data itself, the gradient can be computed with numerical algorithms. The gradient command can be used to compute local or global gradients. Fig. 4.5 shows the gradient terms $\partial f/\partial x$ and $\partial f/\partial y$ for the two functions shown in Fig. 4.3. The code used to produce these critical terms for the gradient descert algorithm is given by

[dfx,dfy] =gradient(f,dx,dy);

where the function $f(t, \gamma)$ is a two-dimensional function compared from a known function or directly from data. The output are marices containing the values of 2/3 ($x_{\rm M}$ and 3/2) over the discretized domain. The gradient dent the bused to approximate either local or global gradients to execute the gradient descent. The following code, whose results are shown in Fig. 4.6, uses the interp2 function to extract the values of the function and gradient of the function in Fig. 4.5(b).

Code 4.7 Gradient descent example using interpolation.

```
bilac();; y(1) y(2) y(1);
bilac(); y(1) y(1) y(1);
diputant(); y(1) y(1);
diputant();
```



Figure 4.5 Computation of the gradient for the two functions illustrated in Fig. 4.3. In the left panels, the gradient terms (a) $\delta f/\delta x$ and (c) $\delta f/\delta y$ are computed for Fig. 4.3(a), while the right panels compute these same terms for Fig. 4.3(b) in (b) and (d), respectively. The gradient command numerically generates the gradient.

In this code, the **fminsearch** command is used to find the correct value of δ . The function to optimize the size of the iterative step is given by

```
function mindel=delmearch(del,x,y,dfx,dfy,X,Y,F)
x0sx-del=dfy;
y0sy-del=dfy;
mindel=interp2(X,Y,F,x0,y0);
```

This discussion provides a rulimetrary introduction to gradient descent. A wide range of innovations have attempted to opeed up this dominant nonlinear optimization procedure, including alternating descent methods. Some of these will be discussed further in the neural network charger where gradient descent phys a critical orbit in training an attevit. For now, one can see that there are a number of issues for this nonlinear optimization procedure, including determining the initial gaves, tops size *i*, and comparing the syndrate reflections?

Alternating Descent

Another common technique for optimizing nonlinear functions of several variables is the discretistic discretistic models (ADM). Interacted comparing the gradient in several variables, optimization is done interactively in one variable at a time. For the example just demostance, this would much the comparison of the gradient matter easier. The basic sample stated, this would much the comparison of the gradient matter easier. The tables many other variables strend the state of the stat



Figure 4.0 Gradient descent applied to the function featured in Fig. 4.3(b). Three initial conditions are shown: $(x_0, y_0) = [(4, 0), (0, -5), (-5, 2)]$. The first of these (red circles) gets stack in a local minima while the other two initial conditions (bits ead magenta) find the global minima. Interrobation of the readem floateoiss of Fig. 4.2 are used to undue the solutions.



Figure 42 Alternating descent applied to the function in Fig. 4.3(b). Three initial conditions are shown: $(n_0, \eta_0) = \{(4, 0), (0, -5), (-5, 2)\}$. The first of these (red circle) gets stack in a local minimus while the other two initial conditions (blue and magnuts) find the global minimus. No gradients are computed to update the solution. Note the rapid convergence in comparison with Fig. 4.6.

procedure for the example of Fig. 4.6. This replaces the gradient computation to produce an iterative update.

Code 4.8 Alternating descent algorithm for updating solution.

```
fxsinterp2(X,Y,F,xs(1),y); xs(2)=xs(1); [-,ind]=min(fx); ys(2)=y
(ind);
fysinterp2(X,Y,F,x,ys(2)); ys(3)=ys(2); [-,ind]=min(fy); xs(3)=x
(ind);
```

Note that the alternating descent only requires a line search along one variable at a time, thus potentially speeding up computations. Moreover, the method is derivative free, which is attractive in many applications.

4.3 Regression and Ax = b: Over- and Under-Determined Systems

Curve fitting, as shown in the previous true section, results is a splittingian political in models in the maximum of the previous case the automatical binding is a shown in the splitting of the sector splitting is a splitting of the sector splitting of the sector splitting is a splitting of the sector splitting is a splitting of the sector splitting of the sector splitting is a splitting of the sector splitting is a splitting of the sector splitti

Before proceeding further, it should be noted that the system $\mathbf{Ax} = \mathbf{b}$ considered here is a restricted instance of $\mathbf{Y} = f(\mathbf{X}, \boldsymbol{\beta})$ in (4.4). Thus the solution \mathbf{x} contains the localings or *leverage scores* relating the relationship between the input data \mathbf{A} and outcome data \mathbf{b} . A simple solution for this linear problem uses the Moose-Penrose pseudo inverse \mathbf{A}^{T} from Sec. 1.4:

$$x = A^{\dagger}b$$
. (4.38)

This operator is computed with the **pinv(A)** command in MATLAB. However, such a solution is restrictive, and a greater degree of flexibility is sought for computing solutions. Our particular aim in this section is to demonstrate the interplay in solving over- and underdetermined systems using the *k* and *k* neuros.

Over-Determined Systems

Fig. 4.8 shows the general structure of an over-determined system. As already stated, there are generally no solutions that satisfy Ax = b. Thus, the optimization problem to be solved involves minimizing the error, for example the least-squares t_2 error E_2 , by finding an appropriate value of \hat{x} :

$$\hat{x} = \underset{x}{\operatorname{argmin}} \|Ax - b\|_2.$$
 (4.39)

This basic architecture does not explicitly enforce any constraints on the loadings x. In order to both minimize the error and enforce a constraint on the solution, the basic optimization methiecture can be modified to the following

$$\hat{\mathbf{x}} = \underset{\mathbf{x}}{\operatorname{argmin}} \|\mathbf{A}\mathbf{x} - \mathbf{b}\|_2 + \lambda_1 \|\mathbf{x}\|_1 + \lambda_2 \|\mathbf{x}\|_2 \qquad (4.40)$$

where the parameters λ_1 and λ_2 control the penalization of the ℓ_1 and ℓ_2 norms, respectively. This now explicitly enforces a constraint on the solution vector itself, not just the error. The ability to design the penalty by adding regularizing constraints is critical for understanding model selective in the following.

In the examples that follow, a particular focus will be given to the role of the ℓ_1 norm. The ℓ_1 norm, as already shown in Chapter 3, promotes sparsity so that many of the loadings



Figure 4.8 Regression framework for overdetermined systems. In this case, Ax = b cannot be satisfied in general. Thus, finding solutions for this system involves minimizing, for instance, the least-square error $||Ax - b||_2$ subject to a constraint on the solution x, such as minimizing the t_2 nem ||x||.

of the solution x are zero. This will play an important role in variable and model selection in the next section. For mow, consider solving the optimization problem (4.40) with $\lambda_2 = 0$. We use the open-source convex optimization package evx in MATLAB [218], to compute our solution to (4.40). The following code consider various values of the (λ_1 pentilaztion in producing solutions to an over-determined systems with 500 constraints, and 100 unknowns.

```
Code 4.9 Solutions for an over-determined system.
```

131



Figure 43 Solutions to an overdetermined system with 500 constraints and 100 meksowns. Panels ($\phi_i(z)$, dows als here for the values of the basings of the vectors. Note that an bet p penalty in increased from (a), $\lambda_2 = 0$ to (b), $\lambda_2 = 0$. In the (a), $\lambda_2 = 0$ to the vector increases, i.e. it becomes more spaces. A histogram of the touching values for $(d_i(z))$ is shown in the panels ($d_i(z)$, respectively. This highlights the role that the t_1 norm plays in promoting sparsity in — the volution.

```
subplot(4,1,j),bar(x)
subplot(4,3,9+j), hist(x,20)
and
```

Fig. 4.9 highlights the results of the optimization process as a function of the parameter λ_1 . Is should be noted that the solution with $\lambda_2 = 0$ is equivalent to the solutions what produced by computing the pseudo-inverse of the matrix A. Note that the ℓ_1 norm promotes a sparse solution where many of the components of the solution vector \mathbf{x} are zero. The histograms of the solution values of \mathbf{x} in Fig. 4/(0, \ell/) are particularly revealing as they show the sparsification process for increasing λ_1 .

The regression for over-determined systems can be generalized to matrix systems as shown in Fig. 4.8. In this case, the evx command structure simply modifies the size of the matrix **b** and solution matrix **x**. Consider the two solutions of an over-determined system generated from the following code.

Code 4.10 Solutions for over-determined matrix system.

```
n=300; m=60; p=20;
Aerand(n,m); berand(n,p);
lam=[0 0.1];
```



Figure 4.10 Solutions to an overdetermined system Ax = b with 300 constraints and 60×20 unknown. Panels (a) and (b) show a plot of the values of the loadings of the matrix x with ℓ_1 penalty (a) $\lambda_2 = 0$ to (b) $\lambda_1 = 0.1$.

```
for jell2
cvx_begin;
variable x(m,p)
minimize (serm(A*x-b,Z) + lam(j)*norm(x,l));
minimize (serm(A*x-b,Z) + lam(j)*norm(x,l));
minpilec(1,1,j), peolor(x.'), colormap(hot), colorbar
mampilec(1,1,j), peolor(x.'), colormap(hot), colorbar
```

Fig. 4.10 shows the results of this matrix over-determined systems for two different values of the added f_2 percently. Note that the addition of the f_2 norm sparsifies the solution and produces a matrix which is dominated by zero curities. The two examples in Figs. 4.9 and 4.10 show the important role that the f_2 and f_1 norms have in generating different types of solutions. In the following sections of this book, these norms will be exploited to produce raminmoisons models from data.

Under-Determined Systems

For undetermined systems, there are an infinite number of possible solutions satisfying Ax = b. The goal in this case is to impose an additional constraint, or set of constraints, whereby a unique solution is generated from the infinite possibilities. The basic mathemati-



Figure 4.11 Regression framework for underdetermined systems. In this case, Ax = b can be satisfied. In fact, there are an infinite number of solutions. Thus pinning down a unique solution for this system involves minimizing a constraint. For instance, from an infinite number of solutions, we choose the one that minimizes the ℓ_2 norm $|x||_2$, which is subject to the constraint Ax = b.

cal structure is shown in Fig. 4.11. As an optimization, the solution to the under-determined system can be stated as

$$\min \|\mathbf{x}\|_{p}$$
 subject to $A\mathbf{x} = \mathbf{b}$ (4.41)

where the p denotes the p-norm of the vector x. For simplicity, we consider the ℓ_2 and ℓ_1 norms only. As has already been shown for over-determined systems, the ℓ_1 norm promotes starsity of the solution.

We again use the convex optimization package evs. to compute our solution to (4.41). The following code considers both ℓ_2 and ℓ_1 peralization in producing solutions to an under-determined systems with 20 constraints and 100 unknowns.

Code 4.11 Solutions for an under-determined matrix systems.

```
n=20; ==100
Averand(n,m); berand(n,1);
cvx_begin;
variable x2(m)
minimize( norm(x2,2));
subject to
Avs2 == b;
cvx_end;
cvx_equin;
variable x1(m)
minimize( norm(x1,1));
```

```
A+x1 == b;
cvx_end;
```

This code produces two solutions vectors x2 and x1 which minimize the f_{x2} and f_{x1} more respectively. Note the ways that ext allows one to impose constraints in the optimization routine, Fig. 4.12 shows a bur plot and histogram of the two solutions produced. As before, the sparsity promoting f_{x1} compared base as used to be a solution were dominated be zeros. In fact, for this case, there are exactly 80 zeros for this linear system since there are only 20 constraints for the 100 nuknows:

As with the over-determined system, the optimization can be modified to handle more general under-determined markiv equations as shown in Fig. 41.1 The ever optimization package may be used for this case as before with over-determined systems. The software engine can also work with more general-pacoms as well as minimize with both t_1 an t_2 penalized simultaneously. For instance, a common optimization modifies (4.41) to the following

$$\min(\lambda_1 \|\mathbf{x}\|_1 + \lambda_2 \|\mathbf{x}\|_2)$$
 subject to $\mathbf{A}\mathbf{x} = \mathbf{b}$ (4.42)



Figure 412 Solutions to an under-determined system with 20 constraints and 100 unknowns. Pancle (a) and (b) show a bur plot of the values of the loadings of the vectors x. In the former panel, the optimization is subject to minimizing the ℓ_2 norm of the solution, while the latter panel is subject to minimizing the ℓ_1 norm. Note that the ℓ_1 penalization produces a sparse solution vector. A histogram of the loading values for (a) and (b) is shown in the punel (c) and (d) respectively.

135



Figure 413 (a) One handed realizations of the parabelic function (4.43) with additive white noise parametrized by $\sigma = 0.1$. Although the noise is small, the least-square fitting procedure produces significant variability when fitting to a polynomial of degree twenty. Pauch (b)-(c) demonstrate the loading (coefficients) for the various polynomial coefficients for four different noise realizations. This demonstrate node

where the weighting between λ_1 and λ_2 can be used to promote a desired sparsification of the solution. These different optimization strategies are common and will be considered further in the following.

4.4 Optimization as the Cornerstone of Regression

In the previous two sections of this chapter, the fitting function f(x) was specified. For ionance, it may be detainable to produce an fit for ohat f(x) = p + p + 2p. The creditions are then found by the regression and optimization methods already discussed. In what follows, our objective is to develop techniques which allow us to objectively steel z a goodmodel for fitting the data, i.e. should one use a quadratic or cohe fit? The error matrix alone does not duct az a good model sections as the none terms that are also chosen for fitting, the more parameters are available for lowering the error, regardless of whether the additional terms have are momention or interventibility.

Optimization strategies will play a foundational role in extracting interpretable results and meaningful models from data. As already shown in previous sections, the interplay of the ξ_2 and ξ_1 norms has a critical impact on the optimization outcomes. To illustrate further the role of optimization and the variety of possible outcomes, consider the simple example of data generated from noisy measurements of a parabola

$$f(x) = x^2 + N(0, \sigma)$$
 (4.43)

where $\mathcal{N}(\theta, \sigma)$ is a normally distributed random variable with mean zero and standard deviation σ . Fig. 4.13(a) shows an example of 100 random measurements of (4.43). The panbolic structure is clearly evident despite the noise added to the measurement. Indeed, a panzholic it is trivial to compute using classic least-square fitting methods outlined in the first section of this chapter.

The goal is to discover the best model for the data given. So instead of specifying a model a priori, in practice, we do not know what the function is and need to discover it. We can begin by positing a regression to a set of polynomial models. In particular, consider framing the model selection problem $Y = f(X, \beta)$ of (4.4) as the following system Ax = b:

$$\begin{bmatrix} | & | & \cdots & | \\ 1 & x_j & x_j^2 & \cdots & x_j^{p-1} \\ | & | & | & \cdots & | \end{bmatrix} \begin{bmatrix} \beta_1 \\ \vdots \\ \beta_p \end{bmatrix} = \begin{bmatrix} f(x_1) \\ f(x_2) \\ \vdots \\ f(x_{100}) \end{bmatrix}$$
(4.44)

where the matrix A contains polynomial models up to degree p - 1 with each row representing a measurement, the β_l are the coefficients for each polynomial, and the matrix be contains the concenses (data 1) $f(x_l)$. In what follows, we will consider a scenario where 100 measurements are taken and 20 term (19th order) polynomial is fit. Thus the matrix system **Ax** = **b** results in an over-determined system as illustrated in Fig. 4.8.

The following code solves the over-determined system (4.44) using least-square regression via the **pinv** function. For this case, four realizations are run in order to illustrate the impact that a small amount of noise has on the regression procedure.

Code 4.12 Least-squares polynomial fit to parabola with noise.

```
[act00: lock
[act00: lock]
[act0...];
[
```

Fig. 4.13(b)-(c) shows four typical loadings β computed from the regression procedure. Note that despite the low-level of noise added, the loadings are significantly different from one another. Thus each noise realization produces a very different model to explain the data.

The variability of the regression results are problematic for model selection. It suggests that even a small amount of measurement noise can lead to significantly different conclusions about the underlying model. In what follows, we quantify this variability while also considering various regression procedures for solving the over-determined linear system Ax = b. Highlighted here are five standard methods: least-square regression (pinv), the backslash operator (\), LASSO (least absolute shrinkage and selection operator) (lasso), robust fit (robustfit), and ridge regression (ridge). Returning to the last section, and specifically (4.40), helps frame the mathematical architecture for these various Ax = b solvers. Specifically, the Moore-Penrose pseudo-inverse (pinv) solves (4.40) with $\lambda_1 = \lambda_2 = 0$. The backslash command (\) for over-determined systems solves the linear system via a OR decomposition [524]. The LASSO (lasso) solves (4.40) with $\lambda_1 > 0$ and $\lambda_2 = 0$. Ridge regression (ridge) solves (4.40) with $\lambda_1 = 0$ and $\lambda_2 > 0$. However, the modern implementation of ridge in MATLAB is a bit more nuanced. The popular elastic net algorithm weights both the ℓ_2 and ℓ_1 penalty, thus providing a tunable hybrid model regression between ridge and LASSO. Robust fit (robustfit) solves (4.40) by a weighted least-squares fitting. Moreover, it allows one to leverage robust statistics methods and penalize according to the Huber norm so as to promote outlier rejection [260]. In the data considered here, no outliers are imposed on the data so that the power of robust fit is not properly leveraged. Regardless, it is an important technique one should consider

Fig. 4.14 shows a serie of box form (100 realizations) of alm humflemme between the process matching strengtons in tellings councils of a last hugflem, similarly different strength of the strengton in tellings councils of the strength s

Code 4.13 Comparison of regression methods.

This code also produces the 100 realizations visualized in Fig. 4.13(a).



Figure 4.3 Comparison of regression methods to $A.4 \approx$ for *i* in one-determined system of heart A and A

Despite the significant variability exhibited in Fig. 4.14 fer most of the loading values by the different regression techniques, the error produced in the tilting sourceafter has little variability. Moreover, the various methods all produce regressions that have comparable relatively apposing the structure of the structure of the structure of the relatively apposing the structure of the regression methods of Fig. 4.14. All the regression features produce comparably for error and no variability evolution in the structure of the

As a final note to this section and the code provided, we can consider instead the regression procedure as a function of the number of polynomials in (4.44). In our example of Fig. 4.14, holynomials up to degree 20 were considered. If instead, we sweep through polynomial degrees, then something interesting and important occurs as illustrated in Fig. 4.15(b)-(c). Specifically, the error of the regression collapses to 10^{-3} after the



Figure 415 (a) Comparison of the error for the six regression methods used in Fig. 4.14. Despite the variability across the optimization methods, all of them produce the owneror solutions. (b) Error using local-square regression as a function of increasing degree of polynomial. The error drops rapidly until the quadratic term is used in the regression, (c) Detail of the error showing that the error actually increases within the waits are hitche-drence of orborometial to fitth data.

quadratic term is added as shown in panel (b). This is expected since the original model was a quadratic function with a small amount of noise. Remarkably, as more polynomial terms are added, the ensemble error actually increases in the regression procedure as highlighted in panel (c). Thus simply adding more terms does not improve the error, which is counter-immitive at first. The code to produce these results are given by the following:

Code 4.14 Model fitting with polynomials of varying degree.

```
Immession(100,W);
for jj=1,jj
pht(:,j)=(x,').^(j-1);
mak(:,j)=(x,').;
for j=1:0;
for j=1:0;
for j=1:0;
manufactorial (j, j);
mod (j, j);
mod
```

Note that we have only swept through polynomials up to degree 10. Note further that panel (c) of Fig. 4.15 is a detail of panel (b). The error produced by a simple parabolic fit is approximately twice as good as a polynomial with degree 10. These results will help frame our model selection framework of the remaining sections.

4.5 The Pareto Front and Lex Parsimoniae

The preceding chapters show that regression is more manaced than simply choosing a model and performing a least-square. It No only may there manrenss metrics for constraining the solution, the model inself should be carefully selected in order to achieve a here, more interpretated sectorism of the data. Such considerations on an approprinte model data back is William of Oceann (c. 1227–1327), who was an English Franciscan first, scheduste philosopher, and thereidogian. Ocean proposed bia issue of paramotopy (in this for garary maniar), commonly taxons as Oceans's ranzo, whereby the stated that among competing hypotheses, the one with the fevent samplings should be selected, or when you here to solve the solution of the solve sampling should be selected or when you here to solve the solve should be solved as a solved back of the solved ba competing theories that make exactly the same predictions, the simpler one is the more likely. The philosophy of Occam's mator has been used extensively throughout the physical and biological sciences for developing governing equations to model observed phenomena.

Pariniony also plays a certain do fei a fei matematical work of Windo Partos (1) Histo-1937). Barous sona halina majeres, escolariza, economica, platical escienta, and philosopher. He ands several impertant combinition to economics, guestically in the start of a conset distribution at the analysis of analysis ("starts, the analysis and philosopher. He and several impertant combinition are commonly and the material data and the analysis of analysis ("starts, the analysis of the bar location for the popular 1020 rate which is quantized by Haurania In-1945). Starts of the analysis of the analysis of analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the analysis of the analysis of the analysis of the starts of the analysis of the starts of the analysis of

Pareto and Occam ultimately advocated the same philosophy: explain the majority of observed data with a parsimonious model. Importantly, model selection is not simply about reducing error, it is about producing a model that has a high degree of interpretability, generalization and predictive capabilities. Fig. 4.16 shows the basic concept of the *Pareto*



Number of Terms

Figure 4.16 For model velocities, the criteria of accuracy (low error) is halanced against parimony. There can be a variety of models with the wanne number of terms (green and magnetin pairs), but the *Parture Forsture* (magneta points) is defined by the envelope of models that produce the lowest error for a given number of terms. The solid line provides an approximation to the Parture Forest *Parture foreignet and the solid line provides an approximation* is the Parture for forest. The *Parture solid line* (shaded region) are those models that produce accurate models while remaining maximus fors. Frantise and Parone Optimal additions, Specifically, for each model considered, the number for any addre nervin multicing the data is compared. The solutions with the lowest enter for a given number of terms define the Parone function. These paraloxies non-blands multimodel and the solution of the solution of the solution of the lowest enter the solution of the solution of the solution of the solution of the lowest enter than cannot be made to perform here aquation tase exposing strategy within performantion and the solution of a solution of the so

Overfitting

The Parties concept tested amending when considering applications to red data. Specifically, when building models with maps for parameters, it is often the case in methics learning applications with high-dimensional data, it is easy to overift a model to the data. Models the increase it never dimensional Fig. 41,459, on methics of parameters in the dimensional data with the dimensional energy data should complexity inflatances the specific test of the dimensional data with the data with t

To illustrate the overfitting that access with read data, consider the simple comple of the state scients. In this complex, our as simply trying to find the correct parabolic model measured with additive noise (4.37). The results of Fig.4.15(b) and 4.15(c) already induced that constraint is a courting for palyomatian models beyond new constrained. The following that produces a raining and to set with the parabolic action of the rain < 1.5 and the produces a raining and to set with the parabolic or (6.47). The training set is on the region < 10.4 with the first set of expendition region yull be far < 1.5 (3.4).

Code 4.15 Parabolic model with training and test data

This code produces the ideal model on two distinct regions: $x \in [0, 4]$ and $x \in [4, 8]$. Once measurement noise is added to the model, then the parameters for a polynomial fit no longer produce the perfect parabolic model. We can compute for given noisy measurements both an interpolation error, where measurements are taken in the data regime of $x \in [4, 8]$. this example, a least squares regression is performed using the pseudo-inverse (pinv) from MATLAB.

Code 4.16 Overfitting a quadratic model.

```
Model = Franchise of model interes
for the second sec
```

This simple cample shows some of the most basic and common frateries movimits in the structure of the struc

The above example shows that some form of model selection to systemically deduce a parimotions model a critical for producing values models that can agreentize costide of where data is collected. Much of machine fearming revolves around (1) using data to generate predictive models, and (1) excess-validation techniques to resource the mone delagenerate predictive models, and (1) excess-validation techniques to resource the mone delaproduce a surgementizable model such as that exhibited in Fig. 4.7.1 in bath follows, we will consider some standard strategies for producing reasonable models.

4.6 Model Selection: Cross-Validation

The previous section highlights many of the fundamental problems with regression. Specifically, it is easy to overfit a model to the data, thus leading to a model that is incapable of generalizing for extrapolation. This is an especially perincious issue in training deep



Figure 4.7 (c) the island model ((r)):= -2^{-1} our the domain $< \xi > 0$, 31, Data is collected in the region $< \xi > 0$, 31, Data is oblicable in the region $< \xi > 0$, 31, Data is oblicable in the domain $< \xi > 0$, 31, Data is oblicable in the domain $< \xi > 0$, 31, Data is oblicable in the domain $< \xi > 0$, 31, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain $< \xi > 0$, 32, Data is oblicable in the domain < 0, Data is oblicable in the domain < 0. Data is oblicable in the domain < 0, Data is oblicable in the domain < 0, Data is oblicable in the domain < 0. Dependence in the domain < 0 and < 0, Data is oblicable in the domain < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0. Dependence in the domain < 0 and < 0. Dependence in the domain < 0. Dependence in the doma

neural nets. To overcome the consequences of overfitting, various techniques have been proposed to mere appropriately select a puriminators model with only a few parameters, thus balancing the error with a model that can more easily generatize, or extrapolate. This provides a realistrepretation of the hareor forta in Fig. 21.6. Specifically, the error increases dramatically with the number of terms due to overfitting, especially when used for extrapolate.

There are two common mathematical strategies for circumventing the effects of overfiting in model electrica cross-studiation and computing jaromatic arrivas. This sector considers the former, while the tare method is considered in the next section. Crossvalidation strategies are perhaps the most common and critical technologies in almost almachine learning algorithms. Infaced, one should never trust a model unless properly crossvalidated. Cross-studiation can be stated gains imply: That random porticon of your distaand build an edd. Do this 4 items and nerage the parameter scores (regression loading) to to produce the cross-validated model. Text the model prediction against withhel (terrary olation) data and evaluate whether the model is actually any good. This commonly used strategy is called k-fold cross-validation. It is simple, intuitively appealing, and the k-fold model building procedure produces a statistically based model for evaluation.

To illustrate the concept of cross-valuations, see will not equipation consider thing polynomial models to the simple function $f(1) = 2^{-1}$ (Ger Fig. 4.13). The previous sections of this chapter we should considered the inputtient in dual, both them the values argument for the simplexity of the



Figure 4.32 Cross-validation using 4-fold strategy with k = 2.20 and 100 (telt, middle and right outman respective). These different regression strategies are access-validated learness-super fitting of peach-siteware (the QR-based backshak, and the sparsity promoting LASSO. Note that the LASSO for this ensuring produces the quadratic model within evaluation action or two fold validation. The backshahm have (QR-diperdim has a strong signature rather 100 fold cross-validated learness), while the DAS fold cross-validation.

```
Code 4.17 k-fold cross-validation using 100 foldings.
```

Fig. 4.18 shows the results of the *k*-fold conv-validation computations. By promoting sparsity (paraimony), the LASSO achieves the desired quadratic model after even a single k = 1 fold (1c. thus this is not even cross-validated). In contrast the least-square regression (pendo-inverse) and QR-based regression both require a significant number of folds to produce the dominant quadratic term. The least-square regression, even after k = 100folds, still includes both a quadratic and cohe term.

The full model selecting process used in Add cross-validation derse an isother standarding of terms that may increase the Add cross-validation of the Add terms and the Add terms and the Add terms and the Add terms and the has a small contributing latter composed. The QH category of heat kalash produces a many the of and increases the add terms and terms a

Code 4.18 Comparison of cross-validated models.



Figure 4.19 Error and loading results for k = 100 field cross-validations. The loadings for the k-fold under with the freeding actual by advance (x, b), and values (1) the methoding as advanced to the strength of the

```
Eni(jj)=morm(ftrain-fnai)/norm(ftrain);
fnaesphi_e=ani;
Ene(jj)=morm(ftest-fnae)/norm(ftest);
end
```

The results of Fig. 4.19 show that the model selection process, and the regression technique used, makes a critical difference in producing a viable model. It further shows that despite a k-fold cross-validation, the extrapolation error, or generalizability, of the model can still be poor. A good model is one that keeps errors small and also generalizes well, as does the LASSO in the previous example.

k-fold Cross-Validation

The process of k-fold cross validation is highlighted in Fig. 4.20. The concept is to partition a data set into a training set and a test set. The test set, or withhold set, is kept separate from any training procedure for the model. Importantly, the test set is where the model produces an extrapolation approximation, which the figures of the last two sections show to be challenging. In k-fold cross-validation, the training data is further partitioned into kfolds, which are typically randomly selected portions of the data. For instance, in standard 10-fold cross validation, the training data is randomly partitioned into 10 partitions (or folds). Each partition is used to construct a regression model $\mathbf{Y}_i = f(\mathbf{X}_i, \boldsymbol{\beta}_i)$ for j =1, 2, ..., 10. One method for constructing the final model is to average the loading values $\tilde{\beta} = (1/k) \sum_{i=1}^{k} \beta_i$, which are then used for the final, cross-validated regression model $\mathbf{Y} = f(\mathbf{X}, \hat{\boldsymbol{\beta}})$. This model is then used on the withhold data to test its extrapolation power. or generalizability. The error on this withhold test set is what determines the efficacy of the model. There are a variety of other methods for selecting the best model, including simply choosing the best of the k-fold models. As for partitioning the data, a common strategy is to break the data into 70% training data, 20% validation data, and 10% withheld data. For very large data sets, the validation and withheld can be reduced provided there is enough data to accurately assess the model constructed.

Leave p-out Cross-Validation

Another standard technique for cross-validation involves the us-called low power crossvalidation (LGO CV). In this case, p-samples of the training data are removed from the data and keep as the validation set. A model is built on the remaining training data and the accuracy of the model is used on the particular data set. This is repeated with a new selection of p-samples unit all the training data has been part of the validation data set. The other data set of the samples of the samples of the samples of the samples of the other data set. The samples of the models and the loadings reduced from the various neutrinos of the data.

4.7 Model Selection: Information Criteria

Three is a different approach to model selection than the cross-validation strategies outlined in the previous section. Indeed, model selection has a rigorous set of mathematical innovations starting from the early 1950s. The Kullhack-Leitler (KL) divergence [314] measures the distance between two probability density distributions (or data sets which represent the truth and a model) and is the core of modem information theory criteria for evaluating the



Figure 4.2) Procedure for k-fold cross-validation of models. The data is initially particular dim as training and text (without k), Tripically be without loss it a generated from a random sample of the overall data. The training data is optically on the value of the star is a strained from the star is a grantered from the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically with the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is optically be an example on the straining data is a straining dat

viability of a model. The KL divergence has deep mathematical connections to statistical methods characterizing eneropy as developed by Ladwig E. Behzmann (E. 1844-1906), as well as a etailona to information theory developed by Claude Shannon [486]. Model stetetion is a well developed field with a large body of literature, most of which is exceptionally well reviewed by Bannham and Anderson [105]. In what follows, only brief highlights will be given to demonstrate source of the standard methods.

The KL divergence between two models $f(\mathbf{X}, \boldsymbol{\beta})$ and $g(\mathbf{X}, \boldsymbol{\mu})$ is defined as

$$I(f, g) = \int f(\mathbf{X}, \boldsymbol{\beta}) \log \left[\frac{f(\mathbf{X}, \boldsymbol{\beta})}{g(\mathbf{X}, \boldsymbol{\mu})} \right] d\mathbf{X}$$
 (4.45)

where g and μ are parameterizations of the the models $f(\cdot)$ and $g(\cdot)$ respectively. From an information theory perspective, the quantity I(f, g) measures the information to twice gis used to represent f. Note that if f = g, then the log term is zero (i.e. $\log(1) = 0$) and I(f, g) = 0 so that there is no information loss. In practice, f will represent the routh, or measurements of an excernineat, while e will be a model romouth of describe f.

Unlike the regression and cross-validation performed pervisorsly, when computing KL divergence a model using different regression strategies (See Fig. 4.2) for instance). Here a number of models will be posited and the loss of information, or KL divergence, of each model will be constanted. The model with the lowest loss of information is generally regarded as the best model. Thus given M proposed models $g_j(\mathbf{X}, \boldsymbol{\mu}_j)$ where $j = 1, 2, \dots, M$, we can compute $I_j(f, g_j)$ for each model. The correct model, or best model, is the one that minimizes the information loss min_j $I_j(f, g_j)$.

As a simple example, consider Fig. 4.21 which shows three different models that are compared to the truth data. To generate this figure, the following code was used. The computation of the KL divergence score is also illustrated. Note that in order to avoid division by zero, a constant offset is and ded to each probability distribution. The runh data generated, f(x), it a simple normally distributed variable. The three models shown are variants of normality and mildrowing distributed functions.

Code 4.19 Computation of KL divergence.

```
n=10000;

n=10; 11; 1; n=0;

n=0; n=n=0; (n=1; 1; n=0; n=0; 1; n=0; 1;
```



Figure 4.21 Comparison of three models $g_1(x)$, $g_2(x)$ and $g_3(x)$ against the truth model f(x). The KL divergence $I_j(f, g_j)$ for each model is computed, showing that the model $g_1(x)$ is closest to statistically representing the true data.

```
gnAnter(s, s)=0.51;
el/(rang(s, t)) = nermalize data
ging()Trang(s, t); phep/(rang(s, t)); ging()Trang(s, g));
phenomena (s)=0.51; el/(s)=0.51; el/(s)=0.51;
```

Information Criteria: AIC and BIC

This simple example shows the basic ideas behind model selection: compute a distance between a proposed mod coping g1(a) and the measured truth [7] (b). In the cashy [790; Hiongan Akale combined Fisher's maximum likelihood comparison [183] with the KL divergence cores to produce what is now called healthic information (Fisher) with the CH [260] with how produces that is now called healthic information (Fisher) with the CH [260] with how produces and in the so-called Baryosian Information Criterion BRC [260] with how produced an information occer that was parameted to converge to the correct model in the large data limit, provided the correct model was included in the set of condidate models.

To be more precise, we turn to Akalar's semilari contribution [7]. Akalar's usa aware that K. divergence cannot be compared in practice since it requires that Bowdelage of the statistics of the truth model (1/c) and of all the parameters in the proposed models (1/c). Thus, Akalar proposed an alternative way to orinniat K. divergence based on the rempricted hysical hardbowner and the state of the state way a calculative state in the state of the state way a paradigmental the state of the state o

$$AIC = 2K - 2 \log [\hat{L}(\hat{\mu}|\mathbf{x})],$$
 (4.46)

where K is the number of parameters used in the model, h is an estimate of the less parameters used (i) closes KL divergence) is $(K_{\rm g})$ compared from a maximum likelihood estimates (MLE); and χ are independent samples of the data to be B. Thus, instead of a direct ensaice of the datance between the outdot, the ALC power data of the test of the estimates of the datance between the approximating model and the two model or data. No 2.7. The providing parameters of the datance between the data to be the single state of the test of the datance between the data test. The data test is the data test of the data test. The data test of the data test. The data test of the data test. The data test of the data test. The data test of the data test. The data test of the data test of the data test of the data test of the data test. The data test of the data test of the data test of the data test of the data test. The data test of the data test of the data test of the data test. The data test of the data test of the data test. The data test of the data test of the data test of the data test. The data test of the data test. The data test of the data test of the data test of the data test of the data test. The data test of the data test. The data test of the data test

AIC is one of the standard model selection criteria used today. However, there are others. Highlighted here is the modification of AIC by Gideon Schwarz to construct BIC [480]. BIC is almost identical to AIC aside from the penalization of the information criteria by the number of terms. Specifically: BIC is defined as

$$BIC = \log(n)K - 2\log \left[\hat{\mathcal{L}}(\hat{\boldsymbol{\mu}}|\mathbf{x}) \right], \quad (4.47)$$

where *n* is the number of data points, or sample size, considered. This slightly different version of the information criteria has one significant consequence. The semial contribution of Schwarz was to prove that if the correct model was included along with a set of candidate models, then it would be theoretically guaranteed to be selected as the best model based upon BIC for sufficiently large set of data x. This is in contrast to AIC for which in certain antibodicul cases, it can asset the wrome model.

Computing AIC and BIC Scores

MATLAB allows us to directly compute the AIC and/or BIC score from the ability mand. This computational tool is embedded in the econometrics toolbox, and it allows one to evaluate a set of models against one another. The evaluation is made from the loglikelihood estimate of the models under consideration. An arbitrary number of models can be compared.

In the specific example considered here, we consider a ground truth model constructed from the autorearessive model

$$x_{n} = -4 + 0.2x_{n-1} + 0.5x_{n-2} + N(0, 2)$$
 (4.48)

where x_1 is the value of the time series at time t_1 and $\mathcal{N}(0,2)$ is a white-noise process with mean zero and variance two. We fit three autoregressive integrated moving average (ARMA) models to the data. The three ARMA models have one, two and three time delays in their models. The following code computes their log-likelihood and corresponding AIC and BIC scores.

Code 4.20 Computation of AIC and BIC scores.

```
[7 : 10] I. Sample size
DD = artisel (Description 1, -4, 202, [0.2, 0.3], "Maximum", 2];
y = simulation (DD, 7];
handlin = artisel (NELGY, 1])
handlin = artisel (NELGY, 1], artisel (Alado)
[--, 1](2]) = artisel forellin, y, form: (Alado)
[--, 1](2]) = artisel forellin, y, for
```

Note that the best model, the one with both the lowest AIC and BIC score, is the second model which has two time delays. This is expected as it corresponds to the ground truth model. The output in this case is given by the following.

```
aic =
301.7732
350.2422
350.0479
bic =
309.5007
```

368.6629

The lowest AIC and BIC score is 358.2422 and 368.6629 respectively. Note that although the correct model was selected, the AIC score provides little distinction between models, especially the two and three time-delay models.

Suggested Reading

Texts

- Model selection and multimodel inference, by K. P. Burnham and D. R. Anderson [105].
- (2) Multivariate analysis, by R. A. Johnson and D. Wichern, 2002 [266].
- (3) An introduction to statistical learning, by G. James, D. Witten, T. Hastie and R. Tibshirani, 2013 [264].

Papers and Reviews

- On the mathematical foundations of theoretical statistics., by R. A. Fischer, Philosophical Transactions of the Royal Society of London, 1922 [183].
- (2) A new look at the statistical model identification., by H. Akaike, IEEE Transactions on Automatic Control, 1974 [7].
- (3) Estimating the dimension of a model., by G. Schwarz et al., The annals of statistics, 1978 [480].
- (4) On information and sufficiency., by S. Kullback and R. A. Leibler, The annals of statistics, 1951 [314].
- (5) A mathematical theory of communication., by C. Shannon, ACM SIGMOBILE Mobile Computing and Communications Review, 2001 [480].

Mathus learning is based good good granting technical t

There are two broad categories for machine learning: supervised machine learning and unsupervised machine learning. In the former, the algorithm is presented with labelled datasets. The training data, as outlined in the cross-validation method of the last chapter, is labeled by a teacher/expert. Thus examples of the input and output of a desired model are explicitly given, and regression methods are used to find the best model for the given labeled data, via optimization. This model is then used for prediction and classification using new data. There are important variants of supervised methods, including semisupervised learning in which incomplete training is given so that some of the input/output relationships are missing, i.e. for some input data, the actual output is missing. Active learning is another common subclass of supervised methods whereby the algorithm can only obtain training labels for a limited set of instances, based on a budget, and also has to optimize its choice of objects to acquire labels for. In an interactive framework, these can be presented to the user for labeling. Finally, in reinforcement learning, rewards or punishments are the training labels that help shape the regression architecture in order to build the best model. In contrast, no labels are given for ansupervised learning algorithms, Thus, they must find patterns in the data in a principled way in order to determine how to cluster data and generate labels for predicting and classifying new data. In unsupervised learning, the goal itself may be to discover patterns in the data embedded in the lowrank subspaces so that feature envincering or feature extraction can be used to build an appropriate model.

In this chapter, we will consider some of the most commonly used supervised and unsupervised machine learning methods. As will be seen, our goal is to highlight how data mining can produce important data features (feature engineering) for later use in model building. We will also show that the machine learning methods can be broadly used for clustering and classification, as well as for building regression models for prediction. Critical to all of this machine learning architecture is finding low-rank feature spaces that are informative and interpretable.

5.1 Feature Selection and Data Mining

To replace that for diagnostics, predictions and costend, dominant features of the dama structured in the sequences of the sequence of the local VD and FCA were instanded as the extracted in the sequences of the sequences of the sequences of the designation example of features. It for instance, the dominant features of large number of the sequences produces of the sequences of the sequences of the sequences produces of the sequences of the sequences of the sequences including highlighting the sequences of the sequences of the sequences including highlighting the sequences of the sequences including highlighting the sequences of the sequences and the sequences of the sequences of the sequences of the sequences in the sequences of the sequences

The got of data mixing and matchical transition is to construct and exploit the istrinuits low-and fatteren opera of a photom data art. The fatter space can be found in an annumprotocal fiscals by an algorithm, or it can be explicitly constructed by expert laxobicity and the exploration of the structure of the exploration of the exploration

Securit examples will be developed that illustrate how to generate a feature space, starting with a standard data set included with MTLAB. The Folderin data set included measurements of 150 trises of there valeties: setons, verticolor, and vitigaics. The 50 samples of each dhower include measurements in continuers on the regal length, samples with, ped length, and pedit with. For this data set, the four features are already defined in terms of interpreture properties of the biology of the plants. For visualization approach, Fig. 31 considers only the first three of these features. The following code accesses the Fielder is data set.

Code 5.1 Features of the Fisher irises.

Fig. 5.1 shows that the properties measured can be used as a good set of features for clustering and classification purposes. Specifically, the three iris varieties are well separated



Figure 3.1 Fisher inis data set with 150 measurements over three variaties including 50 measurements cach of setosa, versicolor, and virginica. Each flower includes a measurement of sepal length, sepa with, petal length, and petal width. The first three of faces are illustrated here showing that these simple biological features are sufficient to show that the data has distinct, quantifiable differences between the species.

in this forature space. The senso airs is most distinctive in its forature profile, while the versicolot and viginitic have a small overlap among the smalles kines. For this data set, machine learning is certainly not required to generate a good classification scheme. Howere, data generally does not so readily classification schemes how the set of the state of t

As a second example, we consider in Fig. 5.2 a selection from an image database of B0 of one and B0 cm. A specific angle of the data cm is to develop an atmosch dashielladin method behaving the compares can distinguish between cm and dogs. In fits case, the data method behaving the compares can distinguish between com and dogs. In fits case, the data method behaving the compares of the data set of the data set. Like experiments of the data and performs a singler value decomposition on the data after the mean n-subtraction. The SFD produces an ordered set of modes characterizing the data for the data and performs a ordered and of modes characterizing the data for the mean n-subtraction. The SFD produces are ordered set of modes characterizing the data for the data and performs (2, 2, 3) shows the fits the OFD modes of the D410 immers (500 dward 80 cm) that (2, 2, 3) shows the fits the OFD modes of the data for the mean set of the data set of set o

Code 5.2 Features of dogs and cats.

```
load dogData.mat
load catData.mat
CDsdouble([dog cat]);
[u,s,v]savd(CD-mean(CD(:)),'econ');
```



Figure 5.2 Example images of dogs (left) and cats (right). Our goal is to construct a feature space where automated classification of these images can be efficiently computed.



Reput 3.1 First from framese (a-b)(d) generated from the SVD of the 160 images of dargs and acts, i.e. the three more there for the common of the U mixer of the SVD. Tyrica to and dargs images are shown in Fig. 5.2. Note that the first two manages (a) and (b) show that the triangular cars are important for the contrady addinguidable frame for the strangest actually addinguidable frame for the strangest show the strangest shows a strangest show the strangest show the strangest shows a strangest show the strangest show the strangest show the strangest shows a strangest show the strangest show t

The original image space, or pixel space, is only one potential set of data to work with. The data can be transformed into a wavelet representation where edges of the images are emphasized. The following code loads in the images in their wavelet representation and computes a new low-rank embedding space.


Figure 5.4 First four features (a)-(d) generated from the SVD of the 160 images of dogs and cats in the wavelet domain. As before, the first two modes (a) and (b) show that the triangular cars are important. This is an alternative representation of the dogs and cats that can help better classify dogs versus cats.

Code 5.3 Wavelet features of dogs and cats.

```
load catData_w.mat
load dogData_w.mat
CD2s[dog_wave cat_wave];
[u2, m2, v2] savd(CD2-mean(CD2(:)), 'scon');
```

The equivalent of Fig. 5.3 in wavelet space is shown in Fig. 5.4. Note that the wavelet representation helps emphasize many key features such as the eyes, nose, and cars, potentially making it easier to make a classification decision. Generating a feature space that enables classification is critical for constructing effective machine learning algorithms.

Whether using the image space directly or a wavelet representation, Fig. 5.3 and 5.7 representerly, the gain is to project the data one the forance space generation by each. A construction of the structure of the structure of the structure of the structure of the of tasks that may include chestering, classification, and predictions. The importance of each former us on individual image is given by bub warms in the STOS predictably, each column of the distribution of the structure of the distribution of the structure of the flagges are three to build only and the structure of the distribution of the data are structured and the structure of the distribution of the data are structured by each former (the Fig. 5.5). The distribution of the distribution data are structured by the structure of the distribution of the distribution of the structure of the distribution of

Code 5.4 Feature histograms of dogs and cats.

```
xbin=linspace(-0.25,0.25,20);
for j=1:4
```



Figure 5.5 Histogram of the distribution of loadings for dogs (blue) and cats (red) on the first four dominant SVD mode: The left panel shows the distributions for the raw images (See Fig. 5.3) while the right panels show the distribution for wavelet transformed data (See Fig. 5.4). The loadings come from the columns of the V matrix of the SVD. Note the goed separability between dogs and cats using the second mode.

```
subplot(4,2,2*j-1)
pdf1shist(v(1:80,j),xbin)
pdf2shist(v(81:160,j),xbin)
plot(xbin,pdf1,xbin,pdf2,'Linewidth',[2])
and
```

Fig. 5.5 shows the distribution of loading scores for the first four modes for both the raw images as well as the wavelet transformed images. For both the sets of images, the distribution of loadings on the second mode clearly shows a strong separability between dogs and cass. The wavelet processed images also show an ice separability on the fourth mode. Note that the first mode for both shows very little discrimination between the distributions and is thus no tweet life of classification and clusterine doscritics.

Features that provide strong separability between different types of data (e.g. dogs and cas) are typically explorited for machine learning tasks. This simple example shows that feature engineering is a process whereby an initial data exploration is used to help detertion of the second strong strong



Figure 3.8 Projection of door (green) and cash (magerda) into feature space. Note that the raw images and their wavelet constrepasts produce different emboddings of the data. Both exhibit clustering around their baleed states of dog and cat. This is republied in the learning algorithms that follow. The wavelet images are especially good for clustering and classification as this feature space more easily separate the data.

wavelet transformed counterparts. As will be seen later, the wavelet transformed images provide a higher degree of separability, and thus improved classification.

5.2 Supervised versus Unsupervised Learning

As previously attack, the goal of that attaining and mathitle learning it is conserved and project the initiative locase. Rearring once a given that acts Goal frames employeding and frame reactions digitations can then the stock to have based to the stock of the formation of the stock of the stock of the stock of the stock of the formation of the stock of the stock of the stock of the stock of the formation of the stock of the stock of the stock of the stock of the formation of the stock of the formation of the stock of

To lithing the difference is supervised events unsupervised hearing, consider Fig. 3.7. This shows a scare first of two formulas distributions. Is one case, the data is well separated but their means are sufficiently far apert and two distinct clusters are observed, the start is seen the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is seen to be a start of the start is the start of the start o



Figure 32 Binaration of unsupervised versus supervised lataming, In the Lift panels (a) and (c), unsupervised lataming latents to find clusters for the data in ond to classify them into two prospec. Five well separated data (c), the task is straightforward and labels on easily be produced. For conclusions, the straightforward of the task is straightforward and labels on easily be produced. For the conclusion of the straightforward and labels on easily be produced. For the straightforward of the straightforward and labels on easily be produced. For the straightforward of the straightforward and labels on easily be produced for balls. The remaining analysis of the straightforward process or supervise To be also also also also (b), bleforth data is any while overlapping that process significant durates, durates of the straightforward data.

distributions that produced the data are well separated, then using the labels in combination with the data provides a simple way to classify all the unlabeled data points. Supervised algorithms also perform poorly if the data distributions have significant overlap.

Supervised and unsupervised learning can be stated mathematically. Let

$$D \subset \mathbb{R}^n$$
(5.1)

so that D is an open bounded set of dimension n. Further, let

$$D' \subset D$$
. (5.2)

The goal of classification is to build a classifier labeling all data in D given data from D'.

To make our problem statement more precise, consider a set of data points $x_j \in \mathbb{R}^n$ and labels y_j for each point where $j = 1, 2, \cdots, m$. Labels for the data can come in many forms, from numeric values, including integer labels, to text strings. For simplicity, we will label the data in a binary way se there plus or minus one so that $y_j \in \{\pm1\}$. For unsupervised learning, the following inputs and outputs are then associated with learning a classification task

Input
data
$$\{x_j \in \mathbb{R}^n, j \in \mathbb{Z} := \{1, 2, \dots, m\}\}$$

(5.3a)

Output

labels
$$\{y_i \in [\pm 1], i \in Z\}$$
. (5.3b)

Thus the mathematical framing of unsupervised learning is focused on producing labels y_j for all the data. Generally, the data x_j used for training the classifier is from D^* . The classifier is then more broadly applied, i.e. it generalizes, to the open bounded domain D. If the data used to build a classifier only samples a small portion of the larger domain, then it is often the uset that the classifier will not researching well.

Supervised learning provides labels for the training stage. The inputs and outputs for this learning classification task can be stated as follows

Inout

data
$$\{\mathbf{x}_j \in \mathbb{R}^n, j \in \mathbb{Z} := \{1, 2, \dots, m\}\}$$
 (5.4a)

labels
$$\{y_j \in \{\pm 1\}, j \in \mathbb{Z}' \subset \mathbb{Z}\}$$
 (5.4b)

Output

labels
$$\{y_i \in \{\pm 1\}, i \in Z\}$$
. (5.4c)

In this case, a subset of the data is labeled and the missing labels are provided for the remaining dath. Technically speaking, this is a semi-supervised learning task incressoner of the training labels are missing. For supervised learning, all the labels are known in order to build the classifier on 7⁻. The classifier is then applied to 70. As with unsupervised learning, if the data used to build a classifier only samples a small portion of the larger domain, then it is often the case that the classifier will not generalize well.

For the data sets considered in our feature selection and data mining section, we can consider in more detail the key components required to build a classification model: $\mathbf{x}_i, \mathbf{y}_j$. D and D^* . The Fisher iris data of Fig. 5.1 is a classic example for which we can detail these quantities. We begin with the data collected

$$\mathbf{x}_i = \{\text{sepal length, sepal width, petal length, petal width}\}.$$
 (5.5)

Thus each iris measurement contains four data fields, or features, for our analysis. The labels can be one of the following

In this case the labels are text strings, and there are three of them. Note that in our formulation of supervised and unsupervised learning, there were only two outputs (binary) which were labeled either ±1. Generally, there can be many labels, and they are often text strings. Finally, there is the domain of the data. For this case

 $D' \in \{150 \text{ iris samples: } 50 \text{ setosa, } 50 \text{ versicolor, and } 50 \text{ virginica}\}$ (5.7)

and

$$D \in$$
 [the universe of setosa, versicolor and virginica irises]. (5.8)

We can similarly assess the dog and cat data as follows:

$$x_i = (64 \times 64 \text{ image} = 4096 \text{ pixels})$$
 (5.9)

where each dog and cat is labeled as

$$y_i = [dog, cat] = [1, -1]$$
. (5.10)

In this case the labels are text strings which can also be translated to numeric values. This is consistent with our formulation of supervised and unsupervised learning where there are only two outputs (binary) labeled either ±1. Finally, there is the domain of the data which is the string of the string of

$$D' \in (160 \text{ image samples}; 80 \text{ does and } 80 \text{ cats})$$
 (5.11)

and

$$D \in$$
 (the universe of dogs and cats). (5.12)

Supervised and unsupervised learning methods aim to either create algorithms for classification, clustering or regression. The discussion above is a general strategy for clusification, the previous chapter discusses regression architectures. For both tasks, the gual is to build a modif form data on 2th that can generalize to 2th, as already-shows in the preceding chapter on regression, generalization can be vary difficult and cross-volidation strategies are critical. Deep neural networks, which are started of the and multiple algorithms for regression and classification, often have difficulty generalizing. Creating strane researchaizon behaves is in the forefront of machine learning research.

Some of the difficulties in generalization can be illustrated in Fig. 5.8. These data sets, although easily classified and clustered through visual inspection can be difficulty for many regression and classification schemes. Essentially, the boundary between the data forms a nonlinear manifold that is often difficult to characterize. Moreover, if the sampling data T' only catures a nortion of the manifold, then a classification or remession model will



Figure 3.2 Classification and regression models for data can be difficult when the data have routinear fractions which separate them. In this case, the function separating the green and magerita balls are be difficult to extract. Moreover, if only a small sample of the data *D'* is available, then a generalizable model may be impossible to construct (*a D'*. The left data set (*d*) represents two half-more shapes that are just superimposed while the concentric ring in (b) requires a circle as a stranulation busches between the data. Both an e collatorium to revolve.

almost survey hall in characterizing 20. These are also only two-dimensional depictions of a classification problem. In its not difficulty issuing how complicated such data structhodings and he in higher dimensional space. Visualization in such cases is essentially impossible and one must bey on algorithms to extract the meaningful boundaries separating data. What follows in this charges and the east are mechadis for classification and argression gives data on 20^T that may or may one buildelikel. There is quite a diversity of mathematical methods available for performing such tasks.

5.3 Unsupervised Learning: k-means Clustering

A variety of supervised and unsupervised algorithms will be highlighted in this chapter. We will start with one of the most promisem unsupervised algorithms in use todays - kneans, clustering. The k-means algorithm assumes one is given a set of vector valued data with the goal of partitioning or observations into the clusters. Each observation is tableed as belonging to a cluster with the nearest mean, which serves as a proxy (prototype) for that cluster. This results in a partitionine of the data sustein the Vortes effects.

Although the number of observations and dimension of the system are known, be manber of partitions it is generally values and and and how the detrimited, the users simply shows a number of clusters to extract from the data. The I-means algorithm the means and the algorithm has conserged. Fig. 5.8 depicts the spatial read of the the means algorithm. The algorithm has conserged Fig. 5.8 depicts the spatial read of the 1.4 deficient strange of the strange of the strange of the spatial read of the strange of absention of the strange of the strange of the strange of the strange of the observation as belonging to the storest mean, (iii) One the lefting is completed, faith the other of strange of the other of strange of the strange of t



Figure 32 Illustration of the k-means algorithm for k = 2. Two initial starting values of the man are given (black +). Each point is labeled as belonging to one of the two means. The green balls are thus labeled as part of the claster with the kft + and the magneta balls are labeled as part of the right +. Once labeled, the mean of the two clasters is recomputed (red +). The process is repeated until the means conserve.

start back at step (i) in the algorithm. This is a heuristic algorithm that was first proposed by Stuart Lloyd in 1957 [339], although it was not published until 1982.

The k-means objective can be stated formally in terms of an optimization problem. Specifically, the following minimization describes this process

$$\operatorname{argmin}_{\mu_j} \sum_{j=1}^{k} \sum_{\mathbf{x}_j \in \mathcal{D}'_j} \|\mathbf{x}_j - \mu_j\|^2 \qquad (5.13)$$

where the μ_j denote the mean of the jth cluster and T_j^j denotes the subdomino of data associated with that cluster. This minimizes the within-cluster sum of squares. In general, solving the optimization problem as stated is NP-hard, making it computationally intractable. However, there a number of heuristic algorithms that provide good performance despite not having a guarantee that they will converge to the globally objimal solution.

Cross-sultation of the k-means algorithm, as well as any machine learning algorithm, is critical for determining its effectiveness. Without lebst be new sultations metcheds in more manaced as there is no ground ruth to compare with. The cross-sultation metcheds the field as sectods, however, can still be used to so the the obstances of the classifier to different generate Lloyd's algorithm for k-means choiceing We fint consider making two clusters of data and the sectod structure of the sectod structure of the sectod structure of the sectod generate Lloyd's algorithm for k-means choiceing We fint consider making two clusters of data and particulum field data is the sectod structure of the sect

Code 5.5 k-means data generation

```
Linear is interpret inter
distribution of the state
is made aligns : constraint of (, c)
is made aligns : c)
is mad
```

Fig. 5.11 shows the data generated from two distinct Gaussian distributions. In this case, we have ground truth data to check the *k*-means clustering against. In general, this is not the case. The Lloyd algorithm gausses the number of clusters and the initial cluster means and then proceeds to oughted them in an iterative fadios. A means is sensitive to the initial guess and many modern versions of the algorithm also provide principled strategies for initialization.



Figure 5.10 Elastration of the A-means iteration procedure based upon Lloyd's algorithm [339]. Two chatters are sought so that A = 2. The initial gamess (black circles in panel (a)) are used to initially likel all the data according to their distance from each initial games for the mean. The means are then updated by computing the means of the newly labeled data. This two-stage heuristic converges after areovinatively four iterations.

Code 5.6 Lloyd algorithm for k-means.

```
[sh(-1, sh), = fractial pass
status
is joint
for joint leased[);
for joint leased[);
disconfigure();
disc
```

Fig. 5.10 shows the iterative procedure of the *k*-means clustering. The two initial guesses are used to initially label all the data points (Fig. 5.10(a)). New means are computed and the data relabeled. After only four iterations, the clusters converge. This algorithm was explicitly developed here to show how the iteration procedure rapidly provides an unsupervised labeling of all of the data. MATL AB has a built in *k*-means algorithm that only requires a



Figure 5.11 4-means clustering of the data using MATLAB's means command. Only the data and number of clusters need be specified. (a) The training data is used to produce a desirion line (black line) separating the clusters. Note that the line is clearly not optimal. The classification line can then be used on withheld data to too the accuracy of the algorithm. For the test data, one (of 50) magnetas hall would be michabed with its wirk (50) mean tabila are mishabed.

data matrix and the number of clusters desired. It is simple to use and provides a valuable diagnostic tool for data. The following code uses the MATLAB command mean and also extracts the *decision lines generated* from the algorithm separating the two clusters.

```
Code 5.7 k-means using MATLAB.
```

```
1 increase contained and a second sec
```

Fig. 5.11 shows the results of the k-means algorithm and depicts the decision lise separing the data into two clusters. The genera and magnetine balls donce the two labels of the data, showing that the k-means line does not correctly extract the labels. Indeed, a supervised algorithm is more prodiction: in extracting the ground truth results, as will be shown later in this chapter. Regardless, the algorithm does get a majority of the data labeled correctly.

The success of k-means is based on two factors: (i) no supervision is required, and (ii) it is a fast heuristic algorithm. The example here shows that the method is not very accurate, but this is often the case in unsupervised methods as the algorithm has limited knowledge of the data. Cross-validation efforts, such as k-fold cross-validation, can help improve the model and make the unsupervised learning more accurate, but it will generally be less accurate than a supervised algorithm that has labeled data.

5.4 Unsupervised Hierarchical Clustering: Dendrogram

Another commonly used unsupervised algorithm for clustering data is a *dowlorgoma*. Like *k*-marce shortering, *domlorgama* are created from a simple hierarchical algorithm, allowing one to efficiently visualize if data is clustered without any labeling or supervision. This hierarchical approach will be applied to the data libratura da Irg6, 512 where a ground truth is hown. Hierarchical clustering methods are generated either from a top-down or a bottom-an anteroach. Stecifically, the are one of two two:

Agglomerative: Each data point x_j is its own cluster initially. The data is merged in pairs as one creates a hierarchy of clusters. The merging of data eventually stops once all the data has been merged into a single über cluster. This is the bottom-up approach in hierarchical clusterine.

Divisive: In this case, all the observations \mathbf{x}_j are initially part of a single giant cluster. The data is then recursively split into smaller and smaller clusters. The splitting continues until the algorithm stops according to a user specified objective. The divisive method can split the data until each data point is its rown node.

In general, the merging and splitting of data is accomplished with a heuristic, greedy algorithm which is easy to execute computationally. The results of hierarchical clustering are usually presented in a denforgram.



Figure 5.12 Example data used for construction of a dendrogram. The data is constructed from two Gaussian distributions (50 points cach) that are easy to discern through a visual inspection. The dendrogram will produce a hierarchy that ideally would separate green halls from magenta bulb.



Figure 3.23 Illustrations of the agglomerative hierarchical clustering scheme applied to four data points. In the algorithm, the distance between the four data points is compared. Intailuly the Eardiant distance between points 2 and 3 is closest. Firstin 2 and 3 are now merged into a point distance and the scheme and the distances are not energiant compared. The dischargement on the right the length of the heardness of the dendorgram trees are directly related to the distance between the merged points.

In this section, we will focus on agglomerative hierarchical clustering and the dendrogram command from MATLAB. Like the Lloyd algorithm for *k*-mass clustering, building the dendrogram proceeds from a simple algorithmic structure based on computing the distance brivere and angoints. Although we typically use a Backlean distance, there are a number of important distance metrics one might consider for different types of data. Some typical distance are given as follow:

- Euclidean distance $||\mathbf{x}_i \mathbf{x}_i||_2$ (5.14a)
- Squared Euclidean distance $||\mathbf{x}_i \mathbf{x}_k||_2^2$ (5.14b)
- Manhattan distance $||\mathbf{x}_i \mathbf{x}_i||_1$ (5.14c)
- Maximum distance $||\mathbf{x}_1 \mathbf{x}_2||_{\infty}$ (5.14d)

Mahalanobis distance
$$\sqrt{(\mathbf{x}_j - \mathbf{x}_k)^T \mathbf{C}^{-1} (\mathbf{x}_j - \mathbf{x}_k)}$$
 (5.14e)

where C⁻¹ is the covariance matrix. As already illustrated in the previous chapter, the choice of norm can make a treamendous difference for exposing patterns in the data that can be extobled for clusterine and classification.

The dendrogram algorithm is shown in Fig. 5.13. The algorithm is as follows: (i) the distance between all *m* data points x_j is computed (the figure illustrates the use of a Euclidian distance), (ii) the closest two data points are merged into a single new data point midway between their original locations, and (iii) repeat the calculation with the new *m* = 1 points.



Figure 5.14 Dendrogram structure produced from the data in Fig. 5.12. The dendrogram shows which points are merged as well as the distance between points. Two clusters are generated for this level of threshold.

The algorithm continues until the data has been hierarchically merged into a single data point.

¹ The following code performs a hierarchical clustering using the dendrogram command from MATLAB. The example we use is the same as that considered for k-means clustering. Fig. 5.12 shows the data under consideration. Visual inspection shows two clear clusters that are easily discernible. As with k-means, our goal is to see how well a dendrogram can extract the two clusters.

Code 5.8 Dendrogram for unsupervised clustering.

```
Yis[X1(1:50,:), X2(1:50,:]);
Y2 = pdimt(Y1,'auclidean');
Z = linkage(Y2,'average');
thresh=0.85=max(Z(:,1));
(N,T,0)=dendrogram (Z,100,'ColorThreshold',thresh);
```

Fig. 5.14 shows the dendrogram associated with the data in Fig. 5.12. The structure of the algorithm shows which points are merged as well as the distance between points. The threshold command is important in labeling where each point belongs in the laterarchical scheme. By setting the threshold at different levels, there can be more or fewer clusters in the dendrogram. The following oed uses the output of the dendrogram to show how



Figure 151 Clustering outcome from dendogram routine. This is a summary of Fig. 5.14, showing how each of the points was chattered through the distance meric. The horizental red dated line shows where the ideal separation should occur. The first 50 points (green dots of Fig. 5.12) should be grouped on that they are body the red horizontal line in the lower left quadrature. The second 50 points (magneta dots of Fig. 5.12) should be grouped above the red horizontal line in the topper right quadratur. In summary, the dendogram on princisionif dot to prease points principal dots of the second state of t

the data was labeled. Recall that the first 50 data points are from the green cluster and the second 50 data points are from the magenta cluster.

Code 5.9 Dendrogram labels for cats and dogs.

```
bar(0), hold on
plot([0 100],[50 50],'r:','Linewidth',2)
plot([50.5 50.5],[0 100],'r:','Linewidth',2)
```

Fig. 5.15 shows how the data was clustered in the dendrogram. If perfect clustering had been achieved, then the first 50 points would have been below the horizontal dotted red line while the second 50 points would have been above the horizontal dotted red line. The vertical dotted red line is the line separating the green dots on the left from the magenta dots on the rint.

The following code shows how a greater number of clusters are generated by adjusting the threshold in the **dendrogram** command. This is equivalent to setting the number of clusters in *k*-means to something greater than two. Recall that one nurly has a ground truth to compare with when doing unsupervised clustering, so tuning the threshold becomes important.

```
threshe0.25*max(Z(:,3));
[N,T,O] edendrogram(Z,100, 'ColorThreshold', thresh);
```



Figure 5.10 Dendrogram structure produced from the data in Fig. 5.12 with a different threshold used than in Fig. 5.14. The dendrogram shows which points are merged as well as the distance between points. In this case, more than a dozen chusters are generated.

Fig. 5.16 shows a new dendrogram with a different threshold. Note that in this case, the hierarchical clustering produces more than a doore clusters. The training parameter can be seen to be critical for unsupervised clustering, much like choosing the number of clusters in k-means. In summary, buck k-means and hierarchical clustering provide a method whereby data can be parsed automatically into clusters. This provides a starting point for interpretations and analysis in data mining.

5.5 Mixture Models and the Expectation-Maximization Algorithm

The third sumpervised method we consider its laws as *pdime statuse model*. Often the models are assumed to be Gaussian distributions in which case the hardhol is known as *Gaussian mixture model*. (GMM). The basic assumption in this method is that distributions at *maximum constrainty model*. (GMM) model we that *may model*. The Like *k* - many *maximum model*. (GMM) model we that *may model*. The model is the *maximum model* (GMM) model we that *may model*. The *maximum maximum model* (GMM) model we that *may model*. The model has *maximum model* model and *maximum model* in a *Gaussian distribution maximum*. model has *Gaussian distribution may model maximum model*. The *Gaussian distribution maximum model* has *Gaussian distribution may model*. The *Gaussian distribution may model*. *maximum model* has *Gaussian distribution may model*. *maximum model* has *Gaussian distribution may maximum model*. The *Maximum model* has *maximum model* has *maximum model*. The *Maximum model* has *maximum maximum model*. This *Maximum model* has *maximum maximum maximum* The algorithm that enables the CMM computes the maximum-likelihood using the more Tope-control-Static Static (148). The IAM agent that is designed to the function of the algorithm that the static s

In any such iteration science, it is not obvious that the solution will coverege, or that the obtains is good, and the relycality full is no local value of the maximum Batchhood Bhat's can be proven that in this content; it does converge, and that the diversative of the observation of the solution of the solution of the solution of the solution of the observation of the solution of the minimum condition of the complexity and the complexity of the solution of

The fundamental assumption of the mixture model is that the probability density function (PDF) for observations of data x₁ is a weighted linear sum of a set of unknown distributions

$$f(\mathbf{x}_j, \mathbf{\Theta}) = \sum_{p=1}^{k} \alpha_p f_p(\mathbf{x}_j, \mathbf{\Theta}_p) \qquad (5.15)$$

where $f(\cdot)$ is the measured PDF, $f_F(\cdot)$ is the PDF of the mixture j, and i is the total mander of mixtures. Each of the PDFs $f_F(\cdot)$ is weighted by a_0 ($a_1 + a_2 + \cdots + a_n = 1$) and parametrized by an automatic wave total of parameters Φ_p . To state the objective of mixture models more precisely these: *Given the barries of \Phi_p*. To state the objective distinct weights a_p and B_n (B_p). More f_p is distribution Φ_p . Note that Φ is a scenar containing the objective distribution $f_1(\cdot)$.

For GMM, the parameters in the vector Θ_p are known to include only two variables: the mean μ_p and variance σ_p . Moreover, the distribution $f_p(\cdot)$ is normally distributed so that (5.15) becomes

$$f(\mathbf{x}_j, \mathbf{\Theta}) = \sum_{p=1}^{k} \alpha_p N_p(\mathbf{x}_j, \mu_p, \sigma_p). \qquad (5.16)$$

This gives a much more tractable framework since there are now a limited set of parameters. Thus once one assumes a number of mixtures k, then the task is to determine a_{μ} along with μ_{μ} and a_{μ} for each mixture. It should be noted that there are many other distributions beides Gaussian that can be imposed, but GMM are common since without prior knowledge, an assumption of Gaussian distribution is tryically assumed. An estimate of the parameter vector Θ can be computed using the maximum likelihood estimate (MLE) of Fisher. The MLE computes the value of Θ from the roots of

$$\frac{\partial L(\Theta)}{\partial \Theta} = 0$$
 (5.17)

where the log-likelihood function L is

$$L(\Theta) = \sum_{j=1}^{n} \log f(\mathbf{x}_{j}|\Theta) \qquad (5.18)$$

and the sum is over all the *n* data vectors **x**_j. The solution to this optimization problem, i.e. when the derivative is zero, produces a local maximizer. This maximizer can be computed using the EM algorithm since derivatives cannot be explicitly computed without an analytic form.

The EM algorithm starts by assuming an initial estimate (guess) of the parameter vector Θ . This estimate can be used to estimate

$$\tau_p(\mathbf{x}_j, \mathbf{\Theta}) = \frac{\alpha_p f_p(\mathbf{x}_j, \mathbf{\Theta}_p)}{f(\mathbf{x}_j, \mathbf{\Theta})} \qquad (5.19)$$

which is the posterior probability of component membership of x_j in the *p*th distribution. In other words, does x_j belong to the *p*th mixture? The E-step of the EM algorithm uses this posterior to compute memberships. For GMM, the algorithm proceeds as follows: Given an initial parametrization of Θ and a_n , compute

$$r_p^{(k)}(\mathbf{x}_j) = \frac{a_p^{(k)} N_p(\mathbf{x}_j, \mu_p^{(k)}, \sigma_p^{(k)})}{N(\mathbf{x}_j, \Theta^{(k)})}.$$
 (5.20)

With an estimated posterior probability, the M-step of the algorithm then updates the parameters and mixture weights

$$a_p^{(k+1)} = \frac{1}{n} \sum_{j=1}^{n} r_p^{(k)}(\mathbf{x}_j)$$
 (5.21a)

$$\mu_{\mu}^{(k+1)} = \frac{\sum_{j=1}^{n} x_j \tau_{\mu}^{(k)}(x_j)}{\sum_{j=1}^{n} \tau_{\mu}^{(k)}(x_j)} \qquad (5.21b)$$

$$\Sigma_{p}^{(k+1)} = \frac{\sum_{j=1}^{n} \tau_{p}^{(k)}(\mathbf{x}_{j}) \left(\mathbf{x}_{j} - \mu_{p}^{(k+1)}\right) \left(\mathbf{x}_{j} - \mu_{p}^{(k+1)}\right)^{T}}{\sum_{j=1}^{n} \tau_{p}^{(k)}(\mathbf{x}_{j})}$$
(5.21c)

where the matrix $\Sigma^{(k+1)}$ is the covariance matrix containing the variance parameters. The E- and M-scope are alternated unal convergence within a specified totarcace. Recall that to initialize the algorithm, the number of mixture models *i* must be specified and an initial gament strateging (gament) and the *i*-manned and the strateging of the str

The GMM is popular since it simply fits & Gaussian distributions to data, which is reasonable for unsupervised learning. The GMM algorithm also has a stronger theoretical base than most unsupervised methods as both k-means and hierarchical clustering are



Figure 5.17 GMM fit of the second and fourth principal components of the dog and cat wavelet image data. The two Gaussian sare well placed over the distinct dog and cat features as shown in (a). The PDF of the Gaussian models extracted are highlighted in (b) in arbitrary units.

simply defined as algorithms. The primary assumption in GMM is the number of clusters and the form of the distribution $f(\cdot)$.

The following code executes a GMM model on the second and fourth principal components of the dog and cat wavelet image data introduced previously in Figs. 5.4-5.6. Thus the features are the second and fourth columns of the right singular vector of the SVD. The **figmidit** command is used to extract the mixture model.

Code 5.10 Gaussian mixture model for cats versus dogs.

```
dogsatev(i,2:2:4);
@Modelsfitgmdist(dogsat,2)
ATCs GMModel.ATC
subplot(2,2:1)
hescontour((x1,x2)pdf(GMModel.[x1 x2]));
mubplot(2,2:2)
hesemash(d(x1,x2)pdf(GMModel.[x1 x2]));
```

The results of the algorithm can be plotted for visual inspection, and the parameters associated with each Gaussian are given. Specifically, the mixing proportion of each model along with the mean in each of the two dimensions of the feature space. The following is displayed to the screen.

```
Component 1:

Mixing proportion: 0.35535

Mean: -0.0290 -0.0753

Component 2:

Mixing proportion: 0.644465

Mean: 0.0076

AIC =

-792.8105
```

The code can also produce an AIC score for how well the mixture of Gaussians explain the data. This gives a principled method for cross-validating in order to determine the number of mixtures required to describe the data.



Figure 3.3 Illustration of linear descriminant analysis (LDA), The LDA optimization method produces an optimal dimensionibility production to a decision find of calsolicitation. The figure illustrates her projection of data onto the second and fourth principal component modes of the dogs and cal wavefet data considered in Fig. 2.5 A without optimization, a general projection can lead to very poor discrimination between the data. However, the LDA separates the probability distribution fractions in an optimal way.

Fig. 5.17 shows the results of the GMM fitting procedure along with the original data of cats and dogs. The Gaussians produced from the fitting procedure are also illustrated. The **fitgmdist** command can also be used with **cluster** to label new data from the feature separation discovered by GMM.

5.6 Supervised Learning and Linear Discriminants

We now more attention to supervise learning methods. One of the actients apprecised methods for elasticitotical dama and onlycelity performs 1956 in the control of transorony [112]. In *Internal Antividual dama and activation* performs the standard beningeretion of the standard state of the state is stated as a state of the state of the

The supervised learning architecture includes a training and withhold set of data. The withhold set is never used to train the classifier. However, the training data can be partituded into i-folds, for instance, to help build a tetter classification model. The large data is the original data is the star of the

For a two-class LDA, this results in the following mathematical formulation. Construct a projection w such that

$$\mathbf{w} = \arg \max_{\mathbf{w}} \frac{\mathbf{w}^T \mathbf{S}_B \mathbf{w}}{\mathbf{w}^T \mathbf{S}_W \mathbf{w}} \qquad (5.22)$$

where the scatter matrices for between-class Sn and within-class Sw data are given by

$$\hat{S}_{B} = (\mu_{2} - \mu_{1})(\mu_{2} - \mu_{1})^{T} \qquad (5.23)$$

$$S_W = \sum_{j=1}^{n} \sum_{\mathbf{x}} (\mathbf{x} - \mu_j) (\mathbf{x} - \mu_j)^T$$
. (5.24)

These quantities essentially measure the variance of the data sets as well as the variance of the difference in the means. The criterion in (5.22) is commonly known as the generalized Rayleigh quotient whose solution can be found via the generalized epicensile problem

$$S_B w = \lambda S_W w$$
 (5.25)

where the maximum eigenvalue \u03c4 and its associated eigenvector gives the quantity of interest and the projection basis. Thus, once the scatter matrices are constructed, the generalized eigenvectors can be constructed with MATLAB.

Performing at LDs analysis in MATAIN is simple. One seeds only segminize the matrix maps of web Matei (see in our less of matrix maps of the matrix web schem and the sequence is not ad any correspondent to the matrix maps of the matrix web schema and the matrix maps of the matrix maps. For implicit, the matrix maps of the matrix maps

Code 5.11 LDA analysis of dogs versus cats.

```
load cstData_w.mat
load dogData_w.mat
CDs[dog_wave cat_wave];
[u, m, v] swvd(CD-mean(CD(:)));
xtrains[v(1:60,2:2:4); v(81:140,2:2:4)];
```



Figure 5.19 Depiction of the performance achieved for classification using the second and fourth principal component modes. The top two punchs are PCA modes (features) used to build a classifier. The labels returned are either $y_j \in \{\pm1\}$. The ground truth answer in this case should produce a vector of 20 ones followed by 20 negative ones.

```
label=[cnex(60,1); -1.cnex(60,1)];
tests(v(61:80,2:2:4); v(141:160,2:2:4)];
class=classify(test,xtrain,label);
truths[cnex(20,1); -1.cnex(20,1)];
Ex100-zum(0.5.abs(class-truth))/40+100
```

Of course, the data is fairly limited and cross-validation should always be performed to evaluate the classifier. The following code runs 100 trials of the **classifi** command where 60 dors and canism the remaining 20 images. Code 5.12 Cross-validation of the LDA analysis.

```
for jpi1160/ GOT ______ remndprem(60);
indire(1:60), indire(1:60)-60;
indire(1:60), indire(1:60)-60;
xtrains[v(ind1,2:24); v(ind2,2:24)];
texts(v(ind1,2:24), v(ind2,2:24)];
labels(onse(60,1), -1,cess(60,1)];
truths(onse(60,1), -1,cess(60,1)];
datasclamify(text,xtrain,1&dd);
d(j);ense(dc)(21, v); -1,cess(20,1)];
d(j);ense(dc)(21, v); -1,cess(21, v); -1,cess(21
```

Fig. 5.20 shows the results of the cross-validation over 100 trials. Note the variability has an occur from trial to util. Specificially, the performance can achieve 100%, the can also be as low as 40%, which is worse than a coint flip. The survey flap coint is the second trial period trial of the strength periliphild in the regression chapter, is character as the second strength of the second trial period trial peri

In addition to a linear discriminant line, a quadratic discriminant line can be found to separate the data. Indeed, the **classify** command in MATLAB allows one to not only produce the classifier, but also extract the line of separation between the data. The following



Figure 520 Performance of the LDA over 100 trials. Note the variability that can occur in the classifier depending on which data is selected for training and testing. This highlights the immediate of cross-validation for building a robust classifier.

commands are used to produce labels for new data as well as the discrimination line between the dogs and cats.

Code 5.13 Plotting the linear and quadratic discrimination lines

Fig. 5.21 shows the dog and cat data along with the linear and quadratic lines separating them. This linear capadratic fit is found in the structured variable coeff which is returned with classify. The quadratic line of separation can often offer a little more flexibility when ruling to fit houndries separating data. A major advantage of LDA based methods: they are casily interpretable and casiy to compute. Thus, they are widely used across many branches of the sciences for classification of data.

5.7 Support Vector Machines (SVM)

One of the most successful data mining methods developed to date is the support vector machine (SVM). It is a core machine learning tool that is used widely in industry and science, often providing results that are better than competing methods. Along with the randow/porst algorithm, they have been pillars of machine learning in the last few decades. With enough running data, the SVM can now be replaced with deep neural nets. But



Figure 521 Classification line for (a) linear discriminant (LDA) and (b) quadratic discriminant (QDA) for dog (green dots) versus car (magenta dots) data projected onto the second and fourth principal components. This two dimensional feature space allows for a good discrimination in the data. The two lines represent the bost line and parabola for separating the data for a given training samele.

otherwise, SVM and random forest are frequently used algorithms for applications where the best classification scores are required.

The original SVM algorithm by Vapnik and Chervonenkis evolved out of the statistical learning literature in 1963, where hyperplanes are optimized to split the data into distinct clusters. Nearly three decades later, Boser, Guyon and Vapnik retraeted nonlinear clussifiers by applying the kernel trick to maximum-margin hyperplanes [70]. The current standard incurration (soft margin) was proposed by Cortes and Vapnik in the mid-1996 [138].

Linear SVM

The key idea of the linear SVM method is to construct a hyperplane

$$w \cdot x + b = 0$$
 (5.26)

where the vector w and constant b parametrize the hyperplane. Fig. 522 shows two potential hyperplanes splitting as of of all. Each has a different value of w and constant A. The optimized potential splitting and the state of the state of the splitting and the state of the state of the splitting and the



Figure 3.2 The SVM classification scheme constructs a hyperplane $\mathbf{w} + \mathbf{s} + \mathbf{b} = 0$ that optimally separates the hidded date. The series of the margin separating the hidded data is maximali (a) and much less (n b). Determining the vector \mathbf{w} and parameter b is the goal of the SVM optimization, such can let for data to the right of the hyperparate $\mathbf{w} + \mathbf{s} > 0$, and the ordination the hidde $\mathbf{v} + \mathbf{s} > 0$. Then the constraints labels $\mathbf{y}_{i} \in [2, 1]$ for the data is the left or right of the hyperplane is given by the left hidde (h) the structure scheme (h) the rest of the neuron scheme (h) the neuron vectors.

the data to the left or right of the hyperplane is given by

$$\mathbf{y}_{j}(\mathbf{w} \cdot \mathbf{x}_{j}+b) = \operatorname{sign}(\mathbf{w} \cdot \mathbf{x}_{j}+b) = \begin{cases} +1 & \operatorname{magenta \ ball} \\ -1 & \operatorname{green \ ball.} \end{cases}$$
(5.27)

Thus the classifier y₁ is explicitly dependent on the position of x₁.

Critical to the success of the SVM is determining w and b in a principled way. As with all machine learning methods, an appropriate optimization must be formulated. The optimization is nued at both minimizing the number of misclassified data points as well as creating the largest margin possible. To construct the optimization objective function, we define a loss function

$$\ell(\mathbf{y}_j, \tilde{\mathbf{y}}_j) = \ell(\mathbf{y}_j, \operatorname{sign}(\mathbf{w} \cdot \mathbf{x}_j + b)) = \begin{cases} 0 & \text{if } \mathbf{y}_j = \operatorname{sign}(\mathbf{w} \cdot \mathbf{x}_j + b) \\ +1 & \text{if } \mathbf{y}_j \neq \operatorname{sign}(\mathbf{w} \cdot \mathbf{x}_j + b) \end{cases}$$
. (5.28)

Stated more simply

$$\ell(\mathbf{y}_j, \hat{\mathbf{y}}_j) = \begin{cases} 0 & \text{if data is correctly labeled} \\ +1 & \text{if data is incorrectly labeled} \end{cases}$$
(5.29)

Thus each mislabeled point produces a loss of unity. The training error over m data points is the sum of the loss functions $\ell(y_i, \hat{y}_i)$.

In addition to minimizing the loss function, the goal is also to make the margin as large as possible. We can then frame the linear SVM optimization problem as

$$\underset{\mathbf{w}, b}{\operatorname{argmin}} \sum_{j=1}^{m} \ell(\mathbf{y}_{j}, \hat{\mathbf{y}}_{j}) + \frac{1}{2} \|\mathbf{w}\|^{2} \quad \text{subject to} \quad \min_{j} |\mathbf{x}_{j} \cdot \mathbf{w}| = 1. \quad (5.30)$$

Although this is a concise statement of the optimization problem, the fact that the loss function is discrete and constructed from ones and zeros makes it very difficult to actually optimize. Most optimization algorithms are based on some form of gradient descent which requires smooth objective functions in order to compute derivatives or gradients to update the solution. A more common formulation then is view to by

$$\operatorname{argmin}_{\mathbf{w},b} \sum_{j=1}^{m} H(\mathbf{y}_{j}, \bar{\mathbf{y}}_{j}) + \frac{1}{2} \|\mathbf{w}\|^{2} \text{ subject to } \min_{j} |\mathbf{x}_{j} \cdot \mathbf{w}| = 1 \quad (5.31)$$

where α is the weighting of the loss function and $H(z) = \max(0, 1-z)$ is called a Hinge loss function. This is a smooth function that counts the number of errors in a linear way and that allows for piecewise differentiation so that standard optimization routines can be employed.

Nonlinear SVM

Although easily interpretable, linear classifiers are of limited value. They are simply too rearries for data methoded in a high-dimensional space and which may have the structured separation as illustrated in Fig. 5.8. To build more sophisticated classification curves, the feature space for SVM must be extricted. SVM does the values *i* and then building hyperplanes in this new space. To do this, one simply maps the data into a nonlinear, higher-dimensional space

$$\mathbf{x} \mapsto \mathbf{\Phi}(\mathbf{x})$$
. (5.32)

We can call the $\Phi(x)$ new observables of the data. The SVM algorithm now learns the hyperplanes that optimally split the data into distinct clusters in a new space. Thus one now considers the hyperplane function

$$f(\mathbf{x}) = \mathbf{w} \cdot \mathbf{\Phi}(\mathbf{x}) + b \qquad (5.33)$$

with corresponding labels $y_i \in \{\pm 1\}$ for each point $f(x_i)$.

This simple idea, of enriching feature space by defining new functions of the data \mathbf{x}_i is exceptionally powerful for clustering and classification. As a simple example, consider two dimensional data $\mathbf{x} = (x_1, x_2)$. One can easily enrich the space by considering polynomials of the data.

$$(x_1, x_2) \mapsto (z_1, z_2, z_3) := (x_1, x_2, x_1^2 + x_2^2).$$
 (5.34)

This gives a new set of polynomial coordinates in x_1 and x_2 that can be used to embed the data. This photopsy is simplet; by embedding the data in a higher dimensional space, it is much more likely to be separable by hyperplanes. As a simple camplet, consider that data influentation if $x_1 \leq 3.00$, A finase calculate (to hyperplane) in the $x_1 > x_2$ plane will there dimensional space which can be easily separated by a hyperplane as illustrated in Fig. 5.23.

The ability of SVM to embed in higher-dimensional nonlinear spaces makes it one of the most successful machine learning algorithms developed. The underlying optimization algorithm (5.31) remains unchanged, except that the previous lubeling function \hat{y}_j = sign(w, x, +b) is now

$$\hat{\mathbf{y}}_i = \operatorname{sign}(\mathbf{w} \cdot \mathbf{\Phi}(\mathbf{x}_i) + b).$$
 (5.35)

The function $\Phi(\mathbf{x})$ specifies the enriched space of observables. As a general rule, more features are better for classification.

Kernel Methods for SVM

Despte its promise, the SVM method of building nonlinear classifiers by enriching in higher-dimensions leads to a computationally intractable optimization. Specifically, the large number of additional features leads to the *curve of dimensionality*. This computing the vectors w is problisively expensive and may not even be represented explicitly in memory. The *lersel trick* solves this problem. In this scenario, the w vector is represented as follows:

$$\mathbf{w} = \sum_{j=1}^{m} \alpha_j \Phi(\mathbf{x}_j) \qquad (5.36)$$

where α_j are parameters that weight the different nonlinear observable functions $\Phi(\mathbf{x}_j)$. Thus the vector w is expanded in the observable set of functions. We can then generalize (5.33) to the following

$$f(\mathbf{x}) = \sum_{j=1}^{m} \alpha_j \Phi(\mathbf{x}_j) \cdot \Phi(\mathbf{x}) + b. \qquad (5.37)$$



Figure 3.21 The nonlinear embedding of Fig. 5.8(b) using the variables (iii) (z_1 , z_2) + (z_1 , z_2) = (z_1 , z_2) + (z_1) = (z_1). An Alyreptimes can now easily separate the green from magneta halb, aboving that former chosofication can be accomplicated intry by the encircling the measurement uses of the data. Varian linear field in the original coordinate separation on he achieved with for gates $z_2 \approx 14$ (halds green plane), the original coordinate system the integer a called coordination in the second plane) is the original coordinate system the integer a called coordination for the second plane). The system is the second plane is

The kernel function [479] is then defined as

$$K(\mathbf{x}_{j}, \mathbf{x}) = \mathbf{\Phi}(\mathbf{x}_{j}) \cdot \Phi(\mathbf{x}).$$
 (5.38)

With this new definition of w, the optimization problem (5.31) becomes

$$\operatorname{argmin}_{\boldsymbol{a},\boldsymbol{b}} \sum_{j=1}^{m} H(\mathbf{y}_{j}, \mathbf{\bar{y}}_{j}) + \frac{1}{2} \| \sum_{j=1}^{m} \alpha_{j} \Phi(\mathbf{x}_{j}) \|^{2} \text{ subject to } \min_{j} |\mathbf{x}_{j} \cdot \mathbf{w}| = 1 \quad (5.39)$$

where α is the vector of α_j coefficients that must be determined in the minimization process. There are different conventions for representing the minimization. However, in this formulation, the minimization is now over α instead of w.

In this formulation, the kernel function R (sq. x) essentially allows us to represent Taylor series expansions of a large (infinite) number of observations in a compart way [FP9]. The kernel function enables one to operate in a high-dimensional, implicit feature space without ever comparing the coordinates of the data in that space, but match by simplicy comparing the interp products between all pairs of data in the feature space. For instance, two of the most commonly would kernel functions are

Radial basis functions (RBF):
$$K(\mathbf{x}_j, \mathbf{x}) = \exp(-\gamma ||\mathbf{x}_j - \mathbf{x}||^2)$$
 (5.40a)

Polynomial kernel:
$$K(\mathbf{x}_i, \mathbf{x}) = (\mathbf{x}_i \cdot \mathbf{x} + 1)^N$$
 (5.40b)

where N is the degree of polynomials to be considered, which is exceptionally large to evaluate without using the kernel trick; and γ is the width of the Gaussian kernel measuring the distance between individual data points x_j and the classification line. These functions can be differentiated in order to optimize (5.39).

This represents the major theoretical underpinning of the SVM method. It allows us to construct higher-dimensional spaces using observables generated by lexentel functions. Moreover, it results in a computationally tractable optimization. The following code shows the basis workings of the kernel method on the example of dog and art classification data. In the first example, a standard linear SVM is used, while in the second, the RBF is executed as an option.

Code 5.14 SVM classification

```
Lad attrict.com
Lad attrict.com
(c)dga even any even
(c)dga even
```

Note that in this code we have demonstrated some of the diagnostic features of the SVM method in MATLAB, including the cross-vulidation and class loss sccees that are associated with training. This is a superficial treatment of the SVM. Overall, SVM is one of the most sophisticated machine learning tools in MATLAB and there are many options that can be executed in order to tune performance and extract accurex/cross-vulidation metrics.

5.8 Classification Trees and Random Forest

Decision new are common in business. They estabilish an algorithmic flow charf for maining decisions bade on circuits har are deemain important and cleated on a decised oncome. Other the decision trees are constructed by experts with knowledge of the workflow intervels in the decision mice argorosom trees are commonline using and atm method have on the discrimination and regressions trees are come nucleus teaming and data method have on the discrimination and regressions trees are come nucleus teaming and data the one of the discrimination and regressions trees are come nucleus teaming and data the first main and the discrimination of the theorem of the discrimination of the theorem of the one of the discrimination of the theorem of the discrimination of the discriminating of the discrimination of the discriminatin The decision true is a hierarchical construct that fooks for optimal ways to split the data dord to provide a subst classification and expression. It is the opposite of the unsuperival dendrogram hierarchical clustering previously demonstrated. In this case, our goal is not to more from home up in the clustering process, but from options in order to create the best splits possible for classification. The fact that it is a supervised algorithm, which uses labeled data, allows us to split the data accordingly.

There are significant advantages in developing decision trees for classification and regression: (1) they often produce interpretable results that can be graphically displayed, making them easy to interpret even for nonexperts, (ii) they can handle numerical or congestical data squadity well, (iii) they nerform well with large data sets at scale, and (') of the model can be assessed, (iv) they perform well with large data sets at scale, and (') weeks.

As one might expect, the success of decision tree learning has produced a large number of innovations and algorithms for how to best split the data. The coverage here will be limited, but we will highlight the basic architecture for data splitting and tree construction. Recall that we have the following:

data
$$\{\mathbf{x}_j \in \mathbb{R}^n, j \in \mathbb{Z} := \{1, 2, \dots, m\}\}$$
 (5.41a)

labels
$$\{y_j \in \{\pm 1\}, j \in Z' \subset Z\}$$
. (5.41b)

The basic decision tree algorithm is futly simple: (i) scan through each component ((i)same) at (i = 1, 2, ... a) of the vector x y to identify the used σ_i that gives the best labeling prediction for γ_i . (ii) (compare the prediction accuracy for each split on the future γ_i . The future gives the best segmentation of the data is selected as the split for the tree. The algorithm terminates once the each individual data point is a unique cluster, known as $1\sigma_i$ on an own transition of the trees is selected by the individual data point is a unique cluster, known as $1\sigma_i$ on an own transition of the tree. This is secondarily the inverse of the dendrogram.

As a specific example, consider the Fisher iris data set from Fig. 5.1. For this data, each flower had four features (petal width and length, sepal width and length), and three labels (setosa, versicolor and virginica). There were fifty flowers of each variety for a total of 150 data points. Thus for this data the vector **x**, has the four components

$$x_1 = \text{sepal width}$$
 (5.42a)

$$x_2 = \text{sepal length}$$
 (5.42b)

$$x_3 = \text{petal width}$$
 (5.42c)

$$x_4 = \text{petal length.}$$
 (5.42d)

The decision tree algorithm scam over these four feature is noder to decide how to be the dust [0, 52] shows the splinging process in the space of the nurshifts x_1 through x_2 . [Instancial are two data planes containing x_1 versus x_2 (madel (b)) and x_2 through x_2 . [Instancial are two data planes containing x_1 versus x_1 (dust (d)) by and the split of the data maximally specarise the data. In fact, the decision tree performs the first split of the data that the split is performed at $x_1 = 4.50$. Other split are shown in Tabu process shows that the split is performed at $x_1 = 4.50$. Other split are shown. This process shows that the split is performed at $x_1 = 4.50$. Other split are shown in the split split are split to the split split is performed at $x_2 = 4.50$. Other split are split are shown in the split split is performed at $x_2 = 4.50$. (b) three split are shown in the split split is performed at $x_2 = 4.50$. (b) three split are shown in the split split is performed at $x_2 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at $x_3 = 4.50$. (b) three split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at $x_3 = 4.50$. (b) the split are shown in the split split is performed at



Figure 2.28 Indication of the splitting procedure for decision tree transing performance on the Fisher indicates. Each variable r_1 shows p_1 is consistent over to determine the boxer split of data which, the splitting of the splitting of



Figure 525 Tree structure generated by the MATLAB filterroe command. Note that only three splits are conducted, creating a classification tree that produces a class error of 4.67%

are clear visible. Moreover, the splitting does not occur on the x_1 and x_2 (width and length) variables as they do not provide a clear separation of the data. Fig. 5.25 shows the tree used for Fig. 5.24.

The following code fits a tree to the Fisher iris data. Note that the **fitctree** command allows for many options, including a cross-validation procedure (used in the code) and parameter tuning (not used in the code). Code 5.15 Decision tree classification of Fisher iris data.

```
lasd fishering:
treasfitteres(mas, species, 'MasHumfplits',1, 'GrossVal', 'on')
view(tres.Trained[1], Node', 'graph');
classError & Moldanes(trea)
xlamsas(1:50,i); & treatous
xlamsas(1:50,i); & treatous
xlamsas(1:50,i); & treatous
xlamsas(1:50,i); & treatous
```

The results of the splitting procedure are demonstrated in Fig. 5.25. The view command generates an interactive window showing the tree structure. The tree can be pruned and other diagnostics are shown in this interactive graphic format. The class error achieved for the Fisher iris data is 4.67%.

As a second example, we construct a decision tree to the classify dogs versus cats using our previously considered wavelet images. The following code loads and splits the data.

Code 5.16 Decision tree classification of dogs versus cats.

Fig. 5.25 shows the resulting classification tree. Note that the decision tree learning algotim identifies the first two splits as a corring along the z_2 and x_4 variables respectively. These two variables have been considered previously since their histograms show them to homo elistinguished the math other PCA components (See Fig. 5.3). For this splitting, which has been cross-validated, the class error achieved is approximately 16%, which can be compared with the 30% error of LDA.

As a final example, we consider census data that is included in MATLAB. The following order shows some important uses of the calsolitation and regression tree architecture. In particular, the variables included can be used to make associations between relationships in this case, the variables included can be used to make associations between relationships of the case of the variables included on the start of particular shows and the start matching of the start of the compared as shown in Fig. 2.37. The following code highlights some of the functionality of the tree architecture.

Code 5.17 Decision tree classification of census data

```
load census1994
X = adultdata(;/age','workClass','education_num','
marital_status','race','sex','capital_gain',...
'capital_loas','hours_per_week','salary'));
```



Figure 5.28 Tree structure generated by the MATLAB fitterine command for dog versus cat data. Note that only two splits are conducted, creating a classification tree that produces a class error of anerosimately 16%

```
M1 fittere(C,'salay','PredictorSelection','curvature','
Surreyste','on')
imp = predictorImportance(Ddl);
her(imp.'saccolor',[C.4.4],'HepGclor','k');
titl('Predictor'); ideal('Fredictors'); k = gen;
yiaka('Fredictors'); k = gen;
histochasteristics = 45'
```

As with the SVM algorithm, there exists a wide variety of tuning parameters for classification trees, and this is a superficial treatment. Overall, such trees are one of the most sophisticated machine learning tools in MATLAB and there are many options that can be executed to tune performance and extract accuracy/cross-validation metrics.

Random Forest Algorithms

Refere colong this section, it is important to mention Breinna's studios from (77) linear variants for decision learning trees. Random forests, are and secretific terming method for classification and regression. This is an important introvaluus is the decision tree could by upfilling and generally not robust for different samples of the data. Thus one can generate two significantly different classifications trees with the same strength of the data of the data of the data of the data of the same strength of the data of the random decision from correct for a decision trees. This of eventing to their training set, then mendition a new resolutions.



Figure 5.27 Importance of variables for prediction of salary data for the US census of 1994. The classification tree architecture allows for sophisticated treatment of data, including understanding how each variable contributes statistically by predicting a classification outcome.

There are many variants of the random forest architecture, including variants with loosing and longing. These will not be considered here except to mention that the MATLAB **figtree** explosis many of these techniques through its options. One ways to think, about member learning its huir allows for nobus classification meres. It often does this by focustions, and the state of the state of the state of the state of the state forests. Jugging and boosting are all extrasive unifocus in their own right, but have already been incorporated into localing orderware with their but focus intening trees.

5.9 Top 10 Algorithms in Data Mining 2008

This chapter has illustrated the tremendom diversity of uppertucts and sumpervised marks of available for the analysis of data. Although the dispatisment are now analy accessible through many commercial and open-source software packages, the difficulty is now early markine heating based based on the HEBE International Conference on Data Manga Manga Manga Markshol, and the HEBE International Conference on Data Manga Manga Manga Markshol, and the HEBE International Conference on Data Manga summary article, each algorithm was briefly described along with its impact and potential fitter directions of research. The 10 algorithm scoreed calsufficture, clustering, attricture learning, association analysis, and link mining, which are all among the nost important polycis in data mining screach and development. Interestingly, deep learning and neural networks, which are the topic of the next chapter, are not meetineed in the article. The landcage of data science would change significantly in 2021 with the ImageNTE data are, and deep convolutional neural networks began to dominate almost any meaningful meric for classification and regression accuracy.

In this section, we highlight their identified top 10 algorithms and the basic mathematical structure of each. Many of them have already been corrected in this chapter. This list is not exhaustive, nor does it rank them beyond their inclusion in the top 10 list. Our objective simply to highlight what was considered by the community as the state-of-the-art data maining took in 2008. We begin with those algorithms already considered previously in this chatter.

k-means

This is one of the workhorse unsupervised algorithms. As already demonstrated, the goal of k-mean is simply to cluster by proximity to a set of k points. By updating the locations of the k points according to the mean of the points closest to them, the algorithm iterates to the k-means. The structure of the MATLAB command is as follows

[labels, centers] skmeans (X, k)

The means command takes in data X and the number of prescribed clusters k. It returns labels for each point labels along with their location centers.

EM (mixture models)

Mixture models are the second workhorse algorithm for unsupervised learning. The assumption underlying the mixture models is that the observed data is produced by a mixture of different poolshilting distribution functions whose weightings are unknown. Moreover, the parameters must be estimated, thus nequiring the Expectation-Maximization (Equ) algorithm. The structure of the MATLAB command is as follows:

Model=fitgmdist(X,k)

where the **fitgmdist** by default fits Gaussian mixtures to the data **X** in k clusters. The **Model** output is a structured variable containing information on the probability distributions (mean, variance, etc.) along with the goodness-of-fit.

Support Vector Machine (SVM)

One of the most powerful and flexible supervised learning algorithms used for most of the 90s and 2005, the SVM is an exceptional of the-shelf method for elassification and regression. The main idea: project the data into higher dimensions and split the data with hyperplanes. Chitcai to maiking this work in practice work herverl with *G* refliciently evaluating inner products of functions in higher-dimensional space. The structure of the MATLAB communit is as follows:

Model = fitcsvm(xtrain, label);

test labels = predict(Model,test);

where the fitesym command takes in labeled training data denoted by train and label, and it produces a structured output Model. The structured output can be used along with the predict command to take test data test and produce labels (test_{labels}). There exist many options and tuning parameters for fitesym, making it one of the best of the-shelf methods.

CART (Classification and Regression Tree)

This was the subject of the last section and was demonstrated to provide another powerful technique of supervised learning. The underlying idea was to split the data in a principled and informed ways on so to produce an interpretable clustering of the data. The data splitting occurs along a single variable at a time to produce branches of the tree structure. The structure of the MATLAB commands is as follows:

tree = fitctree(xtrain, label);

where the fitctree command takes in labeled training data denoted by train and label, and it produces a structured output tree. There are many options and tuning parameters for fitctree, making it one of the best off-the-helf methods.

k-nearest Neighbors (kNN)

This is perhaps the simplest supervised algorithm to understand. It is highly interpretable and easy to execute. Given a new data point x_k which does not have a label, simply find the k nearest neighbors x_k with labels y_k. The label of the new point x_k is determined by a majority vote of the kNN. Given a model for the data, the MATLAB command to execute the kNN search is the following:

label = knnsearch(Mdl,test)

where the knnsearch uses the MdI to label the test data test.

Naive Bayes

The Naire Bayes algorithm provides an intrative framework for supervised learning. It is single to construct and does not require your complicated pursumer estimation, similar to SVM and/or classification trees. If further gives highly interpretable results that are remark, buy good in practice. The method is bused upon Bayes's theorem and the computation of prior probability distributions of the labeled data. The MATLAB command structure for constructing a Naire Bayes model is the following

Model = fitNsiveBayes(xtrain, label)

where the fiteNativeBayes command takes in labeled training data denoted by train and label, and it produces a structured output Model. The structured output can be used with the predict command to label test data test.

AdaBoost (Ensemble Learning and Boosting)

AdaBoost is an example of an ensemble learning algorithm [188]. Broadly speaking, AdaBoost is a form of random forest [77] which takes into account an ensemble of decision tree models. The way all booting algorithms work is to first consider an equal weighing for all raising data x. Boosting re-weights the importance of the data according to how difficult they are to classify. Thus the algorithm focuses on harder to classify data. Thus a family of weak kanness can be trained to yield a strong learner by booting the importance of hard is classify data [470]. This concept and its methaness are hance MTLAB. Communic of hard is a fellow.

ads = fitcensemble(xtrain, label, 'Method', 'AdaBoostM1')

where the fiteensemble command is a general ensemble learner that can do many more things than AdaBoost, including robust boosting and gradient boosting. Gradient boosting is one of the most powerful techniques [189].

C4.5 (Ensemble Learning of Decision Trees)

This algorithm is another variant of decision tree learning developed by J. P. Quinlan [43, 44]. At its core, the algorithm splits the data according to an information entropy score. In its latest version, it supports booting as well as many other well known functionalities to improve performance. Broadly, we can thit of this as a storng performing version of CART. The **fitterenemble** algorithm highlighted with AdaBoot gives a genetic executible summar architecture that can iconome decision trees. Allowing for a (C-3 His algorithm highlighted with AdaBoot gives a genetic executible summar architecture that can iconome decision trees. Allowing for a (C-3 His algorithm highlighted performance) and the summar architecture and ecosion trees.

Apriori Algorithm

The list two methods highlighted here tend to focus on different aspects of data mining, in the Appiori algorithm, for goal is to find frequest itensets from data. Although this may somal trivial, it is not since data sets tend to be very large and can easily produce N-based produce the combinatorial insure of the algorithm. The Appiori algorithm architecture (4), This algorithm can then be used for fast learning of associate rules in the data.

PageRank

The founding of Google by Sergey Brin and Larry Page revolved around the PageBank algorithm [32]. PageBank produces a start familia of variables, tota is web pages. No comparing an off-line value for each variable that does not depend on search querels. The PageBank is associated with graph theory as to originally interpreted a hyperlink from one can one can obsci on the similar to original programment and pageting theory of the similar to can also compare an avec. From this, and valoes modifications of the original algorithm, one can then compare an importance score to end variable and provides an ordered math list. The number of enhancements for this algorithm is quite large. Frobacing scorates cordering of variables (web pages) and their importance remains an antice topic of rescards.
Suggested Reading

Texts

- (1) Machine learning: a probabilistic perspective, by K. P. Murphy, 2012 [396].
- (2) Pattern recognition and machine learning, by C. M. Bishop, 2006 [64].
- (3) Pattern classification, by R. O. Duda, P. E. Hart, and D. G. Stork, 2000 [161].
- (4) An introduction to statistical learning, by G. James, D. Witten, T. Hastie and R. Tibshirani, 2013 [264].
- (5) Learning with kernels: support vector machines, regularization, optimization, and beyond, by B. Schölkopf and A. J. Smola, 2002 [479].
- (6) Classification and regression trees, by L. Breiman, J. Friedman, C. J. Stone and R. A. Olshen, 1984 [79].
- (7) Random forests, by L. Breiman, 2001 [77].

Papers and Reviews

- Top 10 algorithms in data mining, by X. Wu et al., Knowledge and information systems, 2008 [562].
- (2) The strength of weak learnability, by R. E. Schapire, Machine Learning, 1990 [470].
- (3) Greedy function approximation: a gradient boosting machine, by J. H. Friedman, Annals of Statistics, 2001 [189].

Neural networks (NNs) were inspired by the Nobel prize winning work of Hubel and Wiesel on the primary visual cortex of cats [259]. Their seminal experiments showed that neuronal networks were organized in hierarchical layers of cells for processing visual stimulus. The first mathematical model of the NN, termed the Neocornitron in 1980 [193]. had many of the characteristic features of today's deep convolutional NNs (or DCNNs). including a multi-layer structure, convolution, max pooling and nonlinear dynamical nodes. The recent success of DCNNs in computer vision has been enabled by two critical components: (i) the continued growth of computational power, and (ii) exceptionally large labeled data sets which take advantage of the power of a deep multi-layer architecture. Indeed, although the theoretical inception of NNs has an almost four-decade history, the analysis of the ImageNet data set in 2012 [310] provided a watershed moment for NNs and deep learning [324]. Prior to this data set, there were a number of data sets available with approximately tens of thousands of labeled images. ImageNet provided over 15 million labeled, high-resolution images with over 22,000 categories. DCNNs, which are only one potential category of NNs, have since transformed the field of computer vision by dominating the performance metrics in almost every meaningful computer vision task intended for classification and identification.

Although ImageNet has been critically enabling for the field, NNs were textbook material in the early 1990s with a focus typically on a small number of layers. Critical machine learning tasks such as principal component analysis (PCA) were shown to be intimately connected with networks which included back propagation. Importantly, there were a number of critical innovations which established multilayer feedforward networks as a class of universal approximators [255]. The past five years have seen tremendous advances in NN architectures, many designed and tailored for specific application areas. Innovations have come from algorithmic modifications that have led to significant performance gains in a variety of fields. These innovations include pretraining, dropout, inception modules, data augmentation with virtual examples, batch normalization, and/or residual learning (See Ref. [216] for a detailed exposition of NNs). This is only a partial list of potential algorithmic innovations, thus highlighting the continuing and rapid pace of progress in the field. Remarkably, NNs were not even listed as one of the top 10 algorithms of data mining in 2008 [562]. But a decade later, its undeniable and growing list of successes on challenge data sets make it perhaps the most important data mining tool for our emerging generation of scientists and engineers.

As already shown in the last two chapters, all of machine learning revolves fundamentally around optimization. NNs specifically optimize over a compositional function

$$\operatorname{argmin}_{\mathbf{A}_{j}} \left(f_{M}(\mathbf{A}_{M}, \cdots, f_{2}(\mathbf{A}_{2}, f_{1}(\mathbf{A}_{1}, \mathbf{x})) \cdots) + \lambda g(\mathbf{A}_{j}) \right)$$
(6.1)

which is often solved using succhastic gradient descent and back propagation algorithms. Beam hards A₄ denotes the weight concentration the section of the section of the section of the section of the section and regularization and experimental systems which is regularized by $g(A_1)$. Composition and regularization are critical for generating expressive preparations of the data and preventing overfining, respectively. This general optimization framework is at the center of deep kerning algorithms, and its to solution will be considered in this charger. Importantly, NNs have significant potential for overfining of datas to that const-validation must be carefully considered. Result data for dword from sub-induces to air dataset.

6.1 Neural Networks: 1-Layer Networks

The generic architecture of a multi-layer NN is shown in Fig. 6.1. For classification tasks, the goal of the NN is to map a set of input data to a classification. Specificality, we train the NN to accurately map the data x ₁ to their correct label y₁. As shown in Fig. 6.1, the input space has the dimension of the raw data x₁ $\in \mathbb{R}^{+}$. Becoupt layers have dimension of the raw data x₁ $\in \mathbb{R}^{+}$. Becoupt layers have dimension of the raw data x₁ $\in \mathbb{R}^{+}$. Becoupt layers have dimension of the raw data x₁ $\in \mathbb{R}^{+}$. Becoupt layers have dimension of the raw data x₁ $\in \mathbb{R}^{+}$. Becoupt layers have dimension of the following.

Immediately, one can see that three are a great number of design questions regarding NNs. How many hypers should be used? What should be the dimension of the layer? How should the comput layer be designed? Should one use all-to-all or sparsified connections between layer? How should the mapping between layers be performed. Zin *all arear* mapping or a *nonlinear* mapping? Much like the tuning options on SVM and classification trees. NNs have a significant number of design options that can be nucled to improve performance.

Initially, we consider the mapping between layers of Fig. 6.1. We denote the various layers between input and output as x⁽¹⁾ where k is the layer number. For a linear mapping between layers, the following relations hold

$$i^{(1)} = A_1 x$$
 (6.2a)

$$\mathbf{x}^{(2)} = \mathbf{A}_2 \mathbf{x}^{(1)}$$
 (6.2b)

$$y = A_3 x^{(2)}$$
. (6.2c)

This forms a compositional structure so that the mapping between input and output can be represented as

$$y = A_1A_2A_1x.$$
 (6.3)

This basic architecture can scale to M layers so that a general representation between input data and the output layer for a linear NN is given by

$$y = A_M A_{M-1} \cdots A_2 A_1 x.$$
 (6.4)

This is generally a highly underdetermined system that requires some constraints on the solution in order to select a unique solution. One constraint is immediately obvious: The mapping must generate *M* distinct matrices that give the best mapping. It should be noted



Figure 1: Electration of a neural net multiviture mapping an input layer to an order layer, γ the multiviture and the bidded bidded

that linear mappings, even with a compositional structure, can only produce a limited range of functional restonses due to the limitations of the linearity.

Notlinear mappings are also possible, and generally used, in constructing the NN. Indeed, nonlinear activation functions allow for a richer set of functional responses than their linear counterparts. In this case, the connections between layers are given by

$$\mathbf{x}^{(1)} = f_1(\mathbf{A}_1, \mathbf{x})$$
 (6.5a)

$$\mathbf{x}^{(2)} = f_2(\mathbf{A}_2, \mathbf{x}^{(1)})$$
 (6.5b)

$$y = f_3(A_3, x^{(2)}).$$
 (6.5c)

Note that we have used different nonlinear functions $f_j(\cdot)$ between layers. Often a single function is used; however, there is no constraint that this is necessary. In terms of mapping the data between input and output over M layers, the following is derived

$$y = f_M(A_M, \dots, f_2(A_2, f_1(A_1, x)) \dots)$$
 (6.6)

which can be compared with (6.1) for the general optimization which constructs the NN. As a highly underdetermined system, constraints should be imposed in order to extract a desired solution type, as in (6.1). For big data applications such as ImageNET and computer vision tasks, the optimization associated with this compositional framework is expensive given the number of variables that must be determined. However, for moderate sized networks, it can be performed on workstation and laptop computers. Modern stochastic gradient descent and back propagation algorithms enable this optimization, and both are covered in later sections.

A One-Layer Network

To gain insight into how an NN might be constructed, we will consider a single layer network that is optimized to build a classifier between dogs and cats. The dog and cat example was considered extensively in the previous chapter. Recall that we were given images of dogs and cats, or a wavelet version of dogs and cats. Fig. 6.2 shows our construction. To make this as simule as resolible, we consider the simule NN output

$$y = {dog, cat} = {+1, -1}$$
 (6.7)

which labels each data vector with an output $\mathbf{y} \in \{\pm 1\}$. In this case the output layer is a single node. As in previous supervised learning algorithms the goal is to determine a materian so that each data vector \mathbf{x} , is labeled correctly by \mathbf{v}_1 .

The easiest mapping is a linear mapping between the input images $x_j \in \mathbb{R}^n$ and the output layer. This gives a linear system AX = Y of the form

$$\mathbf{A}\mathbf{X} = \mathbf{Y} \rightarrow [a_1 \ a_2 \ \cdots \ a_n] \begin{bmatrix} | & | & | \\ \mathbf{x}_1 & \mathbf{x}_2 & \cdots & \mathbf{x}_n \\ | & | & | & | \end{bmatrix} = [+1 \ +1 \ \cdots \ -1 \ -1] \quad (6.8)$$



Figure 6.2: Single layer network for binary classification between dogs and cats. The output layer for this case is a perceptron with $y \in \{\pm 1\}$. A linear mapping between the input image space and output output layer can be constructed for training data by solving $A = YX^{\dagger}$. This gives a least square recression for the matrix A margine the images to label space.

where each column of the matrix X is a dog or cat image and the columns of Y are its corresponding labels. Since the output layer is a single node, both A and Y reduce to vectors. In this case, our goal is to determine the matrix (vector) A with components a_j . The simplext solution is to take the pseudo-inverse of the data matrix X

$$A = YX^{T}$$
. (6.9)

Thus a single output layer allows us to build a NN using least-square fitting. Of course, we could also solve this linear system in a variety of other ways, including with sparsitypromoting methods. The following code solves this problem through both least-square fitting (piny) and the LASSO.

```
Code 6.1 1-layer, linear neural network.
```

```
Ind contac_vont lead applics_vont (th(d_sum cont_vont)
indicating the sum of the sum of the sum of the sum of the
indicating the sum of the sum of the sum of the sum of the
application (th), is the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
application (the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the sum of the sum of the
sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the sum of the su
```

Figs. 5.3 and 6.4 show the results of this linear single-layer NN with single-node computer by prescription $\beta_{\rm eff}$ is non-single-layer in the single-layer matrix that the single-low computer layer shows that the single-layer layer layer layer in the single-layer layer (REI) probability with single-layer layer layer

6.2 Multi-Laver Networks and Activation Functions

The previous section constructed what is perhaps the simplest NN possible. It was linear, had a single layer, and a single output layer neuron. The potential generalizations are endless, but we will focus on two simple extensions of the NN in this section. The first seturesion concerns the assumption of linearity in which we assumed that there is a linear



Figure 3.2. Choselication of withheld data total on a trained, single-layer network with linear morphic herease many fixed party, and an angle ostpace, i.o. and () care the bar graph of the output layer score y (±1) achieved for the withheld data using a puscida-asswere for training and the LASSO for training properties/b. The results down it hold cases that dogs are more conference of the score of the production of the score of the score of the score of the score of the production of the score of the score of the score of the score of the formers, the suscepting the NN is highly space.

_		
_		
_		
The second second		
The second se		
_		
	_	
	_	
_		

Figure 8.4 Weightings of the matrix A reshoped into 3.2×32 arrays. The left matrix shows the matrix A compared by Lassa sparse preprison the pseudo-inverse while the right matrix shows the matrix A compared by LASSO. Both matrices provide similar classification scores on withheld dats. They finder provide interpretability in the sense that the results from the pseudo-inverse values of the further provide interpretability in the sense that the results from the pseudo-inverse value are alone can give the features required for distinguishing between dego and can.

transform from the image space to the output layer: Ax = y in (6.8). We highlight here common nonlinear transformations from input-to-output space represented by

$$y = f(A, x)$$
 (6.10)

where $f(\cdot)$ is a specified activation function (transfer function) for our mapping.

The linear mapping used previously, although simple, does not offer the flexibility and performance that other mappings offer. Some standard activation functions are given by

$$f(x) = x$$
 — linear (6.11a)

$$f(x) = \begin{cases} 0 & x \leq 0 \\ 1 & x > 0 \end{cases} - \text{binary step} \qquad (6.11b)$$

$$f(x) = \frac{1}{1 + \exp(-x)} - \text{logistic (soft step)} \quad (6.11c)$$

$$f(x) = tanh(x)$$
 – TanH (6.11d)

$$f(x) = \begin{cases} 0 & x \leq 0 \\ x & x > 0 \end{cases} - \text{rectified linear unit (ReLU).} \quad (6.11e)$$

There are other possibilities, but these are perhaps the most commonly considered in practice and they will serve for our purposes. Importantly, the chosen function *f*(*x*) will be difformitated in order to be used in gradient descent algorithms for optimization. Each of the functions above is either differentiable or piecewise differentiable. Perhaps the most commonly used activation function is currently the ReLU, which we denote *f*(*x*) = mRLU(*x*).

With a nonlinear activation function [11], or if there are more than one layer, here instand linear optimization nortices used as the process-inverse and LASCO anno hongers be used. Although this may not seem immediately significant, ceef that we are optimizing optimizations. These moderate is small problems: can be comparisonally expensive to solve optimizations. These moderate is small problems: can be comparisonally expensive to solve the structure optimization of the structure optimization of the solvent of the structure optimization of the structure optimization of the solvent included with the seemi network intension calls in MATLAB. As these methods are origin with examples, both the source of the solvent optimization of the calcurate.

MATLAE's neural network toolbox, much like TensorFow in python, has a vide range of features which makes it exceptionally powerful and convenients for building NNs. In the following code, we will train a NN to classify between dogs and cata as in the previous example. However, in this case, we allow the single layer to have a nonlinear transfer function that maps the input to the output layer. The output layer for this example will be modified to the following.

$$\mathbf{y} = \begin{bmatrix} 1 \\ 0 \end{bmatrix} = \{\text{dog}\} \text{ and } \mathbf{y} = \begin{bmatrix} 0 \\ 1 \end{bmatrix} = \{\text{cat}\}.$$
 (6.12)

Half of the data is extracted for training, while the other half is used for testing the results. The following code builds a network using the **train** command to classify between our images.

Code 6.2 Neural network with nonlinear transfer functions

In the code above, the patternnet command builds a classification network with two outputs (6.12). It also optimizes with the option trainage which is a *scaled conjugate* guidiont backpropagation. The **nt layers** also allows us to specify the transfer function, in this case hyperbolic tangent functions (6.110). The **view(net)** command produces a diagnostic tod shown in Fig. 6.5 that summarizes the optimization and NN.

The results of the classification for a cross-validated training set as well as a withhold set as shown in Fig. 66. Specifically, the desired outputs are given by the vectors (6.12). For both the training and withhold sets, the two components of the vectors are above for the 80 training timages (80 cuts and 40 dogs) and the 80 withhold images (40 cuts and 40 dogs). The training set process a prefect disatility within the set of the tangent trained runcinco (6.11.6). On the withhold data, it incorrectly identifies 6 of 40 dogs and cuts, yielding an accurst of 96. 3% on set data.

The diagonic tool down in Fig. 6.5 allows access to a number of frames critical for evaluating 108 N/16 Q/1 as a summary of the performance scheded by the NN training values of the tool training values of the tool valuation and the tool exist of the the backgroup anise of the tool of the tool of the tool values of the tool values of the tool values of the tool values of the tool values of the tool values of the tool values of the tool of tool of the tool of

There are two other features easily available with the NN diagnostic tool of Fig. 6.5. Fig. 6.8 shows an error histogram associated with the trained network. As with Fig. 6.7, the

Neural Network		
Hidden		Output 2
Algorithms		
Data Division: Random (di Training: Scaled Conju Performance: Cross-Entro Calculations: MEX	viderand) Igate Gradient (trainscg) py (crossentropy)	
Progress		
Epoch: 0	27 iterations	1000
Time:	0:00:01	
Performance: 0.592	5.60e-07	0.00
Gradient: 0.624	0.358-07	1.000-06
Tandation checks. 0		0
Plots		
Performance	(plotperform)	
Training State	(plottrainstate)	
Error Histogram	(ploterrhist)	
Confusion	(plotconfusion)	
Receiver Operating Chara	(plotroc)	
Plot Interval: Suprepay	1 epochs	
✔ Minimum gradient rea	ched.	
	Stop Training	Cancel

Figure 6.5 MATLAB neural network visualization tool. The number of iterations along with the performance can all be accessed from the interactive graphical tool. The performance, error histogram and confusion buttors produce Figure 6.7-6.9 respectively.

data is divided into training, validation, and test sets. This provides an overall assessment of the classification quality that can be achieved by the NN training algorithm. Another view of the performance can be seen in the confusion matrices for the training, validation,



Figure 6.6 Comparison of the output vectors $y = [y_1 \ y_2]^T$ which are ideally (6.12) for the dogs and cats considered here. The NN training stage produces a cross-validated classifier that achieves 100% accuracy in classifying the training data (top two panels for 40 dogs and 40 cats). When applied to a withheld set. S⁴S accuracy is achieved (bottom two marks for 40 doss and 40 cats).

and test data. This is shown in Fig. 6.9. Overall, between Figs. 6.7 to 6.9, high-quality diagnostic tools are available to evaluate how well the NN is able to achieve its classification task. The nerformance limits are easily seen in these fluences.

6.3 The Backpropagation Algorithm

As was shown for the NNs of the last two sections, training data is required to determine the weights of the network. Specifically, the network weights are determined to as to best classify day create cat images. In the 1-layer network, this was done using both least-square regression and 1-2-350. This shows that an is core, an optimization routine and objective regression and 1-2-350. This shows that an is core, an optimization routine and objective a measure of the mixels satisfied images. The optimization, however, can be modified by monstain a constraint, such as the ft- negative interaction is LASSO.

In practice, the objective function chosen for optimization is not the true objective function desired, but rather a proxy for it. Proxies are chosen largely due to the ability to differentiate the objective function in a computationally tractable manner. There are also many different objective functions for different tasks. Instead, one often considers a suitably



Figure 37: Summary of training of the NN over a number of epochs. The NN architecture intermedically separate the data into training, vulndarion and lead sets. The training continues (with a maximum of 1000 epochs) until the validation error curve hirs a minimum. The training them stops and the trained algorithm is then used on the tot set to evaluate performance. The NN trained here has only a limited armount of data (40 dops and 40 cash), than limiting the performance basis fogure is accessed with the performance basis on the NN interactive tool of Fig. 6.6.

chosen loss function so as to approximate the true objective. Ultimately, computational tractability is critical for training NNs.

The backgroupadom algorithm thatkgroup caption the compositional name of NNs in order to frame an outgrainstance problem for determining the weights of the network. Specifically, it produces a formation amenable so standard gradient descent optimization for differentiation. Moreover, it can be proven that the comparison function of the second for differentiation. Moreover, it can be proven that the comparison function of the second standards the gradient is thus. In factor of the second standard comparison during the second standard standard standards and the second standard standards and the second standards of the single standard standard standards and the simple to compare factoring the standard standard standards and standards and standards and how the gradient descent is to be performed. The second standards of the single standard standard standards and standards and

$$y = g(z, b) = g(f(x, a), b).$$
 (6.13)

Thus given a function $f(\cdot)$ and $g(\cdot)$ with weighting constants a and b, the output error produce by the network can be computed against the ground truth as

$$E = \frac{1}{2}(y_0 - y)^2$$
(6.14)

Error Hatogram with 20 Bins

Figure 6.8 Summary of the error performance of the NN architecture for training, validation and test sets. This figure is accessed with the errorhistogram button on the NN interactive tool of Fig. 6.6.

where y_0 is the correct output and y is the NN approximation to the output. The goal is to find a and b to minimize the error. The minimization requires

$$\frac{\partial E}{\partial a} = -(y_0 - y) \frac{dy}{dz} \frac{dz}{dy} = 0. \quad (6.15)$$

A critical observation is that the compositional nature of the network along with the chain rule forces the optimization to backpropagate error through the network. In particular, the terms dy/dz, dz/da show how this backprop occurs. Given functions $f(\cdot)$ and $g(\cdot)$, the chain rule can be explicitly computed.

Backprop results in an iterative, gradient descent update rule

$$a_{k+1} = a_k + \delta \frac{\partial E}{\partial a_k}$$

(6.16a)

$$b_{k+1} = b_k + \delta \frac{\partial E}{\partial b_0}$$

(6.16b)

where δ is the so-called learning rate and $\partial E/\partial a$ along with $\partial E/\partial b$ can be explicitly computed using (6.15). The iteration algorithm is executed to convergence. As with all iterative optimization, a good initial guess is critical to achieve a good solution in a reasonable amount of commutational time.

Backprop proceeds as follows: (i) A NN is specified along with a labeled training set. (ii) The initial weights of the network are set to random values. Importantly, one must not initialize the weights to zero, similar to what may be done in other machine learning



Figure 6.3 Summary of the error performance through confusion matrices of the NN architecture for training, validation and test sets. This figure is accessed with the confusion button on the NN interactive tool of Fig. 6.6.

algorithms. The registra are initialized to zero, there can bupdets, the oraping weights of each merow will be identical. Notes one thergo attent with the identical. Moreover, NNs other get study at location of prima where the gradeent is zero but that are not global misinus, so randou wideptimistizzational shows one hose a chance of currentwing this hypothese strangest and different modes values. (iii) The training data is run mongh, the services to produce are noder s, whose data globand and on ongot a gives fragmentary that the product and meteorist weight is then compared using backpup formulas (15.1); (ii)) For a given fragment detection and an anomaly mode of current and the strategies of the strategies of the meteorist weight is then compared using backpup formulas (15.1); (iii)) For a given fragment detection and an anomaly mode (5) controls or resolved or conversioned is tableted.

As a simple example, consider the linear activation function

$$f(\xi, \alpha) = g(\xi, \alpha) = \alpha\xi.$$
 (6.17)

In this case we have in Fig. 6.10

$$z = ax$$
 (6.18a)

$$y = b_{c}$$
. (6.18b)



Figure 1.9 Instruction of the backgrouppation lapitition on a non-node, one hidden layer network. The compositional nature of the network gives the imple output relationships y = g(t, o, a), by its minimizing the error between the output y and is desired output y, the composition along with the chain in produces an explicit number 1.0.3 for myraking the values of the weights. None that the chain rule backgrouppath the error all the way function that the output of the dimensional transformation of the structure of the structure of the structure output of the values of the dimensional transformation of the structure output of the structure o

We can now explicitly compute the gradients such as (6.15). This gives

$$\frac{\partial E}{\partial a} = -(y_0 - y) \frac{dy}{dz} \frac{dz}{da} = -(y_0 - y) \cdot b \cdot x$$
 (6.19a)

$$\frac{\partial E}{\partial b} = -(y_0 - y)\frac{dy}{db} = -(y_0 - y)z = -(y_0 - y) \cdot a \cdot x.$$
 (6.19b)

Thus with the current values of a and b, along with the input-output pair x and y and target truth y_0 , each derivative can be evaluated. This provides the required information to perform the update (6.16).

The backprop for a deeper net follows in a similar fashion. Consider a network with Mhidden layers labeled z_1 to z_m with the first connection weight a between x and z_1 . The generalization of Fig. 6.10 and (61.51) is given by

$$\frac{\partial E}{\partial a} = -(y_0 - y) \frac{dy}{dz_m} \frac{dz_m}{dz_{m-1}} \cdots \frac{dz_2}{dz_1} \frac{dz_1}{dz_1}.$$
(6.20)

The cascade of derivates induced by the composition and chain rule highlights the backpropagation of errors that occurs when minimizing the classification error.

A full generalization of backprops involves multiple types as well multiple nodes per layer. The general histonics in illustrated in [6, 6.1. The objective is to determine the mutrix chements of each matrix A.7. Thus a significant number of network parameters need to be splated in grander descern. Indeed, tuning a network can other be comparisodarily infrashibe even through the update rules for individual weights is not afflicit. Nice and the supfit from the curve of disminstrating to a schematic at conduct the comparisodarily updating an 'coefficients for an a-dimensional input, assuming the two connected layers are both a-dimensional.

Denoting all the weights to be updated by the vector w, where w contains all the elements of the matrices A₁ illustrated in Fig. 6.1, then

$$\mathbf{w}_{k+1} = \mathbf{w}_k + \delta \nabla E \qquad (6.21)$$

where the gradient of the error VE, through the composition and chain rule, produces the backpropagation algorithm for updating the weights and reducing the error. Expressed in a component-by-component way

$$w_{k+1}^j = w_k^j + \delta \frac{\partial E}{\partial w_i^j}$$

(6.22)

where this equation bolds for the jth component of the vector \mathbf{w} . The term $\partial F_i \partial m_i$ produces the backgrouppation through the chain nucle, i.e. it produces the suggestimate set of functions to evaluate as in (2.0). Methods for obving this optimization more quickly, or even simply evaluating the comparison to be tractable, remain of active research interest. Prohass the most important method is stochastic gradient descent which is considered in the next section.

6.4 The Stochastic Gradient Descent Algorithm

Training neural networks is comparationally expensive due to the size of the NNb biogutandi. Fore NNb of moders is care honcome probibitively expensive it the optimization rotaties used for training are as twell informed. Two algorithms have been especially final days and the straining a NNb enshaped and accord to SDM and hadropson Rackprop abouts for an efficient comparation of the objective function's gradest while of Diprobatis as more organization of the objective function's gradest while hadropson and straining and the straining and the straining alternative hadropson and SGCD are both considered have in detail to as to give the tradest an idea of the constraints of tradition, NNb.

Gradient descent was considered in Section 4.2. Recall that this algorithm was developed for nonlinear regression where the data fit takes the general form

$$f(x) = f(x, \beta)$$
 (6.23)

where β are fitting coefficients used to minimize the error. In NNs, the parameters β are the network weights, thus we can rewrite this in the form

$$f(\mathbf{x}) = f(\mathbf{x}, \mathbf{A}_1, \mathbf{A}_2, \dots, \mathbf{A}_M)$$
 (6.24)

where the A_j are the connectivity matrices from one layer to the next in the NN. Thus A₁ connects the first and second layers, and there are M hidden layers.

The goal of training the NN is to minimize the error between the network and the data. The standard root-mean square error for this case is defined as

$$\underset{\mathbf{A}_{j}}{\operatorname{argmin}} E(\mathbf{A}_{1}, \mathbf{A}_{2}, \cdots, \mathbf{A}_{M}) = \underset{\mathbf{A}_{j}}{\operatorname{argmin}} \sum_{k=1}^{n} (f(\mathbf{x}_{k}, \mathbf{A}_{1}, \mathbf{A}_{2}, \cdots, \mathbf{A}_{M}) - \mathbf{y}_{k})^{2} \quad (6.25)$$

which can be minimized by setting the partial derivative with respect to each matrix component to zero, i.e. we require $\delta E/\delta(a_0)_{\rm H} = 0$ where $(a_0)_{\rm R}$ is the *i*th row and *j*th column of the *k*th matrix $(k = 1, 2, \cdots, M)$. Recall that the zero derivate is a minimum since there is no maximum error. This gives the gradient $V_j(x)$ of the function with respect to the NN parameters. Note further that $f_j(x)$ is the function evaluated at each of the *n* data points.

As was shown in Section 4.2, this leads to a Newton-Raphson iteration scheme for finding the minima

$$\mathbf{x}_{i+1}(\delta) = \mathbf{x}_i - \delta \nabla f(\mathbf{x}_i) \qquad (6.26)$$

where 6 is a parameter determining how far a step should be taken along the gradient direction. In NNs, this parameter is called the *learning rate*. Unlike standard gradient descent, it can be computationally prohibitive to compute an optimal learning rate.

Although the optimization formulation is easily constructed, evaluating (6.25) is often computationally intractable for NNs. This due to two reasons: (i) the number of matrix weighting parameters for each A_j is quite large, and (ii) the number of data points *n* is generally also large.

To render the computation (6.25) potentially tractable, SGD does not estimate the gradient in (6.26) using all n data points. Rather, a single, randomly chosen data point, or a subset for batch gradient descent, is used to approximate the gradient at each step of the iteration. In this case, we can reformulate the least-square fitting of (6.25) so that

$$E(\mathbf{A}_{1}, \mathbf{A}_{2}, \dots, \mathbf{A}_{M}) = \sum_{k=1}^{n} E_{k}(\mathbf{A}_{1}, \mathbf{A}_{2}, \dots, \mathbf{A}_{M})$$
 (6.27)

and

 $E_k(\mathbf{A}_1, \mathbf{A}_2, \dots, \mathbf{A}_M) = (f_k(\mathbf{x}_k, \mathbf{A}_1, \mathbf{A}_2, \dots, \mathbf{A}_M) - \mathbf{y}_k)^2$ (6.28)

where $f_E(\cdot)$ is now the fitting function for each data point, and the entries of the matrices A_i are determined from the optimization process.

The gradient descent iteration algorithm (6.26) is now updated as follows

$$\mathbf{w}_{i+1}(\delta) = \mathbf{w}_i - \delta \nabla f_k(\mathbf{w}_i) \qquad (6.29)$$

where w_j is the vector of all the network weights from M_j ($j = 1, 2, ..., M_j$) at the j/h iteration, and the gradient is computed using only the fluid tag point and $f_j(c)$. Thus instead of computing the gradient with all a points, only a single data point is molecular with and used. At the next intention, another mandow yelected point is used to compete the gradient and update the ostitation. The algorithm may require multiple passes through all the data to converse, but each stop is non easy to valuate versus the requires computation of the Jacobian which is required for the gradient. If instead of a single point, a subset of points is used, then we here following their anglerism.

$$\mathbf{w}_{i+1}(\delta) = \mathbf{w}_i - \delta \nabla f_K(\mathbf{w}_i)$$
 (6.30)

where $K \in [k_1, k_2, \dots, k_p]$ denotes the p randomly selected data points k_j used to approximate the stratient.

The following code is a modification of the code shown in Section 4.2 for gradient descent. The modification here involves taking a significant subsampling of the data to approximate the gradient. Specifically, a batch gradient descent is illustrated with a fitted learning rate of $\delta = 2$. Ten points are used to approximate the gradient of the function at each step.

Code 6.3 Stochastic gradient descent algorithm.

```
lsoC.i; ss-6thif; ys-6thif; sslength(x);
[X,Y]smeshgrid(x,y); clear x, clear y
Full.5-1.6*exp(-0.05*(3*(3*3).^2*(Y-3).^2));
Full.5-1.6*exp(-0.1*(3*(X-3).^2*(Y-3).^2)));
[dTx,dTy]geradiat([v,h,h);
```

Fig. 6.11 shows the convergence of SGD for three initial conditions. As with gradient descent, the algorithm can get stuck in local minima. However, the SGD now approximates the gradient with only 100 points instead of the full 10⁴ points, thus allowing for a computation which is three orders of magnitude smaller. Importantly, the SGD is a scalable algorithm, allowing for significant computational sarvings even as the durat grows to be high-



Figure 413 Suchastic gradient descent applied to the function featured in Fig. 43(b). The convergence can be compared to a full gradient descent algorithm adown in Fig. 45. Each step of the stochastic (batch) gradient descent solection to dones in the fig. 45. Each gradient, instead of the UF adap rosis of the Adar. These initial conditions are shown in type, $y_0 = (4, 6)$, (0, -5), (-5, 1). The first of these (red) circles/gradient and local the random figure (4, 6), (0, -5), (-5, 1). The first of these (red) circles/gradient and local minima. The respection of the random figure (5, -5) are solely a nonline of the respection of the random figure (5, -5), (-5, 1).

dimensional. For this reason, SGD has become a critically enabling part of NN training. Note that the learning rate, batch size, and data sampling play an important role in the convergence of the method.

6.5 Deep Convolutional Neural Networks

With the basis of the NN architecture in hand, along with an antechnolog of the northermalies an optimization innewerk the helpop and neuklos compare the galared sector efficiently (SOCD) see at analy to construct *large consultant* marked and (DCNN) which are the minimizent balared back of *a day* names which, halsed, a kady, when partitions are presented as the sector of the Han as much as we would like to here participled genoma by halding DCNN, are estimated as the sector of the sector of the sector of the sector of the sector conservalue the results. The neuter theories do not be participle of controllow risk of the sector of

Like SVM and random forest adjorithms, the MATLAB package for building NNs has a treencodoss multire of features and tuning curranters. This fielding is both advantagoost and overstellming at the same time. As was pointed out at the beginning of this chapter, it is immediately orient that there are a granumeter of design position regarding NNs. How many layers should be used? What should be the dimension of the layers? How should the output put per designed? Should one use all-solar in possition? contentions between layers? How should he mapping between layers be performed: a *linear mapping* or a *nullear ampung*?

The protopolar downtow of ACONN histometed lor \mathbb{F}_{0} 6.1: Included line is unified of contrast number of controls and the product layers. Also informed in the transmitter of controls and the product layers of the product layers of the layers of the product layers of the layers of the layers of the outer layers of the layers between x 10 layers. More recent clinks have considered the absorber are still into the layers of the layers of the layers of the layers of the layer product layers layers with approximately layers, but here the layers with DCON symplem layers layers are strength and the layers of the layer product layers are still and lay layers. The laborating paragraph highlight some of the anei product layers are layers and dong dongs.

Convolutional Lavers

Convolutional layers are similar to windowed (Gabor) Fourier transforms or wavelets from Chapter 2, in that a small selection of the full high-dimensional input space is extracted and used for feature engineering. Fig. 6.12 shows the convolutional windows (dark gray boxes) that are sidi across the entire layer (light gray boxes). Each convolution window transforms



Figure 6.12 Prototypical DCNN architecture which includes commonly used convolutional and pooling layers. The dark gray boxes show the convolutional sampling from layer to layer. Note that for each layer, many functional transformations can be used to produce a variety of feature spaces. The network dimensional wave all this informations in oth to output layer.

the data into a new node through a given activation function, as shown in Fig. (5.12a). The frame spaces can bus hist from the smaller pathete of the data. Convolutional layers are expectally useful for images as they can extract important features such as adopts. Wavefst are as also known to efficiently vectors useft features and there are depenmathematical connections between wavefest and DCNNs as shown by Malitra and coworkers [358, 12]. Note that in Fig. 6.12, is a lengt layer can be used to construct many layers by simply annipolating the activation function $f(\cdot)$ to the next layer as well the size of the convolutional values.

Pooling Layers

It is common the predictably intert a Pooling tays between successive consolutional graves in a DON with these track is to proposely obscie the applicable and it is of the trapes is and interesting the transfer of the proposely obscie the applicable and the proposely obscie the pro

Fully-Connected Layers

Occasionally, fully-connected layers are inserted into the DCNN so that different regions can be connected. The pooling and convolutional layers are local connections only, while the fully-connected layer restores global connectivity. This is another commonly used layer in the DCNN architecture, providing a potentially important feature space to improve performance.

Dropout

Ordinitian ja asteoloon piloken in ECNNs, backed overfitting in at the ore of why ECNNs the fit is to humanize and generalizability promote See Chapter 4 are approached by the second second second second second second second second second descriptions of many different targe second second second second in the network (along with these constants) in the DCN hadron promotion. Dropped in the discond second second second second second second second discond second second

There are many other techniques that have been devised for training DCNNs, but the above methods highly some of the most commonly used. The most successful applications of these techniques tend to be in computer vision tasks where DCNNs offer unparalledel performance in comparison to other machine learning methods. Importunity, the ImageNET data set is what allowed these DCNN layers to be maximally leveraged for human level recognition performance.

To illustrate how to train and execute a DCNN, we use data from MATLAB. Specifically, we use a data set that has a training and test set with the alphabet characters A, B, and C. The following code loads the data set and plots a representative sample of the characters in Fig. 6.13.

```
Code 6.4 Loading alphabet images
```

```
load letter#TrainSet
perm = randperm(1500,20);
for j = 1:20
    mubplot(4,5,j);
    imshow(XTrain(:,:,:,perm(j)));
end
```

This code loads the training data, XTrain, that contains 1500 28×28 grayscale images of the letters A, B, and C in a 4-D array. There are equal numbers of each letter in the data set. The variable TTrain contains the categorical array of the letter labels, i.e. the truth labels. The following code constructs and trains a DCNN.

Code 6.5 Train a DCNN.

```
layers = [imageInputLayer([28 28 1]);
convolution2dLayer(5.16);
```



Figure 6.13 Representative images of the alphabet characters A, B, and C. There are a total of 1500 28×28 prayscale images (XTrain) of the letters that are labeled (TTrain).



Note the simplicity in how diverse network layers are easily put together. In addition, a ReLu activation layer is specified along with the training method of stochastic gradient descent (sgdm). The trainNetwork command integrates the options and layer specifications to build the best classifier possible. The resulting trained network can now be used on a test data set.

Code 6.6 Test the DCNN performance

```
load lettersTestSet;
YTest = classify(net,XTest);
accuracy = sum(YTest == TTest)/numel(TTest)
```

The resulting classification performance is approximately 93%. One can see by this code structure that modifying the network architecture and specifications is trivial. Indeed, one can probably easily engineer a network to outperform the illustrated DCNN. As already mentioned, artistry and expert intuition are critical for producing the highest performing networks.

6.6 Neural Networks for Dynamical Systems

Neural networks differ an amazingly flexible architecture for performing a diorese set of mineturatical tasks, or terms to S. Multita. Supervised learning is a high-dimensional interpolation problem [153]. Thus if sufficiently rick data can be acquired, NNs offer the ability is interrogate the data for a variety of tasks centered on classification and prediction. To this point, the task demonstrated have primarily been accument, with computer vision. However, NNs can also be used for future state predictions of dynamical systems (See Chapter T).

To demonstrate the usefulness of NNs for applications in dynamical systems, we will consider the Lorenz system of differential equations [345]

$$\dot{x} = \sigma(y - x)$$
 (6.31a)

$$\dot{v} = x(\rho - z) - v$$
 (6.31b)

$$t = xy - \beta z$$
, (6.31c)

where the state of the system is given by $x = [1 y \ 2]^2$ with the parameters $\sigma = 10$, $\rho = 20$, $\lambda = 30$, $\lambda = 10$, $\lambda = 30$. This system will be considered in further details in the next charges. For the present, we will simulate this nonlinear system and use it as a demonstration of how NN can be trained to characterize dynamical systems. Specifically, the pad of about sets a state of the system and the state of the system is a state of the system and the state of the system and the state of the system at many sets and the state of the system at many set and the constraints of the solution in time requires a nonlinear transfer function since Lorenz inelf is nonlinear.

The training data required for the NN is constructed from high-accuracy simulations of the Lorenz system. The following code generates a diverse set of initial conditions. One hundred initial conditions are considered in order to generate one hundred majectories. The sampling time is fixed at $\Delta \tau = 0.01$. Note that the sampling time is not the same as the time-steps taken by the 486-order Range-Katta method [316]. The time-steps are adaptively chosen to meet the samplent lower of accuracy chosens for this example.

Code 6.7 Create training data of Lorenz trajectories

```
[ 1 min.2 = Leres gets

Min.2 = Leres gets

Min.2 = Leres (1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1, 1 = 1
```



Figure 8.14 Evolution of the Lorenz dynamical equations for one handred randomly chosen initial conditions (red circles). For the parameters $\sigma = 10$, $\rho = 28$, and $\beta = 8/3$, all trajectories collapse to an attractor. These trajectories, generated from a diverse set of initial data, are used to train a neural network to learn the mathema mapping from x_1 to x_{k+1} .

The simulation of the Lorenz system produces to key matrices: input and output. The former is a matrix of the system at x_k , while the latter is the corresponding state of the system x_{k-1} advanced $\Delta t = 0.01$.

The NN must learn the nonlinear mapping from x_k to x_{k+1} . Fig. 6.14 shows the various trajectories used to train the NN. Note the diversity of initial conditions and the underlying attractor of the Lorenz system.

We now build a NN trained on trajectories of Fig. 1.4 to advance the solution $\Delta t = 0.01$ into the future for an advarsary initial condition. Here, a three syster network is constructed with an nodes in each alayer and a different activation unit for each layer. The choice of advancion types, nodes in the layer and number of layers are advarsary. It is trivial to make the NN for the advance of the system and the system of the system of the the NN for the advance of the system and the system of the system advance many. The NN is built with the following for time of occles.

Code 6.8 Build a neural network for Lorenz system

```
net = feedforwardnet([10 10 10]);
net.layers[1].transferFon = 'logsig';
net.layers[2].transferFon = 'radbas';
net = layers[3].transferFon = 'purelin'
net = train(net.input.'cutput.');
```



Figure 815 (a) Network architecture used to train the NN on the trajectory data of Fig. 6.14. A three-layer network is constructed with ten nodes in each layer and a different activation unit for a call layer. (b) Performance summary of the NN optimization algorithm. Over 1000 speechs of training, accuracies on the order of 10^{-5} are produced. The NN is also cross-validated in the process.

The code produces a function net which can be used with a new set of data to produce predictions of the funct. Specifically, the function net gives the nonlinear mapping from x_k to x_{k-1} . Fig. 6.15 shows the structure of the network along with the performance of the training over 1000 epochs of training. The results of the cross-valued data are also demonstrated. The NN converges steadily to a network that produces accuracies on the order of 10⁻⁵.

Done the N1 is mained on the trajectory data, the conditioner model mapping us, to us, it on the sured to predict the future start of the system from an initial condition, and a brinner in the start of the start of maximum start is used in taken an initial condition and a brinner and the start of the start 2.2 and into the future. The instruction start gales can produce a prediction for the future start as far into the future an desired. In what follows, the mapping is used to predict the start of the start of the start of the start of the start start of the start of the start condition start of the dwamins.



Figure 6.16 Comparison of the time evolution of the Lorenz system (solid line) with the NN prediction (dotted line) for two randomly chosen initial conditions (red dots). The NN prediction stays close to the dynamical trajectory of the Lorenz model. A more detailed comparison is given in Fig. 6.1.7.

Code 6.9 Neural network for prediction.

```
Imn(1,1)se0;
for jjs2laegth(t)
y0ms(te0);
ynn(jj,1)sy0.'; x0sy0;
end
plot1(ynn(:,1),ynn(1,2),ynn(1,3),':','Linewidth',[2])
```

Fig. 6.16 shows the evolution of two randomly drawn trajectories (solid lines) compared against the NP prediction of its principotics (stoted lines). The NN prediction is remarkably accurate in producing an approximation to the high-accuracy simulations. This was that the data used for training its capable for producing a high-paulity nonlinear model mapping us to \mathbf{x}_{+1} . The quality of the approximation is more classifier and the second state the production of the second state of the second sta

In conclusion, the NN can be trained to learn dynamics. More precisely, the NN seems to kern an algorithm which is approximately equivalent to a 4th-order Range-Khat scheme for advancing the solution a time-step A.r. Indeed, NNh have been used to model dynamical systems [213] and other physical processors [311] for decades. However, grant strides have been made recently in using DNNs to learn Koopman embeddings, resulting in several secretion papers [553, 306, 831, 364, 412, 332]. For example, the VAMPmet architec-



Figure 8.17 Comparison of the time evolution of the Lorenz system for two randomly chosen initial conditions (Also shown in Fig. 6.16). The left column shows that the evolution of the Lorenz differential equations and the NN mapping gives identical results until $1 \approx 5.5$, at which point they diverge. In contrast, the NIP prediction stays on the trajectory of the second initial condition for the entire time window.

ture 1500, 3601 uses a time-larged ano-encoder and a contom vurificand score to identify Koopman coordinates on an impressive procession folding example. In an alternative formalation, vurification and a construction of the second score and the second score and the respectation of the Koopman operator for model. 1304 By Governice, the resulting network is not parsimonious and interpretable, retaining the fexibility of neural networks and the physical interpretation of Koopman heavy. In all of these recent studes, DNN representations have been shown to be more flexible and exhibit higher accuracy than other leading methods on challenging problems.

6.7 The Diversity of Neural Networks

There are a wide variety of NN architecures, with only a few of the most dominant architectures considered thus far. This chapter and book does not attempt to give a comprehensive assessment of the state-of-the-art in neural networks. Ruther, our focus is on liturating some of the key concepts and enabling mathematical architectures that have ked NNs to a dominant position in modern data science. For a more in-depth review, plotsas es [216]. However, to conclude this chapter, we would like to hajdhjuls some of the NN architectures that are used in practice for various data science tasks. This overview is inspired by the *neural network* 200 as highlighted by Fjodor Van Veen of the Asimov Institute (http://www.saimovinstitute.og).

The neural network zoo highlight control of different nethientual structures around No. Some of the networks highlighted are commonly used aroun isolarys, while other serve induces the other structures are structured as a control as interesting and training and training the other structures are structured as a contrast structure and high the processing and training and the structure and high the processing and the structure and the structure and and the structure and the structure and the structure and and the structure and the structure and the structure and and the structure and the structure and the structure and structures and the constructure and the structure and the structure structures and the constructure and the structure structures and the constructure and the structure and the structure structures and the constructure and the structure and the structure and structures and the constructure and the structure and the structure and structures and the constructure and the structure and the structure and structures and the constructure and the structure and the structure and structures and the structure and the structure and the structure and structures and the structure and the structure and the structure and structures and the structure and the

Perceptron

The first mathematical model of NNs by Fukushima was termed the Neccognitron in 1980 [193]. His model had a single layer with a single output cell called the perceptron, which made a categorial decision based on the sign of the output. Fig. 6.2 shows this architecture to classify between dogs and cats. The perceptron is an algorithm for supervised learning of binary classifiers.

Feed Forward (FF)

Feed forward networks concert the input layer to optique layer by forming concercitons there the networks of the phot of them is excised for the straight and a servision of this architecture where the information simply propagates. Them the hist right in the straight excised of the straight excised on the straight excised of the straight excised of the straight excised on the straight excised of the straight excised of the straight excised on the straight excised of the straight excised of the straight excised on the straight excised of the straight excised of the straight excised on the straight e

Recurrent Neural Network (RNN)

Illustrated in Fig. 6. 18(4), RNNs are characterized by connections between units that from a detreed graph along a sequence. This along to its exhibit dynamic temporal behavior for a time sequence [172]. Utilia feedforward neural networks, RNNs can use their internal static immorphy (to press sequences of mission). The prototypical architecture in Fig. 6.18(4), shows that each cell feeds back on neelf. This self-interaction, which is not part of the FF architecture, alongs for a variety of manufactors. Specifically, in allows for the delay and/or fendback kops. Such controlled latters are referred to a gated state or gated memory, and are gated to the symmetries. Such arc for the more more, TGN-TMINENTONA [284] and and grapt of the two framediations. The more more of TGN-TMINENTONA [284] and the more more more more than the temporary of the more more of the temporary of temporary of the temporary of the temporary of the temporary of temporary of the temporary of temporary of temporary of temporary of the temporary of temporary



Figure 6.18 Neural network architectures commonly considered in the literature. The NNs are comprised of input nodes, output nodes, and hidden nodes. Additionally, the nodes can have memory, perform consolution and/or pooling, and perform a kernel transformation. Each network, and their acrossym is explained in the text.

gated recurrent units (GRU) [132]. LSTM is of particular importance as it revolutionized speech recognition, setting a variety of performance records and outperforming traditional models in a variety of speech applications. GRUs are a variation of LSTMs which have been demonstrated to exhibit better performance on smaller datasets.

Auto Encoder (AE)

The aim of an auto encoder, represented in Fig. 6.18(b), is to learn a representation (encoding) for a set of data, typically for the purpose of dimensionality reduction. For AEs, the input and output cells are matched so that the AE is essentially constructed to be a nonlinear transform into and out of a new representation, acting as an approximate identity map on the data. Thus AEs can be thought of as a generalization of linear dimensionality reduction techniques such as PCA. AEs can notentially nurduce nonlinear PCA representations of the data or nonlinear manifolds on which the data should be embedded [71]. Since most data lives in nonlinear subspaces. AFs are an important class of NN for data science, with many innovations and modifications. Three important modifications of the standard AE are commonly used. The variational auto encoder (VAE) [290] (shown in Fig. 6.18(c)) is a popular approach to unsupervised learning of complicated distributions. By making strong assumptions concerning the distribution of latent variables, it can be trained using standard gradient descent algorithms to provide a good assessments of data in an unsupervised fashion. The denoising auto encoder (DAE) [541] (shown in Fig. 6.18(c)) takes a partially corrupted input during training to recover the original undistorted input. Thus noise is intentionally added to the input in order to learn the nonlinear embedding. Finally, the sparse auto encoder (SAE) [432] (shown in Fig. 6.18(d)) imposes sparsity on the hidden units during training, while having a larger number of hidden units than inputs, so that an autoencoder can learn useful structures in the input data. Sparsity is typically imposed by thresholding all but the few strongest hidden unit activations.

Markov Chain (MC)

A Markov chain is a atochatic model describing a sequence of possible events in which her probability or cach event depend only on the state arianticel in the previous event. So although not formally a NN, it shares many common features with RNNs, Markov chains are standard even in undergraduate probability and statistics courses. Fig. 6, 18(1) shows the basic architecture where each cell is connected to the other cells by a probability model for a transition.

Hopfield Network (HN)

A Hopfetd network is a form of a RNN which was popularized by John Hopfetd in 1992 for understanding human memory [24], Fig. 6.18(2) also wise the basic architecture of an all-so-all connected network where each node can act as an input cell. The network networks at trainflash content-addresardle association memory system tilt biamy fluedoid minimum. Sometimes it converge to a fabe pattern, or memory twrong local minimum, rather than the system of pattern (expected basic minimum).

Boltzmann Machine (BM)

The Bokzmann machine, sometimes called a stochastic Hopfield network with hidden units, is a stochastic, generative counterpart of the Hopfield network. They were one of the first neural networks capable of beaming intermal representations, and are able to represent and (given sufficient time) solve difficult combinatoric problems [266]. Fig. 6.18(b) shows the structure of the BM. Note that unlike Markov chains (which have no input units) or Hopfield networks (where all cells are inputs), the BM is a hybrid which has a mixture of input cells and hidden units. Boltzmann machines are irrativityly appealing due to their resemblance to the dynamics of simple physical processes. They are named after the Boltznann distribution in statistical mechanics, which is used in their sampling function.

Restricted Boltzmann Machine (RBM)

Introduced match the name Homoscaniny by Parl Simolensky in 1997; HRM humes proposed for discontingent production, calculation, collaborative information, Reinner, Reinner Mark, Sanger Mark, Sanger Mark, Sanger Mark, Sanger Mark, Terrard, Tank, G. Hittens height bring given in promitence by developing in calculation from such that so as a discont of MM with seven controlsion are imposed on the NN with this costs in dr. NN must from a solution of MM with the solution of the solution of the solution of the solution of MM and the solution of the Haldow match regulation of the solution of the solution of the solution of the constraints with the solution of the solut

Deep Belief Network (DBN)

DBNs are a generative graphical model that are composed of multiple layers of latert hidden variables, with connections between the layers but on between units within case. Using 152, Fig. 6.1800, shows the architecture of the DBNs. The training of the DBNs can be done stack by stack from AE or RBM bingers. Thus each of the layers of ally has to learn to encode the previous network, which is effectively a greedy training algorithm for finding locally optimization introm. The DBNs can be viewed as a composition of simple, sumapervised networks such as RBMs and AEs where each sub-network's hidden layer serves as the visible layer for the next.

Deep Convolutional Neural Network (DCNN)

DCNNs are the workhorse of computer vision and have already been considered in this chapter. They are abstractly represented in Fig. 6.18(j), and in a more specific fashion in Fig. 6.12. Their impact and influence on computer vision cannot be overestimated. They were originally developed for document recognition [325].

Deconvolutional Network (DN)

Decomotional Networks, shown in Fig. 6.18(4), are essentially a revene of DCNNs [567]. The mathematical structure ODM pennith the suspervised construction of hierarchical image representations. These representations can be used for both low-level tasks such as denoising, as well as providing features for object recognition. Each level of the hierarchy groups information from the level beneath to form more complex features that exist over a larger scale in the image. As with DCNNs, it is well suited for computer vision tasks.

Deep Convolutional Inverse Graphics Network (DCIGN)

The DCIGN is a form of a VAE that uses DCNNs for the encoding and decoding [313]. As with the AE/VAE/SAE structures, the output layer shown in Fig. 6.18(1) is constrained to match the input layer. DCIGN combine the power of DCNNs with VAEs, which provides a formative mathematical architecture for computer visions and image processing.

Generative Adversarial Network (GAN)

In a inconverse modification of NNs, the CAN architecture of Pg. 4. Holy mains we determine a multianceal (F1). The attendo, show these are combination of DENNs to produce a structure of DENNs to produce the Section of DENNs to the DENns to the Section of DENNs to the DENNs to the Section of DENNs to the DE

Liquid State Machine (LSM)

The LSM shown in Fig. (18(6)) is a particular kind or spliting neural network [352], An LSM consists of a large collection of node, case of which recreases their waying larger from external acures: (the larger) as well as from other nodes. Nodes are randomly connected to each other. The resurrent nature of the concections turns the time waying larger line on spatio-lengoned pattern of activations in the network nodes. The spatio-lengoned patterns of actuations are read on by linear discriminant units. This architecture is noticeded by splitiking neurons in the brain, thus helping understand how information processing and discrimination mith three neurons.

Extreme Learning Machine (ELM)

With the same underlying architecture of an LSM shown in Fig. 6.18(a), the LEM is a F network for classification, repression, classering, sparse approximation, compression and feature learning with a single layer or multiple layers of hidden nodes, where the parameters of hidden nodes can be randomly assigned and never updated, or can be indirectif form their reactions without bridge langed. In motion case, the outgat weights of hidden nodes, are usually learned in a single step, which essentially amounts to learning a linear model [108].

Echo State Network (ESN)

ESNs are RNNs with a sparsely connected hidden layer (with typically 1% connectivity). The connectivity and weights of hidden neurons have memory and are fixed and randomly assigned (See Fig. 6.18(0)). Thus like LSMs and ELMs they are not fixed into a wellordered layered structure. The weights of output neurons can be learned so that the network can generate specific temporal patterns [263].

Deep Residual Network (DRN)

DBNs took the deep learning workly by storm when Mensoroh Research released Deep Readual Learning for hunge Recognition [27]. These networks do to logicar winning entries in all there main tracks of the ImageNet and COXO 2015 competitions, which covered image classification, object detection, and accumate segmentation. The robustness of ResNets has since been proven by various visual recognition tasks and by nonrisual institution of the start these starts in the start of the start of the start of the start of the start these starts in the start of the start

Kohonen Network (KN)

Kohonen networks are also kanow as self-organizing feature maps [298], KNs use competivice kenning to classify data without separitisalin, partin pissentela to he KNs usi in Fig. 6.18(a), after which the network assesses which of the neurons closely much that input. These self cognizining mays differ from deter NNs as low apply competitive learning a opposed to error-correction learning (such as backpropagation with gradient decent), and in the sense that they use a particular backpropagation with gradient decent, may an its ense that they use a particular backpropagation with singular properties of the input space. This makes KNs useful for low-dimensional visualization of high-dimensional data.

Neural Turing Machine (NTM)

An NTM implements a NN controller coupled to an external memory resource (See Fig. 6.18(r)), which it interacts with through metricoal mechanisms [219]. The memory interactions are differentiable end-to-end, making it possible to optimize them using gradient descent. An NTM with a LSTM controller can infer simple algorithms such as copying, sorting, and associative recall from input and output examples.

Suggested Reading

Texts

- (1) Deep learning, by I. Goodfellow, Y. Bengio and A. Courville, 2016 [216].
- (2) Neural networks for pattern recognition, by C. M. Bishop, 1995 [63].

Papers and Reviews

- (1) Deep learning, by Y. LeCun, Y. Bengio and G. Hinton, Nature, 2015 [324].
- (2) Understanding deep convolutional networks, by S. Mallat, Phil. Trans. R. Soc. A, 2016 [358].
- (3) Deep learning: mathematics and neuroscience, by T. Poggio, Views & Reviews, McGovern Center for Brains, Minds and Machines, 2016 [430].
- (4) Imagenet classification with deep convolutional neural, by A. Krizhevsky, I. Sutskever and G. Hinton, Advances in neural information processing systems, 2012 [310].

Dynamics and Control

Dynamical systems provide a mathematical framework to describe the world around us, modeling the rich interactions between quantities that o covides in this. Formally, dynamical cal systems concerns the analysis; prediction, and understanding of the behavior of systems of differential quantities or interarie mapping that discribe the oriention of the state of phenomena including theore described in the concernst orient of the state of phenomena including theore described in the concernst oriential covients, methoding theore, theorem of the state of the fluids, clinical systems, fluids, excluding as a staggering range of phenomena including theorem ends in the concernst phenomenon ends of the state of the

Modern optimized systems legan with the semial work of Patoacte on the chainsing motion of planes. This root is a location densities, and any be visued in the cohination of matcheol of yours of parotect mixed may hegeline to present the semitering of the system of the semiconductive systems and Leiblar. The event of the system of the system provides on the system and fields and chaining problems. Dynamical systems provides used the sum of appears in the "concerned leibla" of mathematics. Multiple spresses gives may be appeared and vectorison of the for mathematics. Multiple spresses spress matcheol has been over control in the modeling and analysis of systems in source years by did of the property gives and the systems.

Motion's primited systems in correctly undergoing arranisance, with analytical derivations after the projection of the streng systems, but is constructed of of spanning systems in science and engineering. Then are shouldness, which go system is the strength systems in science and engineering. Then are shouldness, which go system is the strength systems in science and engineering. Then are shouldness, which go system is the strength system is science and engineering. Then are shouldness, the strength systems is been approximately as the strength system is a strength system in the system is strength system is strength systems. The system is strength systems is strength systems is strength systems, the strength systems is strength systems with a system is strength system is strength systems in the system is strength system is strength systems, the strength systems is strength systems is strength systems in the strength system system is strength systems in the strength systems is strength and systems is strength systems in the strength systems is strength systems in the strength system system is strength systems in the strength systems is strength as a strength system system is strength systems in the strength systems is strength systems in the strength system system is strength systems in the strength systems is strength systems in the strength system system is strength systems in the strength system system system is strength systems in the strength system system is strength systems in the strength system system system is strength systems in the system system system system system systems in the system system system system system systems in the strength system system system system systems in the strength system system system system systems in the strength system system system system system systems in the strength system system system system system system systems in the strength system systems in the str

In addition, the classical geometric and statistical perspectives on dynamical systems are being complemented by a hind operato-showing repredicts, based on the evolution of measurements of the system. This so-called *Koopmon* operator theory is pixele to capitalize on the increasing availability of measurement data from complex systems. Moreover, Koopman theory provides a path to identify intrinsic coordinate systems to represent nonlinear dynamics in a littler framework. Obtaining litterar representations of
strongly nonlinear systems has the potential to revolutionize our ability to predict and control these systems.

This chapter presents a modern perspective on dynamical systems in the context of current goals and open challenges. Data driven dynamical systems is a raphile verolving field, and therefore, we focus on a mix of established and emerging methods that are driving current developments. In particular, we will focus on the key challenges of discovering dynamics from data and finding data-driven representations that make nonlinear systems anneable to linear analysis.

7.1 Overview, Motivations, and Challenges

Before summarizing recent developments in data-driven dynamical systems, it is important to first provide a mathematical introduction to the notation and summarize key motivations and oeen challenose in dynamical systems.

Dynamical Systems

Throughout this chapter, we will consider dynamical systems of the form:

$$\frac{d}{dt}\mathbf{x}(t) = \mathbf{f}(\mathbf{x}(t), t; \boldsymbol{\beta}). \quad (7.1)$$

where x is the state of the system and f is a vector field that possibly depends on the state x, time t, and a set of parameters β .

For example, consider the Lorenz equations [345]

$$\dot{x} = \sigma(y - x)$$
 (7.2a)

$$\dot{y} = x(\rho - z) - y$$
 (7.2b)

$$\dot{z} = xy - \beta z$$
, (7.2c)

with parameters $\sigma = 10$, $\rho = 28$, and $\beta = 8/3$. A trajectory of the Lorenz system is shown in Fig. 7.1. In this case, the state vector is $\mathbf{x} = \begin{bmatrix} x & y & z \end{bmatrix}^T$ and the parameter vector is $\boldsymbol{\beta} = \begin{bmatrix} \sigma & \rho & \beta \end{bmatrix}^T$.

The Lorenz system is among the simplest and most well-studied dynamical systems that exhibits chaos, which is characterized as a sensitive dependence on initial conditions. Two trajectories with nearby initial conditions will rapidly diverge in behavior, and after long times, only statistical statements can be made.

It is simple to simulate dynamical systems, such as the Lorenz system. First, the vector field f(x, r; 8) is defined in the function **lorenz**:

```
function dx = lorenx(t,x,Heta)
dx = [
Heta(1)*(x(2)-x(1));
x(1)*(Heta(2)-x(3))-x(2);
x(1)*(2)-Heta(3)*x(3);
];
```

Next, we define the system parameters **B**, initial condition x₀, and time span:



Figure 7.1 Chaotic trajectory of the Lorenz system from (7.2).

```
dt = 0.001;
tspansdtidt:50;
options = odeset('RelTol',ie-12,'AbsTol',ie-12,ones(1,3));
```

Finally, we simulate the equations with ode45, which implements a fourth-order Runge Kutta integration scheme with adaptive time step:

[t,x] sode45(@(t,x) lorenx(t,x,Beta),tspan,x0,options);
plot3(x(:,1),x(:,2),x(:,3));

We will often consider the simpler case of an autonomous system without time dependence or parameters:

$$\frac{d}{dt}\mathbf{x}(t) = \mathbf{f}(\mathbf{x}(t)). \quad (7.3)$$

In general, $\mathbf{x}(t) \in \mathbf{M}$ is an *n*-dimensional state that lives on a smooth manifold \mathbf{M} , and *t* is an element of the tangent bundle TM of \mathbf{M} so that $f(\mathbf{x}(t)) \in T_{W/R}$. However, we will typically consider the simpler case where \mathbf{x} is a vector, $\mathbf{M} = \mathbb{R}^n$, and *t* is a Lipschitz continuous function, guaranteeing existence and uniqueness of solutions to (7.3). For the more general formulation, see [1].

Discrete-Time Systems

We will also consider the discrete-time dynamical system

$$x_{k+1} = F(x_k).$$
 (7.4)

Also known as a map, the discrete-time dynamics are more general than the continuoustime formulation in (7.3), encompassing discontinuous and hybrid systems as well.

For example, consider the logistic map:

$$x_{k+1} = \beta x_k (1 - x_k).$$
 (7.5)



Figure 7.2 Attracting sets of the logistic map for varying parameter β .

As the parameter β is increased, the attracting set becomes increasingly complex, shown in Fig. 7.2. A series of period-doubling bifurcations occur until the attracting set becomes fractal.

Discrete-time dynamics may be induced from continuous-time dynamics, where x_k is obtained by sampling the trajectory in (7.3) discretely in time, so that $x_k = x(k \Delta t)$. The discrete-time propagator $\mathbf{F}_{\Delta t}$ is now parameterized by the time step Δt . For an arbitrary time t, the flow may \mathbf{F}_k is defined as

$$\mathbf{F}_{t}(\mathbf{x}(t_{0})) = \mathbf{x}(t_{0}) + \int_{t_{0}}^{t_{0}+\tau} \mathbf{f}(\mathbf{x}(\tau)) d\tau.$$
 (7.6)

The discrete-time perspective is often more natural when considering experimental data and dirital control.

Linear Dynamics and Spectral Decomposition

Whenever possible, it is desirable to work with linear dynamics of the form

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x}.$$
 (7.7)

Linear dynamical systems admit closed-form solutions, and there are a wealth of techniques for the analysis, prediction, numerical simulation, estimation, and control of such systems. The solution of (7.7) is siven by

$$\mathbf{x}(t_0 + t) = e^{At}\mathbf{x}(t_0).$$
 (7.8)

The dynamics are entirely characterized by the eigenvalues and eigenvectors of the matrix **A**, given by the spectral decomposition (eigen-decomposition) of **A**:

$$AT = TA$$
. (7.9)

When A has n distinct eigenvalues, then A is a diagonal matrix containing the eigenvalues λ_1 and T is a matrix whose columns are the linearly independent eigenvectors ξ_1 associated with eigenvalues λ_j . In this case, it is possible to write $A = TAT^{-1}$, and the solution in (7.8) becomes

$$\mathbf{x}(t_0 + t) = \mathbf{T}e^{At}\mathbf{T}^{-1}\mathbf{x}(t_0).$$
 (7.10)

More generally, in the case of repeated eigenvalues, the matrix A will consist of Jordan blocks [427]. See Section 8.2 for a detailed derivation of the above arguments for control systems. Note that the continuous-time system gives rise to a discrete-time dynamical system, with F_g given by the solution map exp(Δr) in (7.8). In this case, the discrete-time eienvalues are or given by ref.

The matrix T^{-1} defines a transformation, $z = T^{-1}x$, into intrinsic eigenvector coordinates, z, where the dynamics become decoupled:

$$\frac{d}{dt}\mathbf{z} = \mathbf{A}\mathbf{z}.$$
 (7.11)

In other words, each coordinate, z1, only depends on itself, with simple dynamics given by

$$\frac{d}{dt}z_j = \lambda_j z_j. \quad (7.12)$$

Thus, it is highly desirable to work with linear systems, since it is possible to easily transform the system into eigenvector coordinates where the dynamics become decoupled. No such closed-form solution or simple linear change of coordinates exist in general for nonlinear systems, motivatine many of the directions described in this chatter.

Goals and Challenges in Modern Dynamical Systems

As we generally use dynamical systems to model real-world phenomena, there are a number of high-priority goals associated with the analysis of dynamical systems:

- Future state prediction. In many cases, such as meteorology and climatology, we seek predictions of the future state of a system. Long-time predictions may still be challenging.
- Design and optimization. We may seek to tune the parameters of a system for improved performance or stability, for example through the placement of fins on a recket.
- Estimation and control. It is often possible to actively control a dynamical system through feedback, using measurements of the system to inform actuation to modify the behavior. In this case, it is often necessary to estimate the full state of the system from limited measurements.
- Interpretability and physical understanding. Perhaps a more fundamental goal of dynamical systems is to provide physical insight and interpretability into a system's behavior through analyzing trajectories and solutions to the governing equations of motion.

Real-world systems are generally nonlinear and exhibit multi-scale behavior in both space and time. It must also be assumed that there is uncertainty in the equations of motion, in the specification of parameters, and in the measurements of the system. Some systems are more sensitive to this uncertainty than others, and probabilistic approaches must be used. Increasingly, it is also the case that the basic equations of motion are not specified and they might be intractable to derive from first principles.

This chapter will cover recent data-driven techniques to identify and analyze dynamical systems. The majority of this chapter addresses two primary challenges of modern dynamical systems:

 Nonlinearity. Nonlinearity remains a primary challenge in analyzing and controlling dynamical systems, giving rise to complex global dynamics. We saw above that linear systems may be complexity characterized in terms of the spectral decomposition (i.e. eigenvalues and eigenvectors) of the matrix A. Euding to generate procedures: nonlinear systems, and developing this general framework is a mathematical grand challenge of the 21 sectors.

The leading properties on nonlinear dynamical systems considers the generative of subspaces of local linearizations around there is measures, and more greated heterocchica is and humocchica orbits connecting these structures, and more greated heterocchica systems, and its associations and the second systems of the systems and its associations in the largely methandin to their erical categories, such as the Harman Gorosan theorem, which excludes when there is it possible to apply the world of littles analysis techniques in a study is in often possible to apply the world of littles analysis techniques in a study in protochical system and its association in the study attemption of protochical systems and its association of the study attemption of the protochical systems and its association of the study attemption of the study of the study littless models, global analysis has remained largely in and counted any study.

Unknown dynamics, Perdapa un even more estrait duttinge artice from the lack favora governing equation for many under mysters of interast. Internatingly, researchers are tackling more complex and realities systems, such as are bond in morroriscience, epistemiologi, and coxing). In these fields, there is a basic lack of known physical laces that provide then phrasical phase that have been provided by the start of the phrase in the physical phase. The physical laces that provide the physical phase that the physical phase in the physical phase that the physical phase in the physical physical phase that the physical phase in the physical phase that the physical phase in the physical physi

Traditionally, physical systems were analyzed by making ideal approximations and then deriving simple differential equation models with Newton's second law. Dramatic simplifications could often be made by exploiting symmetrics and clever coordinate systems, an highlighted by the success of Lagrangian and Hamiltonian dynamics [2,309]. With increasingly complex systems, the paradigms is shifting from the classical approach to dut-driven methods to discover governing equations.

All models are approximations, and with increasing complexity, these approximations often become suspect. Telestrimiting what is the correct model is becoming more subjective, and there is a growing need for automated model discovery techniques that illuminate underlying physical mechanisms. There are also often latent variables that are relevant to the dynamics but may go unmeasured. Uncovering these hidden effects is a major challenge for data-driven methods. Identifying unknown dynamics from data and learning intrinsic coordinates that enable the linear representation of nonlinear systems are two of the most pressing goals of modern dynamical systems. Overcoming the challenges of unknown dynamics and nonlinearity has the promise of transforming our understanding of complex systems, with tremendous potential benefit to nearly all fields of science and engineering.

Throughout this chapter we will explore these issues in further detail and describe a number of the emerging techniques to address these challenges. In particular, there are two key approaches that are defining modern data-driven dynamical systems:

- Operator theoretic representations. To address the issue of nonlinearity, operatortheoretic approaches to dynamical asystems are becoming increasingly used. As we will abow, it is possible to represent nonlinear dynamical systems in terms of infinitedimensional but linear operators, such as the Koopman operator form Societion 7. A that advances measurement functions, and the Perron-Probenius operator functions probability densities and resembles through the dynamics.
- 2. Data-driven regression and machine learning. As data becomes increasingly abundant, and we continue to increasing the approximation of the externation of the e

It is important to note that many of the methods and perspectives described in this chapter are interrelated, and continuing to strengthen and uncover these relationships is the abaject of ongoing research. It is also worth mentioning that a third major challenge is the high-dimensionality associated with many modern dynamical systems, such as are found in population dynamics, basis imitations, and high-defaulty mensical divergentions of partial differential equations. High-dimensionality is addressed extensively in the subsequent chapters or neduced order models (RODh).

7.2 Dynamic Mode Decomposition (DMD)

Tynnine mode choromopolions was accelerately bished [174, 171] in the third dynamic submound in the second of the

Soon after the development of the original DMD algorithm [474, 472], Rowley, Mezic, and collaborators established an important connection between DMD and Koopman theory [456] (see Section 7.4). DMD may be formulated as an algorithm to identify the best-fit lineer dynamical system that advances high-dimensional measurements forward in time [535]. In this way, DMD approximates the Kooptana operator restricted to the set of direct measurements of the state of a high-dimensional system. This concarrison between the computationally straightforward and linear DMD framework and nonlinear dynamical systems has generated considerable interests in these methods [317].

Within a short amout of time, DMD has become a workhore algorithm for the datadirect hardress and the high-dimensional systems. DMD is equivalently wild for expeniential and numerical data, as it is not based on knowledge of the governing equations, but is sinceal based provide on measurement data. The DMD algorithm may also be seen as connecting the favorable aspects of the SVD (see Chapter 1) for spatial dimensionality reduction and her PF1 (see Chapter 2) for temporal fragmenty absorbing the DMD mode is associated with a particular (spronule $\lambda = a + i\hbar$, with a maticular floament α oscillation λ and moth or devar rate a

There are many variants of DMD and it is connected to existing techniques from oyten identification and extraction. DMD has become expectively popular in recent years in large part due to its simple numerical implementation and strong connections to founder dynamical systems with Koopman operand theory? Building and the strong strong feasible platform, both numbratically and manufactures (by characterized to the strong feasible platform). Both numbratically and manufactures (by characterized to the strong strong

The DMD Algorithm

Secaral apportants have been proposed for DMD, although here we present the caract DMD framework developed by 10 r at al. [353], whereas cartier fromations required uniform sampling of the dynamics in time, the approach presented here works with irregularly sampled data and with concentented data from several different experiments or numerical simulations. Moreover, the exact formulation of Ts et al. provides a precise mathematical definition of DMD that allows for digroscore horecritor length. Finally, react DMD is bosed on the efficient and numerically well-conditioned singular value decomposition, as is the original formation by Schuld (472).

DMD is inherently data-driven, and the first step is to collect a number of pairs of anythost of the start of a system air is cover in time. These standpot pairs may be denoted by $[0x(t_2), x(t_2')]_{times}^{times}$, where $t_2' = t_1 + \Delta t_1$ and the timestep Δt is sufficiently small to reache the highest frequencies in the dynamics. At letters, and paped may be bet start of a location, that is recharged into high-dimensional column vector. These snapshots are the samped into two dom numbers, Δt :

$$\mathbf{X} = \begin{bmatrix} \mathbf{x}(t_1) & \mathbf{x}(t_2) & \cdots & \mathbf{x}(t_m) \end{bmatrix} \quad (7.13a)$$

$$\mathbf{X}' = \begin{bmatrix} \mathbf{x}(t'_1) & \mathbf{x}(t'_2) & \cdots & \mathbf{x}(t'_m) \end{bmatrix}$$
. (7.13b)

The original formulations of Schmid [472] and Rowley et al. [456] assumed uniform sampling in time, so that $t_k = kAr$ and $t'_k = k_k + \Delta r = k_{k-1}$. If we assume uniform sampling in time, we will adopt the notation $\mathbf{x}_k = \mathbf{x}(Ar)$.

The DMD algorithm seeks the leading spectral decomposition (i.e., eigenvalues and eigenvectors) of the best-fit linear operator A that relates the two snapshot matrices in time:

$$X' \approx AX.$$
 (7.14)

The best fit operator A then establishes a linear dynamical system that best advances snapshot measurements forward in time. If we assume uniform sampling in time, this becomes:

$$x_{k+1} \approx Ax_k$$
. (7.15)

Mathematically, the best-fit operator A is defined as

$$\mathbf{A} = \underset{\mathbf{A}}{\operatorname{argmin}} \|\mathbf{X}' - \mathbf{A}\mathbf{X}\|_{F} = \mathbf{X}'\mathbf{X}^{\dagger} \qquad (7.16)$$

where $\|\cdot\|_F$ is the Frobenius norm and [†] denotes the pseudo-inverse. The optimized DMD algorithm generalizes the optimization framework of exact DMD to perform a regression to exponential time dynamics, thus providing an improved computation of the DMD modes and their eigenvalues [20].

It is worth noting at this point that the matrix A in (7.15) closely resembles the Koopman operator in (7.53), if we choose direct linear measurements of the state, so that $g(\mathbf{x}) = \mathbf{x}$. This connection was originally established by Rowley, Meric and collaboratori [456], and has sparked considerable interest in both DMD and Koopman theory. These connections will be explored in more depth below.

For a high-mession disc vector $x \in \mathbb{R}^n$, the matrix A has a^n elements, and argues the matrix blass and the comparison is a part of the bimarchike integrations and the disc comparison is a part of the matrix blass and the generations of A and the messations in the star of the matrix blass and the generation of A and the messation is a star of the messation of the matrix S. Since the matrix the messation is a star of the matrix of the messation of the matrix A with more ensembles in hand of comparise the high-flow matrix of the matrix o

Step 1. Compute the singular value decomposition of X (see Chapter 1):

$$\mathbf{X} \approx \tilde{\mathbf{U}} \tilde{\mathbf{\Sigma}} \tilde{\mathbf{V}}^*$$
, (7.17)

where $\hat{U} \in \mathbb{C}^{M'}$, $\hat{E} \in \mathbb{C}^{M'}$, and $\hat{V} \in \mathbb{C}^{M''}$ and $r \geq n$ denotes either the exact or approximate rank of the data matrix **X**. In practice, choosing the approximate rank *r* is one of the most important and subjective steps in TDMD, and in dimensionality reduction in general. We advocue the principled hard-thresholding algorithm of Gavistian **d** Donole **D** (o) determine *r* from looky data (see Scene) r.7). The columns of the matrix \hat{U} are also known as POD modes, and they satisfy $\hat{U} \hat{U} = L$ Similarly, columns of \hat{V} are column of \hat{V} are column.

Step 2. According to (7.16), the full matrix A may be obtained by computing the pseudo-inverse of X:

$$\mathbf{A} = \mathbf{X}^{\prime} \mathbf{\tilde{V}} \mathbf{\tilde{\Sigma}}^{-1} \mathbf{\tilde{U}}^{*}. \quad (7.18)$$

However, we are only interested in the leading r eigenvalues and eigenvectors of A, and we may thus project A onto the POD modes in U:

$$\tilde{A} = \tilde{U}^* A \tilde{U} = \tilde{U}^* X' \tilde{V} \tilde{\Sigma}^{-1}$$
(7.19)

The key observation here is that the reduced matrix Å has the same nonzero eigenvalues as the full matrix A. Thus, we need only compute the reduced Å directly, without ever working with the high-dimensional A matrix. The reduced-order matrix Å defines a linear model for the dynamics of the vector of POD coefficients §:

$$\tilde{x}_{l+1} = \tilde{A}\tilde{x}_{l}$$
. (7.20)

Note that the matrix \hat{U} provides a map to reconstruct the full state x from the reduced state \hat{x} : $x = \hat{U}\hat{x}$.

Step 3. The spectral decomposition of A is computed:

$$AW = WA$$
. (7.21)

The entries of the diagonal matrix A are the DMD eigenvalues, which also correspond to eigenvalues of the full A matrix. The columns of W are eigenvectors of \tilde{A}_{i} and provide a coordinate transformation that diagonalizes the matrix. These columns may be thought of as linear combinations of POD mode amplitudes that behave linearly with a single temporal pattern given by δ_{i} .

Step 4. The high-dimensional DMD modes Φ are reconstructed using the eigenvectors W of the reduced system and the time-shifted snapshot matrix X' according to:

$$\Phi = \mathbf{X} \mathbf{\tilde{V}} \mathbf{\tilde{\Sigma}}^{-1} \mathbf{W}. \quad (7.22)$$

Remarkably, these DMD modes are eigenvectors of the high-dimensional A matrix corresponding to the eigenvalues in A, as shown in Tu et al. [535]:

$$\begin{split} \mathbf{A} \Phi &= (\mathbf{X}^T \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} \underbrace{\tilde{\mathbf{U}}^* (\mathbf{X}^T \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} }_{\tilde{\mathbf{X}}} \mathbf{W} \\ &= \mathbf{X}^T \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} \tilde{\mathbf{A}} \mathbf{W} \\ &= \mathbf{X}^T \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} \mathbf{W} \mathbf{A} \\ &= \Phi \mathbf{A} \end{split}$$

In the original paper by Schmid [472], DMD modes are compared using $\Phi = \hat{U}W_i$ which are known as projected moder, however, these modes are to guaranteed to be cause dispervisors of A Beauce A is defined as $A = XX^2$, eigencentors of A should be in the column space of X, as in the cause IMM definition, intended of the column space of X and V = 0 original DMD algorithm. In practice, the column space of X and X will test to be nearly definited for dynamical systems with low-rank structure, so that the projected and east: DMD mode of nearconcepter.

To find a DMD mode corresponding to a zero eigenvalue, $\lambda = 0$, it is possible to use the exact formulation if $\phi = \mathbf{X}^T \hat{\mathbf{Y}} \hat{\mathbf{\Sigma}}^{-1} \mathbf{w} \neq 0$. However, if this expression is null, then the projected mode $\phi = \tilde{U} \mathbf{w}$ should be used.

Historical Perspective

In the original formulation, the snapshot matrices X and X' were formed with a collection of sequential snapshots, evenly snaced in time:

$$\mathbf{X} = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{x}_1 & \mathbf{x}_2 & \cdots & \mathbf{x}_m \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}$$
(7.23a)

$$\mathbf{X}' = \begin{bmatrix} \mathbf{1} & \mathbf{1} & \cdots & \mathbf{1} \\ \mathbf{x}_2 & \mathbf{x}_3 & \cdots & \mathbf{x}_{m+1} \\ \mathbf{1} & \mathbf{1} & \cdots & \mathbf{1} \end{bmatrix}$$
. (7.23b)

Thus, the matrix X can be written in terms of iterations of the matrix A as:

$$\mathbf{X} \approx \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{x}_1 & \mathbf{A}\mathbf{x}_1 & \cdots & \mathbf{A}^{m-1}\mathbf{x}_1 \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}. \quad (7.24)$$

Thus, the columns of the matrix X belong to a Krylov subspace generated by the propagator A and the initial condition x₁. In addition, the matrix X' may be related to X through the *shift* operator as:

$$X' = XS$$
, (7.25)

where S is defined as

$$\mathbf{S} = \begin{bmatrix} 0 & 0 & 0 & \cdots & 0 & a_1 \\ 1 & 0 & 0 & \cdots & 0 & a_2 \\ 0 & 1 & 0 & \cdots & 0 & a_3 \\ \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ 0 & 0 & 0 & \cdots & 1 & a_m \end{bmatrix}.$$
 (7.26)

Thus, the first w = 1 columns of X are obtained directly by shifting the corresponding columns of X, and the last column is obtained as a best-fit considuation of the set columns of X that minimizes the residual. In this way, the PMD algorithm resembles as Arnolds algorithm used to find the dominant explorations and experisences of a matrix A fromoph fractions. The mark S with future dependences with the high-dimensional A matrix, to this dependence of X is a straightforward of X and X are complete the set of X and X are completed to X are completed to X and X are completed to X and X are completed to X are completed to X and X are completed to X and X are completed to X and X are completed to X are completed to X and X are completed to X are completed to X are completed to X are completed to X and X are completed to X are complete

Spectral Decomposition and DMD Expansion

One of the most important aspects of the DMD is the ability to expand the system state in terms of a data-driven spectral decomposition:

$$\mathbf{x}_{k} = \sum_{j=1}^{r} \phi_{j} \lambda_{j}^{k-1} b_{j} = \Phi \Lambda^{k-1} \mathbf{b},$$
 (7.27)

where ϕ_j are DMD modes (eigenvectors of the A matrix), λ_j are DMD eigenvalues (eigenvalues of the A matrix), and b_j is the mode amplitude. The vector **b** of mode amplitudes is generally computed as

$$b = \Phi^{\dagger} x_{1}$$
. (7.28)

More principled approaches to select dominant and sparse modes have been considered [129, 270]. However, computing the mode amplitudes is generally quite expensive, even using the straightforward definition in (7.28). Instead, it is possible to compute these amplitudes using POD projected data:

$$c_1 = \Phi b$$
 (7.29a)

$$\implies \tilde{U}\tilde{x}_1 = X'\tilde{V}\tilde{\Sigma}^{-1}Wb$$
 (7.29b)

$$\implies \tilde{x}_1 = \tilde{U}^* X' \tilde{V} \tilde{\Sigma}^{-1} W b$$
 (7.29c)

$$\implies \tilde{x}_1 = \tilde{A}Wb$$
 (7.29d)

$$\implies \tilde{x}_1 = WAb$$
 (7.29c)

$$\implies \mathbf{b} = (\mathbf{W}\mathbf{A})^{-1}\tilde{\mathbf{x}}_1.$$
 (7.29f)

The matrices W and A are both size $r \times r$, as opposed to the large Φ matrix that is $n \times r$.

The spectral expansion above may also be written in continuous time by introducing the continuous eigenvalues $\omega = log(\lambda)/\Delta t$:

$$\mathbf{x}(t) = \sum_{j=1}^{t} \phi_j e^{\omega_j t} b_j = \mathbf{\Phi} \exp(\mathbf{\Omega} t) \mathbf{b}, \quad (7.30)$$

where Ω is a diagonal matrix containing the continuous-time eigenvalues ω_1 .

Example and Code

A basic DMD code is provided here:

```
        function [Phi, Lanka, b] = DHD(X, Xprime, r)

        [D_Jigma, v] = work(x, *exemt);
        # Step 1

        Sigmar = Sigma(x, r), r);
        # Step 2

        Sigmar = Sigma(x, r), r);
        W = V(x, 1nr);

        W = V(x, 1nr);
        W = V(x, 1nr);

        W = V(x, 1nr);
        # Step 2

        Sigmar = Sigma(x, r), r);
        # Step 2

        Sigmar = Sigma(x, r), r);
        # Step 3

        Phi = Sprime(Vr/Sigmar)*W;
        # Step 4

        Sight = Sigma(r)*(r);
        # Step 4
```



Figure 7.3 Overview of DMD illustrated on the fluid flow past a circular cylinder at Reynolds number 100. Rewnshood from [317].

This DMD code is demonstrated in Fig. 7.3 for the fluid flow past a circular cylinder at Reynolds number 100, based on the cylinder diameter. The two-dimensional Navier-Sobset equations are simulated using the immersed boundary projection method (IBPM) solver¹ based on the fast multi-domain method of Taira and Colonius [511, 135]. The data required for this example may be downloaded without running the IBPM code at *admlosol.com*.

With this data, it is simple to compute the dynamic mode decomposition:

```
# VORTALL contains flow fields reshaped into column vectors
X = VORTALL;
[Phi, Lambda, b] = IND(X(:,1:end-1),X(:,2:end),21);
```

Extensions, Applications, and Limitations

One of the major advantages of dynamic mode decomposition is its simple framing in terms of linear regression. DMD does not require knowledge of governing equations. For this reason, DMD has been applied by extended to include several methodological innovations and has been widely applied by byout flatdy dynamics (117), where it originates. Here, we pred we also prior of the leading algorithmic extension and promising domains applications, reserved. The term of the tendence of the DMD theory the must be addressed in forme reserved.

Methodological Extensions

 Compression and randomized linear algebra: DDD was originally designed for high-dimensional data sets in fluid dynamics, such as a fluid velocity or verice ity field, which may contain millions of degrees of freedom. However, the fact that DDD often uncorrection-dimensional structure in these high dimensional data implies that there may be more efficient measurement and comparisonal strategies based on principles of paralyity des Chapera 33. There have been several independents and highly successful extrasmiss and modifications of DMD to exploit low-rank structure and source).

1 The IBPM code is publicly available at: https://github.com/cwrowley/hpm.

In 2014, Jovanovic et al. [270] used sparsity promoting optimization to identify the fewest DMD modes required to describe a data set, essentially identifying a few dominant DMD mode amplitudes in b. The alternative approach, of testing and comparing all subsets of DMD modes, represents a computationally intractable brate force search.

Another line of work is based on the fact that DMD modes generally admit a gener representation in Fourier or worker bases. Moreover, the time dynamics of each mode are simple pure tone harmonics, which are the definition of sparse in a Fourier basis. This sparsity has facilitated several efficient measurement strategies that reduce the number of measurements required in time (554) and space [90, 22, 174], based on compressed sensing. This has the broad potential to enable highresolution characterization of systems from under-coorden quesurements.

Related to the use of compressed censing, randomized linear algebra has recently been used to accelerate DBD comparison when this state data in a sublishe. Instead of collecting subsampled measurements and using compressed sensing to line fraidpationessional attractions: randomized methods have a strate with full data and then randomly project into a lower-dimensional and subspace, where comparison may be performed a modeling of the strategies of the strategies of the strategies of the modeling of the strategies of the strategies of the strategies of the modeling of the strategies of the strategies of the strategies of the modeling of the strategies of the straneous of the strategies o

Finally, libraries of DMD modes have also been used to identify dynamical regimes [308], based on the sparse representation for classification [560] (see Section 3.6), which was used earlier to identify dynamical regimes using libraries of POD modes [80, 98].

Inputs and control. A major strength of DDD is the ability to advance may be and high-dimensional systems in terms on sum matters of dominant modes, which represent space-temporal concerns measures. Reducing the dismonsionality of and inverse integra concerns and the strength of the strength of the discovery integra controllers with higher performance and nobusces. Thus, compare and reflection and extination, the corealisery predictioning presulty transfer directly into controllers with higher performance and nobusces. Thus, compare and reflection presentations of compared systems such as that flows have been to compare space of the effect of archiver and controls.

Shortly after the original DMD algorithm, Proctor et al. [434] extended the algorithm to disambiguate between the natural unforced dynamics and the effect of actuation. This essentially amounts to a generalized evolution equation

$$x_{k+1} \approx Ax_k + Bu_k$$
, (7.31)

which results in another linear regression problem (see Section 10.1).

The original motivation for DMD with control (DMDc) was the use of DMD to characterize epidemiological systems (e.g., malaria spreading across a continent), where it is not possible to stop intervention efforts, such as vaccinations and bed nets, in order to characterize the unforced dynamics [433]. Since the original DMDs algorithm, the compressed sensing DMD and DMDs algorithms have been combined, resulting in a new finance of for compressive system identification [30]. In this framework, it is possible to collect undersampled measurements or an entanted system and identify an accurate and efficient loworder model, related to DMD and the eigensystem realization algorithm (ERA; see Section 9.3) 12721.

DMDs models, based on linear and nonlinear measurements of the system, have recerrly been used with model predictive control (MPC) for enhanced control of nonlinear systems by Korda and Mezič [302]. Model predictive control using DMDs models were subsequently used as a benchmark comparison for MPC based on fully nonlinear models in the work of Kaiser et al. [277], and the DMDs models performed surrisingly well. zero for stronger homolinear systems.

Notificate measurements. Much of the excitonent around DBD due to be more sourcestein to nonlinear quarks in the Keensen personal (158), historia excitation and an excitation of the sourcester and the source of the test error system, as long as a sufficient amount of data is obtened. The bits DBD algorithms are loss measurement of the system, bits the argenrithy not field enough to characterize rates possibility of persons, bits the generality not field enough to characterize rates possibility of the system, and the source of the system of the system of the system, measurements were angenred to livelab outdointe measurements of the system, are straight the basis and to prepresent the Koopman operator. The so-called rated DBD (169) and persons and the system and the system and persons and the system of the system of the system of the system of the system.

$$y_{k+1} \approx A_Y y_k$$
. (7.32)

For high-dimensional systems, this augmented state y may be intractably large, motivating the use of kernel methods to approximate the evolution operator A_Y [557]. This kernel DMD has since been extended to include dictionary learning techniouse [332].

It has recently been shown that eDMD is equivalent to the variational approach of conformation dynamics (VAC) [463, vol.763], fand derethe by Noci and Niske in 2013 to simulate molecular dynamics with a broad separation of timescales. Further connections between eDMD and VAC and between DMD and the time lagged indipendent component analysis (TFLA) are explored in a secret network [237]. A key contribution of VAC is a variational score enabling the objective assessment of Koopman models via cross-validation.

Following the extended DMD, it was shown that there are relatively statistics confidence for detaining large regression models that includes the encloqued state of the system [72]. For totalizer systems with multiple fixed points, periode detains, the state share in spherical state of the system [72]. The state of the system [73] and the fixed point of the state share in spherical state of the system [74] and the fixed point of the state share in spherical state of the system [74] and the desay model are spherical state of the system [74] and the state share in spherical state of the state state in spherical state state in the spin of these explorestors, allowing it may be possible to interivity time its state is noted with the spin of these explorestors, allowing its may be possible to interivity time its state is noted with the spin of the state state in the spin of the state state state of the state state is noted with the spin state sta De-ostisting. The DMD algorithm is purely data-driven, and is thus equally applicable to experimental and numerical data. When charactering experimental data with DMD, the effects of assort noise and stochastic disturbances must be accounted for the original DMD algorithm is purchastrally securitive to moste, and it was shown that 28, 147, 2411, Athonoph increased sampling decrements the variance of the eigenvalue disturbances.

There are several approaches to correct for the effect of senson noise and disturbances. Hennel et al [241] use the total least-aquarts regression to account for the possibility of noisy measurements and disturbances to the state, replacing the original least-squares regression. Dawson et al. [147] compate DMD on the data in forward and backward mice and the average the resulting operator, removing the systematic bias. This work also provides an excellent discussion on the sources of noise and a compations of various denoising algorithms.

More recently, Adham and Kura [20] introduced the optimized DMD algorithm, which uses a variable projection method for onelinear less against to compute the DMD for unevenly timed samples, significantly mitigating the bias due to noise. The subspace DMD algorithm of Talcelah et al. [514] alion compensates for measurement noise by computing an orthoganal projection of future samplatos conto the space of previous samphots and then constructing a linear model. Extensions that combine DMD with Breselan anoreonchen have all been developed [512].

- Multirevolution, DMD is often applied to complex, high-dimensional dynamical systems, with a high-dimensional dynamical system, and an abula burdence or optical system, solution and the system and t
- Doty measurements. Although DDD was developed for high-dimensional data where it is assume that one has access to the fiduates of a systems with history that the structure spatio-temporal elements measures in systems with history of the systems of the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of the system (and the structure) and the structure of th
- Streaming and parallelized codes. Because of the computational burden of computing the DMD on high-resolution data, several advances have been made to accel-

ente DAD in streaming applications and with parallelized approxime. DADI is often used in a streaming engine, where a moving window of samphote are processed continuously, resulting in redundant comparations when new data become a valiable. A several approxime confer or treaming DADI housed on the intermetaria DVD [242], a streaming method of samphots SVD [142], and mak-one updates to the DADI MOM Dappeting and the output parallelized, as its based on the SVD. Several parallelized codes are available, based on the QR [466] and SVD [175, 177, 176].

Applications

Fluid dynamics. DND originated in the fluid dynamics community (472), and has since been applied to a wide range of dom generaties (are, covity) fluo, wates, channel flow, boundary layers, etc.), to study mixing accouries, and conhustion, among other phenomena. In the original paper of Schmid (474, 472), both a cavity flow and a jet were considered. In the original paper of Schmid (474, 472), both a cavity flow and a jet were considered. In the softmat paper and schward paper of the softward of the softward paper of Schmid (473, 494, 482, 472), and (etc.), 473, 494, 482, 472.

DMD has also been applied to wake flows, including to investigate frequency lockon [534], the wake past a gurney flap [415], the cylinder wake [28], and dynamic stall [166]. Boundary layers have also been extensively studied with DMD [411, 465, 3831. In acoustics, DMD has been used to canture the near-field and far-field acoustics that result from instabilities observed in shear flows [495]. In combustion, DMD has been used to understand the coherent heat release in turbulent swirl flames [387] and to analyze a rocket combustor [258]. DMD has also been used to analyze nonnormal growth mechanisms in thermoacoustic interactions in a Riike tube, DMD has been compared with POD for reacting flows [459]. DMD has also been used to analyze more exotic flows, including a simulated model of a high-speed train [392]. Shock turbulent boundary layer interaction (STBLI) has also been investigated, and DMD was used to identify a pulsating separation bubble that is accompanied by shockwave motion [222]. DMD has also been used to study self-excited fluctuations in detonation waves [373]. Other problems include identifying hairpin vortices [516], decomposing the flow past a surface mounted cube [393], modeling shallow water equations [65], studying nano fluids past a square cylinder [463], and measuring the growth rate of instabilities in annular liquid sheets [163].

- Epidemiology. DMD has recently been applied to investigate epidemiological systems by Proceed and Exhaft (F3X). This is a particularly interpretable application, as modal frequencies often correspond to yearly or seasonal fluctuations. Morrorer, the phase of DMD mode gives inability into how discass from propagate spatially, potentially informing future intervention efforts. The application of DMD to desay season in order to identify the undready during the theory of the vaccinations in order to identify the undready during the theory of the propagate the theory of the the the theory of the theory of the the t
- Neuroscience. Complex signals from neural recordings are increasingly highfidelity and high dimensional, with advances in hardware pushing the frontiers of data collection. DMD has the potential to transform the analysis of such neural recordings, as evidenced in a recent study that identified dynamically relevant betaures in ECOG data of desping particus [50]. Since then, several works

have applied DMD to neural recordings or suggested possible implementation in hardware [3, 85, 520].

- Video processing. Separating foreground and background objects in video is a common task in surveillance applications. Real-time separation is a challenge that is only exacerbated by ever increasing video resolutions. DMD provides a flexible platform for video separation, as the background may be approximated by a DMD mode with zero eigenvalue [223, 174, 244].
- Other applications. DMD has been applied to an increasingly diverse array of problems, including robotics [56], finance [363], and plasma physics [517]. It is expected that this true will increase.

Challenges

- Traveling waves: DMD is based on the SVD of a data matrix X = UXV* whose columns are optial measurements colving in finite. In this case, the SVD is a spacetime separation of variables into spatial modes, given by the columns of U, and fune dynamics, given by the columns of V. As is PDD. MDD thas has limitations for problems that exhibit traveling waves, where separation of variables is known to fail.
- Transients. Many systems of interest are characterized by transients and intermittent
 phenomena. Several methods have been proposed to identify these events, such as the
 multi-resolution DMD and the use of delay coordinates. However, it is still necessary
 to formalize the choice of relevant timescales and the window size to compute DMD.
- Confinemes spectrum. Related to the above, many systems are characterized by boundant flowparcy contents, as spoose to alse distinct and advector flopeness. This broadbank flopparay contents is also shown as a southname aperciane, where the standard flopparay continuous program, and the standard standard

Secural data-driven approaches have been recently proposed to handle systems with continuous spectra. Applying DDID to a vector of delayed measurements of a system, the so-called *HAVOK* analysis in Section 7.5, has been shown to approcimate the dynamics of chaode systems, such as the Lorenz system, which exhibits a continuous spectrum. In addition, Lasset et al. [309] showed that is is possible to design a deep learning architecture with an auxiliary network to parameterize the continuous frequency.

 Strong nonlinearity and choice of measurements: Although significant progress has been made connecting DMD to molitizen systems [57]. It choicing nonlinear measurements to argument the DMD regression is still not an exact science. Identifying measurements subspaces that remain doesd under the Koopman operator is an engang challeng [70]. Recent progress in deep horming laws the patorial to enable (564, 412, 2017).

7.3 Sparse Identification of Nonlinear Dynamics (SINDy)

Discovering dynamical systems models from data is a central challenge in undernatical physics, with a rich history going back tasta at at as at the line of Koplet and Newton and the discovery of the laws of planetary motion. Historically, this process relied on a combination of high-quality measurements and experi inmition. With wat quantities of data and increasing computational power, the austomated discovery of governing equations and dynamical systems is a new rad ecuting scientific paradigm.

Typically, the form of a candidate model is either constrained via prior knowledge of the governing equipations, as in Garken projection [402, 452, 77, 404, 119, 559, 72, 118] (see Chapter 12), or a handhal of hearistic models are tested and parameters are optimized to in data. Alternatively, best-fit linear models may be doubled using DMD or IERA. Simultaneously identifying the numbers structure and parameters of a model from data. Simultaneously interactivity are a combinated through the structure model structers.

The sparse identification of nonlinear dynamics (SINDy) algorithm [95] bypasses the intractable combinatorial search through all possible model structures, leveraging the fact that many dynamical systems

$$\frac{d}{dt}\mathbf{x} = \mathbf{f}(\mathbf{x}) \quad (7.33)$$

have dynamics \mathbf{f} with only a few active terms in the space of possible right-hand side functions; for example, the Lorenz equations in (7.2) only have a few linear and quadratic interaction terms per equation.

We then seek to approximate f by a generalized linear model

$$\mathbf{f}(\mathbf{x}) \approx \sum_{k=1}^{p} \theta_k(\mathbf{x})\xi_k = \mathbf{\Theta}(\mathbf{x})\xi,$$
 (7.34)

with the fewest nonzero terms in § as possible. It is then possible to solve for the relevant terms that are active in the dynamics using sparse regression [518, 573, 236, 264] that penalizes the number of terms in the dynamics and scales well to larve problems.

First, time-series data is collected from (7.33) and formed into a data matrix:

$$\mathbf{X} = \begin{bmatrix} \mathbf{x}(t_1) & \mathbf{x}(t_2) & \cdots & \mathbf{x}(t_m) \end{bmatrix}^T. \quad (7.35)$$

A similar matrix of derivatives is formed

$$\dot{\mathbf{X}} = \begin{bmatrix} \dot{\mathbf{x}}(t_1) & \dot{\mathbf{x}}(t_2) & \cdots & \dot{\mathbf{x}}(t_m) \end{bmatrix}^T. \quad (7.36)$$

In practice, this may be computed directly from the data in X; for noisy data, the totalvariation regularized derivative tends to provide numerically robust derivatives [125]. Alternatively, it is possible to formulate the SINDy algorithm for discrete-time systems $x_{n+1} = F(x_n)$, as in the DMD algorithm, and avoid derivatives entirely.

A library of candidate nonlinear functions $\Theta(X)$ may be constructed from the data in X:

$$\Theta(\mathbf{X}) = \begin{bmatrix} \mathbf{1} & \mathbf{X} & \mathbf{X}^2 & \cdots & \mathbf{X}^d & \cdots & \sin(\mathbf{X}) & \cdots \end{bmatrix}$$
(7.37)

Here, the matrix X^d denotes a matrix with column vectors given by all possible time-series of d-th degree polynomials in the state x. In general, this library of candidate functions is only limited by one's imagination. The dynamical system in (7.33) may now be represented in terms of the data matrices in (7.36) and (7.37) as

$$\hat{X} = \Theta(X)\Xi$$
. (7.38)

Each column ξ_k in Ξ is a vector of coefficients determining the active terms in the k-th row in (7.33). A parsimonious model will provide an accurate model fit in (7.38) with as few terms as possible in Ξ . Such a model may be identified using a convex ℓ_1 -regularized sparse represents.

$$\xi_k = \operatorname{argmin}_{k'} \| \dot{X}_k - \Theta(X) \xi'_k \|_2 + \lambda \| \xi'_k \|_1.$$
 (7.39)

Here, $\mathbf{\hat{x}}_1$ is the 4-th column of $\mathbf{\hat{x}}_1$ and λ is a spanity-promoting back Sparse regression, such as the LASSO [158] or the sequential interoloded tear-squares (STLS) algorithm used in SINDy [99], improves the numerical robustness of this identification for noisy oredetermined problem, in contrast to carlier methods [258] that used compresed sensing [150, 109, 112, 111, 113, 39, 529]. We advocate the STLS (Code 7.1) to select active terms.

Code 7.1 Sequentially thresholded least-squares

```
Idention 5: = specify/pression(Thirt, ddb)_ideh(s))

Starting (Start, Start, St
```

The sparse vectors & may be synthesized into a dynamical system:

$$\dot{x}_{k} = \Theta(\mathbf{x})\xi_{k}$$
. (7.40)

Note that x_k is the k-th element of x and $\Theta(x)$ is a row vector of symbolic functions of x, as opposed to the data matrix $\Theta(X)$. Fig. 7.4 shows how SINDy may be used to discover the Lorenz equations from data. Code 7.2 generates data and performs the SINDy regression for the Lorenz system.

Code 7.2 SINDy regression to identify the Lorenz system from data.

```
%% Generate Data
Bata = [10:20;s/3], % Lorenz's parameters (chaotic)
n = 3;
m = 3;
m = 2; * 2; * 1; * Initial condition
types(0;10:10;);
mptimes = odeset('shift).is=12, 'Abarb','l=-12=cmes(1,n));
[5,3] ede64(0;k,1) lorenz(1,x,1)=13, 'Abarb','l=-12=cmes(1,n));
[5,3] ede64(0;k,1) lorenz(1,x,1)=13, 'Abarb','l=-12=cmes(1,n));
[5,3] ede64(0;k,1) lorenz(1,x,1)=13, 'Abarb','l=-12=cmes(1,n));
[5,1] edef(1,k,1) lorenz(1,x,1)=13, 'Abarb','l=-12=cmes(1,n));
[5,1] edef(1,k,1)=13, 'Abarb','l=-12=cmes(1,n));
[5,1] edef(1,k,1)=13, 'Abarb','l=-13, 'Abarb','l=
```



Figure 7.4 Schematic of the sparse identification of nonlinear dynamics (SINDy) algorithm [95]. Parsimotions models are selected from a library of candidate nonlinear terms using sparse regression. This library $\Theta(X)$ may be constructed purely from measurement data. *Modified from Browton et al.* [95].

```
dx(i,:) = lorenx(0,x(i,:),Reta);
and
10 Build library and compute generative regression
texts a specification,a); 1 Mg for bird order polynomials
lambds # 0.005; 8 lambds is our generification knob.
Xi = specifyrynmics(Theta, ac,lambda, J)
```

This code also relies on a function **poolData** that generates the library Θ . In this case, polynomials up to third order are used. This code is available online.

	. 3000.		- 200C -
111	[0]	[0]	[0]
' x'	[-10.0000]	[28.0000]	[0]
1.81	[10.0000]	[-1.0000]	[0]
121	[0]	[0]	[-2.6667]
" xx "	[0]	[0]	[0]
' xy'	[0]	[0]	[1.0000]
'xx'	[0]	[-1.0000]	[0]
1771	[0]	[0]	[0]
' yz'	[0]	[0]	[0]
1221	[0]	[0]	[0]
" xxxx "	[0]	[0]	[0]
' xxy'	[0]	[0]	[0]
'xxz'	[0]	[0]	[0]
' ×yy '	[0]	[0]	[0]
'xyz'	[0]	[0]	[0]
'xzz'	[0]	[0]	[0]
' ' ' ' ' '	[0]	[0]	[0]
'yyz'	[0]	[0]	[0]
'yzz'	[0]	[0]	[0]
'222'	[0]	[0]	[0]

The output of the SINDy algorithm is a snarse matrix of coefficients Z:

The result of the SINDy regression is a parsimonious model that includes only the most important terms required to explain the observed behavior. The space regression procedure used to identify the most parsimonious nonlinear model is a convex procedure.



Figure 32 Schematic overview of nonlinear model identification from high-dimensional data using the sparse identification of nonlinear dynamics (SNDy) [95]. This precedure is modular, so that different techniques can be used for the feature extraction and repression steps. In this example of flow past a cylinder, SNDy discovers the model of Noack et al. [402]. Modified from Branton et al. [95].

The alternative appears, back in more regression one very possible space solutions the structure of the optimization and machine terming. It is interesting to note that for detection the structure of the ACN contains only of the structure of the structure possible structure of the SS3, 317 JL structure of the s

Applications, Extensions, and Historical Context

The SINDy algorithm has recently been applied to identify high-dimensional dynamical systems, such as fluid flows, based on POD coefficients (95, 341, 342). Fig. 7.5 illustrates the application of SINDy to the flow past a cylinder, where the generalized meant-field model of Noack et al. [402] was discovered from data. SINDy has also been applied to identify models in nonlinear optics [497] and plasma physics [141].

Because SDRVb is formulated in terms of linear regression in a sunliner library, it is highly stessible. The SDRV functionals has non-recently generalized by Lisonn and Brannen [14] to incorporate lowon physical constraints and symmetries in the equations. In parametring constraints of questiful physical constraints in the Notice Solice include, energy preserving constraints on the quadratic nonlinearities in the Notice Solice means and the straints of the straints and the straints of the straints of the straints of the straints of the straint protocol straints (1953), 211 [31]. The straints which is shown in the physical induces are parameterized by the big straints of the straints of the straints of the straints (1953), 211 [31]. The straints of the straints of the straints (1953), 211 [31]. The straints of the straints of the straints (1953), 211 [31]. The straints of the straints of the straints (1953), 211 [31]. The straints of the straints of the straints (1953), 211 [31]. The straints of the straints of the straints (1953), 211 [31]. The straints of the straints (1953), 211 [31]. The straints of the straints (1953), 211 [31]. The straints (1953), 211 [31]. The straints of the straints (1953), 211 [31]. The s Linear et al. [321] also demonstrated the ability of SMDy is isolatify dynamical type, immodies of high-dimensional systems, and an hidd fraces, from a fee physical ensurmeasurements, as ha thirt and ang measurements on the system's flace 10^{-1} and the model systems. SMDy his best espective to hinder large model on the one physical ensurtions of the system SMDy large model is in the system of the system of the system of the SMDy large model is incorporate information entering the system of system of the system of the inferring durity of the system SMDy large model is incorporate information entering for dispective durity (SMD) years also recently extended to incorporate information entering of the system of system of the system SMDy finances was generalized to introduce the dispective durity of the system of the system SMD physical durity. The system of the syste

More generally, the use of snarsity-promoting methods in dynamics is quite recent 1548. 467, 414, 353, 98, 433, 31, 29, 89, 364, 3661. Other techniques for dynamical system discovery include methods to discover equations from time-series [140], equation-free modeling [288], empirical dynamic modeling [503, 563], modeling emergent behavior [452], the nonlinear autoregressive model with exorenous inputs (NARMAX) [208, 571, 59, 484]. and automated inference of dynamics 1478, 142, 1431, Broadly speaking, these techniques may be classified as system identification, where methods from statistics and machine learning are used to identify dynamical systems from data. Nearly all methods of system identification involve some form of regression of data onto dynamics, and the main distinction between the various techniques is the degree to which this regression is constrained. For example, the dynamic mode decomposition generates best-fit linear models. Recent nonlinear regression techniques have produced nonlinear dynamic models that preserve physical constraints, such as conservation of energy. A major breakthrough in automated nonlinear system identification was made by Bongard and Lipson [68] and Schmidt and Lipson [477], where they used genetic programming to identify the structure of nonlinear dynamics. These methods are highly flexible and impose very few constraints on the form of the dynamics identified. In addition, SINDy is closely related to NARMAX [59], which identifies the structure of models from time-series data through an orthogonal least squares procedure.

Discovering Partial Differential Equations

A major extension of the SINDy inndeling framework generalized the library to include partial derivatives, smalling the identification of partial differential equations [646, 468]. The resulting algorithm, called the partial differential equation functional identification is nonlinear dynamics (PBE-HND), has been demonstrated to successfully identify sevent canonical PBEs from classical physics, parely from noisy data. These PBEs for the extension of the extens

PDE-FND is similar to SINDy, in that it is based on sparse regression in a library constructed from measurement data. The sparse regression and discovery method is shown in Fig. 7.6. PDE-FIND is outlined below for PDEs in a single variable, although the theory is readily generalized to higher dimensional PDEs. The spatial time-series data is arranged into a single commo vector **Y** = C^{omm}, representing data collected over *m* time toxins



Figure 25 Steps in the PEE functional identification of annihume dynamics (PEEFERD) algorithms applied to infort the Netro Subscie equations to make (neuroback der Mohler et al. 1400). Its Data compiled and the second step of the second step of the second step of the second step compiled in a large neutror 0. Incomposition granulation terms for the PEE. Its Sparse regressions to value of indicating steps starts 0. The lenge-neutron step of the second step of the se

and a spatial locations. Additional inputs, such as a known potential for the Schrödinger equation, or the magnitude of complex that, is a stranged into a column vector $Q \in \mathbb{C}^n$. Next, a litterary $\Theta(T,Q) \in \mathbb{C}^{m+2}$ of D candidate linear and nonlinear terms and partial deformations for the Postic is constructed. Environments are taken as in the differences in the Postic is constructed. Environments are taken as in the postic and the postic and the difference in the Postic is constructed. Environ is an environment of the source of the postic is th

$$\Theta(\Upsilon, \mathbf{Q}) = \begin{bmatrix} \mathbf{1} & \Upsilon & \Upsilon^2 & \dots & \mathbf{Q} & \dots & \Upsilon_X & \Upsilon\Upsilon_X & \dots \end{bmatrix}$$
. (7.41)

Each column of Θ contains all of the values of a particular candidate function across all of the nw space-time grid points on which data is collected. The time derivative Υ_{I} is also computed and reshaped into a column vector. Fig. 7.6 demonstrates the data collection and processing. As an example, a column of $\Theta(\Upsilon, Q)$ may be $a\mu_{i}^{2}$.

The PDE evolution can be expressed in this library as follows:

$$\Upsilon_I = \Theta(\Upsilon, Q)\xi.$$
 (7.42)

Each entry in ξ is a coefficient corresponding to a term in the PDE, and for canonical PDEs, the vector θ is source, meaning that only a few terms are active.

If the library Θ has a sufficiently rich column space that the dynamics are in it's span, then the PDE should be well-represented by (7.42) with a sparse vector of coefficients §. To identify the few active terms in the dynamics, a sparsity-promoting regression is employed, as in SINDy: Importantly, the regression problem in (7.42) may be poorly conditioned.

Algorithm 1 STRidge($\Theta, \Upsilon_{I}, \lambda, tol, iters$)							
$\hat{\xi} = \arg \min_{\xi} \ \Theta \xi - \Upsilon_t\ _2^2 + i$	A ≹ ² % ridge regression						
bigcoeffs = { $j : \hat{t}_j \ge tol$ } % select large coefficients							
$\hat{\xi}[\sim bigcoeffs] = 0$	% apply hard threshold						
$\hat{\xi}[bigcoeffs] = STRidge(\Theta[:, bigcoeffs], \Upsilon_1, tol, iters - 1)$							
% recursive call with fewer coefficients							
return ê							

Error in computing the derivatives will be magnified by numerical errors when inverting Θ . Thus a least squares regression radically changes the qualitative nature of the inferred dynamics.

In general, we seek the sparsers vettor **f** that satisfies (7.42) with a small residual. Intensiof an intrastruction combinatorial accessible through all possible sparse vector structures, a common technique is to relax the problem to a convert, (1. regularized leax squares [518]; however, this mesh a poferior poorly with highly correlated data. Intenda, we use right progression with hard thresholding, which we call sequential threshold ridge regression (STMg) et a Agarten approximation to **g**. We learning the approximation threshold 3, this gives a sparse approximation to **g**. We learning the elearne of Agaothm 1 to find the test predictor based on the selection circuit.

$$\bar{\xi} = \operatorname{argmin}_{\xi} \|\Theta(\Upsilon, Q)\xi - \Upsilon_{\Gamma}\|_{2}^{2} + \epsilon \kappa(\Theta(\Upsilon, Q)) \|\xi\|_{0}$$
(7.43)

where $\kappa(\Theta)$ is the condition number of the matrix Θ , providing stronger regularization for ill-posed problems. Penalizing $\|\xi\|_0$ discourages over fitting by selecting from the optimal position in a Pareo front.

As in the SNDY algorithm, it is important to provide sufficiently rich training data to disambigute brevens serveral different models. For example, Fig. 7: Illustrates the use of PRE-FIND algorithm identifying the Kortsweg-de Virus (KdV) equation. If only a single method wave and analysis, the method incontectly identifies the standard linear advection in the standard linear advection of a star tracking waves of afferent amplitudes are analyzed, the KdV equation in correctly identifies the tracking star and the star and analyzed. The KdV equation is correctly identifies the star discuss the different amplitudes deer analyzed. The KdV equation is correctly identifies in a tracking star and the star and analyzed and the star and the star advection of the star and the star and the star and the star and the star advection and the star and the star advection advecting is an extension of the star advection advection advection advecting the star advection the star advection advection advection advecting the star advection advection advection advection advecting the star advection advection advection advection advecting the star advection advection advection advection advection advection advection advection advecting the star advection advection

The PDE-FND algorithm can also be used to identify PDEs based on Lagrangian messurements that follow the pads of individual particles. For example, Fig. 72 Millioratos the identification of the diffusion equation describing Brownian motion of a particle based on a single long time-series measurement of the particle position. In this example, the time series is broken up into several short sequences, and the evolution of the expositions is used to identify the diffusion equation.

Extension of SINDy for Rational Function Nonlinearities

Many dynamical systems, such as metabolic and regulatory networks in biology, contain rational function nonlinearities in the dynamics. Often, these rational function nonlinearities arise because of a separation of time scales. Although the original SINDy algorithm is highly flexible in terms of the choice of the library of nonlinearities, it is not straightforward to identify rational functions, since on spaces linear combi-



Figure 3.7 Inferring nonlinearity via observing solutions at multiple amplitudes (reproduced from Rady et al. [460]). (a) An example 2-soliton solution to the KdV equation. (b) Applying our method to a single soliton solution determines that it solves the standard advection equation. (c) Looking at two commletely semartic solutions reveals nonlinearity.



Figure 7.8 Inferring the diffusion equation from a single Brownian motion (reproduced from Rady et al. [460]). (a) Time series is broken into many short random walks that are used to construct histograms of the displacement. (b) The Brownian motion migretory, (olisening the diffusion equation, (c) Parameter error ($||\mathbf{t}|^2 - \frac{1}{2}||_1$) vs. length of Larowa time series. Blue symbols correspond to correct identification of the structure of the diffusion model. u = correct

nations of a few basis functions. Instead, it is necessary to reformulate the dynamics in an implicit ordinary differential equation and modify the optimization procedure accordingly, as in Mannan et al. [361]. We consider dynamical systems with rational nonlinearities:

$$\dot{x}_k = \frac{f_N(\mathbf{x})}{f_D(\mathbf{x})}$$
(7.44)

where x_k is the k-th variable, and $f_N(\mathbf{x})$ and $f_D(\mathbf{x})$ represent numerator and denominator polynomials in the state variable \mathbf{x} . For each index k, it is possible to multiply both sides by the denominator f_D , resulting in the equation:

$$f_N(\mathbf{x}) - f_D(\mathbf{x})\dot{x}_k = 0.$$
 (7.45)

The implicit form of (7.45) motivates a generalization of the function library Θ in (7.37) in terms of the state x and the derivative \dot{s}_k :

$$\Theta(\mathbf{X}, \dot{x}_k(\mathbf{t})) = \left[\Theta_N(\mathbf{X}) \quad \text{diag} (\dot{x}_k(\mathbf{t})) \Theta_D(\mathbf{X})\right].$$
 (7.46)

The first term, $\Theta_N(\mathbf{X})$, is the library of numerator monomials in \mathbf{x} , as in (7.37). The second term, diag $(\hat{x}_i(t)) \Theta_D(\mathbf{X})$, is obtained by multiplying each column of the library of denominator polynomials $\Theta_D(\mathbf{X})$ with the vector $\hat{x}_i(t)$ in an element-wise fashion. For a single variable x_j , this would give the following:

$$diag(\dot{x}_k(t))\Theta(\mathbf{X}) = [\dot{x}_k(t) \ (\dot{x}_k x_k)(t) \ (\dot{x}_k x_k^2)(t) \ \dots].$$
 (7.47)

In most cases, we will use the same polynomial degree for both the numerator and denominator library, so that $\Theta_N(\mathbf{X}) = \Theta_D(\mathbf{X})$. Thus, the augmented library in (7.46) is only twice the size of the original library in (7.37).

We may now write the dynamics in (7.45) in terms of the augmented library in (7.46):

$$\Theta(\mathbf{X}, \dot{x}_{k}(t))\xi_{k} = 0.$$
 (7.48)

The sparse vector of coefficients ξ_1 will have nonzero entries for the active terms in the dynamics. However, it is not possible to use the same sparse regression procedure as in SINDy, since the sparsest vectors ξ_1 that satisfies (7.48) is the trivial zero vector.

Instead, the spansest nonzero vector ξ_0 that satisfies (7.43) is identified as the spansest vector in the null space of Θ . This is generally a nanoverse problem, although there are recent algorithms developed by Qu et al. [440], based on the alternating directions method (ADM), to identify the spacest vector in a subspace. Unlike the outgrad ISDN ja algorithm, this procedure is quite sensitive to noise, as the null-space is numerically approximated as the span of the singular vectors corresponding to small singular value. When noise in addeed to the data matrix X_i and here to Θ , the noise flow of the singular value decomposition goes up, increasing the rank of the manifestion and space.

General Formulation for Implicit ODEs

The optimization procedure above may be generalized to include a larger class of implicit ordinary differential equations, in addition to those containing rational function nonlineartities. The library $\Theta(\mathbf{X}, \lambda_i(t))$ contains a subset of the columns of the library $\Theta(\mathbf{X} \cdot \mathbf{X})$, which is obtained by building nonlinear functions of the state \mathbf{x} and derivative \mathbf{x} identifying the sparsest vector in the null space of $\Theta(\mathbf{X} \cdot \mathbf{X})$ provides more flexibility in identifying



Figure 7.9 Illustration of model selection using SINDy and information criteria, as in Mangan et al. [362]. The most parsimonious model on the Pareto front is chosen to minimize the AIC score (blue circle), nerventine overfiltine.

nonlinear equations with mixed terms containing various powers of any combination of derivatives and states. For example, the system given by

$$\dot{x}^{2}x^{2} - \dot{x}x - x^{2} = 0 \qquad (7.49)$$

may be represented as a sparse vector in the null space of $\Theta([\mathbf{x} \ \mathbf{\dot{x}}])$. This formulation may be extended to include higher order derivatives in the library Θ library, for example to identify second-order implicit differential causions:

$$\Theta([X \ \dot{X} \ \ddot{X}])$$
. (7.50)

The generality of this approach enables the identification of many systems of interest, including those systems with rational function nonlinearities.

Information Criteria for Model Selection

When performing the space engewoise in the SMD2 algorithm, the spear-by-pointering the spearse λ is in the spearse, all dimension λ with results of the spearse performance λ is in the spearse λ dimension λ with the spearse λ and the spearse λ run the same probability of the spearse λ dimension λ with the spearse λ much and the spearse shows allowing one version comparison in the χ - λ T. Thus, the spearse λ much assimption of the spearse λ much as the spearse λ much and the spearse λ much assimption of the spearse λ much as the spearse λ much assimption of the spearse λ much as the spearse λ much assimption of the spearse λ much as the spearse λ much assimption of the spearse λ much as a spearse λ much as the spearse λ much as the spearse λ much as a spearse λ much as the spearse λ much as the spearse λ much as a spearse λ much as the spearse λ much as the spearse λ much as much as the spearse λ much as the sp

7.4 Koopman Operator Theory

Koopman operator theory has recently emerged as an alternative perspective for dynamical systems in terms of the evolution of measurements g(x). In 1931, Bernard O. Koopman demonstrated that it is possible to represent a nonlinear dynamical system in terms of an infinite-dimensional linear operator acting on a Hilbert space of measurement functions of the state of the system. This so-called Koopman operator is linear, and its spectral decomposition completely characterizes the behavior of a nonlinear system analogous to (7.7). However, it is also infinite-dimensional, as there are infinitely many degrees of freedom required to describe the store of all possible measurement functions e of the state. This poses new challenges. Obtaining finite-dimensional, matrix approximations of the Koonman operator is the focus of intense research efforts and holds the promise of enabling globally linear representations of nonlinear dynamical systems. Expressing nonlinear dynamics in a linear framework is appealing because of the wealth of optimal estimation and control techniques available for linear systems (see Charter 8) and the ability to analytically predict the future state of the system. Obtaining a finite-dimensional approximation of the Koopman operator has been challenging in practice, as it involves identifying a subspace spanned by a subset of eigenfunctions of the Koopman operator.

Mathematical Formulation of Koopman Theory

The Koopman operator advances measurement functions of the state with the flow of the dynamics. We consider real-valued measurement functions $g : \mathbf{M} \to \mathbf{R}$, which are elements of an infinite-dimensional Hilbert space. The functions g are also commonly flown are solverwalking, although this may be construct with the unrelated downrublity from control theory. Typically, the Hilbert space is given by the Lebesgue spars-integrable functions \mathbf{M} is enclosed on the same space are also valid.

The Koopman operator K_{γ} is an infinite-dimensional linear operator that acts on measurement functions ρ as:

$$K_{tR} = g \circ F_t$$
 (7.51)

where \circ is the composition operator. For a discrete-time system with timestep Δr , this becomes:

$$K_{\Delta x}g(\mathbf{x}_{k}) = g(\mathbf{F}_{\Delta x}(\mathbf{x}_{k})) = g(\mathbf{x}_{k+1}).$$
 (7.52)

In other words, the Koopman operator defines an infinite-dimensional linear dynamical system that advances the observation of the state $v_2 = v(x_2)$ to the next time ster:

$$g(\mathbf{x}_{k+1}) = K_{\Delta I} g(\mathbf{x}_{k}).$$
 (7.53)

Note that this is true for any observable function g and for any state x₂.

The Koopman operator is linear, a property which is inherited from the linearity of the addition operation in function spaces:

$$\mathcal{K}_{r}(\alpha_{1}g_{1}(\mathbf{x}) + \alpha_{2}g_{2}(\mathbf{x})) = \alpha_{1}g_{1}(\mathbf{F}_{r}(\mathbf{x})) + \alpha_{2}g_{2}(\mathbf{F}_{r}(\mathbf{x}))$$
 (7.54a)

$$= \alpha_1 \mathcal{K}_{t} g_1(\mathbf{x}) + \alpha_2 \mathcal{K}_{t} g_2(\mathbf{x}).$$
 (7.54b)

For sufficiently smooth dynamical systems, it is also possible to define the continuoustime analogue of the Koopman dynamical system in (7.53):

$$\frac{d}{dt}g = Kg.$$
 (7.55)

The operator K is the infinitesimal generator of the one-parameter family of transformations K_r [1]. It is defined by its action on an observable function r:

$$K_g = \lim_{l \to 0} \frac{K_{lg} - g}{l} = \lim_{l \to 0} \frac{g \circ \mathbf{F}_l - g}{l}. \quad (7.56)$$

The linear dynamical systems in (7.53) and (7.53) are analogous to the dynamical systems in (7.3) and (7.4), respectively. It is important to note that the original state stray be the observable, and the infinite-dimensional operator K_y will advance this function. However, the simple representation of the observable g = x in a chose basis for Hilbert space may become arbitrarily complex once iterated through the dynamics. In other words, finding a representation for K may not be simple or straightforward.

Koopman Eigenfunctions and Intrinsic Coordinates

The Koopman operator is linear, which is appealing, but is infinite dimensional, postigues for representation and comparation. Instead of appring the evolution of all measurement functions in a Hilbert space, applied Koopman analysis attempts to identify legs of the Koopman experime priority in gate that is set of special instancements that behave linearly in time. In fact, a primary motivation to adopt the Koopman framework is the dynamics introduced to equivalent to the eigen-Accomposition of the parameters of the parameters

A discrete-time Koopman eigenfunction $\varphi(\mathbf{x})$ corresponding to eigenvalue λ satisfies

$$\varphi(\mathbf{x}_{k+1}) = \mathcal{K}_{\Delta t}\varphi(\mathbf{x}_k) = \lambda\varphi(\mathbf{x}_k).$$
 (7.57)

In continuous-time, a Koopman eigenfunction $\varphi(\mathbf{x})$ satisfies

$$\frac{d}{dt}\varphi(\mathbf{x}) = \mathcal{K}\varphi(\mathbf{x}) = \lambda\varphi(\mathbf{x}). \quad (7.58)$$

Obtaining Koopman eigenfunctions from data or from analytic expressions is a central applied challenge in modern dynamical systems. Discovering these eigenfunctions enables abbally linear representations of strongly nonlinear systems.

Applying the chain rule to the time derivative of the Koopman eigenfunction $\phi(x)$ yields

$$\frac{d}{dt}\varphi(\mathbf{x}) = \nabla \varphi(\mathbf{x}) \cdot \dot{\mathbf{x}} = \nabla \varphi(\mathbf{x}) \cdot \mathbf{f}(\mathbf{x}). \quad (7.59)$$

Combined with (7.58), this results in a partial differential equation (PDE) for the eigenfunction $\phi(x)$:

$$\nabla \phi(\mathbf{x}) \cdot \mathbf{f}(\mathbf{x}) = \lambda \phi(\mathbf{x}).$$
 (7.60)

With this nonlinear PDE, it is possible to approximate the eigenfunctions, either by solving for the Laurent series or with data via regression, both of which are explored below. This formulation assumes that the dynamics are both continuous and differentiable. The discrete-time dynamics in (7.4) are more general, abhough in many examples the continuous-time dynamics have a simpler representation than the discrete-time map for long times. For example, the simple Lorenz system has a simple continuous-time representation, yet is generally unrepresentable for even moderately long discrete-time updates.

The key takensay from (7.5) and (7.8) is that the nonlinear dynamics become completely linear in eigenfunction coordinance, given by qich. As an aimple cample, any coserved quartity of a dynamical system is a Koopman eigenfunction corresponding to eigenvature $\lambda = 0$. This excitables as Koopman excitation of the finants Notech's theorem [400], implying that any symmetry in the governing equations given: rise to a new Koopman eigenfunction with eigenvalue $\lambda = 0$. The complete the lamilsonic merger function is a Koopman eigenfunction for a concervative system. In addition, the constant function $\varphi = 1$ is durant as this eigenfunction corresponding to $\lambda = 0$ for even dynamical system.

Eigenvalue lattices Interestingly, a set of Koopman eigenfunctions may be used to generate more eigenfunctions. In discrete time, we find that the product of two eigenfunctions w(x) and w(x) is also an eigenfunction

$$K_t(\phi_1(\mathbf{x})\phi_2(\mathbf{x})) = \phi_1(\mathbf{F}_t(\mathbf{x}))\phi_2(\mathbf{F}_t(\mathbf{x}))$$
 (7.61a)

$$=\lambda_1\lambda_2\varphi_1(\mathbf{x})\varphi_2(\mathbf{x})$$
 (7.61b)

corresponding to a new eigenvalue $\lambda_1 \lambda_2$ given by the product of the two eigenvalues of $w_1(x)$ and $w_2(x)$.

In continuous time, the relationship becomes:

$$\mathcal{K}(\varphi_1\varphi_2) = \frac{d}{dt}(\varphi_1\varphi_2) \quad (7.62a)$$

$$= \phi_1 \phi_2 + \phi_1 \phi_2$$
 (7.62b)

$$= \lambda_1 \varphi_1 \varphi_2 + \lambda_2 \varphi_1 \varphi_2 \qquad (7.62c)$$

$$= (\lambda_1 + \lambda_2) \varphi_1 \varphi_2.$$
 (7.62d)

Interestingly, this means that the set of Koopman eigenfunctions establishes a commute the roomsid and the points where multiplications are model has the structure of a pope, except that the elements need to have inverses. Thus, depending on the dynamical system, there multiplications are determined in the end to construct all other eigenfunctions. The corresponding eigenvalues simulative from a lattice, based on the point +1, +2, appending whether the dynamics in disactive and in the product +1, 2 on +1, +2, appending whether the dynamics in the corresponding eigenvalues in the dynamics in the dynamics in the dynamics of the dynamics of the dynamics in the dynamics of the dynamics in the dynamics of t

The continuous time and discrete time lattices are related in a simple way. If the continuous-time eigenvalues are given by λ , then the corresponding discrete-time eigenvalues are given by $e^{i\lambda}$. Thus, the eigenvalue expressions in (7.61b) and (7.62d) are related as:

$$e^{\lambda_{1}t}e^{\lambda_{2}t}\psi_{1}(\mathbf{x})\psi_{2}(\mathbf{x}) = e^{(\lambda_{1}+\lambda_{2})t}\psi_{1}(\mathbf{x})\psi_{2}(\mathbf{x}).$$
 (7.63)

As another simple demonstration of the relationship between continuous-time and discrete-time eigenvalues, consider the continuous-time definition in (7.56) applied to an eigenfunction:

$$\lim_{t\to 0} \frac{\mathcal{K}_t \varphi(\mathbf{x}) - \varphi(\mathbf{x})}{I} = \lim_{t\to 0} \frac{e^{\lambda I} \varphi(\mathbf{x}) - \varphi(\mathbf{x})}{I} = \lambda \varphi(\mathbf{x}). \quad (7.64)$$

Koopman Mode Decomposition and Finite Representations

Until now, we have considered scalar measurements of a system, and we uncovered special eigen-measurements that evolve linearly in time. However, we often take multiple measurements of a system. In extreme cases, we may measure the entire state of a high-dimensional spatial system, such as an evolving fluid flow. These measurements may then be arranged in a vector gr.

$$g(\mathbf{x}) = \begin{bmatrix} g_1(\mathbf{x}) \\ g_2(\mathbf{x}) \\ \vdots \\ g_p(\mathbf{x}) \end{bmatrix}. \quad (7.65)$$

Each of the individual measurements may be expanded in terms of the eigenfunctions $\phi_J(\mathbf{x})$, which provide a basis for Hilbert space:

$$g_i(\mathbf{x}) = \sum_{j=1}^{\infty} v_{ij} \varphi_j(\mathbf{x}).$$
 (7.66)

Thus, the vector of observables, g, may be similarly expanded:

$$\mathbf{g}(\mathbf{x}) = \begin{bmatrix} g_1(\mathbf{x}) \\ g_2(\mathbf{x}) \\ \vdots \\ g_p(\mathbf{x}) \end{bmatrix} = \sum_{j=1}^{\infty} \varphi_j(\mathbf{x}) \mathbf{v}_j, \quad (7.67)$$

where v_i is the *j*-th Koopman mode associated with the eigenfunction φ_i .

For conservative dynamical systems, such as those governed by Hamiltonian dynamics, the Koopman operator is unitary. Thus, the Koopman eigenfunctions are orthonormal for conservative systems, and it is possible to compute the Koopman modes v_j directly by projection:

$$\mathbf{v}_{j} = \begin{bmatrix} \langle \varphi_{j}, g_{1} \rangle \\ \langle \varphi_{j}, g_{2} \rangle \\ \vdots \\ \langle \varphi_{j}, g_{p} \rangle \end{bmatrix}, \quad (7.68)$$

where $\langle \cdot, \cdot \rangle$ is the standard inner product of functions in Hilbert space. These modes have a physical interpretation in the case of direct spatial measurements of a system, g(x) = x, in which case the modes are coherent spatial modes that behave linearly with the same temporal dynamics (i.e., oscillations, possible with linear growth or decay).

Given the decomposition in (7.67), it is possible to represent the dynamics of the measurements g as follows:

$$\mathbf{g}(\mathbf{x}_k) = \mathcal{K}_{\Delta i}^k \mathbf{g}(\mathbf{x}_0) = \mathcal{K}_{\Delta i}^k \sum_{j=0}^{\infty} \varphi_j(\mathbf{x}_0) \mathbf{v}_j$$
 (7.69a)

$$= \sum_{j=0}^{\infty} \mathcal{K}_{\Delta i}^{k} \varphi_{j}(\mathbf{x}_{0}) \mathbf{v}_{j} \qquad (7.69b)$$



Figure 7.10 Schematic illustrating the Koopman operator for nonlinear dynamical systems. The dashed lines from $v_1 \rightarrow x_0$ indicate that we would like to be able to recover the original state.

$$= \sum_{j=0}^{\infty} \lambda_j^k \varphi_j(\mathbf{x}_0) \mathbf{v}_j. \qquad (7.69c)$$

This sequence of triples, $[(\lambda_j, \psi_j, \tau_j)]_{j,0}^{\infty}$ is known as the Koopman mode decomposition, and was introduced by Mexic in 2005 [376]. The Koopman mode decomposition was later connected to data-driven regression via the dynamic mode decomposition [456], which will be discussed in Section 7.2.

Invariant Eigenspaces and Finite-Dimensional Models

Instead of capturing the evolution of all measurement functions in a Hilbert space, applied Koopman analysis approximates the evolution on an invariant subspace spanned by a finite set of measurement functions.

A Koopman-invariant subspace is defined as the span of a set of functions $\{g_1, g_2, \dots, g_p\}$ if all functions g in this subspace

$$g = \alpha_1 g_1 + \alpha_2 g_2 + \cdots + \alpha_p g_p$$
 (7.70)

remain in this subspace after being acted on by the Koopman operator K:

$$K_g = \beta_1 g_1 + \beta_2 g_2 + \cdots + \beta_p g_p.$$
 (7.71)

It is possible to obtain a finite-dimensional matrix representation of the Koopman operator by restricting it to an invariant subspace spanned by a finite number of functions $[g_j]_{j=0}^{p}$. The matrix representation K acts on a vector space \mathbb{R}^p , with the coordinates given by the values of $g_i(\mathbf{x})$. This induces a finite-dimensional linear system, as in (7.53) and (7.55).

Any finite set of eigenfunctions of the Koopman operator will span an invariant subspace. Discovering these eigenfunction coordinates is, therefore, a central challenge, as they provide intrinsic coordinates along which the dynamics behave linearly. In practice, it is more likely that we will identify an approximately invariant subspace, given by a set of functions $\{g_j\}_{j=0}^{p}$, where each of the functions g_j is well approximated by a finite sum of eigenfunctions: $g_j \approx \sum_{j=0}^{p} a_j q_j q_j$.

Examples of Koopman Embeddings Nonlinear System with Single Fixed Point and a Slow Manifold

Here, we consider an example system with a single fixed point, given by:

$$\dot{x}_1 = \mu x_1$$
 (7.72a)

$$\dot{x}_2 = \lambda (x_2 - x_1^2).$$
 (7.72b)

For $\lambda < \mu < 0$, the system exhibits a slow attracting manifold given by $x_2 = x_1^2$. It is possible to augment the state **x** with the nonlinear measurement $g = x_1^2$, to define a three-dimensional Koopman invariant subspace. In these coordinates, the dynamics become linear:

$$\frac{d}{dt} \begin{bmatrix} y_1 \\ y_2 \\ y_3 \end{bmatrix} = \begin{bmatrix} \mu & 0 & 0 \\ 0 & \lambda & -\lambda \\ 0 & 0 & 2\mu \end{bmatrix} \begin{bmatrix} y_1 \\ y_2 \\ y_3 \end{bmatrix} \text{ for } \begin{bmatrix} y_1 \\ y_2 \\ y_3 \end{bmatrix} = \begin{bmatrix} x_1 \\ x_2 \\ x_1^2 \end{bmatrix}. \quad (7.73a)$$

The full first-dimensional Loopenn belowskit revert space is unimited in Fig. 211. The full first-dimension is an array moder of the structure space is unimited in Fig. 211. Second with the interaction of the structure space is unimited. The structure space is unimited by the structure space is the structure space is universe space is the structure space is the st

Intrinsic coordinates defined by eigenfunctions of the Koopman operator The left eigenvectors of the Koopman operator yield Koopman eigenfunctions (i.e., eigenobservables). The Koopman eigenfunctions of (7.73a) corresponding to eigenvalues µ and λ are:

$$\varphi_{\mu} = x_1$$
, and $\varphi_{\lambda} = x_2 - bx_1^2$ with $b = \frac{\lambda}{\lambda - 2\mu}$. (7.74)

The constant b in φ_{λ} captures the fact that for a finite ratio λ/μ , the dynamics only shadow the asymptotically attracting slow manifold $x_{2} = x_{1}^{2}$, but in fact follow neighboring parabolic trajectories. This is illustrated more clearly by the various surfaces in Fig. 7.11 for different ratios λ/μ .

In this way, a set of intrinsic coordinates may be determined from the observable functions defined by the left eigenvectors of the Koopman operator on an invariant subspace. Explicitly,

$$\phi_0(\mathbf{x}) = \xi_- \mathbf{y}(\mathbf{x}), \text{ where } \xi_- \mathbf{K} = \alpha \xi_0.$$
 (7.75)



Figure 211 Visualization of three-dimensional linear Koopman system from (7.73a) along with projection of dynamics onto the x_1 - x_2 plane. The attracting slow manifold is shown in red, the constraint $y_1 = x_1^2$ is shown in blue, and the slow unstable subspace of (7.73a) is shown in green. Black trajectories of the linear Koopman system in y project onto trajectories of the full nonlinear system in x in the y_1 - y_2 plane. Here, $\mu_0 = -0.05$ and $\lambda = 1$. Reproduced from Bravter et al. [92].

These eigen-observables define observable subspaces that remain invariant under the Koopman operator, even after coordinate transformations. As such, they may be regarded as intrinsic coordinates (5556) on the Koopman-invariant subspace.

Example of Intractable Representation

Consider the logistic map, given by:

$$x_{k+1} = \beta x_k (1 - x_k).$$
 (7.76)

Let our observable subspace include x and x2:

$$\mathbf{y}_{k} = \begin{bmatrix} x \\ x^{2} \end{bmatrix}_{k} \triangleq \begin{bmatrix} x_{k} \\ x_{k}^{2} \end{bmatrix}$$
 (7.77)

Writing out the Koopman operator, the first row equation is simple:

$$\mathbf{y}_{k+1} = \begin{bmatrix} x \\ x^2 \end{bmatrix}_{k+1} = \begin{bmatrix} \beta & -\beta \\ \gamma & \gamma \end{bmatrix} \begin{bmatrix} x \\ x^2 \end{bmatrix}_k,$$
 (7.78)

but the second row is not obvious. To find this expression, expand $x_{2,1}^2$

$$x_{k+1}^2 = (\beta x_k (1 - x_k))^2 = \beta^2 (x_k^2 - 2x_k^3 + x_k^4).$$
 (7.79)

Thus, cubic an	d quartic polynomial	terms are required t	to advance x*. Sin	nilarly, these terms
need polynomi	als up to sixth and ei	ghth order, respect	ively, and so on, a	d infinitum:

		х	x ²	x3	- X ⁴	x ⁵	x ⁶	x7	x ⁸	x9	x ¹⁰			
$\lceil x \rceil$		Гв	$-\beta$	0	0	0	0	0	0	0	0]	[x]	
x2		6	β ²	$-2\beta^2$	r2	0	0	0	0	0	0		x2	
x^3		0	0	β ³	$-3\beta^3$	$3\beta^3$	β^3	0	0	0	0		x3	
x^4	-	0	0	0	β4	$-4\beta^4$	684	$-4\beta^4$	β^4	0	0		x^4	
x5		0	0	0	0	B ⁵	$-5\beta^5$	1085	$-10\beta^{5}$	5 ß ⁵	$-\beta^5$		x5	
		. I											·	
1.2.1		12.										- 24	131	

It is interesting to note that the rows of this equation are related to the rows of Pascal's triangle, with the n-th row scaled by rⁿ, and with the omission of the first row:

$$[x^0]_{k+1} = [0] [x^0]_k$$
. (7.80)

The above representation of the Koopman operator in a polynomial basis is somewhat troubling. Not only there no colourse, how the determinant of any finite-mark transmission is very large for $\beta > 1$. This libraries a patiful associated with naive representation of the initiate dimensional Koopman operator for on a simple chaotic system. Truncating the system, or performing a least squares fit on an anymented observable vector (i.e. DMD on a nulmater massement, see Section 7.5) yields poor results, which the truncad systems only appending with the true dynamics for a small handful of iterations, as the complexity of the representation groups optically.

Analytic Series Expansions for Eigenfunctions

Given the dynamics in (7.1), it is possible to solve the PDE in (7.60) using standard techniques, such as recursively solving for the terms in a Taylor or Laurent series. A number of simple examples are exolored below.

Linear Dynamics

Consider the simple linear dynamics

$$\frac{d}{dt}x = x.$$
 (7.82)

Assuming a Taylor series expansion for $\varphi(x)$:

$$g(x) = c_0 + c_1x + c_2x^2 + c_3x^3 + \cdots$$

then the gradient and directional derivatives are given by:

$$\nabla \varphi = c_1 + 2c_2x + 3c_3x^2 + 4c_4x^3 + \cdots$$

 $\nabla \varphi \cdot f = c_1x + 2c_2x^2 + 3c_3x^3 + 4c_4x^4 + \cdots$

Solving for terms in the Koopman eigenfunction PDE (7.60), we see that $c_0 = 0$ must hold. For any positive integer λ in (7.60), only one of the coefficients may be nonzero. Specifically, for $\lambda = k \in \mathbb{Z}^+$, then $\varphi(x) = cx^k$ is an eigenfunction for any constant c. For instance, if $\lambda = 1$, then $\varphi(x) = x$.

Quadratic Nonlinear Dynamics

Consider a nonlinear dynamical system

$$\frac{d}{dt} = x^2$$
. (7.83)

There is no Taylor series that satisfies (7.60), except the trivial solution $\phi = 0$ for $\lambda = 0$. Instead, we assume a Laurent series:

$$\varphi(x) = \cdots + c_{-3}x^{-3} + c_{-2}x^{-2} + c_{-1}x^{-1} + c_0$$

+ $c_1x + c_2x^2 + c_3x^3 + \cdots$.

The gradient and directional derivatives are given by:

$$\nabla \varphi = \cdots - 3c_{-3}x^{-4} - 2c_{-2}x^{-3} - c_{-1}x^{-2} + c_1 + 2c_2x$$

+ $3c_3x^2 + 4c_4x^3 + \cdots$
 $\nabla \varphi \cdot f = \cdots - 3c_{-3}x^{-2} - 2c_{-2}x^{-1} - c_{-1} + c_1x^2 + 2c_2x^3$
+ $3c_3x^4 + 4c_4x^5 + \cdots$.

Solving for the coefficients of the Laurent series that satisfy (7.60), we find that all coefficients with positive index are zero, i.e. $c_k = 0$ for all $k \ge 1$. However, the nonpositive index coefficients are given by the recursion $\lambda c_{k+1} = kc_k$, for negative $k \le -1$. Thus, the Laurent series is

$$\varphi(x) = c_0 \left(1 - \lambda x^{-1} + \frac{\lambda^2}{2}x^{-2} - \frac{\lambda^3}{3}x^{-3} + \cdots\right) = c_0 e^{-\lambda/x}.$$

This holds for all values of $\lambda \in \mathbb{C}$. There are also other Koopman eigenfunctions that can be identified from the Laurent series.

Polynomial Nonlinear Dynamics

For a more general nonlinear dynamical system

$$\frac{d}{dt} = at^{0}$$
, (7.84)

 $\varphi(x) = e^{\frac{1}{(1-\alpha)\mu}x^{1-\alpha}}$ is an eigenfunction for all $\lambda \in \mathbb{C}$.
As mentioned above, it is also possible to generate new eigenfunctions by taking powers of these primitive eigenfunctions; the resulting eigenvalues generate a *lattice* in the complex plane.

History and Recent Developments

The original analysis of Eucopeans in 1911 was introduced to describe the ordinators in the environment of Human Interport [500], and then been yan appendicularly Koopean environment of Human Interport (1900), and then been yan appendicular (1900) care of Human Interport (1900), and the singular value decomposition (2001) was an able decores Fourier transformation (1907) and the singular value decomposition (2001) was an able decores Fourier transformation (1907) and the singular value decomposition (2001) was associated and the singular value decomposition (2001) which is interported and the singular value decomposition (2001) within the singular polytopol

Recently, it has been shown that the operator theoretic framework complements the traditional geometry and probabilities preservences. For example, level sets of Koopaan eigenfunctions form invariant partitions of the state-space of a dynamical system [103] in particular, eigenfunctions of the Koopaan operator may be used to analyze the ergodic partition [130]. TOIS [102] Koopaan analysis has also hear recently shown to generatize the time sostion ergodica could 1322.

At the time of this writing, representing Koopman eigenfunctions for general dynamical systems remains a central unsolved et allenges. Significant research efforts are focused on developing data-driven techniques to identify Koopman eigenfunctions and use these for control, which will be discussed in the following sections and chapters. Recently, new work has emerged that attempts to leverage the power of deep learning to discover and represent circulmations of the data S03, 864, 813, 864, 812, 901.

7.5 Data-Driven Koopman Analysis

Obtaining linear representations for strongly nonlinear systems has the potential to revolutionize our ability to predict and control these systems. The linearization of dynamics near In facto piants operiodics of this has long term complexed for low of harm representation of the imposing [52]. The Kongen on sprarts is appeared however is provide as a phofed lutear representation, via data array from fatta phonis and periodic of the Kongena representation of the start and the start is a provide the start of the start

Extended DMD

The extended DMD algorithm [556] is essentially the same as stundard DMD [535], except that instead of performing regression on direct measurements of the state, regression is performed on an augmented vector containing nonlinear measurements of the state. As discussed earlier, eDMD is equivalent to the variational approach of conformation dynamics [405, 407, 408], which was developed in 2013 by Nocl and Niske.

Here, we will modify the notation slightly to conform to related methods. In eDMD, an augmented state is constructed:

$$\mathbf{y} = \Theta^T (\mathbf{x}) = \begin{bmatrix} \theta_1(\mathbf{x}) \\ \theta_2(\mathbf{x}) \\ \vdots \\ \theta_n(\mathbf{x}) \end{bmatrix}$$
. (7.85)

 Θ may contain the original state x as well as nonlinear measurements, so often $p \gg n$. Next, two data matrices are constructed, as in DMD:

$$\mathbf{Y} = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{y}_1 & \mathbf{y}_2 & \cdots & \mathbf{y}_n \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}, \qquad \mathbf{Y}' = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{y}_2 & \mathbf{y}_3 & \cdots & \mathbf{y}_{n+1} \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}. \quad (7.86a)$$

Finally, a best-fit linear operator Ay is constructed that maps Y into Y':

$$\mathbf{A}_{\mathbf{Y}} = \underset{\mathbf{A}_{\mathbf{Y}}}{\operatorname{argmin}} \|\mathbf{Y} - \mathbf{A}_{\mathbf{Y}}\mathbf{Y}\| = \mathbf{Y}'\mathbf{Y}^{\dagger}.$$
 (7.87)

This regression may be written in terms of the data matrices $\Theta(X)$ and $\Theta(X')$:

$$\mathbf{A}_{\mathbf{Y}} = \underset{\mathbf{A}_{\mathbf{Y}}}{\operatorname{argmin}} \| \mathbf{\Theta}^{T}(\mathbf{X}') - \mathbf{A}_{\mathbf{Y}} \mathbf{\Theta}^{T}(\mathbf{X}) \| = \mathbf{\Theta}^{T}(\mathbf{X}') \left(\mathbf{\Theta}^{T}(\mathbf{X}) \right)^{\top}.$$
 (7.88)

Because the augmented vectory may be significantly larger than the state x, kernel methods are often employed to compare this regression [377]. In principle, the emitfed Bhary Θ provides a larger basis in which to approximate the Koopman operator. It has been shown recently that in the limit of infinite samphoses, the extensed DMD operator converges to the Koopman operator projected onto the subspace, spanned by Θ [303]. However, if Θ does not span a Koopman invirtant subspace, then the projected operator may not have any more than the subspace operator projection of the subspace operator by the subspace operator project operator may not have any recentilance to the original Koopmun operator, as all of the eigenvalues and eigenvectors may be different. In fact, it was abown that the extended DIMD operators with thre septimoeigenvalues and eigenvectors unless it is represented in terms of a Koopmun invariant subspace [92]. Therefore, it is essential to use vuldation and cross-scatiation techniques or the dDMD models are not orefit, as discussed below. For example, it was shown that dDMD models are not orefit, as discussed below. For example, it was shown that dDMD randoc data and the set is an ensumement and represent a system cannot be topologically conjugate to a final dimensional linear yearem [92].

Approximating Koopman Eigenfunctions from Data

In discrete-time, a Koopman eigenfunction $\varphi(\mathbf{x})$ evaluated at a number of data points in X will satisfy:

$$\begin{bmatrix} \lambda \varphi(\mathbf{x}_{2}) \\ \lambda \varphi(\mathbf{x}_{2}) \\ \vdots \\ \lambda \varphi(\mathbf{x}_{m}) \end{bmatrix} = \begin{bmatrix} \varphi(\mathbf{x}_{2}) \\ \varphi(\mathbf{x}_{1}) \\ \vdots \\ \varphi(\mathbf{x}_{m+1}) \end{bmatrix}.$$
(7.39)

It is possible to approximate this eigenfunction as an expansion in terms of a set of candidate functions.

$$\Theta(\mathbf{x}) = \begin{bmatrix} \theta_1(\mathbf{x}) & \theta_2(\mathbf{x}) & \cdots & \theta_n(\mathbf{x}) \end{bmatrix}$$
(7.90)

The Koopman eigenfunctionmay be approximated in this basis as:

$$\varphi(\mathbf{x}) \approx \sum_{k=1}^{p} \theta_{k}(\mathbf{x})\xi_{k} = \Theta(\mathbf{x})\xi.$$
 (7.91)

Writing (7.89) in terms of this expansion yields the matrix system:

$$(\lambda \Theta(X) - \Theta(X')) \xi = 0.$$
 (7.92)

If we seek the best least-squares fit to (7.92), this reduces to the extended DMD [557, 556] formulation:

$$\lambda \xi = \Theta(\mathbf{X})^T \Theta(\mathbf{X}') \xi.$$
 (7.93)

Note that (7.93) is the transpose of (7.38), so that left eigenvectors become right eigenvectors. Time, eigenvectors ξ_i of Φ^0 yield the coefficients of the eigenvalue for a represented in the basis $\Theta(x_i)$. It is absolutely essential to then confirm that prediced eigenfunctions actually behave linearly on uniperiories. By comparing them with the prediced dynamics $y_{i+1} = \lambda_{ip}$, because the regression above will result in spurious eigenvalues and eigenvectors unless the basis cleanest θ , span a Rooman invariant subpace [92].

Sparse Identification of Eigenfunctions

It is possible to leverage the SINDy regression [95] to identify Koopman eigenfunctions corresponding to a particular eigenvalue λ_s , selecting only the few active terms in the library $\Theta(\mathbf{x})$ to avoid overfitting. Given the data matrices, \mathbf{X} and $\hat{\mathbf{X}}$ from above it is possible to construct the library of basis functions $\Theta(\mathbf{X})$ as well as a library of directional derivatives. representing the possible terms in $\nabla \phi(\mathbf{x}) \cdot \mathbf{f}(\mathbf{x})$ from (7.60):

$$\Gamma(\mathbf{x}, \dot{\mathbf{x}}) = \left[\nabla \theta_1(\mathbf{x}) \cdot \dot{\mathbf{x}} \quad \nabla \theta_2(\mathbf{x}) \cdot \dot{\mathbf{x}} \quad \cdots \quad \nabla \theta_p(\mathbf{x}) \cdot \dot{\mathbf{x}}\right].$$
 (7.94)

It is then possible to construct Γ from data:

$$\Gamma(X, \hat{X}) = \begin{bmatrix} \nabla \theta_1(x_1) \cdot \dot{x}_1 & \nabla \theta_2(x_1) \cdot \dot{x}_1 & \cdots & \nabla \theta_p(x_1) \cdot \dot{x}_1 \\ \nabla \theta_1(x_2) \cdot \dot{x}_2 & \nabla \theta_2(x_2) \cdot \dot{x}_2 & \cdots & \nabla \theta_p(x_3) \cdot \dot{x}_2 \\ \vdots & \vdots & \vdots \\ \nabla \theta_1(x_n) \cdot \dot{x}_n & \nabla \theta_2(x_n) \cdot \dot{x}_n & \cdots & \nabla \theta_p(x_n) \cdot \dot{x}_n \end{bmatrix},$$

For a given eigenvalue λ , the Koopman PDE in (7.60) may be evaluated on data:

$$(\lambda \Theta(\mathbf{X}) - \Gamma(\mathbf{X}, \dot{\mathbf{X}})) \mathbf{i} = 0.$$
 (7.95)

The formulations in (759) is implicit, so that ξ will be in the analyses of 10.051. Next, The T C S were transformed by first gluon probability of the signature cosmotion (2000). The transformed cosmological structure is the signature cosmological structure is signature cosmological structure is signature cosmological structure is a signature cosmological structure is signature cosmological structure is signature cosmological structure is signature cosmological structure cosmological struc

From a practical standpoint, data in X does not need to be sampled from full trajectories, but can be obtained using more sophisticated strategies such as latin hypercube sampling or sampling from a distribution over the phase space. Moreover, reproducing kernel Hilbert spaces (RKHY) can be enableded to describe of Vi locally in mothes of state space.

Example: Duffing System (Kaiser et al [276])

We demonstrate the sparse identification of Koopman eigenfunctions on the undamped Duffing oscillator:

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} x_2 \\ x_1 - x_1^3 \end{bmatrix}$$

where x_1 is the position and x_2 is the velocity of a particle in a double well potential with equilibria (0, 4) and (±1, 0). This system is conservative, with Hamiltonian $H = \frac{1}{2}x_2^2 - \frac{1}{2}x_1^2 + \frac{1}{2}x_1^2$. The Hamiltonian, and in general any conserved quantity, is a Koopman eigenfunction with zero eigenvalue.

For the eigenvalue $\lambda = 0$, (7.95) becomes $-\Gamma(\mathbf{X}, \dot{\mathbf{X}}) \mathbf{\xi} = 0$, and hence a sparse $\mathbf{\xi}$ is sought in the null-space of $-\Gamma(\mathbf{X}, \dot{\mathbf{X}})$. A library of candidate functions is constructed from data, employing polynomials up to fourth order:

$$\Theta(\mathbf{X}) = \begin{bmatrix} | & | & | & | \\ x_1(t) & x_2(t) & x_1^2(t) & x_1(t)x_2(t) & \cdots & x_2^4(t) \\ | & | & | & | & | \\ \end{vmatrix}$$

$$\Gamma(\mathbf{X}, \mathbf{\hat{X}}) = \begin{bmatrix} | & | & | & | \\ \dot{x}_1(t) & \dot{x}_2(t) & 2x_1(t)\dot{x}_1(t) & x_2(t)\dot{x}_1(t) + \dot{x}_1(t) + \dot{x}_2(t) & \cdots & 4x_2(t)^3 \dot{x}_2(t) \\ | & | & | & | & | \\ \end{bmatrix}$$

A sparse vector of coefficients ξ may be identified, with the few nonzero entries determining the active terms in the Koopman eigenfunction. The identified Koopman eigenfunctionassociated with $\lambda = 0$ is

$$\varphi(\mathbf{x}) = -2/3x_1^2 + 2/3x_2^2 + 1/3x_1^4. \quad (7.96)$$

This eigenfunction matches the Hamiltonian perfectly up to a constant scaling.

Data-Driven Koopman and Delay Coordinates

The time-delay measurement scheme is shown schematically in Fig. 7.12, as illustrated on the Lorenz system for a single time-scrise measurement of the first variable, *x*(1). The conditions of the Takens embedding theorem are satisfied [515], so it is possible to obtain a diffeomophium between a delay neubodied attractor and the attractor in the original coordinates. We then obtain eigen-time-delay coordinates from a time-series of a single measurement *x*(1) by using the SYO to the Hankel mutrix **B**:

$$\mathbf{H} = \begin{bmatrix} x(t_1) & x(t_2) & \cdots & x(t_{m_i}) \\ x(t_2) & x(t_3) & \cdots & x(t_{m_i+1}) \\ \vdots & \vdots & \ddots & \vdots \\ x(t_{m_i}) & x(t_{m_i+1}) & \cdots & x(t_m) \end{bmatrix} = \mathbf{U} \mathbf{\Sigma} \mathbf{V}^*. \quad (7.97)$$

The columns of U and V from the SVD are arranged hierarchically by their shifty to model the columns and rows of H, respectively. Often, H may admit a low-rank approximation by the first *z*-columns of U and V. Note that the Hankel matrix in (7.57) is the basis of the eigensystem realization algorithm [272] in linear system identification (see Section 9.3) and simular spectrum analysis (SSA11881 in climme time-teries analysis).

The low-rank approximation to (7.97) provides a data-driven measurement system that is approximately invariant to the Koopman operator for states on the attractor. By definition, the dynamics map the attractor into itself, making it invariant to the flow. In other words, the columns of U form a Koopman invariant subspace. We may re-write (7.97) with the

and



Figure 3.22 Decomposition of chaos into a literar system with forcing. A time series x(t) is stacked into a Hankel more N. H. SeVD of H. Jiski da harmedy of relative time series that produce a delay-senhedded attractor. A borel di literar regression model is obtained on the delay coordinates y_i is an input for the distribution of the dist

Koopman operator $\mathcal{K} \triangleq \mathcal{K}_{Av}$:

$$\mathbf{H} = \begin{bmatrix} x(t_1) & \mathcal{K}^{x}(t_1) & \cdots & \mathcal{K}^{m_n-1}x(t_1) \\ \mathcal{K}x(t_1) & \mathcal{K}^{-2}x(t_1) & \cdots & \mathcal{K}^{m_n}x(t_1) \\ \vdots & \vdots & \ddots & \vdots \\ \mathcal{K}^{m_n-1}x(t_1) & \mathcal{K}^{m_n}x(t_1) & \cdots & \mathcal{K}^{m-1}x(t_1) \end{bmatrix}.$$
 (7.98)

The columns of (7.97) are well-approximated by the first r columns of U. The first r columns of V provide a time series of the magnitude of each of the columns of U Σ in the data. By plotting the first three columns of V, we obtain an embedded attractor for the Lorenz system (See Fig. 7.12).

The connection between eigen-time-delay coordinates from (7.97) and the Koopnan operater motivates a linear regression model on the variable is tr. Urse with an approximately Koopnan-invariant measurement system, there remain challenges to identifyinga linear model for a choice system. A linear model, however duelled, cannot capturemultiple fixed points or the supredictable behavior characteristic of chois with a positive $Lyapmov exponent [27]. Instead of constructing a closed linear model for the fitty <math>\tau$ variables in V_{ν} we build a linear model on the first r - 1 variables and recuss the last variable, ν_{ν} as in forcing term:

$$\frac{d}{dt}\mathbf{v}(t) = \mathbf{A}\mathbf{v}(t) + \mathbf{B}v_r(t), \quad (7.99)$$

where $\mathbf{v} = \begin{bmatrix} v_1 & v_2 & \cdots & v_{r-1} \end{bmatrix}^T$ is a vector of the first r - 1 eigen-time-delay coordinates. Other work has investigated the splitting of dynamics into deterministic linear, and chaotic stochastic dynamics [376].

In all of the example ceptoet all [91], the linear model on the first r - 1 terms is contrast, while to hister model represents r_1 , the start planet from r_2 the dynamics in (75%) which approximates the nonlinear dynamics. The startings of r_1 planet dynamics is related to the start planet dynamics of the start planet dynamics is the locarest system; this related to an event dynamics of the locare systems of the start dynamics and the start planet dynamics of the locare systems (16%)

In addition, the use of delay coordinates as intrinsis measurements for Koopman analysis asgeness that Koopman theory may also be used to improve spatially distributed sensor technologies. A spatial array of sensors, for example the C/1000 strain sensors on the wings of physig insects, may use place delay coordinates to provide nearly optimal embeddings to detect and control convective structures (e.g., stall from a gust, leading edge vortex formation and convection, etc.).

HAVOK Code for Lorenz System

Below is the code to generate a HAVOK model for the same Lorenz system data generated in Code 7.2. Here we use $\Delta t = 0.01$, $m_o = 10$, and r = 10, although the results would be more accurate for $\Delta t = 0.001$, $m_o = 100$, and r = 15.

Code 7.3 HAVOK code for Lorenz data renerated in Section 7.1.

```
\label{eq:constraints} \begin{split} & is states - rate (interval of the state o
```

Neural Networks for Koopman Embeddings

Despite the promise of Koopman embeddings, obtaining tractable representations has remained a central challenge. Recall that even for relatively simple dynamical systems, the eigenfunctions of the Koopman operator may be arbitrarily complex. Deep learning, which is well-suited for representing arbitrary functions, has recently emerged as a promising approach for discovering and representing Koopman eigenfunctions [550, 368, 513, 564, 412, 332, 349], providing a data-driven embedding of strongly nonlinear systems into intrinsic linear coordinates. In particular, the Koopman perspective fits naturally with the deep auto-encoder structure discussed in Chanter 6, where a few key latent variables $\mathbf{v} = \boldsymbol{\varphi}(\mathbf{x})$ are discovered to parameterize the dynamics. In a Koopman network, an additional constraint is enforced so that the dynamics must be linear on these latent variables, forcing the functions $\varphi(\mathbf{x})$ to be Koopman eigenfunctions, as illustrated in Fig. 7.13. The constraint of linear dynamics is enforced by the loss function $\|\varphi(\mathbf{x}_{l+1}) - \mathbf{K}\varphi(\mathbf{x}_{l})\|$, where K is a matrix. In general, linearity is enforced over multiple time steps, so that a trajectory is captured by iterating K on the latent variables. In addition, it is important to be able to map back to physical variables x, which is why the autoencoder structure is favorable [349]. Variational autoencoders are also used for stochastic dynamical systems, such as molecular dynamics, where the map back to physical configuration space from the latent variables is probabilistic [550, 368].

For simple systems with a discrete eigenvalue spectrum, a compact representation may be obtained in terms of a lew antencoender variables. However, dynamical alyzense with continuous eigenvalue spectra dely low-dimensional representations using many existing neuration and the simple pendulum to nonlinear optics and broadband turbulence. For example, the classical pendulum, prive by

$$\bar{x} = -\sin(\omega x)$$
 (7.100)



Figure 7.13 Deep neural network architecture used to identify Koopman eigenfunctions $\varphi(x)$. The network is based on a deep auto-encoder (a), which identifies intrinsic coordinates $y \equiv \varphi(x)$. Additional loss functions are included to enforce linear dynamics in the auto-encoder variables (bc). Recondensel with remaining from Lands et al. [349].



Figure 7.14 Modified network architecture with saxiliary network to parameterize the continuous eigenvalue spectrum. A continuous eigenvalue λ enables aggressive dimensionality reduction in the auto-encoder, avoiding the need for higher harmonics of the fundamental frequency that are senerated by the nonlinearity. *Recordsoci* with *worksion* (and *et al.*) 2001.

exhibits a continuous range of frequencies, from so to 0, as the amplitude of the pendulum oscillation is increased. Thus, the continuous spectrum confounds a simple description in terms of a few Koopman eigenfunctions [378]. Indeed, away from the linear regime, an infinite Fourier sum is required to approximate the shift in frequency.

In a creat work by Lande et al. (1949), an auxiliary network is used to parameterize the origination of the systep general cancel and a structure that the land horizontomics and interpretation. This parameterized network is depicted structure that the land horizontomics works and the system of the structure of the structure of the structure of the system of the structure of the structure of the structure of the structure of the system of the structure of the structure of the structure of the structure of the system of the structure of the struc

It is expected that neural network representations of dynamical systems, and Koopman embeddings in particular, will remain a growing area of interest in data-driven dynamics. Combining the representational power of deep learning with the elegance and simplicity of Koopman embeddings has the potential to transform the analysis and control of complex systems.

Suggested Reading

Texts

- Nonlinear oscillations, dynamical systems, and bifurcations of vector fields, by P. Holmes and J. Guckenheimer, 1983 (252).
- (2) Dynamic mode decomposition: Data-driven modeling of complex systems, by J. N. Kutz, S. L. Brunton, B. W. Brunton, and J. L. Proctor. 2016 [317].
- (3) Differential equations and dynamical systems, by L. Perko, 2013 [427].



Figure 7.15 Neural network embedding of the nonlinear pendulum, using the parameterized network in Fig. 7.14. As the pendulum amplitude increases, the frequency continuously changes (I). In the Koopman eigenfunction coordinates (III), the dynamics become linear, given by perfect circles (IIC). Removaled with permission from Larke et al. [249].

Papers and reviews

- Distilling free-form natural laws from experimental data, by M. Schmidt and H. Lipson, Science, 2009 [477].
- (2) Discovering governing equations from data by sparse identification of nonlinear dynamical systems, by S. L. Brunton, J. L. Procter, and J. N. Kutz, Proceedings of the National Acadews of Sciences 2016 1951.
- (3) On dynamic mode decomposition: theory and applications, by J. H. Tu, C. W. Rowley, D. M. Luchterburg, S. L. Brunten, J. N. Kutz, *Journal of Computational Dynamics*, 2014 (535).
- (4) Hamiltonian systems and transformation in Hilbert Space, by B. O. Koopman, Proceedings of the National Academy of Sciences, 1931 [300].
- (5) Spectral properties of dynamical systems, model reduction and decompositions, by I. Mezić, Nonlinear Dynamics, 2005 13761.
- (6) Data-driven model reduction and transfer operator approximation, by S. Klus, F. Nuske, P. Kolui, H. Wu, I. Kevrekidis, C. Schutte, and F. Noe, *Journal of Nonlin*or Dynamics, 2018 12931.
- (7) Hidden physics models: Machine learning of nonlinear partial differential equations, by M. Raissi and G. E. Karnindakis, *Journal of Computational Physics*, 2018 [445].

The focus of the look has lengtly been enhancertaing couples systems through lines: the straight probability, and provide the systems maching lines between an overgime engineering objective. The ends of practice of maniphating dynamic dynamics and practice dynamics and practice of maniphating dynamics dynamics and practice of the system. The straight and practice of maniphating dynamics and and the system and a straight dynamics dynamics and practice of applied and another and a straight of applied and the simulation of applied and applied and applied and applied and applied and practice of applied and applied and applied and applied and straight dynamics of the system, since changing the system change applied model practice interactions are straight.

Common theory has helped shape the modern technological and industrial landscape langungle abound, heating rules carried in anamobiles, position course of a constrution regiment, by by storing analysis in already industrial measuration, packet moting in a straight of the storage of the straight of the storage of the storage for of the many applications. In the finance, coursed will be increasingly applied to high dimensional, straight positionar and maticical positions, using and they are finance, epideonology, anneousness storage, and the storage of the dimensional traignly means constrain the variable of the storage of Compared 2014.

This chapter will introduce the key concepts from closed-loop feedback control. The goal is to build intuition for how and when to use feedback control, movineed by practical real-world challenges. Most of the theory will be developed for linear systems, where a wealth of powerful techniques circl [162, 492]. This theory will then be demonstrated on simple and intuitive examples, such as to develop a cruise controller for an automobile or shibite an inverse development.

Types of Control

There are many ways to manipulate the behavior of a dynamical system, and these control approaches are organized schematicality in Fig. 8.1. Passive control does not neglize input energy, and when sufficient, it is desirable because of its simplicity, reliability, and low cost. For example, stop signs at a traffic intersection regulate the flow of traffic. Active control requires input energy, and these controllers are divided into two broad categories based on whether or not services are used to inform the controller. In the first category, eque-loop whether or not services are used to inform the controller. In the first category, eque-loop to the service of the services are used to inform the controller.



Figure 8.1 Schematic illustrating the various types of control. Most of this chapter will focus on closed-loop feedback control.

cound effect on a per-popuration's cound sequence, in the traffic counds, is justice as the propursion of a sequence of the sequence of the sequence of the sequence counds in the second sequence of the sequence of the sequence counds in the second sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence of the sequence distribution of the sequence of the sequence of the sequence of the sequence distribution of the sequence of the

8.1 Closed-Loop Feedback Control

The main faces of this chapter is closed-loop foreflack cosmol, which is the method of chapter is the method of the method of the method of the method of depicts the general feedback cosmol framework, where seems measurements, y, et a system are fold-scale in a controller, which increases the method of the method methods and the dynamics and part for the system of the method of the method of the system of the system of the system of the system of the dynamics of the system of the dynamics of the system disturbances to be state of the system, w₁ is measurement using, and w₁ is a reference reaction with advacuation that system of the s

Mathematically, the system and measurements are typically described by a dynamical system:

$$\frac{d}{dt}\mathbf{x} = \mathbf{f}(\mathbf{x}, \mathbf{u}, \mathbf{w}_d) \quad (8.1a)$$



Figure 32: Standard framework for feedback control. Measurements of the system, $y_0^{(i)}$, are fold back into a controller, which then decides on the appropriate actuation signal w(t) to control the system. The control law is designed to modify the system dynamics and provide good performance, quantified by the cost J, despite exopensors disturbances and noise in w. The exogenous input w may also include a reference trainscieve w. It this should be tracked.

$$y = g(x, u, w_a).$$
 (8.1b)

The goal is to construct a control law

$$u = k(y, w_r)$$
 (8.2)

that minimizes a cost function

$$J \cong J(\mathbf{x}, \mathbf{u}, \mathbf{w}_{f}).$$
 (8.3)

Thus, modern control relies heavily on techniques from optimization [74]. In general, the controller in (6.2) will be a dynamical system, rather than a static function of the inputs. For example, the Kalman filter in Section 8.5 dynamically estimates the full state x from measurements of u and y. In this case, the control law will become $\mathbf{u} = \mathbf{k}(\mathbf{y}, \hat{\mathbf{x}}, \mathbf{w}_f)$, where $\hat{\mathbf{x}}$ is the full-state estimate.

To motivate the added com and computing of senses based feedback cosmet, it is height incompare this problem, be construct a height commutation is the commutation of the commutation o

In contrast, closed-loop feedback control, shown in Fig. 8.4 uses sensor measurements of the system to inform the controller about how the system is actually responding. These sensor measurements provide information about unmodeled dynamics and disturbances that would degrade the performance in open-loop control. Further, with feedback it is



Figure 8.3 Open-loop control diagram. Given a desired reference signal w_r , the open-loop control law constructs a control protocol u to drive the system based on a model. External disturbances (w_d) and sensor noise (w_m), as well as unmodeled system dynamics and uncertainty, are not accounted for and derrade performance.



Figure 8.4 Closed-loop feedback control diagram. The sensor signal y is fed back and subtracted from the reference signal w₁, providing information about how the system is respecting to actuation and extramid disturbances. The controller uses the resulting error w to determine the correct actuation signal u for the desired response. Feedback is often able to stabilize unstable dynamics while effectively rescirctim disturbances w₂ and attenuate movies movies.

often possible to modify and stabilize the dynamics of the closed-loop system, something which is not possible with open-loop control. Thus, closed-loop feedback control is often able to maintain high-performance operation for systems with unstable dynamics, model uncertainty, and external disturbances.

Examples of the Benefits of Feedback Control

To summarize, closed-loop feedback control has several benefits over open-loop control:

- It may be possible to stabilize an unstable system;
- It may be possible to compensate for external disturbances;
- . It may be possible to correct for unmodeled dynamics and model uncertainty.

These issues are illustrated in the following two simple examples.

Inverted pendulum Consider the unstable inverted pendulum equations, which will be derived later in Section 8.2. The linearized equations are:

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ g/L & d \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u \qquad (8.4)$$

where $x_1 = \theta$, $x_2 = \theta$, u is a torque applied to the pendulum arm, g is gravitational acceleration, L is the length of the pendulum arm, and d is damping. We may write this system in standard form as

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}u.$$

If we choose constants so that the natural frequency is $\omega_n = \sqrt{g/L} = 1$ and d = 0, then the system has eigenvalues $\lambda = \pm 1$, corresponding to an unstable saddle-type fixed point.

No open-loop control strategy can change the dynamics of the system, given by the eigenvalues of A. However, with full-state feedback control, given by $u = -\mathbf{K}\mathbf{x}$, the closedloop system becomes

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} = (\mathbf{A} - \mathbf{B}\mathbf{K})\mathbf{x}.$$

Choosing $\mathbf{K} = \begin{bmatrix} 4 & 4 \end{bmatrix}$, corresponding to a control law $u = -4x_1 - 4x_2 = -4\theta - 4\dot{\theta}$, the closed loop system $(\mathbf{A} - \mathbf{B}\mathbf{K})$ has stable eigenvalues $\lambda = -1$ and $\lambda = -3$.

Determining when it is possible to change the eigenvalues of the closed-loop system, and determining the appropriate control law K to achieve this, will be the subject of future sections.

Cruise control To appreciate the ability of closed-loop control to compensate for unmodeled dynamics and disturbances, we will consider a simple model of cruise control in an automobile. Let u be the rate of gas fed into the engine, and let y be the car's speed. Neglecting transients, a crude model 1 is:

Thus, if we double the gas input, we double the automobile's speed.

Based on this model, we may design an open-loop cruise controller to track a reference speed w, by simply commanding an input of $u = w_r$. However, an incorrect automobile model (i.e., in actuality y = 2u), or external disturbances, such as rolling hills (i.e., if $y = u + \sin(t)$), are not accounted for in the simple open-loop design.

In contrast, a closed-loop control luw, based on measurements of the speed, is able to compressite for unmodeled dynamics and disturbances. Consolider the closed-loop control law $u = Kr_{0-} - y_1$, so that gas is increased when the measured velocity is too low, and decreased when it is too high. Then if the dynamics are actually y = 2u instead of y = u, the open-boxy system will have 50% steady-state tracking error, while the performance of the closed-loop system can be significantly improved for large K:

$$y = 2K(w_r - y) \implies (1 + 2K)y = 2Kw_r \implies y = \frac{2K}{1 + 2K}w_r.$$
 (8.6)

For K = 50, the closed-loop system only has 1% steady-state tracking error. Similarly, an added disturbance w_d will be attenuated by a factor of 1/(2K + 1).

As a concrete example, consider a reference tracking problem with a deciden efference speed of 00 mph. The model is $y = u_{\rm ex}$ and the true system is y = 0.5 u. In addition, there is a disturbance in the forms of refling hills that increase and decrease the speed by j = 10 mph at a flat primery of 0.1 keV. As ones how conclude is i compared with a closed-loop proportional and the speed of the speed significantly better performance, we will see later that a large proportional gain may come at the cost of robuscness. Adding an imaged new will improve performance.

¹ A more realistic model would have acceleration dynamics, so that $\dot{x} = -x + u$ and y = x.



Figure 8.5 Oren-loop vs. closed-loop cruise control.

Code 8.1 Commare open-loop and closed-loop cruise control.

```
files sli, close sli, clc
t = 0.02102 t intervent
s = 0.02102 t intervent
```

8.2 Linear Time-Invariant Systems

The most complete theory of counts has been developed for linear systems (492, 165, 23); Linear systems are generally obtained by linearizing a nonlinear system hours a fued point or a periodic other. However, instability may epickly take a trajectory far away from the fixed point. Forumately, an effective subhizing controller will keep the state of the system in south neighbord of the fixed point where the linear approximation is viable. For example, in the case of the inverted pendulum, feedback count on my keep the pendulum sublicited in the verture points where the Juna: points and the pendulum sublicited in the verture points where the Juna: points where the pendulum sublicited in the verture points where the Juna: behave linearly.

Linearization of Nonlinear Dynamics

Given a nonlinear input-output system

$$\frac{d}{dt}\mathbf{x} = \mathbf{f}(\mathbf{x}, \mathbf{u}) \quad (8.7a)$$

$$y = g(x, u)$$
 (8.7b)

it is possible to linearize the dynamics near a fixed point (\hat{x}, \hat{u}) where $f(\hat{x}, \hat{u}) = 0$. For small $\Delta x = x - \hat{x}$ and $\Delta u = u - \hat{u}$ the dynamics f may be expanded in a Taylor series about the point (\hat{x}, \hat{u}) :

$$\mathbf{f}(\hat{\mathbf{x}} + \Delta \mathbf{x}, \hat{\mathbf{u}} + \Delta \mathbf{u}) = \mathbf{f}(\hat{\mathbf{x}}, \hat{\mathbf{u}}) + \underbrace{\frac{d\mathbf{f}}{d\mathbf{x}}\Big|_{(\mathbf{x}, \hat{\mathbf{u}})}}_{\mathbf{A}} \cdot \Delta \mathbf{x} + \underbrace{\frac{d\mathbf{f}}{d\mathbf{u}}\Big|_{(\mathbf{x}, \hat{\mathbf{u}})}}_{\mathbf{B}} \cdot \Delta \mathbf{u} + \cdots .$$
 (8.8)

Similarly, the output equation g may be expanded as:

$$g(\hat{\mathbf{x}} + \Delta \mathbf{x}, \hat{\mathbf{u}} + \Delta \mathbf{u}) = g(\hat{\mathbf{x}}, \hat{\mathbf{u}}) + \underbrace{\frac{dg}{dx}}_{C} \cdot \Delta \mathbf{x} + \underbrace{\frac{dg}{du}}_{D} \cdot \Delta \mathbf{u} + \cdots$$
 (8.9)

For small displacements around the fixed point, the higher order terms are negligibly small. Dropping the Δ and shifting to a coordinate system where \hat{x} , \hat{u} , and \hat{y} are at the origin, the linearized dynamics may be written as:

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} \qquad (8.10a)$$

$$y = Cx + Du$$
. (8.10b)

Note that we have neglected the disturbance and noise inputs, w_d and w_a, respectively; these will be added back in the discussion on Kalman filtering in Section 8.5.

Unforced Linear System

In the absence of control (i.e., u = 0), and with measurements of the full state (i.e., y = x), the dynamical system in (8.10) becomes

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x}.$$
 (8.11)

The solution $\mathbf{x}(t)$ is given by

$$\mathbf{x}(t) = e^{\mathbf{A}t}\mathbf{x}(0),$$
 (8.12)

where the matrix exponential is defined by:

$$e^{At} = \mathbf{I} + \mathbf{A}t + \frac{\mathbf{A}^{2}t^{2}}{2} + \frac{\mathbf{A}^{3}t^{3}}{3} + \cdots$$
 (8.13)

The solution in (8.12) is determined entirely by the eigenvalues and eigenvectors of the matrix A. Consider the eigendecomposition of A:

$$AT = TA$$
. (8.14)

In the simplest case, A is a diagonal matrix of distinct eigenvalues and **T** is a matrix whose columns are the corresponding linearly independent eigenvalues. A may be written in loodan form, with entries above the diagonal for degenerate eigenvalues of multiplicity ≥ 2 ; the corresponding columns of **T** will be generalized eigenvectors.

In either case, it is easier to compute the matrix exponential e^{Ar} than e^{Ar}. For diagonal A, the matrix exponential is given by:

$$e^{At} = \begin{bmatrix} e^{\lambda_1 t} & 0 & \cdots & 0 \\ 0 & e^{\lambda_2 t} & \cdots & 0 \\ \vdots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & e^{\lambda_m} \end{bmatrix}.$$
 (8.15)

In the case of a nontrivial Jordan block in A with entries above the diagonal, simple extensions exist related to nilpotent matrices (for details, see Perko [427]).

Rearranging the terms in (8.14), we find that it is simple to represent powers of A in terms of the eigenvectors and eigenvalues:

$$A = TAT^{-1}$$
 (8.16a)

$$A^{2} = (TAT^{-1})(TAT^{-1}) = TA^{2}T^{-1}$$

(8.16b)

$$\mathbf{A}^{2} = (\mathbf{T}\mathbf{A}\mathbf{T}^{-1})(\mathbf{T}\mathbf{A}\mathbf{T}^{-1})\cdots(\mathbf{T}\mathbf{A}\mathbf{T}^{-1}) = \mathbf{T}\mathbf{A}^{2}\mathbf{T}^{-1}.$$
 (8.16c)

Finally, substituting these expressions into (8.13) yields:

$$e^{\mathbf{A}t} = e^{\mathbf{T}\mathbf{A}\mathbf{T}^{-1}t} = \mathbf{T}\mathbf{T}^{-1} + \mathbf{T}\mathbf{A}\mathbf{T}^{-1}t + \frac{\mathbf{T}\mathbf{A}^{2}\mathbf{T}^{-1}t^{2}}{2} + \frac{\mathbf{T}\mathbf{A}^{3}\mathbf{T}^{-1}t^{3}}{3} + \cdots$$
 (8.17a)

$$= \mathbf{T} \left[\mathbf{I} + \mathbf{A}t + \frac{\mathbf{A}^{2}t^{2}}{2} + \frac{\mathbf{A}^{3}t^{3}}{3} + \cdots \right] \mathbf{T}^{-1} \qquad (8.17b)$$

$$= Te^{At}T^{-1}$$
. (8.17c)

Thus, we see that it is possible to compute the matrix exponential efficiently in terms of the eigendecomposition of A. Moreover, the matrix of eigenvectors T defines a change of coordinates that dramatically simplifies the dynamics:

$$\mathbf{x} = \mathbf{T}\mathbf{z} \implies \dot{\mathbf{z}} = \mathbf{T}^{-1}\dot{\mathbf{x}} = \mathbf{T}^{-1}\mathbf{A}\mathbf{x} = \mathbf{T}^{-1}\mathbf{A}\mathbf{T}\mathbf{z} \implies \dot{\mathbf{z}} = \mathbf{A}\mathbf{z}.$$
 (8.18)

In other words, changing to eigenvector coordinates, the dynamics become diagonal. Combining (8.12) with (8.17c), it is possible to write the solution x(r) as

$$\mathbf{x}(t) = \mathbf{T} e^{\mathbf{A}t} \frac{\mathbf{T}^{-1}\mathbf{x}(0)}{\mathbf{z}(t)}$$
. (8.19)

In the first step, T^{-1} maps the initial condition in physical coordinates, x(0), into eigenvector coordinates, z(0). The next step advances these initial conditions using the diagonal update e^{At} , which is considerably simpler in eigenvector coordinates z. Finally, multiplying by T mans z(t) back to theyical coordinates x(t).

In addition to making it possible to compute the matrix exponential, and hence the solution x(t), the eigendecomposition of A is even more useful to understand the dynamics and stability of the system. We see from (8.19) that the only time-varying portion of the solution is h^{2k} . In general, these eigenvalues $\lambda = a + b \max p = compte numbers, so that$ $the solutions are given by <math>h^{2k} = h^{2k}(costh)^{2k} + 1 \operatorname{sidis}(h)$. Thus, the other real part (i.e., Re(1)) = a < 0), then the system is stable, and solutions all decay to $\lambda = 0$ at $\lambda = -\infty$. However, if there a single significant the positive real part, then the system is unstable and will diverge from the fixed point along the corresponding unstable eigenvectors. Any motion minimal condition is likely be used as h^{2k} eigenvectors of the system.

Forced Linear System

With forcing, and for zero initial condition, $\mathbf{x}(0) = \mathbf{0}$, the solution to (8.10a) is

$$\mathbf{x}(t) = \int_{0}^{t} e^{\mathbf{A}(t-\tau)} \mathbf{B} \mathbf{u}(\tau) d\tau \triangleq e^{\mathbf{A} t} \mathbf{B} * \mathbf{u}(t).$$
 (8.20)

The control input u(t) is convolved with the kernel $e^{At}B$. With an output y = Cx, we have $y(t) = Ce^{At}B * u(t)$. This convolution is illustrated in Fig. 8.6 for a singleinput, single-output (SISO) system in terms of the impulse response $g(t) = Ce^{At}B = \int_{0}^{\infty} C^{A(t-1)}B(t) dt$ given a Dirac delta input $u(t) = \delta(t)$.

Discrete-Time Systems

In many real-world applications, systems are sampled at discrete instances in time. Thus, digital control systems are typically formulated in terms of discrete-time dynamical systems:

$$x_{k+1} = A_d x_k + B_d u_k$$
 (8.21a)

$$y_k = C_d x_k + D_d u_k$$
, (8.21b)

where $x_k = x(k\Delta t)$. The system matrices in (8.21) can be obtained from the continuoustime system in (8.10) as

$$A_d = e^{A\Delta t}$$

(8.22a)

$$\mathbf{B}_{d} = \int_{0}^{\Delta t} e^{\mathbf{A} \mathbf{r}} \mathbf{B} d\tau \qquad (8.22b)$$

$$C_d = C$$
 (8.22c)

$$D_d = D.$$
 (8.22d)

The stability of the discrete-time system in (8.21) is still determined by the eigenvalues of A_x , although now a system is stable if and only if all discrete-time eigenvalues are inside the unit circle in the complex plane. Thus, exp(AAT) defines a conformal mapping on the complex plane from continuous-time to discrete-time, where eigenvalues in die the unit circle.



Figure 8.8 Convolution for a single-input, single-output (SISO) system.

Example: Inverted Pendulum

Consider the inverted pendulum in Fig. 8.8 with a torque input u at the base. The equation of motion, derived using the Euler–Lagrange equations², is:

$$\ddot{\theta} = -\frac{g}{L}\sin(\theta) + u. \quad (8.23)$$

Introducing the state x, given by the angular position and velocity, we can write this second order differential equation as a system of first order equations:

$$\mathbf{x} = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} \theta \\ \theta \end{bmatrix} \implies \frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} x_2 \\ -\frac{s}{2} \sin(x_1) + u \end{bmatrix}.$$
 (8.24)

² The Lagrangian is $\mathcal{L} = \frac{10}{2}L^2\dot{\theta}^2 - \exp L\cos(\theta)$, and the Euler-Lagrange equation is $\frac{d}{dt}\partial \mathcal{L}/\partial\dot{\theta} - \partial \mathcal{L}/\partial\theta = \tau$, where τ is the input torque.



Figure 8.7 The matrix exponential defines a conformal map on the complex plane, mapping stable eigenvalues in the left half plane into eigenvalues invide the unit circle.



Figure 8.8 Schematic of inverted pendulum system

Taking the Jacobian of f(x, u) yields

$$\frac{df}{dx} = \begin{bmatrix} 0 & 1\\ -\frac{g}{2} \cos(x_1) & 0 \end{bmatrix}, \quad \frac{df}{du} = \begin{bmatrix} 0\\ 1 \end{bmatrix}. \quad (8.25)$$

Linearizing at the pendulum up $(x_1 = \pi, x_2 = 0)$ and down $(x_1 = 0, x_2 = 0)$ equilibria gives

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ \frac{d}{2} & 0 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u \qquad \qquad \frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 0 & 1 \\ -\frac{d}{2} & 0 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u$$
Pendalam days, $\lambda = i \sqrt{dt}$

Thus, we see that the down position is a stable center with eigenvalues $\lambda = \pm i \sqrt{g/L}$ corresponding to oscillations at a natural frequency of $\sqrt{g/L}$. The pendulum up position is an unstable saddle with eigenvalues $\lambda = \pm \sqrt{g/L}$.

8.3 Controllability and Observability

A natural question arises in linear control theory. To what extent can closed-loop forefluck un — Kx maniphate the behavior of the system in (8.10a)? We already use in Section 8.1 that it was possible to modify the eigenvalues of the mandhe inverted pendulum system Vi and the system of the system in a new system matrix ($A \rightarrow BK$) with stable eigenvalues. This section will provide correct confidence on when and how the system dynamic syma be maniplated through feedback control. The dual question, of when it is possible to entime the full state whom measurements will also be addressed.

Controllability

The ability to design the eigenvalues of the closed-loop system with the choice of K relies on the system in (8.10a) being controllable. The controllability of a linear system is determined entirely by the column space of the controllability matrix C:

$$C = [B \ AB \ A^2B \ \cdots \ A^{n-1}B].$$
 (8.26)

If the matrix C has n inscript independent columns, so that it spans all of \mathbb{R}^{N} , then the system in (8.10a) is controllable. The span of the columns of the controllability matrix C forms a Krylov subspace that determines which state vector directions in \mathbb{R}^{N} may be marginalized with context. Thus, in addition to controllability implying arbitrary eigenvalue placement, it also implies that any state $\xi \in \mathbb{R}^{N}$ is reachable in a finite time with some arturation signal ut/).

The following three conditions are equivalent:

1. Controllability. The span of C is R*. The matrix C may be generated by

>> ctrb(A,B)

and the rank may be tested to see if it is equal to n, by

>> rank(ctrb(A,B))

 Arbitrary eigenvalue placement. It is possible to design the eigenvalues of the closedloop system through choice of feedback u = -Kx:

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} = (\mathbf{A} - \mathbf{B}\mathbf{K})\mathbf{x}. \quad (8.27)$$

Given a set of desired eigenvalues, the gain K can be determined by

>> K = place (A, B, neweigs);

Designing K for the best performance will be discussed in Section 8.4.

 Reachability of ℝⁿ. It is possible to steer the system to any arbitrary state x(t) = § ∈ ℝⁿ in a finite time with some actuation signal u(t).

Note that reachability also applies to open-loop systems. In particular, if a direction ξ is not in the span of C, then it is impossible for control to push in this direction in either open-loop or closed-loop. Examples The notion of controllability is more easily understood by investigating a few simple examples. First, consider the following system

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 1 & 0 \\ 0 & 2 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u \implies C = \begin{bmatrix} 0 & 0 \\ 1 & 2 \end{bmatrix}. \quad (8.28)$$

This system is not controllable, because the controllability matrix C consists of two linearly dependent vectors and does not span \mathbb{R}^2 . Even before checking the rank of the controllability matrix, it is easy to see that the system won't be controllable since the states x_1 and x_2 are completely decoupled and the actuation input a only effects the second state.

Modifying this example to include two actuation inputs makes the system controllable by increasing the control authority:

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 1 & 0 \\ 0 & 2 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix} \begin{bmatrix} u_1 \\ u_2 \end{bmatrix} \implies C = \begin{bmatrix} 1 & 0 & 1 & 0 \\ 0 & 1 & 0 & 2 \end{bmatrix}.$$
(8.29)

This fully actuated system is clearly controllable because x_1 and x_2 may be independently controlled with u_1 and u_2 . The controllability of this system is confirmed by checking that the columns of C do som \mathbb{R}^2 .

The most interesting cases are less obvious than these two examples. Consider the system

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 1 & 1 \\ 0 & 2 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 0 \\ 1 \end{bmatrix} u \implies C = \begin{bmatrix} 0 & 1 \\ 1 & 2 \end{bmatrix}. \quad (8.30)$$

This two-state system is controllable with a single actuation input because the states x_1 and x_2 are now coupled through the dynamics. Similarly,

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} 1 & 0 \\ 0 & 2 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 1 \\ 1 \end{bmatrix} u \implies C = \begin{bmatrix} 1 & 1 \\ 1 & 2 \end{bmatrix}. \quad (8.31)$$

is controllable even though the dynamics of x_1 and x_2 are decoupled, because the actuator $\mathbf{B} = \begin{bmatrix} 1 & 1 \end{bmatrix}^T$ is able to simultaneously affect both states and they have different timescales.

We will see in Section 8.3 that controllability is intimately related to the alignment of the columns of **B** with the eigenvector directions of **A**.

Observability

Mathematically, observability of the system in (8.10) is nearly identical to controllability, although the physical interpretation differs somewhat. A system is observable if it is possible to estimate any start $\theta \in \mathbb{R}^n$ from a time-history of the measurements y(t).

Again, the observability of a system is entirely determined by the row space of the observability matrix O:

$$\mathcal{O} = \begin{bmatrix} \mathbf{C} \\ \mathbf{CA} \\ \mathbf{CA}^2 \\ \vdots \\ \mathbf{CA}^{n-1} \end{bmatrix}.$$
(8.32)

In particular, if the rows of the matrix O span \mathbb{R}^n , then it is possible to estimate any fulldimensional state $x \in \mathbb{R}^n$ from the time-history of y(t). The matrix O may be generated by

>> obsv(A,C);

The motivation for full-state estimation is relatively straightforward. We have already end may the Hult are foreback, as $m = -K_{\rm c}$, the possible to modify the behavior of a constraints by symm. However, if Hult state measurements in the area or available, it is necesary to rotinear. Years the measurements, This is possible when the system is observable, in Section 15.5, we will use that it is possible to find any space of the system of the system is observable. It is possible to find any space of the system is observable, the possible to find any space of the system of the system is the system is the system. It is being an abservable may also space of the system of the system is the size is system is the size is system is the size is system is the system is the system is the size is system is size is system is the size is system is the size is system is size is size is system is size is si

Interestingly, the observability criterion is mathematically the dual of the controllability criterion. In fact, the observability matrix is the transpose of the controllability matrix for the pair (A^T, C^T) :

>> 0 = ctrb(A',C')'; % 'obsv' is dual of 'crtb'

The PBH Test for Controllability

There are many tests to determine whether or not a system is controllable. One of the most useful and liminating is the Popor-Beckvich-Harator (BH) test. The PBH test states that the pair (A, B) is controllable if and easily if the column rank of the matrix $|(A \rightarrow I)| = B_i$ equal to N for all $A \in C$. This test is particularly factanismi focusne it contexts controllability³ to a relationship between the columns of **B** and the eigenspace of **A**.

First, the PBH test only needs to be checked at λ that are eigenvalues of \mathbf{A} , since the rank of $\mathbf{A} - \lambda \mathbf{I}$ is equal to n except when λ is an eigenvalue of \mathbf{A} . In fact, the characteristic equation det $(\mathbf{A} - \lambda \mathbf{I}) = 0$ is used to determine the eigenvalues of \mathbf{A} as exactly those values where the matrix $\mathbf{A} - \lambda \mathbf{I}$ becomes rank deficient, or determine.

Now, given that $(A - \lambda)$ is only rank deficient for eigenvalues λ , it also follows that the null-space, or kernel, of $A - \lambda 1$ is given by the span of the eigenvectors corresponding to that particular eigenvalue. Thus, for $(A - \lambda 1)$ B hoker mak *n*, the columns in B mush have some component in each of the eigenvector directions associated with A to complement the null-space of $A - \lambda 1$.

If A has a distinct eigenvalues, then the system will be controllable with a single actuation input, since the matrix A \rightarrow 3 will have at most one eigenvector direction in the mill-space. In particular, we may choose B as the sum of all of the n linearly independent eigenvectors, and it will be guaranteed to have some composite in each direction. It is be controllable with high probability, since it will be exceedingly unlikely that B will be candomly choose to but it has zero controllable from any que concentration.

If there are degenerate eigenvalues with multiplicity ≥ 2 , so that the null-space of $\mathbf{A} - \lambda \mathbf{I}$ is multidimensional, then the actuation input must have as many degrees of freedom. In other words, the only time that multiple actuators (columns of **B**) are strictly required is for

³ There is an equivalent PBH test for observability that states that $\begin{bmatrix} (A - \lambda I) \\ C \end{bmatrix}$ must have row rank *n* for all $\lambda \in \mathbb{C}$ for the votem to be observable.

systems that have degenerate eigenvalues. However, if a system is highly nonnormal, it may helpful to have multiple actuators in practice for better control authority. Such nonnormal systems are characterized by large transient growth due to destructive interference between nearly parallel eigenvectors, often with similar eigenvalues.

The Caylev-Hamilton Theorem and Reachability

To provide insight into the relationship between the controllability of the pair (A, B) and the reachability of any vector $\xi \in \mathbb{R}^n$ via the actuation input u(r), we will leverage the Coyley-Hamilton theorem. This is a gern of linear algebra that provides an elegant way to represent solutions of X = Ax in terms of a finite sum of powers of A, rather than the infinite sum resulted for the matrix economial in S(3).

The Cayley–Hamilton theorem states that every matrix A satisfies its own characteristic (circervalue) equation. $det(A - \lambda I) = 0$:

$$det(\mathbf{A} - \lambda \mathbf{I}) = \lambda^{n} + a_{n-1}\lambda^{n-1} + \cdots + a_{2}\lambda^{2} + a_{1}\lambda + a_{0} = 0 \qquad (8.33a)$$

$$\Rightarrow \mathbf{A}^{n} + a_{n-1}\mathbf{A}^{n-1} + \cdots + a_{2}\mathbf{A}^{2} + a_{1}\mathbf{A} + a_{0}\mathbf{I} = \mathbf{0}.$$
 (8.33b)

Although this is relatively simple to state, it has profound consequences. In particular, it is possible to express Aⁿ as a linear combination of smaller powers of A:

$$\mathbf{A}^{n} = -a_0\mathbf{I} - a_1\mathbf{A} - a_2\mathbf{A}^2 - \cdots - a_{n-1}\mathbf{A}^{n-1}.$$
 (8.34)

It is straightforward to see that this also implies that any higher power $A^{2 \ge n}$ may also be expressed as a sum of the matrices $\{I, A, \dots, A^{n-1}\}$:

$$\Lambda^{k \ge n} = \sum_{j=0}^{n-1} \alpha_j \Lambda^j.$$
 (8.35)

Thus, it is possible to express the infinite sum in the exponential e^{At} as:

$$e^{At} = \mathbf{I} + \mathbf{A}t + \frac{\mathbf{A}^2 t^2}{2} + \cdots$$
 (8.36a)

$$= \beta_0(t)\mathbf{I} + \beta_1(t)\mathbf{A} + \beta_2(t)\mathbf{A}^2 + \cdots + \beta_{n-1}(t)\mathbf{A}^{n-1}. \quad (8.36b)$$

We are now equipped to see how controllability relates to the reachability of an arbitrary vector $\xi \in \mathbb{R}^n$. From (8.20), we see that a state ξ is reachable if there is some u(r) so that:

$$\xi = \int_{0}^{\tau} e^{\mathbf{A}(t-\tau)} \mathbf{B} \mathbf{u}(\tau) d\tau.$$
 (8.37)

Expanding the exponential in the right hand side in terms of (8.36b), we have:

$$\begin{split} \xi &= \int_0^t (\beta_0(t-\tau)\mathbf{IB}\mathbf{u}(\tau) + \beta_1(t-\tau)\mathbf{AB}\mathbf{u}(\tau) + \cdots \\ & \cdots + \beta_{n-1}(t-\tau)\mathbf{A}^{n-1}\mathbf{B}\mathbf{u}(\tau))d\tau \\ &= \mathbf{B}\int_0^t \beta_0(t-\tau)\mathbf{u}(\tau)\,d\tau + \mathbf{AB}\int_0^t \beta_1(t-\tau)\mathbf{u}(\tau)\,d\tau + \cdots \\ & \cdots + \mathbf{A}^{n-1}\mathbf{B}\int_0^t \beta_{n-1}(t-\tau)\mathbf{u}(\tau)\,d\tau \end{split}$$

$$= \begin{bmatrix} \mathbf{B} & \mathbf{A}\mathbf{B} & \cdots & \mathbf{A}^{n-1}\mathbf{B} \end{bmatrix} \begin{bmatrix} \int_{0}^{T} \beta_0(r-\tau)\mathbf{u}(\tau) d\tau \\ \int_{0}^{T} \beta_1(r-\tau)\mathbf{u}(\tau) d\tau \\ \vdots \\ \int_{0}^{T} \beta_{n-1}(r-\tau)\mathbf{u}(\tau) d\tau \end{bmatrix}$$

Note that the matrix on the left is the controllability matrix C_{n} and we see that the only with all of \mathbb{R}^{2} is reachable is if the coulom space of C spans all of \mathbb{R}^{2} . It is somewhat more difficult to see that if C has rank n them is its possible to design a only to neach any advantage starting starting E_{n}^{2} . The high relies control fact, that the R is not constant in the fact that the R integration to the control integration $[g_{1}(r)]_{n=0}^{2}$ and $[g_{2}(r)]_{n=0}^{2}$ the distribution of R is the distribution of the fact that the R integration integration is provided in the relation of the rank $[g_{1}(r)]_{n=0}^{2}$ and $[g_{2}(r)]_{n=0}^{2}$ the distribution of the rank $[g_{2}(r)]_{n=0}^{2}$ is the rank $[g_{2}(r)]_{n=0}^{2}$ the rank $[g_$

Gramians and Degrees of Controllability/Observability

The previous tests for controllability and observability are binary, in the sense that the rank of C (resp. O) is either n, or it isn't. However, there are degrees of controllability and observability, as some states x may be easier to control or estimate than others.

To identify which states are more or less controllable, one must analyze the eigendecomposition of the controllability Gramian:

$$\mathbf{W}_{\varepsilon}(t) = \int_{0}^{t} e^{\mathbf{A} \mathbf{\tau}} \mathbf{B} \mathbf{B}^{*} e^{\mathbf{A}^{*} \mathbf{\tau}} d\tau. \qquad (8.38)$$

Similarly, the observability Gramian is given by:

$$\mathbf{W}_{o}(t) = \int_{0}^{t} e^{\mathbf{A}^{t}\tau} \mathbf{C}^{*} \mathbf{C} e^{\mathbf{A}\tau} d\tau. \qquad (8.39)$$

These Gramians are often evaluated at infinite time, and unless otherwise stated, we refer to $W_c = \lim_{t\to\infty} W_c(t)$ and $W_o = \lim_{t\to\infty} W_o(t)$.

The controllability of a state is in measured by t-Wa_A, which will be larger for more controllabilities. It is break of t-Wa_A is large, then it is possible to avoign the system far in the state direction with a stat control larger. The observability of a state is similarly measured by t-Wa_A. Both Gramma are apresented; and positive semi-difficult, hencing nonnegative eigenvalues. Thus, the eigenvalues and eigenvectors may be ordered hierarchically, while eigenvalues. Thus, the eigenvalues are larger eigenvalues being more easily controllable or observable. In this way, the Gramman are used eigenvalues being more active controllable or observable. In this way, the Gramman induce a new inner-product over state-space in terms of the controllabilitor or observables of the states.

Cannians may be visualized by ellipoids in state-space, with the principal area given by directions that are hierarchically outerain in terms of controllinity or observability. An example of this visualization is shown in Fig. 9.2 in Chapter 9. In fact, Cannians may be used to design freder-older models for high-dimensional optersm. Thoogo Jabalancing transformation, a key subspace is identified with the most jointy controllable and observable modes. These modes these there a good projection basis to define a model that a opteres the dominant input-coupt dynamics. This form of balanced model reduction will be investigated fraction is Section 9.2. Gramians are also useful to determine the minimum-energy control $\mathbf{u}(t)$ required to navigate the system to $\mathbf{x}(t_f)$ at time t_f from $\mathbf{x}(0) = \mathbf{0}$:

$$\mathbf{u}(t) = \mathbf{B}^{*} \left(e^{\mathbf{A}(t_{f}-t)} \right)^{*} \mathbf{W}_{c}(t_{f})^{-1} \mathbf{x}(t_{f}).$$
 (8.40)

The total energy expended by this control law is given by

$$\int_{0}^{l_{f}} \|\mathbf{u}(\tau)\|^{2} d\tau = \mathbf{x}^{*} \mathbf{W}_{c}(t_{f})^{-1} \mathbf{x}. \quad (8.41)$$

It can now be seen that if the controllability matrix is nearly singular, then there are directions that require extreme actuation energy to manipulate. Conversely, if the eigenvalues of W_c are all large, then the system is easily controlled.

It is generally impractical to compute the Gramians directly using (8.38) and (8.39). Instead, the controllability Gramian is the solution to the following Lyapunov equation:

$$AW_c + W_c A^* + BB^* = 0,$$
 (8.42)

while the observability Gramian is the solution to

$$A^*W_o + W_oA + C^*C = 0.$$
 (8.43)

Obtaining Gramians by solving a Lyapunov equation is typically quite expensive for high-dimensional systems [213, 231, 496, 489, 55]. Instead, Gramians are often approximated empirically using snapshot data from the direct and adjoint systems, as will be discussed in Section 9.2.

Stabilizability and Detectability

In practice, full-state controllability and observability may be too much to expect in highdimensional systems. For example, in a high-dimensional fluid system, it may be unrealistic to manipulate every minor fluid vortex; instead control authority over the large, energycontaining coherent structures is often enough.

Stabilizability refers to the ability to control all unstable eigenvector directions of A, so that they are in the span of C. In practice, we might relax this definition to include lightly damped eigenvector modes, corresponding to eigenvalues with a small, negative real part. Similarly, if all unstable eigenvectors of A are in the span of O^* , then the system is detectable.

There may also be states in the model description that are superfluous for control. As an example, consider the control system for a commercial passenger jet. The state of the system may include the passenger seat positions, although this will surely not be controllable by the pilot, nor should it be.

8.4 Optimal Full-State Control: Linear Quadratic Regulator (LQR)

We have seen in the previous sections that if (A, B) is controllable, then it is possible to arbitrarily manipulate the eigenvalues of the closed-loop system (A - BK) through choice of a full-state feedback control law u = -Kx. This implicitly assumes that full-state measurements are available (i.e., C = I and D = 0, so that y = x). Although full-state measurements are not always available, especially for high-dimensional systems, we will show in the next section that if the system is observable, it is possible to build a full-state estimate from the sensor measurements.

Over a controllable system, and other measurements of the fill data of a subsection of the system with in file and emission, from a result objects of adultizing count laws with the system with the file adult of the system of the system of the trainy study, placing them as it as a shorted in the fill shaft of the couplex junce. However, we can be able adult of the system of the system of the system of the data search in a study of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the system of the system of the system of the laws of the system of the laws of the system of the laws of the system of the laws of the system of th

Choosing the best gain matrix \mathbf{k} to stabilize the system without expending too much correct efforts in an important gain al organization control. A shall have must be struct the stability of the closed-loop system and the aggressiveness of control. It is important to also control expenditure into account 10 to prevent the controller from over-reacting allowed amplitudes, and 3) so that control is not prohibitively expensive. In particular, the cone frances

$$J(t) = \int_{0}^{t} \mathbf{x}(\tau)^{*} \mathbf{Q} \mathbf{x}(\tau) + \mathbf{u}(\tau)^{*} \mathbf{R} \mathbf{u}(\tau) d\tau \qquad (8.44)$$

balances the cost of effective regulation of the state with the cost of control. The matrices Q and R weight the cost of deviations of the state from zero and the cost of actuation, respectively. The matrix Q is positive semi-definite, and R is positive definite; these matrices are often diagonal, and the diagonal elements may be tuned to change the relative importance of the control objectives.

Adding such a cost function makes choosing the control law a well-poord optimization problem, for which there is a weahl of heteroricia and manufactula techniques [74]. The linear-quadratic-segulator (LQR) control law $u = -K_x$ is designed to minimize $J = -K_x$ is to designed to minimize $J = -K_x$ is designed to minimize $J = -K_x$ is a dimensional technique (LQR) and the second s

$$K_r = R^{-1}B^*X$$
, (8.45)

where X is the solution to an algebraic Riccati equation:

$$A^*X + XA - XBR^{-1}B^*X + Q = 0.$$
 (8.46)

Solving the above Riccati equation for X, and hence for K_r, is numerically robust and already implemented in many programming languages [323, 55]. In Matlab, K_r is obtained via

However, solving the Riccati equation scales as $O(n^3)$ in the state-dimension n, making it prohibitively expensive for large systems or for online computations for slowly changing



Figure 8.9 Schematic of the linear quadratic regulator (LQR) for optimal full-state feedback. The optimal controller for a linear system given measurements of the full state, y = x, is given by proportional control $u = -K_r x$ where K_r is a constant gain matrix obtained by solving an algebraic Riccati catuation.

state equations or linear parameter varying (LPV) control. This motivates the development of reduced-order models that capture the same dominant behavior with many fewer states. Control-oriented reduced-order models will be developed more in Chapter 9.

The LQR controller is shown schematically in Fig. 8.9. Out of all possible control laws $\mathbf{u} = \mathbf{K}(\mathbf{x})$, including nonlinear controllers, the LQR controller $\mathbf{u} = -\mathbf{K}, \mathbf{x}$ is optimal, as we will show in Section 8.4. However, it may be the case that a linearized system is linearly uncontrollable while the full nonlinear system in (8.7) is controllable with a nonlinear control law $\mathbf{u} = \mathbf{K}(\mathbf{x})$.

Derivation of the Riccati Equation for Optimal Control

It is worth taking a theoretical desur here to derive the Riccati equation in (3.46) for the problem of optimal field-state regulation. This derivation well provide an example of how to solve convex optimization problems using the calculus of variations, and it will also provide a template for computing the optimal control solution for monlinear systems. Because of the similarity of optimal control to the formulation of Lagrangian and Hamiltonian classical mechanics in the optimal control to the dominant of the anti-anti-anti-anti-antisolution of the variational principal, we adopt similar language and notation.

First, we will add a terminal cost to our LQR cost function in (8.44), and also introduce a factor of 1/2 to simplify computations:

$$J = \int_{0}^{t_{f}} \frac{1}{2} \frac{(\mathbf{x}^{*} \mathbf{Q} \mathbf{x} + \mathbf{u}^{*} \mathbf{R} \mathbf{u})}{\operatorname{Lagrangian} \mathcal{L}} d\tau + \underbrace{\frac{1}{2} \mathbf{x}(t_{f})^{*} \mathbf{Q}_{f} \mathbf{x}(t_{f})}_{\text{Terminal cost}}.$$
 (8.47)

The goal is to minimize the quadratic cost function J subject to the dynamical constraint:

$$\dot{x} = Ax + Bu.$$
 (8.48)

We may solve this using the calculus of variations by introducing the following augmented cost function

$$J_{\text{aug}} = \int_{0}^{t_{f}} \left[\frac{1}{2} \left(\mathbf{x}^{*} \mathbf{Q} \mathbf{x} + \mathbf{u}^{*} \mathbf{R} \mathbf{u} \right) + \mathbf{\lambda}^{*} \left(\mathbf{A} \mathbf{x} + \mathbf{B} \mathbf{u} - \dot{\mathbf{x}} \right) \right] d\tau + \frac{1}{2} \mathbf{x}(t_{f})^{*} \mathbf{Q}_{f} \mathbf{x}(t_{f}). \quad (8.49)$$

The variable λ is a Lagrange multiplier, called the *co-state*, that enforces the dynamic constraints. λ may take any value and J_{aure} = J will hold.

Taking the total variation of Jane in (8.49) yields:

$$\delta J_{aug} = \int_{0}^{t_{f}} \left[\frac{\partial \mathcal{L}}{\partial \mathbf{x}} \delta \mathbf{x} + \frac{\partial \mathcal{L}}{\partial \mathbf{u}} \delta \mathbf{u} + \lambda^{*} \mathbf{A} \delta \mathbf{x} + \lambda^{*} \mathbf{B} \delta \mathbf{u} - \lambda^{*} \delta \dot{\mathbf{x}} \right] d\tau + \mathbf{Q}_{f} \mathbf{x}(t_{f}) \delta \mathbf{x}(t_{f}). \quad (8.50)$$

The partial derivatives⁴ of the Lagrangian are $\partial \mathcal{L}/\partial x = x^*Q$ and $\partial \mathcal{L}/\partial u = u^*R$. The last term in the integral may be modified using integration by parts:

$$-\int_{0}^{t_{f}} \lambda^{*} \delta \dot{\mathbf{x}} dr = -\lambda^{*}(t_{f}) \delta \mathbf{x}(t_{f}) + \lambda^{*}(0) \delta \mathbf{x}(0) + \int_{0}^{t_{f}} \dot{\lambda}^{*} \delta \mathbf{x} dr.$$

The term $\lambda^{*}(0)\delta x(0)$ is equal to zero, or else the control system would be non-causal (i.e., then future control could change the initial condition of the system).

Finally, the total variation of the augmented cost function in (8.50) simplifies as follows:

$$\delta J_{aug} = \int_{0}^{r_f} (\mathbf{x}^* \mathbf{Q} + \mathbf{\lambda}^* \mathbf{A} + \dot{\mathbf{\lambda}}^*) \, \delta \mathbf{x} \, d\tau + \int_{0}^{r_f} (\mathbf{u}^* \mathbf{R} + \mathbf{\lambda}^* \mathbf{B}) \, \delta \mathbf{u} \, d\tau + (\mathbf{x}(r_f)^* \mathbf{Q}_f - \mathbf{\lambda}^*(r_f)) \, \delta \mathbf{x}(r_f).$$
 (8.51)

Each variation term in (8.51) must equal zero for an optimal control solution that minimizes J. Thus, we may break this up into three equations:

$$x^*Q + \lambda^*A + \lambda^* = 0$$
 (8.52a)

$$u^*R + \lambda^*B = 0$$
 (8.52b)

$$\mathbf{x}(t_f)^* \mathbf{Q}_f - \mathbf{\lambda}^*(t_f) = \mathbf{0}.$$
 (8.52c)

Note that the constraint in (8.52c) represents an initial condition for the reverse-time equation for λ starting at r_j . Thus, the dynamics in (8.48) with initial condition $X(0) = x_0$ and (8.52) with the final-time condition $\lambda(r_j) = \mathbf{Q}_j r \delta(r_j)$ from a two-point boundary value problem. This may be integrated numerically to find the optimal control solution, even for nonlinear systems

Because the dynamics are linear, it is possible to *posit* the form $\lambda = Px$, and substitute into (8.52) above. The first equation becomes:

$$(Px + P\dot{x})^* + x^*Q + \lambda^*A = 0.$$

Taking the transpose, and substituting (8.48) in for x, yields:

$$Px + P(Ax + Bu) + Qx + A^*Px = 0$$

From (8.52b), we have

$$u = -R^{-1}B^*\lambda = -R^{-1}B^*Px$$
.

⁴ The derivative of a matrix expression Ax with respect to x is A, and the derivative of x^{*}A with respect to x is A^{*}.

Finally, combining yields:

$$\dot{P}x + PAx + A^*Px - PBR^{-1}B^*Px + Qx = 0.$$
 (8.53)

This equation must be true for all x, and so it may also be written as a matrix equation. Dropping the terminal cost and tetting time go to infinity, the P term disappears, and we recover the alterbraic Riccari contain:

$$PA + AP^* - PBR^{-1}B^*P + Q = 0.$$

Although this precedure is somewhat involved, each step is relatively straightforward. In addition, the dynamics in Eq (8.48) may be replaced with nonlinear dynamics $\dot{\mathbf{x}} = \mathbf{f}(\mathbf{x}, \mathbf{u})$, and a similar mollinear two-point boundary value problem may be formulated with $d/\gamma \mathbf{u}$ replacing **A** and $d/\gamma \mathbf{u}$ replacing **B**. This procedure is extremely general, and may be used to numerically obtain nonlinear ordinan control repetories.

Hamiltonian Formulation Similar to the Lagrangian formulation above, it is also possible to solve the optimization problem by introducing the following Hamiltonian:

$$\mathcal{H} = \underbrace{\frac{1}{2} \left(\mathbf{x}^* \mathbf{Q} \mathbf{x} + \mathbf{u}^* \mathbf{R} \mathbf{u} \right)}_{\mathcal{L}} + \lambda^* \left(\mathbf{A} \mathbf{x} + \mathbf{B} \mathbf{u} \right). \quad (8.54)$$

Then Hamilton's equations become:

$$\begin{split} \ddot{\mathbf{x}} &= \left(\frac{\partial \mathcal{H}}{\partial \lambda}\right)^* = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} & \mathbf{x}(0) = \mathbf{x}_0 \\ &- \dot{\boldsymbol{\lambda}} = \left(\frac{\partial \mathcal{H}}{\partial \mathbf{x}}\right)^* = \mathbf{Q}\mathbf{x} + \mathbf{A}^*\boldsymbol{\lambda} & \boldsymbol{\lambda}(t_f) = \mathbf{Q}_f \mathbf{x}(t_f). \end{split}$$

Again, this is a two-point boundary value problem in x and λ . Plugging in the same expression $\lambda = Px$ will result in the same Riccati equation as above.

8.5 Optimal Full-State Estimation: The Kalman Filter

The optimal LQR controller from Section 8.4 relies on full-state measurements of the system. However, full-state measurements may either be prohibitively expensive or technologically intenable to obtain, especially for high-dimensional systems. The computational burden of collecting and processing full-state measurements may also introduce unaccepttable time delays that will limit robust performance.

Instead of measuring the full state x, it may be possible to estimate the state from limited noisy measurements. J. fact, full-state estimation is mathematically possible as long at the pair (AC,) are observable, although the effectiveness of estimation depends on the depert observable) say squartifiely the deservability Granian. The Kalama filler [27], 551, 2211 is the most commonly used full-state estimator, as it optimally blankers the competing effects or homesurement noise, disturbances, and no departually compared and the shown in the next section, it is possible to use the full-state estimate from a Kalman filter in incontancion with the original full-state loss for Reshark law.

When deriving the optimal full-state estimator, it is necessary to re-introduce disturhances to the state wy and sensor noise way

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u} + \mathbf{w}_d \qquad (8.56a)$$

$$y = Cx + Du + w_a$$
. (8.56b)

The Kalman filter assumes that both the disturbance and noise are zero-mean Gaussian processes with known covariances:

$$\mathbb{E}(\mathbf{w}_{d}(t)\mathbf{w}_{d}(\tau)^{*}) = \mathbf{V}_{d}\delta(t - \tau),$$
 (8.57a)

$$\mathbb{E}(\mathbf{w}_{n}(t)\mathbf{w}_{n}(\tau)^{*}) = \mathbf{V}_{n}\delta(t - \tau).$$
 (8.57b)

Here E is the expected value and $\delta(\cdot)$ is the Dirac delta function. The matrices V_d and V₈ are positive semi-definite with entries containing the covariances of the disturbance and noise terms. Extensions to the Kalman filter exist for correlated, biased, and unknown noise and disturbance terms [498-372]

It is possible to obtain an estimate x of the full-state x from measurements of the input u and output y, via the following estimator dynamical system:

$$\frac{d}{dt}\hat{\mathbf{x}} = \mathbf{A}\hat{\mathbf{x}} + \mathbf{B}\mathbf{u} + \mathbf{K}_{f}(\mathbf{y} - \hat{\mathbf{y}}) \qquad (8.58a)$$

$$\hat{y} = C\hat{x} + Du.$$
 (8.58b)

The matrices A, B, C, and D are obtained from the system model, and the filter gain K (is determined via a similar procedure as in LQR. K (is given by

$$K_f = Y C^* V_s$$
, (8.59)

where y is the solution to another algebraic Riccati equation:

$$YA^* + AY - YC^*V_-^{-1}CY + V_d = 0.$$
 (8.60)

This solution is commonly referred to as the Kalman filter, and it is the optimal full-state estimator with respect to the following cost function:

$$J = \lim_{t\to\infty} \mathbb{E}\left(\left(\mathbf{x}(t) - \hat{\mathbf{x}}(t)\right)^* \left(\mathbf{x}(t) - \hat{\mathbf{x}}(t)\right)\right). \quad (8.61)$$

This cost function implicitly includes the effects of disturbance and noise, which are required to determine the optimal balance between appressive estimation and noise attenuation. Thus, the Kalman filter is referred to as linear awadratic estimation (LOE). and has a dual formulation to the LQR optimization. The cost in (8.61) is computed as an ensemble average over many realizations.

The filter gain K (may be determined in Matlab via

Optimal control and estimation are mathematical dual problems, as are controllability and observability, so the Kalman filter may also be found using LOR:

The Kalman filter is shown schematically in Fig. 8.10.



Figure 8.10 Schematic of the Kalman filter for full-state estimation from noisy measurements $y = Cx + w_0$ with process noise (disturbances) w_d . This diagram does not have a feed/through term D. although it may be included.

Substituting the output estimate ŷ from (8,58b) into (8,58a) yields:

$$\frac{d}{dt}\ddot{\mathbf{x}} = (\mathbf{A} - \mathbf{K}_f \mathbf{C})\dot{\mathbf{x}} + \mathbf{K}_f \mathbf{y} + (\mathbf{B} - \mathbf{K}_f \mathbf{D})\mathbf{u} \qquad (8.62a)$$

$$= (\mathbf{A} - \mathbf{K}_f \mathbf{C}) \hat{\mathbf{x}} + [\mathbf{K}_f, (\mathbf{B} - \mathbf{K}_f \mathbf{D})] \begin{bmatrix} \mathbf{y} \\ \mathbf{u} \end{bmatrix}.$$
 (8.62b)

The estimator dynamical system is expressed in terms of the estimate \hat{x} with inputs y and u if the system is observable it is possible to place the eigenvalues of $A = K_{C}$ arbitrarily with choice of K_{f} . When the eigenvalues of the estimator are stable, then the state estimate \hat{x} correspons to the full-state x symptotically, also gas as the model faithfull coputers the true system dynamics. To see this convergence, consider the dynamics of the estimation error $et = x = x^{-1}$.

$$\begin{split} & \frac{d}{dt}e = \frac{d}{dt}\mathbf{x} - \frac{d}{dt}\hat{\mathbf{x}} \\ & = (\mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{a} + \mathbf{w}_c) - \left[(\mathbf{A} - \mathbf{K}_f \mathbf{C})\hat{\mathbf{x}} + \mathbf{K}_f \mathbf{y} + (\mathbf{B} - \mathbf{K}_f \mathbf{D})\mathbf{u} \right] \\ & = \mathbf{A}e + \mathbf{w}_c + \mathbf{K}_f \mathbf{C}\hat{\mathbf{x}} - \mathbf{K}_f \mathbf{H} \mathbf{D}\mathbf{u} \\ & = \mathbf{A}e + \mathbf{w}_c + \mathbf{K}_f \mathbf{C}\hat{\mathbf{x}} - \mathbf{K}_f \frac{|\mathbf{C}\mathbf{x} + \mathbf{D}\mathbf{u} + \mathbf{w}_c|}{2} + \mathbf{K}_f \mathbf{D}\mathbf{u} \\ & = (\mathbf{A} - \mathbf{K}_f \mathbf{C})\mathbf{e} + \mathbf{w}_c - \mathbf{K}_f \mathbf{w}_c. \end{split}$$

Therefore, the estimate k will converge to the true full state when $A - K_{f}C$ has stable eigenvalues. As with LQR, there is a tradeoff between over-stabilization of these eigenvalues and the amplification of sensor noise. This is similar to the behavior of an inexperienced driver who may hold the steering wheel too tightly and will overreact to every minor bump and disturbance on the road.

There are many variants of the Kalman filter for nonlinear systems [724, 275, 303] including the extension and uncextend Kalman filters. The senselber Kalman filter [14] is an extension that works well for high-dimensional systems, such as in geophysical data similation [149]. All of these methods will assume Cassions modes processes, and the particle filters provides a store general, although more experimentally interview. Interview biliness the efficiency of the Kalman filter and accenters of the nortice filter.

8.6 Optimal Sensor-Based Control: Linear Quadratic Gaussian (LOG)

The full-state estimate from the Kalman filter is generally used in conjunction with the full-state feedback control law from LQR, resulting in optimal sensor-based feedback. Remarkably, the LQR gain K_s and the Kalman filter gain K_f may be designed separately, and the resulting sensor-based feedback will remain optimal and retain the closed-loop eigenvalues when combined.

Combining the LQR full-state feedback with the Kalman fitter full-state estimator results in the linear-quadratic Gaussian (LQG) controller. The LQG controller is a dynamical system with inor v. output u. and internal state k:

$$\frac{d}{dt}\hat{\mathbf{x}} = (\mathbf{A} - \mathbf{K}_f \mathbf{C} - \mathbf{B}\mathbf{K}_r)\hat{\mathbf{x}} + \mathbf{K}_f \mathbf{y} \qquad (8.63a)$$

$$u = -K_f \dot{x}$$
. (8.63b)

The LQG controller is optimal with respect to the following ensemble-averaged version of the cost function from (8.44):

$$J(t) = \left\langle \int_{0}^{t} \left[\mathbf{x}(\tau)^{*} \mathbf{Q} \mathbf{x}(\tau) + \mathbf{u}(\tau)^{*} \mathbf{R} \mathbf{u}(\tau) \right] d\tau \right\rangle. \quad (8.64)$$

The controller $\mathbf{u} = -\mathbf{K}_x \hat{\mathbf{x}}$ is in terms of the state estimate, and so this cost function must be averaged over many realizations of the disturbance and noise. Applying LQR to $\hat{\mathbf{x}}$ results in the following state dynamics:

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} - \mathbf{B}\mathbf{K}_{t}\hat{\mathbf{x}} + \mathbf{w}_{d} \qquad (8.65a)$$

$$= Ax - BK_{r}x + BK_{r}(x - \hat{x}) + w_{d}$$
 (8.65b)

$$= \mathbf{A}\mathbf{x} - \mathbf{B}\mathbf{K}_r\mathbf{x} + \mathbf{B}\mathbf{K}_r\mathbf{\epsilon} + \mathbf{w}_d. \quad (8.65c)$$

Again $\epsilon = \mathbf{x} - \hat{\mathbf{x}}$ as before. Finally, the closed-loop system may be written as

$$\frac{d}{dt}\begin{bmatrix} \mathbf{x} \\ \mathbf{\epsilon} \end{bmatrix} = \begin{bmatrix} \mathbf{A} - \mathbf{B}\mathbf{K}_r & \mathbf{B}\mathbf{K}_r \\ \mathbf{0} & \mathbf{A} - \mathbf{K}_f \mathbf{C} \end{bmatrix} \begin{bmatrix} \mathbf{x} \\ \mathbf{\epsilon} \end{bmatrix} + \begin{bmatrix} \mathbf{I} & \mathbf{0} \\ \mathbf{I} & -\mathbf{K}_f \end{bmatrix} \begin{bmatrix} \mathbf{w}_d \\ \mathbf{w}_u \end{bmatrix}. \quad (8.66)$$



Figure 8.11 Schematic illustrating the linear quadratic Gaussian (LQG) controller for optimal closed-loop feedback based on noisy measurements y. The optimal LQR and Kalman filter gain matrices K_r and K_f may be designed independently, based on two different algebraic Riccati countions. When combined, the resulting sensor-based feedback remains softimal.

Thus, the closed-loop eigenvalues of the LQG regulated system are given by the eigenvalues of $A - BK_c$ and $A - K_f C$, which were optimally chosen by the LQR and Kahman filter gain matrices, respectively.

The LQG framework, shown in Fig. 8.11, relies on an accurate model of the system and knowledge of the magnitudes of the distributions can discussment with what are assumed to be Gaussian processor. In real-world system, each of these assumptions may be model, and even sum lime delays and model meetinging any descent protocol and the strength of the strength of the strength of the strength of LQG and reach in insubility (155). The lack of robustness of LQG regulators to model assumed to the strength of the strength of the strength of the strength discretion meeting and the strength of the strength of the strength of the discretion meeting and the strength of t

In contrast to classical control approaches, such as proportional-integral-derivative (PID) control and designing faster inner-loop control and slow outer-loop control assuming a separation of timescales, LQG is able to handle multiple-input, multiple output (MIMO) systems with overlapping timescales and multi-objective cost functions with no additional combedivi in the aborithm or immementation.

8.7 Case Study: Inverted Pendulum on a Cart

To consolidate the concepts of optimal control, we will implement a stabilizing controller for the inverted pendulum on the cart, shown in Fig. 8.12. The full nonlinear dynamics are given by

$$\dot{x} = v$$
 (8.67a)

$$\dot{v} = \frac{-m^2 L^2 g \cos(\theta) \sin(\theta) + mL^2 (mL\omega^2 \sin(\theta) - \delta v) + mL^2 u}{mL^2 (M + m(1 - \cos(\theta)^2))}$$
(8.67b)





Figure 8.12 Schematic of inverted pendulum on a cart. The control forcing acts to accelerate or decelerate cart. For this example, we assume the following parameter values: pendulum mass (m = 1), cart mass (M = 5), pendulum length (L = 2), gravitational acceleration (g = -10), and cart damping ($\delta = 1$).

$$\dot{\theta} = \omega$$

$$(8.67c)$$
 $\dot{\omega} = \frac{(m + M)mgL\sin(\theta) - mL\cos(\theta)(mL\omega^2\sin(\theta) - \delta v) + mL\cos(\theta)u}{mL^2(M + m(1 - \cos(\theta)^2))}$

$$(8.67d)$$

where x is the cart position, v is the velocity, θ is the pendulum angle, ω is the angular velocity, m is the pendulum mass, M is the cart mass, L is the pendulum arm, g is the gravitational acceleration, δ is a friction damping on the dart, and u is a control force applied to the cart.

The following Matlab function, pendcart, may be used to simulate the full nonlinear system in (8.67):

Code 82 Right-hand side function for inverted pendulum on cart.

```
function dx = pendcart(x, m, M, L, q, d, u)
Cx = cos(x(3));
dx(1,1) = x(2):
dx(2,1) = (1/D) * (-n^2*L^2*g*Cx*Sx + n*L^2*(n*L*x(4))^2*Sx - d*x
  (2))) + m+L+L+(1/D)+u;
dx(3,1) = x(4):
```

There are two fixed points, corresponding to either the pendulum down ($\theta = 0$) or pendulum up ($\theta = \pi$) configuration: in both cases, $y = \omega = 0$ for the fixed point, and the cart position x is a free variable, as the equations do not depend explicitly on x. It is possible to linearize the equations in (8.67) about either the up or down solutions, vielding the following linearized dynamics:
$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \end{bmatrix} = \begin{bmatrix} 0 & 1 & 0 & 0 \\ 0 & -\frac{d}{M} & b\frac{\partial w}{M} & 0 \\ 0 & 0 & 0 & 1 \\ 0 & -b\frac{d}{M} & -b\frac{\partial w + M }{M} & 0 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \end{bmatrix} + \begin{bmatrix} x_1 \\ 0 \\ 0 \\ \frac{d}{M} \end{bmatrix} u, \text{ for } \begin{bmatrix} x_1 \\ x_3 \\ x_4 \end{bmatrix} = \begin{bmatrix} v \\ \theta \\ \theta \\ \theta \end{bmatrix},$$

where b = 1 for the pendulum up fixed point, and b = -1 for the pendulum down fixed point. The system matrices A and B may be entered in Matlab using the values for the constants given in Fig. 8.12:

Code 8.3 Construct system matrices for inverted pendulum on a cart.

clear all, close all, clc m = 1; M = 5; L = 2; g = -10; d = 1; b = 1; # Penchulum up (bw1) A = [0 1 = 0; 0; 0 = -3/k baimsg/M 0; 0 = 0 - 2; 0 = 0 - 1; 0 = 0 - 0; 0 = 0 - 0; 0 = 0 - 0; 0 = 0 - 0; 0 = 0; 0

We may also confirm that the open-loop system is unstable by checking the eigenvalues of A:

```
>> lambda = eig(A)
lambda =
-2.4311
-0.2336
-0.6665
```

In the following, we will test for controllability and observability, develop full-state feedback (LQR), full-state estimation (Kalman filter), and sensor-based feedback (LQG) solutions.

Full-state Feedback Control of the Cart-Pendulum

In this section, we will design an LQR controller to stabilize the inverted pendulum configuration ($\theta = \pi$) assuming full-state measurements, $\mathbf{y} = \mathbf{x}$. Before any control design, we must confirm that the system is linearly controllable with the given A and B matrices:

```
>> rank(ctrb(A,B))
ans =
```

Thus, the pair (A, B) is controllable, since the controllability matrix has full rank. It is then possible to specify given Q and R matrices for the cost function and design the LQR controller axis matrix K:

Cade 8.4 Design LQR controller to stabilize inverted pendulum on a cart.

```
20 Design LQR controller
Q = eye(4); % 4x4 identify matrix
```



Figure 8.13 Closed-loop system response of inverted pendulum on a cart stabilized with an LQR controller.

R = .0001; K = lqr(A,B,Q,R);

We may then simulate the closed-loop system response of the full nonlinear system. We will initialize our simulation slightly off equilibrium, at $x_0 = \begin{bmatrix} -1 & 0 & \pi + 1 & 0 \end{bmatrix}^2$, and we also impose a desired step change in the reference position of the cart, from x = -1 to x = 1.

Code 8.5 Simulate closed-loop inverted pendulum on a cart system.

```
b% fimulate closed-loop system
trpam = 0:001:10;
x0 = [-1; 0; pi+.1; 0]; % initial condition
wr = [1; 0; pi, 0]; % reference position
uw0(x)-X*(x - wr); % control law
[t,x] = odd*(0(t,x), q,d,u(x)), tmpan, x0);
```

In this code, the actuation is set to:

$$u = -\mathbf{K} (\mathbf{x} - \mathbf{w}_{r}),$$
 (8.69)

where $\mathbf{w}_r = \begin{bmatrix} 1 & 0 & \pi & 0 \end{bmatrix}^T$ is the reference position. The closed-loop response is shown in Fig. 8.13.

In the above procedure, specifying the system dynamics and simulating the closed-loop system response is considerably more involved than a statuly designing the controller, which amounts to a single function call in Mattah. It is also helpful to compare the LQR response to the response obtained by mosophimal eigenvalue platement. In particular, Fig. 8.14 dows the system response and cost function for 100 nandomly generated sets of stable eigenvalues. A close in the interval -1.5.-5...5.-5.1 The LQR controller has the lower overfall coat, as it is chosen to minimize J. The code to fort the pendulum-cast system is reorided online.

Non-minimum phase systems It can be seen from the response that in order to move from x = -1 to x = 1, the system initially moves in the wrong direction. This behavior



Figure 8.14 Comparison of LQR controller response and cost function with other pole placement locations. Bold lines represent the LOR solutions.

indicates that the system is non-minimum phase, which introduces challenges for robust control, as we will soon see. There are many examples of non-minimum phase systems in control. For instance, parallel parking an automobile first involves moving the center of mass of the car away from the curb before it then moves closer. Other examples include increasing altitude in an aircraft, where the elevators must first move the center of mass down to increase the angle of attack on the main wings before lift increases the altitude. Adding cold fuel to a turbine may also initially drop the temperature before it eventually increases.

Full-State Estimation of the Cart-Pendulum

Now we turn to the full-state estimation problem based on limited noisy measurements S_1 For this example, we will develop the Kaiman filter (for the pendulum own condition $(\theta = 0)$, since without feedback the system in the pendulum-ap condition will quickly leave the fixed point where the linear model is valid. When we combine the Kaiman filter with LQB in the text example, it will be possible to control to the usuable inserted pendulum confuturation. Switching to the readmain down continuous is similar in the code:

b = -1: # pendulum down (b=-1)

Before designing a Kalman filter, we must choose a sensor and test for observability. If we measure the cart position, $y = x_1$,

C = [1 0 0 0]; # measure cart position, x

then the observability matrix has full rank:

```
>> rank(obav(A,C))
```

Because the cart position ar, does not appear explicitly in the dynamics, the system is not fully observable for any measurement that doesn't include x₁. Thus, it is impossible to estimate the cart position with a measurement of the pendalum angle. However, if the eart position is not important for the work function (*i.e.*, if we only want to sublitze the pendalum, and don't care where the cart is located), then other choices of sensor will be admissible.

Now we design the Kalman filter, specifying disturbance and noise covariances:

```
H* Opecify disturbance and noise magnitude
VM eve(4): # disturbance covariance
Ym e 1; # disturbance covariance
# Build Kalman filter
(Kf.P.K] = log(A,eve(4), C.Vd.Ym); # design Kalman filter
# alternatively, possible to design using "LGR" code
K = (Lgr(A', C', Vd, Ym)');
```

The Kalman filter gain matrix is given by

```
Xf =
1.9222
1.3474
-0.6182
-1.8016
```

Finally, to simulate the system and Kalman filter, we must augment the original system to include disturbance and noise inputs:

```
If Augenci system with additional inputs

E_{samp} = (E_{sym}(d) \otimes O_{2}), i () read Deen

E_{samp} = (E_{sym}(d) \otimes O_{2}), i () read Deen

E_{samp} = (O \otimes O \otimes O_{2}), i () read Deen

system sets E_{samp} = E_{samm} = E_{samm} = E_{samm} = E_{samm} = E_{samm} = E_{samm} = E_{sa
```

We now simulate the system with a single output measurement, including additive disturbances and noise, and we use this as the input to a Kalman filter estimator. At time t = 1and t = 15, we give the system a large positive and negative impulse in the actuation, respectively.

```
11 Definite linearies any entry in "deve" position

+ attack

+ attack

att
```

Fig. 8.15 shows the noisy measurement signal used by the Kalman filter, and Fig. 8.16 shows the full noiseless state, with disturbances, along with the Kalman filter estimate.

To build intuition, it is recommended that the reader investigate the performance of the Kalman filter when the model is an imperfect representation of the simulated dynamics. When combined with full-state control in the next section, small time delays and changes to the system model may cause fragility.

Sensor-Based Feedback Control of the Cart-Pendulum

To apply an LQG regulator to the inverted pendulum on a curt, we will simulate the full nonlinear system in Simulita, as aboven in Fig. 8.17. The nonlinear dynamics are encapsulated in the block' varigend_sim, and the inputs consist of the actuation signal and disturbance were listed to fit states for performance analysis, shihough only and disturbance were listed to the state for the performance analysis, shihough the fifter. The full-state estimates is then passed to the LQB Mock, which commands the disturbance accuration signal. For the cample, we use the following LQB and LQB weighting matrices:

```
Q = eye(4); 

R = .000001; 

Vd = .04,eye(4); 

Va = .00c2; 

noise covariance
```



Figure 8.15 Noisy measurement that is used for the Kalman filter, along with the underlying noiseless signal and the Kalman filter estimate.



Figure 8.16 The true and Kalman filter estimated states for the pendulum on a cart system.

The system starts near the vertical equilibrium, at $\mathbf{x}_0 = \begin{bmatrix} 0 & 0 & 3.14 & 0 \end{bmatrix}^7$, and we command step in the cart position from x = 0 to x = 1 at t = 10. The resulting response is shown in Fig. 33. IS begine noisy measurements (Fig. 8.19) and distrubunce (Fig. 8.20), the controller is able to effectively track the reference cart position while stabilizing the inverted pendudum.



Figure &17 Matlab Simulink model for sensor-based LQG feedback control.



Figure 8.18 Output response using LQG feedback control.

8.8 Robust Control and Frequency Domain Techniques

Until now, we have described control systems in terms of state-space systems of ondinary differential equations. This perspective readily lends itself to stability analysis and design via placement of closed-loop eigenvalues. However, it a seminal paper by John Doyle in 1978 [155], it was shown that LQG regulators can have arbitrarily small stability margins, making them *fraperio* to model uncertainties, itme designs, and other model imperfections.

Fortunately, a short time after Doyle's famous 1978 paper, a rigorous mathematical theory was developed to design controllers that promote robustness. Indeed, this new theory

5 Title: Guaranteed margins for LQG regulators; Abstract: There are none.



Figure 8.19 Noisy measurement used for the Kalman filter, along with the underlying noiseless simal and the Kalman filter estimate.



Figure 8.20 Noisy measurement used for the Kalman filter, along with the underlying noiseless simula and the Kalman filter estimate.

of robust control generalizes the optimal control framework used to develop LQR/LQG, by incorporating a different cost function that penalizes worse-case scenario performance.

To understand and design controllers for robust performance, it will be helpful to look at frequency domain transfor functions of various signals. In particular, we will consider the sensitivity, complementary sensitivity, and loop transfer functions. These enable quantistative and visual approaches to assess robust performance, and they enable intuitive and compact representations of control systems.

Robust control is a natural perspective when considering uncertain models obtained from noisy or incomplete data. Moreover, it may be possible to manage system nonlinearity as a form of structured model uncertainty. Finally, we will discuss known factors that limit robust performance, including time delays and non-minimum phase behavior.

Frequency Domain Techniques

To understand and manage the tradeoffs between robustness and performance in a control system, it is helpful to design and analyze controllers using frequency domain techniques.

The Laplace transform allows us to go between the time-domain (state-space) and frequency domain:

$$\mathcal{L}{f(t)} = f(s) = \int_{0^{-}}^{\infty} f(t)e^{-st}dt.$$
 (8.70)

Here, *i* is the complex-valued Laplace variable. The Laplace transform may be hough or as a one-sided generational Fourier transform that is valid for functions that don't converge to zero as $t \rightarrow \infty$. The Laplace transform is particularly useful because it transforms ifferential equations in adjective capations, and convolvation imegrals in the time domain become simple products in the frequency domain. To see how time derivatives pass through the Laplace transform, we use integrations by parts:

$$\mathcal{L} \left\{ \frac{d}{dt} f(t) \right\} = \int_{0^{-\infty}}^{\infty} \frac{d}{dt} \frac{f(t)}{t} e^{-s} dt$$

$$= \left[f(t)e^{-s} \right]_{sd0^{--}}^{sm} - \int_{0^{-\infty}}^{\infty} f(t)(-se^{-st}) dt$$

$$= f(0^{--}) + s\mathcal{L}[f(t)].$$

Thus, for zero initial conditions, $\mathcal{L}[df/dt] = sf(s)$.

Taking the Laplace transform of the control system in (8.10) yields

$$sx(s) = Ax(s) + Bu(s)$$
 (8.71a)

$$y(s) = Cx(s) + Du(s).$$
 (8.71b)

It is possible to solve for $\mathbf{x}(s)$ in the first equation, as

$$(s\mathbf{I} - \mathbf{A})\mathbf{x}(s) = \mathbf{B}\mathbf{u}(s) \implies \mathbf{x}(s) = (s\mathbf{I} - \mathbf{A})^{-1}\mathbf{B}\mathbf{u}(s).$$
 (8.72)

Substituting this into the second equation we arrive at a manning from inputs u to outputs v:

$$\mathbf{y}(s) = \left[\mathbf{C}(s\mathbf{I} - \mathbf{A})^{-1}\mathbf{C} + \mathbf{D}\right]\mathbf{u}(s). \quad (8.73)$$

We define this mapping as the transfer function:

$$G(s) = \frac{y(s)}{u(s)} = C (sI - A)^{-1} B + D.$$
 (8.74)

For linear systems, there are three equivalent representations: 1) time-dsmain, in terms of the impulse response; 2) frequency domain, in terms of the transfer function; and 3) start-apace, in terms of a system of differential equations. These representations are shown schematically in Fig. 82.1. As we will see, there are many benefits to analyzing control systems in the frequency domain.

Frequency Response

The transfer function in (8.74) is particularly useful because it gives rise to the frequency response, which is a graphical representation of the control system in terms of measurable duat. To illustrate this, we will consider a single-input (SISO) system. It is a



Figure 8.21 Three equivalent representations of linear time invariant systems.

property of linear systems with zero initial conditions, that a sinusoidal input will give rise to a sinusoidal output with the same frequency, perhaps with a different magnitude A and phase 6:

$$u(t) = sin(\omega t) \implies v(t) = A sin(\omega t + \phi),$$
 (8.75)

This is true for long-times, after initial transients die out. The amplitude A and phase ϕ of the output sinusoid depend on the input frequency so. These functions $A(\omega)$ and $\phi(\omega)$ may be mapped out by running a number of experiments with sinusoidal input at different frequencies so. Alternatively, this information is obtained from the complex-valued transfer function $G(\omega)$:

$$A(\omega) = [G(i\omega)], \qquad \phi(\omega) = \langle G(i\omega). \qquad (8.76)$$

Thus, the amplitude and phase angle for input $sin(\omega t)$ may be obtained by evaluating the transfer function at $s = i\omega$ (i.e., along the imaginary axis in the complex plane). These quantities may then be plotted, resulting in the frequency response or *Bode plot*.

For a concrete example, consider the spring-mass-damper system, shown in Fig. 8.22. The equations of motion are given by:

$$m\ddot{x} = -\delta\dot{x} - kx + \mu$$
 (8.77)

Choosing values m = 1, $\delta = 1$, k = 2, and taking the Laplace transform yields:

$$G(s) = \frac{1}{s^2 + s + 2}.$$
 (8.78)

Here we are assuming that the output y is a measurement of the position of the mass, x. Note that the denominator of the transfer function G(y) is the characteristic equation of (8.77), written in state-space form. Thus, the poles of the complex function G(x) are eigenvalues of the state-space system.



Figure 8.22 Sprine-mass-damper system.



Figure 8.23 Frequency response of spring-mass-damper system. The magnitude is plotted on a logarithmic scale, in units of decibel (dB), and the frequency is likewise on a log-scale.

It is now possible to create this system in Mattha and plot the frequency response, and shown in Fig. 82.3 Note that the frequency response is readily interpretable and provides physical intuition. For example, the zero stope of the magnitude at the relief of matchares that show frequent granulates after hybrid physical intuition. The intuition of the stope of the mass while the relief of the stope of the most of the mass. Moreover, the resonance frequency is seen as peak in the magnitude, indication and matching and the frequency.

Cade 8.8 Create transfer function and plot frequency response (Bode) plot.

Given a state-space realization,

>> A = [0 1; -2 -1]; >> B = [0; 1];

```
p = c = (1 = 0)
> 0 = 0 = 0
It is imple to obtain A frequency domain representation:
(s) (may, dm) = set(0, Sp, C, 0) + 0 tests space to transf. fun.
so 0 = tfound, dm) = Create transfer function
0 = 1
1 = 1
1 = 1
1 = 1
1 = 1
```

Similarly, it is possible to obtain a state-space system from a transfer function, although this representation is not unique:

Notice that this representation has switched the ordering of our variables to $\mathbf{x} = \begin{bmatrix} v & x \end{bmatrix}^T$, although it still has the correct input–output characteristics.

The frequency-domain is also useful because impulsive or step inputs are particularly simple to represent with the Laplace transform. These are also simple in Matlab. The impulse response (Fig. 8.24) is given by

and the step response (Fig. 8.25) is given by

>> step(G);
 \$ Step response

Performance and the Loop Transfer Function: Sensitivity and Complementary Sensitivity

Consider a slightly modified version of Fig. 8.4, where the disturbance has a model, \mathbf{P}_d . This new diagram, shown in Fig. 8.26, will be used to derive the important transfer functions relevant for assessing robust performance.

$$y = GK(w_r - y - w_n) + G_d w_d$$
 (8.79a)

$$\implies$$
 $(I + GK)y = GKw_r - GKw_n + G_dw_d.$ (8.79b)

$$\implies y = \underbrace{(\mathbf{I} + \mathbf{GK})^{-1}\mathbf{GK}}_{\mathbf{T}} \mathbf{w}_r - \underbrace{(\mathbf{I} + \mathbf{GK})^{-1}\mathbf{GK}}_{\mathbf{T}} \mathbf{w}_n + \underbrace{(\mathbf{I} + \mathbf{GK})^{-1}}_{\mathbf{S}} \mathbf{G}_d \mathbf{w}_d. \quad (8.79c)$$

Here, S is the sensitivity function, and T is the complementary sensitivity function. We may denote L = GK the loop transfer function, which is the open-loop transfer function in the



Figure 8.24 Impulse response of spring-mass-damper system.



Figure 8.25 Step response of spring-mass-damper system.

absence of feedback. Both S and T may be simplified in terms of L:

$$S = (I + L)^{-1}$$
 (8.80a)

$$T = (I + L)^{-1}L.$$
 (8.80b)

Conveniently, the sensitivity and complementary sensitivity functions must add up to the identity: $\mathbf{S} + \mathbf{T} = \mathbf{I}$.

In practice, the transfer function from the exogenous inputs to the noiseless error 6 is more useful for design:

$$\epsilon = \mathbf{w}_r - \mathbf{y} = \mathbf{S}\mathbf{w}_r + \mathbf{T}\mathbf{w}_n - \mathbf{S}\mathbf{G}_d\mathbf{w}_d. \quad (8.81)$$



Figure 8.28 Closed-loop feedback control diagram with reference input, noise, and disturbance. We will consider the various transfer functions from exogenous inputs to the error e, thus deriving the loop transfer function, as well as the sensitivity and complementary sensitivity functions.



Figure 8.27 Loop transfer function along with sensitivity and complementary sensitivity functions.

Thus, we see that the sensitivity and complementary sensitivity functions provide the maps from reference, disturbance, and noise inputs to the tracking error. Since we deaire small tracking error, we may then specify **S** and **T** to have desirable properties, and ideally we will be able to ablive these specifications by designing the loop transfer function **L** in practice, we will choose the coartoller **K** with insortedge of the model **G** so that the loop transfer function has beneficial properties in the frequency domain. For example, small transfer function has beneficial properties in the frequency domain. gain at high frequencies will attenuate sensor noise, since this will result in T being small. Similarly, high gain at low frequencies will provide good reference tracking performance, as S will be small at low frequencies. However, S and T cannot both be small everywhere, since S + T = L from (8.30), and so these design objectives may compete.

For performance and obtainess, we want the maximum peak of K $M_{H} = |\mathbf{S}|_{h_{H}}$. Do its mining a possible (From (31)) in its cert has the tabence of most K. Defined, constrained the rest $\mathbf{T} = K$ and $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = K$) and $\mathbf{T} = K$. The rest $\mathbf{T} = K$ is the rest $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = X$), which we have not the rest $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = X$), which we have $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = X$). The rest $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = X$), which we have $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = X$). The rest $\mathbf{T} = K$ is the rest $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = K$) and $\mathbf{T} = K$ are planess in $|\mathbf{T}|^2$ ($\mathbf{T} = K$). The rest $\mathbf{T} = K$ is the rest \mathbf{T}

The controller handwidth way is the frequency below which feedback control is effective. This is a subjected definition. Often, and is the frequency where $S(i, \alpha)$ filter corses -3 dB from below. We would ideally like the controller bandwidth to be as large as possible without amplifying scores noise, which hypically has a slip frequency. However, there are fundamental bandwidth limitations that are imposed for systems that have time delays or right half of the reso 1920.

Inverting the Dynamics

What is model of the form in (13) or (17.2), it may be possible to obtain an equivalence of the state of the

For open-loop control using system inversion, **G** must also be table. It is impossible to indumentally change the dynamics of a linear system through appen-loop control, and then an unable system cannot be stabilized without feedback. Attempting to stabilize an unable system by interring the dynamics of with pysical byte distances consequences. For instance, consider the following unstable system with a pole $a_I = s$ and a zero at $a_I = -0$. (G) $a_I = 10$ (G) $a_I = 10$, (G) $a_I = 10$,

$$G_{trac}(s)K(s) = \frac{s-5}{s-5+\epsilon}$$

This system is still unstable, despite the attempted pole cancelation. Moreover, the unstable mode is now nearly unobservable.

In addition to sublity, G must not have any time delays or zeros in the right-ful figure. The start of the start star

Combined, these restrictions on G imply that model-based open-loop control should only be used when the system is well-behavior, accurately characterized by a model, when distribances are characterized, and when the additional feedback control hardware is unnecessarly expensive. Oblewise, performance goals must be model. Open-loop model inversion is often used in manufacturing and robotics, where systems are well-characterized and constrained in a stundard operative environment.

Robust Control

As discussed previously, LQG controllers are known to have arbitrarily poor robustness margins. This is a serious problem in systems such as turbulence control, neuromechanical systems, and epidemiology, where the dynamics are wrought with uncertainty and time delays.

Fig. 5.2 shows the most general schematic for closel-loop feedback counds, neuronsmalls theorem and mosts counds antigraphics. In the generalized loop of modern domains and the schematic schematic schematic schematic schematic schematic disturbances, noise, etc.) to an antih-schecher cost lineation [21] accuracy, assumption costdisturbances, etc.). Schematic control (eq. 1), RE, LQE (12) is optimal with the repect to the R_{12} neuron, a bounded from the repect to the R_{12} however, Rest, consisting of stable and is initially optimal with respect to the N_{n2} bounded linking-scene, consisting of stable and its initially optimal with respect to the N_{n2} bounded linking-scene, consisting of stable and its initially optimal with respect to the N_{n2} bounded linking-scene, consisting of stables and the most indefined as:

$$\|\mathbf{G}\|_{\infty} \triangleq \max \sigma_1 (\mathbf{G}(i\omega))$$
. (8.82)

Here, a_i denotes the maximum singular value. Since the $\| \cdot \|_{a_i}$ norm is the maximum value of the transfer function at any frequency, it is often called a worst-actual constraints of the singular transfer and the singular transfer and the nobstates is important. There are many connections between H_2 and H_{a_i} counted, as they exist within the same framework and gravity in the singular transfer and trans

If we let G_{n-1} denote the transfer function from we 0, 1, then the goal of W_{n-2} control to construct a consolitor to maintaine the initialy norm: mile $(G_{n-2})_{n-1}$. This is typiccally difficult, and no analytic closed-form solution exists for the optimal controller in general. However, them are learitively efficient intenivie methods to find a controller with the $|G_{n-2}|_{n-2} \sim \tau$, as described in [156]. There are numerous conditions and cavears that describe when this method can be used in addition, there are comparisonily officient algorithms implemented in both Mattih and Python, and these methods require relatively low overhead from the user.

Selecting the cost function J to meet dusp neglectization is a critically important pure of orban cound duspic. Considerations was also disturbance rejection, these attenuations, controller bandwidth, and actuation cost may be accounted for by a weighted sum of the matter functions. The AMS Is the neural and number of these counderations are unated to matching and the balance the relative importance of these counderations are have the sense of the sense to the sense of the sense treports at high-frequency is discourged. A general cost function may consist of the weighting the the sense of the sense sense of the sense of

H	F ₁ S	н.	
ŧ	F ₂ T	1	
	F ₃ KS	Ш.,	

Another possible relates counted design is called $T_{\rm ce}$ besp-sharped $T_{\rm ce}$ by proceedings and the structure structure of the structure structure

 H_2 optimal control (e.g., LQR, LQR, LQR) has been an extremely popular control paradigm because of its single multimetratical formulation and its mushibity by user input. However, the advantages of $H_{4,2}$ control are being increasingly realized. Additionally, there are numerous consume solutions: advantages of the multiple solution of the solution control toolbox. Similarly, keep-damping is accompliabed using the **loopeyn** command in the robust control toolbox.

Fundamental Limitations on Robust Performance

As discussed above, we want to minimize the peaks of S and T to improve robustness. Some peakedness is inevitable, and there are certain system characteristics that significantly limit performance and robustness. Most notably, time delays and right-half plane zeros of the open-loop system will limit the effective control bandwidth and will increase the antianable lower-bound for peaks of S and T. This contributes to both degrading performance and decreasing robustness.

Similarly, a system will suffer from robust performance limitations if the number of poles exceeds the number of zeros by more than 2. These fundamental limitations are quantified in the waterbed integrals, which are so named because if you push a waterbed down in one location, it must rise in a another. Thus, there are limits to how much one can push down neaks in S without causing other neaks to non-un-

Time delays are relatively easy to understand, since a time delay r will introduce an additional phase lag of Tay at the frequency or, limiting how fast the controller can respond effectively (i.e. bandwidth). Thus, the bandwidth for a controller with acceptable phase margins is typically $\omega n < 1/\tau$.

Following the discussion in 14921, these fundamental limitations may be understood in relation to the limitations of open-loop control based on model inversion. If we consider high-gain feedback $\mathbf{u} = \mathbf{K}(\mathbf{w}_r - \mathbf{v})$ for a system as in Fig. 8.26 and (8.81), but without disturbances or noise, we have

$$u = K_{f} = KSw_{f}$$
. (8.83)

We may write this in terms of the complementary sensitivity T, by noting that since T = I - S, we have $T = L(I + L)^{-1} = GKS$

$$u = G^{-1}Tw_{r}$$
. (8.84)

Thus, at frequencies where T is nearly the identity I and control is effective, the actuation is effectively inverting G. Even with sensor-based feedback, perfect control is unattainable. For example, if G has right-half plane zeros, then the actuation signal will become unbounded if the gain K is too aggressive. Similarly, limitations arise with time delays and when the number of poles of G exceed the number of zeros, as in the case of open-loop

As a final illustration of the limitation of right-half plane zeros, we consider the case of proportional control u = Ky in a single-input, single output system with G(s) = N(s)/D(s). Here, roots of the numerator N(s) are zeros and roots of the denominator D(s) are poles. The closed-loop transfer function from reference w, to sensors s is given by:

$$\frac{y(s)}{w_r(s)} = \frac{GK}{1+GK} = \frac{NK/D}{1+NK/D} = \frac{NK}{D+NK}.$$
 (8.85)

For small control gain K, the term NK in the denominator is small, and the poles of the closed-loop system are near the poles of G, given by roots of D. As K is increased, the NK term in the denominator begins to dominate, and closed-loop poles are attracted to the roots of N, which are the open-loop zeros of G. Thus, if there are right-half plane zeros of the open-loop system G, then high-gain proportional control will drive the system unstable. These effects are often observed in the root locus plot from classical control theory. In this way, we see that right-half plane zeros will directly impose limitations on the gain margin

Suggested Reading

Texts

- Feedback Systems: An Introduction for Scientists and Engineers, by K. J. Aström and R. M. Murray, 2010 [22].
- (2) Feedback Control Theory, by J. C. Doyle, B. A. Francis, and A. R. Tannenbaum, 2013 [157].
- (3) Multivariable Feedback Control: Analysis and Design, by S. Skogestad and I. Postlethwaite, 2005 [492].
- (4) A Course in Robust Control Theory: A Convex Approach, by G. E. Dullerud and F. Paganini, 2000 [165].
- (5) Optimal Control and Estimation, by R. F. Stengel, 2012 [501].

Papers and Reviews

 Guaranteed margins for LQG regulators, by J. C. Doyle, IEEE Transactions on Automatic Control, 1978 [155]. Many systems of interest are exceedingly high dimensional, making them difficult to characterize. High dimensionality also limits controller tobutness due to significant computational time delays. For example, for the governing equations of hild dynamics, the reading discretized equations may have millions or Millions of dialess of hild dynamics, the models that capture the most relevant mechanism and are suitable for feedback, control.

Unlike reduced-outer models based on parger orthogonal decomposition (see Chapters 11 and 12), which our drunds based on energy content in the data, here we will discoss a class of holmood reduced-order models that employ a different inner product to order modes based on injust-organger energy. Thus, so only models that are both highly controllible and highly observable are selected, making balanced models ideal for course applications. In this chapter we add beachers deread proceedures for model reduction and system ideatification, depending on whether or not the user starts with a high-fidelity model or simply has access to measurement data.

9.1 Model Reduction and System Identification

In many nonlinear systems, it is still possible to use linear control techniques. For example, in find dynamic three numerous success of their models have flow control (2), 180, 941, for example to delay maniform laminar to turbulent flow in a spatially developing bounding type, to meleta skif-informa laminar to turbulence, and to stabilize the flow pain appen cavity. However, many linear control approaches do not scale well to lange state spaces, and they may be problemby expensive to easily for all easily and the pain spen cavity. However, many linear control approaches do not scale well to on short intercales. Thus, it is often necessary to develop low-dimensional approximations of the system for use in real-time feedbac cornol.

 (DMD; Section 7.2) [472, 456, 535, 317], the eigensystem realization algorithm (ERA; Section 9.3) [272, 351], the observer-kalman fiber identification (OKID; Section 9.3) [273, 428, 271], NARMAX [59], and the sparse identification of nonlinear dynamics (SINDy; Section 7.3) [95].

After a linear model has been identified, effere by model reduction or system is iterativtian itera prise base sources and exciting the linear term and the origination of the linear term of the linear

9.2 Balanced Model Reduction

The high dimensionling and short time-tasks associated with couples systems any predicts the model hand of most images starthylics in Carpert of maintain for any phyinteraction of the start of the interaction, and pay moles either a high dimensional Riccia quarks, see a regress on interaction in high physics, there may maintain dimposited the host does not in the maintain dimension is high, there may all the a 16 solutions of the start of the maintain dimension is high, there may all the a 16 solutions of the start of the maintain dimension is high physics. The start of the start o

Model reduction is essentially data reduction that respects the fact that the data is generated by a dynamic process. If the dynamical system is a linear time-instrum (LT) inputoutput system, then there is a wealth of machinery available for model reduction, and performance bounds may be quantifict. The texchinagoe explored here are based on the singular used accomposition (SVP). Chapter 1) [212, 166, 211], and the minimal realization theory of Ho and Khamin [247, 383]. The general deals is notemine a hearenchaid model decomposition of the system size that may be transited at some model order, only locenin the coherent motivum that are most innormal.

The Goal of Model Reduction

Consider a high-dimensional system, depicted schematically in Fig. 9.1,

$$\frac{d}{dt}\mathbf{x} = \mathbf{A}\mathbf{x} + \mathbf{B}\mathbf{u}, \quad (9.1a)$$

$$y = Cx + Du$$
, (9.1b)



Figure \$1 Input-output system. A control-oriented reduced-order model will capture the transfer function from u to y.

for example from a spatially discretized simulation of a PDE. The primary goal of model reduction is to find a coordinate transformation $\mathbf{x} = \Psi \hat{\mathbf{x}}$ giving rise to a related system ($\Lambda, \mathbf{B}, \hat{\mathbf{C}}, \hat{\mathbf{D}}$) with similar input-output characteristics,

$$\frac{d}{dt}\tilde{\mathbf{x}} = \tilde{\mathbf{A}}\tilde{\mathbf{x}} + \tilde{\mathbf{B}}\mathbf{u},$$
 (9.2a)

$$y = \tilde{C}\tilde{x} + \tilde{D}u$$
, (9.2b)

in terms of a state $\hat{\mathbf{x}} \in \mathbb{R}^d$ with reduced dimension, $r \ll n$. Note that \mathbf{u} and \mathbf{y} are the same in (9.1) and (9.2) even though the system states are different. Obtaining the projection operator Ψ will be the focus of this section.

As a motivating example, consider the following simplified model:

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} -2 & 0 \\ 0 & -1 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 1 \\ 10^{-10} \end{bmatrix} u \quad (9.3a)$$

$$y = \begin{bmatrix} 1 & 10^{-10} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}$$
. (9.3b)

In this case, the state z_2 is barely controllable and barely observable. Simply choosing $i = z_1$ will result in a reduced-order model that faithfull captures the imput-output dynamics. Although the choice $\tilde{x} = z_1$ seems intraitive in this extreme case, many model reduction techniques would encroncourdly from the state $z = z_2$, since it is more lightly damped. Throughout this section, we will investigate how to accurately and efficiently find the transformation matrix Ψ that best captures the impact output dynamics.

The proper outbagonal decomposition [57, 231] from Chapter 11 provides a mutufied mark of the contrast of the second based on every contrast mark of the second based on every contrast mark of the second based on the second ba

¹ When the training data consists of velocity fields, for example from a high-dimensional discretized fluid system, then the singular values literally indicate the latentic energy content of the associated mode. It is common to refer to POD modes as being ordered by *energy* content, even in other applications, although survisors is more technically correct.

Instead ordering modes hade on energy, it is possible to determine a hierarchy of modes hat are non committal and solvershifts, therefore quaring the molt page-support information. These modes gives its to balanced models, giving equal weighting to the controllability and benefative to balanced models using traditional methods have been externed by an external traditional methods and any second second second is publicity experiments. The second second second second second balanced modes methods for efficient comparison of balanced halances procedure, as well as moders methods for efficient comparison of balance distribution of publicity experiments for additional publicity determines and the second second second second second second balances procedure, as well as moders methods for efficient comparison of balances individual second second second second second second second distribution on the total second second

A balanced reduced-order model should map inputs to outputs as faithfully as possible for a given model order r. It is therefore important to introduce an operator norm to quantify how similarly (9.1) and (9.2) act on a given set of inputs. Typically, we take the infinity norm of the difference between the transfer functions G(s) and $G_s(s)$ obtained from the full system (9.1) and reduced system (9.2), respectively. This nom is given by:

$$\|\mathbf{G}\|_{\infty} \triangleq \max \sigma_1 (\mathbf{G}(i\omega)).$$
 (9.4)

See Section 8.8 for a primer on transfer functions. To summarize, we seek a reduced-order model (9.2) of low order, $r \ll n$, so the operator norm $\|\mathbf{G} - \mathbf{G}_r\|_{\infty}$ is small.

Change of Variables in Control Systems

The balanced model reduction problem may be formulated in terms of first finding a coordinate transformation

the hierarchically orders the states in z in terms of heir ability to capture the input-output characteristics of the system. We will begin by considering an investible transformation $T \in \mathbb{R}^{n_{i}}$, and then provide a method to compare just the first r columns, which will comprise the transformation Ψ in (22). Thus, it will be possible to retain only the first rmore constructionable/blowerscales, which recreate the term. This is similar to the change and the structure of the structure output of the observable transformation of the structure of the dynamic constraints of the structure output of the structure output of the structure output of the structure of the dynamic constraints of the dynamic constraints of the dynamic of the dynamic constraints of the dynamic constraints of the structure output of the structure output of the dynamic constraints of the dynamic constrain

Substituting Tz into (9.1) gives:

$$\frac{d}{dt}Tz = ATz + Bu$$
 (9.6a)

$$y = CTz + Du.$$
 (9.6b)

Finally, multiplying (9.6a) by T-1 yields:

$$\frac{d}{dt}\mathbf{z} = \mathbf{T}^{-1}\mathbf{A}\mathbf{T}\mathbf{z} + \mathbf{T}^{-1}\mathbf{B}\mathbf{u} \qquad (9.7a)$$

$$y = CTz + Du.$$
 (9.7b)

This results in the following transformed equations:

$$\frac{d}{dt}\mathbf{z} = \hat{\mathbf{A}}\mathbf{z} + \hat{\mathbf{B}}\mathbf{u} \qquad (9.8a)$$

$$y = \hat{C}z + Du$$
, (9.8b)

where $\hat{A} = T^{-1}AT$, $\hat{B} = T^{-1}B$, and $\hat{C} = CT$. Note that when the columns of T are orthonormal, the change of coordinates becomes:

$$\frac{d}{dt}z = T^{*}ATz + T^{*}Bu$$
(9.9a)

$$y = CTz + Du.$$
 (9.9b)

Gramians and Coordinate Transformations

The controllability and observability Graminas each establish an inner product on state space in terms of how controllable or observable a given state is, respectively. As such, Graminas depend on the particular choice of coordinate system and will transform under a change of coordinates. In the coordinate system z given by (9.5), the controllability Gramina becomes:

$$\hat{W}_{c} = \int_{0}^{\infty} e^{\hat{k}\tau} \hat{B}\hat{B}^{*} e^{\hat{k}^{*}\tau} d\tau \qquad (9.10a)$$

$$= \int_{0}^{\infty} e^{\mathbf{T}^{-1}\mathbf{A}\mathbf{T}_{T}}\mathbf{T}^{-1}\mathbf{B}\mathbf{B}^{*}\mathbf{T}^{-*}e^{\mathbf{T}^{*}\mathbf{A}^{*}\mathbf{T}^{-*}\tau} d\tau \qquad (9.10b)$$

$$= \int_{0}^{\infty} \mathbf{T}^{-1} e^{\mathbf{A}\tau} \mathbf{T} \mathbf{T}^{-1} \mathbf{B} \mathbf{B}^{*} \mathbf{T}^{-*} \mathbf{T}^{*} e^{\mathbf{A}^{*}\tau} \mathbf{T}^{-*} d\tau \qquad (9.10c)$$

$$= \mathbf{T}^{-1} \left(\int_{0}^{\infty} e^{\mathbf{A} \tau} \mathbf{B} \mathbf{B}^{*} e^{\mathbf{A}^{*}} d\tau \right) \mathbf{T}^{-*} \qquad (9.10d)$$

$$= T^{-1}W_cT^{-*}$$
. (9.10e)

Note that here we introduce $\mathbf{T}^{-*} := (\mathbf{T}^{-1})^* = (\mathbf{T}^*)^{-1}$. The observability Gramian transforms similarly:

$$\hat{W}_{e} = T^{*}W_{e}T$$
, (9.11)

which is an exercise for the reader. Both Gramians transform as tensors (i.e., in terms of the transform matrix T and its transpose, rather than T and its inverse), which is consistent with them inducine an inner reoduct on state-space.

Simple Rescaling

This example, modified from Moore 1981 [388], demonstrates the ability to balance a system through a change of coordinates. Consider the system

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} = \begin{bmatrix} -1 & 0 \\ 0 & -10 \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} + \begin{bmatrix} 10^{-3} \\ 10^3 \end{bmatrix} u \quad (9.12a)$$

$$y = \begin{bmatrix} 10^3 & 10^{-3} \end{bmatrix} \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}$$
. (9.12b)

In this example, the first state x_1 is barely controllable, while the second state is barely observable. However, under the change of coordinates $z_1 = 10^3 x_1$ and $z_2 = 10^{-3} x_2$, the system becomes balanced:

$$\frac{d}{dt}\begin{bmatrix} z_1\\ z_2\end{bmatrix} = \begin{bmatrix} -1 & 0\\ 0 & -10\end{bmatrix}\begin{bmatrix} z_1\\ z_2\end{bmatrix} + \begin{bmatrix} 1\\ 1\end{bmatrix}u$$
(9.13a)

$$y = \begin{bmatrix} 1 & 1 \end{bmatrix} \begin{bmatrix} z_1 \\ z_2 \end{bmatrix}$$
. (9.13b)

In this example, the coordinate change simply rescales the state x. For instance, it may be that the first state had units of millimeters while the second state had units of kilometers. Writing both states in meters balances the dynamics; that is, the controllability and observability formations are equal and diaronal.

Balancing Transformations

Now we are ready to derive the balancing coordinate transformation T that makes the controllability and observability Gramians equal and diagonal:

$$\bar{W}_c = \bar{W}_o = \Sigma.$$
 (9.14)

First, consider the product of the Gramians from (9.10) and (9.11):

$$\hat{W}_c \hat{W}_o = T^{-1} W_c W_o T.$$
 (9.15)

Plugging in the desired $\hat{W}_c = \hat{W}_o = \Sigma$ yields

$$T^{-1}W_cW_oT = \Sigma^2 \implies W_cW_oT = T\Sigma^2$$
. (9.16)

The latter expression in (9.16) is the equation for the eigendecomposition of $W_i W_{s,h}$ the product of the Grammins in the original coordinates. Thus, the balancing transformation T is related to the eigendecomposition of $W_i W_{s,h}$. The expression 9.16 is valid for any eigenvectors, and the correst rescaling must be chosen to exactly balance the $W_i W_{s,h} = 2^2$, but where the individual Grammian are not equal (for example diagonal Grammian $W_{s,h} = 2_{s,h}$ and $W_{s,h} = 2_{s,h}$ with $V_{s,h} = 2^{s,h}$.

We will introduce the matrix $S = T^{-1}$ to simplify notation.

Scaling Eigenvectors for the balancing Transformation

To find the correct scaling of eigenvectors to make $\hat{W}_c = \hat{W}_o = \Sigma$, first consider the simplified case of balancing the first diagonal element of Σ . Let \hat{k}_{μ} denote the unscaled first column of Σ and let μ_{μ} denote the unscaled first row of $S = T^{-1}$. Then

$$q_\mu W_c q^*_\mu = \sigma_c$$
 (9.17a)

$$\xi_{\mu}^{*}W_{\mu}\xi_{\mu} = \sigma_{\mu}$$
. (9.17b)

The first element of the diagonalized controllability Gramian is thus σ_c , while the first element of the diagonalized observability Gramian is σ_c . If we scale the eigenvector $\frac{2}{N_c}$ by σ_c , then the inverse eigenvector $\frac{1}{N_c}$ is scaled by σ_c^{-1} . Transforming via the new scaled eigenvectors $\frac{2}{N_c} = \sigma_c \frac{2}{N_c}$ and $\frac{1}{N_c} = \sigma_c^{-1} \frac{2}{N_c}$, visit:

$$\eta_i W_c \eta_i^* = \sigma_i^{-2} \sigma_c$$
, (9.18a)

$$\xi_{i}^{*}W_{i}\xi_{i} = \sigma_{i}^{2}\sigma_{i}$$
. (9.18b)

Thus, for the two Gramians to be equal,

$$\sigma_i^{-2}\sigma_c = \sigma_i^2 \sigma_o \implies \sigma_i = \left(\frac{\sigma_c}{\sigma_o}\right)^{1/4}$$
. (9.19)

To balance every diagonal entry of the controllability and observability Gramians, we first consider the unscaled eigenvector transformation \mathbf{T}_a from (9.16); the subscript a simply denotes usucated. As an example, we use the standard scaling in most computational software so that the columns of \mathbf{T}_a have unit norm. Then both Gramians are diagonalized, but are not necessarily equal:

$$T_{\mu}^{-1}W_{c}T_{\mu}^{-n} = \Sigma_{c}$$
 (9.20a)

$$T_{\mu}^{*}W_{\mu}T_{\mu} = \Sigma_{\mu}.$$
 (9.20b)

The scaling that exactly balances these Gramians is then given by $\Sigma_z = \Sigma_c^{1/4} \Sigma_o^{-1/4}$. Thus, the exact balancing transformation is given by

$$T = T_{\mu}\Sigma_{\mu}$$
. (9.21)

It is possible to directly confirm that this transformation balances the Gramians:

$$(\mathbf{T}_{u} \Sigma_{z})^{-1} \mathbf{W}_{c} (\mathbf{T}_{u} \Sigma_{z})^{-s} = \Sigma_{z}^{-1} \mathbf{T}_{u}^{-1} \mathbf{W}_{c} \mathbf{T}_{u}^{-s} \Sigma_{z}^{-1} = \Sigma_{z}^{-1} \Sigma_{c} \Sigma_{z}^{-1} = \Sigma_{c}^{1/2} \Sigma_{0}^{1/2}$$
 (9.22a)

$$(\mathbf{T}_{\mu}\Sigma_{\lambda})^{*}\mathbf{W}_{\nu}(\mathbf{T}_{\mu}\Sigma_{\lambda}) = \Sigma_{\lambda}\mathbf{T}_{\mu}^{*}\mathbf{W}_{\nu}\mathbf{T}_{\mu}\Sigma_{\lambda} = \Sigma_{\lambda}\Sigma_{\nu}\Sigma_{\lambda} = \Sigma_{\lambda}^{1/2}\Sigma_{\nu}^{1/2}.$$
 (9.22b)

Manipulations 9.22a and 9.22b rely on the fact that diagonal matrices commute, so that $\Sigma_c \Sigma_o = \Sigma_c \Sigma_c$, etc.

Example of the Balancing Transform and Gramians

Before confronting the practical challenges associated with accurately and efficiently computing the balancing transformation, it is helpful to consider an illustrative example.

In Matlab, computing the balanced system and the balancing transformation is a simple one-line command:

```
[sysb,q,Ti,T] = balreal(sys); & Balance system
```

In this code, T is the transformation, Ti is the inverse transformation, sysb is the balanced system, and g is a vector containing the diagonal elements of the balanced Gramians.

The following example illustrates the balanced realization for a two-dimensional system. First, we generate a system and compute its balanced realization, along with the Gramians for each system. Next, we visualize the Gramians of the unbalanced and balanced systems in Fig. 9.2.

Code 9.1 Obtaining a balanced realization

```
A = [-,75 1; -3 -,75];
B = [2; 1];
C = [1 2];
D = 0;
Hyg = am(A,B,C,D);
Wc = qram(sys,*c'); € Controllability Gramian
Wo = qram(sys,*c'); € Controllability Gramian
```



Figure 5.2 Illustration of balancing transformation on Gramians. The reachable set with unit control input is shown in red, given by $W_c^{1/2}$ x for |y| = 1. The corresponding observable set is shown in blue. Under the balancing transformation T, the Gramians are causal, shown in search.

```
[sysb,g,Ti,T] = baireal(sys); % Balance the system
BMC = gram(sysb,'c') % Balanced Gramians
BMC = gram(sysb,'c')
```

The resulting balanced Gramians are equal, diagonal, and ordered from most controllable/observable mode to least:

```
>>BMC =

1.9439 -0.0000

-0.0000 0.3207

>>BMC =

1.9439 0.0000

0.0000 0.3207
```

To visualize the Gramians in Fig. 9.2, we first recall that the distance the system can go in a direction x with a unit actuation input is given by x^*W_rx . Thus, the controllability Gramian may be visualized by plotting $W_r^{1/2}x$ for x on a sphere with $\|x\| = 1$. The observability cramian may be similarly visualized.

In this example, we see that the most controllable and observable directions may not be well aligned. However, by a change of coordinates, it is possible to find a new direction that is the most jointly controllable and observable. It is then possible to represent the system in this one-dimensional subspace, while still capturing a significant portion of the inparoutput energy. If the red and blue Gramians were exactly perpendicular, so that the most controllable direction was the least observable direction, and vice versa, then the balanced Gramian would be a circle. In this case, there is no preferred state direction, and both directions are equally important for the input-output behavior.

Instead of using the **bahreal** command, it is possible to manually construct the balancing transformation from the eigendecomposition of $W_c W_o$, as described earlier and provided in code available online.

Balanced Truncation

We have now shown that it is possible to define a change of coordinates so that the controlhabitity and observability Graniana are equal and diagonal. Mercorer, these new coordinates may be ranked hierarchically in terms of their joint controllability and observabilityity may be possible to truncate these coordinates and keep only the most controllability servable directions, resulting in a reduced-order model that faitfully captures input-output organism.

Given the new coordinates $z=T^{-1}x\in \mathbb{R}^n,$ it is possible to define a reduced-order state $\tilde{x}\in \mathbb{R}^r,$ as

$$\mathbf{z} = \begin{bmatrix} z_1 \\ \vdots \\ z_r \\ z_{r+1} \\ \vdots \\ z_n \end{bmatrix} \hat{\mathbf{x}}$$
(9.23)

in terms of the first r most controllable and observable directions. If we partition the balancing transformation T and inverse transformation $S = T^{-1}$ into the first r modes to be retained and the last n - r modes to be truncated,

$$\Gamma = \begin{bmatrix} \Psi & T_r \end{bmatrix}, \quad S = \begin{bmatrix} \Phi^* \\ S_r \end{bmatrix},$$
 (9.24)

then it is possible to rewrite the transformed dynamics in (9.7) as:

$$\frac{d}{dt} \begin{bmatrix} \tilde{\mathbf{x}} \\ \mathbf{z}_t \end{bmatrix} = \begin{bmatrix} \Phi^* \mathbf{A} \Psi & \Phi^* \mathbf{A} T_t \\ \mathbf{S}_t \mathbf{A} \Psi & \mathbf{S}_t \mathbf{A} T_t \end{bmatrix} \begin{bmatrix} \tilde{\mathbf{x}} \\ \mathbf{z}_t \end{bmatrix} + \begin{bmatrix} \Phi^* \mathbf{B} \\ \mathbf{S}_t \mathbf{B} \end{bmatrix} \mathbf{u} \quad (9.25a)$$

$$\mathbf{y} = \begin{bmatrix} \mathbf{C} \Psi & \mathbf{C} \mathbf{T}_r \end{bmatrix} \begin{bmatrix} \tilde{\mathbf{X}} \\ \mathbf{z}_r \end{bmatrix} + \mathbf{D} \mathbf{u}.$$
 (9.25b)

In balanced truncation, the state z_c is simply truncated (i.e., discarded and set equal to zero), and only the x equations remain:

$$\frac{d}{dt}\tilde{\mathbf{x}} = \Phi^* \mathbf{A} \Psi \tilde{\mathbf{x}} + \Phi^* \mathbf{B} \mathbf{u} \qquad (9.26a)$$

$$y = C\Psi \hat{x} + Du.$$
 (9.26b)

Only the first r columns of T and S^{*} = T⁻¹ are required to construct Ψ and Φ , and thus computing the entire balancing transformation T is unnecessary. Note that the matrix Φ here is different than the matrix of DMD modes in Section 7.2. The computation of Ψ and Φ without T will be discussed in the following sections. A key benefit of balanced trancation is the existence of upper and lower bounds on the error of a given order transitor.

Upper bound:
$$\|\mathbf{G} - \mathbf{G}_{r}\|_{\infty} \le 2 \sum_{j=r+1}^{n} \sigma_{j}$$
, (9.27a)

Lower bound:
$$\|\mathbf{G} - \mathbf{G}_{r}\|_{\infty} > \sigma_{r+1}$$
, (9.27b)

where σ_j is the *j*th diagonal entry of the balanced Gramians. The diagonal entries of Σ are also known as *Hankel singular values*.

Computing Balanced Realizations

In the previous section we demonstrated the feasibility of obtaining a coordinate transformation that balances the controllability and observability Graminas. However, the comtrol of this balancing transformation is nontrivial, and significant work has gone tono obtaining accurate and efficient methods, starting with Moore in 1981 [138], and contining with Lall, Mandera, and Garaskii a 2002 [21]. Willows and Previne in 2002 [51] and Romby in 2005 [458]. For an excellent and complete treatment of balanced realizations and model robactions, exclusions [17].

In practice, computing the Gramians W_{μ} and W_{ν} and the eigendecomposition of the product $W_{\nu}W_{\mu}$ in (9.16) may be prohibitively expensive for high-dimensional systems. Instead, the bulancing transformation may be approximated from impulse-response data, utilizing the singular value decomposition for efficient extraction of the most relevant subspaces.

We will first show that Gramians may be approximated via a snapshet matrix from impulse-response experiments/simulations. Then, we will show how the balancing transformation may be obtained from this data.

Empirical Gramians

In practice, computing Gramians via the Lyapunov equation is computationally expensive, with computational complexity of $O(n^3)$. Instead, the Gramians may be approximated by full-state measurements of the discrete-time direct and adjoint systems:

direct:
$$x_{k+1} = A_d x_k + B_d u_k$$
, (9.28a)

adjoint:
$$\mathbf{x}_{k+1} = \mathbf{A}_{d}^{*} \mathbf{x}_{k} + \mathbf{C}_{d}^{*} \mathbf{y}_{k}$$
. (9.28b)

(9.28a) is the discrete-time dynamic update equation from (8.21), and (9.28b) is the adjoint equation. The matrices A_1 , B_4 , and C_2 are the discrete-time system matrices from (8.22). Note that the adjoint equation is generally nonphysical, and must be simulated, thus the methods here apply to analytical equations and simulations, but not to experimental data. An alternative formulation that does not rely on adjoint data, and therefore generalizes to experiments, will be provided in Section 9.3. Computing the impulse-response of the direct and adjoint systems yields the following discrete-time snapshot matrices:

$$C_d = \begin{bmatrix} \mathbf{R}_d & \mathbf{A}_d \mathbf{B}_d & \cdots & \mathbf{A}_d^{m_d-1} \mathbf{B}_d \end{bmatrix}$$
 $O_d = \begin{bmatrix} \mathbf{C}_d \\ \mathbf{C}_d \mathbf{A}_d \\ \vdots \\ \mathbf{C}_d \mathbf{A}_d^{m_d-1} \end{bmatrix}$. (9.29)

Note that when $m_c = n$, C_d is the discrete-time controllability matrix and when $m_a = n$, O_d is the discrete-time observability matrix; however, we generally consider $m_c, m_a \ll n$. These matrices may also be obtained by sampling the continuous-time direct and adjoint systems at a regular interval Δt .

It is now possible to compute *empirical* Gramians that approximate the true Gramians without solving the Lyapunov equations in (8.42) and (8.43):

$$W_c \approx W'_c = C_d C_d^*$$
, (9.30a)

$$W_{\rho} \approx W_{\rho}^{2} = O_{A}^{*}O_{A}$$
. (9.30b)

The empirical Grammins essentially comprise a Riemans run approximation of the integral in the continuous-time Grammins, which Boromes extra at the time-stop of the discretistime system becomes arbitrarily small and the duration of the impulse response becomes durationally large In parameters due to a the method of empirical Grammins in quite efficient, highly dumped transients data out. The method of empirical Grammins in quite efficient, and is viably and GRA 20, 231, 254, 544, 544, 304. When the argulated methods are hard to gramming the structure of the structure of the structure of the structure of the number of comparison for a structure of the structu

Balanced POD

Instead of computing the eigendecomposition of W_cW_e , which is an $n \times n$ matrix, it is possible to compute the balancing transformation via the singular value decomposition of the product of the snapshot matrices,

$$O_d C_d$$
. (9.31)

reminiscent of the method of snapshots from Section 1.3 [490]. This is the approach taken by Rowley [458].

First, define the generalized Hankel matrix as the product of the adjoint (O_d) and direct (C_d) snapshot matrices from (9.29), for the discrete-time system:

$$\mathbf{H} = \mathcal{O}_d \mathcal{C}_d = \begin{bmatrix} \mathbf{C}_d \\ \mathbf{C}_t \mathbf{A}_d \\ \vdots \\ \mathbf{C}_d \mathbf{A}_d^{m-1} \end{bmatrix} \begin{bmatrix} \mathbf{B}_d & \mathbf{A}_d \mathbf{B}_d & \cdots & \mathbf{A}_d^{m-1} \mathbf{B}_d \end{bmatrix}$$
(9.32a)

$$= \begin{bmatrix} \mathbf{C}_{d}\mathbf{A}_{d} & \mathbf{C}_{d}A_{d}\mathbf{B}_{d} & \cdots & \mathbf{C}_{d}A_{d}^{n_{d}-1}\mathbf{B}_{d} \\ \mathbf{C}_{d}A_{d}\mathbf{B}_{d} & \mathbf{C}_{d}A_{d}^{n_{d}}\mathbf{B}_{d} \\ \vdots & \vdots & \ddots & \vdots \\ \mathbf{C}_{d}A_{d}^{n_{d}-1}\mathbf{B}_{d} & \mathbf{C}_{d}A_{d}^{n_{d}}\mathbf{B}_{d} & \cdots & \mathbf{C}_{d}A_{d}^{n_{d}-n_{d}-2}\mathbf{B}_{d} \end{bmatrix}, \quad (9.32b)$$

Next, we factor H using the SVD:

$$\mathbf{H} = \mathbf{U}\Sigma\mathbf{V}^* = \begin{bmatrix} \tilde{\mathbf{U}} & \mathbf{U}_t \end{bmatrix} \begin{bmatrix} \tilde{\Sigma} & \mathbf{0} \\ \mathbf{0} & \Sigma_t \end{bmatrix} \begin{bmatrix} \tilde{\mathbf{V}}^* \\ \mathbf{V}_t^* \end{bmatrix} \approx \tilde{\mathbf{U}}\tilde{\Sigma}\tilde{\mathbf{V}}^*.$$
 (9.33)

For a given desired model order $r \ll n$, only the first r columns of U and V are retained, along with the first $r \times r$ block of Σ ; the remaining contribution from $U_t \Sigma_t V_t^*$ may be truncated. This yields a bi-orthogonal set of modes given by:

direct modes:
$$\Psi = C_d \tilde{V} \tilde{\Sigma}^{-1/2}$$
, (9.34a)

adjoint modes:
$$\Phi = O_d^* \tilde{U} \tilde{\Sigma}^{-1/2}$$
. (9.34b)

The direct modes $\Psi \in \mathbb{R}^{n\times r}$ and adjoint modes $\Phi \in \mathbb{R}^{n\times r}$ are bi-orthogonal, $\Phi^*\Psi = \mathbf{I}_{rerr}$, and Rowley [458] showed that they establish the change of coordinates that balance the truncated empirical Gramians. Thus, Ψ approximates the first *r*-observed for full $n \times n$ balancing transformation, $\mathbf{T} = \mathbf{T}^{-1}$.

Now, it is possible to project the original system onto these modes, yielding a balanced reduced-order model of order r:

$$\tilde{A} = \Phi^* A_d \Psi$$
, (9.35a)

$$B = \Phi^* B_d$$
, (9.35b)

$$\dot{C} = C_d \Psi$$
. (9.35c)

It is possible to compute the reduced system dynamics in (9.35a) without having direct access to A_d . In some cases, A_d may be exceedingly large and unwieldy, and instead it is only possible to evaluate the action of this matrix on an input vector. For example, in many modern hald dynamics codes the matrix A_d is not actually represented, but because it is sparse, it is possible to implement efficient routints to multiply this matrix by a vector.

It is important to note that the reduced-order model in (9.35) is formulated in discrete time, as it is based on discrete-time empirical snapshot matrices. However, it is simple to obtain the corresponding continuous-time system:

In this example, D is the same in continuous time and discrete time, and in the full-order and reduced-order models.

Note that a BPOD model may not exactly satisfy the upper bound from balanced truncation (see (9.27)) due to errors in the empirical Gramians.

Output Projection

Often, in high-dimensional simulations, we assume full-state measurements, so that p = nis exceedingly large. To avoid computing p = n adjoint simulations, it is possible instead to solve an output-projected adjoint equation [458]:

$$x_{k+1} = A_d^* x_k + C_d^* U y$$
 (9.36)

where \hat{U} is a mark: containing the first *r* singular vectors of C_{π} . Thus, we let kidently a low-dimensional POD subspace \hat{U} from a direct impate response, and then only perform adjoint impaties response simulations by exciting these for POD coefficient measurements. More generally, if r_{μ} is high dimensional but does not measure the first star, it is possible to use a POD subspace trained on the measurements, given by the first *r* singular vectors \hat{U} for $C_{2\pi}$. Adjoint impaties responses may the the performed in these compareDO directions.

Data Collection and Stacking

The powers m_c and m_b in (9.23) signify that data must be collected until the matrices L_p and C_p^2 are full match after which the controllable/observable subspaces have been sampled. Unless we collect data until transients decay, the true Gramians are only approximately balanced. Instead, it is possible to collect data until the Hankel naturk is full rank, balance the resulting model, and then truncate. This more efficient approach is developed in [533] and [1546].

The snapshot matrices in (9.29) are generated from impulse-response simulations of the direct (9.28a) and adjoint (9.36) systems. These time-series snapshots are then interfeaved to form the snapshot matrices.

Historical Note

The balanced POD method described in the previous subsection originated with the seminal work of Moore in 1981 [388], which provided a data-driven generalization of the minimal realization theory of Ho and Kalama [247]. Uttil them, minimal realizations were defined in terms of idealized controllable and observable subspaces, which neglected the subtlety of degrees of controllability and observables.

More reper introduced a number of oriend onceps the tripped regular balance of the gap from three design (FeIA) is calcular by respective design of the strength on the strength one strength on the strength one strength on the strength of the strength on the strength on the strength on the strength on the strength of the strength on the strength on the strength of the strength of the strength on the strength of the strength of

Due derwheck of Moore's approach is that he compared the entire $n \times n$ balancies transformation, which is not satisfied for exceedingly high dimensional systems. In 2002, Willcox and Perzine [554] generalized the method to high-dimensional systems, introducing avatiant based on the rathe' decomposition of W, and W, obtained from the direct W, W, using efficient eigenvalues are been strateging and W and W and W and W and matrices. However, this approach has the drawback of requiring a many adaption implacments. response simulations as the number of output equations, which may be exceedingly large for full-state measurements. In 2005, Robely [458] address diffusion for anordening the output projection, discussed previously, which limits the number of adjoint simulations to the number of relevant POD modes in the data. He also showed that it is possible to use the eigendecomposition of the product $O_{n}C_{d}$. The product $O_{n}C_{d}$ is often smaller, and these computations may be more accurate.

It is intersting to note that a nearly equivalent formulation was developed twenty years caller in the field of system identification. The so-called eigensystem realization algorithm (EAA) [272], introduced in 1986 by Joang and Pappa, obtains equivalent balanced models without the need for adjoint data, making it useful for system identification in experiments. This connection between ERA and BPOD was scabilished by Mae et al. n2011 15311.

Balanced Model Reduction Example

In this example we will demonstrate the computation of balanced truncation and balanced POD models on a random state-space system with n = 100 states, q = 2 inputs, and p = 2outputs. First, we generate a system in Matab:

Next, we compute the Hankel singular values, which are plotted in Fig. 9.3. We see that r = 10 modes captures over 90% of the input-output energy.

```
have = havd(svsFull): # Mankel singular values
```

Now we construct an exact balanced truncation model with order r = 10:

```
## Exact balanced truncation
#ysBT = balred(#ysFull,r); # Balanced truncation
```

The full-order system, and the balanced truncation and balanced POD models are compared in Fig. 9.4. The BPOD model is computed using Code 9.2. It can be seen that the



Figure 2.3 Hankel singular values (left) and cumulative energy (right) for random state space system with n = 100, p = q = 2. The first r = 10 HSVs contain 92.9% of the energy.



Figure 8.4 Impulse response of full-state model with n = 100, p = q = 2, along with balanced truncation and balanced POD models with r = 10.

balanced model accurately captures the dominant input-output dynamics, even when only 10% of the modes are kept.

Code \$2 Balanced proper orthogonal decomposition (BPOD).

```
press = REDS [symbil: space[symbil: spa
```

```
Phi = Xdata+V+Sig^ (-1/2);
Pwi = Ydata+U+Sig^ (-1/2);
Ar = Pwi(i,1rr) *ayaPull.a,Phi(i,1rr);
Br = Pwi(i,1rr) *ayaPull.b;
Cr = ayaPull.c+Phi(1rr);
Dr = ayaPull.c+Phi(1rr);
Dr = ayaPull.c+Phi(r,Cr,Dr,-1);
```

9.3 System Identification

In contrast to model relation, where the system model (a, B, C, D) we also say, years distribution in period particular characteristic structures of the structure of matchine length of a characteristic structure of the system is largered from maning data structures are structured by the system in the structure of the structure of the system of a system of system (assistic) and (1). Single structures of the localing methods are been of a form of dynamic segression that fits models the structure of a system of the system of system (assistic) and (1). Single structures of the localing methods are been of a form of dynamic segression that fits models because of algorithm (1). Single-models are structured by the system relation algorithm (1) and observer K-Mann thick identifications (GOM) methods because of and chore here the system of the system structures and the system structures and chore here the system of the system structures are structured and chore here the system of the system structure of the structures the model system (2). The TEACKERD procedures is also applicable to model application. In high-density of the MAN system (3) and the structures the model system (3) and structures are barried and the structures the model system (3) and structures are barried and the structures the model method and dis SNNN system. Structures are structures and structures are also applicable to model and the structure of the structure of the structures are also applicable to models and the structure of the structures are also applicable to models are also applicable to models and a structure of the structures are also applicable to models are also as a structure of the structures are also applicable to the structure are also applicable to models are also as a structure of the structure are also applicable to the structure are also applicable to the structure are also applicable to the structure and the structure and the structure are also applicable to the structure are also applicable to the structure are also applicable

Eigensystem Realization Algorithm

The eigensystem realization algorithm produces low dimensional linear impart-output models of from encore measurements of an impaine response reperiment, based on the "minimal realization" theory of Ho and Kahuma [247]. The modern theory was developed to identify structural models for various space-traffic 2723, and it has been shown by Ma er al [351] that ERA models are equivalent to BPOD models". However, ERA is haved entering on impate response measurements and does not require ports havedenge of a model.

We consider a discrete-time system, as described in Section 8.2:

$$x_{k+1} = A_d x_k + B_d u_k$$
 (9.37a)

$$y_k = C_d x_k + D_d u_k$$
. (9.37b)

A discrete-time delta function input in the actuation u:

$$\mathbf{u}_{k}^{1} \triangleq \mathbf{u}^{1}(k\Delta t) = \begin{cases} \mathbf{I}, & k = 0\\ \mathbf{0}, & k = 1, 2, 3, \cdots \end{cases}$$
(9.38)

² BPOD and ERA models both balance the empirical Gramians and approximate balanced trancation [388] for birb-dimensional systems, given a sufficient volume of data.

gives rise to a discrete-time impulse response in the sensors y:

$$\mathbf{y}_{k}^{\delta} \triangleq \mathbf{y}^{\delta}(k\Delta t) = \begin{cases} \mathbf{D}_{d}, & k = 0\\ \mathbf{C}_{d}\mathbf{A}_{d}^{k-1}\mathbf{B}_{d}, & k = 1, 2, 3, \cdots \end{cases}$$
(9.39)

In an experiment or simulation, typically a impulse responses are performed, one for each of the a spearent input channel. The output responses are collected for each impulsive input, and a given time-step 1, the output vectors in response to the j-th impulsive input will form the j-th oolmom of g_{1}^{2} . Thus, each of the g_{1}^{2} is a p < q matrix (CA⁴ B. Note that the system matrices (A, B, C, D) don't actually need to exist, as the method in the next section is purely dura-driven.

The Hankel matrix H from (9.32), is formed by stacking shifted time-series of impulseresponse measurements into a matrix, as in the HAVOK method from Section 7.5:

$$\mathbf{H} = \begin{bmatrix} \mathbf{y}_{1}^{*} & \mathbf{y}_{2}^{*} & \cdots & \mathbf{y}_{n-1}^{*} \\ \mathbf{y}_{2}^{*} & \mathbf{y}_{2}^{*} & \cdots & \mathbf{y}_{n-1}^{*} \\ \vdots & \vdots & \ddots & \vdots \\ \mathbf{y}_{n-1}^{*} & \mathbf{y}_{n-1}^{*} & \cdots & \mathbf{y}_{n-1}^{*} \\ \mathbf{C}_{n} \mathbf{A}_{n} \mathbf{B}_{n} & \mathbf{C}_{n} \mathbf{A}_{n}^{*} \mathbf{B}_{n} & \cdots & \mathbf{C}_{n} \mathbf{A}_{n}^{*-1} \mathbf{B}_{n} \\ \mathbf{C}_{n} \mathbf{A}_{n} \mathbf{B}_{n} & \mathbf{C}_{n} \mathbf{A}_{n}^{*} \mathbf{B}_{n} & \cdots & \mathbf{C}_{n} \mathbf{A}_{n}^{*-1} \mathbf{B}_{n} \\ \mathbf{C}_{n} \mathbf{A}_{n}^{*-1} \mathbf{B}_{n} & \mathbf{C}_{n} \mathbf{A}_{n}^{*} \mathbf{B}_{n} \cdots & \mathbf{C}_{n} \mathbf{A}_{n}^{*-1} \mathbf{B}_{n} \end{bmatrix}, \quad (9.400)$$

The matrix H may be constructed purely from measurements y^{δ} , without separately constructing O_d and C_d . Thus, we do not need access to adjoint equations.

Taking the SVD of the Hankel matrix yields the dominant temporal patterns in the timeseries data:

$$\mathbf{H} = \mathbf{U} \mathbf{\Sigma} \mathbf{V}^{*} = \begin{bmatrix} \mathbf{U} & \mathbf{U}_{t} \end{bmatrix} \begin{bmatrix} \mathbf{\hat{\Sigma}} & \mathbf{0} \\ \mathbf{0} & \mathbf{\Sigma}_{t} \end{bmatrix} \begin{bmatrix} \mathbf{\hat{V}}^{*} \\ \mathbf{V}_{t}^{*} \end{bmatrix} \approx \mathbf{\hat{U}} \mathbf{\hat{\Sigma}} \mathbf{\hat{V}}^{*}.$$
 (9.41)

The small small singular values in Σ_r are truncated, and only the first r singular values in $\tilde{\Sigma}$ are retained. The columns of \tilde{U} and \tilde{V} are eigen-time-delay coordinates.

Until this point, the ERA algorithm closely resemble: the BPOD precedure from Section 9.2. However, we don't require direct access to O_d and C_d or the system (A, B, C, D) to construct the direct and adjoint balancing transformations. Instead, with sensor measurements from an impulse-response experiment, it is also possible to create a second, shifted Huntel matrix H:

$$\begin{split} \mathbf{H}' = \begin{bmatrix} \mathbf{y}_{2} & \mathbf{y}_{1}^{2} & \cdots & \mathbf{y}_{n+1}^{2} \\ \mathbf{y}_{3}^{2} & \mathbf{y}_{4}^{2} & \cdots & \mathbf{y}_{n+2}^{2} \\ \vdots & \vdots & \ddots & \vdots \\ \mathbf{y}_{n+1}^{4} & \mathbf{y}_{n+2}^{4} & \cdots & \mathbf{y}_{n+m_{n}}^{4} \end{bmatrix} \end{split} \tag{9.42a}$$
$$= \begin{bmatrix} C_d A_d B_d & C_d A_d^2 B_d & \cdots & C_d A_d^{n_c} B_d \\ C_d A_d^2 B_d & C_d A_d^2 B_d & \cdots & C_d A_d^{n_c} + B_d \\ \vdots & \vdots & \ddots & \vdots \\ C_d A_d^{n_c} B_d & C_d A_d^{n_c+1} B_d & \cdots & C_d A_d^{n_c+n_c-1} B_d \end{bmatrix} = \mathcal{O}_d A \mathcal{C}_d. \quad (9.42b)$$

Based on the matrices H and H', we are able to construct a reduced-order model as follows:

$$\tilde{A} = \tilde{\Sigma}^{-1/2} \tilde{U}^* H' \tilde{V} \tilde{\Sigma}^{-1/2};$$
 (9.43a)

$$\tilde{\mathbf{B}} = \tilde{\boldsymbol{\Sigma}}^{1/2} \tilde{\mathbf{V}}^{*} \begin{bmatrix} \mathbf{I}_{p} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} \end{bmatrix}; \quad (9.43b)$$

$$\tilde{C} = \begin{bmatrix} I_{ij} & 0 \\ 0 & 0 \end{bmatrix} \tilde{U} \tilde{\Sigma}^{1/2}.$$
 (9.43c)

Here I_p is the $p \times p$ identity matrix, which extracts the first p columns, and I_q is the $q \times q$ identity matrix, which extracts the first q rows. Thus, we express the input-output dynamics in terms of a reduced system with a low-dimensional state $\hat{s} \in \mathbb{R}^{1}$:

$$\tilde{x}_{k+1} = \tilde{A}\tilde{x}_k + \tilde{B}u$$
 (9.44a)

$$y = C \tilde{x}_{g}$$
. (9.44b)

H and H' are constructed from impulse response simulationsteeperiments, without the need for storing direct or adjoint snaphots, as in other balanced model reduction techniques. However, if full-state snaphots are available, for example, by collecting velocity fields in simulations or PIV experiments, it is then possible to construct direct modes. These full-state snaphots form *Cx*, and modes can be constructed by:

$$\Psi = C_d \tilde{V} \tilde{\Sigma}^{-1/2}. \qquad (9.45)$$

These modes may then be used to approximate the full-state of the high-dimensional system from the low-dimensional model in (9.44) by:

$$x \approx \Psi \tilde{x}$$
. (9.46)

If enough data is collected when constructing the Haadi matrix H, then ERA balances the empirical constructionity and observability standing of the matrix $d_{\rm eff}$ and $d_{\rm eff}/d_{\rm eff}$. However, if less data is collected, so that light damped transients do takes time to decay, there ERA will and approximately balance the system. It is instand possible to collect just enough data so other the Haadi matrix H searches numerical full face of the matrix angle of the system of the transient of the standing of the system of the system possible values as before the backet balance of the system of the system of the system constrained by and behaviorability and hence half by a system of the balance balance of transients on this simulation models is alth docted in (3331 and [146].

```
The code to compute ERA is provided in Code 9.3.
```

Code 9.3 Electrovytem realization algorithm.

```
function [Ar, Br, Cr, Dr, HSVs] = ERA(YY, m, n, nin, nout, r)
     for jel:nin
         Dr(i, i) = Tr(i, i, 1);
         Y(i,j,:) = YY(i,j,2:end);
assert(length(Y(1,:,1))==nin);
assert(length(Y(1,1,:))>=n+n);
        for O=1:nout
            for Pal:nin
[U.S.V] = avd(H.'econ');
Sioma = S(1:r.1:r);
Ar = Sigma^(-.5) • Ur' • H2 • Vr • Sigma^(-.5);
Br = Sigma^ (-.5) + Ur' + H(:,1:nin);
HSVs = diag(S);
```

Observer Kalman Filter Identification

ORD was developed to complement the ERA for lightly damped experimental systems with noise [273], in particle, performing isolated implate response experiments is challenging, and the effect of measurement noise can commander tesults. Mereover, if there is a large vehanism of instructures, the system constrained and must be collected to use a disrupt synthesis of the system of the system of the system of the system arbitrary input-stopet data. Typically, one would identify reduced order models according to the following error procedure:



Figure 25 Schematic overview of OKID procedure. The output of OKID is an impulse response that can be used for system identification via ERA.

- 1. Collect the output in response to a pseudo-random input.
- This information is passed through the OKID algorithm to obtain the de-noised linear impulse response.
- The impulse response is passed through the ERA to obtain a reduced-order statespace system.

The output y_k in response to a general input signal u_k , for zero initial condition $x_0 = 0$, is given by:

$$y_0 = D_d u_0$$
 (9.47a)

$$y_1 = C_d B_d u_0 + D_d u_1$$
 (9.47b)

$$y_2 = C_d A_d B_d u_0 + C_d B_d u_1 + D_d u_2$$
 (9.47c)

$$y_k = C_d A_d^{k-1} B_d u_0 + C_d A_d^{k-2} B_d u_1 + \cdots + C_d B_d u_{k-1} + D_d u_k.$$
 (9.47d)

Note that there is no C term in the expression for y_0 since there is zero initial condition $x_0 = 0$. This progression of measurements y_k may be further simplified and expressed in terms of impulso-response measurements $y_k^{(1)}$:

$$\underbrace{[\underbrace{y_0 \quad y_1 \quad \cdots \quad y_m}_{\mathcal{S}}] = \underbrace{[\underbrace{y_0^{*} \quad y_1^{*} \quad \cdots \quad y_m^{*}}_{\mathcal{S}^{*}}]}_{\mathcal{S}^{*}} \underbrace{[\underbrace{\begin{matrix} u_0 \quad u_1 \quad \cdots \quad u_m \\ 0 \quad u_0 \cdots \quad u_m \\ \vdots \quad \vdots \quad \ddots \quad \vdots \\ 0 \quad 0 \quad \cdots \quad u_m \\ \mathcal{S}}}_{\mathcal{S}}.$$
(9.48)

It is often possible to invert the matrix of control inputs, B to solve for the Markov parameters S^3 . However, B may either be un-inventible, or inversion may be ill-conditioned. In addition, B is large for lightly damped systems, making inversion computationally expensive. Finally, noise is not optimally filtered by simply inverting B to solve for the Markov parameters.

The OKID method addresses each of these issues. Instead of the original discrete-time system, we now introduce an optimal observer system:

$$\hat{\mathbf{x}}_{k+1} = \mathbf{A}_d \hat{\mathbf{x}}_k + \mathbf{K}_f (\mathbf{y}_k - \hat{\mathbf{y}}_k) + \mathbf{B}_d \mathbf{u}_k$$
 (9.49a)

$$\hat{y}_{k} = C_{d}\hat{x}_{k} + D_{d}u_{k}$$
, (9.49b)

which may be re-written as:

$$\hat{\mathbf{x}}_{k+1} = \underbrace{(\mathbf{A}_d - \mathbf{K}_f \mathbf{C}_d)}_{\hat{\mathbf{A}}_d} \hat{\mathbf{x}}_k + \underbrace{[\mathbf{B}_d - \mathbf{K}_f \mathbf{D}_d, \mathbf{K}_f]}_{\mathbf{B}_d} \begin{bmatrix} \mathbf{u}_k \\ \mathbf{y}_k \end{bmatrix}. \quad (9.50)$$

Recall from earlier that if the system is observable, it is possible to place the poles of $A_d = S_t C_d$ approaches well like. However, depending on the amount of noise in the measurements, the magnitude of process noise, and uncertainty in our model, there are optimal pole bactions that are given by the *Kolama filter* (recall Section 8.5). We may now solve for the observer Markov parameters S^2 of the system in (9.50) in terms of measured inputs and outputs according to the following algorithm from [273]:

- 1. Choose the number of observer Markov parameters to identify. 1.
- 2. Construct the data matrices here:

$$S = [y_0 \ y_1 \ \cdots \ y_l \ \cdots \ y_m]$$
 (9.51)

$$\mathcal{V} = \begin{bmatrix} u_0 & u_1 & \cdots & u_r & \cdots & u_m \\ 0 & v_0 & \cdots & v_{r-1} & \cdots & v_{m-1} \\ \vdots & \vdots & \ddots & \vdots & \ddots & \vdots \\ 0 & 0 & \cdots & v_0 & \cdots & v_{m-r} \end{bmatrix}$$
(9.52)

where $\mathbf{v}_i = \begin{bmatrix} \mathbf{u}_i^T & \mathbf{y}_i^T \end{bmatrix}^T$.

The murity V resembles B_c except that is has been augmented with the outputs y_c . In this way, we are working with a system that is augmented to include a Kalman filter. We are now identifying the observer Markov parameters of the augmented system, B^1 , using the equation $S = S^1V$. It will be possible to identify these observer Markov parameters from the data and then extract the impulse response (Markov parameters) of the original system.

- Identify the matrix S¹ of observer Markov parameters by solving S = S¹V for S¹ using the right pseudo-inverse of V (i.e., SVD).
- Recover system Markov parameters, S³, from the observer Markov parameters, S³:
 - (a) Order the observer Markov parameters \tilde{S}^{δ} as:

$$\tilde{S}_{0}^{1} = \mathbf{D},$$
 (9.53)

$$\hat{S}_{k}^{1} = \left[(S^{1})_{k}^{(1)} (S^{1})_{k}^{(2)} \right] \text{ for } k \ge 1,$$
 (9.54)

where
$$(\tilde{S}^{\delta})_{k}^{(1)} \in \mathbb{R}^{q \times p}$$
, $(\tilde{S}^{\delta})_{k}^{(2)} \in \mathbb{R}^{q \times q}$, and $y_{0}^{\delta} = \tilde{S}_{0}^{\delta} = \mathbf{D}$.

(b) Reconstruct system Markov parameters:

$$\mathbf{y}_{k}^{i} = (\mathbf{S}^{i})_{k}^{(1)} + \sum_{i=1}^{k} (\mathbf{S}^{i})_{i}^{(2)} \mathbf{y}_{k-i}^{i} \text{ for } k \ge 1.$$
 (9.55)

Thus, the OKID method identifies the Markov parameters of a system augmented with an asymptotically stable Kalman filter. The system Markov parameters are extracted from the observer Markov parameters by (9.55). These system Markov parameters approximate the impulse response of the system, and may be used directly as inputs to the ERA algorithm. A code to compute OKID is provided in Code 9.4. ERAONID has been widely applied across a range of system identification tasks, including to identify models of aeroclastic structures and fluid dynamic systems. There are numerous extensions of the ERAONID methods. For example, there are generalizations for linear parameter varying (LPV) systems and systems linearized about a limit cycle.

Code 9.4 Observer Kalman filter identification (OKID).

```
function H = OKID(v.u.r)
$ Step 0, check shapes of y,u
p = size(y,1); * p is the number of outputs
g = size(u,1); } g is the number of inputs
1 = r+5;
V = zeros(q + (q+p)+1.m);
    for i=1:m+1-i
        y_{temp} = \{u(1, i), y(1, i)\}
        V(q+(i-2)*(q+p)+1:q+(i-1)*(q+p),i+j-1) = vtemp;
Thar = v*pinv(V,1,e-3);
D = Ybar(:,1:q); & Feed-through term (D) is first term
    Ybar1(1:p,1:g,i) = Ybar(:,g+1+(g+p)+(i-1):g+(g+p)+(i-1)+g):
    Ybar2(1:p.1:g.i) = Ybar(:.g+1+(g+p)*(i-1)+g:g+(g+p)*i);
 Y(:,:,1) = Ybar1(:,:,1) + Ybar2(:,:,1)*D;
    Y(:,:,k) = Ybar1(:,:,k) + Ybar2(:,:,k) \cdot D:
    for i=1:k-1
     Y(:,:,k) = Y(:,:,k) + Ybar2(:,:,i) * Y(:,:,k-i):
   H(:...,k) = Y(:...,k-1):
```

Combining ERA and OKID

Here we demonstrate ERA and OKID on the same model system from Section 9.2. Because ERA yields the same balanced models as BPOD, the reduced system responses should be the same.

First, we compute an impulse response of the full system, and use this as an input to ERA:

```
82 Obtain imputer response of tuil system
(yrbil); = langules (systul), cir(s)=1;);
TT = permete(yrbil); [2 > 1]); # Roorder to be size p x q x m
(default is n x p x q)
# Computer EAA from impulse response
mos = floar((langthyrbil)=1)/(2); # n q = n a = (n-1)/2
[An, Br.C.pte, HAVI] = EAA(TY, non, non, miniputs, nonOutputs, r);
norEAA = an (Ar.E.C.F.D.-1);
```

Next, if an impute response is unavailable, it is possible to excite the system with a random input signal and use OKID to extract an impute response. This impute response is then used by ERA to extract the model.

```
14 Compute random input simulation for OKTD
Windoms = random (unsingut, solo) + 8 Andom forcing input
yfacdom = laim(systHil, ufandom, 11200)'; + Output
14 Compute Coll and Inton EMPL
15 Compute Coll and Inton EMPL
16 compute Coll and Inton EMPL
16 constitute (unsingthEM)-1/2(); + 8 c = 8.0
[Ar_Br(c,D,D,TMVM] = SA(N,m,c,mc),meninguts,memOutputs,r);
yes/SACATO = e(Ar,Br,c,C,c,I);
```

Figure 9.6 shows the input-output data used by OKID to approximate the impulse response. The impulse responses of the resulting systems are computed via

```
[y1,t1] = impulse (sysFull,0:1:200);
[y2,t2] = impulse (sysERA,0:1:100);
[y3,t3] = impulse (sysERAONID,0:1:100);
```

Finally, the system responses can be seen in Fig. 9.7. The low-order FEAA and ERA/ORD models closely mutch the full model and have similar performance to the IPOD models described previously. Because ERA and IPOD are mathematically equivalent, this agreement is no assigning. However, the ability of EEA/ORD to settine a realead-graning and the system of the ability of EEA/ORD to be setting a realacidgrant setting of the system of the system of the system of the system IBOD, these methods are scalily opticable to represent an ensurements, as they do not require nonphytical adjoint equations.



Figure 9.6 Input-output data used by OKID.



Figure 27 Impulse response of full-state model with n = 100, p = q = 2, along with ERA and ERA/OKID models with r = 10.

Suggested Reading

Papers and Reviews

- Principal component analysis in linear systems: Controllability, observability, and model reduction, by B. C. Moore, *IEEE Transactions on Automatic Control*, 1981 13881.
- (2) Identification of linear parameter varying models, by B. Bamich and L. Giarré, International Journal of Robust and Nonlinear Control, 2002 [34].
- (3) Balanced model reduction via the proper orthogonal decomposition, by K. Willcox and J. Peraise, AIAA Journal, 2002 [554].
- (4) Model reduction for fluids using balanced proper orthogonal decomposition, by C. W. Rowley, International Journal of Bifurcations and Chaos, 2005 [458].
- (5) An eigensystem realization algorithm for modal parameter identification and model reduction, by J. N. Juang and R. S. Pappa, *Journal of Guidance, Control,* and Dynamics, 1985 [272].

As described in Charter 8, control design often begins with a model of the system being controlled. Notable exceptions include model-free adaptive control strategies and many uses of PID control. For mechanical systems of moderate dimension, it may be possible to write down a model (e.g., based on the Newtonian, Lagrangian, or Hamiltonian formalism) and linearize the dynamics about a fixed point or periodic orbit. However, for modern systems of interest, as are found in neuroscience, turbulence, epidemiology, climate, and finance, typically there are no simple models suitable for control design. Chapter 9 described techniques to obtain control-oriented reduced-order models for high-dimensional systems from data, but these approaches are limited to linear systems. Real-world systems are usually nonlinear and the control objective is not readily achieved via linear techniques. Nonlinear control can still be posed as an optimization problem with a high-dimensional, nonconvex cost function landscape with multiple local minima. Machine learning is complementary, as it constitutes a growing set of techniques that may be broadly described as performing nonlinear optimization in a high-dimensional space from data. In this chapter we describe emerging techniques that use machine learning to characterize and control strongly nonlinear, highdimensional, and multi-scale systems, leveraging the increasing availability of high-quality measurement data

Broadly speaking, machine learning techniques may be used to 1) characterize a system for later use with model-based control, or 2) directly characterize a control law that effectively interacts with a system. This is illustrated schematically in Fig. 10.1, where data-driven techniques may be applied to either the System or Controller blocks. In addition, related methods may also be used to identify good sensors and actuators, as discussed previously in Section 3.8. In this chapter, Section 10.1 will explore the use of machine learning to identify nonlinear input-output models for control, based on the methods from Chapter 7. In Section 10.2 we will explore machine learning techniques to directly identify controllers from input-output data. This is a rapidly developing field. with many powerful methods, such as reinforcement learning, iterative learning control, and genetic algorithms. Here we provide a high-level overview of these methods and then explore an example using genetic algorithms. However, it is important to emphasize the breadth and depth of this field, and the fact that any one method may be the subject of an entire book. Finally, in Section 10.3 we describe the adaptive extremum-seeking control strategy, which optimizes the control signal based on how the system responds to perturbations.



Figure 10.1 In the standard control framework from Chapter 8, machine learning may be used 1) to develop a model of the system or 2) to learn a controller.

10.1 Nonlinear System Identification for Control

The data driven modeling and counted or complex systems is an adapting a resolution, there have have been dependent on machine learning and optimizations, and monoton computational hardwares. Despite the interacting use of equitorism is the strength of the strength optimization of the strength optimization of the transformation of the strength optimization of the measurement of the strength optimization of the strength optimization of the based strength optimization of the strength optimization of the strength optimization of the based strength optimization of the strength optimization of the strength optimization of the based strength optimization of the strength optimization of the strength optimization of the based strength optimization of the based strength optimization of the strength optimization optimization of the strength optimization of the strength optimization of the strength optimization of the strength optimization opt

The goal of system identification is to identify a low-order model of the input-output dynamics from actuation u to measurements y. If we are able to measure the full state x of the system, then this reduces to identifying the dynamics if that satisfy:

$$\frac{d}{dt}\mathbf{x} = \mathbf{f}(\mathbf{x}, \mathbf{u}). \quad (10.1)$$

This problem may be formulated in discrete-time, since data is typically collected at discrete instances in time and control laws are often implemented digitally. In this case, the dynamics read:

$$\mathbf{x}_{p,\pm 1} = \mathbf{F}(\mathbf{x}_{p}, \mathbf{u}_{p}).$$
 (10.2)

When the dynamics are approximately linear, we may identify a linear system

$$x_{k+1} = Ax_k + Bu_k$$
, (10.3)

which is the approach taken in the DMD with control (DMDc) algorithm below.

It may also be advantageous to identify a set of measurements y = g(x), in which the unforced nonlinear dynamics appear linear:

$$y_{k+1} = A_Y y_k$$
. (10.4)

This is the approach taken in the Koopman control method below. In this way, nonlinear dynamics may be estimated and controlled using standard textbook linear control theory in the intrinsic coordinates y [302, 276].

Finally, the nonlinear dynamics in (10.1) or (10.2) may be identified directly using the SINDY with control algorithm. The resulting models may be used with model predictive control for the control of fully nonlinear systems [277].

DMD with Control

The DMDc method seeks to identify the best-fit linear operators A and B that approximately satisfy the following dynamics on measurement data:

$$x_{k+1} \approx Ax_k + Bu_k$$
. (10.5)

In addition to the snapshot matrix $\mathbf{X} = \begin{bmatrix} \mathbf{x}_1 & \mathbf{x}_2 & \cdots & \mathbf{x}_m \end{bmatrix}$ and the time-shifted snapshot matrix $\mathbf{X} = \begin{bmatrix} \mathbf{x}_2 & \mathbf{x}_3 & \cdots & \mathbf{x}_{m+1} \end{bmatrix}$ from (7.23), a matrix of the actuation input history is assembled:

$$\Upsilon = \begin{bmatrix} | & | & | & | \\ u_1 & u_2 & \cdots & u_m \\ | & | & | & | \end{bmatrix}.$$
 (10.6)

The dynamics in (10.5) may be written in terms of the data matrices:

$$X' \approx AX + BY$$
. (10.7)

As in the DMD algorithm (see Section 7.2), the leading eigenvalues and eigenvectors of the best-fit linear operator A are obtained via dimensionality reduction and regression. If the actuation matrix B is known, then it is straightforward to correct for the actuation and identify the spectral decomposition of A by replacing \mathbf{X}' with $\mathbf{X}' = \mathbf{B}\mathbf{Y}$ in the DMD algorithm:

$$(X' - BY) \approx AX.$$
 (10.8)

When B is unknown, both A and B must be simultaneously identified. In this case, the dynamics in (10.7) may be recast as:

$$\mathbf{X}' \approx \begin{bmatrix} \mathbf{A} & \mathbf{B} \end{bmatrix} \begin{bmatrix} \mathbf{X} \\ \mathbf{Y} \end{bmatrix} = \mathbf{G} \mathbf{\Omega},$$
 (10.9)

and the matrix $\mathbf{G} = \begin{bmatrix} \mathbf{A} & \mathbf{B} \end{bmatrix}$ is obtained via least-squares regression:

$$G \approx X' \Omega^{\uparrow}$$
. (10.10)

The matrix $\Omega = [X^* \ \Upsilon^*]^*$ is generally a high-dimensional data matrix, which may be approximated using the SVD:

$$Ω = UΣV^*$$
. (10.11)

The matrix \tilde{U} must be split into two matrices, $\tilde{U} = \begin{bmatrix} \tilde{U}_1^* & \tilde{U}_2^* \end{bmatrix}^n$, to provide bases for X and Y. Unlike the DMD algorithm, \tilde{U} provides a reduced basis for the *input space*, while \hat{U} from

$$X' = \hat{U}\hat{\Sigma}\hat{V}^{*}$$
 (10.12)

defines a reduced basis for the *output space*. It is then possible to approximate $G = \begin{bmatrix} A & B \end{bmatrix}$ by projecting onto this basis:

$$\tilde{\mathbf{G}} = \hat{\mathbf{U}}^* \mathbf{G} \begin{bmatrix} \hat{\mathbf{U}} \\ \mathbf{I} \end{bmatrix}$$
(10.13)

The resulting projected matrices A and B in G are:

$$\tilde{\mathbf{A}} = \hat{\mathbf{U}}^* \mathbf{A} \hat{\mathbf{U}} = \hat{\mathbf{U}}^* \mathbf{X}' \tilde{\mathbf{V}} \tilde{\boldsymbol{\Sigma}}^{-1} \tilde{\mathbf{U}}_1^* \hat{\mathbf{U}}$$
(10.14a)

$$\tilde{B} = \hat{U}^* B = \hat{U}^* X' \tilde{V} \tilde{\Sigma}^{-1} \tilde{U}_2^*$$
(10.14b)

More importantly, it is possible to recover the DMD eigenvectors Φ from the eigendecomposition $\hat{A}W = WA$:

$$\Phi = \mathbf{X} \mathbf{\tilde{V}} \mathbf{\tilde{\Sigma}}^{-1} \mathbf{\tilde{U}}_{1}^{*} \mathbf{\hat{U}} \mathbf{W}. \quad (10.15)$$

Ambiguity in Identifying Closed-Loop Systems

For systems that are being actively controlled via feedback, with u = Kx,

$$x_{k+1} = Ax_k + Bu_k$$
 (10.16a)

$$= Ax_k + BKx_k$$
 (10.16b)

$$= (A + BK)x_0$$
, (10.16c)

it is impossible to disambiguate the dynamics A and the actuation BK. In this case, it is important to add perturbations to the actuation signal u to provide additional information. These perturbations may be a white noise process or occasional impulses that provide a kick to the system, providing a signal to disambiguate the dynamics from the feedback signal.

Koopman Operator Nonlinear Control

For nonlinear systems, it may be advantageous to identify data-driven coordinate transformations that make the dynamics appear linear. These coordinate transformations are related to intrivice coordinates defined by eigenfunctions of the Koopinan operator (see Section 7.4). Koopinan analysis has thus been leveraged for nonlinear estimation [504, 505] and control [302, 276, 423]. It is possible to decign estimators and controller directly from DMD or eXMD models. In GMD and GMD effective control MMC (> to control and model predetive control MC (> to control and > to control MC (> to control and > to control MC (> to control and > to control model >

In Section 7.5, we described several strategies to approximate Koopman eigenfunctions, a(x), where the dynamics become linear:

$$\frac{d}{dt}\varphi(\mathbf{x}) = \lambda \varphi(\mathbf{x}).$$
 (10.17)

In Kaiser et al. [276] the Koopman eigenfunction equation was extended for control-affine nonlinear systems:

$$\frac{d}{dt}\mathbf{x} = \mathbf{f}(\mathbf{x}) + \mathbf{B}\mathbf{u}. \quad (10.18)$$

For these systems, it is possible to apply the chain rule to $\frac{d}{d} \varphi(\mathbf{x})$, yielding:

$$\frac{d}{dt}\varphi(\mathbf{x}) = \nabla \varphi(\mathbf{x}) \cdot (\mathbf{f}(\mathbf{x}) + \mathbf{B}\mathbf{u}) \quad (10.19a)$$

$$= \lambda \varphi(\mathbf{x}) + \nabla \varphi(\mathbf{x}) \cdot \mathbf{B}\mathbf{u}.$$
 (10.19b)

Note that even with actuation, the dynamics of Koopman eigenfunctions remain linear, and the effect of actuation is still additive. However, now the actuation mode $\nabla e_i (x_i) = B may be$ state dependent. In fact, the actuation will be state dependent unless the directional derivtive of the eigenfunction is constant in the B direction. Fortunately, there are many powerfulgeneralizations of standard Riccati-based linear control theory (e.g., LOR, Kalman filters,etc.) for systems with a nature dependent Riccati equation.

SINDy with Control

Abough its appealing to identify intrainsic coordinates along which nonlinear dynamics program funct, these conductions are challenging to discover, even for hardney's jangle program funct, these conductions are challenging to discover even for the intrained product the standard model based corner. Using the prove identification of nonlinear dynamics (SDNP) methods (the SCHATT 2) remarks in comparison (2017). However, the models model is interained with the standard model production (2017) methods the SCHATT 2) remarks in a standard model interained and the standard model production (2017). However, the models much identified from relativity and an annual relativity of the characterist of mine and a response to attract charge to the function of the standard in the standard model charge to the standard model production of the standard standard in the standard model production of the standard standard in the standard model production of the standard standard in the standard standard standard in the standard stand

The SINDy algorithm is readily extended to include the effects of actuation [100, 277]. In addition to collecting measurements of the state snapshots x in the matrix X, actuation inputs u are collected in the matrix \hat{T} from (10.6) as in DMDx. Next, an augmented library of candidate right hand side functions $\Theta([X, T])$ is constructed:

$$\Theta([X \ \Upsilon]) = [1 \ X \ \Upsilon \ X^2 \ X \otimes \Upsilon \ \Upsilon^2 \ \cdots].$$
 (10.20)

Here, $X \otimes \Upsilon$ denotes quadratic cross-terms between the state x and the actuation u, evaluated on the data.

In SIMDy with control (SINDYc), the same sparse regression is used to determine the fewest active terms in the library required to describe the observed dynamics. As in DMDe, if the system is being actively controlled via feedback **u** = **K**(**x**), then it is impossible to disambiguate from the internal dynamics and the actuation, unless an addition perturbation simal is added to the actuation to newide additional information.

Model Predictive Control (MPC) Example

In this campie, we will use SNNT's to identify a model of the fored Lunca equiliton time data and the control imm glood effective control (MVC) MPC [107, 105, 408, 191, 447, 409, 196, 256, 271 [108] as bosone a concentrate of modern process models and the structure of the stru

Model predictive control determines the next immediate control action by solving an optimal control problem over a receding hieroism. In particular, the open-show paramiton signal with coprimized on a receding time-horizon $t_{\mu} = m_{\mu} \Delta t$ to minimum z and m equal none prediction borning $t_{\mu} = m_{\mu} \Delta t$. The control horizon is by ryshigh less than or equal control is then applied for one time step, and the procedure is repeated and the receding control is then applied for one time step, and the procedure is repeated and the receding horizon counter σ_{eq} relations and the control horizon to the results in the counter lates:

$$K(x_i) = u_{i+1}(x_i),$$
 (10.21)

where up.,11 she first time step of the optimized actuation starting at up., This is show schematically in [76] [50, 21. It is possible to optimize highly contonized contained subject to nonlinear dynamics, with constraints on the actuation and state. However, the comparison of experiments of re-optimizing at each time-schema pare considerable, putting limits on the complexity of the model and optimization techniques. Fortunately, neglitime contrast of the product of the schema pare on the complexity of the model and optimization techniques. Thermately, neglicottend, optimized power and optimization are enabled particle to contain a contrast contrast.



Figure 10.2 Schematic overview of model predictive control, where the actuation input u is iteratively optimized over a receding horizon. Reproduced with permission from Kaiser et al. [277].

MPC to Control the Lorenz Equations with SINDYc

The following example illustrates how to identify a model with SINDYc for use in MPC. The basic code is the same as SINDy, except that the actuation is included as a variable when building the library Θ .

We test the SINDYc model identification on the forced Lorenz equations:

$$\dot{x} = \sigma(y - x) + g(u)$$
 (10.22a)

$$\dot{y} = x(\rho - z) - y$$
 (10.22b)

$$\dot{z} = xy - \beta z$$
, (10.22c)

In this example, we train a model using 20 time units of controlled data, and willdate it on another 20 time units where we switch the foreing to a periodic signal $w(r) = 50 \sin(10t)$. The SINUY algorithm does not capture the effect of actuation, while SINUY correctly identifies the forced model and predicts the behavior in response to a new actuation that was not used in the training data, as shown in Fig. 10.3.

Finally, SINDYc and neural network models of Lorenz are both used to design model predictive controllers, as shown in Fig. 10.4. Both methods identify accurate models that



Figure 10.3 SINDY and SINDYc predictions for the controlled Lorenz system in (10.22). Training data consists of the Lorenz system with state feedback. For the training period the input is a(t) = 26 - x(t) + d(t) with a Gaussian disturbance d. Afterward the input a switches to a periodic signal $a(t) = 50 \sin(10t)$. Reproduced with permission from /100.

capture the dynamics, although the SINDY's procedure requires less data, identifies models more rapidly, and is more robust to noise than the neural network model. This added efficiency and robustness is due to the sparsity promoting optimization, which regularizes the model identification problem. In addition, identifying a sparse model requires less data.

10.2 Machine Learning Control

Machine learning is a rapidly developing field that is transforming our ability to describe complex systems from observational data, rather than first-principles modeling [382, 161, 64, 396]. Until recently, these methods have largely been developed for static data, although



Figure 10.4 Model predictive control of the Lorenz system with a neural network model and a SINDy model. Reansdaced with permission from Kaiser et al. (277).



Figure 18.3 Schematic of machine learning control wrapped around a complex system using noisy sensor-based feedback. The control objective is to minimize a well-defined cost function J within the space of possible control lows. An off-line learning loop provides experiential data to train the controller. Genetic programming provides a particularly flexible algorithm to search out effective controller. Genetic programming provides a particularly flexible algorithm to be cost.

there is a growing emphasis on using machine learning to characterize dynamical systems. The use of massiles results to like our cost to $(L_{22}, L_{23}, L_{23}, R_{23}, R_$ It is important to note that model-free control methodologies may be applied to numerical or experimental systems with linet modification. All of these medle-free methods have somes not on macroscopic objective function, typically based on sensor massurements: (past and present). Some challenging real-world example objectives in different disciplines include:

- Fluid dynamics: In aerodynamic applications, the goal is often some combination of drag reduction, lift increase, and noise reduction, while in pharmaceusical and chemical engineering applications the goal may involve mixing enhancement.
- Finance: The goal is often to maximize profit at a given level of risk tolerance, subject to the law.
- Epidemiology: The goal may be to effectively suppress a disease with constraints of sensing (e.g., blood samples, clinics, etc.) and actuation (e.g., vaccines, bed nets, etc.).
- Industry: The goal of increasing productivity must be balanced with several constraints, including labor and work safety laws, as well as environmental impact, which other have significant uncertainty.
- Autonomy and robotics: The goal of self-driving cars and autonomous robots is to achieve a task while interacting safely with a complex environment, including cooperating with human agents.

In the examples above, the objective involve some minimization or maximization of a given quarkin splice) to some contraints. The plane that, as in the case of discase apprecision on a fixed badget, or they may involve a complex multi-objective minimized. Other, constraint optimizations will real in solutions that there are boundary of the constraint, which may explain why many comparisot optimized that the solution of the constraint, which may explain why many comparisot optimized that the solution of the constraint, which may explain why many comparisot optimized that the solution of the constraint, which may explain why many comparisot optimized that the solution of the constraint, which may explain with the solution of the solution, hance is a powersed by human behavior and constraint, and discase sprands is the result of a complex interaction of thology, multion behavior, and georegrades.

These real-world control problems are extremely challenging for a number of reasons. They are high-dimensional and strongly nonlinear, cflow with millions or billions of degrees of freedom that evolve according to possibly unknown nonlinear interactions. In addition, it may be exceedingly expensive or infansible to rand different scenarios for system identification, for example, there are serious ethical issues associated with testing different vaccination strateneises when human lives are at stake.

Increasingly, challenging optimization problems are being volved with machine learning, beeraging the availability of vust and increasing quantities of data. Many of the recent successes have been on static data (e.g., image classification, speech recognition, etc.), and marketing tasks (e.g., online sales and a dplacement). However, current efforts are applying machine learning to analyze and control complex systems with dynamics, with the potential to revolutionize our addity to interact, with and maniplated these systems.

The following sections describe a handful of powerful learning techniques that are being widely applied to control complex systems where models may be unavailable. Note that the relative importance of the following methods are not proportional to the amount of space dedicated.

Reinforcement Learning

Reinforcement harming (RL) is an important discipline at the intersection of machine harming and county (C)(n) and in its currently bring used horshy by companies such as Cougle for generatined artificial integrace, antonomous robos, and self-draving cars. In advection theory despection of the system and the county of the Markov Acision porcess, where the dynamics of the system and the county of the Markov Acision porcess, where the dynamics of the system and the county of a possibility of entires, possibility of entires, possibility of entires of possibilities entires, possibilities entires, possibility of the system and the county of possibility of entires of possibilities of entires of the system and the county of possibility of entires of the system and the county of possibility of entires of the system and the county of the system of the system and the county of the system of the system of the system and the county of the system of the system and the county of the system of the system of the system and the county of the system and the county of the system of the syste

Reinforcement learning may be viewed as partially supervised, since it is not always known immediately if a control action was effective or not. In RL, a control policy is enacted by an avent, and this agent may only receive partial information about the effectiveness of their control strategy. For example, when learning to play a game like tic-tac-toe or chess, it is not clear if a specific intermediate move is responsible for winning or losing. The player receives binary feedback at the end of the game as to whether or not they win or lose. A major challence that is addressed by RL is the development of a value function, also known as a quality function O, that describes the value or quality of being in a particular state and making a particular control policy decision. Over time, the agent learns and refines this O function, improving their ability to make good decisions. In the example of chess, an expert player begins to have intuition for good strategy based on board position, which is a complex value function over an extremely high-dimensional state space (i.e., the space of all possible board configurations). Q-learning is a model-free reinforcement learning strategy, where the value function is learned from experience. Recently, deep learning has been leveraged to dramatically improve the Q-learning process in situations where data is readily available [336, 385, 386, 384]. For example, the Google DeepMind algorithm has been able to master many classic Atari video games and has recently defeated the best players in the world at Go. We leave a more in-denth discussion of reinforcement learning for other books, but emphasize its importance in the growing field of machine learning control

Iterative Learning Control

Turitive learning control (RL) [55, 75, 18, 18, 18) [6] is a solidy out cloudlage multiturning the second of times. Its contrast is the feedback control second second



Figure 10.6 Depiction of parameter cube for PID control. The genetic algorithm represents a given parameter value as a generic response that concatenates the various parameters. In this example, the parameters are expressed in binary representation that is scaled so that **000** is the minimum bound and 111 is the users bound. Color indicates the cost associated with each treatmeter value.

Genetic Algorithms

The generic algorithm (Ed.) is one of the cartest and implete algorithms for parameter parameters and the solution of the cartest and implete algorithm. In the solution and finance [202, 104, 202] CGA is Regardly used to use and algorithm (Ed.) and intermediate the solution of the solution of the solution of the solution of the solutions are used as a solution of the solution of the solution of the solution of the HO more solution of the solution of the solution of the solution of the HO more solution of the filter solution of the solu

- Elitism (optional): A set number of the most fit individuals with the best performance are advanced directly to the next generation.
- Replication: An individual is selected to advance to the next generation.
- Crossover: Two individuals are selected to exchange a portion of their code and then advance to the next generation; crossover serves to exploit and enhance existing successful strategies.
- Mutation: An individual is selected to have a portion of its code modified with new values; mutation promotes diversity and serves to increase the exploration of parameter space.

For the replication, crossover, and mutation operations, individuals are randomly selected to advance to the next generation with the probability of selection increasing with fitness. The genetic operations are illustrated for the HD control example in Fig. 10.7. These generations are evolved until the fitness of the top individuals converges or other stopping criteria are met.



Figure 10.2 Schematic illustrating evolution in a generic algorithm. The individuals in generators k are card e-valuated and madei an accending order based on their con function, which is inservely proportional to their probability of a schecing for generic operations. Then, individuals are chosen based on this weighted probability for advancement to generations k + 1 using the force operations: clisiss, reglectation, crossover, and mutation. This forms generations k + 1 such the togenese is exceeded with the condition table togenese in second-

Cenctic algorithms are generally used to find nearly globally optimal parameter values, as they are capible exploring and explosing local wells in the cont function. CA you vides a middle ground between a botte-force search and a convex optimization, and is an alternative to expensive. Most: Carlo sampling, wilch does not scale to high-dimensional parameter spaces. However, there is no gaarantee that genetic algorithms will converge to a globally optimal solution. There are also a sumber of hype-parameters that any affect performance, including the size of the populations, muther of generations, and relative selection nites of the vision stenetic neuration.

Genetic algorithms have been widely used for optimization and control in nonlinear systems [184]. For example, GA was used for parameter tuning in open loop control [394], with applications in jet mixing [304], combustion processes [101], wake control [431, 192], and drag reduction [201]. GA has also been employed to tune an H_{∞} controller in a combusion experiment [233].

Genetic Programming

Geneics programming (CP) 1027, 300, 300 is a powerful generalization of genetic algorithm that simultaneously operations both the structure and parameters of an impur-coupts maps. Recently, genetic programming has also been used to obtain counted have that may seems defective that the encoding of complexity defective transmission of the structure and parameters. Seems on all counting are complexed and the event counted larger barries and parameters. Seems on all counting are complexed and probability are shown about the transmission of consource matation, and probability and structure that the parameters of the counting are complexed and probability and with the counted parameters. Seems on the counting are complexed and probability and the next parameters of the counted probability of the counted parameters and target and there, as desired and probability of the counted parameters and target parameters of the counted parameters of the counted parameters and target parameters of the counted parameters of the counted parameters and target parameters. Seeming and the counted parameters of the counted parameters and target parameters of the counter of the parameters of the parameters

Genetic programming has been recently used with impressive results in turbulence control experiments, led by Bernd Noack and collaborators [403, 417, 199, 168, 169, 416].



Figure 10.8 Illustration of function tree used to represent the control law u in genetic programming control.

Example: Genetic Algorithm to Tune PID Control

In this example, we will use the genetic algorithm to tune a proportional-integral-derivative (PID) controller. However, it should be noted that this is just a simple demonstration of evolutionary algorithms, and such heavy machinery is not recommended to tune a PID controller in practice, as there are far simpler techniques.

PID control is among the simplest and most videly used control architectures in indution control systems, including for motor position and velocity control, for tuning of various sub-systems in an automobile, and for the pressure and temperature controls in modern expression machines, to sume only a few of the myridal applications. As its name suggests, and and the integral and derivative in time. A schematic of PID control is shown in Fig. 10,100.

In the cruise control example in Section 8.1, we saw that it was possible to reduce reference tracking error by increasing the proportional control gain K_P in the control law $u = -K_P(w_r - y)$. However, increasing the gain may eventually cause instability in some systems, and it will not completely eliminate the steady-state tracking error. The addition



Figure 10.9 Genetic operations used to advance function trees across generations in genetic programming control. The relative selection rates of replication, erossover, and mutation are p(R) = 0.1, p(C) = 0.7, and p(M) = 0.2, respectively.

of an integral control term, $K_I \int_0^x (w_r - y)$ is useful to eliminate steady-state reference tracking error while alleviating the work required by the proportional term.

There are formal rules for how to choose the PID gains for various design specifications, such as fast response and minimal overshoot and ringing. In this example, we explore the use of a genetic algorithm to find effective PID gains to minimize a cost function. We use an LOR cost function

$$J = \int_{0}^{T} Q(w_r - y)^2 + Ra^2 dr$$

with Q = 1 and R = 0.001 for a step response $w_r = 1$. The system to be controlled will be given by the transfer function

$$G(s) = \frac{1}{s^4 + s}$$
.



Figure 10.10 Proportional-integral-derivative (PID) control schematic. PID remains ubiquitous in industrial control.

The first step is to write a function that evaluates a given PID controller, as in Code 10.1. The three PID mins are stored in the variable **parms**.

Code 10.1 Evaluate cost function for PID controller

Next, it is relatively simple to use a genetic algorithm to optimize the PID control gains, as in Code 10.2. In this example, we run the GA for 10 generations, with a population size of 25 individuals per generation.

Code 10.2 Genetic algorithm to tune PID controller.

```
dt = 0.001;
Profile = 2.15;
North (1.15);
a = 1/(ne(nessen1));
a = 1/(ne(nessen1));
Automatic art, North Networks, (output for, inty(int);
North (nessen1);
(n, Nu1) = ga((d) picture((n, N, 1), -sym(1), secon(3, 1),
...(1, 0, 1, 0, 0, 0, options);
```

The results from intermediate generations are saved using the custom output function in Code 10.3.

```
Code 10.3 Special output function to save generations.
```

```
fmaction (rists, grt, opthalpad) mython(grts, state, flag)
protingent = false;
mython (rist, state, flag)
fase 'site'
case 'site', 'sater, provide (rist, 'sater, flag)
fase 'site', 'sater, grt(')
hardrow(', sater, flag)
fase 'site (hardrow(', sater, sater
```

The evolution of the cost function across various generations is shown in Fig. 10.11. As the generations progress, the cost function steadily decreases. The individual gains are shown in Fig. 10.12, with redder dots corresponding to early generations and bluer generations corresponding to later generations. As the genetic algorithm progresses, the PDP rains beein to obser around the contant solution (bluck circle).

Fig. 10.13 shows the output in response to the PID controllers from the first generation. It is clear from this plot that many of the controllers fail to stabilize the system, resulting in large deviations in y. In contrast, Fig. 10.14 shows the output in response to the PID controllers from the last generation. Overall, these controllers are more effective at producing a suble step resonance.

The best controllers from each generation are shown in Fig. 10.15. In this plot, the controllers from early generations are redder, while the controllers from later generations



Figure 10.11 Cost function across generations, as GA optimizes PID gains.



Figure 10.12 PID gains generated from genetic algorithm. Red points correspond to early generations while blue points correspond to later generations. The black point is the best individual found by GA.



Figure 10.13 PID controller response from first generation of genetic algorithm.

are bluer. As the GA progresses, the controller is able to minimizes output oscillations and achieves fast rise time.

10.3 Adaptive Extremum-Seeking Control

Although there are many powerful techniques for model-based control design, there are also a number of animoback. First, in many systems, there may not be access to a model, or the model may not be withhelp for control (i.e., there may be strong nonlinearities or the model may be represented in a normational form). Nexe, were after an attraction has been identified and the dynamics characterized, control may invalidate this model by modifying the attractor, griving rise to new and mextracterized dynamics. The obvious exception is the attractor, griving rise to new and mextracterized dynamics. The obvious exception is



Figure 10.14 PID controller response from last generation of genetic algorithm.



Figure 10.15 Best PID controllers from each generation. Red trajectories are from early generations, and blue trajectories correspond to the last generation.

stabilizing a fixed point or a periodic orbit, in which case effective control keeps the system in a neighborhood where the linearized model remains accurate. Finally, there may be slow changes to the system that modify the underlying dynamics, and it may be difficult to measure and model these effects. The field of adaptive countel breadly addresses these challenges, by allowing the coatrol law the field/billy to modify its action based on the changing dynamics of a system. Extremum-secking control (BSC) [12], 19] is a particularly attractive form of adaptive control for complex system because it does not rely on an underlying model and it has guaranteed convergence and stability under a set of well-defined conditions. Extremumseeking may be used to rack local nations of an objective function, cheptie disturbance, varying system parameters, and atominearities. Adaptive control may be ingitemented for isdance control or user for down maint of parameters in a working controller.

Extremess-tecking control may be thought of as an advanced perturb-and-observe method, whereby a sinusoidal perturbation is addively injected in the actuation signal and used to estimate the gradient of an objective function. If that should be maximized or minimized, The objective function is generally compared based on scores measurements of the system, although it utilized depends on the internal dynamics and the choice of the inject signal. The strength endowed the objective function is greater of the site of the system, score measurements of the control of the system, and the strength endowed the strength endowed the system seture is a strength endowed the strength endowed the strength endowed the relation of the strength endowed the strength endowed

The extremum-seeking control architecture is shown in Fig. 10.16. This schematic depicts ESC for a scalar input *a*, albody the methods resulting generalize for vector-valued inputs *u*. A convex objective function $I_{(A)}$, is shown in Fig. 10.17 for static plant dynamics (i.e., for $\gamma = u$). The extremum-secking controller uses an input perturbation to estimate the gradient of the objective function J and steer the mean actuation signal towards the comminism values.



Figure 10.16 Schematic illustrating an extremum-secking controller, A sinusoidal perturbation is added to the best gauses of the input a, and it gauses through the plant, resulting in a sinusoidal output perturbation that may be observed in the sensor signal y and the cost J. The high-pass filter results in a zero-moust oper transform, which is then multiplied (demodulated) by the same input perturbation testing in the signal g. This demodulated signal is finally integrated into the best mess i. for the origination rate a.



Figure 10.17 Schematic illustrating extremum-seeking control on for a static objective function J(u). The output perturbation (red) is in phase when the input is left of the peak value (i.e. $u < u^{+}$) and out of phase when the input is to the right of the peak (i.e. $u > u^{+}$). Thus, integrating the predicat of input and output sinusoids moves it towards u^{+} .

Three distinct time-scales are relevant for extremum-seeking control:

- slow external disturbances and parameter variation:
- medium perturbation frequency ω;
- fast system dynamics.

In many systems, the internal system dynamics evolve on a fast time-scale. For example, turbulent fluctuations may equilibrate rapidly compared to actuation time-scales. In optical systems, such as a fiber laser [93], the dynamics of light inside the fiber are extremely fast compared to the time-scales of actuation.

In extremum-seeking control, a sinusoidal perturbation is added to the estimate of the input that maximizes the objective function, \hat{w} :

$$u = \hat{u} + a \sin(\omega t)$$
. (10.23)

This input perturbation passes through the system dynamics and output, resulting in an objective function J that varies simusoidally about some mean value, as shown in Fig. 10.17. The output J is high-pass filtered to remove the mean OEC componenty, resulting in the oscillatory signal p. A simple high-pass filter is represented in the frequency domain as

$$\frac{s}{s + a_{0}}$$
 (10.24)

where s is the Laplace variable, and ω_0 is the filter frequency. The high-pass filter is chosen to pass the perturbation frequency ω . The high-pass filtered output is then multiplied by the input sinuscial, possibly with a phase shift ϕ_i resulting in the devokadated signal ξ :

$$\xi = a \sin(\omega t - \phi)\rho.$$
 (10.25)

This signal ξ is mostly positive if the input *u* is to the left of the optimal value *u*^{*} and it is mostly negative if *u* is to the right of the optimal value *u*^{*}, shown as red curves in Fig. 10.17. Thus, the demodulated signal ξ is integrated into \hat{u} , the best estimate of the optimizing value

$$\frac{d}{dt}\hat{u} = k \xi,$$
 (10.26)

so that the system estimate \hat{u} is steered towards the optimal input u^* . Here, k is an integral gain, which determines how aggressively the actuation climbs gradients in J.

Roughly speaking, the demodulated signal g measures gradients in the objective function, so that the algorithm elimbs to the optimum more rapidly when the gradient is larger. This is simple to see for constant plant dynamics, where J is simply a function of the input $J(a) = J(\hat{a} + a \sin(ac))$. Expanding J(a) in the perturbation amplitude a, which is assumed to be small, yields:

$$J(u) = J(\hat{u} + a \sin(\omega t))$$
 (10.27a)

$$= J(\hat{u}) + \frac{\partial J}{\partial u}\Big|_{u=\hat{u}} \cdot a \sin(\omega t) + O(a^2).$$
 (10.27b)

The leading-order term in the high-pass filtered signal is $\rho \approx \partial J/\partial u|_{u=\hat{u}} \cdot a \sin(\omega t)$. Averaging $\xi = a \sin(\omega t - \phi)\rho$ over one period yields:

$$\xi_{avg} = \frac{\omega}{2\pi} \int_{0}^{2\pi/\omega} a \sin(\omega t - \phi) \rho \, dt$$
 (10.28a)

$$= \frac{\omega}{2\pi} \int_{0}^{2\pi/\omega} \frac{\partial J}{\partial u} \Big|_{u=0} a^2 \sin(\omega t - \phi) \sin(\omega t) dt \quad (10.28b)$$

$$= \frac{a^2}{2} \frac{\partial J}{\partial u} \bigg|_{u=0} \cos(\phi). \quad (10.28c)$$

Thus, for the case of trivial plant dynamics, the average signal ξ_{avg} is proportional to the gradient of the objective function J with respect to the input u.

In general, extremum-socking counter may be applied to systems with nonlinical dynamic relating the large at a low endrys of that can estimation the simulation merging analysis of the complications the simplication merging analysis of the simulation of the simula



Figure 10.18 Extremum-seeking control response for cost function in (10.29).

used for frequency selection and slope-seeking is used for amplitude selection when tuning an open-loop periodic forcing.

It is important to note that extremum-seeking control will only find local maxima of the objective function, and there are so guarantees that this will correspond to a global maxima. Thus, it is important to start with a good initial condition for the optimization. In a number of studies, extremum-seeking control is used in conjunction with other global optimization techniques, such as a genetic algorithm, or sparse representation for classification[191,99].

Simple Example of Extremum-Seeking Control

Here we consider a simple application of extremum-seeking control to find the maximum of a static quadratic cost function.

$$J(u) = 25 - (5 - u)^2$$
, (10.29)

This function has a single global maxima at w = 5. Starting at u = 0, we apply extremomseeking control with a perturbation frequency of $\omega = 10$ Hz and an amplitude of a = 0.2. Fig. 10.18 shows the controller response and the rapid tracking of the optimal value $u^{+} = 5$. Code 10.4 shows how extremum-seeking may be implemented using a simple Butterworth hist-bases filter.

Notice that when the gradient of the cost function is larger (i.e., closer to u = 0), the oscillations in J are larger, and the controller climbs more rapidly. When the input u gets close to the oortimum value at $u^* = 5$, even though the input be entroling has the same amplitude a, the output perturbation is nearly zero (on the order of a^2), since the quadratic cost function is flat near the peak. Thus we achieve fast tracking far away from the optimum value and small deviations near the peak.

Code 10.4 Extremum-seeking control code

```
J = @(u,t)(25-(5-(u)),^2);
freq = 10.2.pi; } sample frequency
T = 10; } total period of simulation (in seconds)
A = .2; # amplitude
omega = 10+2+pi; $ 10 Ma
   t = (i-1) • dt:
       HPF(k) = HPF(k+1):
   vs(butterorder+1) = vvals(i);
       HPFnew = HPFnew - a(k) .HPF(butterorder+2-k);
   HPF(butterorder+1) = HPFnew;
   xi = HPFnew+sin(omega+t + phase);
   uhat = uhat + xi .K.dt:
   u = uhat + A+sin(omega+t + phase);
   uhats(i) = uhat;
```

To see the ability of extremum-seeking control to handle varying system parameters, consider the time-dependent cost function given by

$$J(u) = 25 - (5 - u - \sin(t))^2. \quad (10.30)$$

The varying parameters, which oscillate at $1/2\pi$ Hz, may be consider slow compared with the perturbation frequency 10 Hz. The response of extremm—seeking control for this slowly varying system is shown in Fig. 10.19. In this response, the actuation signal is able



Figure 10.19 Extremum-seeking control response with a slowly changing cost function J(u, t).

to maintain good performance by oscillating back and forth to approximately track the oscillating optimal *uv*, which oscillates between 4 and 6. The output function *J* remains close to the optimal value of 25, despite the unknown varying parameter.

Challenging Example of Extremum-Seeking Control

Here we consider an example inspired by a challenging benchmark problem in Section 1.3 of [19]. This system has a time-varying objective function J(t) and dynamics with a right-half plane zero, making it difficult to control.

In one formulation of extremum-secking [133, 19], there are additional guidelines for designing the controller if the plant can be split into three blocks that define the input dynamics, at line-varying objective functions with no internal dynamics, and the output dynamics, as shown in Fig. 10.20. In this case, there are procedures to design the high-pass fiber and integraptor blocks.

In this example, the objective function is given by

$$J(\theta) = .05\delta(t - 10) + (\theta - \theta^{*}(t))^{2}$$

where δ is the Dirac delta function, and the optimal value $\theta^{*}(t)$ is given by

$$\theta^* = .01 + .001t$$
.



Figure 10.20 Schematic of a specific extremum-seeking control architecture that benefits from a wealth of desire techniques [133, 19].

The optimal objective is given by $J^* = .058(t - 10)$. The input and output dynamics are taken from the example in [19], and are given by

$$F_{in}(s) = \frac{s-1}{(s+2)(s+1)},$$
 $F_{out}(s) = \frac{1}{s+1}.$

Using the design procedure in [19], one arrives at the high-poss filter s/(a + 5) and an integrate-like bek (yruch § 50(a - 10/(a - 0.1)). In addition, a perturbation with a = 5 and a = 0.05 is used, and the demodulating perturbation is phase-ahilted by $\phi = .7985$. The high particular by evaluating the input function $T_{\rm eff}$ at $a_{\rm eff}$ becomes of this controller is shown in Fig. 10.21, along with the Simulatik implementation Fig. 10.22. The controller is able to accurately runk the optimizing input, despite additive sensor noise.

Applications of Extremum-Seeking Control

Because of the lack of assumptions and ease of implementation, extremum-seeking control has been widely applied to a number of complex systems. Although ESC is generally applicable for instime control of dynamical systems, it is also widely used as an online optimization algorithm that can adapt to slow changes and disturbances. Among the many uses of extremum-secking control, here we highlight only a few.



Figure 18.21 Extremum-seeking control response for a challenging test system with a right-half plane zero, inspired by [19].



Figure 10.22 Simulink model for extremum-seekine controller used in Fig. 10.21.

bioreactors [546], PID [289] and PI [311] tuning, active braking systems [568], and control of Tokamaks [413].

Extremum-seeking has also been broadly applied in turbulent flow control. Despite the ability to control dynamics in-time with ESC, it is often used as a slow feedback optimization to ture the parameters of a working open-long controller. This is how feedback has may benefits, such as maintaining performance despite dow changes to environment contains. Recommon sceking has been used to control at a staff flow compresses [273], the appendix performance of the provide scenario of the sequence and contains. The staff of the provide scenario of the sequence and contains and the staff of the sequence and the staff of the squence and the staff of the staff of the staff of the staff of the squence contains. If a staff of the staff of the staff of the staff of the squence of the staff of the staff of the staff of the staff of the squence of the staff of the

There are numerous extensions to extremum-seeking that improve performance. For example, extended Kalman filters were used as the filters in [2021 to control thermozocousic instabilities in a combustor experiment, reducing pressure fluctuations by nearly 404B, Kalman filters were also used with EVC to reduce the flow separation and increase the pressure ratio in a high-pressure axia fan using an injected pulsed air stream [553]. Including the Kalman filters were novel the controller bandwidth by a factor of 10 over transitional ESC.

Suggested Reading

Texts

- Reinforcement learning: An introduction, by R. S. Satton and A. G. Barto, 1998 15071.
- (2) Real-time optimization by extremum-seeking control, by K. B. Ariyur and M. Krstić. 2003 [19].
- (3) Machine learning control: Taming nonlinear dynamics and turbulence, by T. Duricz, S. L. Brunton, and B. R. Noack, 2016 [167].
- (4) Model predictive control, by E. F. Camancho, C. B. Alba, 2013 [107].

Papers and Reviews

- Stability of extremum seeking feedback for general nonlinear dynamic systems, by M. Krstić and H. H. Wang, Automatica, 2000 [312].
- (2) Dynamic mode decomposition with control, by J. L. Proctor, S. L. Brunton, and J. N. Kutz, SIAM Journal on Applied Dynamical Systems, 2016 [434].
- (3) Model predictive control: theory and practice a survey, by C. E. Garcia, D. M. Prett, and M. Morari, Automatica, 1989 [195].
- (4) Closed-loop turbulence control: Progress and challenges, by S. L. Brunton and B. R. Noack, Applied Mechanics Reviews, 2015 [94].

Reduced Order Models
The proper orthogonal decomposition (POD) is the SVD algorithm applied to partial differential equations (PDEs). As such, it is one of the most important dimensionality reduction techniques available to study complex, spatio-temporal systems. Such systems are typically exemplified by nonlinear partial differential equations that prescribe the evolution in time and space of the quantities of interest in a given physical, engineering and/or biological system. The success of the POD is related to the seeminaly ubiquitous observation that in most complex systems, meanineful behaviors are encoded in low-dimensional patterns of dynamic activity. The POD technique seeks to take advantage of this fact in order to produce low-rank dynamical systems canable of accurately modeling the full spatiotemporal evolution of the governing complex system. Specifically, reduced order models (ROMs) leverage POD modes for projecting PDE dynamics to low-rank subspaces where simulations of the governing PDE model can be more readily evaluated. Importantly, the low-rank models produced by the ROM allow for significant improvements in computational speed, potentially enabling prohibitively expensive Monte-Carlo simulations of PDE systems, optimization over parametrized PDE systems, and/or real-time control of PDEbased systems. POD has been extensively used in the fluids dynamics community [251]. It has also found a wide variety of applications in structural mechanics and vibrational analysis [287, 23, 232, 329], optical and MEMS technologies [333, 488], atmospheric sciences (where it is called empirical orthogonal functions (EOEs)) [116-117] wind engineering applications [494], acoustics [181], and neuroscience [33, 519, 284]. The success of the method relies on its ability to provide physically interpretable spatio-temporal decompositions of data [316, 57, 181, 286, 126, 333].

11.1 POD for Partial Differential Equations

Throughout the engineering, physical and bulkopical sciences, may system are known boots prescribed attractionals, between time and prove that draws parsets of dynamic observes, dynamics that survivate a spectra of the spectra of the spectra observes. A parameters have a spectra of the same fragments of the spectra of to a desired level of accuracy, with higher accuracy requiring a larger dimension of the discretized system. Hist technique, the high demicultural, sarring that simply a consequence of the underlying numerical solution scheme. In contrast, asymptotic reduction effects are larger the complex system with an aimple set of equations presently that are frame as not be annealed to analysis. Before the 1966s and the rise of computing and arguments includence in order that the strength of the argument includence and arguments includence in the strength of the argument interaction. A symptotic includence are covered in this book, her the computational methods the analyst ender strength ended are as a covered on this book, her the computational methods.

To be more mathematically precise about our study of complex systems, we consider generically a system of nonlinear PDEs of a single spatial variable that can be modeled as

$$u_r = N(u, u_r, u_{rr}, \dots, x, t; \beta)$$
 (11.1)

where the subscripts denote partial differentiation and N₀ prescribes the generically nonline evolution. The partial probability of the structure parameter for on later conduction and the structure of the structure parameters of the structure parameters of the structure of the structure parameters of the structure parameters of the structure of the structure of the structure parameters of the structure parameters (111) as a ster of outline y differential equations (ODEs). The standard PER methods of the structure of the structure of the structure parameters of the structure parameters (111) as a ster of outline y differential equations (ODEs). The standard PER methods of Date is the form of an ODEs is bounded variety of analytic methods, can be applied in an or other amounted on each in its dimension behavior.

Although a number of potential solution strategies have been mentioned, (111) does not admit a closed from colution in general. Even the simplest nonlinearity or a squitable dependent coefficient can reader the standard analytic solutions strategies useless. However, comparational strategies for solving (111) are admadat and have provided transformative insights across the physical, engineering and biological sciences. The various compartional techniques devide lead to a approximate numerical solution of (111), which is to high-dimension. Consider, for instance, a standard spatial discretization of (11.1) whereby the spatial variable is a volumed at n p topions.

$$u(x_k, t)$$
 for $k = 1, 2, \dots, n$ (11.2)

with spacing $\Delta x = x_{k+1} - x_k = 2L/n$. Using standard finite-difference formulas, spatial derivatives can be evaluated using neighboring spatial points so that, for instance,

$$\mathbf{u}_{t} = \frac{\mathbf{u}(x_{k+1}, t) - \mathbf{u}(x_{k-1}, t)}{2\Delta x}$$
(11.3a)

$$\mathbf{u}_{ax} = \frac{\mathbf{u}(x_{k+1}, t) - 2\mathbf{u}(x_k, t) + \mathbf{u}(x_{k-1}, t)}{\Delta x^2}.$$
 (11.3b)

Such spatial discretization transforms the governing PDE (11.1) into a set of n ODEs

$$\frac{d\mathbf{u}_{k}}{dt} = \mathbf{N} \left(\mathbf{u}(x_{k+1}, t), \mathbf{u}(x_{k}, t), \mathbf{u}(x_{k-1}, t), \cdots, x_{k}, t, \beta \right), \quad k = 1, 2, \cdots, n. \quad (11.4)$$

This process of discretization produces a more manageable system of equations at the expense of rendering (11.1) high-dimensional. It should be noted that as accuracy requirements become more stringent, the resulting dimension no of the system (11.4) also increases, since $\Delta x = 2L/n$. Thus, the dimension of the underlying computational scheme is artificially determined by the accuracy of the finite-difference differentiation schemes.

The spatial discretization of (11.1) illustrates how high-dimensional systems are enered. The article production of high-dimensional systems is obligationa across computational schemes and presents significant challenges for scientific computing efforts. To their illustrate high photometors, we consider a second computational scheme for solving (11.1) Is particular, we consider the most common technique for analytically solving (11.1) and particular, we consider the most common technique for analytically using (11.1) in particular, we consider the most common technique for analytically solving in are independent, to that

$$u(x, t) = a(t)\psi(x)$$
 (11.5)

where the variable a(r) subsumes all the time dependence of (11.1) and $\psi(x)$ characterizes the spatial dependence. Separation of variables is only guaranteed to work amplically if (11.1) is linear with constant coefficients. In that restrictive case, two differential equations can be derived that separately characterize the spatial and temporal dependences of the complex system. The differential equations are related by a constant parameter that is present in each.

For the general form of (11.1), separation of variables can be used to yield a computational algorithm capable of producing accurate solutions. Since the spatial solutions are not known a priori, it is typical to assume a set of basis modes which are used to construct (y(1), Indeed, such assumptions on basis modes underlast the critical lates as of the method of eigenfunction expansions. This yields a separation of variables solution ansatz of the form

$$\mathbf{u}(x, t) = \sum_{k=1}^{n} \mathbf{a}_{k}(t)\psi_{k}(x)$$
 (11.6)

where $\psi_i(x)$ form a set of $n \gg 1$ basis modes. As before, this expansion attificially renders a high dimensional system of equations since n modes are required. This separation of variables solution approximates the true solution, provided n is large enough. Increasing the number of modes n is equivalent to increasing the spatial discretization in a finitedifference scheme.

The orthogonality properties of the basis functions $\psi_1(x)$ enable us to make use of (11.6). To illustrate this, consider a scalar version of (11.1) with the associated scalar separable solution $u(x, t) = \sum_{k=1}^{n} a_k(t)\psi_k(x)$. Inserting this solution into the governing equations gives

$$\sum \psi_k \frac{da_k}{dt} = \mathbf{N} \left(\sum a_k \psi_k, \sum a_k (\psi_k)_z, \sum a_k (\psi_k)_{zz}, \cdots, x, t, \beta \right) \quad (11.7)$$

where the sums are from $k = 1, 2, \dots, n$. Orthogonality of our basis functions implies that

$$\langle \psi_k, \psi_j \rangle = \delta_{kj} = \begin{cases} 0 & j \neq k \\ 1 & j = k \end{cases}$$
(11.8)

where δ_{kl} is the Kronecker delta function and (ψ_k, ψ_l) is the inner product defined as:

$$|\psi_k, \psi_j| = \int_{-L}^{L} \psi_k \psi_j^* dx$$
 (11.9)

where * denotes complex conjugation.

Once the modal basis is decided on, the governing equations for the $a_k(t)$ can be determined by multiplying (11.7) by $\psi_j(x)$ and integrating from $x \in [-L, L]$. Onthogonality then results in the temporal governing equations, or Galerkin projected dynamics, for each mode

$$\frac{du_k}{dt} = \left\langle N \left(\sum a_j \psi_j, \sum a_j (\psi_j)_k, \sum a_j (\psi_j)_{kl}, \cdots, x, l, \beta \right), \psi_k \right\rangle \quad k = 1, 2, \cdots, n.$$
(11.10)

The given form of $N(\cdot)$ determines the mode-coupling that occurs between the various n modes. Indeed, the hallmark feature of nonlinearity is the production of modal mixing from (11.10).

Numerical schemes based on the Galarkin projection (11.16) are a commonly used to prevent approximation for the full preventing system (11.1). Conserptore to the nucleablance of the resolutions can be accomplianted by both judicious choice of the model have functions μ_{ij} as well as more than the standard state of the standard state. The state of th

Fourier Mode Expansion

The most profile basis used for the Galarkia projection technique is Fourier modes. More preciscly, the fait Poweir transfrom (FTP) and its vurtains the dominated scientificour pluting applied to the engineering, physical, and biological sciences. There are two pimury reasons for this: (5) There is a strong intrinsition developed around the meaning of Fourier modes as it directly relates to spatial wavelengths and frequencies, and more importantly, (6) the algorithm encessary to compare the relational developed around the properturb, (6) the algorithm encessary to compare the relational developed around projections. The scenar of the loss construm and a foundational correstored or clearithic computing.

The Fourier mode basis elements are given by

$$\psi_{E}(x) = \frac{1}{L} \exp \left(i \frac{2\pi kx}{L}\right)$$
 $x \in [0, L]$ and $k = -\pi/2, \dots, -1, 0, 1, \dots, \pi/2 - 1.$

(11.11)

It should be noted that in most software packages, including Matlab, the FFT command assumes that the spatial interval is $x \in [0, 2\pi]$. Thus one must rescale a domain of length L to 2π before using the FFT.

Obviously the Fourier modes (11.11) are complex periodic functions on the interval $x \in [0, L]$. However, they are applicable to a much broader class of functions that are not mecessarily periodic. For instance, consider a localized Gaussian function

$$u(x, t) = \exp(-\sigma x^2) \qquad (11.12)$$

whose Fourier transform is also a Gaussian. In representing such a function with Fourier modes, a large number of modes are often required since the function itself isn't periodic. Fig. 11.1 shows the Fourier mode representation of the Gaussian for three values of σ . Of note is the fact that a large number of modes is required to represent this simple function,



Figure 11.1 Illustration of Fourier modes for representing a bacelized Gaussian pubs. (a) n = 80Fourier modes are used to represent the Gaussian $u(x) = ueg(-nx^2)$ in the domain $x \in [-1, 0]$ to for $\sigma = 0.1$ (red), n = 1 (black) and $\sigma = 10$ (black). (b) The Fourier mode representation of the Gaussian, showing the modes required for an accurate representation of the bacelized function. (c) The convergence of the n mode solution to the actual Gaussian $(\sigma = 1)$ with the (d) L² error from the true value for the three values of σ .

especially as the Gaussian withh is decreased. Although the FPT algorithm is extremely fut and widsly applied, one can see immediately that a large number of modes are generically required to represent simple functions of interest. Thus, solving problems using the FPT for nequires high-dimensional representations (i.e., $n \gg 1$) to accommodate generic, localized spatial behaviors. Ultimately, our aim is to move away from artificially creating such high-dimensional problems.

Special Functions and Sturm-Liouville Theory

In the 1800s, and early 1900s, mathematical physics developed many of the governing principles beind near low, electromagnetism and quantum mechanics, for instance. Many of the hallmark problems considered were driven by *linux* dynamics, allowing for analyically metalbe solutions. And since these problems arose before the advect of computing, nonlinearities were typically treated as perturbations to an underlying linear equation. Thus one or ehne considered complex systems of the form

$$u_r = Lu + \epsilon N (u, u_r, u_{xx}, \dots, x, t, \beta)$$
 (11.13)

where L is a linear operator and $\epsilon \ll 1$ is a small parameter used for perturbation calculations. Often in mathematical physics, the operator L is a Sturm-Lioaville operator which guarantees many advantageous properties of the eigenvalues and eigenfunctions.

To solve equations of the form in (11.13), special modes are often used that are ideally suited for the problem. Such modes are eigenfunctions of the underlying linear operator L in (11.13):

$$L\psi_k = \lambda_k \psi_k$$
 (11.14)

where $\psi_{2}(x)$ are orthonormal eigenfunctions of the operator **I**. The eigenfunctions allow for an eigenfunction expansion solution whereby $\mathbf{u}(x, t) = \sum a_{k}(t)\psi_{2}(x)$. This leads to the following solution form

$$\frac{da_k}{dt} = (\mathbf{Lu}, \psi_k) + \epsilon \langle \mathbf{N}, \psi_k \rangle . \quad (11.15)$$

The key kess in such as requestors in the the eigenfunctions processingly are lated for the object of the step of an extra product to the product to the step of the step of

Dimensionality Reduction

The examples above and solution methods for PDFs illutrate a common problem of scientific computing: the generation of a degree high-dimensional systems. For many complex PDFs with several spatial dimensions, it is not amcommon for discretization or modul expansion techniques to yield systems of differential equations with millions or billions of degrees of flexods. Such large systems are accurredly demanding for even the latest computational architectures, limiting accuracies and run-times in the modeling of many complex systems, such as high Reynolds mutter hand the flax.

To add in comparation, the selection of a set of optimal bioss mode is critical, at in our probability due to the matter of the distribution of parameters. Here, so show its behaviour probability due to the matter of the distribution of the selection of the selection of the distribution of the distribution of the distribution of mathematic bios. Thus, booking a is of permanent importance. Does can advance on an advance the dynamic of the complex system. However, for ramply availance, complex systems the dynamics of the complex system. However, the ramply availance, complex systems advances in the distribution of the dynamics of the distribution of the dynamics and the distribution of the complex system. However, the ramply availance, complex systems advances are major persons models with an evaluated specifically for the dynamics and distribution of the distribution of the dynamic and the distribution of the dynamics and distribution of the complex models in the ary system of the responsing and the distribution of the dynamics and distribution of the distribution of the dynamics and the distribution of the dynamics and distribution of the distribution of the dynamics and the distribution of the dynamics and distribution of the distribution of the dynamics and the dynamics and the dynamics and the dynamics and measurement data, potentially allowing for significant reduction of the number of modes n required to model the behavior of (11.1) for a given accuracy [57, 542, 543].

11.2 Optimal Basis Elements: The POD Expansion

A illustrated in the previous section, the electronic of a good model have for webly (11)). A influence of the global constraints in (114) is certained from the order, however, however, a massning the Global consequences in (114) is certained as a straint of the order of t

Two options exist for extracting the optimal basis mode from a given complex system. One can diret corlect and directly from an eperiment, or one can induce the complex system and assumple the state of the system as it evolves according to the dynamics. It has the can be steme uses a simulated to extract modes, one can argue that a complexity and the can be the system is a simulated to extract modes, one can argue that comparison and the state of the system of the system is an exception of the simulation of the state of the system is an exception of the simulation of the state of the system of the state of the system is an exception of the simulation of the state of the system is a state

To proceed with the construction of the optimal POO modes, the dynamics of (11.1) are sampled at some prescribed time interval. In particular, a sampled w₁ consists of samples of the complex system, with subscript k indicating sampling at time k_1^* u_k := $[u(x_1, x_1) u(x_2, x_1) \cdots u(x_k, x_k)]^T$. Now, the continuous functions and modes will be evaluated at discress systemical locations, resulting in a high-dimensionil vector representation, these will be denoted by bold symbols. We are generally interested in marking the computational by or experimentally secreted large data set **X**:

$$\mathbf{X} = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{u}_1 & \mathbf{u}_2 & \cdots & \mathbf{u}_w \end{bmatrix}$$
(11.16)

where the columns $\mathbf{u}_k = u(r_k) \in \mathbb{C}^n$ may be measurements from simulations or experiments. **X** consists of a time-series of data, with *m* distinct measurement instances in time. Often the state-dimension *n* is very large, on the order of millions or billions in the case of fluid systems. Typically *n* $\gg m$, resulting in a *tall-skinny* matrix, as opposed to a *short-far* matrix when *n* $\ll m$.

As discussed previously, the singular value decomposition (SVD) provides a unique matrix decomposition for any complex valued matrix $X \in \mathbb{C}^{n \times m}$:

$$X = U\Sigma V^{*}$$
 (11.17)

where $U \in \mathbb{C}^{n+m}$ and $V \in \mathbb{C}^{n+m}$ are an anotymetry matters and $Z \in \mathbb{C}^{n+m}$ is a matrix with manyors. The subsequence metrix on the disparal. Here a "assume the complex compact matrix matrixs." The matrix of the subsequence of the star of the subsequence of the star of the star is the star of the star matrix of the star matrix of the star matrix of the star matrix of the star matrix of the star matrix of the star matrix of the star matrix of the star of

The total number of modes generated is typically determined by the number of snaphots: m taken is constrainty **X** (where normally $n \gg \infty$), care objective is to determine the minimal number of modes necessary to accurately represent the dynamics of (1.1), with a Galerkin projection (1.6). Thus we are interested in a rank-r approximation to the true dynamics where typically $r \ll m$. The quantity of interest is then the low-rank decomposition of the SVD given by

$$\hat{X} = \hat{U}\hat{\Sigma}\hat{V}^{*}$$
 (11.18)

where $\|\mathbf{X} - \hat{\mathbf{X}}\| < \epsilon$ for a given small value of epsilon. This low-rank truncation allows us to construct the modes of interest ψ_2 from the columns of the truncated matrix $\hat{\mathbf{U}}$. In particular, the optimal basis modes are given by

$$\tilde{\mathbf{U}} = \Psi = \begin{bmatrix} 1 & 1 & 1 \\ \psi_1 & \psi_2 & \cdots & \psi_r \\ 1 & 1 & 1 \end{bmatrix}$$
(11.19)

where the truncation preserves the r most dominant modes used in (11.6). The truncated r modes $[\psi_1, \psi_2, \cdots, \psi_r]$ are then used as the low-rank, orthogonal basis to represent the dynamics of (11.1).

The above suspition based methods for extracting the low cash, *e-dimensional unbugger* of symanic resultion associated with (111,11) and durkness comparisonal architecture. Indeed, inprovides an equation-fore method, i.e. the governing quarkins (111,11) may attuible to all solutions. In the event that the underlying plusmins case auknows, the extra solution of the low-radi, space allows case to half potential models in an *e-dimensional unbugger* as opposed to avanisating in a halp-dimensional unsue where as *y*. The low laws with the optimum distribution of the low case of the low case where a *y*. The low laws with the optimum distribution of the low case of the low case of the low case of the low case of the optimum distribution of the low case of the low case of the low case of the optimum distribution of the low case of the low case of the low case of the optimum distribution of the low case of the low case of the low case of the low case of the optimum distribution of the low case of the low case of the low case of the low case of the optimum distribution of the low case of the low case of the low case of the low case of the optimum distribution of the low case of the law case of the low case of the low case of the law case of the low case of the law case of the law

Galerkin Projection onto POD Modes

It is possible to approximate the state u of the PDE using a Galerkin expansion:

$$u(t) \approx \Psi a(t)$$
 (11.20)

where $\mathbf{a}(t) \in \mathbb{R}^{d}$ is the time-dependent coefficient vector and $r \ll n$. Plugging this modal expansion into the governing equation (11.13) and applying orthogonality (multiplying by Ψ^{2}) gives the dimensionality reduced evolution

$$\frac{d\mathbf{a}(t)}{dt} = \Psi^T \mathbf{L} \Psi \mathbf{a}(t) + \Psi^T \mathbf{N}(\Psi \mathbf{a}(t), \beta). \quad (11.21)$$

By solving this system of much smaller dimension, the solution of a high-dimensional nonlinear dynamics lystem can be approximated. Of critical importance is evaluating the nonlinear terms in an efficient way using the gappy POD or DEIM mathematical architecture in Chapter 12. Otherwise, the evaluation of the nonlinear terms sall requires calculation of functions and inter products with the original dimension n. In certain cases, such as the application continuity of Nortice-Stokes, the nonlinear terms call requires once in an off-line manner. However, parametrized systems generally require repeated contains of the nonlinear terms as the POD modes change with β .

Example: The Harmonic Oscillator

To linuxen the POD method for selecting optimal basis elements, we will consider a classic problem of methaenutical physics: the quantum harmonic flattars. Although the deal basis functions (Gauss-Hernite functions) for this problem are already known, we would like to infer these special functions from any physical drawn with the we deduce these special functions from any physical drawn and and the harmonic coefficient and the study of sphig-mass systems. In particular, one often some that the studye flow of the study of sphig-mass systems. In particular, one often assume that the studye flow of the study of sphig-mass systems. In particular, one often some that the studye flow of the study of sphig-mass systems. In particular, one often some that the studye flow of the study of sphig-mass systems. In particular, one often some that the studye flow of the study of sphig-mass systems. In a particular, one often some that the studye flow of the study of sphig-mass systems. In a particular, one often some that the studye flow of the study of sphig-mass systems. In a particular, one often some that the studye flow of the studye of sphig-mass systems. In a particular, one often some that the studye flow of the study of sphig-mass systems. In a particular, one often some that the studye flow of the studye of sphig-mass systems. In a particular, one often some that the studye is studyed as a studyed base of the st

$$F(t) = -kx$$
 (11.22)

where k is the spring constant and x(t) represents the displacement of the spring from its equilibrium position. Such a force gives rise to a potential energy for the spring of the form $V = kx^2/2$.

In considering quantum mechanical systems, such a restoring force (with k = 1 without loss of generality) and associated potential energy gives rise to the Schrödinger equation with a parabolic potential

$$iu_t + \frac{1}{2}u_{xx} - \frac{x^2}{2}u = 0$$
 (11.23)

where the second term in the partial differential equation represents the kinetic energy of a quantum particle while the last term is the parabolic potential associated with the linear restoring force.

The solution for the quantum harmonic oscillator can be easily computed in terms of special functions. In particular, by assuming a solution of the form

$$u(x, t) = a_k \psi_k(x) \exp \left[-i(k + 1/2)t\right]$$
(11.24)

with a_k determined from initial conditions, one finds the following boundary value problem for the eigenmodes of the system

$$\frac{d^2 \psi_k}{dx^2} + (2k + 1 - x^2)\psi_k \qquad (11.25)$$

with the boundary conditions $\psi_L \rightarrow 0$ as $x \rightarrow \pm \infty$. Normalized solutions to this equation can be expressed in terms of *Herwite polynomials*, $H_E(x)$ or the Gaussian-Hermite functions

$$\psi_k = \left(2^k k \sqrt{\pi}\right)^{-1/2} \exp(-x^2/2) H_k(x)$$
 (11.26a)

$$= (-1)^k \left(2^k k \sqrt{\pi}\right)^{-1/2} \exp(-x^2/2) \frac{d^4}{dx^k} \exp(-x^2).$$
 (11.26b)

The Gauss-Hermite functions are typically thought of as the optimal basis functions for the harmonic oscillator as they naturally represent the underlying dynamics driven by the Schrödinger equation with parabolic potential. Indeed, solutions of the complex system (11.23) can be represented as the sum

$$u(x, t) = \sum_{k=0}^{\infty} a_k \left(2^k k \sqrt{\pi}\right)^{-1/2} \exp(-x^2/2) H_k(x) \exp\left[-i(k+1/2)t\right].$$
 (11.27)

Such a solution strategy is ubiquitous in mathematical physics as is evidenced by the large number of special functions, often of Snurn-Liouville form, for different geometries and boundary conditions. These include Bessel functions, Laguerre polynomials, Legendre polynomials, parabolic cylinder functions, spherical harmonics, etc.

A numerical solution to the governing PDE (11.23) based on the fast Porierit random is easy to implement 10(4). The following occe excess to a flux numerical solutions with the initial conditions w(r, 0) = exp(-0.2)($-\alpha_0 \overline{0}$), which is a Gaussian public centered at $x = \alpha_0$. This initial condition generically excites a number of Gauss-Hermite functions. In particular, the initial projection onto the eigenmodes is computed from the orthogonality conditions of the solution of the s

$$a_k = (u(x, 0), \psi_k)$$
. (11.28)

This inner product projects the initial condition onto each mode ψ_k .

Cade 11.1 Harmonic oscillator code.

The right-hand side function, pod_harm_rhs.m associated with the above code contains the governing equation (11.23) in a three-line MATLAB code:

Code 11.2 Harmonic oscillator right-hand side.

```
function rhsepod_harm_rhs(t,ut,dummy,k,V)
usifft(ut);
rhse-(i/2).(k.^2)..ut - 0.5.i.fft(V..u);
```



Figure 12: Dynamics of the quantum harmonic socilitor (11.23) given the initial condition of (1, 0) = csp(-(2, -(m_2))^2 for n = 0 of left parel and n = 1 (right parel). The symmetric initial data circles a dominant for mode response which the initial condition with initial offset n = 1 advances to mode. The bettom parels show the singular values of the SVD of their corresponding top panels along with the presentage of energy (m L² newn jin each mode. The dynamics are clored boso-cash picent the read decay of the singular values.

The two codes together produce dynamics associated with the quantum harmonic particular, Eq. 11.2 above the dynamical conduct of an initial Gaussian are, $(n) = nq^{(-1)}(2,1)$, $n_{q_{-}} = 0$ (d) parely harmonic hybrid parely. Then the particle Aroma (Harmonic and Aroma) and a set of the particle Aroma (Harmonic and Aroma) and a set of the Aroma (Harmonic Aroma

The singular value decomposition not only gives the distribution of energy within the first set of modes, but it also produces the optimal basis elements as columns of the matrix. U. The distribution of singular values is highly suggestive of how to tunnear with a lowrank subspace of *r* modes, thus allowing us to construct the dimensionally reduced space (11.19) anoromica for a Galerkin-2000 extonsion.

The modes of the quantum harmonic oscillator are illustrated in Fig. 11.3. Specifically, the first five modes are shown for (i) the Gauss-Hernite functions representing the special function solutions, (ii) the modes of the SVD for the symmetric $(x_0 = 0)$ initial conditions, and (iii) the modes of the SVD for the offset (asymmetric, $x_0 = 1$) initial conditions. The Gauss-Hernite functions, by construction, are arranged from lowset citererable



Figure 13. First for modes of the quantum harmonic scattarts is the top panel, the first for disconstruction of the quantization of the grant scattart of the thermoscie scattart of the scat

of the Sum-Lassville problem (1122). The eigenmodes alternate between symmetry of the Sum-Lassville problem (1122). The eigenmode alternate 1 = 0 initial conditions pixel by $a(t, 0) = cop(-0.25^{+})$, the first free modes are all symmetries and assaylable based of the dynamics and the heap run of observable t = 0 minimized in the exclusion. In contrast, with a right officer, $a(t, 0) = cop(-0.25, -17)^{+}$, mapping of the observable modes any symmetries model in the symmetry observable of the conduct like linear symmetry observable and the symmetry observable of the conduct like linear integration of the symmetry observable of the Gaussilike linear integration of the symmetry observable of the Gaussilike linear integration of the symmetry observable of the symmetry polynomials that provide all generating or early such taken symmetry modes the fittening polynomial characteristic of the system linearized heat is conliferent expression discontensities of the system linearized heat is not heat worked.

11.3 POD and Soliton Dynamics

To illustrate a full implementation of the Galerkin-POD method, we will consider an illustrative complex system whose dynamics are strongly nonlinear. Thus, we consider the nonlinear Schrödinger (NLS) equation

$$iu_t + \frac{1}{2}u_{xx} + |u|^2 u = 0$$
 (11.29)

with the boundary conditions $u \rightarrow 0$ as $x \rightarrow \pm \infty$. If not for the nonlinear term, this equation could be solved easily in closed form. However, the nonlinearity mixes the eigenfunction components in the expansion (11.6), and it is impossible to derive a simple analytic solution.

To solve the NLS computationally, a Fourier mode expansion is used. Thus the standard fast Fourier transform may be leveraged. Rewriting (11.29) in the Fourier domain, i.e. taking the Fourier transform, gives the set of differential equations

$$\hat{u}_{t} = -\frac{i}{2}k^{2}\hat{u} + i\widehat{|u|^{2}u} \qquad (11.30)$$

where the Fourier mode mixing occurs due to the nonlinear mixing in the cubic term. This gives the system of differential equations to be solved in order to evaluate the NLS behavior.

The following code formulates the PDE solution as an eigenfunction expansion (11.6) of the NLS (11.29). The first step in the process is to define an appropriate spatial and temporal domain for the solution along with the Fourier frequencies present in the system. The following code produces both the time and space domains of interest:

Code 11.3 Nonlinear Schrödinger equation solver.

The right-hand side function, pod_sol_rhs.m associated with the above code contains the governing equation (11.29) in a three-line MATLAB code:

Code 11.4 NLS right-hand side.

```
function rhsspod_sol_rhs(t,ut,dummy,k)
usifft(ut);
rhss-(i/2)*(k,^2).*ut + i*fft( (abs(u).^2).*u );
```

It now remains to consider a specific spatial configuration for the initial condition. For the NLS, there are a set of special initial conditions called solitons where the initial conditions are given by

$$u(x, 0) = Nsech(x)$$
 (11.31)



Figure 11.4 Evolution of the $(a) N \equiv 1$ and $(b) N \equiv 2$ solitons. Here steady-state $(N \equiv 1, left panels (a) and <math>(c)$) and periodic $(N \equiv 2, right panels (b) and (d))$ dynamics are observed and approximately 50 and 200 Fourier modes, respectively, are required to model the behaviors.

where N is an integer. We will consider the soliton dynamics with N = 1 and N = 2. First, the initial condition is projected onto the Fourier modes with the fast Fourier transform.

The dynamics of the $\dot{N} = 1$ and N = 2 solitons are demonstrated in Fig. 11.4. During evolution, the N = 1 soliton only undergoes phase changes while its amplitude remains stationary. In contrast, the N = 2 soliton undergoes profied a solillations. In bot cases, a large number of Fourier modes, about 50 and 200 respectively, are required to model the simple behaviors linkarated.

The obvious question to ask in light of our dimensionality reduction thinking is this is the soliton dynamics really a 50 or 200 degrees of freedom system as required by the Fourier mode solution technics. The answer is no. Indeed, with the appropriate basis, Le He POD modes generated from the SVD1 c can be shown that the dynamics is a simple reduction to it or 2 modes respectively. Indeed, it can easily be shown that $b^{r} = 1$ and the ecolutions theory in Fig. 11.4.

Fig. 11.2 explicitly denovarizes the low-dimensional name of the numerical solutions by comparing the singular values, along with the modes to be used in our we eigenfunction expansion. For both of these cases, the dynamics are neithy low dimensional with the N = 1solution being modeled with y_0 single POD mode with the N = 2 solutions are modeled quite well with two POD modes. Thus, in performing an eigenfunction expansion, the modes choses should be the POD mode segrented from the simulation themselves, to these accessions, we will derive the dynamics of the model interaction for these two cases, which are low-dimensional and memble to analysio.



Figure 113 Projection of the N = 1 and N = 2 evolutions onto their POD modes. The top two figure (1) and (b) are the singular values or p on a longitudine scale of the two evolutions demonstrated in (11.4). This demonstrates that the N = 1 and N = 2 solution dynamics are primarily low-rank, with the N = 1 lowing a single mode evolution and N = 2 being unimated by two modes that contain approximately 95% of the evolution variance. The first three modes in both cases are shown in the bottom two mades (10 and (6).

Soliton Reduction (N - 1)

To take advantage of the low dimensional structure, we first consider the N = 1 soliton dynamics. Fig. 11.5 shows that a single mode in the SVD dominates the dynamics. This is the first column of the U matrix. Thus the dynamics are recast in a single mode so that

$$u(x, t) = a(t)\psi(x).$$
 (11.32)

Plugging this into the NLS equation (11.29) yields the following:

$$ia_t\psi + \frac{1}{2}a\psi_{xx} + |a|^2a|\psi|^2\psi = 0.$$
 (11.33)

The inner product is now taken with respect to ψ which gives

$$ia_t + \frac{\alpha}{2}a + \beta |a|^2 a = 0$$
 (11.34)

where

$$\alpha = \frac{\langle \psi_{xx}, \psi \rangle}{\langle \psi, \psi \rangle}$$
(11.35a)

$$\beta = \frac{\langle |\psi|^2 \psi, \psi \rangle}{\langle \psi, \psi \rangle}. \quad (11.35b)$$

This is the low-rank approximation achieved by the POD-Galerkin method.

The differential equation (11.34) for a(t) can be solved explicitly to yield

$$a(t) = a(0) \exp \left(i \frac{\alpha}{2} t + \beta |a(0)|^2 t\right)$$
 (11.36)

where a(0) is the initial condition for a(t). To find the initial condition, recall that

$$u(x, 0) = \operatorname{sech}(x) = a(0)\psi(x)$$
. (11.37)

Taking the inner product with respect to $\psi(x)$ gives

$$a(0) = \frac{(\text{sech}(x), \psi)}{(\psi, \psi)}$$
. (11.38)

Thus the one mode expansion gives the approximate PDE solution

$$u(x, t) = a(0) \exp \left(i \frac{\alpha}{2}t + \beta |a(0)|^2 t\right) \psi(x).$$
 (11.39)

This solution is the low-dimensional POD approximation of the PDE expanded in the best basis possible, i.e. the SVD basis.

For the N = 1 soliton, the spatial profile remains constant while its phase undergoes a nonlinear rotation. The POD solution (11.39) can be solved exactly to characterize this phase rotation.

Soliton Reduction (N = 2)

The N = 2 soliton case is a bit more complicated and interesting. In this case, two modes clearly dominate the behavior of the system, as they contain 96% of the energy. These two modes, ψ_1 and ψ_2 , are the first two columns of the matrix U and are now used to approximate the dynamics observed in Fig. (11.4). In this case, the two mode expansion takes the form

$$u(x, t) = a_1(t)\psi_1(x) + a_2(t)\psi_2(x).$$
 (11.40)

Inserting this approximation into the governing equation (11.29) gives

$$i(a_{1_{1}}\psi_{1} + a_{2_{1}}\psi_{2}) + \frac{1}{2}(a_{1}\psi_{1_{XX}} + a_{2}\psi_{2_{XX}}) + (a_{1}\psi_{1} + a_{2}\psi_{2})^{2}(a_{1}^{*}\psi_{1}^{*} + a_{2}^{*}\psi_{2}^{*}) = 0.$$
 (11.41)

Multiplying out the cubic term gives

$$i (a_{11}\psi_1 + a_{21}\psi_2) + \frac{1}{2} (a_1\psi_{1x} + a_2\psi_{2x})$$

+ $(|a_1|^2a_1|\psi_1|^2\psi_1 + |a_2|^2a_2|\psi_2|^2\psi_2 + 2|a_1|^2a_2|\psi_1|^2\psi_2 + 2|a_2|^2a_1|\psi_2|^2\psi_1$
+ $a_1^2a_2^*\psi_1^2\psi_1^2 + a_2^2a_1^*\psi_2^2\psi_1^*)$. (11.42)

All that remains is to take the inner product of this equation with respect to both $\psi_1(x)$ and $\psi_2(x)$. Recall that these two modes are orthogonal, resulting in the following 2×2 system of nonlinear equations:

$$ia_{1t} + a_{11}a_1 + a_{12}a_2 + (\beta_{111}|a_1|^2 + 2\beta_{211}|a_2|^2)a_1$$
 (11.43a)

$$\begin{split} &+ \left(\beta_{121}|a_1|^2 + 2\beta_{221}|a_2|^2\right)a_2 + \sigma_{212}a_1^2a_2^2 + \sigma_{211}a_2^2a_1^* = 0\\ &ia_{22} + a_{21}a_1 + a_{22}a_2 + \left(\beta_{121}|a_1|^2 + 2\beta_{212}|a_2|^2\right)a_1 \\ &+ \left(\beta_{222}|a_1|^2 + 2\beta_{222}|a_2|^2\right)a_2 + \sigma_{212}a_1^2a_2^2 + \sigma_{212}a_2^2a_1^* = 0 \end{split}$$
(11.43b)

where

$$x_{jk} = (\psi_{j_{11}}, \psi_k)/2$$
 (11.44a)

$$\beta_{jkl} = \langle |\psi_j|^2 \psi_k, \psi_l \rangle$$
 (11.44b)

$$\eta_{kl} = \langle \psi_l^2 \psi_k^*, \psi_l \rangle$$
 (11.44c)

and the initial values of the two components are given by

$$a_1(0) = \frac{(2 \operatorname{sech}(x), \psi_1)}{(\psi_1, \psi_2)}$$
(11.45a)

$$a_2(0) = \frac{(2 \operatorname{sech}(x), \psi_2)}{(\psi_2, \psi_2)}$$
. (11.45b)

This gives a complete description of the two mode dynamics predicted from the SVD analysis.

The two mode dynamics accurately approximates the solution. However, there is a phase drift than occurs in the dynamics that words require higher periods in to the the integers of of the HTPE and more accurate integration of the integer products for the coefficients. Indeed, the most simple trapeoidal rela has been used to compute the integration of the accuracy is somewhat suspect, this issue will be addressed in the following section. Higher order schemes could cathol help. In other case, this demonstrates how one would use the low dimensional structures to approximate PDE dynamics in practice.

11.4 Continuous Formulation of POD

Thus far, the POD reduction has been constructed to accommodate discrete data measuremere snapshots X as given by (11.16). The POD reduction generates a set of low-rank basis modes Ψ so that the following least-squares error is minimized:

$$\underset{\Psi s.t. rask(\Psi)=r}{\operatorname{argmin}} \|\mathbf{X} - \Psi \Psi^T \mathbf{X}\|_F. \quad (11.46)$$

Recall that $X \in \mathbb{C}^{n \times m}$ and $\Psi \in \mathbb{C}^{n \times r}$ where r is the rank of the truncation.

In many cases, measurements are performed on a continuous time process over a prescribed spatial domain, thus the data we consider are constructed from trajectories

$$u(x, t) \quad t \in [0, T], x \in [-L, L].$$
 (11.47)

Such data require a continuous time formulation of the POD reduction. In particular, an equivalent of (11.6) must be constructed for these continuous time trajecories. Note that instead of a spatially dependent function w(x, t), one can also consider a vector of trajectories $w(t) \in \mathbb{C}^n$. This may rise when a PDE is discretized so that the infinite dimensional spatial variable x is finite dimensional. Wolkwein [542, 543] gives an excellent, technical overview of the PD0 method and in continuous formulation. To define the continuous formulation, we prescribe the inner product

$$(f(x), g(x)) = \int_{-L}^{L} f(x)g^{*}(x)dx.$$
 (11.48)

To find the best fit function through the entire temporal trajectory u(x, t) in (11.47), the following minimization problem must be solved

$$\min_{\psi} \frac{1}{T} \int_{0}^{T} ||u(x, t) - \langle u(x, t), \psi(x) \rangle \psi ||^{2} dt \text{ subject to } ||\psi||^{2} = 1 \quad (11.49)$$

where the normalization of the temporal integral by 1/T averages the difference between the data and its lows-rank approximation using the function ψ over the time $t \in [0, T]$. Equation $(1, 4\phi)$ is equivalent to maximizing the inner product between the data u(x, t) and the function $\psi(x)$, i.e. they are maximally parallel in function space. Thus the minimization problem can be restarted as

$$\max_{\psi} \frac{1}{T} \int_{0}^{T} |\langle u(x, t), \psi(x) \rangle|^{2} dt \text{ subject to } \|\psi\|^{2} = 1. \quad (11.50)$$

The constrained optimization problem in (11.50) can be reformulated as a Lagrangian functional

$$\mathcal{L}(\psi, \lambda) = \frac{1}{T} \int_{0}^{T} |\langle u(x, t), \psi(x) \rangle|^{2} dt + \lambda (1 - ||\psi||^{2})$$
 (11.51)

where λ is the Lagrange multiplier that enforces the constraint $\|\psi\|^2 = 1$. This can be rewritten as

$$\mathcal{L}(\psi, \lambda) = \frac{1}{T} \int_0^T \left(\int_{-L}^L u(\xi, t) \psi^*(\xi) d\xi \int_{-L}^L u^*(x, t) \psi(x) dx \right) dt$$

+ $\lambda \left(1 - \|\psi\|^2 \right) + \lambda \left(1 - \int_{-L}^L \psi(x) \psi^*(x) dx \right).$ (11.52)

The Lagrange multiplier problem requires that the functional derivative be zero:

$$\frac{\partial \mathcal{L}}{\partial \psi^{+}} = 0.$$
 (11.53)

Applying this derivative constraint to (11.52) and interchanging integrals yields

$$\frac{\partial \mathcal{L}}{\partial \psi^*} = \int_{-L}^{L} d\xi \left[\frac{1}{T} \int_{0}^{T} \left(u(\xi, t) \int_{-L}^{L} u^*(x, t) \psi(x) dx \right) dt - \lambda \psi(x) \right] = 0. \quad (11.54)$$

Setting the integrand to zero, the following eigenvalue problem is derived

$$\langle R(\xi, x), \psi \rangle = \lambda \psi$$
 (11.55)

where $R(\xi, x)$ is a two-point correlation tensor of the continuous data u(x, t) which is averaged over the time interval where the data is sampled

$$R(\xi, x) = \frac{1}{T} \int_{0}^{T} u(\xi, t)u^{*}(x, t)dt.$$
 (11.56)



Figure 11.8 Illustration of an implementation of the quadrature rule to evaluate the integrals $\int_{0}^{T} f(t) dt$. The rectangles of height $f(t_{f}) = f_{f}$ and width δt are summed to approximate the integral.

If the spatial direction x is discretized, resulting in a high-dimensional vector $\mathbf{u}(t) = \begin{bmatrix} u(x_1, t) & u(x_2, t) & \cdots & u(x_n, t) \end{bmatrix}^T$, then $R(\tilde{\xi}, x)$ becomes:

$$\mathbf{R} = \frac{1}{T} \int_{0}^{T} \mathbf{u}(t) \mathbf{u}^{*}(t) dt. \qquad (11.57)$$

In practice, the function R is evaluated using a quadrature rule for integration. This will allow us to connect the method to the snapshot based method discussed thus far.

Quadrature Rules for R: Trapezoidal Rule

The evaluation of the integral (11.57) can be performed by numerical quadrature [156]. The simplest quadrature rule is the trapezoidal rule which evaluates the integral via summation of approximation for examples. Fig. 11.6 illustrates a version of the trapezoidal rule where the integral is approximated by a summation over a number of rectangles. This gives the approximation of the two-point correlation tensor:

$$\mathbf{R} = \frac{1}{T} \int_{0}^{T} \mathbf{u}(t)\mathbf{u}^{*}(t)dt$$

$$\approx \frac{\Delta t}{T} \left[\mathbf{u}^{*}(t_{1})\mathbf{u}(t_{1}) + \mathbf{u}^{*}(t_{2})\mathbf{u}(t_{2}) + \dots + \mathbf{u}^{*}(t_{m})\mathbf{u}(t_{m})\right] \qquad (11.58)$$

$$= \frac{\Delta t}{T} \left[\mathbf{u}^{*}\mathbf{u}_{1} + \mathbf{u}^{*}_{2}\mathbf{u}_{2} + \dots + \mathbf{u}^{*}_{m}\mathbf{u}_{m}\right]$$

where we have assumed u(x, t) is discretized into a vector $\mathbf{u}_j = \mathbf{u}(t_j)$, and there are m rectangular bins of width Δt so that $(m)\Delta t = T$. Defining a data matrix

$$X = [u_1 \ u_2 \ \cdots \ u_m]$$
 (11.59)

we can then rewrite the two-point correlation tensor as

$$\mathbf{R} \approx \frac{1}{m} \mathbf{X}^* \mathbf{X}$$
 (11.60)

which is exactly the definition of the covariance matrix in (1.27), i.e. $C \approx R$. Note that the role of 1/T is to average over the various trajectories so that the average is subtracted out, giving rise to a definition consistent with the covariance.

Higher-order Quadrature Rules

Numerical integration simply calculates the area under a given curve. The basic ideas for performing such an operation come from the definition of integration

$$\int_{a}^{b} f(t)dt = \lim_{\Delta t \to 0} \sum_{j=0}^{m-1} f(t_j)\Delta t \qquad (11.61)$$

where $b - a = (m - 1)\Delta t$. The area under the curve is a limiting process of summing up an ever-increasing number of rectangles. This process is known as numerical quadrature. Specifically, any sum can be represented as follows:

$$Q[f] = \sum_{j=0}^{m-1} w_j f(t_j) = w_0 f(t_0) + w_1 f(t_1) + \dots + w_{m-1} f(t_{m-1})$$
 (11.62)

where $a = t_0 < t_1 < t_2 < \cdots < t_{m-1} = b$. Thus the integral is evaluated as

$$\int_{a}^{b} f(t)dt = Q[f] + E[f] \qquad (11.63)$$

where the term E[f] is the error in approximating the integral by the quadrature sum (11.62). Typically, the error E[f] is due to truncation error. To integrate, we will use polynomial fits to the y-values $f(t_j)$. Thus we assume the function f(t) can be approximated by a polynomial

$$P_n(t) = a_n t^n + a_{n-1} t^{n-1} + \dots + a_1 t + a_0$$
 (11.64)

where the truncation error in this case is proportional to the $(u + 1)^{46}$ derivative $E[ff] = Af^{(u+1)}(c)$ and A is a constant. This process of polynomial fitting the data gives the Newton-Cores Forwardsz.

The following integration approximations result from using a polynomial fit through the data to be integrated. It is assumed that

$$t_k = t_0 + \Delta t k$$
 $f_k = f(t_k)$. (11.65)

This gives the following integration algorithms:

Trapezoid Rule
$$\int_{t_0}^{t_1} f(t)dt = \frac{\Delta t}{2}(f_0 + f_1) - \frac{\Delta t^3}{12}f''(c)$$
 (11.66a)

Simpson's Rule
$$\int_{t_0}^{t_2} f(t)dt = \frac{\Delta t}{3}(f_0 + 4f_1 + f_2) - \frac{\Delta t^5}{90}f^{sv}(c)$$
 (11.66b)

Simpson's 3/8 Rule
$$\int_{t_0}^{t_1} (t) dt = \frac{3\Delta t}{8} (f_0 + 3f_1 + 3f_2 + f_3) - \frac{3\Delta t^3}{80} f^{'''}(c)$$
 (11.66c)

Boole's Rule
$$\int_{t_0}^{t_1} (t) dt = \frac{2\Delta t}{45} (7f_0 + 32f_1 + 12f_2 + 32f_3 + 7f_4) - \frac{8\Delta t^2}{945} f^{(6)}(c).$$
 (11.66d)

These algorithms have varying degrees of accuracy. Specifically, they are $O(\Delta r^3)$, $O(\Delta r^4)$, $O(\Delta r^4)$ and $O(\Delta r^6)$ accurate schemes respectively. The accuracy condition is determined from the transaction terms of the polynomial fit. Note that the trapezoidal rule uses a sum of simple trapezoids to approximate the integral. Suppose r and r is a quadratic curve through three points and calculates the area under the quadratic curve. Suppose r 3σ rate uses for 3σ rate for the transaction terms of the polynomial fit. Note that the trapezoids of the transaction terms of the polynomial fit. Note that the trapezoids of r 3σ rate uses for 3σ rate r is the source of the transaction terms of the polynomial fit. Note that the trapezoids of the transaction terms of the polynomial fit. Note that the trapezoids of the transaction terms of the polynomial fit. Note that the trapezoids of the terms of the polynomial fit. Note that the trapezoids of the terms of the polynomial fit. Note that the trapezoids of the terms of the terms of the terms of the polynomial fit. Note that the terms of the terms of the polynomial fit. Note that the terms of terms of the terms of points and a cubic polynomial to evaluate the area, while Boole's rule uses five points and a quartic polynomial fit to generate an evaluation of the integral.

The integration methods (11.66) give values for the integrats over only a small part of the integration domain. The trapezoidal rule, for instance, only gives a value for $t \in \{0, n\}$. Howevere, our fundamental aim is to evaluate the integral over the entire domain $t \in \{a, b\}$. Assuming once again that our interval is divided as $a = t_0 < t_1 < t_2 < \cdots < t_{n-1} = b$, then the trapezoidal rule applied over the interval gives the total integral.

$$\int_{a}^{b} f(t)dt \approx Q[f] = \sum_{j=1}^{m} \frac{\Delta t}{2} (f_{j} + f_{j+1}). \quad (11.67)$$

Writing out this sum gives

$$\sum_{j=1}^{m} \frac{\Delta t}{2} (f_j + f_{j+1}) = \frac{\Delta t}{2} (f_0 + f_1) + \frac{\Delta t}{2} (f_1 + f_2) + \dots + \frac{\Delta t}{2} (f_m + f_{m-1})$$

$$= \frac{\Delta t}{2} (f_0 + 2f_1 + 2f_1 + \dots + 2f_m + f_{m-1}) \quad (11.68)$$

$$= \frac{\Delta t}{2} \left(f_0 + f_{m-1} + 2\sum_{j=1}^{m} f_j \right).$$

The final expression no longer double counts the values of the points between f_0 and f_{m-1} . Instead, the final sum only counts the intermediate values once, thus making the algorithm about twice as fast as the previous sum expression. These are computational savings which should always be exploited if possible.

POD Modes from Quadrature Rules

Any of these algorithms could be used to approximate the two-point correlation tensor $\mathbf{R}(\tilde{g}, x)$. The method of snapshots implicitly uses the trapezoidal rule to produce the snapshot matrix \mathbf{X} . Specifically, recall that

$$\mathbf{X} = \begin{bmatrix} \mathbf{I} & \mathbf{I} & \mathbf{I} \\ \mathbf{u}_1 & \mathbf{u}_2 & \cdots & \mathbf{u}_n \\ \mathbf{I} & \mathbf{I} & \mathbf{I} \end{bmatrix}$$
(11.69)

where the columns $u_k \in \mathbb{C}^n$ may be measurements from simulations or experiments. The SVD of this matrix produces the modes used to produce a low-rank embedding Ψ of the data.

One could alternatively use a higher-order quadrature rule to produce a low-rank decomposition. Thus the matrix (11.69) would be modified to

$$\mathbf{X} = \begin{bmatrix} | & | & | & | & | & | & | & | \\ \mathbf{u}_1 & 4\mathbf{u}_2 & 2\mathbf{u}_3 & 4\mathbf{u}_4 & 2\mathbf{u}_5 & \cdots & 4\mathbf{u}_{m-1} & \mathbf{u}_m \\ | & | & | & | & | & | & | & | \end{bmatrix}$$
(11.70)

where the Simpson's rule quadrature formula is used. Simpson's rule is commonly used in practice as it is simple to execute and provides significant improvement in accuracy over the trapezoidal rule. Producing this matrix simply involves: multiplying the data matrix on the right by $\begin{bmatrix} 1 & 4 & 2 & 4 & \cdots & 2 & 4 & 1 \end{bmatrix}^T$. The SVD can then be used to construct a lowrank embedding Ψ . Before approximating the low-rank solution, the quadrature weighting matrix must be undown. To our knowledge, very little work has been done in quantifying the merits of various quadrature rules. However, the interested reader should consider the optimal samphots sampling strategy developed by Knnisch and Volkwein [35].

11.5 POD with Symmetries: Rotations and Translations

The POD method is not without its shortcomings, It is well known in the POD community that the underlying JOD algorithm does Mandie invariances in the data is an anginand way. The most common invariance arises from translational or rotational invariances in the data. Translational invariance is observed in the simple phenomenon of wave propagation, making it difficult for correlation to be computed since critical features in the data are no longer aligned support to support.

In what follows, we will consider the effects of both translation and rotation. The examples are motivated from physical problems of practical interest. The important observation is that unless the invariance structure is accounted for, the POD reduction will give an artificially inflated dimension for the underlying dynamics. This challenges our ability to use the POD as a diagnostic tool or as the platform for reduced order models.

Translation: Wave Propagation

To illustrate the impact of translation on a POD analysis, consider a simple translating Gaussian propagating with velocity c.

$$u(x, t) = \exp \left[-(x - ct + 15)^2\right].$$
 (11.71)

We consider this solution on the space and time intervals $x \in [-20, 20]$ and $t \in [0, 10]$. The following code produces the representative translating solution and its low-rank representation.

Code 11.5 Translating wave for POD analysis.

```
nr200; L200; selimspace(-L,L,n); yxx; } space
medi; T400; tilinspace(0,T,m); b time
c12; b vave speed
Xe[];
for jel;n
for jel;n
for jel;n;
```

Figure 11.7(a) demonstrates the simple evolution to be considered. As is clear from the figure, the translation of the pulse will clearly affect the correlation as a given spatial location. Naive application of the SVD does not account for the translating nature of the data. As a result, the singular values produced by the SVD dccay slowly as shown in Fig. 11.7(b) and (c). In fact, the first few modes each contain approximately 8% of the variance.

The slow decay of singular values suggests that a low-rank embedding is not easily constructed. Moreover, there are interesting issues interpreting the POD modes and their



Figure 11.7 (a) Translating Gaussian with speed c = 3. The singular value decomposition produces a slow decay of the singular values which is shown on a (b) normal and (c) logarithmic plot.



Figure 11.8 First four spatial modes (a) (first four columns of the U matrix) and temporal modes (b) (first four columns of the V matrix). A wave translating at a constant speed produces Fourier mode structures in both snace and time.

time dynamics. Fig. 11.8 shows the first four spatial (U) and temporal (V) modes generated by the SVD. The spatial modes are global in that they span the entire region where the puble propagation occurred. Interestingly, they appear to be Fourier modes over the region where the puble propagated. The temporal modes illustrate a similar Fourier mode basis for this specific example of a translating wave propagating at a constant velocity.

The failure of POD in this case is due simply to the translational invariance. If the invariance is removed, or factored out [457], before a data reduction is attempted, then the POD method can once again be used to produce a low-rank approximation. In order



Figure 11.9 Spiral waves (a) u(x, y), (b) |u(x, y)| and (c) $u(x, y)^5$ on the domain $x \in [-20, 20]$ and $y \in [-20, 20]$. The spirals are made to spin clockwise with angular velocity u.

to remove the invariance, the invariance must first be identified and an auxiliary variable defined. Thus we consider the dynamics rewritten as

$$u(x, t) \rightarrow u(x - c(t))$$
 (11.72)

where (cr) corresponds to the translational invariance in the system responsible for limiting the POD method. The parameter c can be found by a number of methods. Rowley and Manden (4-27) propose a template based technique for factoring out the invariance. Alternatively, a simple center-of-mass calculation can be used to compute the location of the wave and the variable (cr) [316].

Rotation: Spiral Waves

A second invariance commonly observed in simulations and data is associated with retation. Much like translation, rotation moves a coherent, low-rank structure in such a way that correlations, which are produced at specific spatial locations, are no longer produced. To illustrate the effects of rotational invariance, a localized spiral wave with rotation will be considered.

A spiral wave centered at the origin can be defined as follows

$$u(x, y) = \tanh \left[\sqrt{x^2 + y^2} \cos \left(A_{d}(x + iy) - \sqrt{x^2 + y^2} \right) \right]$$
 (11.73)

where A is the number of arms of the spiral, and the \downarrow denotes the phase angle of the quantity (x + iy). To localize the spiral on a spatial domain, it is multiplied by a Gaussian centered at the origin so that our function of interest is given by

$$f(x, y) = u(x, y) \exp \left[-0.01(x^2 + y^2)\right].$$
 (11.74)

This function can be produced with the following code.

Code 11.6 Spiral wave for POD analysis.

```
[act00]

[bcc] xallangaac(-L,L,R); ywa;

[bcc] xallangaac(-L,L,R); ywa;

[bcc] xallangaac(-L,L,R); ywa;

[bcc] xallangaac(-L,T,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T,T);

[bcc] xallangaac(-L,T);

[bcc] xalla
```



Figure 11.10 (a) First four temporal modes of the matrix V. To numerical precision, all the variance is in the first two modes as shown by the singular value decay on a neemal (b) and logarithmic (c) plot. Remarkably, the POD extracts exactly two modes (See Fig. 11.11) to represent the rotating spiral wave.

pcolor(x, y, uf), shading interp, colormap(hot) and

Note that the code produces snapshots which advance the phase of the spiral wave by //10 each pass through the for loop. This creates the *notation* structure we wish to consider. The rate of spin can be made faster or slower by lowering or raising the value of the denominator respectively.

In addition to considering the function u(x, y), we will also consider the closely related functions |u(x, y)| and $u(x, y)^3$ as shown in Fig. 11.9. Although these three functions clearly have the same underlying function that rotates, the change in functional form is shown to produce quite different low-rank approximations for the rotating waves.

To begin our analysis, consider the function u(x, y) illustrated in Fig. 11.9(a). The SVD of this matrix can be computed and its low-rank structure evaluated using the following code.

Code 11.7 SVD decomposition of spiral wave.

```
[[1,5,v] ared(Xd,0);
figure(2)
aubplot(4,3)
plot(100,diag(5)/sum(diag(5)),'ko','Linewidth',[2])
sumblot(4,1,4)
sumblot(2,1,4)
sumblot(2,1,4);
sumblot(2,1,4);
```



Figure 11.11 First four POD modes associated with the rotating spiral wave $\alpha(x, y)$. The first two modes capture all the variance to numerical precision while the third and fourth mode are noisy due to numerical round-off. The domain considered is $x \in [-20, 20]$ and $x \in [-20, 20]$.

Two figures are produced. The first assesses the ratix of the observed dynamics and the temporal behavior of the first from rules (so V. Figs. 11.10) of us of (s) show the decoy of singular values on a regular and logarithmic scale respectively. Remarkably, the first two modes capture affle twitzmics of the data to numerical previous. This is further illustrated in the time dynamics of the first four modes. Specifically, the first two modes of the eleven or discussion of the original strain the second strain the size of the second strain Fig. 11.00a [here a class collicity signature associated with the rotation of modes or and two of Fig. 1.11.1. Modes there and four resemble noise in both time and space as a result of unserial cont and eff.

The spin we (1) 13.4 allows for a row mode transition that is scenario to manufacture of the scenario form of the



Figure 11.12 Decay of the singular values on a normal (a) and logarithmic (b) scale showing that the function |u(x, t)| produces a slow decay while $u(x, t)^2$ produces as r = 6 approximation to numerical accuracy. The first four temporal modes of the matrix V are shown for these two functions in (c) and (d) respectively.



Figure 11.13. First four POD modes associated with the rotating spinal wave |u(x, y)| (top row) and $u(x, t)^2$ (bottom row). Unlike our previous example, the first four modes do not capture all the variance to numerical precision, thus requiring more modes for accurate approximation. The domain considered is $x \in [-20, 20]$ and $y \in [-20, 20]$.

After all, the only difference between the three rotating solutions is the actual shape of the rotating function as they are all rotating with the same speed.

To conclude, invariances can severely limit the POD method, Most notably, it can artificially inflate the dimension of the system and lead to compromised interpretability: Expert Knowledge of a given system and its potential invariances can help frame mathematical strategies to remove the invariances, i.e. re-aligning the data [154, 457]. But this strategy also has limitations, especially if two or more invariant structures are present. For instance, if two waves of different speeds are observed in the data, then the methods proposed for removing invariances will fail to capture both wave speeds simultaneously. Ultimately, dealing with invariances remains an open research question.

Suggested Reading

Texts

- Certified reduced basis methods for parametrized partial differential equations, by J. Hesthaven, G. Rozza and B. Stamm, 2015 [244].
- (2) Reduced basis methods for partial differential equations: An introduction, by A. Quarteroni, A. Manzoni and N. Federico, 2015 [442].
- (3) Model reduction and approximation: Theory and algorithms, by P. Benner, A. Cohen, M. Ohlberger and K. Willcox. 2017 1541.
- (4) Turbulence, coherent structures, dynamical systems and symmetry, by P. Holmes, J. L. Lumley, G. Berkooz and C. W. Rowley, 2012 [251].

Papers and Reviews

- A survey of model reduction methods for parametric systems, by P. Benner, S. Gugercin and K. Wilkox, SIAM Review, 2015 [53].
- (2) Model reduction using proper orthogonal decomposition, by S. Volkwein, Lecture Notes, Institute of Mathematics and Scientific Computing, University of Graz, 2011 [542].
- (3) The proper orthogonal decomposition in the analysis of turbulent flows, by G. Berkooz, P. Holmes and J. L. Lumley, Annual Review of Fluid Mechanics, 1993 [57].

In the last charge the mathematical furneesses, of RDMs uses onlined. Specifically, Days the PL has already highlighted Br PD models of projecting TRE dynamics to loss and softwares. The comparison of project provide structures and the sensitivstructures of the comparison of project provide structures and the sensitivty of the sensitivity of project provide structures and the sensitivity of the sensitivity of project provide structures and the sensitivity of the sensitivity of project provide structures and the sensitivity of the sensitivity and an outprovide structures and the comparison of the sensitivity of the sensitivity and comparison sampling and exhaning the comparison of the sensitivity of the sensitivity and comparison sampling digitations of Capitra 3 balance at an allow of a subject and comparison complexity of RMS and a loss distancial contractions of mathy respectively. The sensitivity of the sensitity of the sen

12.1 Gappy POD

The success of nonlinear model order reduction is largely dependent upon two key innovations: (i) the well-known POD-Galerkin method [251, 57, 542, 543], which is used to project the high-dimensional nonlinear dynamics onto a low-dimensional subspace in a principled way, and (ii) sparse sampling of the state space for interpolating the nonlinear terms required for the subspace projection. Thus sparsity is already established as a critically enabling mathematical framework for model reduction through methods such as gappy POD and its variants [179, 555, 565, 120, 159]. Indeed, efficiently managing the computation of the nonlinearity was recognized early on in the ROMs community, and a variety of techniques were proposed to accomplish this task. Perhaps the first innovation in snarse sampling with POD modes was the technique proposed by Everson and Sirmich for which the gappy POD moniker was derived [179]. In their sparse sampling scheme, random measurements were used to approximate inner products. Principled selection of the interpolation points, through the gapoy POD infrastructure [179, 555, 565, 120, 159] or missing point (best points) estimation (MPE) [400, 21], were quickly incorporated into ROMs to improve performance. More recently, the empirical interpolation method (EIM) [41] and its most successful variant, the POD-tailored discrete empirical interpolation method (DEIM) [127], have provided a greedy algorithm that allows for nearly optimal reconstructions of nonlinear terms of the original high-dimensional system. The DEIM approach combines projection with interpolation. Specifically, DEIM uses selected interpolation indices to specify an interpolation-based projection for a nearly optimal ℓ_2 subspace approximating the nonlinearity.

The low-rank approximation provided by PDD allows for a reconstruction of the solution u(x, t) in (12.9) with r measurements of the *n*-dimensional state. This viewpoint has profound consequences on how we might consider measuring our dynamical system (179). In particular, only $r \ll n$ measurements are required for reconstruction, allowing us to define the sparse representation variable $\tilde{u} \in \mathbb{C}^{2}$

$$\tilde{u} = Pu$$
 (12.1)

where the measurement matrix $\mathbf{P} \in \mathbb{R}^{r \times n}$ specifies r measurement locations of the full state $\mathbf{u} \in \mathbb{C}^n$. As an example, the measurement matrix might take the form

	Г	1	0							0	1
	Į.	0		0	1	0				0	
$\mathbf{P} =$		0					0	1	0	0	(12.2)
			0				0	0	1		
	L	0					0	0	0	1	J

where measurement locations take on the value of unity and the matrix elements are zero elsewhere. The matrix \mathbf{P} defines a projection onto an *r*-dimensional space $\hat{\mathbf{u}}$ that can be used to approximate solutions of a PDE.

The insight and observation of (12.1) forms the basis of the gappy POD method introduced by Everson and Strovich [179]. In particular, one can use a small number of measurements, or gappy data, to reconstruct the full state of the system. In doing so, we can overcome the complexity of evaluating higher-order nonlinear terms in the POD reduction.

Sparse Measurements and Reconstruction

The measurement matrix P allows for an approximation of the state vector u from r measurements. The approximation is obtained by using (12.1) with the standard POD projection:

$$\tilde{\mathbf{u}} \approx \mathbf{P} \sum_{k=1}^{I} \tilde{a}_k \psi_k$$
(12.3)

where the coefficients \hat{a}_{ii} minimize the error in approximation: $[\hat{a}_{ii} - Pu]_{ii}$. The challenge more is how to determ for \hat{a}_{ii}_{ii} given that thing inter products of (127), can no longer be performed. Specifically, the vector \hat{a}_{ii} has dimension v whereas the POD modes have dimension n_i . Let miser product regregive information from the fall range of v, and superative order specifically on the vector \hat{a}_{ii} has a single specific dimensioned on the fall range of v. The spectrum of the order point of the spectrum of the order point of the spectrum of the order of the order of the order of the order with the dotted as i_{jij} . More precisely, orthogonality must be considered on the fall range vecus the support spece. Thus the following two estimations hold

$$M_{kj} = \langle \psi_k, \psi_j \rangle = \delta_{kj}$$
 (12.4a)

$$M_{ki} = (\psi_k, \psi_j)_{ini} \neq 0 \text{ for all } k, j$$
 (12.4b)

where M_{ij} are the entries of the Hermitian matrix **M** and δ_{ij} is the Kroenecker delta function. The fact that the POD modes are not orthogonal on the support $s[\hat{u}]$ leads us to consider informatives for evaluating the vectore \hat{u} .

To determine the \tilde{a}_k , a least-squares algorithm can be used to minimize the error

$$E = \int_{t[\mathbf{k}]} \left[\hat{\mathbf{u}} - \sum_{k=1}^{r} \tilde{a}_k \psi_k \right]^2 d\mathbf{x} \qquad (12.5)$$

where the inner product is evaluated on the support $s(\hat{u})$, thus making the two terms in the integral of the same size r. The minimizing solution to (12.5) requires the residual to be orthoronal to each mode Ψ , so that

$$\left\langle \bar{\mathbf{u}} - \sum_{k=1}^{r} \tilde{a}_{k} \psi_{k}, \psi_{j} \right\rangle_{s[\mathbf{k}]} = 0 \quad j \neq k, j = 1, 2, \cdots, r.$$
 (12.6)

In practice, we can project the full state vector u onto the support space and determine the vector a:

$$M\ddot{a} = f$$
 (12.7)

where the elements of M are given by (12.4b) and the components of the vector f are given by

$$f_k = (\mathbf{u}, \psi_k)_{(01)}$$
. (12.8)

Note that if the measurement space is sufficiently dense, or if the support space is the entire space, then $M = L_1$ implying the eigenvalues of M approach unity as the number of measurements become dense. Once the vector \tilde{a} is determined, a reconstruction of the solution can be performed as

$$u(x, t) \approx \Psi \hat{a}$$
. (12.9)

As the measurements become dense, not only does the matrix **M** converge to the idenity, but $\hat{a} \rightarrow a$. Interestingly, these observations lead us to consider the efficacy of the method and/or annovationation by considering the condition number of the matrix **M** 15241:

$$\kappa(\mathbf{M}) = \|\mathbf{M}\| \|\mathbf{M}^{-1}\| = \frac{\sigma_1}{\sigma_m}.$$
 (12.10)

Here the 2-norm has been used, if $\epsilon(M)$ is small then the matrix is said to be wellconditioned. A minima value of $\epsilon(M)$ is achieved with the identify matrix M = 1. Thus, as the sampling space become dense, the condition number also approaches unity. This can be used as a neutric of determining how with the space sampling is performing. Large condition numbers suggest poor reconstruction while values tending toward unity should perform well.

Harmonic Oscillator Modes

To demonstrate the gappy sampling method and its reconstruction efficacy, we apply the technique to the Gauss-Hermite functions defined by (11.25) and (11.26). In the code that follows, we compute the first ten modes as given by (11.26). To compute the second devivative, we use the fact that the Fourier transform F can produce a spectrally accurate approximation, i.e. $u_m = F^{-1}[(m)^2/a_2]$. For the task of producing accurate derivatives, we consider the domains $x \in [-1, 0, 10]$ but then work with the number domain of interests $x \in [-4, 4]$. Recall further that the Fourier transform assumes a 2*x*-periodic domain. This is handled by a scaling factor in the *k*-vurvectors. The first first modes have been demonstrated in Fig. 11.3. In the code that follows, we view the first 10 modes with a proview code pto bin order highlight the vurvous features of the modes.

Code 12.1 Harmonic oscillator modes.

```
 \begin{array}{l} \label{eq:sets} \left[ \begin{array}{c} \operatorname{sets} \operatorname{sets
```

The mode construction is shown in the top panel of Fig. 12.1. Each colored cell represents the discrete value of the mode in the interval $x \in [-4, 4]$ with $\Delta x = 0.1$. Thus there



Figure 12.1 The top panel shows the first 10 modes of the quantum harmonic oscillator considered in (11.25) and (11.26). Three randomly generated measurement matrices, P_j with j = 1, 2 and 3, are depicted. There is a 20% chance of performing a measurement at a given spatial location x_j in the interval $x \in [-4, 4]$ with a spacing of $\Delta x = 0.1$.



Figure 12: The top panel shows the original function block is along with a 10 mode reconstruction of the tota function $(\rho_1, \rho_2) = \exp(-c_1 - e_2)^{-1} + \exp(-c_2)^{-1} + \exp$

are 81 discrete values for each of the modes ψ_k . Our objective is to reconstruct a function outside of the basis modes of the harmonic oscillator. In particular, consider the function

$$f(x) = \exp[-(x - 0.5)^2] + 3\exp[-2(x + 3/2)^2]$$
 (12.11)

which will be discretized and defined over the same domain as the model basis of the large mode constitute. The following code basis that following a first hard for a final effect manually constrained the possicities of the functions on the basis functions Ψ_{i} . The original function is plotted by E_{i} . Zo Note that the galo model is E_{i} . Zo Note that the galo model is E_{i} . Zo Note that the galo model is E_{i} . Zo Note that the galo model is the function basis functions is the function of the function of

Code 12.2 Test function and reconstruction.

```
 \begin{array}{l} f_{11} \left( \exp\left(-(x-0,s\right), ^{-2}\right) + 3 \cdot \exp\left(-3 \cdot (x+1,s), ^{-2}\right) \right) \times i \\ for j=1:10 \quad \& \quad full \quad reconstruction \\ = i(j,2) \quad & \text{strenge} \left(x, f, \, v_{\text{pharm}}(i,j)\right) \\ \text{and} \\ f = v_{\text{pharm}}, a_{j} \\ \quad & \text{subplot}(2,1,1), \quad & \text{plot}(x, f2, ^{+}r^{+}) \\ \text{Reconstruction error} \end{array}
```

```
for j=1:10 # matrix M reconstruction
      Areastrapz(x,yharn(:,j).,yharn(:,jj));
      M(i,ii)=Area:
cond(N) & get condition number
```

Results of the low-rank and gappy reconstruction are shown in Fig. 12.2. The lowrank reconstruction is performed using the full measurements projected to the 10 leading harmonic oscillator modes. In this case, the inner product of the measurement matrix is given by (12.4a) and is approximately the identify. The fact that we are working on a limited domain $x \in [-4, 4]$ with a discretization step of $\Delta x = 0.1$ is what makes $M \approx I$ versus being exactly the identify. For the three different sparse measurement scenarios P₁ of Fig. 12.1, the reconstruction is also shown along with the least-square error and the logarithm of the condition number $log(\kappa(M_i))$. We also visualize the three matrices M₁ in Fig. 12.3. The condition number of each of these matrices helps determine its reconstruction accuracy.







Figure 12.3 Demonstration of the deterioration of the orthogonality of the modal basis in the support space xlul as given by the matrix M defined in (12.4). The top left shows that the identity matrix is produced for full measurements, or nearly so but with errors due to truncation of the domain over $x \in [-4, 4]$. The matrices M₁, which longer look diagonal, correspond to the sparse sampling matrices P1 in Fig. 12.1. Thus it is clear that the modes are not orthogonal in the support space of the measurements

```
Code 12.3 Gappy sampling of harmonic oscillator.
```

```
[advg=(*,*,*)]; i lowe different measurement make
fragment), major(1,1,1); advg=120 routen measurements
internet[main], i lowed [i low 200 routen measurements
fragment]), mass[i lowed 201 routen measurements
fragment]), mass[i lowed 201 routen measurements
fragment]), mass[i lowed 201 routen measurements
fragment], measurements[]
measurements[], measurements[], measurements[],
measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[], measurements[
```

12.2 Error and Convergence of Gappy POD

As was shown in the previous section, the ability of the gampy sampling strategy to accurately reconstruct a given function depends critically on the placement of the measurement (userso) locations. Given the importance of his issue, we will discuss a variety of principled methods for placing a limited number of sensors in detail in subsequent sections. Our agoin in this section is to investigate the covergence properties and error associated with the gappy method as a function of the percentage of sampling of the full system. Random sumitaria locations will be used.

Given our random sampling strategy, the results that follow will be statistical in nature, computing averages and variances for batches of randomly selected sampling. The modal basis for our numerical experiments are again the Gauss-Hermite functions defined by (11.23) and (11.26), generated by Code 12.1 and shown in the top panel of Fig. 12.1.

Random Sampling and Convergence


Figure 12.4 Logarithm of the least-square error, $\log(E + 1)$ (unity is added to avoid negative numbers), and the log of the condition number, $\log(e(M))$, as a function of percentage of random measurements. For 10% measurements, the error and condition number are largest as expected. However, the variance of the results, depicted by the red bars is also quite large, suggesting that the performance for a small number of sensors is highly sensoria vitoria to the dement.

Fig. 12.4 depicts the average over 1000 trials of the logarithm of the lass-square energy, log(E+1) (unity is added to avoid negative numbers), and the log of the condition number, log(E+1) (unity is a direction of percentage of random measurements. Also depicted is the rotations or with the edb has donoting in a when μ is the average value. The error and energy of the edb has donoting in μ when μ is the average value. The error and error does not approach zero states only a 16-mode basis expansion is used, thus limiting the accuracy of the PDD expansion and arcsonatischic even with Hill measurements.

The following code, which is the basis for constructing Fig. 12.4, draws over 1000 random sensor configurations using 10% 2, 50%, 50%, 60% and 50% wangling. The full reconstruction (100% sampling) is actually performed in Code 12.2 and is used to make the final graphic for Fig. 12.3. Note that are accreticit, the error and condition number trends are similar, thus supporting the hypothesis that the condition number can be used to evaluate the efficacy of the gaven measurement. Indeed, this clearly shows that the condition number provides an evaluation that does not require knowledge of the function in (22.1).

Code 12.4 Convergence of error and condition number

```
for threatw1:5;
for jloopw1:1000 $ 1000 random trials
n2srandsample(n,%.threat); $ random sampling
Pwaros(n,1); P(n2)=1;
for js1:10
for jis1:10 $ compute N matrix
```



Figure 123 Statistics of 20% random measurements considered in Fig. 1.2.4. The top panel (a) depices 200 random times and the condition number lengt (M) (M) or the time A strangement of (b) the logarithm of the East-square error, lengt E + 1, and (c) condition number. Lengt(M)), are also depiced for the 200 min. The figures literature the extremely high variability generated from the results and extremely peop spectrum error of the strategies of the time of the time of the strategies of the strat

Gappy Measurements and Performance

We can continue this statistical analysis of the gappy reconstruction method by looking more carefully at 200 random trials of 20% measurements. Fig. 12.5 shows three key features of the 200 random trials. In particular, as shown in the top panel of this figure,



Figure 12.6 Depiction of the 200 random 20% measurement vectors \mathbf{P}_j considered in Fig. 12.5. Each row is a randomly generated measurement trial (from 1 to 200) while the columns represent their spatial location on the domain $x \in [-4, 4]$ with $\Delta x = 0.1$.

there is a large variance in the distribution of the condition number a (M) for 20% sampling. Specifically, the condition number can change by detect of nugminds with the same number of sensors, but simply placed in different locations. A histogram of the distribution of the log ereo (log (1 - 1)) and the log of the condition number are achieved in the bottom row panels. The error appears to be distributed in an exponentially decaying fashion whereas the condition numbers are exceptionally high, suggesting sensor configurations to be avaded.

In order to visualize the random, gappy measurements of the 200 samples used in the statistical analysis of Fig. 12.5, we plot the **P**₂ measurement masks in each row of the matrix is Fig. 12.6. The while regions represent regions where no measurements state the Mack regions are where the measurements are taken. These are the measurements that generate the orders of magnitude variance in the error and condition number.

As a final analysis, we can slit through the 200 random measuremers of Fig. 12.6 and pick out both the ten best and ten worst measurement vectors \mathbf{P}_j . Fig. 12.7 shows the results of this sitting process. The top two parels depict the best and worst measurement configurations. Interestingly, the worst measurements have long stretches of missing measurements near the center of the domain where much of the modul variance occurs.



Figure 32.2 Depiction of the 10 best and 10 west random 20% measurement vectors **P**₂ considered in Figure 1.2 and 1.2.5. The top pasel shows fut the best measurement vectors using the figure 1.2 and 1.2.5. The second shows fut the best measurement (undisting the mach have tage scanning figures on the descent of the domain scales to a large combinion number e MD. The bettom panel shows a bare that of the best and wears taking to a scale s

In contrast, the best measurements have well sampled domains with few long gaps between measurement locations. The bottom panel shows that the best measurements (on the left) offer an improvement of two orders of magnitude in the condition number over the poor performing counterparts (on the right).

12.3 Gappy Measurements: Minimize Condition Number

The preceding section librarias that the placement of gappy measurements is critical for accurately reconstructing the PDD solution. This suggests that a principied way to determine measurement becautons is of great importance. In what follows, we confine a method originally proposed by Wilking (SSS) dessersing the gappy measurement locations. The methods whose the conditions of the second second

The algorithm proposed [555] is computationally costly, but it can be performed in an offline training stage. Once the sensor locations are determined, they can be used for online reconstruction. The algorithm is as follows:

 Place sensor k at each spatial location possible and evaluate the condition number κ(M). Only points not already containing a sensor are considered.



Figure 12. Despition of the first four intension of the paper measurement beation algorithm of Witton (555). The advanced in the paper measurement beation algorithm function (11.25) and (11.26) discritted on the interval $x \in \{-4, 4\}$ with Ax = 0.1. The uppend shows the confidem means (A) as a single series in consolicated at each beat B intervel to series of the series measurement of the single series of the 11 discret Weiter x_2 . The first mean minimum the conducton number (shows in nucl at x_{22} . A second series a norconsidered at all emission for some discretion of the minimul conduction more location for its series of the series of the series of the single series of the series o

- Determine the spatial location that minimizes the condition number κ(M). This spatial location is now the kth sensor location.
- Add sensor k + 1 and repeat the previous two steps.

The algorithm is not optimal, nor are there guaranteed. However, it works quite well in practice since sensor configurations with low condition number produce good reconstructions with the POD modes.

We apply this algorithm to construct the gappy measurement matrix \mathbf{P} . As before, the model basis for our numerical experiments are the Gauss-Hennitz functions defined by (1123) and (11.50). The gappy measurement matrix algorithm for constructing \mathbf{P} is shown in Norke that the algorithm outfind above sets down one sense on a time, thus with the 10 POD mode expansion, the system is underdetermined until 10 sensors are placed. This is suggests that the first 10 sensor locations may be generated from inaccurate calculations of the condition numbers on the outer of 10% or the first 9 sensor placements. It also suggests that the first 10 sensor locations may be generated from inaccurate calculations of

The following code builds upon Code 12.1 which is used to generate the 10-mode expansion of the Gauss-Hermite functions. The code minimizes the condition number and identifies the first 20 sensor locations. Specifically, the code provides a principled way of producing a measurement matrix P that allows for good reconstruction of the POD mode expansion with limited measurements.

Code 12.5 Garny placement: Minimize condition number

In addition to identifying the placement of the first 20 sensors, the code also reconstructs the example function given by (12.11) at each iteration of the routine. Note the use of the setdiff command which removes the condition number minimizing sensor location from consideration in the next iteration.

To column for gapty sense bosonic algorithm, we task the condition marker as a minimum of the marker for the property in 10 Survey. Additushy, at each horizon, a final single and the sense of the sense by constructions of the algorithm mainteen in the condition marker of the sense by constructions the gaptimum single sense by constructions of the sense sense sectors after sense the spectra in the sense marker sense that the sense is the sense in the spectra of the spectra in the spectra of the sense marker sense that the sense is not submarker shows that the sense is a sense that all descense that the sense marker sense is not submarker sense that the sense is not submarker sense that the sense is not submarker sense that the sense is not submarker sense the sense is not submarker sense the sense is not submarker sense that the sense marker sense the sense marker sense that the sense marker sense that the sense marker sense the sen



Figure 12: Condition number and lass-space error (logarithm) as a function of the number of interaints on the graym sour plearcent alogarithm. The log of the condition number high (M) high statistical states and the states of the states of the states of the states of the logarithm of the states of the field rank and the conditions mattered drops by orders of magnitude. The bottom part droves the full rank and the conditions mattered drops by orders of magnitude. The bottom part droves the full rank and the conditions mattered drops by orders of magnitude. The bottom part droves the conditions of the states of the states of the states between the states of the states. The states of the states. The states of the states of

10 measurements are made. In general, if an r-mode POD expansion is to be considered, then reasonable results using the gappy reconstruction cannot be achieved until r sensors are placed.

We now consider the placement of the sensors as a function of iteration in the bottometer panel of Fig. 12.5, Specifically, we depict when sensors are identified in the iteration. The first sensor location is 32, followed by 32, 32 and 32, 32 perceively. The processing is continued until the first 20 sensors are identified. The pattern of sensors depicted is important as it illustrates a fairly uniform sampling of the domain. Alternative schemes with the considered in the following.

As a final illustration of the gapy algorithm, we consider the resonancesion of the test function (12.11) as the number of herizoin (scenosi) increase. As expected, he more sensors that are used in the gappy framework, the better the reconstruction is, expectably in they are placed in a principide way as coulding by Wilco (S23), Fig. 12.10 shows the reconstructed function with increasing increasion number, in the left panel, increasion are network ways of the increasing increasion matters in the left panel, increasion from instruction there is thereased on the interaction from the the reconstruction from iteration in the orienty and on a none illumid r-axis value where the reconstruction from the resonance of the interaction. The right panel highlights the reconstruction from iteration into the orienty and on a none illumid r-axis value where the reconstruction



Figure 12:10 Convergence of the reconstruction to the test function (12.11). The left parel shows iterations one through brendys and the significant reconstructions errors of the early iterations and limited number of sensors. Indeed, for the first into iterations, the condition number and least-square errors in quick large size the system is not fift attact. The right parallel dows a zone-in of the solution from iteration mice to twenty where the convergence is clearly observed. Comparison in both paralect can be made to the test function.



Figure 211 Sum of diagonals minus off-diagonals (top (td)) and least-square error (dopriming to a function of the market of interation in the scenard paper streng characterial approximation. The area proxy metric for condition markets monotonically increases since this is being maximized at each interitors where the scenard strength strength strength strength strength strength strength strength strength towards improvement as the member of seman and strength show at strength strength

converges to the test function. The true test function is also shown in order to visualize the comparison. This illustrates in a tangible way the convergence of the iteration algorithm to the test solution with a orienteded nlacement of sensors.

Proxy Measures to the Condition Number

We end this section by considering alternative measures to the condition number (Al). The computation of the condition number itself can be computationally expensive. Moreover, until s statistics are chosen in an >400 mode expansion, the condition number computation is itself numerically unstable. However, it is clear what the condition number minimization algorithm is trying to achiever, make the measurement matrix M as near to the identity are start to the identity and the measurement. possible. This suggests the following alternative algorithm, which was also developed by Wilcox [555].

- Place sensor k at each spatial location possible and evaluate the difference in the sum of the diagonal entries of the matrix M minus the sum of the off-diagonal components, call this k₂(M). Only points not already containing a sensor are considered.
- Determine the spatial location that generates the maximum value of the above quantify. This spatial location is now the kth sensor location.
- Add sensor k + 1 and repeat the previous two steps.

This algorithm provides a simple modification of the original algorithm which minimizes the condition number. In particular, the following lines of code provide modifications to Code 12.5. Specifically, where the condition number is computed, the following line is now included:

nallssetdiff(nall,ns); % new sensor indeces

Additonally, the sensor locations are now considered at the maximal points so that the following line of code is applied

Pereros(n,1); P(ns)=1;

Thus the modification of two lines of code can enact this new metric which circumvents the computation of the condition number.

To evaluate this new gapsy sensor location algorithm, we track the new proxy numerican evaluation of the method of the sensor of our test function of the 110.1 h this case, up to 60 sensors are considered to the test of test

As before, we also consider the placement of the sensors as a function of iteration in the right panel of P_1 [1.1.1. specialized) we depict the turning on process of the sensors. The first sensor location is x_2 followed by x_{2k} , x_{2k} and x_2 respectively. The process is continued until the first of the sensors turned on. The purpose of sensors depicted is algorithm, and with these modes, turns on sensors in local locations without sampling uniformly from the domain.

12.4 Gappy Measurements: Maximal Variance

The previous section developed principled ways to determine the location of sensors for gappy POD measurements. This was a significant improvement over simply choosing sensor locations randomly. Indeed, the minimization of the condition ramber through location selection performed quite well, quickly improving accuracy and least-square reconstruction neuror. The drankstic to the proposed method was two-fold the algorithm it neitif is expensive to implement, requiring a computation of the condition number for every sensor location selected under an exhaustive search. Secondly, the algorithm it neitif and the method until the rth sensor was chosen in an *i*-POD mode expansion. Thus the condition number was theoretically infinite, but on the order of 10¹⁷ for computational purposes.

Karniadakis and co-workers [565] peopsed an alternative to the Willcox [555] algorithm to overcome the computational issues outlined. Specifically, instead of placing one sensor at a time, the new algorithm placer s sensors, for an r-POD mode expansion, at the first step of the iteration. Thus the matrix generated is no longer ill-conditioned with a theoretically infinite condition muther.

The algorithm by Karniadakis further proposes a principled way to select the original r sensor locations. This method selects locations that are extrema points of the POD modes, which are designed to maximally capture variance in the data. Specifically, the following algorithm is suggested:

- Place r sensors initially.
- Determine the spatial locations of these first r sensors by considering the maximum of each of the POD modes w_n.
- 3. Add additional sensors at the next largest extrema of the POD modes.

The following code determines the maximum of each mode and constructs a gappy measurement matrix P from such locations.

Code 12.6 Garrey placement: Maximize variance

```
ns=[1;
for j=1:10 % welk through the modes
[s1,n1]smax(yharm(:,j)); % pick max
ns=[ns n1];
and
Paraeros(n,1); P(ns)=1;
```

The performance of this algorithm is not strong for only r measurements, but it at least produces stable condition number calculations. To improve performance, one could also use the minimum of each of the modes Ψ_{ℓ} . Thus the maximal value and minimal value of variance are considered. For the humonic isocilitate code, the first mode produces no minimum as the minimum are at $x \rightarrow \pm \infty$. Thus 19 sensor locations are chosen in the following code:

Code 12.7 Garry placement: Max and min variance

```
nms(j)
for juli0 % walk through the modes
[sl,n],max(yharm(:,j)); % pick max
and
for juli0
[sl,n],min(yharm(:,j)); % pick max
mas[n n 2];
and
Powerce(n,l): P(ns)el;
```



Figure 12. The top panel shows the mode structures of the Gauss-Hermite polynomials Ψ in the nor-mark approximation of a POD expansion. The discretization interval is $z \in [-1, 4]$ with a spacing of $\Delta x \equiv 0.1$. The color map shows the maximum (which) and minimum (black) that occurs in the mode structures. The leston panel shows the grad coll concepting to maximum and minimum (colorum) of POD mode variance. The extremus are candidate for senses locations, or the marker marker

Note that in this case, the number of sensors is almost double that of the previous case. Moreover it only searches for the the locations where variability is highest, which is intutively appealing for measurements.

More generally, the Karaiadaki adgorthm [563] advocates randomly selecting p eson from M potential extrema, and the modifying the search positions with the goal of improving the condition number. In this case, one must identify all the maxima and minima of the POD modes is order to make the selection. The harmonic socillator modes and their maxima and minima are illustrated in Fig. 12.12. The algorithm used to produce the extrema of each mode, and in potential for use in the grapy algorithm, is as follows:

Code 12.8 Gappy placement: Extrema locations.

```
pmacelp = memin(1)
from the present (1) = memin(1) = memin(1)
```



Figure 12:3 Condition number and least-square error to test function (12.11) over 100 random trials that draw 20 sensor locations from the possible 55 extrema depixted in Fig. 12.12. The 100 trials produce a ramber of sensor configurations that particular control to the level of the condition number minimization algorithm of the last section. However, the computational costs in generating such trials can be significantly lower.

```
ns=[nmax nmin];
nisrandsample(length(ns),20);
nsrens(ni);
Pereros(n,1); P(nsr)=1;
```

Note that the resulting vector ns contains all 55 possible extrema. This computation assumes the data is sufficiently smooth so that extrema are simply found by considering neighboring points, i.e. a maxima exists if its two neighbors have a lower value whereas an minima exists if its neighbors have a higher value.

The maximal variance algorithms suggests trying different configurations of the sensors in the extramo joics. In particular, if 29 appre massurements are disciled, due use would need to search through various configurations of the 55 locations using 20 sensors. This combinated at search, its instrachle. However, if we simply attempt 100 random trials and select the best performing configuration, it is quite close to the performance of the condition number minimizing algorithm. A filt exceeding of the location allow with a compartion of the condition number and least-square fit error with (12.11), is generated by the following code:

Code 12.9 Gappy placement: Random selection.

```
ntot=length(ns);
for jtrials=1:100
```



Figure 12. A Performance metrics for placing memors based upon the externm of the variance of the POD modes. Both the data sequarater rend for the reconstraints of the test functions (12.11) and the reconstraints of the test sequences of the test sectors of the test set of the sectors because the sectors of 20 of the Sectors and the test sectors of the POD modes. These are compared appairs (d) the Sectors of 20 of the Sectors minimization and (e) the confident method method method methods Sectors of 20 of the test of the test sectors and (e) the confident method method method sectors of 20 of the file of the confident methods and (e) the confident method method method methods and the file of the confident methods and (e) the confident method method method methods methods and the file of the confident methods method method method method methods and methods and the file of the confident method method method method method methods and methods and the file of the confident method method method method method methods and methods and the file of the confident method method method method method methods and methods and the file of the confident method methods and methods and the file of the confident method meth

```
and
subplot(2,1,1), bar(log(cos_tri),'Facecolor',[0.7 0.7 0.7])
subplot(2,1,2), bar(log(E_tri+1),'Facecolor',[0.7 0.7 0.7])
```

The condition number and least-square error for the 100 trials is shown in Fig. 12.13. The configurations perform well compared with random measurements, although some have excellent performance.

A direct comparison of all these remodels is shown in Fig. 1.1.3. Specificility, whit is immediated are the results in using (in the maximum hockniss of the PDO modes, (b) the maximum at minimum bocation of eta PDO mode, b) and the maximum at minimum bocation of eta PDO mode. The eta PDO mode is the PDO mode is the

12.5 POD and the Discrete Empirical Interpolation Method (DEIM)

The POD method illustrated thus far aims to exploit the underlying low-dimensional dynamics observed in many high-dimensional computations. POD is often used for reduced-order models (ROMs), which are of growing importance in scientific applications and computing. ROMS reduce the computational complexity and time needed to solve large-scale, complex systems [53, 442, 244, 17]. Specifically, ROMs provide a principled approach to approximating high-dimensional spatio-temporal systems [139], typically generated from numerical discretization, by low-dimensional subspaces that produce nearly identical input/output characteristics of the underlying nonlinear dynamical system. However, despite the significant reduction in dimensionality with a POD basis, the complexity of evaluating higher-order nonlinear terms may remain as challenging as the original problem [41, 127]. The empirical interpolation method (EIM), and the simplified discrete empirical interpolation method (DEIM) for the proper orthogonal decomposition (POD) [347, 251], overcome this difficulty by providing a computationally efficient method for discretely (sparsely) sampling and evaluating the nonlinearity. These methods ensure that the computational complexity of ROMs scale favorably with the rank of the approximation, even with complex nonlinearities.

¹EM has been developed for the purpose of efficiently managing the computation of the nonlinearity in discussionality relevance scenes, with DEM weekforthy unleaded to FOD with Galerkin projection, Indeed, DEM approximates the nonlinearity by mainly small, discrete sampling of points that are determined in an algorithmic way. This resource that the computational cost of evaluating the nonlinearity projection of the memory of the computational cost of evaluating the nonlinearity projection transmission. A simple the confidence frequencies in the FOO Galerkin proprimities to endue, to example the confidence frequencies in the FOO Galerkin programming to endue, considing in "operations to evaluate the nonlinear term. IEEM approximation to endue scaling and "operations to evaluate the nonlinear term. TEEM approximation to endue scaling and the scaling scale scale projection of the scale scalence of the scale scalence of the scale Table 12.1 DEIM algorithm for finding approximation basis for the noninearity and its interpolation indices. The algorithm first constructs the nonlinear basis modes and initializes the first measurement location, and the matrix \mathbf{P}_1 , as the maximum of $\boldsymbol{\beta}_1$. The algorithm then successively constructs columns of \mathbf{P}_1 by considering the location of the maximum of the residual \mathbf{B}_1 .

DEIM algorithm Basis Construction and Initialization	
Interpolation Indices and It	eration Loop (j = 2, 3,, p)
calculate e _j compute residual find index of maximum residual add new column to measurement matrix	$\begin{array}{l} \mathbf{P}_{j}^{T} \boldsymbol{\Xi}_{j} \mathbf{c}_{j} = \mathbf{P}_{j}^{T} \boldsymbol{\xi}_{j+1} \\ \mathbf{R}_{j+1} = \boldsymbol{\xi}_{j+1} - \boldsymbol{\Xi}_{j} \mathbf{c}_{j} \\ \left[\boldsymbol{\rho}, \boldsymbol{\gamma}_{j} \right] = \max \left[\mathbf{R}_{j+1} \right] \\ \mathbf{P}_{j+1} = \left[\mathbf{P}_{j} \ \mathbf{e}_{\gamma_{j}} \right] \end{array}$

preceiving a low-dimensional (O(1)) comparison, as desired. The DBM approach containes projection with interplotation. Specifically, DBM and scienced marginalism containes projection with interplotation. Specifically, DBM and science and approximation approximating the nonlinearity, IDM/DBM are not fire only methods developed to roltest the complexity of e-submarkary, DBM/DBM are not fire only methods. However, they have been accorded in a large number of diverse applications and models (127). In any case, the MMC, gappy POD, and EMM/DBM are a small selected set of quantal gap case, the MMC, gappy POD, and EMM/DBM are a small selected are of quantal gap errors.

POD and DEIM

Consider a high-dimensional system of nonlinear differential equations that can arise, for example, from the finite difference discretization of a partial differential equation. In addition to constructing a supplot matrix (12.12) of the solution of the PDE so that POD modes can be extracted, the DEIM algorithm also constructs a snapshot matrix of the nonlinear term of the PDE:

$$\mathbf{N} = \begin{bmatrix} \mathbf{I} & \mathbf{I} \\ \mathbf{N}_1 & \mathbf{N}_2 & \cdots & \mathbf{N}_m \end{bmatrix}$$
(12.12)

where the columns $N_k \in \mathbb{C}^n$ are evaluations of the nonlinearity at time t_k .

To achieve high accuracy solutions, n is typically very large, making the computation of the solution expensive and/or intractable. The POD-Galerkin method is a principled idmensionally-reduction scheme that anoroximates the function u(r) with rank-r outimal



Figure 32.3 Demonstration of the first three iterations of the DEM algorithm. For illustration only, the nonlinearity marks $N = \Sigma_{\rm T} \sqrt{2}$ as anomaly to be composed of harmonic oscillar mode, with the first ten mode c, ratherwards, there is a three at ppresent for action is chosen at the maximum of the first mode 1, Alexironali, there is a three at ppresents of acticing subsequent masurements locations based upon the location of the maximum of the residual vector **B**. The first (red), scenario the subsections are shown along with the construction of the summarized based marks and the constraint of the maximum of the maximum of the state of th

basis functions where $r \ll n$. As shown in the previous chapter, these optimal basis functions are computed from a singular value decomposition of a series of temporal snapshots of the complex system.

The standard POD procedure [231] is a subiquitons algorithm in the reduced order modiding community, however, it also helps limitarise the need for innovations such as DEBM. Gappy POD and/or MPE: Cossider the nonlinear component of the low-dimensional volution (121): ψ (W)(ψ)) for a single nonlinearity synthem λ (w(t), t) = w(t, t), consider in singlet on a spatially-discretized, through ΔR (w(t), t) = w(t), t, consider in singlet on a spatially-discretized, through ΔR (w(t), t) regions the probability of the pr

$$r(x, t)^3 = a_1^3 \psi_1^3 + 3a_1^2 a_2 \psi_1^2 \psi_2 + 3a_1 a_2^2 \psi_1 \psi_2^2 + a_3^2 \psi_3^3$$
. (12.13)

The dynamics of $a_1(t)$ and $a_2(t)$ would then be compated by projecting onto the lowdimensional basis by taking the inner product of this nonlinear term with respect to both ψ_1 and ψ_2 . Thus the number of computations nor only doubles, but the inner products must be compated with the *n*-dimensional vectors. Methods such as DEIM overcome this high-dimensional computation.

DEIM

As outlined in the previous section, the shortcomings of the POD-Galerkin method are generally due to the evaluation of the nonlinear term $N(\Psi a(t))$. To avoid this difficulty, DEBM approximates $N(\Psi a(t))$ through projection and interpolation instead of evaluating it directly. Specifically, a low-rank representation of the nonlinearity is computed from the simular value decomposition

$$N = \Xi \Sigma_N V_N^*$$
(12.14)

where the matrix Ξ contains the optimal basis for spanning the nonlinearity. Specifically, we consider the rank-p basis

$$\Xi_p = [\xi_1, \xi_2, \dots, \xi_n]$$
 (12.15)

that approximates the nonlinear function ($p \ll n$ and $p \sim r$). The approximation to the nonlinearity N is given by:

$$N \approx \Xi_{\mu} c(t)$$
 (12.16)

where c(t) is similar to a(t) in (11.20). Since this is a highly overdetermined system, a suitable vector c(t) can be found by selecting p rows of the system. The DEIM algorithm was developed to identify which p rows to evaluate.

The DEIM algorithm begins by considering the vectors $\mathbf{e}_{r_j} \in \mathbf{R}^*$ which are the γ_j -th column of the *n* dimensional identity matrix. We can then construct the projection matrix $\mathbf{P} = [\mathbf{e}_{r_1} \mathbf{e}_{r_2} \cdots \mathbf{e}_{r_p}]$ which is chosen so that $\mathbf{P}^T \equiv_{r_j}$ is nonsingular. Then $\mathbf{e}(t)$ is uniquely defined from $\mathbf{P}^T \mathbf{N} = \mathbf{P}^T \equiv_{\mathbf{n}}(t)$, and thus,

$$\mathbf{N} \approx \Xi_{\rho} (\mathbf{P}^T \Xi_{\rho})^{-1} \mathbf{P}^T \mathbf{N}.$$
 (12.17)

The tremendous advantage of this result for nonlinear model reduction is that the term \mathbf{P}^{T} N requires evaluation of the nonlinearity only at $p \ll n$ indices. DEIM further proposes a principled method for choosing the basis vectors $\mathbf{\xi}_{j}$ and indices y_{j} . The DEIM algorithm, which is based on a greedy search, is detailed in [127] and further demonstrated in Table 12.1.

POD and DEIM provide a number of advantages for nonlinear model reduction of complex systems. POD provides a principled way no commercian and antipacte & characterizing the dynamics. DEIM augments POD by providing a method to evaluate the problematic nonlinear terms using an *p*-dimensional subspace \mathbb{Z}_p that represents the nonlinearity. Thus a small number of points can be sampled to approximate the nonlinear terms in the ROM.

12.6 DEIM Algorithm Implementation

To demonstrate model reduction with DEIM, we again consider the NLS equation (11.29). Recall that the numerical method for solving this equation is given in Codes 11.3 and 11.4. The output of this code is a matrix used whose rows represent the time snapshots and whose columns represent the spatial discretization points. As in the first section of this chapter, our first are jot to transpose this data so than the time snapshots are columns instead of rows. The following code transposes the data and also performs a singular value decomposition to get the POD modes.

Code 12.10 Dimensionality reduction for NLS.

```
Xsumol.'; % data matrix X
[U,S,W] savd(X,0); % SVD reduction
```

In addition to the standard POD modes, the singular value decomposition of the nonlinear term is also required for the DEIM algorithm. This computes the low-rank representation of $N(u) = |u|^2 u$ directly as $N = \Xi_{x} \nabla_{x}^{N}$.

Code 12.11 Dimensionality reduction for nonlinearity of NLS.

```
NL=i*(abs(X).^2).*X;
[XI,S NL,W]=svd(NL,0);
```

Once the low-rank structures are computed, the rank of the system is chosen with the parameter r. In what follows, we choose r = p = 3 so that both the standard POD modes and nonlinear modes, $\dot{\Psi}$ and Ξ_p have three columns each. The following code selects the POD modes for $\dot{\Psi}$ and projects the initial condition onto the POD subspace.

Code 12.12 Rank selection and POD modes

```
r=3; % select rank truncation
PsisU(:,1:r); % select POD modes
s=Psi'_su0; % project initial conditions
```

We now build the interpolation matrix P by executing the DEIM algorithm outlined in the last section. The algorithm starts by selecting the first interpolation point from the maximum of the first most dominant mode of Ξ_n .

Code 12.13 First DEIM point

```
[Xi_max, nmax] =max(abs(XI(:,1)));
XI_meXI(:,1);
x=eros(n,1);
Pez: P(nmax)=1;
```

The algorithm iteratively builds **P** one column at a time. The next step of the algorithm is compute the second so the iterapolation point via the greed DEM algorithm. Specifically, the vector e_1 is compared from $\mathbf{P}^T_{2} e_{ij} = \mathbf{P}^T_{2} e_{ij}$, where k_{ji} are the columns of the model sources \mathbf{x}_{ij} . The second iteratively dimensional region and the column is of the source interpolation point counts from the body and the source \mathbf{x}_{ij} is the second source \mathbf{x}_{ij} . The source \mathbf{x}_{ij} is the source \mathbf

Code 12.14 DEIM points 2 through r.

```
for j=2:r
    cs(P*sII_m)\(P*sII(:,j));
    ressII(:,j)-XI_mec;
    [X1_max,nmax]emax(abs(res));
    XI_max[xI_mXI(:,j)];
    P=[P,z]; P(rmax,j)=1;
    and
```



Figure 12.18 Comparison of the (a) full simulation dynamics and (b) rank r = 3 ROM using the three DEIM interpolation points, (c) A detail of the three POD modes used for simulation are shown along with the first, second and third DEIM interpolation point location. These three interpolation points are carable of accurately treeroducian the constitution of the full POE vystem.

With the interpolation matrix, we are ready to construct the ROM. The first part is to construct the linear term $\Psi^T L \Psi$ of (11.21) where the linear operator for NLS is the Laplacian. The derivatives are computed using the Fourier transform.

Code 12.15 Projection of linear terms.

```
for j=lir % linear derivative terms
Lox(:,j)=ifft(-k.^2.*fft(Pai(:,j)));
end
Le(i/2)*(Pai')*Lox: % projected linear term
```

The projection of the nonlinearity is accomplished using the interpolation matrix \mathbf{P} with the formula (12.17). Recall that the nonlinear term in (11.21) is multiplied by Ψ^T . Also computed is the interpolated version of the low-rank subspace spanned by Ψ .

Code 12.16 Projection of nonlinear terms.

```
P_NL=Pmi' + ( XI_m + inv(P' + XI_m) ); 
    nonlinear projection
    P Pmi=P' + Pmi;
    interpolation of Pmi
```

It only remains now to advance the solution in time using a numerical time stepper. This is done with a 4th-order Runge-Kutta routine.

Code 12.17 Time steering of ROM

```
[tt,s]=ode45('rom deim_rhs',t,s,[],P_NL,P_Psi,L);
XtildesPsiss'; * DELM approximation
waterfal(x,t,abe(Xtilde')), shading interp, colormap gray
```

The right hand side of the time stepper is now completely low dimensional.

Code 12.18 Right hand side of ROM.

```
function rhssrom_deim_rhs(tspan, s,dummy,P_NL,P_Psi,L)
NsP_Psiss;
rhssItss + i*P NL*((abs(N).^2).*N);
```

A comparison of the full simulation dynamics and rank r = 3 ROM using the three DEIM interpolation points is shown in Fig. 12.16. Additionally, the location of the DEIM points relative to the POD modes is shown. Aside from the first DEIM point, the other locations are not on the minima or maxima of the POD modes. Rather, the algorithms relaces them to mountimize the residual.

QDEIM Algorithm

Alhough DEM is an efficient greedy algorithm for selecting interpolation points, there are other techniques that are equally efficient. The recercity proposed DEM algorithm [192] leverages the QR decomposition to provide efficient, greedy interpolation locations. This has been show to be a cohost nutlementatic architecture for sense projecture in a many applications [566]. See Section 3.2 for a more general discussion. The QR decomposition can also provide a greedy staturegy to identify interpolation points. Others, the QR point locations are sense locations. The following code can replace the DEM algorithm to produce the interpolation matrix **P**.

Code 12.19 QR based interpolation points

[Q,R,pivot]sqr(NL.'); Pspivot(:,1:r);

Using this interpolation matrix gives identical interpolation locations as shown in Fig. 12.16. More generally, there are estimates that show that the QDEIM may improve error performance over standard DEIM [159]. The case of use of the QR algorithm makes this an attractive method for source interpolation.

12.7 Machine Learning ROMs

Inspired by machine learning methods, the various TOD bases for a parametrized system are negod into a match Broys of POD modes by which contain all the loss rads ado spaces calibrated by the dynamical system. This leverages the fact that POD provides a tripicable valy to contrast as -d-messional analogue e_{γ} , duracterizing the dynamics while space sampling anguments the POD method by providing a anthol to evaluate the space of the problem in the space of the space space of the space of

The method introduced here capitalizes on these methods by building low dimensional hieraries associated with the full nonlinear system dynamics as well as the specific nonlinearities. Interpolation points, as will be shown in what follows, can be used with space representation and compressive sensities (to (1) dentify dynamical regimes, (1)) reconstruct the full state of the system, and (iii) provide an efficient nonlinear model reduction and POOP Galekin metication for the future state.



Figure 12.1 Library construction from numerical simulations of the governing equations (11.1). Simulations are performed of the parametrized system for different values of a bifurcation parameter µ. For each regime, low-dimensional POD modes Ψ_{μ} are computed via an SVD decomposition. The various nake-r transcated subspaces are stored in the library of modes matrix Ψ_{μ} . This is the learning stage of the algorithm. corporationed from Kater et al. [310]

The concept of library building of low-rank features from data is well established in the computer science community. In the reduced-order modeling community, it has recently become an enabling computational strategy for parametric systems. Indeed, a variety of recent works have produced libraries of ROM models [80, 98, 462, 10, 134, 422, 421, 420] that can be selected and/or interpolated through measurement and classification. Alternatively, cluster-based reduced order models use a k-means clustering to build a Markov transition model between dynamical states [278]. These recent innovations are similar to the ideas advocated here. However, our focus is on determining how a suitably chosen P can be used across all the libraries for POD mode selection and reconstruction. One can also build two sets of libraries: one for the full dynamics and a second for the nonlinearity so as to make it computationally efficient with the DEIM strategy [462]. Before these more formal techniques based on machine learning were developed, it was already realized that parameter domains could be decomposed into subdomains and a local ROM/POD computed in each subdomain. Patera and co-workers [171] used a partitioning based on a binary tree whereas Amsallem et al. 191 used a Voronoi tessellation of the domain. Such methods were closely related to the work of Du and Gunzburger [160] where the data snarshots were partitioned into subsets and multiple reduced bases computed. The multiple bases were then recombined into a single basis, so it doesn't lead to a library, per se. For a review of these domain partitioning strategies, please see Ref. [11].

POD Mode Selection

Although there are a number of techniques for velecting the correct POD library elements to use, including the workhore *i* – neares clustering algorithm [10, 134, 422, 421, 420], one can also instead make use of sparse sampling and the sparse representation for classification (SRC) innovations confiled in Chapter 3 to characterize the nonlinear dynamical system [80, 98, 462]. Specifically, the goal is to use a limited number of sensors (interpolation points) to classify the dynamical ergine of the system from a nange of potential POD



Figure 12.17 The space representation for classification (SRC) algorithm for library mode selection; we selection 3.6 for most details. Is this multitude of lamowske, a possere measurement in state of the system (1.1) and a lap lip), most details of the state of lamowske and the system (1.1) and the lip lip). The state of lamowske and the system of lamowske and lamokske and lamowske and lamokske and lamowske and lamokske and lamowske and lamokske an

library elements characterized by a parameter β . Once a correct classification is a achieved, a standard c_2 reconstruction of the full state space can be accomplished with the selected subset of POD modes, and a POD-Galerkin prediction can be computed for its future.

In general, we will have a sparse measurement vector $\tilde{\mathbf{u}}$ given by (12.1). The full state vector \mathbf{u} can be approximated with the POD library modes ($\mathbf{u} = \Psi_L \mathbf{a}$), therefore

$$\tilde{u} = P \Psi_L a$$
, (12.18)

where Ψ_{ii} is the low-mark matrix whose columns are POD basis vectors concententle across all *i* genges and *i* is the coefficient vector giving the projection of *i* at onto these POD modes, if **P** Ψ_{ii} obeys the restricted isometry property and a is atfliciantly space in Ψ_{ii} , then it is possible to solve the highly-moderization given provide the spacestar of the provided structure of the space structure of the space of the Housevect under certain conditions, a space solution of equation (12.18) can be found (Sie Chapter 2) by minimizing the *i*, proministed to that

$$\mathbf{c} = \arg \min ||\mathbf{a}'||_1$$
, subject to $\tilde{\mathbf{u}} = \mathbf{P} \Psi_L \mathbf{a}$. (12.19)

The last equation can be solved through standard convex optimization methods. Thus the ℓ_1 norm is a proxy for sparsity. Note that we only use the sparsity for classification, not reconstruction. Fig. 12.18 demonstrates the sparse sampling strategy and prototypical results for the sparse solution **a**.



Figure 12.19 Time dynamics of the pressure field (top panels) for flow around a cylinder for Reynolda number $R_e = 40, 150, 300$ and 1000. Collecting samphots of the dynamics reveals low-dimensional instrumers dominant the dynamics. The dominant three POD pressure modes for each Reynolda number regime are shown in polar coredinates. The pressure scale is in reagenta (bottom left). (revealued from Katz et al. [319))

Example: Flow around a Cylinder

The data we consider comes from numerical simulations of the incompressible Navier-Stokes equation:

$$\frac{\partial u}{\partial t} + u \cdot \nabla u + \nabla p - \frac{1}{Re} \nabla^2 u = 0$$
 (12.20a)

$$7 \cdot u = 0$$
 (12.20b)

where $u(x, y, t) \in \mathbb{R}^2$ represents the 2D velocity, and $p(x, y, t) \in \mathbb{R}^2$ the corresponding pressure field. The boundary condition are a follows: (i) Constant flow of $u = (1, 0)^7$ at x = -15, i.e., the entry of the channel, (ii) Constant pressure of p = 0 at x = 25, i.e., the end of the channel, and (iii) Neumann boundary conditions, i.e., $\frac{2\pi}{m} = 0$ on the boundary of the channel and the cvinder (externed at (x, y) = 0) on an do f radius unity).

For each relevant value of the parameter Re we perform an SVD on the data matrix in order to extract POD modes. It is well known that for relatively low Reynolds number, a first decay of the singular values is observed so that only a few POD modes are needed to characterize the dynamics. Fig. 12.19 shows the 3 most dominant POD modes for Reynolds number Re = 40, 150, 300, 1000. Note that 9% of the total energy variance is scheduler.



Figure 12.20 Illustration of m sparse sensor locations (left panel) for classification and reconstruction of the flow field. The selection of sensory/interpolation locations can be accomplished by various algorithms [80, 98, 462, 281, 374, 89, 540]. For a selected algorithm, the sensing matrix P determines the classification and reconstruction performance. (removalcod from Kate et al. J219)

for the POD mode selection cut-off, giving a total of 1, 3, 3, and 9 POD modes to represent the dynamics in the regimes shown. For a threshold of 99.9%, more modes are required to account for the variability.

Classification of the Royalda number is accompliable by solving the optimization of the Royalda number is accompliable by solving the response configuration of the Royalda number of Roy

Although the classification accuracy h high, many of the filter classifications are due to comparing a Reyrolds number from an endpoint glow, i.e. Reyrolds 1000 is often minickate for Reyrolds, number 100. This is due to the fact that these two Reyrolds number (100, 110) is indexed to the fact that these two Reyrolds number 100, 110 is indexed to the fact that the rest of th

Finally, to vinalize the static spaces sensing and ecconstruction process more carefully, fig. 12.21 also so both Recycloads number communition for the nine-vinage flow field along with the SPCs discussion of the Recycload number of the SPCs discussion of the SPCs discussion of the SPC discussion discussion of the SPC discussion discussion discussion discussion discussion





Suggested Reading

Texts

- Certified reduced basis methods for parametrized partial differential equations, by J. Hesthaven, G. Rozza and B. Stamm, 2015 [244].
- (2) Reduced basis methods for partial differential equations: An introduction, by A. Quarteroni, A. Manzoni and N. Federico, 2015 [442].
- (3) Model reduction and approximation: Theory and algorithms, by P. Benner, A. Cohen, M. Ohlberger and K. Willcox, 2017 [54].

Papers and Reviews

- A survey of model reduction methods for parametric systems, by P. Benner, S. Gugercin and K. Willcox, SIAM Review, 2015 [53].
- (2) Model reduction using proper orthogonal decomposition, by S. Volkwein, Lecnare Notes, Institute of Mathematics and Scientific Computing, University of Graz, 2011 [542].
- (3) Nonlinear model reduction for dynamical systems using sparse sensor locations from learned libraries, by S. Sargsyan, S. L. Brunton and J. N. Kutz, *Physical Review E*, 2015 [462].
- (4) An online method for interpolating linear parametric reduced-order models, by D. Amsallem and C. Farhat, SIAM Journal of Scientific Computing, 2011 [10].

Glossary

Adjoint – For a finite-dimensional linear map (i.e., a matrix A), the adjoint A* is given by the complex conjugate transpose of the matrix. In the infinite-dimensional context, the adjoint A* of a linear operator A is defined so that $(Af, g) = (f, A^*g)$, where $(., \cdot)$ is an inner reduct.

Akaike information criterion (AIC) – An estimator of the relative quality of statistical models for a given set of data. Given a collection of models for the data, AIC estimates the quality of each model, relative to each of the other models. Thus, AIC provides a means for model selection.

Backpropagation (Backprop) – A method used for computing the gradient descent required for the training of neural networks. Based upon the chain rule, backprop exploits the compositional nature of NNs in order to frame an optimization problem for updating the weights of the network. It is commonly used to train deen neural networks.

Balanced input-output model – A model expressed in a coordinate system where the states are ordered hierarchically in terms of their joint controllability and observability. The controllability and observability (raminatis are equal and diagonal for such a system.

Bayesian information criterion (BIC) – An estimator of the relative quality of statistical models for a given set of data. Given a collection of models for the data, BIC estimates the quality of each model, relative to each of the other models. Thus, BIC provides a means for model selection.

Classification – A general process related to categorization, the process in which ideas and objects are recognized, differentiated, and understood. Classification is a common task for machine learning algorithms.

Closed-loop control – A control architecture where the actuation is informed by sensor data about the output of the system.

Clustering – A task of grouping a set of objects in such a way that objects in the same group (called a cluster) are more similar (in some sense) to each other than to those in other groups (clusters). It is a primary goal of exploratory data mining, and a common technique for statistical data analysis.

Coherent structure - A spatial mode that is correlated with the data from a system.

Compression – The process of reducing the size of a high-dimensional vector or array by approximating it as a sparse vector in a transformed basis. For example, MP3 and JPG compression use the Fourier basis or Wavelet basis to compress and/or or image signals. Compressed sensing – The process of reconstructing a high-dimensional vector signal from a random under sampling of the data using the fact that the high-dimensional signal is sparse in a known transform basis, such as the Fourier basis.

Control theory - The framework for modifying a dynamical system to conform to desired engineering specification through sensing and actuation.

Controllability - A system is controllable if it is possible to steer the system to any state with actuation. Degrees of controllability are determined by the controllability Gramian.

Convex optimization - An algorithmic frameworks for minimizing convex functions over convex sets.

Convolutional neural network (CNN) – A class of deep, feed-forward neural networks that is especially amenable to analyzing natural images. The convolution is typically a spatial filter which synthesizes local (neiphboring) spatial information.

Cross-validation - A model validation technique for assessing how the results of a statistical analysis will generalize to an independent (withheld) data set.

Data matrix – A matrix where each column vector is a snapshot of the state of a system at a particular instance in time. These snapshots may be sequential in time, or they may come from an ensemble of initial conditions or experiments.

Deep kearing – A class of machine learning algorithms that typically uses deep CNNs for feature extraction and transformation. Deep learning can leverage supervised (e.g., classification) and/or unsupervised (e.g., pattern analysis) algorithms, learning multiple levels of representations that correspond to different levels of abstraction; the levels form a hierarchy of concents.

DMD amplitude – The amplitude of a given DMD mode as expressed in the data. These amplitudes may be interpreted as the significance of a given DMD mode, similar to the power spectrum in the FFT.

DMD eigenvalue - Eigenvalues of the best-fit DMD operator A (see dynamic mode decomposition) representing an oscillation frequency and a growth or decay term.

DMD mode (also dynamic mode) — An eigenvector of the best-fit DMD operator A (see dynamic mode decomposition). These modes are spatially coherent and oscillate in time at a fixed frequency and a growth or decay rate.

Dynamic mode decomposition (DMD) – The leading eigendecomposition of a best-fit linear operator $A = XX^{+}$ that propagates the data matrix X into a future data matrix X. The eigenvectors of A are DMD modes and the corresponding eigenvalues determine the time dynamics of these modes.

Dynamical system – A mathematical model for the dynamic evolution of a system. Typically, a dynamical system is formulated in terms of ordinary differential equations on a state-space. The resulting equations may be linear or nonlinear and may also include the effect of actuation instats and represent outours as sensor measurements of the state.

Eigensystem realization algorithm (ERA) – A system identification technique that produces balanced input-output models of a system from impulse response data. ERA has been shown to produce equivalent models to balanced proper orthogonal decomposition and dynamic mode decomposition under some circumstances.

Emission - The measurement functions for a hidden Markov model.

Feedback control – Closed-loop control where sensors measure the downstream effect of actuators, so that information is fed back to the actuators. Feedback is essential for robust control where model uncertainty and instability may be counteracted with fast sensor feedback.

Feedforward control – Control where sensors measure the upstream disturbances to a system, so that information is fed forward to actuators to cancel disturbances proactively.

Fast Fourier transform (FFT) – A numerical algorithm to compute the discrete Fourier transform (DFT) in O(n log(n)) operations. The FFT has revolutionized modern computations, sirnal processing, compression, and data transmission.

Fourier transform – A change of basis used to represent a function in terms of an infinite series of sines and cosines.

Galerkin projection – A process by which governing partial differential equations are reduced into ordinary differential equations in terms of the dynamics of the coefficients of a set of orthogonal basis modes that are used to approximate the solution.

Gramian – The controllability (resp. observability) Gramian determines the degree to which a state is controllable (resp. observable) via actuation (resp. via estimation). The Gramian establishes an inner product on the state space.

Hidden Markov model (HMM) – A Markov model where there is a hidden state that is only observed through a set of measurements known as emissions.

Hilbert space – A generalized vector space with an inner product. When referred to in this text, a Hilbert space typically refers to an infinite-dimensional function space. These spaces are also complete metric spaces, providing a sufficient mathematical framework to enable calculus on functions.

Incoherent measurements — Measurements that have a small incer product with the basis vectors of a sparsifying transform. For instance, single pixel measurements (i.e., spatial delta function) are incoherent with respect to the spatial Fourier transform basis, since these single pixel measurements excite all frequencies and do not preferentially align with any single frequency.

Kalman filter – An estimator that reconstructs the full state of a dynamical system from measurements of a time-series of the sensor outputs and accuation inputs. A Kalman filter is itself a dynamical system that is constructed for observable systems to stably converge to the true state of the system. The Kalman filter is optimal for linear systems with Gaussian process and measurement noise of a known magnitude.

Koopman eigenfunction – An eigenfunction of the Koopman operator. These eigenfunctions correspond to measurements on the state-space of a dynamical system that form intrinsic coordinates. In other words, these intrinsic measurements will evolve linearly in time despite the underlying system being nonlinear. Koopman operator – An infinite-dimensional linear operator that propagates measurement functions from an infinite dimensional Hilbert space through a dynamical system.

Least squares regression - A regression technique where a best-fit line or vector is found by minimizing the sum of squares of the error between the model and the data.

Linear quadratic regulator (LQR) – An optimal proportional feedback controller for full-state feedback, which balances the objectives of regulating the state while not expending too much control energy. The proportional gain matrix is determined by solving an algebraic Riscati equation.

Linear system – A system where superposition of any two inputs results in the superposition of the two corresponding outputs. In other words, doubling the input doubles the output. Linear time-invariant dynamical systems are characterized by linear operators, which are represented as matrices.

Low rank – A property of a matrix where the number of linearly independent rows and columns is small compared with the size of the matrix. Generally, low-rank approximations are sought for larce data matrices.

Machine learning – A set of statistical tools and algorithms that are capable of extracting the dominant patterns in data. The data mining can be supervised or unsupervised, with the goal of clustering, classification and prediction.

Markov model – A probabilistic dynamical system where the state vector contains the probability that the system will be in a given state; thus, this state vector must always sum to unity. The dynamics are given by the Markov transition marks, which is constructed so that each row sums to unity.

Markov parameters - The output measurements of a dynamical system in response to an impulsive input.

Max pooling – A data down-sampling strategy whereby an input representation (image, hidden-layer output matrix, etc.) is reduced in dimensionality, thus allowing for assumptions to be made about features contained in the down-sampled sub-regions.

Model predictive control (MPC) – A form of optimal control that optimizes a control policy over a finite-time horizon, based on a model. The models used for MPC are typically linear and may be determined empirically via system identification.

Moore's law – The observation that transistor density, and hence processor speed, increases exponentially in time. Moore's law is commonly used to predict future computational power and the associated increase in the scale of problem that will be computationally feasible.

Multiscale – The property of having many scales in space and/or time. Many systems, such as turbulence, exhibit spatial and temporal scales that vary across many orders of magnitude.

Observability – A system is observable if it is possible to estimate any system state with a time-history of the available sensors. Degrees of observability are determined by the observability Gramian. Observable function – A function that measures some property of the state of a system. Observable functions are typically elements of a Hilbert space.

Optimization — Concratly use of algorithms that find the "best available" values of some objective function given a defined domain (or input), including a variety of different types of objective functions and different types of domains. Mathematically, optimization a nime to maximize or minimum real function by systematically, hoosing input values from within an allowed set and computing the value of the function. The generalization of optimization is there are not effect to the systematical type area or a defined immediant.

Overdetermined system – A system **Ax** = **b** where there are more equations than unknowns. Usually there is no exact solution **x** to an overdetermined system, unless the vector **b** is in the column space of **A**.

Pareto front – The allocation of resources from which it is impossible to reallocate so as to make any one individual or preference criterion better off without making at least one individual or preference criterion worse off.

Perron-Frobenius operator – The adjoint of the Koopman operator, the Perron-Frobenius operator is an infinite-dimensional operator that advances probability density functions through advantical system.

Power spectrum – The squared magnitude of each coefficient of a Fourier transform of a signal. The power corresponds to the amount of each frequency required to reconstruct a given signal.

Principal component - A spatially correlated mode in a given data set, often computed using the singular value decomposition of the data after the mean has been subtracted.

Principal components analysis (PCA) – A decomposition of a data matrix into h herarchy of principal component vectors that are ordered from most correlated to least correlated with the data. PCA is compated by taking the singular value decomposition of the data after subtracting the mean. In this case, each singular value represents the variance of the corresponding principal component (singular vector) in the data.

Proper orthogonal decomposition (POD) – The decomposition of data from a dynamical system into a hierarchical set of orthogonal modes, often using the singular value decomposition. When the data consists of velocity measurements of a system, such as an incompressible fluid, then the proper orthogonal decomposition orders modes in terms of the amount of energy these modes contain in the given data.

Pseudo-inverse – The pseudo-inverse generalizes the matrix inverse for non-square matrices, and is often used to compute the least-squares solution to a system of equations. The SVD is a common method to compute the pseudo-inverse: given the SVD $\mathbf{X} = \mathbf{U} \mathbf{\Sigma} \mathbf{V}^*$, the pseudo-inverse is $\mathbf{X}^T = \mathbf{V} \mathbf{\Sigma}^{-1} \mathbf{U}^*$.

Recurrent neural network (RNN) – A class of neural networks where connections between units form a directed graph along a sequence. This allows it to exhibit dynamic temporal behavior for a time sequence. Reduced-order model (ROM) – A model of a high-dimensional system in terms of a lowdimensional state. Typically, a reduced-order model balances accuracy with computational cost of the model.

Regression – A statistical model that represents an outcome variable in terms of indicators variables. Least-squares regression is a linear regression that finds the line of best fit to data; when generalized to higher dimensions and multi-linear regression, this generalizes to principal components regression. Nonlinear regression, downmaire regression, and functional or semantic regression are used in system identification, model reduction, and machine learning.

Restricted isometry property (RIP) – The property that a matrix acts like a unitary matrix, or an isometry map, on sparse vectors. In other words, the distance between any two sparse vectors is preserved if these vectors are mapped through a matrix that satisfies the restricted isometry property.

Robust control – A field of control that penalizes worst case scenario control outcomes, thus promoting controllers that are robust to uncertainties, disturbances, and unmodeled dynamics.

Robust statistics – Methods for producing good statistical estimates for data drawn from a wide range of probability distributions, especially for distributions that are not normal and where outliers compromise predictive canabilities.

Singular value decomposition (SVD) – Given a matrix X $\in \mathbb{C}^{n\times n}$, the SVD is forw by A = UEV where $0 \in \mathbb{C}^{n\times n} X \in \mathbb{C}^{n\times n}$, and $0 \in \mathbb{C}^{n\times n}$. Intensities U and V are unitary, so that UU = UU = 1 and VV = VV = 1. The matrix X has emission and the disquadies composition of the singular values in that ero offset from largest to matrix 1 more singular the singular values of the around the singular values of matrix 1 matrix singular values of the other poster of the singular values ordered by the singular values of the first r matrix - matrix matrix matrix matrix ordered by the singular values of the first r matrix - matrix

Snaphot – A single high-dimensional measurement of a system at a particular time. A number of snapshots collected at a sequence of times may be arranged as column vectors in a data matrix.

Sparse identification of nonlinear dynamics (SINDy) – A nonlinear system identification framework used to simultaneously identify the nonlinear structure and parameters of a dynamical system from data. Various sparse optimization techniques may be used to determine SINDy models.

Sparsity – A vector is sparse if most of its entries are zero or nearly zero. Sparsity refers to the observation that most data are sparse when represented as vectors in an appropriate transformed basis, such as Fourier or POD bases.

Spectrogram – A short-time Fourier transform computed on a moving window, which results in a time-frequency plot of which frequencies are active at a given time. The spectrogram is useful for characterizing nonperiodic signals, where the frequency content evolves over time, as in music. State space – The set of all possible system states. Often the state-space is a vector space, such as \mathbb{R}^n , although it may also be a smooth manifold \mathcal{M} .

Stochastic gradient descent – Also known as incremental gradient descent, it allows one to approximate the gradient with a single data point instead of all available data. At each step of the gradient descent, a randomly chosen data point is used to compute the gradient direction.

System identification – The process by which a model is constructed for a system from measurement data, possibly after perturbing the system.

Time delay coordinates – An augmented set of coordinates constructed by considering a measurement at the current time along with a number of times in the past at fixed intervals from the current time. Time delay coordinates are often useful in reconstructing attractor dynamics for systems that do not have enough measurements, as in the Takens embedding theorem.

Total least squares – A least-squares regression algorithm that minimizes the error on both the inputs and the outputs. Geometrically, this corresponds to finding the line that minimizes the sum of squares of the total distance to all points, rather than the sum of sources of the vertical distance to all points.

Uncertainty quantification (UQ) – The principled characterization and management of uncertainty in engineering systems. Uncertainty quantification often involves the application of powerful tools from probability and statistics to do ynamical systems.

Underdetermined system – A system Ax = b where there are fewer equations than unknowns. Generally the system has infinitely many solutions x unless b is not in the column space of A.

Unitary matrix — A matrix whose complex conjugate transpose is also its inverse. All eigenvalues of a unitary matrix are on the complex unit circle, and the action of a unitary matrix may be through of as a change of coordinates that preserves the Euclidean distance between any two vectors.

Wavelet – A generalized function, or family of functions, used to generalize the Fourier transform to approximate more complex and multiscale signals.

Bibliography

- Ralph Abraham, Jerrold E. Marsden, and Tudor Ratiu. Manifolds, Tensor Analysis, and Applications, volume 75 of Applied Mathematical Sciences. Springer-Verlag, 1988.
- [2] Ralph Abraham and Jerrold E. Marsden. Foundations of Mechanics, volume 36. Benjamin/Cummines Publishine Computer Reading. Massachusetts. 1978.
- [3] Mradul Agraval, Sandeep Vidyashankar, and Ke Huang. On-chip implementation of ECoG signal data decoding in brain-computer interface. In Mixed-Signal Testing Workshop (IMSTW), 2016 IEEE 21st International, pages 1–6. IEEE, 2016.
- [4] Rakesh Agrawal, Ramakrishnan Srikant, et al. Fast algorithms for mining association rules. In Proc. 20th int. conf. very large data bases, VLDB, volume 1215, pages 487–499, 1994.
- [5] Hyo-Sung Ahn, YangQuan Chen, and Kevin L. Moore. Iterative learning control: Brief survey and categorization. IEEE Transactions on Systems, Man, and Cybernetice, Part C (Applications and Review), 37(6):109–1121, 2007.
- [6] Hirotagu Akaike. Fitting autoregressive models for prediction. Annals of the Institute of Statistical Mathematics, 21(1):243–247, 1969.
- [7] Hirotagu Akaike. A new look at the statistical model identification. Automatic Control, IEEE Transactions on, 19(6):716–723, 1974.
- [8] W. O. Amrein and Anne-Marie Berthier. On support properties of Lp-functions and their Fourier transforms. Journal of Functional Analysis, 24(3):258–267, 1977.
- [9] David Amsallen, Jalien Cottial, and Charbel Farhat. On-demand cfd-based aeroelastic predictions using a database of reduced-order bases and models. In 47th AIAA Aerospace Sciences Meeting Including The New Horizone Forum and Aerospace Exposition, page 800, 2009.
- [10] David Amsallem and Charbel Fashat. An online method for interpolating linear parametric reduced-order models. SIAM Journal on Scientific Computing, 33(5):2169–2198, 2011.
- [11] David Armallem, Matthew J. Zahr, and Kyle Washahaugh. Fast local reduced basis updates for the efficient reduction of nonlinear systems with hyper-reduction. Advances in Computational Mathematics, 41(5):1187–1230, 2015.
- [12] Joakim Andén and Stéphane Mallat. Deep scattering spectrum. IEEE Transactions on Signal Processing, 62(16):4114–4128, 2014.
- [13] Edward Anderson, Zhaojim Bai, Christian Bischof, Sasan Blackford, James Demmel, Jack Dongarn, Jeremy Da Croz, Ame Gecenbarn, S. Hammerling, Alan McKenney, et al. LAPACK Urers' Guide, volume 9, SLMM, 1999.
- [14] Jeffrey L. Anderson. An ensemble adjustment Kalman filter for data assimilation. Monthly Weather Review, 129(12):2884–2903, 2001.
- [15] Chaus A. Andersson and Rasmus Bro. The n-way toolbox for matlab. Chemometrics and Intelligent Laboratory Systems, 52(1):1–4, 2000.
- [16] Marc Antonini, Michel Barlaud, Pierre Mathieu, and Ingrid Daubechies. Image coding using wavelet transform. IEEE Transactions on Image Processing, 1(2):205–220, 1992.
- [17] Athanasios C. Antoulas. Approximation of large-scale dynamical systems. SIAM, 2005.

- [18] Hassan Arbabi and Igor Mezić. Ergodic theory, dynamic mode decomposition and computation of spectral properties of the Koopman operator. SIAM Journal on Applied Dynamical Systems, 16 (4):206–2126, 2017.
- [19] Kartik B. Ariyur and Minoslav Krotić. Real-Time Optimization by Extremum-Seeking Control. Wiley, Hoboken, New Jersey, 2003.
- [20] Travis Askham and J. Nathan Kutz. Variable projection methods for an optimized dynamic mode decomposition. SIAM Journal on Applied Dynamical Systems, 17 (1):380–416, 2018.
- [21] Patricia Astrid. Fast reduced order modeling technique for large scale LTV systems. In American Control Conference, 2004. Proceedings of the 2004, volume 1, pages 762–767. IEEE: 2004.
- [22] Karl Johan Aström and Richard M. Murray. Feedback Systems: An Introduction for Scientists and Engineers. Princeton University Press, 2010.
- [23] M. F. A. Azeez and A. F. Vakakis. Proper orthogonal decomposition (POD) of a class of vibroimpact oscillations. *Journal of Sound and Vibration*, 240(5):859–889, 2001.
- [24] K. Bache and M. Lichman. UCI machine learning repository, 2013.
- [25] Brett W. Bader and Tamara G. Kolda. Efficient MATLAB computations with sparse and factored tensors. SIAM Journal on Scientific Computing, 30(1):205–231, December 2007.
- [26] S. Bagheri, L. Brandt, and D. S. Henningson. Input-output analysis, model reduction and control of the flat-plate boundary layer. Journal of Fluid Mechanics, 620:263–298, 2009.
- [27] S. Bagheri, J. Hoepffner, P. J. Schmid, and D. S. Henningson. Input-output analysis and control design applied to a linear model of spatially developing flows. *Applied Mechanics Reviews*, 612(2):00803-1.27, 2009.
- [28] Shervin Bagheri. Koopman-mode decomposition of the cylinder wake. Journal of Fluid Mechanics. 726:596–623, 2013.
- [29] Z. Bai, S. L. Brunten, B. W. Brunten, J. N. Katz, E. Kaiser, A. Spohn, and B. R. Noack. Datadriven methods in fluid dynamics: Sparse classification from experimental data. In *Invited Chapter for Whither Transhances and Big Data in the 21st Constary*, 2015.
- [30] Z. Bai, E. Kaiser, J. L. Proctor, J. N. Kutz, and S. L. Brunton. Dynamic mode decomposition for compressive system identification. arXiv preprint arXiv:1710.07737, 2017.
- [31] Zhe Bai, Thakshila Wimalajeewa, Zachary Berger, Guannan Wang, Mark Glauser, and Pramod K. Varshney. Low-dimensional approach for reconstruction of airfoil data via contressive sensitir. AMA Journal 53(4):1920-933. 2014.
- [32] Maciej J. Balajewicz, Earl H. Dowell, and Bernd R. Noack. Low-dimensional modelling of high-Reynolds-number shear flows incorporating constraints from the Navier-Stokes constitut. Journal of Fluid Mechanics. 79:2285–338. 2013.
- [33] Malhuudhaan Balasbeamanian, Stanialar Zahie, Christopher Bowd, Hilary W. Thompson, Peter Wolenski, S. Sithurama Iyengar, Bijaya B. Karki, and Linda M. Zangwill. A framework for detecting glancomatons progression in the optic nerve lead of an ope using proper orthogonal decomposition. *IEEE Transactions on Information Technology in Biomedicine*, 13(5):781–793, 2009.
- [34] B. Barnich and L. Giarré. Identification of Enear parameter varying models. International Journal of Robust and Nonlinear Control, 12:841–853, 2002.
- [35] Andrzej Banaszuk, Kartik B. Ariyar, Miroslav Krstić, and Clas A. Jacobson. An adaptive algorithm for control of combustion instability. Automatica, 40(11):1965–1972, 2004.
- [36] Andrzej Banasouk, Satish Narayanan, and Youping Zhang. Adaptive control of flow separation in a planar diffuser. AIAA paper, 617:2003, 2003.
- [37] Andrzej Banzsnak, Youping Zhang, and Clas A. Jacobson. Adaptive control of combustion instability using extremum-seeking. In American Control Conference, 2000. Proceedings of the 2000, volume 1. nases 146–422. IEEE, 2000.

- [38] S. P. Banks. Infinite-dimensional Carleman linearization, the Lie series and optimal control of non-linear partial differential equations. *International Journal of Systems Science*, 23(5):863– 675, 1992.
- [39] R. G. Baraniuk. Compressive sensing. IEEE Signal Processing Magazine, 24(4):118–120, 2007.
- [40] R. G. Baraniuk, V. Cevher, M. F. Duarte, and C. Hegde. Model-based compressive sensing. *IEEE Transactions on Information Theory*, 56(4):1982–2001, 2010.
- [41] Maxime Barrault, Yvon Maday, Ngoc Cuong Nguyen, and Anthony T. Patera. An empirical interpolation method: application to efficient reduced-basis discretization of partial differential equations. *Computer Review Methornation*, 3309;1667–672, 2004.
- [42] J. Basley, L. R. Pastur, N. Delprat, and F. Lusseyran. Space-time aspects of a three-dimensional multi-modulated open cavity flow. *Physics of Fluids (1994-present)*, 25(6):064105, 2013.
- [43] J. Basley, L. R. Pastar, F. Lusseyran, T. M. Faure, and N. Delprat. Experimental investigation of global structures in an incompressible cavity flow using time-resolved PIV. *Experiments in Fukitic*, 90(4):005–918, 2011.
- [44] Walter Baar and Volker Strassen. The complexity of partial derivatives. Theoretical Computer Science, 22(3):317–330, 1983.
- [45] J. F. Beaudoin, O. Cadot, J. L. Aider, and J. E. Wesfreid. Bluff-body drag reduction by extremum-seeking control. *Journal of Fluids and Structures*, 22:973–978, 2006.
- [46] Jean-Francois Beaudoin, Olivier Cadot, Jean-Luc Aider, and José-Eduardo Wesfreid. Drag reduction of a bluff body using adaptive control methods. *Physics of Fluids*, 18(8):085107, 2006.
- [47] R. Becker, R. King, R. Petz, and W. Nitsche. Adaptive closed-loop control on a high-lift configuration usine extremum seeking. AIAA Journal, 45(6):1382–92, 2007.
- [48] P. N. Belhameur, J. P. Hespanha, and D. J. Kriegman. Eigenfaces vs. Fisherfaces: Recognition using class specific linear projection. *IEEE Transactions on Pattern Analysis and Machine Intelligence (PMI)*, 1997;711–720. 1997.
- [49] G. Bellani. Experimental studies of complex flows through image-based techniques. 2011.
- [50] Brandt A. Belson, Jonathan H. Tu, and Clarence W. Rowley. Algorithm 945: modred—a parallelized model reduction library. ACM Transactions on Mathematical Software, 40(4):30, 2014.
- [51] Michael Benedicks. On Fourier transforms of functions supported on sets of finite Lebesgue measure. Journal of Mathematical Analysis and Applications. 106(1):180–183, 1985.
- [52] Yoshua Bengio, Pascal Lamblin, Dan Popovici, and Hugo Larochelle. Greedy layer-wise training of deep networks. In Advances in Neural Information Processing Systems, pages 1531–160, 2007.
- [53] P. Benner, S. Gugercin, and K Willcox. A survey of projection-based model reduction methods for parametric dynamical systems. SIAM Review, 57(4):483–531, 2015.
- [54] Peter Benner, Albert Cohen, Mario Ohlberger, and Karen Willcox. Model Reduction and Antroximation: Theory and Algorithms, volume 15. SIAM, 2017.
- [55] Peter Benner, Jing-Rebecca Li, and Thilo Penzl. Numerical solution of large-scale Lyapunov equations, Riccati equations, and linear-quadratic optimal control problems. *Numerical Linear Algobravini Replications*, 15(9):755–777, 2008.
- [56] E. Berger, M. Sastuba, D. Vogt, B. Jung, and H. B. Amor. Estimation of perturbations in robotic behavior using dynamic mode decomposition. *Journal of Advanced Robotics*, 29(5):331–342, 2015.
- [57] G. Berkooz, P. Holmes, and J. L. Lumley. The proper orthogonal decomposition in the analysis of turbulent flows. Annual Review of Fluid Mechanics, 25:539–575, 1993.
- [58] Gregory Beylkin, Renald Coifman, and Vladimir Rohhlin. Fast wavelet transforms and numerical algorithms i. Communications on Proce and Applied Mathematics, 44(2):141–183, 1991.
- [59] Stephen A. Billings. Nonlinear System Identification: NARMAX Methods in the Time, Frequency, and Spatio-Temporal Domains. John Wiley & Sens, 2013.
- [60] P. Binetti, K. B. Ariyur, M. Krstić, and F. Bernelli. Formation flight optimization using extremum seeking feedback. *Journal of Gaidance, Control, and Dynamics*, 26(1):132–142, 2003.
- [61] G. D. Birkhoff and B. O. Koopman. Recent contributions to the ergodic theory. Proceedings of the National Academy of Sciences, 18(3):279–282, 1932.
- [62] George D. Birkhoff. Proof of the ergodic theorem. Proceedings of the National Academy of Sciences, 17(12):656–660, 1931.
- [63] Christopher M. Bishop. Neural Networks for Pattern Recognition. Oxford university press, 1995.
- [64] Christopher M. Bishop. Pattern Recognition and Machine Learning. Springer New York, 2006.
- [65] D. A. Bistrian and I. M. Navon. An improved algorithm for the shallow water equations model reduction: Dynamic mode decomposition vs POD. International Journal for Numerical Methods in Flaids, 2015.
- [66] D. A. Bistrian and I. M. Navon. Randomized dynamic mode decomposition for non-intrusive reduced order modelling. *International Journal for Numerical Methods in Engineering*, 2016.
- [67] Paola Bondi, Giuseppe Casalino, and Lucia Gambardella. On the iterative learning control theory for robotic manipulators. *IEEE Journal on Robotics and Automation*, 4(1):14–22, 1988.
- [68] Josh Bongaed and Hod Lipson. Automated reverse engineering of nonlinear dynamical systems. Proceedings of the National Academy of Sciences, 104(24):9943–9948, 2007.
- [69] Jorre Luis Borres. The library of Babel. Collected Fictions, 1998.
- [70] Bernhard E. Boser, Isabelle M. Gayon, and Vladimir N. Vapnik. A training algorithm for optimal margin classifiers. In *Proceedings of the Fifth Annual Workshop on Computational Learning Theory*, pages 144–152. ACM, 1992.
- [71] H. Boulard and Y. Kamp. Autoassociative memory by multilayer perceptron and singular values decomposition. Biol Cybern, 59:291–294, 1989.
- [72] George E. P. Box, Gwilym M. Jenkins, Gregory C. Reinsel, and Greta M. Ljung. Time Series Analysis: Forecasting and Control. John Wiley & Sons, 2015.
- [73] Stephen Boyd, Leon O. Chua, and Charles A. Desoer. Analytical foundations of Volterra series. IMA Journal of Mathematical Control and Information, 1(3):243–282, 1984.
- [74] Stephen Boyd and Lieven Vandenberghe. Convex Pptimization. Cambridge University Press, 2009.
- [75] Antoneta Iuliana Brateu, Iulian Munteanu, Seddik Bacha, and Bertrand Raisen. Maximum power point tracking of grid-connected photoroltaic arrays by using extremum seeking control. CEAI, 10(4):3–12, 2008.
- [76] Leo Breiman. Better subset regression using the nonnegative garrote. Technometrics, 37(4):373–384, 1995.
- [77] Leo Breiman. Random forests. Machine Learning, 45(1):5-32, 2001.
- [78] Leo Breiman et al. Statistical modeling: The two cultures (with comments and a rejoinder by the author). Statistical Science, 16(3):199–231, 2001.
- [79] Leo Breiman, Jerome Friedman, Charles J. Stone, and Richard A. Olshen. Classification and Regression Trans. CRC press, 1984.

- [80] I. Bright, G. Lin, and J. N. Kutz. Compressive sensing and machine learning strategies for characterizing the flow around a cylinder with limited pressure measurements. *Physics of Fluids*, 25 (121102):1–15, 2013.
- [81] Ido Bright, Guang Lin, and J. Nathan Katz. Classification of spatio-temporal data via asynchronous sparse sampling: Application to flow around a cylinder. *Maltincale Modeling & Simulation*, 14 (2), 823–838, 2016.
- [82] Sergey Brin and Lawrence Page. The anatomy of a large-scale hypertextual web search engine. Computer Networks and ISDN Systems, 30(1-7):107–117, 1998.
- [83] Douglas Bristow, Marina Tharayil, Andrew G. Alleyne, et al. A survey of iterative learning control. Control Systems, IEEE, 26(3):96–114, 2006.
- [84] Rasmus Bro. Parafac. tutorial and applications. Chemometrics and Intelligent Laboratory Systems, 38(2):149–171, 1997.
- [85] Alexander Broad, T. D. Murphey, and Brenna Argall. Learning models for shared control of human-machine systems with unknown dynamics. *Robotics: Science and Systems Proceeding*, 2017.
- [86] Roger W. Brockett. Volterra series and geometric control theory. Automatica, 12(2):167–176, 1976.
- [87] David S. Broomhead and David Lowe. Radial basis functions, multi-variable functional interpolation and adaptive networks. Technical report, Royal Signah and Radar Establishment Malvern (United Kirnelem). 1988.
- [88] D. S. Broomhead and R. Jones. Time-series analysis. In Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciencer, volume 423, pages 103–121. The Royal Society. 1999.
- [89] B. W. Brunton, S. L. Brunton, J. L. Proctor, and J. N. Kutz. Sparse sensor placement optimization for classification. *SIAM Journal on Applied Mathematics*, 76(5):2009–2122, 2016.
- [90] B. W. Brunton, L. A. Johnson, J. G. Ojemann, and J. N. Kutz. Extracting spatial-temporal coherent patterns in large-scale neural recordings using dynamic mode decomposition. *Journal of Neuroscience: Methods*, 258, 1–15, 2016.
- [91] S. L. Brunton, B. W. Brunton, J. L. Proctor, E. Kaiser, and J. N. Kutz. Chaos as an intermittently forced linear system. *Nature Communications*, 8(19):1–9, 2017.
- [92] S. L. Brunton, B. W. Brunton, J. L. Proctor, and J. N Kutz. Koopman invariant subspaces and finite linear representations of nonlinear dynamical systems for control. *PLoS ONE*, 11(2):e0510717, 2016.
- [93] S. L. Brunton, X. Fu, and J. N. Kutz. Extremum-seeking control of a mode-locked laser. IEEE Journal of Quantum Electronics, 49(10):852–861, 2013.
- [94] S. L. Brunton and B. R. Noack. Closed-loop turbulence control: Progress and challenges. Applied Mechanics Reviews, 67:050801–1–050801–48, 2015.
- [95] S. L. Bennton, J. L. Proctor, and J. N. Kutz. Discovering governing equations from data by sparse identification of nonlinear dynamical systems. *Proceedings of the National Academy* of *Science*, 113 (15):3923–3937, 2016.
- [96] S. L. Brunton, J. L. Proctor, J. H. Tu, and J. N. Kutz. Compressed sensing and dynamic mode decomposition. *Journal of Computational Dynamics*, 2(2):165–191, 2015.
- [97] S. L. Brunton and C. W. Rowley. Maximum power point tracking for photovoltaic optimization using ripple-based extremum seeking control. *IEEE Transactions on Power Electronics*, 25(10):2531–2540, 2010.
- [98] S. L. Bounton, J. H. Tu, I. Bright, and J. N. Kutz. Compressive sensing and low-rank libraries for classification of bifurcation regimes in nonlinear dynamical systems. SIAM Journal on Applied Dynamical Systems, 13(4):1716–1732, 2014.

- [99] Steven L. Brunton, Xing Fu, and J. Nathan Kutz. Self-tuning fiber lasers. IEEE Journal of Selected Toxics in Output Electronics, 20(5), 2014.
- [100] Steven L Brunton, Joshua L Proctor, and J Nathan Kutz. Sparse identification of nonlinear dynamics with control (SINDYc). *IFAC NOLCOS*, 49(18):710–715, 2016.
- [101] D. Buche, Peter Stell, Rolf Dornberger, and Petros Kournoutsakos. Multiobjective evolutionary algorithm for the optimization of noisy combustion processes. Systems, Man, and Cyberneitez, Part C. Applications and Reviews, IEEE Transactions on, 32(4):660–737, 2002.
- [102] Marko Badišć and Igor Mezić. An approximate parametrization of the ergodic partition using time averaged observables. In Decision and Control, 2009 hold jointly with the 2009 28th Chinase Control Conference. CDC/CCC 2009. Proceedings of the 48th IEEE Conference on, pages 3162–3168. IEEE, 2009.
- [103] Marko Budišić and Igor Mezić. Geometry of the ergodic quotient reveals coherent structures in flows. Physica D: Nonlinear Physicano, 241(15):1255–1269, 2012.
- [104] Marko Badišić, Ryan Mohr, and Igor Mezić. Applied Koopmanism a). Chaos: An Interdisciplinary Journal of Nonlinear Science, 22(4):047510, 2012.
- [105] Kenneth P. Burnham and David R. Anderson. Model Selection and Multimodel Inference: A Practical Information-Theoretic Annovach. Springer Science & Business Media, 2003.
- [106] Peter A. Businger and Gene H. Golub. Algorithm 358: Singular value decomposition of a complex matrix [11, 4, 5]. Communications of the ACM, 12(10):564–565, 1969.
- [107] Eduardo F. Camacho and Carlos Bordons Alba. Model Predictive Control. Springer Science & Business Media, 2013.
- [108] Erik Cumbria, Guang-Bin Huang, Liyanaarachchi Lekamalage Chamara Kasun, Hongming Zhou, Chi Man Veng, Jaran Lin, Jiarping Yin, Zhiping Cai, Qiang Liu, Kuan Li, et al. Extreme learning machines [trends & controversios]. *IEEE Intelligent Systems*, 28(6):30–59, 2013.
- [109] E. J. Candès. Compressive sensing. Proceedings of the International Congress of Mathematics, 2006.
- [110] E. J. Candès, X. Li, Y. Ma, and J. Wright. Robust principal component analysis? Journal of the ACM, 58(3):11–1–11–37, 2011.
- [111] E. J. Candès, J. Romberg, and T. Tao. Robust uncertainty principles: exact signal reconstruction from highly incomplete frequency information. *IEEE Transactions on Information Theory*, 52(2):489–502, 2006.
- [112] E. J. Candès, J. Romberg, and T. Tao. Stable signal recovery from incomplete and inaccurate measurements. Communications in Pure and Applied Mathematics, 8(1207–1223), 59.
- [113] E. J. Candès and T. Tao. Near optimal signal recovery from random projections: Universal encoding strategies? IEEE Transactions on Information Theory, 52(12):5406–5425, 2006.
- [114] E. J. Candès and M. B. Wakin. An introduction to compressive sampling. IEEE Signal Processing Magazine, pages 21–30, 2008.
- [115] Emmanuel J. Candes and Terence Tao. Decoding by linear programming. Information Theory, IEEE Transactions on, 51(12):4203–4215, 2005.
- [116] Yanhua Cao, Jiang Zhu, Zhendong Luo, and IM Navon. Reduced-order modeling of the upper tropical pacific occan model using proper orthogonal decomposition. *Computers & Mathematics with Academatics*, vol. 84, 005, 2006.
- [117] Yanhua Cao, Jiang Zhu, I Michael Navon, and Zhendong Luo. A reduced-order approach to four-dimensional variational data assimilation using proper orthogonal decomposition. *Intermedisional Journal for Nanorical Methods in Fluids*, 53(10):1571–1583, 2007.
- [118] Kevin Carlberg, Matthew Barone, and Harbir Antil. Galerkin v. least-squares Petrov-Galerkin projection in nonlinear model reduction. *Journal of Computational Physics*, 330:693–734, 2017.

- [119] Kevin Carlberg, Charbel Boa-Moskh, and Charbel Farhat. Efficient non-linear model reduction via a least-squares Petrov-Galerkin projection and compressive tensor approximations. *International Journal for Numerical Methods in Engineering*, 86(2):155–181, 2011.
- [120] Kevin Carlberg, Charbel Farhat, Julien Cortial, and David Amsallem. The GNAT method for nonlinear model reduction: effective implementation and application to computational fluid dynamics and turbulent loss. Journal of Computational Physics, 242-263–467, 2013.
- [121] Torsten Carleman. Application de la théorie des équations intégrales linéaires aux systèmes d'équations différentielles non linéaires. Acta Mathematica, 59(1):63–87, 1932.
- [122] Torsten Carleman. Sur la théorie de l'équation intégrodifférentielle de boltzmann. Acta Mathematica, 60(1):91–146, 1933.
- [123] Torsten Carleman. Sur les systemes lineaires aux dérivées partielles du premier ordrea deux variables. CR Acad. Sci. Paris, 197:471–474, 1933.
- [124] J. Douglas Carroll and Jib-Jie Chang. Analysis of individual differences in multidimensional scaling via an N-way generalization of "Eckart-Young" decomposition. Psychometrika, 35:283–310, 1970.
- [125] Rick Chartrand. Numerical differentiation of noisy, nonsmooth data. ISBN Applied Mathematics, 2011, 2011.
- [126] Anindya Chatterjee. An introduction to the proper orthogonal decomposition. Current Science, 78(7):808–817, 2000.
- [127] S. Chaturantabat and D. C. Sorensen. Nonlinear model reduction via discrete empirical interpolation. SIAM Journal on Scientific Computing, 32(5):2737–2764, 2010.
- [128] K. K. Chen and C. W. Rowley. Normalized coprime robust stability and performance guarantees for reduced-order controllers. *IEEE Transactions on Automatic Control*, 58(4):1068– 1073, 2013.
- [129] K. K. Chen, J. H. Tu, and C. W. Rowley. Variants of dynamic mode decomposition: Boundary condition, Koopman, and Fourier analyses. *Journal of Nonlinear Science*, 22(6):887–915, 2012.
- [130] Yangquan Chen, Kevin L Moore, and Hyo-Sung Ahn. Iterative learning control. In Encyclorodia of the Sciences of Learnine, pages 1648–1652. Springer, 2012.
- [131] Steve Cherry. Singular value decomposition analysis and canonical correlation analysis. Journal of Climate, 9(9):2003–2009, 1996.
- [132] Kyunghyun Cho, Bart Van Memirhboer, Caglar Gulcehre, Dzmitry Bahdanau, Fethi Bougares, Holger Schwenk, and Yoshua Bengio. Learning phrase representations using rm encoderdecoder for statistical machine translation. arXiv provint arXiv:1406.1078, 2014.
- [133] J. Y. Choi, M. Krytić, K. B. Ariyur, and J. S. Lee. Extremum seeking control for discrete-time systems. *IEEE Transactions on Automatic Control*, 47(2):318–323, FEB 2002.
- [134] Youngsoo Choi, David Amsallem, and Charbel Farhat. Gradient-based constrained optimization using a database of linear reduced-order models. arXiv preprint arXiv:1506.07849, 2015.
- [135] T. Colonius and K. Taira. A fast immersed boundary method using a nullspace approach and multi-domain far-field boundary conditions. *Computer Methods in Applied Mechanics and Engineering*, 197:2131–2146, 2008.
- [136] James W. Cooley, Peter A. W. Lewis, and Peter D. Welch. Historical notes on the fast Fourier transform. Proceedings of the IEEE, 55(10):1675–1677, 1967.
- [137] James W. Cooley and John W Takey. An algorithm for the machine calculation of complex Fourier series. Mathematics of Commutation, 19(90):297–301, 1965.
- [138] Corinna Cortes and Vladimir Vapnik. Support-vector networks. Machine Learning, 20(3):273–297, 1995.
- [139] Mark C. Cross and Pierre C. Hohenberg. Pattern formation outside of equilibrium. Reviews of Modern Physics, 65(3):851, 1993.

- [140] James P. Crutchfield and Brace S. McNamara. Equations of motion from a data series. Complex Systems, 1:417–452, 1987.
- [141] Magnus Dam, Morten Bryns, Jens Jual Rasmussen, Volker Naulin, and Jan S. Hesthaven. Sparse identification of a produtor-prey system from simulation data of a convection model. *Physics of Planuses*, 24(2):022310, 2017.
- [142] Bryan C. Daniels and Ilya Nemenman. Automated adaptive inference of phenomenological dynamical models. Nature Communications, 6, 2015.
- [143] Bryan C. Daniels and Ilya Nementman. Efficient inference of parsimonious phenomenological models of cellular dynamics using s-systems and alternating regression. *PloS one*, 1033;e0119821, 2015.
- [144] Suddhasattwa Das and Dimitrios Giannakis. Delay-coordinate maps and the spectra of Koopman operators. arXiv preprint arXiv:1706.08544, 2017.
- [145] Ingrid Daubechies. The wavelet transform, time-frequency localization and signal analysis. *IEEE transactions on information theory*, 36(5):961–1005, 1990.
- [146] Lawrence Davis et al. Handbook of Genetic Algorithms, volume 115. Van Nostrand Reinhold New York, 1991.
- [147] Scott T. M. Dawson, Maziar S. Hemati, Matthew O. Williams, and Charence W. Rowley. Characterizing and correcting for the effect of sensor noise in the dynamic mode decomposition. *Eurorison tris Flaids*, 57(3):1–19, 2016.
- [148] Arthur P. Dempster, Nan M. Laird, and Donald B. Rabin. Maximum likelihood from incomplete data via the EM algorithm. Journal of the Royal Statistical Society. Series B (methodological), pages 1–38, 1977.
- [149] Santosh Devasia, Degang Chen, and Brad Paden. Nonlinear inversion-based output tracking. Automatic Control. IEEE Transactions on, 41(7):930–942, 1996.
- [150] D. L. Doncho. Compressed sensing. IEEE Transactions on Information Theory, 52(4):1289– 1306, 2006.
- [151] D. L. Donoho and M. Gavish. Code supplement to "The optimal hard threshold for singular values is 4/\sqrt{5}", http://purl.stanford.edu/vr705ar/9070.2014.
- [152] David Donoho. 50 years of data science. In Based on a Presentation at the Takey Centennial Workshop. NJ Princeton, 2015.
- [153] David L. Doroho, Iain M Johmstone, Jeffrey C Hoch, and Alan S Stern. Maximum entropy and the nearly black object. Journal of the Royal Statistical Society. Series B (Methodological), neares 41–81, 1992.
- [154] David L. Donoho and Jain M. Johnstone. Ideal spatial adaptation by wavelet shrinkage. Biometrika, 81(3):425–455, 1994.
- [155] J. C. Doyle. Guaranteed margins for LQG regulators. IEEE Transactions on Automatic Control, 23(4):756–757, 1978.
- [156] J. C. Doyle, K. Glover, P. P. Khargonekar, and B. A. Francis. State-space solutions to standard H₂ and H₅₀ control problems. *IEEE Transactions on Automatic Control*, 34(8):831–847, 1989.
- [157] John C. Doyle, Bruce A. Francis, and Allen R. Tannenbaum. Feedback Control Theory. Courier Corporation, 2013.
- [158] Petros Drincus and Michael W. Mahoney. A randomized algorithm for a tensor-based generalization of the singular value decomposition. *Linear Algebra and Its Applications*, 420(2-3):553–571. 2007.
- [159] Zhako Drmac and Serkan Gupercin. A new selection operator for the discrete empirical interpolation method—improved a priori error bound and extensions. SIAM Journal on Scientific Computing, 38(2):A631–A648, 2016.

- [160] Qiang Du and Max Gandwarger. Model reduction by proper orthogonal decomposition coupled with centroidal voronei tessellations (keynote). In ASME 2002 Joint US-European Fluide Engineering Division Conference, pages 1401–1406. American Society of Mechanical Engineers, 2002.
- [161] R. O. Duda, P. E. Hart, and D. G. Stork. Pattern Classification. Wiley-Interscience, 2000.
- [162] Jed A. Duersch and Ming Gu. Randomized QR with column pivoting. SIAM Journal on Scientific Computing, 39(4):C263–C291, 2017.
- [163] D. Duke, D. Honnery, and J. Soria. Experimental investigation of nonlinear instabilities in annular liquid sheets. Journal of Fluid Mechanics, 691:594–604, 2012.
- [164] Daniel Duke, Julio Soria, and Damon Honnery. An error analysis of the dynamic mode decomposition. *Europeiments in Fluids*, 52(2):529–542, 2012.
- [165] Geir, E. Dullerud and Fernando Paganini. A Course in Robust Control Theory: A Convex Annovach. Texts in Ambied Mathematics. Springer, Berlin, Heidelberg, 2000.
- [166] R. Dunne and B. J. McKeen. Dynamic stall on a pitching and surging airfoil. Experiments in Fluids, 56(8):1–15, 2015.
- [167] T. Dariez, S. L. Branton, and B. R. Noack. Machine Learning Control: Taming Nonlinear Dynamics and Tarbulence. Springer, 2016.
- [168] Thomas Daricz, Vladimir Parezanović, Laurent Cordier, Bernd R. Noack, Joël Delville, Jean-Paul Bornet, Marc Segond, and Markus Abel. Closed-loop turbulence control using muchine learning. arXiv revenit arXiv:1404.4559. 2014.
- [169] Thomas Dazier, Vladmir Parezanovic, Joan-Charles Larrentis, Carine Forement, Joël Delville, Jean-Paul Bonnet, Laurent Cordier, Bernd R Noack, Marc Segond, Markus Abel, Nicolan Gazine, Jean-Luc Aider, Carlie Raihuada, Christophe Cavier, Michel Stanisha, and Steven L Bratton. Closed-loop control of experimental shear flows using machine learning. AIAA Paper 2014-2219, 7th Free Content Conference, 2014.
- [170] C. Eckart and G. Yoang. The approximation of one matrix by another of lower rank. Psychometrika, 1(3):211–218, 1936.
- [171] Jens L. Eftang, Anthony T. Patera, and Einar M. Rønquist. An' hp' certified reduced basis method for parametrized elliptic partial differential equations. *SIAM Journal on Scientific Convention*, 32(6):137–2300, 2010.
- [172] Jeffrey L. Elman. Finding structure in time. Cognitive Science, 14(2):179-211, 1990.
- [173] Ulku Eren, Anna Prach, Başaran Bahader Koçer, Sala V. Raković, Erdal Kayacan, and Behyet Açıkmeye. Model peedictive control in aerospace systems: Current state and oppertunities. *Journal of Guidance, Control, and Dynamics*, 40(7):1541–1565, 2017.
- [174] N. B. Erichson, S. L. Brunton, and J. N. Kutz. Compressed dynamic mode decomposition for real-time object detection. *Journal of Real-Time Image Processing*, 2016.
- [175] N. B. Erichson, S. L. Brunton, and J. N. Kutz. Randomized dynamic mode decomposition. arXiv preprint arXiv:1702.02912, 2017.
- [176] N. B. Erichson, K. Manohar, S. L. Brunton, and J. N. Kutz. Randomized CP tensor decomposition. arXiv preprint arXiv:1703.09074.
- [177] N. B. Erichson, S. Voronin, S. L. Branton, and J. N. Kutz. Randomized matrix decompositions using R. arXiv preprint arXiv:1608.02148, 2016.
- [178] Trishan Esram, Jonathan W. Kimball, Philip T Krein, Patrick L. Chapman, and Pallab Midya. Dynamic maximum power point tracking of photoroltaic arrays using ripple correlation control. *Iner Transactions On Power Electronics*, 21(5):1282–1293, September 2006.
- [179] Richard Everson and Lawrence Sirovich. Karhunen–Loeve procedure for gappy data. JOSA A, 12(8):1657–1664, 1995.

- [180] N. Fabbiane, O. Semeraro, S. Bagheri, and D. S. Henningson. Adaptive and model-based control theory applied to convectively unstable flows. Appl. Mech. Rev., 66(6):060801–1–20, 2014.
- [181] B. F. Feeny. On proper orthogonal co-ordinates as indicators of modal activity. Journal of Sound and Vibration. 255(5):805–817, 2002.
- [182] Ronald A. Fisher. The use of multiple measurements in taxonomic problems. Annals of Human Genetics, 7(2):179–188, 1936.
- [183] Ronald Ayhner Fisher. On the mathematical foundations of theoretical statistics. Philosophical Transactions of the Royal Society of London. Series A, Containing Papers of a Mathematical or Prinzical Character. 222:309–368, 1922.
- [184] P. J. Fleming and R. C. Purshouse. Evolutionary algorithms in control systems engineering: a survey. Control Engineering Practice, 10:1223–1241, 2002.
- [185] Jean Baptiste Joseph Fourier. The Analytical Theory of Heat. The University Press, 1878.
- [186] Joseph Fourier. Theorie analytique de la chaleur, par M. Fourier. Chez Firmin Didot, père et fils, 1822.
- [187] J. E. Fowler. Compressive-projection principal component analysis. IEEE Transactions on Image Processing, 18(10):2230–2242, 2009.
- [188] Yoav Freund and Robert E. Schapire. A decision-theoretic generalization of on-line learning and an application to boosting. *Journal of Computer and System Sciences*, 55(1):119–139, 1997.
- [189] Jerome H. Friedman. Greedy function approximation: a gradient boosting machine. Annals of Statistics, pages 1189–1232, 2001.
- [190] Alan Fricze, Ravi Kannan, and Santosh Vempala. Fast Monte-Carlo algorithms for finding low-rank approximations. Journal of the ACM, 51(6):1025–1041, 2004.
- [191] Xing Fu, Steven L. Brunton, and J. Nathan Kutz. Classification of birefringence in modelocked fiber lasers using machine learning and sparse representation. *Optics Express*, 2277:8585–8597, 2014.
- [192] Koji Fukagata, Stefan Kern, Philippe Chatelain, Petros Kournoutsakos, and Nobuhide Kasagi. Evolutionary optimization of an anisotropic compliant surface for turbulent friction drag reduction. Journal of Zarehalence, 9(8):21–17, 2008.
- [193] F. Fukushima. A self-organizing neural network model for a mechanism of pattern recognition unaffected by shift in position. *Biological Cybernetic*, 36:193–202, 1980.
- [194] H. Gao, J. Lam, C. Wang, and Y. Wang. Delay-dependent output-feedback stabilisation of discrete-time systems with time-varying state delay. *IEE Proceedings-Control Theory and AmVications*, 151(6):691–698, 2004.
- [195] Carlos E. Garcia, David M. Prett, and Manfred Morari. Model predictive control: theory and practice: A survey. Automatica, 25(3):335–348, 1989.
- [196] Jorge L. Garriga and Masoud Scroush. Model predictive control tuning methods: A review. Industrial & Enviroering Chemistry Research, 49(8):3505–3515, 2010.
- [197] Carl-Friedrich Gauss. Theoria combinationis observationam erroribus minimis obsostiae, volume 1. Henricus Dieterich. 1823.
- [198] C. F. Gauss. Nachlass: Theoria interpolationis methodo nova tractata, volume werke. Königliche Gesellschaft der Wissenschaften, Göttingen, 1866.
- [199] Nicolas Gantier, J.L. Aider, Thomas Duricz, B. R. Noack, Maec Segond, and Markus Abel. Closed-loop separation control using machine learning. *Journal of Fluid Mechanics*, 770:442– 457, 2015.
- [200] M. Gavish and D. L. Donoho. The optimal hard threshold for singular values is 4/√3. IEEE Transactions on Information Theory, 60(8):5040–5053, 2014.

- [201] Mattia Gazzola, Oleg V. Vasilyev, and Petros Koumostvakos. Shape optimization for drag reduction in linked bodies using evolution strategies. *Computers & Structures*, 89(11):1224– 1231, 2011.
- [202] Greger Gelbert, Jonas P. Moeck, Christian O. Paschereit, and Rudibert King. Advanced algorithms for gradient estimation in one-and two-parameter extremum seeking controllers. *Journal of Process Control*, 2244:700–709, 2012.
- [203] A. S. Georghiades, P. N. Belhameur, and D. J. Kriegman. From few to many: Illumination come models for face recognition under variable lighting and pose. *IEEE Transactions on Pattern Analysis and Machine Intelligence (PMM)*, 23(6):643–660, 2001.
- [204] Jan J. Gerbrands. On the relationships between SVD, KLT and PCA. Pattern Recognition, 14(1):375–381, 1981.
- [205] A. C. Gilbert and P. Indyk. Sparse recovery using sparse matrices. Proceedings of the IEEE, 98(6):937–947, 2010.
- [206] A. C. Gilbert, J. Y. Park, and M. B. Wakin. Sketched SVD: Recovering spectral features from contrective measurements. ArXiv e-prints. 2012.
- [207] A. C. Gilbert, M. J. Strauss, and J. A. Tropp. A tutorial on fast Fourier sampling. IEEE Signal Processing Magazine, pages 57–66, 2008.
- [208] Bryan Glaz, Li Liu, and Peretz P Friedmann. Reduced-order nonlinear unsteady aerodynamic modeling using a surrogate-based recurrence framework. AIAA Journal, 48(10):2418–2429, 2010.
- [209] Philip J Goddard and Keith Glover. Controller approximation: approaches for preserving H_∞ performance. IEEE Transactions on Automatic Control, 43(7):858–871, 1998.
- [210] David E Goldberg. Genetic Algorithms. Pearson Education India, 2006.
- [211] G. H. Golub and C. Reinsch. Singular value decomposition and least squares solutions. *Numerical Mathematics*, 14:403–420, 1970.
- [212] Gene Golub and William Kahan. Calculating the singular values and pseudo-inverse of a matrix. Journal of the Society for Industrial & Applied Mathematics, Series B: Numerical Anardvin, 2021;05–224, 1985.
- [213] Gene Golab, Stephen Nash, and Charles Van Loan. A Hessenberg-Schur method for the problem ax + xb = c. IEEE Transactions on Automatic Control, 24(6):909–913, 1979.
- [214] Gene H. Golub and Charles F. Van Loan. Matrix Computations, volume 3. JHU Press, 2012.
- [215] R. Genzalez-Garcia, R. Rico-Martinez, and I. G. Kevrekidis. Identification of distributed parameter systems: A neural net based approach. Comp. & Chem. Eng., 22:S065–S968, 1998.
- [216] Ian Goodfellow, Yoshua Bengio, and Aaron Courville. Deep Learning. MIT Press, 2016. http://www.deeplearningbook.org.
- [217] Ian Goodfellow, Jean Ponget-Abadie, Mehdi Mizza, Bing Xu, David Wande-Farley, Sherjil Ozair, Aaron Courville, and Yoshna Bengio. Generative adversarial nets. In Advances in Neural Information Processing Systems, mares 2672–2680, 2014.
- [218] Michael Grant, Stephen Boyd, and Yinyu Ye. Cvx: Matlab software for disciplined convex programming, 2008.
- [219] Alex Graves, Greg Wayne, and Ivo Danihelka. Neural turing machines. arXiv preprint arXiv:1410.5401, 2014.
- [220] Anne Greenhaum. Iterative Methods for Solving Linear Systems. SIAM, 1997.
- [221] Mohinder S. Grewal. Kalman filtering. In International Encyclopedia of Statistical Science, pages 705–708. Springer, 2011.
- [222] M. Grilli, P. J. Schmid, S. Hickel, and N. A. Adams. Analysis of unsteady behaviour in shockwave tarbalent boundary layer interaction. *Journal of Fluid Mechanics*, 700:16–28, 2012.

- [223] Jacob Grosek and J. Nathan Kutz. Dynamic mode decomposition for real-time background/ foreground separation in video. arXiv preprint arXiv:1404.7592, 2014.
- [224] Ming Gu. Subspace iteration randomization and singular value problems. SIAM Journal on Scientific Computing, 37(3):1139–1173, 2015.
- [225] F. Gueniat, L. Mathelin, and L. Pastar. A dynamic mode decomposition approach for large and arbitrarily sampled systems. *Physics of Fluids*, 27(2):025113, 2015.
- [226] Fredrik Gastafssen, Fredrik Gurmarsson, Niclas Bergman, Urban Forssell, Jonas Jansson, Rickard Karlsson, and P-J. Needland. Particle filters for positioning, navigation, and tracking. *IEEE Transactions on ximal processing*, 50(2):425–437, 2002.
- [227] Alfred Haar. Zur theorie der orthogonalen funktionensysteme. Mathematische Annalen, 69(3):331–371, 1910.
- [228] N. Halko, P. G. Martinsson, and J. A. Tropp. Finding structure with randomness: Probabilistic algorithms for constructing approximate matrix decompositions. *SIAM Review*, 53(2):217– 288, 2011.
- [229] Nathan Halko, Per-Gurmar Martinsson, Yoel Shkolnisky, and Mark Tygert. An algorithm for the principal component analysis of large data sets. *SIAM Journal on Scientific Computing*, 33:2580–2594, 2011.
- [230] Nathan Halko, Per-Gannar Martinsson, and Joel A. Tropp. Finding structure with randomness: Probabilistic algorithms for constructing approximate matrix decompositions. SIAM Review, 53(2):217–288, 2011.
- [231] Sven J. Hammarling. Numerical solution of the stable, non-negative definite Lyapunov equation. IMA Journal of Numerical Analysis, 2(3):303–323, 1982.
- [232] Sangbo Han and Brian Feeny. Application of proper orthogonal decomposition to structural vibration analysis. Mechanical Systems and Signal Processing, 17(5):989–1001, 2003.
- [233] Nikolaus Hansen, André S. P. Niederberger, Lino Guzzella, and Petros Koamoutsakos. A method for handling uncertainty in evolutionary optimization with an application to feedback control of combustion. *IEEE Transactions on Evolutionary Computation*, 13(1):180–197, 2009.
- [234] David Harrison Jr. and Daniel L. Rubinfeld. Hedonic housing prices and the demand for clean air. Journal of Environmental Economics and Management, 5(1):81–102, 1978.
- [235] Richard A. Harshman. Foundations of the PARAFAC procedure: Models and conditions for an "explanatory" multi-modal factor analysis. *UCLA working papers in photorics*, 16:1–84, 1970. Available at www.spychology.uww.caffaculty.htmshmar/hyppfictd.pdf.
- [236] Trevor Hastie, Robert Tibshirani, Jerome Friedman, T Hastie, J. Friedman, and R. Tibshirani. The Elements of Statistical Learnine, volume 2. Springer, 2009.
- [237] Kaiming He, Xiangyu Zhang, Shaoqing Ren, and Jian Sun. Deep residual learning for image necognition. In Proceedings of the IEEE conference on computer vision and pattern recognition, pages 770–778, 2016.
- [238] M. T. Heath, A. J. Laub, C. C. Paige, and R. C. Ward. Computing the singular value decomposition of a product of two matrices. *SIAM Journal on Scientific and Statistical Computing*, 7(4):1147–1159, 1986.
- [239] Michael Heideman, Don Johnson, and C Barrus. Gauss and the history of the fast Fourier transform. IEEE ASSP Manazine, 1(4):14–21, 1984.
- [240] Werner Heisenberg. Über den anschaulichen inhalt der quantentheoretischen kinematik und mechanik. In Original Scientific Papers Wissenschaftliche Originalarbeiten, pages 478–504. Streituner. 1985.
- [241] Maziar S. Hernati, Clarence W. Rowley, Eric A. Deem, and Louis N. Cattafesta. De-biasing the dynamic mode decomposition for applied Koopman spectral analysis. *Theoretical and Computational Fluid Dynamics*, 31(4):491–368, 2017.

- [242] Maziar S. Hemati, Matthew O. Williams, and Charence W. Rowley. Dynamic mode decomposition for large and streaming datasets. *Physics of Fluids (1994-present)*, 26(11):111701, 2014.
- [243] Kyle K. Herrity, Anna C. Gilbert, and Joel A. Tropp. Sparse approximation via iterative thresholding. In Accusatic, Speech and Signal Processing 2006. ICASSP 2006 Proceedings. 2006 IEEE International Conference on, volume 3, pages III-III. IEEE, 2006.
- [244] Jan S. Hesthaven, Gianluigi Rozza, and Benjamin Stamm. Certified reduced basis methods for parametrized partial differential equations. SpringerBriefs in Mathematics, 2015.
- [245] Tony Hey, Stewart Tansley, Kristin M. Tolle, et al. The Fourth Paradigm: Data-Intensive Scien tific Discovery, volume 1. Microsoft research Redmond, WA, 2009.
- [246] Geoffrey E. Hinton and Terrence J. Sejnowski. Learning and relearning in boltzmann machines. Parallel Distributed Processing: Explorations in the Microstructure of Cognition, 1(282-317):2, 1986.
- [247] B. L. Ho and R. E. Kalman. Effective construction of linear state-variable models from input/sulpst data. In Proceedings of the 3rd Annual Allerion Conference on Circuit and System Theory. Immes 440–450, 1965.
- [248] Sepp Hochreiter and Jürgen Schmidhuber. Long shoet-term memory. Neural Computation, 9(8):1735–1780. 1997.
- [249] Arthur E. Hoerl and Robert W. Kennard. Ridge regression: Biased estimation for nonorthogonal problems. *Technometrics*, 12(1):55–67, 1970.
- [250] John H. Holland. Adaptation in natural and artificial systems: An Introductory Analysis with Applications to Biology. Control, and Artificial Intelligence. University of Michigan Press, 1975.
- [251] P. Holmes, J. L. Lumley, G. Berkosz, and C. W. Rowley. Turbulonce, Coherent Structures, Dynamical Systems and Symmetry. Cambridge University Press, Cambridge, 2nd paperback edition, 2012.
- [252] Philip Holmes and John Guckenheimer. Nonlinear oscillations, dynamical systems, and higherations of vector fields, volume 42 of Applied Mathematical Sciences. Springer-Verlag, Berlin, Heidelbere, 1983.
- [253] Eberhard Hopf. The partial differential equation u₁ + au₈ = µu₈₈. Communications on Pure and Applied Mathematics, 3(3):201–230, 1950.
- [254] John J. Hopfield. Neural networks and physical systems with emergent collective computational abilities. Proceedings of the National Academy of Sciences, 79(8):2554–2558, 1982.
- [255] Kurt Hornik, Maxwell Stinchcombe, and Halbert White. Multilayer feedforward networks are universal approximators. Neural Networks, 2(5):359–366, 1989.
- [256] H. Hotelling. Analysis of a complex of statistical variables into principal components. Journal of Educational Psychology, 24:417–441, September 1933.
- [257] H. Hotelling. Analysis of a complex of statistical variables into principal components. Journal of Educational Psychology, 24:498–520, October 1933.
- [258] C. Huang, W. E. Anderson, M. E. Harvazinski, and V. Sankaran. Analysis of self-excited combustion instabilities using decomposition techniques. In 51st AIAA Aerospace Sciences Montine, names 1–18, 2013.
- [259] D. H. Hubel and T. N. Wiesel. Receptive fields, binocular interaction and functional architecture in the cat's visual cortex. *Journal of Physiology*, 160:106–154, 1962.
- [260] Peter J. Huber. Robust statistics. In International Encyclopedia of Statistical Science, pages 1248–1251. Springer, 2011.
- [261] S. J. Illingworth, A. S. Morgans, and C. W. Rowley. Feedback control of flow resonances using balanced reduced-order models. *Journal of Sound and Vibration*, 330(8):1567–1581, 2010.

- [262] Eric Jacobsen and Richard Lyons. The sliding DFT. IEEE Signal Processing Magazine, 20(2):74–80, 2003.
- [263] Herbert Jaeger and Harald Haas. Harnessing nonlinearity: Predicting chaotic systems and saving energy in wireless communication. Science, 304(5667):78–80, 2004.
- [264] Gareth James, Daniela Witten, Trevor Hastie, and Robert Tibshirani. An Introduction to Statistical Learning. Springer, 2013.
- [265] M. C. Johnson, S. L. Brunton, N. B. Kundtz, and J. N. Kutz. Extremam-seeking control of a beam pattern of a reconfigurable holographic metamaterial antenna. *Journal of the Optical Society of America A*, 33(1):59–68, 2016.
- [266] Richard A. Johnson and Dean Wichern. Multivariate Analysis. Wiley Online Library, 2002.
- [267] W. B Johnson and J. Lindenstrauss. Extensions of Lipschitz mappings into a Hilbert space. Contemporary Mathematics, 26(189-206); 1, 1984.
- [268] Ian Jolliffe. Principal Component Analysis. Wiley Online Library, 2005.
- [269] Siddharth Joshi and Stephen Boyd. Sensor selection via convex optimization. IEEE Transactions on Signal Processing, 57(2):451–462, 2009.
- [270] Mihailo R. Jovanović, Peter J. Schmid, and Joseph W. Nichols. Sparsity-promoting dynamic mode decomposition. *Physics of Flaids*, 26(2):024103, 2014.
- [271] J. N. Juang. Applied System Identification. Prentice Hall PTR, Upper Saddle River, New Jersey, 1994.
- [272] J. N. Juang and R. S. Pappa. An eigensystem realization algorithm for modal parameter identification and model reduction. *Journal of Guidance, Control, and Dynamics*, 8(5):620– 627, 1985.
- [273] J. N. Juang, M. Phan, L. G. Horta, and R. W. Longman. Identification of observer/Kalman filter Markov parameters: Theory and experiments. Technical Memorandum 104069, NASA, 1991.
- [274] Simon J. Julier and Jeffrey K. Uhlmann. A new extension of the Kalman filter to nonlinear systems. In Int. symp. aerospace/deforce sensing, simul. and controls, volume 3, pages 182– 193. Obtando. FL. 1997.
- [275] Simon J. Julier and Jeffrey K. Uhlmann. Unscented filtering and nonlinear estimation. Proceedings of the IEEE, 92(3):401–422, 2004.
- [276] E. Kaiser, J. N. Kutz, and S. L. Brunton. Data-driven discovery of Koopman eigenfunctions for control. arXiv preprint arXiv:1707.01146, 2017.
- [277] Eurika Kaiser, J. Nathan Kutz, and Steven L. Bounton. Sparse identification of nonlinear dynamics for model predictive control in the low-data limit. To appear in *Proceedings of the Royal Society A., arXiv proprint arXiv:1711.05501*, 2017.
- [278] Eurika Kaiser, Bernd R. Noack, Laurent Cornier, Andreas Spohn, Marc Segned, Markus Abel, Guillaume Daviller, Jan Östh, Sinika Krajnović, and Robert K. Niven. Clauter-based reducedorder modelling of a mixing layer. Journal of Floid Mechanicz, 754:365–414, 2014.
- [279] Rudolph Emil Kalman. A new approach to linear filtering and prediction problems. Journal of Flaids Engineering, 82(1):35–45, 1960.
- [280] K. Karhunen. Über lineare methoden in der wahrscheinlichkeitsrechnung, vol. 37. Annales Academia:Scientiarum Fennicar, Ser. A. I, 1947.
- [281] Kévin Kasper, Lionel Mathelin, and Hisham Abou-Kandil. A machine learning approach for constrained sensor placement. In American Control Conference (ACC), 2015, pages 4479– 4484, IEEE. 2015.
- [282] A. K. Kassam and L. N. Trefethen. Fourth-order time-stepping for stiff PDEs. SIAM Journal on Scientific Commun. 26(4):1214–1233, 2005.
- [283] Michael Kearns and Leslie Valiant. Cryptographic limitations on learning boolean formulae and finite automata. Journal of the ACM (JACM), 41(1):67–95, 1994.

- [284] Arthony R. Kellems, Saifon Chatarantabat, Danny C. Sorensen, and Steven J. Cox. Morphologically accurate reduced order modeling of spiking neurons. *Journal of Computational Neuroscience*, 28(3):477–494, 2010.
- [285] J. Kepler. Tabulae Rudolphinae, quibus Astronomicae scientiae, temporum longinquitate collamae Restauratio continetur. Ultr: Jonas Saur. 1627.
- [286] Gaitan Kerschen and Jean-Claude Golinval. Physical interpretation of the proper orthogonal modes using the singular value decomposition. *Journal of Sound and Vibration*, 249(5):849– 865, 2002.
- [287] Gaetan Kerschen, Jean-claude Golinval, Alexander F. Vakakis, and Lawrence A. Bergman. The method of proper orthogonal decomposition for dynamical characterization and order reduction of mechanical systems: an overview. *Workforcer Dynamics*. 41(1-3):147–169, 2005.
- [288] I. G. Kevrekidis, C. W. Gear, J. M. Hyman, P. G. Kevrekidis, O. Runberg, and C. Theodoropoulos. Equation-free, coarse-grained multiscale computation: Enabling microscopic simulators to perform system-level analysis. *Communications in Mathematical Science*, 1(4):715– 762, 2003.
- [289] N. J. Killingsworth and M. Krste. PID tuning using extremum seeking: online, model-free performance optimization. *IEEE Control Systems Manazine*, February 70–79, 2006.
- [290] Diederik P. Kingma and Max Welling. Auto-encoding variational bayes. arXiv preprint arXiv:1312.6114, 2013.
- [291] M. Kirby and L. Sirovich. Application of the Karhunen-Loève procedure for the characterization of human faces. *IEEE Transactions on Pattern Analysis and Machine Intelligence* (PMM), 12(1):103–108, 1990.
- [292] V. C. Klema and A. J. Laub. The singular value decomposition: Its computation and some applications. IEEE Transactions on Automatic Control, 25(2):164–176, 1980.
- [293] Stefan Klus, Feldas Näske, Peter Koltai, Hao Wu, Ioannis Kevrekidis, Christof Schütte, and Frank Noć. Data-driven model neduction and transfer operator approximation. *Journal of Nonlinear Science*, pages 1–265, 2018.
- [294] Richard Koch. The 80/20 Principle. Nicholas Brealey Publishing, 1997.
- [295] Richard Koch. Living the 80/20 Way. Audio-Tech Basiness Book Summaries, Incorporated, 2006.
- [296] Richard Koch. The 80/20 Principle: The Secret to Achieving More with Less. Crown Business, 2011.
- [297] Richard Koch. The 80/20 Principle and 92 other Powerfal Laws of Nature: the Science of Success. Nicholas Breakev Publishing, 2013.
- [298] Teavo Kohonen. The self-organizing map. Neurocomputing, 21(1-3):1-6, 1998.
- [299] Tamara G. Kolda and Brett W. Bader. Tensor decompositions and applications. SIAM Review, 51(3):455–500. September 2009.
- [300] B. O. Koopman. Hamiltonian systems and transformation in Hilbert space. Proceedings of the National Academy of Sciences, 17(5):315–318, 1931.
- [301] B. O. Koopman and J.-v. Neumann. Dynamical systems of continuous spectra. Proceedings of the National Academy of Sciences of the United States of America. 18(3):255, 1932.
- [302] Milan Korda and Igor Mezić. Linear predictors for nonlinear dynamical systems: Koopman operator meets model predictive control. Automatica, 93:149–160, 2018.
- [303] Milan Konla and Igor Mezić. On convergence of extended dynamic mode decomposition to the Koopman operator. Journal of Nonlinear Science, 28(2): 687–710, 2018.
- [304] Petros Koumoutsakos, Jorathan Freund, and David Parckh. Evolution strategies for automatic ontimization of jet mixing. AIAA Journal, 39(5):967–969, 2001.
- [305] Krzysztof Kowalski, Willi-Hans Steeb, and K. Kowalksi. Nonlinear Dynamical Systems and Carleman Linearization. World Scientific. 1991.

- [306] John R. Koza. Genetic Programming: On the Programming of Computers by Means of Natural Selection. volume 1. MIT press. 1992.
- [307] John R. Koza, Forrest H. Bennett III, and Oscar Stiffelman. Genetic programming as a darwinian invention machine. In Genetic Programming, pages 93–108. Springer, 1999.
- [308] Boris Kramer, Piyush Grover, Petros Bouforanos, Moshacine Benosman, and Saleh Nabi. Sparse sensing and dnal based identification of flow regimes and bifurcations in complex flows. SIAM Journal on Aerolind Dramatical Systems, 162:11164–1106, 2017.
- [309] J. P. Krieger and M. Krstic. Extremum seeking based on atmospheric turbulence for aircraft endurance. Journal of Guidance, Control, and Dynamics, 34(6):1876–1885, 2011.
- [310] Alex Krizhevsky, Ilya Sutskever, and Geoffrey E. Hinton. Imagenet classification with deep convolutional neural networks. In Advances in Neural Information Processing Systems, pages 1007–1105, 2012.
- [311] M. Krstic, A. Krupadanam, and C. Jacobson. Self-tuning control of a nonlinear model of combustion instabilities. *IEEE Tr. Contr. Syst. Technol.*, 7(4):424–436, 1999.
- [312] M. Krstić and H. H. Wang. Stability of extremum seeking feedback for general nonlinear dynamic systems. Automatica, 36:595–601, 2000.
- [313] Tejas D. Kulkami, William F. Whitney, Pushmeet Kohli, and Josh Tenenbaum. Deep convolutional inverse graphics network. In Advances in Neural Information Processing Systems, name 2539–2547, 2015.
- [314] Solomon Kullback and Richard A Leibler. On information and sufficiency. The Annals of Mathematical Statistics, 22(1):79–86, 1951.
- [315] Karl Kunisch and Stefan Volkwein. Optimal snapshot location for computing pod basis functions. ESAIM: Mathematical Modelling and Numerical Analysis. 44(3):509–529, 2010.
- [316] J. N. Kutz. Data-Driven Modeling & Scientific Compatation: Methods for Complex Systems & Big Data: Oxford University Press, 2013.
- [317] J. N. Kutz, S. L. Branton, B. W. Branton, and J. L. Proctor. Dynamic Mode Decomposition: Data-Driven Modeling of Complex Systems, SIAM, 2016.
- [318] J. N. Kutz, X. Fu, and S. L. Brunton. Multi-resolution dynamic mode decomposition. SIAM Journal on Arabical Dynamical Systems, 15(2):713–735, 2016.
- [319] J. Nathan Kutz, Syuzama Sargsyan, and Steven L Brunton. Leveraging sparsity and compressive serving for reduced order modeling. In Model Robertion of Parametrized Systems, marcs 301–315. Servinez, 2017.
- [320] Sanjay Lall, Jerrold E. Marsden, and Sonja Glavaški. Empirical model reduction of controlled nonlinear systems. In *IEEC World Congress*, volume F, pages 473–478. International Federation of Automatic Control. 1999.
- [321] Sanjay Lall, Jerrold E. Marsden, and Sonja Glavaiki. A subspace approach to balanced transcation for model reduction of notilinear control systems. *International Journal of Robust* and Womlinear Control. 12(6):519–552, 2002.
- [322] Yueheng Lan and Igor Mezić. Linearization in the large of nonlinear systems and Koopman operator spectrum. *Physica D: Nonlinear Phenomena*, 242(1):42–53, 2013.
- [323] Alan Laub. A Schur method for solving algebraic Riccati equations. IEEE Transactions on automatic control, 24(6):913–921, 1979.
- [324] Yann LeCun, Yoshua Bengio, and Geoffrey Hinton. Deep learning. Nature, 521(7553):436, 2015.
- [325] Yann LeCun, Léon Bottou, Yoshua Bengio, and Patrick Haffner. Gradient-based learning applied to document recognition. Proceedings of the IEEE, 86(11):2278–2324, 1998.
- [326] Jay H. Lee. Model predictive control: Review of the three decades of development. International Journal of Control, Automation and Systems, 9(3):415–424, 2011.

- [327] K.C. Lee, J. Ho, and D. Kriegman. Acquiring linear subspaces for face recognition under variable lighting. *IEEE Transactions on Pattern Analysis and Machine Intelligence (PAMI)*, 21(5):584–698. 2005.
- [328] Adrien Marie Legendre. Nouvelles méthodes pour la détermination des orbites des comètes. F. Didot. 1805.
- [329] V. Lenaerts, Gaitan Kerschen, and Jean-Claude Golinval. Proper orthogonal decomposition for model updating of non-linear mechanical systems. *Mechanical Systems and Signal Processing*, 15(1):31–33, 2001.
- [330] Ian Lenz, Ross A. Knepper, and Ashutosh Saxena. Deepmpe: Learning deep latent features for model predictive control. In *Robotics: Science and Systems*, 2015.
- [331] R. Leyea, C. Alorso, I. Queinnec, A. Cid-Pastor, D. Lagrange, and L. Martinez-Salamero. MPPT of photovoltaic systems using extremum-seeking control. *Inver Transactions On Acomputer and Electronic Systems*, 42(1):240–258, January 2006.
- [332] Qianxiao Li, Felix Dietrich, Erik M. Bollt, and Ioannis G. Kevrekidis. Extended dynamic mode decomposition with dictionary learning: A data-driven adaptive spectral decomposition of the Koopman operator. *Chaos: An Interduciplinary Journal of Nonlinear Science*, 27(10):103111, 2017.
- [333] Y. C. Liang, H. P. Lee, S. P. Lim, W. Z. Lin, K. H. Lee, and C. G. Wu. Proper cethogenal decomposition and its applications- part i: Theory. *Journal of Sound and vibration*, 253(3):527–544, 2002.
- [334] Edo Liberty. Simple and deterministic matrix sketching. In Proceedings of the 19th ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, pages 581– 588. ACM, 2013.
- [335] Edo Liberty, Franco Woolfe, Per-Gumar Martinsson, Vladimir Rokhlin, and Mark Tygert. Randomized algorithms for the low-rank approximation of matrices. *Proceedings of the National Academy of Sciences*, 104:20167–20172, 2007.
- [336] Timothy P. Lillierap, Jonathan J. Hunt, Alexander Pritzel, Nicolas Heess, Tom Erez, Yaval Tasoa, David Silver, and Daan Wirestra. Continuous control with deep reinforcement learning. arXiv preprint arXiv:1500.02571, 2015.
- [337] Zhouchen Lin, Minming Chen, and Yi Ma. The augmented lagrange multiplier method for exact recovery of comuted low-rank matrices. arXiv preprint arXiv:1009.5055, 2010.
- [338] L. Ljung, System Identification: Theory for the User. Prentice Hall, 1999.
- [339] Stuart Lloyd. Least squares quantization in PCM. IEEE Transactions on Information Theory, 28(2):129–137, 1982.
- [340] M. Loeve. Probability Theory. Van Nostrand, Princeton, NJ, 1955.
- [341] J.-C. Loiseau and S. L. Brunton. Constrained sparse Galerkin regression. Journal of Fluid Mechanics, 838:42–67, 2018.
- [342] J.-C. Leiseau, B. R. Noack, and S. L. Brunton. Sparse reduced-order modeling: sensor-based dynamics to full-state estimation. Journal of Fluid Mechanics, 844:459–490, 2018.
- [343] Richard W. Longman. Iterative learning control and repetitive control for engineering practice. International Journal of Control, 73(10):930–954, 2000.
- [344] E. N. Lorenz. Empirical orthogonal functions and statistical weather prediction. Technical report. Massachusetts Institute of Technology. December 1956.
- [345] Edward N. Lorenz. Deterministic nonperiodic flow. Journal of the Atmospheric Sciences, 20(2):130-141, 1963.
- [346] D. M. Luchtenburg and C. W. Rowley. Model reduction using snapshot-based realizations. Bulletin of the American Physical Society, 56, 2011.
- [347] J. L. Lamley. Toward a turbulent constitutive relation. Journal of Fluid Mechanics, 41(02):413–434, 1970.

- [348] Bethany Lusch, Eric C. Chi, and J. Nathan Kutz. Shape constrained tensor decompositions using sparse representations in over-complete libraries. arXiv preprint arXiv:1608.04674, 2016.
- [349] Bethary Lusch, J. Nathan Kutz, and Steven L. Branten. Deep learning for universal linear embeddings of nonlinear dynamics. *Nature Communications. arXiv preprint* arXiv:1712.0907, 2018.
- [350] F. Lusseyran, F. Gueniat, J. Basley, C. L. Douay, L. R. Pastar, T. M. Faure, and P. J. Schmid. How coherent structures and frequency signature: application of the dynamic modes decomposition to open cavity flow. In *Journal of Physics: Conference Series*, volume 318, nare 04205. IOP Publishin: 2011.
- [351] Z. Ma, S. Ahuja, and C. W. Rowley. Reduced order models for control of fluids using the eigensystem realization algorithm. *Theor. Comput. Fluid Dyn.*, 25(1):233–247, 2011.
- [352] Wolfgang Maass, Thomas Natschläger, and Henry Markram. Real-time computing without stable states: A new framework for neural computation based on perturbations. *Neural Computation*, 14(11):2331–2580, 2002.
- [353] Alan Mackey, Hayden Schaeffer, and Stanley Osher. On the compressive spectral method. Multiscule Modeline & Simulation. 12(4):1800–1827. 2014.
- [354] Michael W. Mahoney. Randomized algorithms for matrices and data. Foundations and Trends in Machine Learning, 3:123–224, 2011.
- [355] Andrew J. Majda and John Harlim. Physics constrained nonlinear regression models for time series. Nonlinearity, 26(1):201, 2012.
- [356] Andrew J. Majda and Yoomang Lee. Conceptual dynamical models for turbulence. Proceedinex of the National Academy of Sciences. 111(18):6548-6553, 2014.
- [357] Stéphane Mallat. A Wavelet Tour of Signal Processing. Academic Press, 1999.
- [358] Stéphane Mallat. Understanding deep convolutional networks. Phil. Trans. R. Soc. A, 374(2065):20150203, 2016.
- [359] Stephane G. Mallat. A theory for multiresolution signal decomposition: the wavelet representation. *IEEE Transactions on Pattern Analysis and Machine Intelligence*, 11(7):874– 693, 1989.
- [360] John Mandel. Use of the singular value decomposition in regression analysis. The American Statistician, 36(1):15–24, 1982.
- [361] Niall M. Mangan, Steven L. Brunton, Joshna L. Proctor, and J. Nathan Kutz. Inferring biological networks by sparse identification of nonlinear dynamics. *IEEE Transactions on Molecular Biological and Multi-Scale Communications*, 21 (32–43, 2016).
- [362] Niall M. Mangan, J. Nathan Kutz, Steven L. Brunton, and Joshna L. Proctor. Model selection for dynamical systems via sparse repression and information criteria. *Proceedings of the Royal Society*, A, 473(204):1–16, 2017.
- [363] Jordan Mann and J. Nathan Kutz. Dynamic mode decomposition for financial trading strategies. Quantitative Finance, pages 1–13, 2016.
- [364] K. Manohar, S. L. Brunton, and J. N. Kutz. Environmental identification in flight using sparse approximation of wine strain. Journal of Fluids and Structures, 70:162–180, 2017.
- [365] K. Manohar, J. N. Kutz, and S. L. Brunton. Greedy Sensor and Actuator Placement Using Balanced Model Reduction. Balletin of the American Physical Society, 2018.
- [366] Krithika Manohar, Bingni W. Brunton, J. Nathan Kutz, and Steven L. Brunton. Data-driven sparse sensor placement. *IEEE Control Systems Magazine*, 38:63–86, 2018
- [367] Krithika Manchar, Eurika Kaiser, S. L. Bruaton, and J. N. Kutz. Optimized sampling for multiscale dynamics. SIAM Multiscale Modeling and Simulation. arXiv preprint arXiv:1712.05085, 2017.
- [368] Andreas Mandt, Luca Pasquali, Hao Wu, and Frank Noć. VAMPnets: Deep learning of molecular kinetics. Nature Communications, 9(1), 2018.

- [369] J. E. Marsden and T. S. Ratiu. Introduction to Mechanics and Symmetry. Springer-Verlag, 2nd edition, 1999.
- [370] Per-Gunnar Martinisson. Randomized methods for matrix computations and analysis of high dimensional data. arXiv preprint arXiv:1607.01649, 2016.
- [371] Per-Gannar Martinsson, Vladimir Rokhlin, and Mark Tygert. A randomized algorithm for the decomposition of matrices. Applied and Computational Harmonic Analysis, 30:47–68, 2011.
- [372] John L. Maryak, James C. Spall, and Bryan D. Heydon. Use of the Kalman filter for inference in state-space models with unknown noise distributions. *IEEE Transactions on Automatic Control*, 40(1):87–90, 2004.
- [373] L. Massa, R. Kumar, and P. Ravindran. Dynamic mode decomposition analysis of detonation waves. *Physics of Fluids (1994-present)*, 24(6):066101, 2012.
- [374] Lionel Mathelin, Kévin Kasper, and Hisham Abou-Kandil. Observable dictionary learning for high-dimensional statistical inference. Archiver of Computational Methods in Engineering, 25(1):103–120, 2018.
- [375] R. Maury, M. Keonig, L. Cattafesta, P. Jondan, and J. Delville. Extremum-seeking control of jet noise. Aeroacountics, 11(384):459–474, 2012.
- [376] I. Mezić. Spectral properties of dynamical systems, model reduction and decompositions. Nonlinear Drammicz, 41(1-3):309–325, 2005.
- [377] I. Mezić. Analysis of fluid flows via spectral properties of the Koopman operator. Ann. Rev. Fluid Mech., 45:357–378, 2013.
- [378] I. Mezić. Spectral Operator Methods in Dynamical Systems: Theory and Applications. Springer, 2017.
- [379] Igor Mezić and Andrzej Banaszuk. Comparison of systems with complex behavior. Physica D: Nonlinear Phenomena. 197(1):101–133, 2004.
- [380] Igor Mezić and Stephen Wiggins. A method for visualization of invariant sets of dynamical systems based on the ergodic partition. *Chaos: An Interdisciplinary Journal of Nonlinear Science*, 2011;213–218, 1999.
- [381] Michele Milano and Petros Koumoutsakos. Neural network modeling for near wall turbulent flow. Journal of Computational Physics, 182(1):1–26, 2002.
- [382] T. M. Mitchell. Machine Learning. McGraw Hill, 1997.
- [383] Y. Mizuno, D. Duke, C. Atkinson, and J. Soria. Investigation of wall-bounded turbulent flow using dynamic mode decomposition. In *Journal of Physics: Conference Series*, volume 318, neuro 02(304). IOP Publishims. 2011.
- [384] Volodymyr Mnih, Adria Puigdomenech Badia, Mehdi Mirza, Alex Graves, Timothy Lillicrap, Tim Harley, David Silver, and Koray Kavakusoplu. Asynchronous methods for deep reinforcement learning. In *International Conference on Machine Learning*, pages 1928–1937, 2016.
- [385] Volodymyr Mnih, Koray Kavukcuoght, David Silver, Alex Geaves, Ioannis Antonoglou, Daan Wierstra, and Martin Riedmiller. Playing atari with deep reinforcement learning. arXiv proprint arXiv:1312.5602, 2013.
- [386] Volodymyr Mnih, Koray Kavakcaogla, David Silver, Andrei A. Rusu, Joel Veness, Marc G. Bellemare, Alex Graves, Martin Riedmiller, Andreas K. Fidjeland, Georg Ostrovski et al. Human-level control through deep reinforcement learning. *Natures*, 180(7540):529, 2015.
- [387] J. P. Meeck, J.-F. Bourgoain, D. Darox, T. Schuller, and S. Candel. Tomographic reconstruction of heat release rate perturbations induced by helical modes in turbulent swird flames. *Experiments in Flaids*, 54(4):1–17, 2013.
- [388] B. C. Moore. Principal component analysis in linear systems: Controllability, observability, and model reduction. *IEEE Transactions on Automatic Control*, AC-26(1):17–32, 1981.

- [389] Calvin C. Moore. Ergodic theorem, ergodic theory, and statistical mechanics. Proceedings of the National Academy of Sciences, 112(7):1907–1911, 2015.
- [390] Kevin L. Moore. Iterative Learning Control for Deterministic Systems. Springer Science & Business Media, 2012.
- [391] Manfred Morari and Jay H. Lee. Model predictive control: past, present and future. Computers & Chemical Environment, 23(4):667–682, 1999.
- [392] T. W. Muld, G. Efrainsson, and D. S. Henningsen. Flow structures around a high-speed train extracted using proper orthogenal decomposition and dynamic mode decomposition. *Computers & Fluids*, 75:87–97, 2012.
- [393] T. W. Muld, G. Efraimsson, and D. S. Henningson. Mode decomposition on surface-mounted cube. Flow. Technience and Combustion, 88(3):279–310, 2012.
- [394] S. D. Müller, M Milano, and P. Koumoutsakos. Application of machine learning algorithms to flow modeling and optimization. Annual Research Briefe, pages 169–178, 1999.
- [395] Iulian Munteani, Antoneta Iuliana Brateu, and Emil Ceanga. Wind turbulence used as searching signal for MPPT in variable-speed wind energy conversion systems. *Renovable Energy*, 34(1):322–327, January 2009.
- [396] Kevin P. Murphy. Machine Learning: A Probabilistic Perspective. MIT press, 2012.
- [397] Vinod Nair and Geoffrey E. Hinton. Rectified linear units improve restricted boltzmann machines. In Proceedings of the 27th international conference on machine learning (ICML-10), pages 807–814, 2010.
- [398] D. Needell and J. A. Tropp. CoSaMP: iterative signal recovery from incomplete and inaccurate samples. Communications of the ACM, 53(12):93–100, 2010.
- [399] J. v Neumann. Proof of the quasi-ergodic hypothesis. Proceedings of the National Academy of Sciences, 18(1):70–82, 1932.
- [400] N. C. Nguyen, A. T. Patera, and J. Penaire. A best points interpolation method for efficient approximation of parametrized functions. *International Journal for Numerical Methods in Engineering*, 73(4):521–543, 2008.
- [401] Yves Nievergelt and Y. Nievergelt. Wavelets Made Easy, volume 174. Springer, 1999.
- [402] B. R. Noack, K. Afanasiev, M. Merzynski, G. Tadmor, and F. Thiele. A hierarchy of lowdimensional models for the transient and post-transient cylinder wake. *Journal of Fluid Mechanics*, 497:335–363, 2003.
- [403] B. R. Noack, T. Duriez, L. Cordier, M. Segond, M. Abel, S. L. Bunton, M. Mozzyński, J.-C. Laurenie, V. Parezanovic, and J.-P. Bonnet. Closed-loop turbulence control with machine learning methods. Bullytin Am. Phys. Soc., 58(18):M25009, p. 418, 2013.
- [404] Bernd R. Noack, Marek Morzynski, and Gilead Tadmor. Reduced-Order Modelling for Flow Control, volume 528. Springer Science & Business Media, 2011.
- [405] Frank Noé and Feliks Naske. A variational approach to modeling slow processes in stochastic dynamical systems. *Multircale Modeling & Simulation*, 11(2):635–655, 2013.
- [406] E. Noether. Invariante variationsprobleme nachr. d. könig. gesellsch. d. wiss. zu göttingen, math-phys. klasse 1918: 235-257. Explick Reprint: physica0503066, https://dx.doi.org/ 10.1080/04/11457106231446, name 57.1918.
- [407] Feliks Noske, Bettina G. Keller, Guillermo Pérez-Hernández, Antonia S. J. S. Mey, and Frank Noć. Variational approach to molecular kinetics. *Journal of Chemical Theory and Communicip*, 10(4):1739–1752, 2014.
- [408] Feliks Niske, Reinhold Schneider, Francesca Vitalini, and Frank Noć. Variational tensor approach for approximating the rare-event kinetics of macromolecular systems. J. Chem. Phys., 144(5):051105, 2016.
- [409] H. Nyquist. Certain topics in telegraph transmission theory. Transactions of the A. I. E. E., pages 617–644, FEB 1928.

- [410] Goro Obinata and Brian D. O. Anderson. Model reduction for control system design. Springer Science & Business Media. 2012.
- [411] C. M. Ostoich, D. J. Bodony, and P. H. Geubelle. Interaction of a Mach 2.25 turbulent boundary layer with a flattering panel using direct numerical simulation. *Physics of Fluids* (1994-person), 25(11):110806, 2013.
- [412] Samuel E. Otto and Clarence W. Rowley. Linearly-recurrent autoencoder networks for learning dynamics. arXiv preprint arXiv:1712.01378, 2017.
- [413] Y. Ou, C. Xu, E. Schuster, T. C. Luce, J. R. Ferron, M. L. Walker, and D. A. Humphreys. Design and simulation of extremum-seeking open-loop optimal control of current profile in the DIII-Dtokamak. *Planne Phritics and Controlled Torius*, 56:115001-1-115001-24, 2008.
- [414] Vidvuds Ozolini, Rongjie Lai, Russel Caffisch, and Stanley Osher. Compressed modes for variational problems in mathematics and physics. *Proceedings of the National Academy of Sciences*, 10(46):18868–18873, 2013.
- [415] C. Pan, D. Yu, and J. Wang. Dynamical mode decomposition of Garney flap wake flow. Theoretical and Applied Mechanics Letters, 1(1):012002, 2011.
- [416] V. Parezanovic, J.-C. Laurentie, T. Dariez, C. Foarment, J. Delville, J.-P. Bonnet, L. Cordier, B. R. Noack, M. Segond, M. Abel, T. Shnaparin, and S. L. Brenton. Mixing layer manipulation experiment – from periodic forcing to machine learning closed-loop control. *Journal Flow Turbulence and Combustion*, 94(1):155–173, 2015.
- [417] Vladimir Parezanović, Thomas Duriez, Laurent Cordier, Bernd R. Noack, Joël Delville, Jean-Paul Bornet, Marc Segond, Markus Abel, and Steven L. Brunton. Closed-Joop centrel of an experimental mixing layer using machine learning control. arXiv preprint arXiv:1408.3259, 2014.
- [418] K. Pearson. On lines and planes of closest fit to systems of points in space. Philosophical Magazine, 2(7–12):559–572, 1901.
- [419] B. Peherstorfer, D. Batnaru, K. Willcox, and H.-J. Bungartz. Localized discrete empirical interpolation method. SIAM Journal on Scientific Computing, 36(1):A168–A192, 2014.
- [420] Benjamin Peherstorfer and Karen Willcox. Detecting and adapting to parameter changes for reduced models of dynamic data-driven application systems. *Proceedia Computer Science*, 51:2552–2562, 2015.
- [421] Benjamin Peherstorfer and Karen Willcox. Dynamic data-driven reduced-order models. Computer Methods in Applied Mechanics and Envincering, 291:21–41, 2015.
- [422] Benjamin Peherstorfer and Karen Willcox. Online adaptive model reduction for nonlinear systems via low-eask updates. SIAM Journal on Scientific Computing, 37(4):A2123–A2150, 2015.
- [423] Sebastian Peitz and Stefan Klus. Koopman operator-based model reduction for switchedsystem control of PDEs. arXiv preprint arXiv:1710.06759, 2017.
- [424] S. D. Pendergrass, J. N. Kutz, and S. L. Brunton. Streaming GPU singular value and dynamic mode decompositions. arXiv preprint arXiv:1612.07875, 2016.
- [425] Roger Penrose. A generalized inverse for matrices. In Mathematical proceedings of the Cambridge philosophical society, volume 51, pages 406–413. Cambridge University Press, 1955.
- [426] Roger Pereose and John Arthur Tedd. On best approximate solutions of linear matrix equations. In Mathematical Proceedings of the Cambridge Philosophical Society, volume 52, pages 17–19. Cambridge Univ Press, 1966.
- [427] Lawrence Perko. Differential Equations and Dynamical Systems, volume 7. Springer Science & Business Media, 2013.

- [428] M. Phan, L. G. Horta, J. N. Juang, and R. W. Lengman. Linear system identification via an asymptotically stable observer. *Journal of Optimization Theory and Applications*, 79:59–86, 1903.
- [429] Mark A. Pinsky. Introduction to Fourier analysis and wavelets, volume 102. American Mathematical Soc., 2002.
- [430] T. Poggio. Deep learning: mathematics and neuroscience. Views & Reviews, McGovern Center for Brains, Minds and Machines, pages 1–7, 2016.
- [431] Philippe Poncet, Georgei-Henri Cottet, and Petros Kournoutsakos. Control of threedimensional wakes using evolution strategies. Compter Results Mecanique, 333(1):65–77, 2005.
- [432] Christopher Poultney, Sumit Chopen, Yann L Can, et al. Efficient learning of sparse representations with an energy-based model. In Advances in Neural Information Processing systems, pages 1137–1144, 2007.
- [433] J. L. Procher, S. L. Brunton, B. W. Brunton, and J. N. Kutz. Exploiting sparsity and equationfree architectures in complex systems (invited review). *The European Physical Journal Special Towics*, 223(13):2665–2684, 2014.
- [434] Joshua L. Proctor, Steven L. Brunton, and J. Nathan Kutz. Dynamic mode decomposition with control. SIAM Journal on Applied Dynamical Systems, 15(1):142–161, 2016.
- [435] Joshua L. Proctor and Philip A. Eckhoff. Discovering dynamic patterns from infectious disease data usine dynamic mode decomposition. *International Health*, 7(2):139–145, 2015.
- [436] H. Qi and S. M. Hughes. Invariance of principal components under low-dimensional random projection of the data. IEEE International Conference on Image Processing, October 2012.
- [437] Shie Qian and Dapang Chen. Discrete Gabor transform. IEEE Transactions on Signal Processing, 41(7):2429–2438, 1993.
- [438] S. J. Qin and T. A. Badgwell. An overview of industrial model predictive control technology. In AIChE Symposium Series, volume 93, pages 232–256, 1997.
- [439] S. Joe Qin and Thomas A. Badgwell. A survey of industrial model predictive control technology. Control Engineering Practice, 11(7):733–764, 2003.
- [440] Qing Qu, Ju Sun, and John Wright. Finding a sparse vector in a subspace: Linear sparsity using alternating directions. In Advances in Neural Information Processing Systems 27, pages 3401–3409, 2014.
- [441] A. Quarteroni and G. Rozza. Reduced Order Methods for Modeling and Computational Reduction, volume 9 of MS&A – Modeling, Simulation & Appplications, Springer, 2013.
- [442] Alfio Quarteroni, Andrea Manzoni, and Federico Negri. Reduced Basis Methods for Partial Differential Equations: An Introduction, volume 92, Springer, 2015.
- [443] J. Ross Ouinlan. Induction of decision trees. Machine Learning, 1(1):81-106, 1986.
- [444] J Ross Quinlan. C4. 5: Programs for Machine Learning. Elsevier, 2014.
- [445] Maziar Raissi and George Em Karniadakis. Hidden physics models: Machine learning of nonlinear partial differential equations. *Journal of Computational Physics*, 357:125–141, 2018.
- [446] C. Radhakrishna Rao. The utilization of multiple measurements in problems of biological classification. Journal of the Royal Statistical Society. Series B (Methodological), 10(2):159– 203. 1948.
- [447] James B. Rawlings. Tutorial overview of model predictive control. IEEE Control Systems, 20(3):38–52, 2000.
- [448] Soumya Raychaudhuri, Joshua M. Stuart, and Russ B. Altman. Principal components analysis to summarize microarray experiments: application to sporulation time series. In Parific Symposium on Biocomputing. Pacific Symposium on Biocomputing, page 455. NIH Public Access. 2000.

- [449] Rolf H. Reichle, Dennis B. McLaughlin, and Dara Entekhabi. Hydrologic data assimilation with the ensemble Kalman filter. *Monthly Woather Review*, 130(1):103–114, 2002.
- [450] B. Ren, P. Frihauf, R. J. Rafac, and M. Krstić. Laser pulse shaping via extremum seeking. Control Engineering Practice, 20:674–683, 2012.
- [451] Branko Ristic, Sanjeev Aralampalam, and Neil James Gordon. Beyond the Kalman Filter: Particle Filters for Tracking Amfications. Attach house, 2004.
- [452] Anthony John Roberts. Model Emergent Dynamics in Complex Systems. SIAM, 2014.
- [453] Charles A. Rohde. Generalized inverses of partitioned matrices. Journal of the Society for Industrial & Applied Mathematics, 13(4):1033–1035, 1965.
- [454] Vladimir Rokhlin, Arthur Szham, and Mark Tygert. A randomized algorithm for principal component analysis. SIAM Journal on Matrix Analysis and Applicatione, 31:1100–1124, 2009.
- [455] C. W. Rowley, T. Colonius, and R. M. Murray. Model reduction for compressible flows using POD and Galerkin projection. *Physica D*, 189:115–129, 2004.
- [456] C. W. Rowley, I. Mezić, S. Bagheri, P. Schlatter, and D. S. Henningson. Spectral analysis of nonlinear flows. J. Fluid Meck., 645:115–127, 2009.
- [457] Clarence W. Rowley and Jerrold E. Massden. Reconstruction equations and the Karhunen– Lobve expansion fee systems with symmetry. *Physica D: Nonlinear Phenomena*, 142(1):1–19, 2000.
- [458] C.W. Rowley. Model reduction for fluids using balanced proper orthogonal decomposition. Int. J. Bifurcation and Chaon, 15(3):997–1013, 2005.
- [459] S. Roy, J.-C. Hua, W. Barnhill, G. H. Gumaratne, and J. R. Gord. Deconvolution of reactingflow dynamics using proper orthogonal and dynamic mode decompositions. *Physical Review E*, 91(1):013001, 2015.
- [460] S. H. Rudy, S. L. Brunton, J. L. Proctor, and J. N. Kutz. Data-driven discovery of partial differential equations. *Science Advances*, 3(e)1602614), 2017.
- [461] Themistuklis P. Sapsis and Andrew J. Majda. Statistically accurate low-order models for uncertainty quantification in turbulent dynamical systems. *Proceedings of the National* Academy of Sciences, 10(3):41:3706–3710, 2013.
- [462] S. Sargsyan, S. L. Brunton, and J. N. Kutz. Nonlinear model reduction for dynamical systems using sparse sensor locations from learned libraries. *Physical Review E*, 920033304), 2015.
- [463] S. Sarkar, S. Ganguly, A. Dalal, P. Saha, and S. Chakraborty. Mixed convective flow stability of nanoflisik past a square cylinder by dynamic mode decomposition. *International Journal of Hot Flow*, 44:624–634, 2013.
- [464] Tamas Sarlos. Improved approximation algorithms for large matrices via random projections. In Foundations of Computer Science, 47th Annual IEEE Symposium on, pages 143–152, 2006.
- [465] T. Sayadi, P. J. Schmid, J. W. Nichols, and P. Moin. Reduced-order representation of near-wall structures in the late transitional boundary layer. *Journal of Fluid Mechanics*, 748:278–301, 2014.
- [466] Taraneh Sayadi and Peter J. Schmid. Parallel data-driven decomposition algorithm for large-scale datasets: with application to transitional boundary layers. *Theoretical and Communicational Fluid Dynamics*, nares 1–14, 2016.
- [467] H. Schaeffer, R. Caflisch, C. D. Hauck, and S. Osher. Sparse dynamics for partial differential equations. Proceedings of the National Academy of Sciences USA, 110(17):8634–6639, 2013.
- [468] Hayden Schaeffer. Learning partial differential equations via data discovery and sparse optimization. In Proc. R. Soc. A, volume 473, page 20160446. The Royal Society, 2017.
- [469] Hayden Schaeffer and Scott G. McCalla. Sparse model selection via integral terms. Physical Review E, 96(2):023302, 2017.
- [470] Robert E. Schapire. The strength of weak learnability. Machine learning, 5(2):197-227, 1990.

- [471] M. Schlegel, B. R. Noack, and G. Tadmor. Low-dimensional Galerkin models and control of transitional channel flow. Technical Report 01/2004, Hermann-Föttinger-Institut f
 ür Strömangsmechanki, Technische Universität Berlin, Germany, 2004.
- [472] P. J. Schmid. Dynamic mode decomposition for numerical and experimental data. J. Fluid. Mech, 656:5–28, 2010.
- [473] P. J. Schmid, L. Li, M. P. Juniper, and O. Past. Applications of the dynamic mode decomposition. Theoretical and Computational Fluid Dynamics, 25(1-4):249–259, 2011.
- [474] P. J. Schmid and J. Sesterhens. Dynamic mode decomposition of numerical and experimental data. In 61st Annual Meeting of the APS Division of Fluid Dynamics. American Physical Society, November 2008.
- [475] P. J. Schmid, D. Violato, and F. Scarano. Decomposition of time-resolved tomographic PIV. Experiments in Flaids, 52:1567–1579, 2012.
- [476] E. Schmidt. Zur theorie der linearen und nichtlinearen integralgleichungen. i teil. entwicklung willkärlichen funktionen nach system vorgeschriebener. Math. Ann., 3:433–476, 1907.
- [477] Michael Schmidt and Hod Lipson. Distilling free-form natural laws from experimental data. Science, 324(5923):81–85, 2009.
- [478] Michael D. Schmidt, Ravishnakar R. Vallabbajosyala, Jerry W. Jenkins, Jonathan E. Hoed, Abhishek S. Soni, John P. Wikowo, and HoL Lipson. Automated refinement and inference of analytical models for metabolic networks. *Physical Biology*, 85:5055011, 2011.
- [479] Bernhard Schölkopf and Alexander J. Smola. Learning with Kernels: Support Vector Machines, Regularization, Optimization, and beyond. MIT press, 2002.
- [480] Gideon Schwarz et al. Estimating the dimension of a model. The Annals of Statistics, 6(2):461–464, 1978.
- [481] A. Seena and H. J. Sang. Dynamic mode decomposition of turbulent cavity flows for selfsustained oscillations. International Journal of Heat and Fluid Flow, 32(6):1098–1110, 2011.
- [482] Ervin Sejdić, Iger Djarović, and Jin Jiang. Time-frequency feature representation using energy concentration: An overview of recent advances. *Digital Signal Processing*, 19(1):153– 183, 2009.
- [483] O. Semeraro, G. Bellani, and F. Landell. Analysis of time-resolved PIV measurements of a confined turbulent jet using POD and Keopman modes. *Experiments in Fluids*, 53(5):1203– 1220, 2012.
- [484] Onofrio Semeraro, Francois Lusseyran, Luc Pastur, and Peter Jordan. Qualitative dynamics of wavepackets in turbulent jets. *Physical Review Fluids*, 2(9):004605, 2017.
- [485] Gil Shabat, Yaniv Shmueli, Yariv Aizenbud, and Amir Averbuch. Randomized LU decomposition. Availed and Commutational Harmonic Analysis, 2016.
- [486] C. E. Shannon. A mathematical theory of communication. Bell System Technical Journal, 27(3):379–423, 1948.
- [487] Ati S. Sharma, Igor Mezić, and Beverley J. McKeon. Correspondence between Koopman mode decomposition, resolvent mode decomposition, and invariant solutions of the Navier-Stokes equations. *Physical Review Fluids*, 1(5):302402, 2016.
- [488] Eli Shlizerman, Edwin Ding, Matthew O. Williams, and J. Nathan Kutz. The proper orthogonal decomposition for dimensionality reduction in mode-locked lasers and optical systems, *International Journal of Orbits*, 2012. 2011.
- [489] Valeria Simoncini. A new iterative method for solving large-scale Lyapanov matrix equations. SIAM Journal on Scientific Computing, 29(3):1268–1288, 2007.
- [490] L. Sirovich. Turbulence and the dynamics of coherent structures, parts I-III. Q. Appl. Math., XLV(3):561–590, 1987.
- [491] L. Sirovich and M. Kirby. A low-dimensional procedure for the characterization of human faces. Journal of the Optical Society of America A, 4(3):519–524, 1987.

- [492] S. Skogestad and I. Postlethwaite. Multivariable Feedback Control. Wiley, Chichester, 1996.
- [493] Paul Smolensky. Information processing in dynamical systems: Foundations of harmony theory. Technical report, Colorado Univ at Boulder Dept of Computer Science, 1986.
- [494] Giovanni Solari, Laigi Carasale, and Federica Tubino. Proper orthogonal decomposition in wind engineering. part I: A state-of-the-art and some prospects. Wind and Structures, 10(2):153–176. 2007.
- [495] G. Song, F. Alizard, J.-C. Robinet, and X. Gloerfelt. Global and Koopman modes analysis of sound generation in mixing layers. *Physics of Fluids (1994-present)*, 25(12):124101, 2013.
- [496] Danny C. Sorensen and Yunkai Zhou. Direct methods for matrix Sylvester and Lyapunov equations. Journal of Applied Mathematics, 2003(6):277–303, 2003.
- [497] Mariia Sorokina, Stylianos Sygletos, and Sergei Turitsyn. Sparse identification for nonlinear optical communication systems: SINO method. Optics Express, 24(26):30433–30443, 2016.
- [498] James C. Spall. The Kantorovich inequality for error analysis of the Kalman filter with unknown noise distributions. Automatica. 31(10):1513–1517, 1995.
- [499] Nitish Srivastava, Geoffrey Hinton, Alex Krizhevsky, Ilya Suskever, and Ruslan Salakhutdinov. Dropout: A simple way to prevent neural networks from overfitting. *The Journal of Machine Learning Research*, 15(1):1929–1958, 2014.
- [500] W-H. Steeb and F. Wilhelm. Non-linear autonomous systems of differential equations and Carleman linearization procedure. *Journal of Mathematical Analysis and Applications*, 77(2):691–611, 1990.
- [501] Robert F. Stenzel. Optimal Control and Estimation. Courier Corporation, 2012.
- [502] Gilbert W. Stewart. On the early history of the singular value decomposition. SIAM Review, 35(4):551–566, 1993.
- [503] George Sughara, Robert May, Hao Ye, Chih-hao Hsieh, Ethan Deyle, Michael Fogarty, and Stephan March. Detecting causality in complex ecosystems. *Science*, 338(6106):496–500, 2012.
- [504] A. Sarana. Koopman operator based observer synthesis for control-affine nonlinear systems. In 55th IEEE Conference on Decision and Control (CDC, pages 6492–6499, 2016.
- [505] Amit Surana and Andrzej Banaszak. Linear observer synthesis for nonlinear systems using Koorman operator framework. IEAC-PapersOnLine, 49(18):716–723, 2016.
- [506] Yoshihiko Susuki and Igor Mezić. A prony approximation of Koopman mode decomposition. In Decision and Control (CDC), 2015 IEEE 54th Annual Conference on, pages 7022–7027. IEEE, 2015.
- [507] Richard S. Sutton and Andrew G. Barto. Reinforcement Learning: An Introduction, volume 1. MIT press Cambridge, 1998.
- [508] Adam Svenkeson, Bryan Glaz, Samuel Stanton, and Bruce J. West. Spectral decomposition of nonlinear systems with memory. Phys. Rev. E, 93:022211, Feb 2016.
- [509] S. A. Svoronos, D. Papageorgiou, and C. Tsiligiannis. Discretization of nonlinear control systems via the Carleman linearization. *Chemical Engineering Science*, 40(19):3263–3267, 1994.
- [510] D. L. Swets and J. Weng. Using discriminant eigenfeatures for image retrieval. IEEE Transactions on Pattern Analysis and Machine Intelligence (PAMI), 18(8):831–836, 1996.
- [511] K. Taira and T. Colonius. The immersed boundary method: a projection approach. Journal of Computational Physics, 225(2):2118–2137, 2007.
- [512] Naoya Takrishi, Yoshinobu Kawahara, Yasua Tahei, and Takehisa Yaini. Bayesian dynamic mode decomposition. *Trenty-Sixth International Joint Conference on Artificial Intelligence*, 2017.

- [513] Naoya Takeishi, Yoshinobu Kawahara, and Takehisa Yaini. Learning Koopman invariant subspaces for dynamic mode decomposition. In Advances in Neural Information Processing Systems, pages 1130–1140, 2017.
- [514] Naoya Takeishi, Yoshinobu Kawahara, and Takehisa Yairi. Subspace dynamic mode decomposition for stochastic Koreman analysis. Physical Review. E 96:3:033310, 2017.
- [515] F. Takens. Detecting strange attractors in turbulence. Lecture Notes in Mathematics, 898:366– 381, 1981.
- [516] Z. Q. Tang and N. Jiang. Dynamic mode decomposition of hairpin vortices generated by a henisphere protoberance. Science China Physics, Mechanics and Astronomy, 55(1):118–124, 2012.
- [517] Roy Taylor, J. Nathan Kutz, Kyle Morgan, and Brian Nelson. Dynamic mode decomposition for plasma diagnostics and validation. *Review of Scientific Instruments*, 39 (5):053501, 2018.
- [518] Robert Tibshirani. Regression shrinkage and selection via the lasso. Journal of the Royal Statistical Society. Series B (Methodolorical), pages 267–288, 1996.
- [519] Zhou Ting and Jiang Hui. Eeg signal processing based on proper orthogonal decomposition. In Audio, Language and Image Processing (ICALIP), 2012 International Conference on, pages 636–640. IEEE. 2012.
- [520] Santosh Tirunagani, Norman Puh, Kevin Wells, Miroslaw Boher, Isky Gorden, and David Windridge. Movement correction in DCE-MRI through windowed and reconstruction dynamic mode decomposition. Metchine Vision and Applications, 282–419: 393–407, 2017.
- [521] Christopher Torrence and Gilbert P. Compo. A practical guide to wavelet analysis. Bulletin of the American Meteorological Society, 79(1):61–78, 1998.
- [522] Giang Tran and Rachel Ward. Exact receivery of chaotic systems from highly corrupted data. Multiscule Modeling & Simulation, 15 (3):1108–1129, 2017.
- [523] Lloyd N. Trefethen. Spectral methods in MATLAB. SIAM, 2000.
- [524] Lloyd N. Trefethen and David Bau III. Numerical linear alrebra, volume 50. SIAM, 1997.
- [525] J. A. Tropp. Greed is good: Algorithmic results for sparse approximation. IEEE Transactions on Information Theory, 50(10):2231–2242, 2004.
- [526] J. A. Tropp. Recovery of short, complex linear combinations via l₁ minimization. IEEE Transactions on Information Theory, 51(4):1568–1570, 2005.
- [527] J. A. Tropp. Algorithms for simultaneous sparse approximation. part ii: Convex relaxation. Simul Processive, 86(3):589–602, 2006.
- [528] J. A. Tropp. Just relax: Convex programming methods for identifying sparse signals in noise. IEEE Transactions on Information Theory, 52(3):1030–1051, 2006.
- [529] J. A. Tropp and A. C. Gilbert. Signal recovery from random measurements via orthogonal matching pursuit. IEEE Transactions on Information Theory, 53(12):4655–4666, 2007.
- [530] J. A. Tropp, A. C. Gilbert, and M. J. Straass. Algorithms for simultaneous sparse approximation. nart i: Greedy pursuit. Simul Processing, 86(3):572–588, 2006.
- [531] J. A. Tropp, J. N. Laska, M. F. Duarte, J. K. Romberg, and R. G. Baraniak. Beyond Nyquist: Efficient sampling of sparse bandlimited signals. *IEEE Transactions on Information Theory*, 56(1):520–544, 2010.
- [532] Joel A. Tropp, Alp Yurtsever, Madeleine Udell, and Volkan Cevher. Randomized single-view algorithms for low-rank matrix approximation. arXiv preprint arXiv:1609.00048, 2016.
- [533] J. H. Tu and C. W. Rowley. An improved algorithm for balanced POD through an analytic treatment of impulse response tails. J. Comp. Phys., 231(16):5317–5333, 2012.
- [534] J. H. Tu, C. W. Rowley, E. Aram, and R. Mittal. Koopman spectral analysis of separated flow over a finite-thickness flat plate with elliptical leading edge. AIAA Paper 2011, 2864, 2011.
- [535] J. H. Tu, C. W. Rowley, D. M. Luchtenburg, S. L. Brunten, and J. N. Kutz. On dynamic mode decomposition: theory and applications. J. Comp. Dyn., 1(2):391–421, 2014.

- [536] Jonathan H. Tu, Clarence W. Rowley, J. Nathan Kutz, and Jessica K. Shang. Spectral analysis of fluid flows using sub-Nyquist-rate PIV data. Experiments in Fluids, 55(9):1–13, 2014.
- [537] M. Tark and A. Pentland. Eigenfaces for recognition. Journal of Cognitive Neuroscience, 3(1):71–86, 1991.
- [538] Rudolph Van Der Merwe. Sigma-point Kalman Filters for Probabilistic Inference in Dynamic State-Source Models. 2004.
- [539] Charles Van Loan. Computational Frameworks for the Fast Fourier Transform. SIAM, 1992.
- [540] Daniele Venturi and George Em Karniadakis. Gappy data and reconstruction procedures for flow root a cylinder. Journal of Fluid Mechanics, 519:315–336, 2004.
- [541] Pascal Vincent, Hugo Larochelle, Yoshna Bengio, and Pierre-Antoine Manzagol. Extracting and composing robust features with denoising autoencoders. In Proceedings of the 25th international conference on Machine Learning, pages 1006–1103. ACM, 2008.
- [542] Stefan Volkwein. Model reduction using proper orthogonal decomposition. Lecture Notex, Intrinste of Mathematics and Scientific Computing, University of Graz, see http://www.awigoz.ard/instruct/volkeni/POD.pdf, 1025, 2011.
- [543] Stefan Volkwein. Proper orthogonal decomposition: Theory and reduced-order modelling. Lecture Notes, University of Konstanz, 4:4, 2013.
- [544] Sergey Veronin and Per-Gunnar Martinsson. RSVDPACK: Subcostines for computing partial singular value decompositions via randomized sampling on single core, multi core, and GPU architecture. arXiv preprint arXiv:1502.03366. 2015.
- [545] Avery Wang et al. An industrial strength audio search algorithm. In Invir, volume 2003, pages 7–13. Washinston, DC, 2003.
- [546] H. H. Wang, M. Krstić, and G. Bastin. Optimizing bioreactors by extremum seeking. Adaptive Control and Signal Processing, 13(8):651–669, 1999.
- [547] H. H. Wang, S. Yeung, and M. Kestić. Experimental application of extremum seeking on an axial-flow compressor. *IEEE Transactions on Control Systems Technology*, 8(2):300–309, 2000.
- [548] W. X. Wang, R. Yang, Y. C. Lai, V. Kovanis, and C. Grebogi. Predicting catastrophes in nonlinear dynamical systems by compressive sensing. *Physical Review Letters*, 106:154101– 1–154101–4, 2011.
- [549] Zhu Wang, Imran Akhtar, Jeff Borggaard, and Traian Ilicscu. Proper orthogonal decomposition closure models for turbulent flows: a numerical comparison. Computer Methods in Americal Netchastics and Environment, 237:10–28. 2012.
- [550] Christoph Wehmeyer and Frank Noć. Time-lagged autoencoders: Deep learning of slow collective variables for molecular kinetics. *The Journal of Chemical Physics*, 148(24):241703, 2018.
- [551] Greg Welch and Gary Bishop. An introduction to the Kalman filter, 1995.
- [552] Peter Whitle. Hypothesis Testing in Time Series Analysis, volume 4. Almqvist & Wiksells, 1951.
- [553] O. Wiederhold, R. King, B. R. Noack, L. Neuhaus, L. Neise, W. an Enghard, and M. Swobeda. Extensions of extremum-seeking control to improve the aerodynamic performance of axial turbemachines. In 39th AIAA Physic Dynamics Conference, pages 1–19, San Antonio, TX, USA, 2009. AIAA-Paper 002407.
- [554] K. Willcox and J. Peraire. Balanced model reduction via the proper orthogonal decomposition. AIAA Journal. 40(11):2323–2330, 2002.
- [555] Karen Willcox. Unsteady flow sensing and estimation via the gappy proper orthogonal decomposition. Computers & Flaidr, 35(2):208–226, 2006.

- [556] Matthew O. Williams, Ioannis G. Kevrekidis, and Chaence W. Rowley. A data-driven approximation of the Keopman operator: extending dynamic mode decomposition. *Journal of Nonlinear Science*, 6: 1307–1346, 2015.
- [557] Matthew O. Williams, Clarence W. Rowley, and Ioannis G. Kevrelidis. A kernel approach to data-driven Koopman spectral analysis. *Journal of Computational Dynamics*, 2(2):247–265, 2015.
- [558] Daniela M. Witten and Robert Tibshirani. Perulized classification using Fisher's linear discriminant. Journal of the Royal Statistical Society: Series B (Statistical Methodology), 73(5):753–772, 2011.
- [559] Franco Woolfe, Edo Liberty, Vladimir Rokhlin, and Mark Tygert. A fast randomized algorithm for the approximation of matrices. *Journal of Applied and Computational Harmonic Analysis*, 25:335–366, 2008.
- [560] J. Wright, A. Yang, A. Ganesh, S. Sastry, and Y. Ma. Robust face recognition via sparse representation. *IEEE Transactions on Pattern Analysis and Machine Intelligence (PAMI)*, 31(2):210–227. 2009.
- [561] C. F. Jeff Wu. On the convergence properties of the EM algorithm. The Annals of Statistics, pages 95–103, 1983.
- [562] Xindeng Wu, Yupin Kumar, J. Ross Qainlan, Joydeep Ghosh, Qiang Yang, Hiroshi Motoda, Geoffrey J. McLachlan, Angus Ng, Bing Liu, S. Yu Philip et al. Top 10 algorithms in data mining. *Knowlodge and Information Systems*, 14(1):1–37, 2008.
- [563] Hao Ye, Richard J. Beamish, Sarah M. Glaser, Sue C. H. Grant, Chih-hao Hsieh, Laura J. Richards, Jon T. Schratte, and George Sughara. Equation-free mechanistic ecosystem forecasting using empirical dynamic modeling. *Proceedings of the National Academy of Sciences*, 10(3):181505–1876, 2015.
- [564] Enoch Yeung, Soumya Kunda, and Nathan Hodas. Learning deep neural network representations for Kooprum operators of nonlinear dynamical systems. arXiv preprint arXiv:1708.06850, 2017.
- [565] B. Yildirim, C. Chryssostomidis, and G. E. Kamiadakis. Efficient sensor placement for ocean measurements usine low-dimensional concerts. *Ocean Modelline*, 27(3):160–173, 2009.
- [566] Xiaoming Yuan and Junfeng Yang. Sparse and low-rank matrix decomposition via alternating direction methods. preprint, 12, 2009.
- [567] M. D. Zeiler, D. Krishnan, G. W. Taylor, and R. Fergus. Deconvolutional networks. In IEEE Computer Vision and Pattern Recognition (CVPR), pages 2528–2535, 2010.
- [568] C. Zhang and R. Ordó nez. Numerical optimization-based extremum seeking control with application to ABS design. *IEEE Transactions on Automatic Control*, 52(3):454–467, 2007.
- [569] Hao Zhang, Clarence W. Rowley, Eric A. Deem, and Louis N. Cattafesta. Online dynamic mode decomposition for time-varying systems. arXiv preprint arXiv:1707.02876, 2017.
- [570] T. Zhang, G. Kahn, S. Levine, and P. Abbeel. Learning deep control policies for autonomous aerial vehicles with MPC-guided policy search. In *IEEE Robotics and Automation (ICRA)*, neuro 528–515, 2016.
- [571] Weiwei Zhang, Bobin Wang, Zhengyin Ye, and Jingge Quan. Efficient method for limit cycle flatter analysis based on nonlinear aerodynamic reduced-order models. AIAA Journal, 50(5):1019–1028, 2012.
- [572] Sanjo Zlobec. An explicit form of the moore-penrose inverse of an arbitrary complex matrix. SIAM Review, 12(1):132–134, 1970.
- [573] Hui Zou and Trevor Hastie. Regularization and variable selection via the elastic net. Journal of the Royal Statistical Society: Series B (Statistical Methodology), 67(2):301–320, 2005.

Index

N-way arrays, 41, 42 k-fold cross-validation, 145, 146, 148 £-means, 164, 165, 167, 191 k-nearest neighbors (kNN), 192 Akaike information criterion (AIC), 151, 175, 256 alternating descent method, 128 backpropagation (backprop), 201, 202, 204, 205, 208 Bayesian information criterion (BIC), 151, 152 boosting, 190, 192, 193 classification. 25, 28, 72, 103, 105, 106, 190-192. 195, 196, 198, 200, 202, 203, 208, 215, 242, 367 classification trees, 185, 189 clustering, 24, 25, 36, 186, 191, 192 compression, 8, 10, 59, 75, 77, 80, 84, 85, 88, 95, 241 control. 7, 47, 56, 89, 155, 233, 242, 257, 266, 292, control theory, 89, 236, 276, 319, 322, 346, 349 controllability, 47, 287-291, 324-326, 329, 333 convex optimization, 92, 131, 250, 294, 357 consolutional neural network (CNN), 212 convolutional neural network (CNN), 191 cost function, 278, 294, 295, 297, 304, 318, 345, 353, 359, 367 CP tensor decomposition, 43-45 173, 179, 202, 243, 268 data arrays, 41 data labely 154 160-167 206 214 215 data matrix, 31, 35, 42, 44, 107, 167, 199, 237, 244, 348, 393 deep convolutional neural network (DCNN), 212-214 deep learning, 195, 196, 212 dendrogram, 168, 169 dimensionality reduction, 4, 7, 176, 235, 237, 238,

discrete empirical interpolation method (DEIM), 403. 423 discrete wavelet transform, 75, 77, 81 DMD eigenvalue, 238, 240 DMD mode (also dynamic mode), 236. 237. 242 dynamic mode decomposition (DMD), 235, 241 dynamical system, 8, 47, 216, 229-236, 247, 251. 257, 276, 277, 297, 322, 404 circosystem realization algorithm (ERA), 243, 270. ensemble learning, 189, 190, 192, 193 expectation-maximization algorithm (EM), 172-174. entreman socking control (ESC), 362, 364, 365, 369 fast Fourier transform, 3, 47, 56, 59, 85, 378 feature envincering, 154, 159, 212, 430 feedback control. 272, 276, 277, 279, 321 feedforward control, 277, 316 Fourier transform, 15, 47, 54, 55, 266, 310, 387, 405, Gabor transform, 69, 70 Galerkia projection 247 250 371 378 387 473 gappy POD, 383, 403, 424 Gaussian mixture models (GMM), 172-174 eradient descent, 117, 123, 125, 182, 198, 201, 206 gramian, 291, 292, 324, 325, 327, 329 Hankel matrix, 270, 331, 333, 337 incoherent measurements, 96, 97 information theory, 148, 149 inner product, 36, 42, 48, 96, 260, 266, 321, 325, 383, 354, 392, 404, 408 Johnson-Lindenstrauss (IL), 98 Kalman filter. 296, 297, 299, 305, 339, 341, 372 kernel methods, 183, 243, 267 Koopman electroperation, 268, 270 Koopman operator, 260, 261, 267 least square fit, 98, 118, 143 linear quadratic regulator (LOR), 292-294, 349, 359

low-rank, 7, 108, 109, 154, 155, 321, 323, 382, 385, 391, 396, 403, 404 machine learning, 19, 24, 77, 119, 142, 156, 158, 165, Markov parameters, 340 max pooling, 213 minture models, 172-174, 191 model predictive control (MPC), 346, 347, 349 Moore's law, 89 multi-resolution analysis (MRA), 75 multiscale, 58, 244, 276 naive Bayes, 190, 192 neural networks, 125, 191, 195, 196, 209, 273 noise, 92 observability, 47, 257, 287, 288, 292, 305, 324-326, observable function, 257, 258, 266 optimization eradient descent, 125 steepest descent, 125 outliers, 19, 98, 100, 107, 120, 138 overdetermined system, 4, 16, 101, 117, 132, 248 pagerank, 190, 193 Perron-Frobenius operator, 235, 266 polynomial basis functions, 183, 185 principal component, 21-23, 25, 108 principal component analysis (PCA), 3, 7, 21, 37, 97, proper orthogonal decomposition, 97, 245 proper orthogonal decomposition (POD), 3, 235, 321. 323, 375, 380, 423 radial basis functions, 221 radial basis functions (RBE) 184 random forest, 185, 189 randomized algorithms, 37, 39

randomized SVD, 37, 38 rank trancation, 155, 264 reduced order model (ROM), 242, 396, 423 reference tracking, 278, 280, 316, 358, 399 nerrossion, 15, 17, 98, 117-119, 121, 160, 163, 204, 242, 246, 248, 251, 336 restricted isometry property (RIP), 91, 97 ridee recression, 138 robust control, 300, 304, 308, 309, 317 pobust fit, 138 robust principal component analysis (rPCA), 107 sampline rate, 91, 94 senser, 244, 272, 302, 305, 306, 319, 336, 337, 353, singular value decomposition (SVD), 3, 37, 39, 97, 156, 235-237, 266, 322, 330, 381, 385, stratshot, 41, 236, 252, 292, 330-333, 347, 382, 386. stratuse identification of nonlinear dynamics (SINDy). sparse regression, 98, 247-249. 251. 256 sparse representation, 84, 88, 103, 242 sparsity, 92 spectrogram, 69-72 state space, 269, 321, 334 stochastic gradient descent (SGD), 201, 202, 209 support vector machine (SVM), 180, 191 system identification (System ID), 236, 243, 251, 270. 321, 322, 324, 336, 346, 354 total least-squares, 244 training data, 176-178, 180, 212, 214 uncertainty principle, 74.75 underdetermined system, 4, 16, 89, 92, 103, 117, 134, unitary matrix, 14, 15 unsupervised learning, 154, 160, 190, 191 windowed FFT (Gabor transform), 69